

## TMS34020

# User's Guide

1990

**Computer Video Products** 

## TMS34020 User's Guide

2564006-9721 revision \* August 1990



#### **IMPORTANT NOTICE**

Texas Instruments (TI) reserves the right to make changes to or to discontinue any semiconductor product or service identified in this publication without notice. TI advises its customers to obtain the latest version of the relevant information to verify, before placing orders, that the information being relied upon is current.

TI warrants performance of its semiconductor products to current specifications in accordance with TI's standard warranty. Testing and other quality control techniques are utilized to the extent TI deems necessary to support this warranty. Unless mandated by government requirements, specific testing of all parameters of each device is not necessarily performed.

TI assumes no liability for TI applications assistance, customer product design, software performance, or infringement of patents or services described herein. Nor does TI warrant or represent that license, either express or implied, is granted under any patent right, copyright, mask work right, or other intellectual property right of TI covering or relating to any combination, machine, or process in which such semiconductor products or services might be or are used.

#### TRADEMARKS

Apollo is a trademark of Apollo Computer Incorporated. DEC, Ultrix, VAX, and VMS are trademarks of Digital Equipment Corporation. IBM-PC is a registered trademark of International Business Machines Corporation. Macintosh is a trademark of Apple Computer Incorporated. MS-DOS is a trademark of Microsoft Corporation. SUN-3 is a trademark of Sun Microsystems Incorporated. UNIX is a registered trademark of AT&T.

XDS is a trademark of Texas Instruments Incorporated.

Copyright © 1990, Texas Instruments Incorporated

### Preface

## **Read This First**

This document describes the TMS34020 Graphics System Processor. It focuses on the TMS34020's role in applications that involve CRT-based bit-mapped graphics systems.

### If You Need Assistance...

If you want to	Do this
Receive more information about Texas Instruments graphics products	Call the CRC <sup>†</sup> hotline: (800) 232–3200
	Or write to: Market Communications Manager P.O. Box 1443, MS 736 Houston, Texas 77251–1443
Order Texas Instruments documentation	Call the CRC <sup>†</sup> hotline: (800) 232–3200
Ask questions about product operation or report suspected problems	Call the graphics hotline: (713) 274–2340
Report mistakes in this document or in any other TI documentation	Send your comments to: Technical Publications Manager Texas Instruments Incorporated
	P.O. Box 1443, MS640 Houston, Texas 77251–9879

† Texas Instruments Customer Response Center

### **Related Documentation from Texas Instruments**

The following TMS34010 and TMS34020 documents are available from Texas Instruments. To obtain a copy of any of these TI documents, please call the Texas Instruments Customer Response Center (CRC) at (800) 232–3200. When ordering, please identify the book by its title and its literature number.

**Pixel Perspectives** is a quarterly newsletter, published by the Graphics Products group of Texas Instruments Incorporated. This newsletter describes new products, discusses support for existing products, and identifies new documentation releases.

- The **TMS34020 Data Sheet** (lit. number SPVS004) contains electrical specifications, timing information, and mechanical data for the TMS34020.
- The **TMS340 Family Code-Generation Tools User's Guide** (lit. number SPVU004) describes the C compiler, assembler, linker, archiver, and auxiliary tools that are available for developing TMS34010 or TMS34020 code.
- The **TIGA-340 Interface User's Guide** (lit. number SPVU015) describes the Texas Instruments Graphics Architecture (TIGA), a software interface that standardizes communication between application software and TMS340-based hardware for IBM-compatible PCs.
- The **TMS34010 Software Development Board User's Guide** (lit. number SPVU002) describes a high-performance graphics card that aids in understanding TI graphics products. Read Pixel Perspectives for discussions of a TMS34020 version of this product and its documentation.
- The **TMS34010 User's Guide** (lit. number SPVU001) describes the TMS34010, which is the first-generation graphics system processor in the TMS340 family of graphics products.

### Notational Conventions

The following terms and conventions are used throughout this manual.

Term/Convention	Description
Rs, Rd	source register, destination register.
Rs.XY, Rd.XY	source or destination register in XY form.
Rs.X, Rd.X	X half of source or destination register.
Rs <b>.Y</b> , Rd <b>.Y</b>	Y half of source or destination register.
R	is a bit in an instruction opcode that identifies which reg- ister file the register operands are in. R=0 identifies file A; R=1 identifies file B.
PC'	is the address of the next instruction (current PC plus the length of the current instruction).
Rp	register pair.
CC	condition codes for a jump instruction.
IW, IL	16-bit immediate value (short), 32-bit immediate value (long).
SAddress, DAddress	source address, destination address.
SOffset, DOffset	source offset, destination offset.
LSB, MSB	least significant bit, most significant bit.
LSbyte, MSbyte	least significant byte, most significant byte.
LSW, MSW	least significant word, most significant word.

Term/Convention	Description			
n	identifies a number	r that may hav	e several valu	es.
An, Bn	identifies register r	in register file	A or file B.	
TOS	top of stack.			
F	optional field selec tions. F=0 selects t the move; F=1 sele 1 for the move.	t parameter fo he field size & ects the field size	r MOVE-field extension of fi ze & extensior	instruc- eld 0 for n of field
xxx[[REGISTER]]	identifies a bit with CBP[[CONFIG]]	in a register. F	or example,	
	refers to the CBP b	it in the CONF	IG register.	
special font	identifies program l symbol names. Fo	istings, coding r example,	examples, file	names, an
	0011 00000210	0001	.field	1, 2
	0012 00000212	0003	.field	3, 4
	0013 00000215	0006	.field .even	6, 3
italic text	PIXBLT B,XY exactly as shown (I serves two purpos	PIXBLT B, XY es. In text, ita	). <i>Ilics</i> emphasiz	e importa
	explanations. In in holders"that identif for a parameter. Fo CVXYL Rs, Rd	struction synt y the type of inf or example,	ax, <i>italics</i> ide formation you	ntify "place should ente
	CVXYL is an instru Rd—you must repl destination register	iction that has ace <i>Rs</i> and <i>R</i> s (CVXYL A0,	two paramet d with actual	ers, <i>Rs</i> ar source ar
[]	identify an optional struction with an op CMPI <i>IW, Rd [</i> , W	parameter. H otional parame 1	ere's an exam eter:	ple of an i
	CMPI has 3 parame optional. Note that t parameter, you mus	ters; the first tw he <b>, W</b> is bold– st type it as sh	vo are required —so if you use own.	d, the third the option
$\rightarrow$	means <i>becomes th</i> description, for exa	<i>e contents of</i> . ample,	In an instructio	on executio
	$Rs \rightarrow PC$ the contents of Rs contents of) the pro	become the c gram counter.	ontents of (or	replace th
value	means take the abs allel bars.	solute value of	the item betw	een the pa

۷

Term/Convention	Description
{choice1 choice2}	identifies a list; you can enter choice 1 or choice 2.
:	indicates concatenation. For example, Rd:Rd+1 forms a 64-bit register area of Rd and the next register in the same file.
<i>value</i> b , <i>value</i> B	identifies a binary integer. For example, 01b 1111B
<i>value</i> h , <i>value</i> H	identifies a hexadecimal integer. A hex number can't start with a letter—start it with a 0 instead. For example, 0FFFFh 123H

### Information About Cautions and Warnings



The information in a caution or a warning is provided for your protection. Please read each caution and warning carefully.

### Suggested References

The following books and articles provide further background information about graphics and system concepts associated with graphics:

- Artwick, Bruce A. *Applied Concepts in Microcomputer Graphics*. Englewood Cliffs, New Jersey: Prentice-Hall, 1984.
- Asal, Short, Preston, Simpson, Roskell, and Guttag. "The Texas Instruments 34010 Graphics System Processor." *IEEE Computer Graphics and Applications* vol.6, no.10, pp. 24–39.

- Bresenham, J.E. "Algorithm for Computer Control of a Digital Plotter." *IBM Systems Journal* 4, No.1 (1965): 25–30.
- Bresenham, J.E. "A Linear Algorithm for Incremental Display of Digital Arcs." Communications of the ACM 20 (Feb. 1977): 100-106.
- Cody, William J. Jr., and William Waite. *Software Manual for the Elementary Functions*. Englewood Cliffs, New Jersey: Prentice-Hall, 1980.
- Foley, James, and Andries van Dam. *Fundamentals of Interactive Computer Graphics*. Reading, Massachusets: Addison-Wesley, 1982.
- Gupta, Satish. "Architectures and Algorithms for Parallel Updates of Raster Scan Displays." Tech. Report CMU-CS-82-111, Computer Science Dept., Carnegie Mellon University, 1981.
- Ingalls, D.H. "The Smalltalk Graphics Kernel." Special Issue on Smalltalk, *Byte*, August 1981, pp. 168–194.
- Kernighan, B., and D. Ritchie. *The C Programming Language*. Englewood Cliffs, New Jersey: Prentice-Hall, 1978.
- Killebrew, C.R. Jr., "The TMS34010 Graphics System Processor." *Byte*, December 1986, pp. 193–204.
- Kochan, Stephen G. *Programming in C*. Hasbrouck Heights, New Jersey: Hayden Book Company, 1983.
- Newman, W.M., and R.F. Sproull. *Principles of Interactive Computer Graphics*. 2nd ed. New York: McGraw-Hill, 1979.
- Pike, Rob. "Graphics in Overlapping Bitmap Layers." ACM Transactions On Graphics 2 (April 1983): 135–160.
- Pinkham, R., M. Novak, and K. Guttag. "Video RAM Excels at Fast Graphics." Electronic Design, August 18, 1983, pp. 161–168.
- Pitteway, M.L.V. "Algorithm for Drawing Ellipses or Hyperbolae with a Digital Plotter." *Computer Journal* 10 (November 1967): 24–35.
- Porter, T. and T. Duff. "Composing Digital Images." *Computer Graphics*, July 1984, pp. 253–259.
- Sproull, R.F. and I.E. Sutherland. "A Clipping Divider." *Fall Joint Computer Conference*. Washington, DC: Thompson Books, 1968.
- Van Aken, Jerry R. "An Efficient Ellipse-Drawing Algorithm." *IEEE Computer Graphics & Applications* 4 (Sept. 1984): 24–35.
- Wientjes, Guttag, and Roskell. "First Graphics Processor Takes Complex Orders to Run Bit-Mapped Displays." *Electronic Design* Vol. 34, No.2 (January 23, 1986): 73—80.



# Contents

1	Overv Provid applid	view of the TMS34020
	1.1 1.2 1.3	Key Features of the TMS340201-2Typical Applications of the TMS340201-3Major Components of the TMS34020 Architecture1-41.3.1Internal Functions1-51.3.2Major Interfaces1-8
	1.4	System Development Tools1-101.4.1Code-Generation Tools1.4.2Supported Systems1.4.3Packages1.4.4TIGA-340 Graphics Interface1-13
	1.5 1.6	Processors for a Graphics System
2	Pinou Illustra ciated	Its and Signal Descriptions
	2.1 2.2 2.3	Pinouts2-2The TMS34020's Major Interfaces2-8Signal Descriptions2-92.3.1Local-Memory Interface Signals2-112.3.2DRAM and VRAM Control Signals2-122.3.3Multiprocessor Interface Signals2-132.3.4Host Interface Signals2-132.3.5Video Interface Signals2-152.3.6System Control Signals2-162.3.7Power Signals2-16
3	Memo Discu struct and lit	<b>bry Organization and Data Structures</b>
	3.1 3.2 3.3	Memory Map         3-2           Memory Addressing         3-3           Fields         3-5

.....

	3.4	Pixels		-10
		3.4.1 Pixe	els in Memory	-10
		3.4.2 Pixe	els on the Screen	3-11
		3.4.3 Disp	play Pitch	i-13
	3.5	XY Address	sing	J-14
	3.6	Converting a	an XY Address to a Linear Address 3	-15
	3.7	<b>Pixel Arrays</b>	s 3	-18
	3.8	<b>Big-Endian</b>	and Little-Endian Addressing 3	-20
		3.8.1 Sele	ecting Big-Endian or Little-Endian Mode	-20
		3.8.2 Hov	$ u$ the TMS34020 Accesses Memory in These Modes $\dots \dots \dots \dots$ 3	-21
		3.8.3 Ass	embling Code for Big-Endian or Little-Endian Addressing	-24
		3.8.4 Wiri	ing VRAMs to the LAD Bus 3	-25
		3.8.5 Big-	-Endian Effects on Instruction Timing 3	-25
	3.9	Stacks		-26
		3.9.1 Sys	stem Stack	-26
		3.9.2 Aux	(iliary Stacks	-29
4	TMS3	4020 Reaist	ers	4-1
-	Provid	les a detailed	d discussion the TMS34020's registers, including on-chip registers and I/O	• •
	registe	ers; also prov	vides an alphabetical reference of I/O registers combined with the B-file regis-	
	ters th	at graphics i	instructions use as implied operands.	
	4.1	The Status I	Register (ST)	4-2
	4.2	The Program	m Counter (PC)	4-4
	4.3	The Stack F	Pointer (SP)	4-5
	4.4	General-Pu	rpose Registers (Register Files A and B)	4-6
	4.5	I/O Register	rs	4-9
		4.5.1 CPI	U Control Registers	-12
		4.5.2 Hos	st Communications Registers	-12
		4.5.3 Loc	al-Memory and DRAM/VRAM Interface Registers	-12
		4.5.4 Inte	errupt Registers	-12
		4.5.5 Vide	eo Timing and Screen-Refresh Registers 4	-13
		4.5.6 Late	ency of Writes to I/O Registers 4	-13
	4.6	Alphabetica	al Summary of I/O Registers and B-File Registers 4	-14
	Inchar	atian Oacha	and Internal Devellations	
Ð	Instru		e and internal Parallelism	5-1
	Provid	les a detailed	a description of TMS34020 cache architecture and operation.	
	5.1	Cache Arch		5-2
	5.2	Cache Repl		5-4
	5.3	Cache Oper	ration	5-5
		5.3.1 Cac	che Hits	5-5
		5.3.2 Cac	che Misses	5-5
		5.3.3 Feto	ching Data into the Cache Following a Cache Miss	5-6
		5.3.4 Self	t-Moditying Code	5-8
		5.3.5 Flue	shing the Cache	5-8
	-	5.3.6 Disa	abling the Cache	5-8
	5.4	Performanc	e with Cache Enabled vs. Cache Disabled	5-9
	5.5	Internal Par	rallelism	j-10

6	Interr Descr	rupts, Traps, and Reset	i <b>-1</b>
	rupts,	and reset operation.	
	6.1	Related Signals	5-2
	6.2	Related Registers	j-2
	6.3	Enabling and Disabling Interrupts	j-6
	6.4	Interrupt Priorities and Vector Addresses	5-7
	6.5	Interrupt Processing	;-9
		6.5.1 Returning from an Interrupt Service Routine	10
		6.5.2 Interrupt Latency 6-	11
	6.6	Interrupting Instruction Execution	13
	6.7	External Interrupts 1 and 2 6-	15
	6.8	Internal Interrupts	16
		6.8.1 The Nonmaskable Interrupt (NMI) 6-	16
		6.8.2 The Host Interrupt (HI) 6-	16
		6.8.3 The Display Interrupt (DI) 6-	17
		6.8.4 Window-Violation Interrupt (WV) 6-	17
		6.8.5 The Single-Step Interrupt 6-	17
		6.8.6 Illegal-Opcode Interrupts 6-	18
	6.9	The Bus-Fault Interrupt 6-	19
		6.9.1 Activity During a Bus-Fault Interrupt	19
		6.9.2 Bus Fault System Considerations	20
	6.10	Interrupting a Host Processor	21
	6.11	Iraps	21
	0.12	Resei	22
		6.12.1 Activity During Reset	22
		6.12.2 Initial State Following Reset	23 24
		6.12.4 System Configuration Following Reset	24
		6 12 5 BESET and Multiprocessor Synchronization	20
		6 12 6 State of VCLK During Reset	27
	6 13	An Application for Interrupts: Debugging Code	28
	00	6.13.1 How a Debugger Works	28
		6.13.2 Using a Debugger	28
		6.13.3 Entering Single-Step Mode	28
		6.13.4 Clearing the Single-Step Bit	29
		6.13.5 A Few Things to Keep in Mind 6-	29
7	Comr	nunicating with a Host Processor7	<b>'-1</b>
	Descr	ribes methods for transferring information between the TMS34020 and a host processor.	
	7.1	Related Signals	'-2
	7.2	Related Registers	'-3
	7.3	A Basic Block Diagram for the Host Interface 7	'-6
	7.4	Basic Communication: How a Host Processor Reads from	
		and Writes to TMS34020 Local Memory 7	'-7
		7.4.1 How a Host Processor Requests a Read Cycle	-8
		7.4.2 How a Host Processor Requests a Write Cycle	-9
		7.4.3 Local-Memory Faults and Retries	-9

	7.5	Feature	s That Improve Performance of the Host Interface	7-10
		7.5.1	Prefetching Data from the TMS34020's Local Memory	7-10
		7.5.2	Autoincrementing (Implicit Addressing)	7-12
		7.5.3	The TMS34020's Default Memory Cycle	7-15
	7.6	Comple	ting Host Accesses	7-16
		7.6.1	Activating HRDY for Host Reads	7-16
		7.6.2	Activating HRDY for Host Writes	7-16
		7. <u>6</u> .3	Activating HRDY for Host Reads and Writes after Prefetches	7-17
	7.7	Timing I	Examples	7-18
	7.8	Halting	TMS34020 Execution and Downloading New Code	7-32
	7.9	Host-Int	terface Data Throughput (Bandwidth)	7-34
		7.9.1	Achieving Maximum Bandwidth	7-34
		7.9.2	Timing Considerations for Optimizing Host-Interface Bandwidth	7-35
	7.10	Delays	to Host Accesses	7-37
		7.10.1	Worst-Case Delay	7-37
	<b>_</b>	7.10.2	Halt Latency	7-39
	7.11	System	s with Multiple TMS34020s	7-40
	7.12	System	s with 16-Bit Memory Devices	7-42
	7.13	System	s with Big-Endian Addressing	/-44
8	Local	-Memory	y and DRAM/VRAM Interfaces	. 8-1
	Discu	sses the	local memory interface timing, addressing mechanisms, and special topic	s
	relate	d to DRA	Ms and VRAMs.	
	8.1	Related	Signals	. 8-2
	8.2	Related	Registers	. 8-4
	8.3	Prioritie	s of Memory Bus Requests	. 8-6
	8.4	General	I Form of a Local-Memory Cycle	. 8-8
		8.4.1	The Address/Status Subcycle	. 8-8
	0.5	8.4.2		. 8-9
	8.5	Local-M	lemory Cycle Status Codes	8-10
	8.6	Ending		8-12
		8.6.1	Extending a Local-Memory Cycle with Walt States	8-12
		8.6.2	Detrying a Successful Local-Memory Cycle	8-13
		0.0.3		8-13
	07	0.0.4 Dorform	bus Faulting a Local-Memory Cycle	8-14
	0.7		Colorting Dage Mode Operation	8-15
		0.7.1	Selecting Page-Wode Operation	0-10
	0 0	0./.2	How life TMS34020 Uses Page Mode	8-10
	0.0	LUCal-IV	Least Memory Dood Cycles	0-10
		0.0.1	Local-Memory Mead Cycle Timing (with Page Mode)	8-20
		0.0.2	Local-Memory Write-Cycle Timing (with Page Wode)	8-20
		0.0.3	Local-Memory Read/write or Read-Modify-Write Cycle Timing	8-22
	0.0	0.0.4	Host-Initiated Local-Memory Read and Write Cycles	8-24
	0.9		Dete Transfer Lising Dunamic Due Sizing	8-25
		0.9.1	Data Transier Using Dynamic Bus Sizing	8-26 0.00
		0.9.2	Page Node and Dynamic Bus Sizing	8-28
		o.y.3	Bus-Locked Operation and Dynamic Bus Sizing	8-29

	8.10	VRAM	Serial-Register Transfers	8-29
		8.10.1	Memory-to-Serial-Data-Register Cycle (VRAM Read Transfer)	8-30
		8.10.2	Memory-to-Split-Serial-Data-Register Cycle	
			(VRAM Split-Register Midline-Reload Transfer)	8-31
		8.10.3	Serial-Data-Register-to-Memory Cycle (VRAM Write Transfer and Pseudo-Write Transfer)	8-32
		8.10.4	Serial-Data-Register-to-Memory Cycle (VRAM Alternate-Write Transfer)	8-33
	8.11	VRAM	Write-Mask Local-Memory Cycles	8-34
		8.11.1	Load-Write-Mask Cycle	8-34
		8.11.2	Write Cycle (with Mask)	8-36
	8.12	VRAM	Block-Write Local-Memory Cycles	8-37
		8.12.1	VRAM Support of Block-Write Cycles	8-37
		8.12.2	TMS34020 Support of VRAM Block-Write Cycles	8-37
		8.12.3	Load-Color-Register Cycle	8-38
		8.12.4	Block-Write Cycle (Without Mask)	8-39
		8.12.5	Block-Write Cycle (with Mask)	8-40
		8.12.6	Data Mapping During Block-Write Cycles	8-41
	8.13	DRAM-	Refresh Local-Memory Cycles	8-44
	8.14	Local-M	Memory Cycles with Wait States	8-46
		8.14.1	Adding Wait States in Read and Write Cycles	8-46
		8.14.2	Adding Wait States in VRAM Serial-Register Transfers	8-48
	8.15	The Ho	st-Default Local-Memory Cycle	8-49
	8.16	Addres	sing Mechanisms	8-50
		8.16.1	Nonmultiplexed Addressing	8-50
		8.16.2	Multiplexed Addressing	8-51
		8.16.3	Display Memory Requirements for Multiplexed Addressing	8-54
		8.16.4	Example Connections for Multiplexed Addressing	8-54
		8.16.5	Memory Organization and Bank Selecting	8-55
		8.16.6	Display Memory Hardware Requirements	8-56
	8.17	Double	-Buffered Display Example (2×1280×1024)	8-57
		8.17.1	Display Memory Implementation Using Midline Reload	8-58
		8.17.2	Display Memory Implementation Without Midline Reload	8-59
9	Video	Timina	and Screen Refresh	. 9-1
	Descr	ibes the	TMS34020's video timing mechanisms, including separate and composite syn	
	and bl	anking,	interlaced and noninterlaced video, and screen refreshes.	-
	9.1	Related	Signals	. 9-2
	9.2	Related	Registers	. 9-4
	9.3	Relatio	nship Between Horizontal and Vertical Timing Signals	. 9-9
	9.4	Horizor	tal Video Timing (Internal)	9-11
	9.5	Vertical	Video Timing (Internal)	9-13
	9.6	Compo	site Video Timing	9-15
		9.6.1	Theory Behind Serration and Equalization Pulses	9-15
		9.6.2	Serration Pulses on CSYNC	9-16
		9.6.3	Equalization Pulses on CSYNC	9-17

	9.7	Noninte	erlaced Video Timing	9-18
		9.7.1	Activity in Noninterlaced Mode	9-18
		9.7.2	Programming the Vertical Timing Registers for Noninterlaced Video	9-20
	9.8	Interlac	ed Video Timing	9-21
		9.8.1	Activity in Interlaced Mode	9-21
		9.8.2	Programming the Vertical Timing Registers for Interlaced Video	9-24
		9.8.3	American and European Video Standards	9-27
	9.9	Externa	al Synchronization Modes	9-29
		9.9.1	Odd and Even Field Alignment in Interlaced Mode	9-31
		9.9.2	Synchronizing External Syncs to VCLK	9-32
		9.9.3	Loading the Video Counters	9-32
		9.9.4	Synchronization Conversion	9-34
		9.9.5	Programming Flexibility and Limitations	9-34
		9.9.6	External Synchronization Pulse Widths	9-35
	9.10	Screen	Sizes and Dot Rate	9-36
	9.11	Display	Interrupts and Applications	9-37
	9.12	Video T	Timing Programming Examples	9-38
		9.12.1	Noninterlaced 1024 × 768 Display	9-38
		9.12.2	Composite Interlaced NTSC Display Example	9-40
	9.13	Video F	RAM Control	9-42
		9.13.1	Screen Refreshes During Horizontal Blanking	9-42
		9.13.2	Screen Refreshes During the Active Display Time (Midline Reload)	9-43
		9.13.3	Why Use Midline Reload?	9-46
		9.13.4	VRAM Bulk Initialization	9-47
		9.13.5	Video Capture	9-48
		9.13.6	Disabling Screen Refreshes	9-49
	9.14	Schedu	Iling Screen-Refresh Cycles	9-50
	9.15	Genera	ting Screen-Refresh Addresses	9-51
		9.15.1	Horizontal-Blanking Screen-Refresh Addresses	9-52
		9.15.2	Screen-Refresh Addressing Sequence for Noninterlaced Displays	9-53
		9.15.3	Screen-Refresh Addressing Sequence for Interlaced Displays	9-53
		9.15.4	Midline-Reload Screen-Refresh Addresses	9-55
		9.15.5	Display Magnification and Y-Zoom	9-56
		9.15.6	Panning the Display	9-57
10	Com	nunicati	ing with a Coprocessor	10-1
••	Descr	ihes a c	meneral protocol for interfacing with a conrocessor and describes use of the	- • •
	TMS3	4020's d	reneral-purpose coprocessor instructions.	
	10.1	Relator	l Signale	10-2
	10.1	Overvie	a signals	10-2
	10.2	Format	of Commands Passed to a Conrocessor	10-5
	10.0	10 2 1	Contropessor ID	10-5
		10.3.2	Coprocessor Command	10-6
		10.3.2	Conrocessor Parameter Size (size)	10-6
		10.0.0		10-0

		10.3.4 Coprocessor Parameter Index (I) 1	10-7
		10.3.5 16-Bit Word Select (S) 1	10-7
		10.3.6 Coprocessor Status Code (BCST) 1	10-7
	10.4	Local-Memory Coprocessor Cycles 1	10-8
		10.4.1 Passing Commands to a Coprocessor 1	10-8
		10.4.2 Transferring Data to or from a Coprocessor 1	10-8
		10.4.3 Data Transfer Sequences to or from a Coprocessor 1	10-9
		10.4.4 Ending a Local-Memory Coprocessor Cycle 1	10-9
		10.4.5 Coprocessor Command Cycle 10	D-10
		10.4.6 Transferring Values from TMS34020 Registers to a Coprocessor 10	0-11
		10.4.7 Transferring Values from a Coprocessor to TMS34020 Registers 10	)-12
		10.4.8 Transferring Values from Local Memory to a Coprocessor 10	)-14
		10.4.9 Transferring Values from a Coprocessor to Local Memory 10	D-15
	10.5	Coprocessor Aborts and Status Checks 10	)-17
	10.6	System Configuration 10	D-18
11	Multi	processing and System Architecture	11_1
	Deco	ribes the TMS34020 multiprocessor interface and gives examples of using multiple pro-	
	Cesso	nees the TMSS4020 multiplocessor interface and gives examples of using multiple pro-	
	11 1		11-2
	11.1	Avania Signals	11-2
	11.2	Basic Multiprocessor System Configuration	11-2
	11.5	11.3.1 Connecting Multiple Processors Together	11-3
		11.3.2 Synchronizing Multiple TMS24020s at Paset	11-3
	11 /	Protocole for Communicating in a Multiprocessor System	11-5
	11.4	11 / 1 How a Processor Bequests Control of the Local-Memory Bus	11-5
		11.4.2 How a Processor Releases Control of the Local-Memory Bus	11-5
		11.4.3 Passing Control of the Local-Memory Bus	11-6
		11.4.4 Functional Timing Examples	11-7
	11.5	Arbitration Logic Requirements	1-13
	11.0	11.5.1 Passing Control of the Local-Memory Bus	1-13
		11.5.2 Wait States Retries and High-Priority Bus Requests	1-15
	11.6	Multiprocessor Arbitration Examples	1-15
		11.6.1 Arbitration Scheme for Two TMS34020s	1-15
		11.6.2 Arbitration Scheme for One TMS34020 and a Hold Device	1-17
	11 7	Initializing Multiple TMS34020s	1-19
	11.8	Configuration with a Host Processor	1-20
12	Grapl	hics Instructions and Operations 1	12-1
	Offers	s a detailed look at the TMS34020's graphics instructions and their special capabilities.	
	12.1	An Overview of Graphics Instructions 1	12-2
	12.2	An Overview of Graphics Operations 1	12-3
	12.3	Single-Pixel Instructions 1	12-6
	12.4	Line Instructions	12-7

	12.5	Pixel-Array Instructions 1	12-8
		12.5.1 PIXBLTs with XY and Linear Addressing 1	12-9
		12.5.2 Binary (Color Expanding) PIXBLTs	2-12
		12.5.3 Masked PIXBLT	2-14
		12.5.4 VRAM Block-Mode PIXBLT (VBLT)	2-14
		12.5.5 FILLs	2-15
		12.5.6 Horizontal Pattern Fill (PFILL)	2-16
		12.5.7 VRAM Block-Mode Fill (VFILL)	2-16
	12.6	Auxiliary Graphics Instructions	2-17
	12.7	Window Checking	2-19
		12.7.1 Defining a Window	2-19
		12.7.2 Window-Violation Interrupt	2-20
		12.7.3 Window Checking for Single-Pixel Instructions	2-21
		12.7.4 Window Checking for Pixel-Array Instructions	2-21
		12.7.5 Window Checking for the LINE Instruction	2-23
	12.8	Pixel Processing	2-27
		12.8.1 Boolean Processing Examples	2-28
		12.8.2 Multiple-Bit Pixel Operations	2-30
	12.9	Transparency	2-36
	12.10	Plane Masking	2-39
	12.11	Setting Up the Implied Operands for Graphics Instructions	2-43
	12.12	Converting an XY Address to a Linear Address	2-47
		12.12.1 Manual XX to Linear Conversion	2_17
			2-4/
		12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping	2-49
		12.12.1 Manual X1-l0-Linear Conversion	2-49
13	TMS3	12.12.1 Manual X1-0-Linear Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       1	2-49 1 <b>3-1</b>
13	TMS3 Explai	12.12.1 Manual X1-0-Linear Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       12         ins TMS34020 addressing modes and provides an alphabetical reference of the	2-49 1 <b>3-1</b>
13	<b>TMS3</b> Explai TMS3	12.12.1 Manual X1-0-Linear Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12 <b>4020 Assembly Language Instruction Set</b> 1         ins TMS34020 addressing modes and provides an alphabetical reference of the 4020 instruction set.	2-49 1 <b>3-1</b>
13	<b>TMS3</b> Explai TMS3 13.1	12.12.1 Manual X1-0-Linear Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       1         ins TMS34020 addressing modes and provides an alphabetical reference of the 4020 instruction set.       1         Addressing Modes and Operand Formats       1	2-49 1 <b>3-1</b>
13	<b>TMS3</b> Explai TMS3 13.1	12.12.1 Manual X1-0-Linear Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       1         ins TMS34020 addressing modes and provides an alphabetical reference of the 4020 instruction set.       1         Addressing Modes and Operand Formats       1         13.1.1 Immediate Values and Constants       1	13-2 13-2
13	<b>TMS3</b> Explai TMS3 13.1	12.12.1 Manual X1-0-Linear Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       1         ins TMS34020 addressing modes and provides an alphabetical reference of the 4020 instruction set.       1         Addressing Modes and Operand Formats       1         13.1.1 Immediate Values and Constants       1         13.1.2 Absolute Addresses       1	2-49 13-1 13-2 13-2 13-3
13	<b>TMS3</b> Explai TMS3 13.1	12.12.1 Manual X1-0-Linear Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       1         ins TMS34020 addressing modes and provides an alphabetical reference of the 4020 instruction set.       1         Addressing Modes and Operand Formats       1         13.1.1 Immediate Values and Constants       1         13.1.2 Absolute Addresses       1         13.1.3 Register-Direct Operands       1	13-2 13-2 13-3 13-4
13	TMS3 Expla TMS3 13.1	12.12.1 Manual X1-0-Linear Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       1         ins TMS34020 addressing modes and provides an alphabetical reference of the 4020 instruction set.       1         Addressing Modes and Operand Formats       1         13.1.1 Immediate Values and Constants       1         13.1.2 Absolute Addresses       1         13.1.3 Register-Direct Operands       1         13.1.4 Register-Indirect Operands       1	2-49 13-1 13-2 13-2 13-3 13-4 13-5
13	<b>TMS3</b> Expla TMS3 13.1	12.12.1 Manual X1-0-Linear Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       1         ins TMS34020 addressing modes and provides an alphabetical reference of the 4020 instruction set.       1         Addressing Modes and Operand Formats       1         13.1.1 Immediate Values and Constants       1         13.1.2 Absolute Addresses       1         13.1.3 Register-Direct Operands       1         13.1.4 Register-Indirect Operands       1         13.1.5 Register-Indirect with Offset       1	<b>13-2</b> 13-2 13-2 13-3 13-4 13-5 13-6
13	<b>TMS3</b> Explai TMS3 13.1	12.12.1 Manual X1-0-Linear Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       1         ins TMS34020 addressing modes and provides an alphabetical reference of the       1         4020 instruction set.       1         Addressing Modes and Operand Formats       1         13.1.1 Immediate Values and Constants       1         13.1.2 Absolute Addresses       1         13.1.3 Register-Direct Operands       1         13.1.4 Register-Indirect Operands       1         13.1.5 Register-Indirect with Offset       1         13.1.6 Register-Indirect with Postincrement       1	13-2 13-2 13-2 13-3 13-4 13-5 13-6 13-7
13	<b>TMS3</b> Explai TMS3 13.1	12.12.1 Manual X Pro-Linear Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       1         ins TMS34020 addressing modes and provides an alphabetical reference of the       1         4020 instruction set.       1         Addressing Modes and Operand Formats       1         13.1.1 Immediate Values and Constants       1         13.1.2 Absolute Addresses       1         13.1.3 Register-Direct Operands       1         13.1.4 Register-Indirect Operands       1         13.1.5 Register-Indirect with Offset       1         13.1.6 Register-Indirect with Postincrement       1         13.1.7 Register-Indirect with Predecrement       1	13-2 13-2 13-2 13-3 13-4 13-5 13-6 13-7 13-8
13	<b>TMS3</b> <i>Expla</i> . <i>TMS3</i> 13.1	12.12.1 Manual X Pro-Linear Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       1         ins TMS34020 addressing modes and provides an alphabetical reference of the 4020 instruction set.       1         Addressing Modes and Operand Formats       1         13.1.1 Immediate Values and Constants       1         13.1.2 Absolute Addresses       1         13.1.3 Register-Direct Operands       1         13.1.4 Register-Indirect Operands       1         13.1.5 Register-Indirect with Offset       1         13.1.6 Register-Indirect with Postincrement       1         13.1.7 Register-Indirect with Predecrement       1         13.1.8 Register-Indirect in XY Mode       1	2-49 13-1 13-2 13-2 13-3 13-4 13-5 13-6 13-7 13-8 13-9
13	<b>TMS3</b> <i>Explat</i> <i>TMS3</i> 13.1	12.12.1 Manual X1-0-Linear Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       1         ins TMS34020 addressing modes and provides an alphabetical reference of the 4020 instruction set.       1         Addressing Modes and Operand Formats       1         13.1.1 Immediate Values and Constants       1         13.1.2 Absolute Addresses       1         13.1.3 Register-Direct Operands       1         13.1.4 Register-Indirect Operands       1         13.1.5 Register-Indirect with Offset       1         13.1.7 Register-Indirect with Predecrement       1         13.1.8 Register-Indirect in XY Mode       1         13.1.8 Register-Indirect in XY Mode       1	2-49 <b>13-1</b> 13-2 13-2 13-3 13-4 13-5 13-6 13-7 13-8 13-9 13-9 13-9
13	<b>TMS3</b> <i>Explat</i> <i>TMS3</i> 13.1	12.12.1 Manual X1-to-Linear Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       1         ins TMS34020 addressing modes and provides an alphabetical reference of the 4020 instruction set.       1         Addressing Modes and Operand Formats       1         13.1.1 Immediate Values and Constants       1         13.1.2 Absolute Addresses       1         13.1.3 Register-Direct Operands       1         13.1.4 Register-Indirect Operands       1         13.1.5 Register-Indirect with Offset       1         13.1.7 Register-Indirect with Predecrement       1         13.1.8 Register-Indirect in XY Mode       1         13.1.8 Register-Indirect in XY Mode       1         13.1.9 Move Instructions Summary       1	<b>13-2</b> <b>13-1</b> <b>13-2</b> <b>13-3</b> <b>13-4</b> <b>13-5</b> <b>13-6</b> <b>13-7</b> <b>13-8</b> <b>13-9</b> <b>13-9</b> <b>13-9</b> <b>3-19</b>
13	<b>TMS3</b> <i>Explai</i> <i>TMS3</i> 13.1	12.12.1 Manual X1-0-Lineal Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       1         ins TMS34020 addressing modes and provides an alphabetical reference of the 4020 instruction set.       1         Addressing Modes and Operand Formats       1         13.1.1 Immediate Values and Constants       1         13.1.2 Absolute Addresses       1         13.1.3 Register-Direct Operands       1         13.1.4 Register-Indirect Operands       1         13.1.5 Register-Indirect with Offset       1         13.1.6 Register-Indirect with Postincrement       1         13.1.8 Register-Indirect in XY Mode       1         13.1.8 Register-Indirect in XY Mode       1         13.3.1 Register-to-Register Moves       1	<b>13-2</b> <b>13-1</b> <b>13-2</b> <b>13-2</b> <b>13-3</b> <b>13-4</b> <b>13-5</b> <b>13-6</b> <b>13-7</b> <b>13-8</b> <b>13-9</b> <b>13-9</b> <b>3-19</b> <b>3-19</b> <b>3-19</b>
13	<b>TMS3</b> <i>Explai</i> <i>TMS3</i> 13.1	12.12.1 Manual X1-to-Enlear Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12 <b>4020 Assembly Language Instruction Set</b> 1 <i>ins TMS34020 addressing modes and provides an alphabetical reference of the</i> 1 <i>4020 instruction set.</i> 1         Addressing Modes and Operand Formats       1         13.1.1 Immediate Values and Constants       1         13.1.2 Absolute Addresses       1         13.1.3 Register-Direct Operands       1         13.1.4 Register-Indirect Operands       1         13.1.5 Register-Indirect with Offset       1         13.1.6 Register-Indirect with Postincrement       1         13.1.7 Register-Indirect with Predecrement       1         13.1.8 Register-Indirect in XY Mode       1         Summary Table       1         13.3.1 Register-to-Register Moves       1         13.3.2 Value-to-Register Moves       1	<b>13-2</b> <b>13-1</b> <b>13-2</b> <b>13-3</b> <b>13-4</b> <b>13-5</b> <b>13-6</b> <b>13-7</b> <b>13-8</b> <b>13-9</b> <b>13-9</b> <b>13-9</b> <b>13-9</b> <b>13-9</b> <b>3-19</b> <b>3-19</b> <b>3-19</b> <b>3-19</b>
13	<b>TMS3</b> <i>Expla</i> <i>TMS3</i> 13.1	12.12.1 Manual X1-0-Linear Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       1         ins TMS34020 addressing modes and provides an alphabetical reference of the       1         4020 instruction set.       1         Addressing Modes and Operand Formats       1         13.1.1 Immediate Values and Constants       1         13.1.2 Absolute Addresses       1         13.1.3 Register-Direct Operands       1         13.1.4 Register-Indirect Operands       1         13.1.5 Register-Indirect with Offset       1         13.1.6 Register-Indirect with Predecrement       1         13.1.7 Register-Indirect in XY Mode       1         13.1.8 Register-Indirect in XY Mode       1         13.3.1 Register-to-Register Moves       1         13.3.2 Value-to-Register Moves       1         13.3.3 XY Moves       1	<b>13-2</b> <b>13-1</b> <b>13-2</b> <b>13-3</b> <b>13-3</b> <b>13-4</b> <b>13-5</b> <b>13-6</b> <b>13-7</b> <b>13-8</b> <b>13-9</b> <b>13-9</b> <b>13-9</b> <b>13-9</b> <b>13-9</b> <b>13-9</b> <b>13-9</b> <b>13-19</b> <b>3-19</b> <b>3-19</b> <b>3-19</b> <b>3-19</b> <b>3-19</b>
13	<b>TMS3</b> <i>Explat</i> <i>TMS3</i> 13.1	12.12.1 Manual X1-to-Lineal Conversion       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12 <b>4020 Assembly Language Instruction Set</b> 1 <i>ins TMS34020 addressing modes and provides an alphabetical reference of the 4020 instruction set.</i> 1         Addressing Modes and Operand Formats       1         13.1.1 Immediate Values and Constants       1         13.1.2 Absolute Addresses       1         13.1.3 Register-Direct Operands       1         13.1.4 Register-Indirect Operands       1         13.1.5 Register-Indirect With Offset       1         13.1.6 Register-Indirect with Predecrement       1         13.1.7 Register-Indirect in XY Mode       1         13.1.8 Register-Indirect in XY Mode       1         13.3.1 Register-to-Register Moves       1         13.3.2 Value-to-Register Moves       1         13.3.3 XY Moves       1         13.3.4 Multiple-Register Moves       1	2-49 <b>13-1</b> 13-2 13-2 13-3 13-4 13-5 13-6 13-7 13-8 13-9 13-9 3-19 3-19 3-19 3-19 3-20
13	<b>TMS3</b> <i>Explai</i> <i>TMS3</i> 13.1	12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       1         ins TMS34020 addressing modes and provides an alphabetical reference of the       1         4020 instruction set.       1         Addressing Modes and Operand Formats       1         13.1.1 Immediate Values and Constants       1         13.1.2 Absolute Addresses       1         13.1.3 Register-Direct Operands       1         13.1.4 Register-Indirect Operands       1         13.1.5 Register-Indirect with Offset       1         13.1.6 Register-Indirect with Postincrement       1         13.1.7 Register-Indirect in XY Mode       1         13.1.8 Register-Indirect in XY Mode       1         13.3.1 Register-to-Register Moves       1         13.3.2 Value-to-Register Moves       1         13.3.3 XY Moves       1         13.3.4 Multiple-Register Moves       1         13.3.5 Byte Moves       1	<b>13-2</b> <b>13-1</b> <b>13-2</b> <b>13-3</b> <b>13-4</b> <b>13-5</b> <b>13-6</b> <b>13-7</b> <b>13-8</b> <b>13-9</b> <b>3-19</b> <b>3-19</b> <b>3-19</b> <b>3-19</b> <b>3-19</b> <b>3-19</b> <b>3-20</b> <b>3-20</b>
13	<b>TMS3</b> <i>Explai</i> <i>TMS3</i> 13.1	12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping       12         4020 Assembly Language Instruction Set       1         ins TMS34020 addressing modes and provides an alphabetical reference of the       1         4020 instruction set.       1         Addressing Modes and Operand Formats       1         13.1.1 Immediate Values and Constants       1         13.1.2 Absolute Addresses       1         13.1.3 Register-Direct Operands       1         13.1.4 Register-Indirect Operands       1         13.1.5 Register-Indirect with Offset       1         13.1.6 Register-Indirect with Predecrement       1         13.1.7 Register-Indirect with Predecrement       1         13.1.8 Register-Indirect in XY Mode       1         13.3.1 Register-to-Register Moves       1         13.3.2 Value-to-Register Moves       1         13.3.3 XY Moves       1         13.3.4 Multiple-Register Moves       1         13.3.5 Byte Moves       1         13.3.6 Field Moves       1	<b>13-2</b> <b>13-1</b> <b>13-2</b> <b>13-3</b> <b>13-3</b> <b>13-3</b> <b>13-4</b> <b>13-5</b> <b>13-6</b> <b>13-7</b> <b>13-6</b> <b>13-7</b> <b>13-8</b> <b>13-9</b> <b>3-19</b> <b>3-19</b> <b>3-19</b> <b>3-19</b> <b>3-19</b> <b>3-19</b> <b>3-19</b> <b>3-19</b> <b>3-19</b> <b>3-20</b> <b>3-20</b> <b>3-20</b>

Table of Contents

	13.4       Arithmetic, Logical, and Compare Instructions       13         13.5       Program-Control and Context-Switching Instructions       13         13.5.1       Subroutine Calls and Returns       13         13.5.2       Interrupt Handling       13         13.5.3       Setting, Saving, and Restoring Status Information       13         13.5.4       Jump Instructions       13         13.6       Shift Instructions       13         13.7       XY Instructions       13         13.8       Instructions New to the TMS34020       13         13.9       Alphabetical Instruction Reference       13			
14	TMS3 Provic gener sembl 14.1 14.2	4082 Pseudo-ops14-1des a general description of the TMS34082 and the TMS34020's implementation of its al-purpose coprocessor instructions in a manner that directly supports TMS34082 as- ly language instructions.Overview and Key Features of the TMS3408214-2 Pseudo-op FormatPseudo-op Format14-3		
	14.3	Register Operands		
15	Instru Sumn	ction Timing		
	15.1 15.2	Timing for All Instructions Except MOVEs and MOVBs         15-2           Timing for MOVE and MOVB Instructions         15-10		
Α	<b>Test a</b> A.1 A.2 A.3 A.4 A.5 A.6	And Emulation ConsiderationsA-1Overview of an Emulation SystemA-2Emulation Connector (12-Pin Header)A-3Signal BufferingA-4Buffer DelaysA-5Design ConsiderationsA-7Mechanical DimensionsA-9		
в	Gloss	ary B-1		

# Figures

1–1	TMS34020 Block Diagram	1-5
1–2	TMS34020 Software Development Flow	1-10
1–3	Graphics Processing Shared Between TMS340 and Host Processors	1-13
1–4	Graphics Products Roadmap	1-14
1–5	TMS34020 1K×1K×8 PC Display System	1-15
2–1	TMS34020 Pinout, 145-Pin PGA Package (Bottom View)	2-2
2–2	TMS34020 Pinout, 132-Pin QFP Package	2-5
2–3	The TMS34020's Major Interfaces	2-8
3–1	TMS34020 Memory Map	3-2
3–2	Logical Memory Address Space	3-3
3–3	Physical Memory Addressing	3-4
3–4	Status Bits That Control Field 0 and Field 1	3-5
3–5	Field Storage in External Memory	3-6
3–6	Field Alignment in Memory	3-7
3–7	Field Insertion	3-9
3–8	Pixel Storage in External Memory	3-10
3–9	Mapping of Pixels to a Monitor Screen	3-11
3–10	Configurable Screen Origin	3-12
3–11	Display Memory Dimensions	3-12
3–12	Display Memory Coordinates	3-13
3–13	Pixel Addressing in Terms of XY Coordinates	3-14
3–14	Conversion from XY Coordinates to Memory Address	3-17
3–15	Pixel Array	3-18
3–16	How BEN CONFIG Determines the Endian Mode	3-20
3–17	How CBP CONFIG Write-Protects CONFIG's LSbyte	3-21
3–18	How Data Is Represented in Little-Endian Mode	3-22
3–19	Addressing a Field in a Long-Word (Little-Endian)	3-22
3–20	Moving a Field into a General-Purpose Register (Little-Endian)	3-22
3–21	How Data Is Represented in Big-Endian Mode	3-23
3–22	Addressing a Field in a Long-Word (Big-Endian)	3-23
3–23	Moving a Field into a General-Purpose Register (Big-Endian)	3-23
3–24	Sample Listing File (Assembler Output) for Little-Endian and Big-Endian Code	3-24
3–25	Loading Object Code into Memory	3-25

326	Connecting VRAMs to the LAD Bus	3-25
3–27	System Stack	3-26
3–28	Stack Operations	3-28
3–29	An Auxiliary Stack That Grows Toward Lower Addresses	3-30
3–30	An Auxiliary Stack That Grows Toward Higher Addresses	3-31
4–1	Status Register	4-2
4–2	Program Counter	4-4
4–3	The Stack-Pointer Register	4-5
4–4	The Register Files	4-6
4–5	I/O Register Memory Map	4-9
4–6	How DPYMSK Maps to the Logical Screen-Refresh Address	4-44
4–7	The Functions of the Different Fields of DPYMSK	4-45
4–8	Replicating the Mask Value for an 8-Bit Pixel	4-76
5–1	TMS34020 Instruction Cache	5-2
5–2	Segment Start Address	5-3
5–3	Internal Data Paths	5-10
5–4	Parallel Operation of Cache, Execution Unit, and Memory Interface	5-11
6–1	Vector Address Map	6-8
6–2	Actions Performed When the TMS34020 Takes an Interrupt	6-9
6–3	Actions Performed When the TMS34020 Executes a RETI or RETM Instruction	6-10
7–1	Block Diagram with a Host System, a TMS34020, and External Transceivers	7-6
7–2	How a Host Processor Uses the Host Byte-Select Signals to Access Data in TMS34020 Memory	7-8
7–3	How the Values of HINC [HSTCTLH] and HPFW [HSTCTLH] Affect Prefetching	7-10
7–4	How the Value of HINC [HSTCTLH] Affects Address Comparison	7-10
7–5	How the Value of HLB [HSTCTLH] Affects Prefetching	7-11
7–6	Legal Host Byte-Select Combinations for Autoincrementing	7-13
7–7	How the Values of HINC [HSTCTLH] and HPFW [HSTCTLH] Affect Autoincrementing	7-13
7–8	Single Host Read Cycle: HCS Used as Strobe	7-19
7–9	Single Host Read from I/O Registers: HREAD Used as Strobe	7-20
7–10	Single Host Read with One Wait State: HCS Used as Strobe	7-21
7–11	Host Read Back-to-Back with Prefetch of Next Word: HCS Used as Strobe	7-22
7–12	Back-to-Back Host Read Cycles with Implicit Addressing: HREAD as Strobe	7-23
7–13	Successive Reads to Same 32-Bit Location: HCS and HREAD Strobed Together	7-24
7–14	Single Host Write Cycle: HCS Used as Strobe	7-25
7–15	Single Host Write Cycle to I/O Registers: HWRITE Used as Strobe	7-26
7–16	Single Host Write Cycle with One Wait State: HCS Used as Strobe	7-27
7–17	Back-to-Back Host Write Cycles: HCS Used as Strobe	7-28
7–18	Back-to-Back Host Write Cycles with Implicit Addressing; HWRITE as Strobe	7-29
7–19	Host Write Cycle Back-to-Back with Prefetch of Next Word: HCS Used as Strobe	7-30
7–20	Host Write Cycle Back-to-Back with Prefetch of Next Word and Implicit Addressing;	7-31
7-21	Host Bequest Synchronization	7-35
· _ · 7_22	Host-to-TMS34020 Transceiver Wiring with 16-Rit Memory	7-43
	neer to manual manual thing with to bit wondry the terminer	

~

7-23	Big-Endian and Little-Endian Byte Addressing Modes 7-44
8-1	The Two Parts of a Local-Memory Cycle 8-8
8-2	Multiple Local-Memory Cycles Using Page Mode 8-15
8-3	General Timing of the Local-Memory Read and Write Cycles
8-4	Local-Memory Read-Cycle Timing (with Page Mode)
8–5	Local-Memory Write-Cycle Timing (with Page Mode)
8–6	Local-Memory Read/Write or Read-Modify-Write-Cycle Timing
8–7	Dynamic Bus <u>Sizing</u> for a Read Cycle (Connection to LAD0—LAD15, Indicated by SIZE16 Low During 2 <sup>nd</sup> Data Cycle)
8–8	Dynamic Bus <u>Sizing</u> for a Write Cycle (Connection to LAD16—LAD31, Indicated by SIZE16 High During 2 <sup>nd</sup> Data Cycle)
8–9	Memory-to-Serial-Data-Register Cycle (VRAM Read Transfer)
8–10	Memory-to-Split-Serial-Data-Register Cycle (VRAM Split-Register Midline-Reload Transfer)
8–11	VRAM Write Transfer and Pseudo-Write Transfer
8–12	VRAM Alternate-Write Transfer
8–13	Load-Write-Mask Cycle
8–14	Write Cycle (with Mask)
8–15	Load-Color-Register Cycle
8–16	Block-Write Cycle (Without Mask)
8–17	Block-Write Cycle (with Mask) 8-40
8–18	Refresh Cycle Timing
8–19	Local-Memory Read Cycle with 1 Wait State
8–20	Memory-to-Serial-Data-Register Cycle with Wait State (VRAM Read Transfer) 8-48
8–21	The Host-Default Cycle
8–22	Logical Address Output on LAD
8–23	VRAM Address Decode for Example System
8–24	DRAM Address Decode for Example System
8–25	Example Display Memory Dimensions (with Midline Reload)
8–26	Example Display Memory Dimensions (Without Midline Reload)
9–1	Horizontal and Vertical Timing Relationship
9–2	The Porches
9–3	Horizontal Timing
9-4	Horizontal Timing Logic—Equivalent Circuit
9–5	Example of Horizontal Signal Generation
9–6	Vertical Timing for Noninterlaced Display
9–7	Vertical Timing Logic—Equivalent Circuit
9–8	Regions of Vertical Blanking Where Equalization and           Serration Pulses Occur on CSYNC
99	Composite Sync During Serration Region (Interlaced)
9–10	Composite Sync During Equalization Regions (Interlaced)
9–11	Electron Beam Pattern for Noninterlaced Video
9–12	Noninterlaced Video Timing Waveform Example
9–13	Programming the Video Timing Registers for Noninterlaced Video

Electron Beam Pattern for a Typical Interlaced Display	9-22
Interlaced Video Timing Waveform Example	9-23
The Two Regions of Vertical Blanking Used for Programming Calculations	9-25
Programming the Video Timing Registers for Interlaced Video	9-26
Vertical Blanking for NTSC and PAL Standards	9-27
Synchronization Delay Compensation	9-33
Local-Memory Memory-to-Register Transfer Cycle	9-43
Local-Memory Split-Serial-Register VRAM Memory-to-Register Cycle	9-44
Local-Memory Split-Serial-Register VRAM Horizontal-Blanking Memory-to-Register Cycles	9-45
Local-Memory Register-to-Memory Transfer Cycle	9-48
Screen-Refresh Address Fields	9-51
Screen-Refresh Address Generation Flow	9-54
Mapping Relationship Between DPYMSK, DPYST, and DPYNX	9-56
Coprocessor Instruction Format	10-5
Coprocessor Command Cycle	10-10
Transferring a TMS34020 Register to a Coprocessor	10-11
Transferring from a Coprocessor to a TMS34020 Register	10-13
Transfer Memory to Coprocessor	10-14
Transferring from Coprocessor to Memory	10-16
Synchronization of Multiple TMS34020s	11-4
Releasing Control of the Local-Memory Bus and Control Signals (GI Driven High During a Host-Default Idle Cycle)	11-8
Releasing Control of the Local-Memory Bus and Control Signals (GI Driven High at the End of a Write Cycle)	11-9
Releasing Control of the Local-Memory Bus and Control Signals (GI Driven High at the End of a Host Read Cycle)	11-10
Releasing Control of the Local-Memory Interface (GI Driven High During a Page-Mode Sequence)	11-11
Regaining Control of the Local-Memory Bus and Control Signals	11-12
Passing Control of the Local-Memory Between Two TMS34020s	11-14
Graphics Operations Interaction	12-5
How XY and Linear Arrays are Stored in Off-Screen Memory	2-10
Possible Starting Corners	12-11
An Example of the Color-Expand Operation 1	2-13
A Trapezoidal Fill 1	2-18
Setting the W[CONTROL] Bits to Select a Window-Checking Mode 1	2-19
Specifying Window Limits 1	2-20
Outcodes for Line Endpoints	2-24
Using Midpoint Subdivision to Determine Which Portion of a Line Lies Within a Window	2-25
The PPOP [CONTROL] Bits 1	2-27
Examples of Operations on Single-Bit Pixels 1	2-28
Examples of Boolean and Arithmetic Operations 1	2-31
Examples of Operations on Pixel Intensity 1	2-33
	Electron Beam Pattern for a Typical Interlaced Display

12–14	Enabling Transparency and Selecting a Transparency Mode
12–15	Replicating the Plane-Mask Value Through PMASK 12-39
12–16	Read Cycle with Plane Masking, Transparency on Result = 0 12-40
12–17	Write Cycle with Transparency on Result=0 and Plane Masking
12–18	Filled Area for Example 12–4    12-43
12–19	How an XY Address Is Represented 12-47
12–20	How Values Are Contained in a CONVxP Register 12-49
13–1	An Example of Immediate Addressing 13-2
13–2	An Example of Absolute Addressing 13-3
13–3	An Example of Register-Direct Addressing 13-4
13–4	An Example of Register-Indirect Addressing
13–5	An Example of Register-Indirect with Offset Addressing
13–6	An Example of Register-Indirect with Postincrement Addressing
13–7	An Example of Register-Indirect with Predecrement Addressing
13–8	Register-to-Memory Moves
13–9	Memory-to-Register Moves
13–10	Memory-to-Memory Moves
13–11	A Trapezoidal Fill
13–12	Vector Address Map 13-254
13–13	Vector Address Map 13-257
14–1	Coprocessor Instruction Information on the LAD bus
14–2	How General Coprocessor Instruction Syntax
	Corresponds to TMS34082 Pseudo-ops 14-5
14–3	TMS34082 Registers That Can Be Used as Pseudo-op Operands 14-6
14-4	TMS34082 Register Sequence List 14-7
A–1	Typical Setup Using the TMS34020 Emulator and Your Target System A-2
A–2	Connecting the TMS34020 Emulator to Your Target System A-2
A–3	12-Pin Header Signals A-3
A–4	LCLK1 Buffer Restrictions A-5
A5	Emulator Pod Interface A-6
A6	Target Cable A-9
A–7	Pod Dimensions A-9
A8	12-Pin Connector Dimensions A-10

# Tables

.....

......

1-1	Quick Comparison of TMS34010 and TMS34020 Features	1-16
2-1	Numerical List of TMS34020 Pin Assignments (145-Pin PGA)	2-3
2-2	Alphabetical List of TMS34020 Pin Assignments (145-Pin PGA)	2-4
2-3	Numerical List of TMS34020 Pin Assignments (132-Pin QFP)	2-6
2-4	Alphabetical List of TMS34020 Pin Assignments (132-Pin QFP)	2-7
2-5	TMS34020 Pin Descriptions	2-9
2-6	Bus-Cvcle Completion Conditions	2-12
3–1	Decoding the Field-Size Bits in the Status Register	3-5
4–1	Definitions of Bits in the Status Register	4-2
4–2	How Instruction Execution Affects the PC	4-4
4–3	Summary of B-File Registers' Implied-Operand Functions	4-8
4-4	Summary of I/O Registers	4-10
6–1	Interrupt Priorities	6-7
6–2	Sources of Interrupt Delay	6-12
6–3	External Interrupt Vectors	6-15
6–4	Interrupts That are Associated with Internal Events	6-16
65	Initial State of Output Pins while RESET and GI are Low	6-22
7–1	Host Interface Estimated Maximum Bandwidth	7-34
7–2	Sources of Delay	7-37
8—1	Priorities for Memory Cycle Requests	8-6
8–2	LAD-Bus Status Codes	8-10
8–3	Bus Cycle Completion Conditions	8-12
8–4	Interpretation of SIZE16	8-25
8–5	Connections of 4-Bit VRAMs to the TMS34020 LAD Bus for 4 Bits per Pixel	8-41
8–6	Data Remapping for Block Write at 4 Bits per Pixel	8-42
8–7	Block-Write Data Expansion	8-42
8–8	Connections of 4-Bit VRAMs to the TMS34020 LAD Bus for 8 Bits per Pixel 8	8-43
8–9	Data Remapping for Block Write at 8 Bits per Pixel	8-43
8–10	DRAM Array Sizes	8-52
8–11	Logical Addresses Output on the RCA Bus	8-53
8–12	Example Connections to the RCA Bus	8-55
9—1	Screen-Refresh Latency	9-50
9–2	Minimum Horizontal-Blanking Duration	9-51

\*

9–3	Y-Zoom Control	9-57
10–1	TMS34020 General Coprocessor Instructions	
10–2	Suggested Coprocessor ID Assignments	10-6
11–1	Bus Request Codes for the Multiprocessor Interface	
11–2	Arbitration Scheme for Two TMS3402020s	11-16
11–3	Arbitration Scheme for One TMS3402020 and a Hold Device	11-17
12–1	Summary of Graphics Instructions	12-2
12–2	Summary of Graphics Operations	
12–3	PIXBLTs That Can Start from Any Corner	
12–4	Window-Checking Modes for Single-Pixel Instructions	12-21
12–5	Window-Checking Modes for Pixel Array Instructions	12-21
12–6	Window-Checking Modes for the LINE Instruction	12-23
12–7	Pixel-Processing Options	12-27
12–8	Summary of Implied Operands Used by the Graphics Instructions	12-45
12–9	TMS34020 Conversion (CONVxP) Registers	12-48
13–1	Summary of MOVE Instructions	13-19
13–2	Condition Codes for JRcc and JAcc Instructions	13-27
13–3	Summary for XY Instructions	13-29
13–4	Summary of Operand Formats for the MOVB Instruction	13-154
13–5	Summary of Operand Formats for the MOVE Instruction	13-159
13–6	Summary of Array Types for the PIXBLT Instruction	13-191
13–7	Summary of B-File Registers for PIXBLT Instructions	13-191
13–8	Summary of I/O Registers for the PIXBLT Instructions	13-192
13–9	Summary of Operand Formats for the PIXT Instructions	13-206
13–10	Summary of B-File Registers for PIXT Instructions	13-206
13–11	Summary of I/O Registers for the PIXT Instructions	13-207
14–1	Symbols Used in Pseudo-op Syntax Listings	
15–1	Effects of Pixel-Processing Options on Graphics Instructions	15-2
15–2	Cases Table for MOVE and MOVB Timings	15-10
15–3	Source/Destination Alignment for MOVE and MOVB Timings	15-12

# Examples

5—1	Code Without Branches or Immediate Data 5-	-6
5–2	Code with Branches 5-	-7
5–3	Code with Immediate Data 5-	-7
12–1	Transparency on Result = 0 for PIXT *Rs, *Rd 12-3	37
12–2	Transparency on Source = COLOR0 for PIXT *Rs, *Rd	38
12–3	Transparency on Destination = COLOR0 for PIXT *Rs, *Rd 12-3	38
12–4	Setting Up Implied Operands for a FILL Instruction 12-4	14



## **Chapter 1**

## **Overview of the TMS34020**

The TMS34020 Graphics System Processor (GSP) is an advanced, 32-bit microprocessor, optimized for graphic display systems. The TMS34020 is the second generation of the TMS340 family of computer graphics products from Texas Instruments.

The TMS34020 provides a high-performance, cost-effective solution for applications that require efficient data manipulation in a graphics environment. The TMS34020 can be configured to serve in a host-based, stand-alone, or multiprocessing system. The TMS34020 has host and multiprocessor interfaces to facilitate implementation of multiple TMS34020 systems.

The TMS34020 is well supported by a full set of hardware and software development tools, including an optimizing C compiler, an assembler, software libraries, a PC-based development board, and an emulator. In addition, the TMS34020 is fully compatible with and supported by the Texas Instruments Graphics Architecture (TIGA-340).

	Sect	ion	Page
TMS34020-specific information	1.1	Key Features	1-2
describes characteristics of the	1.2	Typical Applications	1-3
TMS34020 processor.	1.3	Major Components of	
		the TMS34020 Architecture	1-4
Information about related products	1.4	System Development Tools	1-10
describes the development tools	1.5	Processors for a Graphics System	1-14
and devices for supporting the	1.6	Compatibility Between the	
compatibility with earlier devices.		TMS34010 and TMS34020	1-16

Topics covered in this introductory section include

### 1.1 Key Features of the TMS34020

- Fully programmable 32-bit general-purpose processor with 512-Mbyte linear address range (bit addressable)
- Second-generation graphics system processor
  - Object code compatible with the TMS34010
  - Enhanced instruction set
  - Optimized graphics instructions
  - Direct coprocessor interface to TMS34082 floating-point processor
- Instruction cycle times:
  - TMS34020-40 ..... 100 ns
  - TMS34020-32 .... 125 ns
- On-chip peripheral functions include
  - Programmable CRT control
  - Direct DRAM/VRAM interface
  - Direct communication with an external (host) processor
  - Communication with multiple TMS34020s
  - Functional expansion with the coprocessor interface
  - Automatic CRT display refresh
- Instruction set supports special graphics functions such as pixel processing, XY addressing, and window checking
- Programmable 1-, 2-, 4-, 8-, 16-, or 32-bit pixel size
- 16 Boolean and 6 arithmetic pixel processing options (raster-ops)
- 30 general-purpose 32-bit registers
- **512-byte LRU on-chip instruction cache**
- Optimized DRAM/VRAM interface
  - Page mode for burst memory operations up to 40 Mbytes per second
  - Dynamic bus sizing (16-bit and 32-bit transfers)
  - Byte-oriented CAS strobes
  - Automatic CRT display refresh
- Flexible host processor interface
  - Supports host transfers at up to 20 Mbytes per second
  - Direct access to all of the TMS34020 address space
  - Implicit addressing (autoincrementing)
  - Prefetching for enhanced read access
- Flexible multiprocessor interface

- Programmable CRT control
  - Composite sync mode
  - Separate sync mode
  - Synchronization to external sync
- Direct support for special features of 1M VRAMs
  - Load write mask
  - Load color mask
  - Block write
  - Write using the write mask

### 1.2 Typical Applications of the TMS34020

The TMS34020's 32-bit processing power and its ability to handle complex data structures make it well suited for a variety of applications. Typical applications that take advantage of the TMS34020's features include

Computers	Industrial Control
Terminals and CRTs	Robotics
Windowing systems	Process control
Electronic publishing	Instrumentation
Laser printers	Motor control
Personal computers	Navigation
<ul> <li>Printers and plotters</li> <li>Engineering workstations</li> <li>Copiers</li> <li>Document readers</li> <li>FAX</li> <li>Imaging</li> </ul>	Consumer Electronics Automotive displays Information terminals Cable TV Home control Video games
Data processing	Telecommunications           Video phones
	PBX

### 1.3 Major Components of the TMS34020 Architecture

The TMS340 family of processors from Texas Instruments combines the best features of general-purpose processors and graphics controllers to create a range of cost-effective, flexible, powerful graphics systems. The key features of the TMS340 family are speed, a high degree of programmability, and efficient manipulation of hardware-supported data types such as pixels and 2-dimensional pixel arrays.

With a built-in instruction cache, the ability to simultaneously access memory and registers, and an instruction set that enhances raster graphics operations, the TMS34020 provides programmable control of the CRT interface as well as the memory interface (both standard DRAM and multiport VRAM). The TMS34020's 4-gigabit (512 Mbyte) physical address space is completely bit-addressable on bit boundaries using variable-width data fields. Graphics addressing modes support 1-, 2-, 4-, 8-, 16-, and 32-bit pixels.

The TMS34020's unique memory interface reduces the time needed to perform tasks such as bit alignment and masking while supporting advanced DRAM access modes. The 32-bit architecture supplies the large blocks of continuously addressable memory that are necessary in graphics applications. Systems designed with the TMS34020 can take advantage of VRAM technology to facilitate applications such as high-bandwidth frame buffers; this circumvents the bottleneck often encountered when using conventional DRAMs in graphics systems.

The TMS34020 instruction set includes a full complement of general- purpose instructions, as well as graphics functions, that you can use to construct efficient high-level functions. The instructions support arithmetic and Boolean operations, data moves, conditional jumps, and subroutine calls and returns. The TMS34020 instruction set also supports the TMS34082 as a coprocessor.

The TMS34020 architecture supports a variety of pixel sizes, frame buffer sizes, and screen sizes. On-chip functions have been carefully selected so that no functions tie the TMS34020 to a particular display resolution. This enhances the portability of graphics software and allows the TMS34020 to adapt to graphics standards such as TIGA, MIT's X, CGI/CGM, GKS, NAPLPS, PHIGS, and evolving industry and display-management standards.

Figure 1–1 illustrates the TMS34020's internal architecture.



Figure 1–1. TMS34020 Block Diagram

### **1.3.1 Internal Functions**

The center portion of Figure 1–1 highlights the main internal functions of the TMS34020 CPU.

- The 32-bit status register (ST) contains several bits that indicate the CPU status. Section 4.1 (page 4-2) discusses the status register.
- The 32-bit program counter (PC) points to the next instruction word to be fetched. The PC's four LSBs are always 0. Section 4.2 (page 4-4) discusses the program counter.
- Register files A and B each contain fifteen 32-bit general-purpose registers. The B-file registers are also used as implied operands for the graphics instructions. Section 4.4 (page 4-6) discusses the register files.
- ❑ The 32-bit stack pointer (SP) contains the bit address of the top of the system stack. The SP is also available to instructions that operate on either register file. For more information, refer to Section 4.3 (page 4-5).

- The 32-bit barrel shifter shifts or rotates 32-bit operands from 1 to 32 bit positions in a single machine cycle. This user's guide does not discuss barrel-shifter operation because the operation is transparent.
- □ The 32-bit ALU allows the TMS34020 to perform most register-to-register operations in a single machine state. (Accessing external memory requires a minimum of two states.) The following actions can occur in parallel during a single machine state:
  - Two operands are transferred from the selected general-purpose register file to the ALU.
  - The ALU performs the specified operation on the operands.
  - The result is routed back to the general-purpose register file.

### Instruction cache

The TMS34020 contains a 512-byte instruction cache that can contain up to 256 instruction words (an instruction word may be an entire single-word instruction or 16 bits of a multiple-word instruction). When the cache is enabled, the TMS34020 provides single-cycle execution of general-purpose instructions and of most integer arithmetic and Boolean operations.

Chapter 5 discusses cache operation.

### I/O registers

Fifty-four 16-bit, on-chip registers are dedicated to peripheral control functions. The I/O registers are divided into five categories:

- ❑ Local-memory registers are dedicated to controlling functions such as bigendian/little-endian addressing, refresh rate, row/column mode, plane masking, refresh address, and recovery from bus faults.
- □ Video timing and screen-refresh registers generate the sync and blanking signals used to drive a CRT, schedule screen refreshes, and allow external synchronization.
- □ *Host-interface registers* help the TMS34020 to communicate with a host processor.
- □ Interrupt-control registers provide status information about interrupt requests.
- CPU-control registers configure the TMS34020 to operate with specific characteristics.

Section 4.6 (page 4-14) provides individual descriptions of each I/O register.

Microcontrol ROM	
	The TMS34020 transfers decoded instructions to the microcontrol ROM for in- terpretation.
Clock timing logic -	
	The clock timing logic converts the clock-input signal (CLKIN) to internal timing signals and generates the clock-output signals, LCLK1 and LCLK2, used by external devices. The machine state is a fundamental time unit of the TMS34020's graphics processor; it is the time interval during which the processor is in a particular microinstruction state. The instruction timing for each assembly-language instruction is specified in multiples of machine states. The TMS34020's machine state is a single local clock period in duration (the time from one LCLK1 low-to-high transition to the next). The local clock period is four times the period of CLKIN.
Host control logic -	
	The host control logic allows a host processor to communicate with the TMS34020 and allows access to TMS34020 local memory. Commands, data, and status information are communicated through this logic.
Page-mode register	rs
	The page-mode registers buffer data to and from the local-memory interface so that data may be temporarily stored during processing. This enhances data flow to memory.
Other special proce	essing hardware
	The TMS34020 CPU also supports the following special processing functions in hardware:
	<ul> <li>Detecting whether a pixel lies within a specified display window</li> <li>Detecting the leftmost or rightmost 1 within a 32-bit register</li> <li>Expanding a black-and-white pattern to a variable pixel-depth pattern</li> <li>Rotating and merging variable-width fields</li> </ul>

- □ Individual byte strobes for partial word writes to memory
- Dynamic bus sizing
- Data bus swizzling for special VRAM block modes
- Big-endian and little-endian addressing modes

### 1.3.2 Major Interfaces

### Local-memory and DRAM/VRAM interfaces -

The TMS34020's local-memory interface consists of a 32-bit, bidirectional address/data bus, various control signals, and row/column address control. During a local-memory cycle, address and status information are output on the local address/data (LAD) bus; then, data is transferred over the same LAD lines. The TMS34020 can transfer data over 16-bit or 32-bit buses.

The TMS34020 interfaces directly to DRAMs and VRAMs, providing address multiplexing for  $64K \times n$ ,  $256K \times n$ ,  $1M \times n$ , and  $4M \times n$  devices. The row and column addresses necessary for accessing DRAMs and VRAMs are available directly from the TMS34020's RCA bus, eliminating the need for external multiplexing hardware.

For more information, refer to Chapter 8.

### Video interface ·

The TMS34020's video interface is extremely flexible and programmable, allowing you to choose between

- Separate sync and blanking or composite sync and blanking
- Synchronization to externally or internally generated video signals
- Interlaced or noninterlaced video

The video interface directly supports VRAMs by generating the serial-register transfers necessary for refreshing a display.

For more information, refer to Chapter 9.

### Host interface

The host interface allows you to map the TMS34020's local memory into a host's memory address space. This allows you to transfer data, commands, and status information between the TMS34020 and the host processor.

For more information, refer to Chapter 7.

#### Coprocessor interface –

The coprocessor interface allows you to extend the TMS34020's basic architecture. Most coprocessor interfaces require a memory-mapped approach, so that a processor treats a coprocessor as a peripheral device. The TMS34020, however, allows direct connection to a coprocessor and provides special instructions that allow you to send instructions and data between the TMS34020 and a coprocessor. The TMS34020 provides extended coprocessor support for the TMS34082 Floating-Point Processor, which is specially designed to serve in a TMS34020 system.

For more information, refer to Chapter 10.

#### Multiprocessor interface ·

The multiprocessor interface allows multiple TMS34020s (as well as other processors) to share the same local memory. The TMS34020's grant-in and request-priority signals provide a flexible method of passing control from one processor to another. The multiprocessor interface requires external arbitration logic to

inform a TMS34020 when it can take control of the bus, and

decode the priorities of requests from the multiple processors.

This scheme allows back-to-back memory cycles even when control passes from one TMS34020 to another.

Any number of devices can be configured together within a single system. However, system performance is not increased significantly when a system contains more than three TMS34020s.

For more information, refer to Chapter 11.

### **Emulation interface**

The TMS34020 supports a 4-wire interface that simplifies connections between a debugger and a target system. For details about emulation, refer to Appendix A.
## 1.4 System Development Tools

The TMS34020 is well supported by a complete set of hardware and software development tools, including a C compiler, an assembler/linker, software libraries, and a PC-based development board. In addition, the TMS34020 is fully compatible with and supported by the Texas Instruments Graphics Architecture (TIGA-340).

#### 1.4.1 Code-Generation Tools

Figure 1–2 illustrates the TMS34020 code development flow. The figure highlights the most common paths of software development; the other portions are optional.

Figure 1–2. TMS34020 Software Development Flow



These tools use common object files format (COFF), which encourages modular programming. COFF allows you to divide your code into logical blocks, define your system's memory map, and then link code into specific memory areas. COFF also provides rich support for source-level debugging.

The following list describes the tools shown in Figure 1–2.

The TMS34020 **C compiler** is a full-featured optimizing compiler that translates standard Kernighan-and-Ritchie C programs into TMS34020 assembly-language source. Key characteristics include

- Standard Kernighan-and-Ritchie C with extensions. The compiler compiles standard C programs as defined by Kernighan and Ritchie's The C Programming Language (first edition). The compiler supports these standard extensions: enumeration types, structure assignments, passing structures to functions, and returning structures from functions. A future release of the compiler will support the full ANSI standard.
- Big-endian or little-endian code.
- Optimization. The compiler uses several advanced techniques for generating efficient, compact code from C source.
- Assembly-language output. The compiler generates assembly- language source that is easily inspected, enabling you to see the code generated from the C source files.
- ANSI standard runtime support. The compiler package comes with a complete runtime library that conforms to the ANSI C library standard. The library includes functions for string manipulation, dynamic memory allocation, data conversion, timekeeping, trigonometry, exponential, and hyperbolic functions. Functions for I/O and signal handling are not included because they are application-specific.
- Flexible assembly-language interface. The compiler has straight-forward calling conventions, allowing you to easily write assembly and C functions that call each other.
- Shell program. The compiler package includes a shell program that enables you to compile, assemble, and link programs in a single step.
- Source interlist utility. The compiler package includes a utility that interlists your original C source statements into the assembly-language output of the compiler. This utility provides you with an easy method for inspecting the assembly code generated for each C statement.

The **assembler** translates assembly-language source files into machine language object files.

The **archiver** allows you to collect a group of files into a library. It also allows you to modify a library by deleting, replacing, extracting, or adding members.

#### C compiler

assembler

archiver

One of the most useful applications of the archiver is to build a library of object modules. Several object libraries and a source library are included with the C compiler.

You can also use application-specific object libraries, available as separate products:

- □ The **math/graphics function library** contains math functions for performing algebraic, trigonometric, and transcendental operations as well as graphics functions for performing viewport management, bitmapped text, graphics output, color-palette control, 3-dimensional transformations, and graphics initialization.
- The font library contains a variety of proportionally spaced and monospaced fonts. You can use the functions in the graphics library to display the fonts.
- The CCITT data compression function library contains CCITT-compatible routines for compressing and decompressing monochrome image data.
- The **8514 adaptor emulation function library** contains routines for emulating IBM PS/2 high-resolution display.

These functions and routines can be called from C programs. You can also create your own object libraries.

The **linker** combines object files into a single, executable object module. As the linker creates the executable module, it performs relocation and resolves external references. The linker is a tool that allows you to define your system's memory map and associate blocks of code with defined memory areas.

The main purpose of the development process is to produce a module that can be executed in a **TMS34020 target system**. You can use one of several debugging tools to refine and correct your code. Available products include a PC-based **software development board** (SDB) and a realtime in-circuit **emula-tor**.

An **object format converter** is also available; it converts a COFF object file into an Intel, Tektronix, or TI-tagged object-format file that can be downloaded to an EPROM programmer.

## 1.4.2 Supported Systems

The TMS34020 C compiler and assembly language tools are available for these systems:

- IBM-PC with PC-DOS
- VAX:
  - VMS
  - Ultrix

linker

debugging

tools

object format

converter

- Apollo workstations:
  - Domain/IX
  - AEGIS
- Sun-3 workstations with UNIX
- Macintosh with MPW

#### 1.4.3 Packages

Texas Instruments supplies development tools in several packages.

#### Assembly language tools package



## 1.4.4 TIGA-340 Graphics Interface

The Texas Instruments Graphics Architecture (TIGA-340) is a software interface standard for the TMS340 family of graphics system processors. TIGA enhances the performance of MS-DOS-based PCs that contain a TMS34010 or TMS34020 and an 8088/86 or 80286/80386 host microprocessor by optimizing communications between the graphics processor and the host processor. The TIGA interface allows the host and graphics processors to share execution of the application.

#### Figure 1–3. Graphics Processing Shared Between TMS340 and Host Processors



## 1.5 **Processors for a Graphics System**

Texas Instruments offers a broad line of graphics and video products. The TMS34020 Graphics System Processor, TMS34082 Floating-Point Processor, and TMS44C251 1-Mbit Video RAM bring workstation performance to the PC and other small systems. Figure 1–4 shows all of the Texas Instruments Graphics products.





The following paragraphs describe the TMS34082 and TMS44C251, which are included in the TMS34020 sample system shown in Figure 1–5.

#### TMS34082-

Many TMS34020 applications require floating-point operations. The TMS34082 floating-point processor is designed to interface directly with the TMS34020, allowing the TMS34020 to perform computation-intensive functions more than 100 times faster than a software implementation. The TMS34082 performs single- and double-precision floating-point operations, conforming fully to the IEEE 754 standard.

In addition to normal floating-point operations, the TMS34082 performs complex 2- and 3-dimensional operations such as 3×3 convolution, 4×4 matrix, and cubic spline operations.

Additional TMS34082 features include

- 32-bit data path
- 32-bit integer and logical operations
- 40-MFLOPS sustained operation
- Single-instruction divide/square-root operations
- External microcode memory interface for defining custom instructions

## TMS44C251

A video RAM (VRAM) is a special memory device, optimized for use in graphics systems. The TMS44C251 multiport VRAM is a high-speed, dual-ported memory. It consists of DRAM organized as 262,144 4-bit words, interfaced to a serial data register.

## Sample system

A typical graphics system designed with the TMS34020 uses several types of memory, as well as external latches, buffers, and transceivers to connect the TMS34020 to the memories, a coprocessor, or a host processor. Figure 1–5 shows a representative TMS34020 design for a PC display system. Note that this system uses the TMS34082 as a coprocessor, a palette, and VRAMs, DRAMs, and ROM memories.

Figure 1–5. TMS34020 1K×1K×8 PC Display System



## 1.6 Compatibility Between the TMS34020 and TMS34010

The information in this section is for readers who are familiar with the TMS34010 graphics system processor. If you are not familiar with the TMS34010, you may want to skip this section. Note that this user's guide *does not* require you to be familiar with the TMS34010.

The TMS34020 is the second generation of the TMS340 family of graphics system processors; the TMS34010 is the first generation. The TMS34010 was the building block for the TMS34020; however, the TMS34020 greatly extends the TMS34010's capabilities by adding new features and enhancing existing features. Table 1–1 shows a sample comparison of the TMS34010 and TMS34020 features.

Note that the TMS34010 and TMS34020 are not pin-for-pin compatible.

Table 1–1. Quick Comparison of TMS34010 and TMS34020 Features

Feature	ТМ	S34010	ТМ	S34020
External bus size		16 bits		32 bits
Cycle time		130, 160, or 200 ns		100 or 125 ns
Cache size		256 bytes		512 bytes
Horizontal pitch		Power of 2		Unlimited
Word addressing		Little endian		Little or big endian
VRAM support		Serial registers		Serial registers Block writes Split serial registers Enhanced page mode
Interfaces		Host		Host
	D	Hold		Coprocessor Multiprocessor
Coprocessor support		Memory mapped		Direct connection

Throughout this user's guide you'll find descriptions of compatibility between the TMS34010 and TMS34020. Such passages are marked with this symbol in the margin:



TMS34010 object code is upward compatible with the TMS34020. If new TMS34020 features would prevent TMS34010 code from running, the TMS34020 provides you with a method of switching these features off. At reset, these features are off to provide compatibility with the TMS34010.

In general, if you followed the compatibility notes in the *TMS34010 User's Guide*, your TMS34010 code should be object-code compatible with the TMS34020.

The following list describes restrictions that TMS34010 code must adhere to in order to be compatible with the TMS34020.

- Color information. The TMS34020 uses all 32 bits of the COLOR0 (B8) and COLOR1 (B9) values. The TMS34010 used only the 16 LSBs of these values. Although the TMS34010 will ignore the 16 MSBs of these values, TMS34010 code should replicate the color information throughout all 32 bits of these registers.
- Plane mask. All 32 bits of the PMASK register, at addresses C000 0160h (16 LSBs) and C000 0170h (16 MSBs), are valid for the TMS34020. TMS34010 could should copy the 16-bit PMASK value at address C000 0160h to address C000 0170h.
- Reserved bits. TMS34010 code should not use any reserved bits in the status register or the I/O registers.
- Register B13. The TMS34020 uses register B13 as a pattern register. TMS34010 code should load B13 with all 1s, causing the code to draw a solid line instead of an unexpected patterned line.
- CONVSP & SPTCH, CONVDP & DPTCH. The TMS34020 uses SPTCH and DPTCH to determine the values of CONVSP and CONVDP, respectively. TMS34010 code should be sure that SPTCH and DPTCH agree with CONVSP and CONVDP. That is, the 5 LSBs of CONVxP must equal the 1s complement of log₂(xPTCH), which is given by the LMO of xPTCH. Set the 11 MSBs of CONVxP to 0.

If an instruction uses CONVSP or CONVDP, then the MSB of CONVxP should be 0 and xPTCH should contain 2<sup>CONVxP</sup> before instruction execution.

- □ **Timing loops.** TMS34010 code should avoid timing loops; obtain timing via the video logic (using DPYINT) or via external interrupt 1 or 2.
- Data alignment. For optimum TMS34020 performance, TMS34010 code should align to 32-bit boundaries (instead of 16-bit boundaries).
- **Cache**. TMS34010 code should not depend on cache-load order.
- Saving the graphics context. If TMS34010 code requires saving/restoring of the graphics context, the code should store the I/O registers at addresses C000 00B0h and C000 0130—C000 01A0h (inclusive).
- Reset vector. At reset, the TMS34020 loads the 4 LSBs of the reset vector into the 4 LSBs of the CONFIG register. TMS34010 code should not depend on values in the 4 LSBs of the reset vector.

- Video registers. All accesses to video timing registers should be separate from other code. Particularly, HESYNC, HEBLNK, HSBLNK, HTOTAL, HCOUNT, VESYNC, VEBLNK, VSBLNK, VTOTAL, and VCOUNT should be manipulated through symbolic names (not by addresses) because their addresses have changed.
- Interrupt routines. Interrupt service routines for the TMS34010 should make no assumptions about the state of the stack, except that the PC and ST are stacked after any extra words. The interrupt routine must return with an RETI instruction, which will pop any extra words to the correct internal registers.
- □ Illegal opcodes. TMS34010 code should not depend on any of the TMS34010's illegal opcodes (except 0000h) to cause a TRAP 30.
- **Traps.** Trap FFFF FBC0h is the TMS34020's bus-fault trap. Trap FFFF FBE0h is the TMS34020's single-step trap.
- ❑ Host interface. To TMS34010 code, the TMS34020's host interface appears the same as the TMS34010's host interface. It is desirable for data shared with the host to be aligned on 32-bit boundaries. In general, code written for the TMS34010 host interface will need to be changed because the TMS34020's host interface is different from the TMS34010's.

# **Pinouts and Signal Descriptions**

This chapter illustrates the TMS34020 pinouts and provides detailed descriptions of the TMS34020's signals. For mechanical dimensions of TMS34020 packages, refer to the *TMS34020 Data Sheet*.

	Section	Page
The <b>pinouts</b> section illustrates the two packages that the TMS34020 is available in, and associates the signal names with the correct pin numbers for each device.	2.1 Pinouts	2-2
The TMS34020's pins are divided among the TMS34020's <b>major</b> interfaces.	2.2 The TMS34020's Major Interfaces	. 2-8
The TMS34020's signals are described, in detail, as they apply to the various interfaces.	2.3 Signal Descriptions	. 2-9

## 2.1 Pinouts

The TMS34020 is offered in two packages:

a 145-pin grid array (PGA) package and

a 132-pin quad flat package (QFP).

Figure 2–1 shows the pinout of the 145-pin PGA, and Figure 2–2 shows the pinout for the 132-pin QFP.

Figure 2–1. TMS34020 Pinout, 145-Pin PGA Package (Bottom View)

	A	В	C	D	E	H	G	н	J	ĸ	L	М	N	Р	н	
1	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	
2	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	
3	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	
4	0	0	0	0									0	0	0	
5	0	0	0										0	0	0	
6	0	0	0										0	0	0	
7	0	0	0										0	0	0	
8	0	0	0										0	0	0	
9	0	0	0										0	0	0	
10	0	0	0										0	0	0	
11	0	0	0										0	0	0	
12	0	0	0										0	0	0	
13	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	
14	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	
15	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	

Pin #	Signal								
A1	V <sub>SS</sub>	B15	LAD12	F1	HRDY	K15	LAD20	P2	HWRITE
A2	ALTCH	C1	CAS0	F2	R0	L1	LINT1	P3	HCS
A3	CBLNK/ VBLNK	C2	V <sub>cc</sub>	F3	V <sub>SS</sub>	L2	CAMD	P4	HA30
A4	HSYNC	СЗ	DDOUT	F13	LAD24	L3	LRDY	P5	HA27
A5	TR/QE	C4	DDIN	F14	LAD8	L13	LAD1	P6	HA24
A6	RCA2	C5	V <sub>SS</sub>	F15	V <sub>SS</sub>	L14	LAD2	P7	HA22
A7	RCA3	C6	SF	G1	HINT	L15	LAD19	P8	HA18
A8	V <sub>cc</sub>	C7	RCA4	G2	HOE	M1	BUSFLT	P9	HA14
A9	RCA6	C8	V <sub>SS</sub>	G3	HDST	M2	PGMD	P10	HA13
A10	RCA7	C9	RCA8	G13	LAD7	MЗ	VCLK	P11	HA10
A11	RCA10	C10	RCA12	G14	V <sub>SS</sub>	M13	V <sub>SS</sub>	P12	HA7
A12	SCLK	C11	LAD30	G15	LAD23	M14	LAD16	P13	HA5
A13	LAD15	C12	V <sub>SS</sub>	H1	LCLK1	M15	LAD18	P14	HBS0
A14	LAD29	C13	V <sub>SS</sub>	H2	EMU3	N1	SIZE16	P15	LAD0
A15	V <sub>SS</sub>	C14	V <sub>cc</sub>	H3	LCLK2	N2	V <sub>cc</sub>	R1	HREAD
B1	CAS3	C15	LAD26	H13	LAD22	N3	CLKIN	R2	HA31
B2	WE	D1	RAS	H14	LAD21	N4	V <sub>SS</sub>	R3	HA28
B3	V <sub>SS</sub>	D2	CAS2	H15	LAD6	N5	HA29	R4	HA26
B4	CSYNC/ HBLNK	D3	V <sub>SS</sub>	J1	EMU0	N6	HA25	R5	HA23
B5	VSYNC	D4	NC	J2	GI	N7	HA21	R6	HA20
B6	RCA0	D13	LAD28	J3	EMU1	N8	V <sub>SS</sub>	R7	HA19
B7	RCA1	D14	LAD11	J13	LAD4	N9	V <sub>SS</sub>	R8	HA17
B8	RCA5	D15	LAD10	J14	V <sub>cc</sub>	N10	HA12	R9	HA16
B9	RCA9	E1	R1	J15	LAD5	N11	HA6	R10	HA15
B10	RCA11	E2	V <sub>cc</sub>	K1	EMU2	N12	HBS2	R11	HA11
B11	LAD31	E3	CAS1	K2	RESET	N13	HBS1	R12	HA9
B12	LAD14	E13	LAD27	КЗ	LINT2	N14	V <sub>cc</sub>	R13	HA8
B13	V <sub>CC</sub>	E14	LAD25	K13	V <sub>SS</sub>	N15	LAD17	R14	HBS3
B14	LAD13	E15	LAD9	K14	LAD3	P1	V <sub>cc</sub>	R15	V <sub>SS</sub>

Table 2–1. Numerical List of TMS34020 Pin Assignments (145-Pin PGA)

Note: Pin D4 is NC (not internally connected). You may use this pin for package alignment, but do not connect it.

Signal	Pin #	Signal	Pin #	Signal	Pin #	Signal	Pin #	Signal	Pin #
ALTCH	A2	HA17	R8	LAD2	L14	LAD31	B11	V <sub>cc</sub>	A8
BUSFLT	<b>M</b> 1	HA18	P8	LAD3	K14	LCLK1	H1	V <sub>cc</sub>	B13
CAMD	L2	HA19	R7	LAD4	J13	LCLK2	HЗ	V <sub>cc</sub>	C2
CAS0	C1	HA20	R6	LAD5	J15	LINT1	L1	V <sub>cc</sub>	C14
CAS1	E3	HA21	N7	LAD6	H15	LINT2	КЗ	V <sub>cc</sub>	E2
CAS2	D2	HA22	P7	LAD7	G13	LRDY	L3	V <sub>cc</sub>	J14
CAS3	B1	HA23	R5	LAD8	F14	NC	D4	V <sub>cc</sub>	N2
CBLNK/ VBLNK	A3	HA24	P6 .	LAD9	E15	PGMD	M2	V <sub>cc</sub>	N14
CLKIN	N3	HA25	N6	LAD10	D15	R0	F2	V <sub>cc</sub>	P1
CSYNC/ HBLNK	B4	HA26	R4	LAD11	D14	R1	E1	VCLK	МЗ
DDIN	C4	HA27	P5	LAD12	B15	RAS	D1	V <sub>SS</sub>	A1
DDOUT	C3	HA28	R3	LAD13	B14	RCA0	B6	V <sub>SS</sub>	A15
EMU0	J1	HA29	N5	LAD14	B12	RCA1	B7	V <sub>SS</sub>	B3
EMU1	J3	HA30	P4	LAD15	A13	RCA2	A6	V <sub>SS</sub>	C5
EMU2	K1	HA31	R2	LAD16	M14	RCA3	A7	V <sub>SS</sub>	C8
EMŲ3	H2	HBS0	P14	LAD17	N15	RCA4	C7	V <sub>SS</sub>	C12
GI	J2	HBS1	N13	LAD18	M15	RCA5	B8	V <sub>SS</sub>	C13
HA5	P13	HBS2	N12	LAD19	L15	RCA6	A9	V <sub>SS</sub>	D3
HA6	N11	HBS3	R14	LAD20	K15	RCA7	A10	V <sub>SS</sub>	F3
HA7	P12	HCS	P3	LAD21	H14	RCA8	C9	V <sub>SS</sub>	F15
HA8 <sup>®</sup>	R13	HDST	G3	LAD22	H13	RCA9	B9	V <sub>SS</sub>	G14
HA9	R12	HINT	G1	LAD23	G15	RCA10	A11	V <sub>SS</sub>	K13
HA10	P11	HOE	G2	LAD24	F13	RCA11	B10	V <sub>SS</sub>	M13
HA11	R11	HRDY	F1	LAD25	E14	RCA12	C10	V <sub>SS</sub>	N4
HA12	N10	HREAD	R1	LAD26	C15	RESET	K2	V <sub>SS</sub>	N8
HA13	P10	HSYNC	A4	LAD27	E13	SCLK	A12	V <sub>SS</sub>	N9
HA14	P9	HWRITE	P2	LAD28	D13	SF	C6	V <sub>SS</sub>	R15
HA15	R10	LAD0	P15	LAD29	A14	SIZE16	N1	VSYNC	B5
HA16	R9	LAD1	L13	LAD30	C11	TR/QE	A5	WE	B2

Table 2–2. Alphabetical List of TMS34020 Pin Assignments (145-Pin PGA)

Note: Pin D4 is NC (not internally connected). You may use this pin for package alignment, but do not connect it.

Pinouts

		1
2-2. TMS34020 Pinout, 132-Pin QFP Package 1180 1191 1191 1191 1191 1191 1191 1191	o	83         84         85         87         87         87         87         87         88         88         88         88         88         89         89         80         81         81         81         81         81         81         81         81         81         81         81         82         83         84         84         84         85         86         86         86         86         86         86         86         87         86         87         87         87         87         87         87         87         87         87         87         87         87         87
б С		

Figure

and the second s							
Pin #	Signal	Pin #	Signal	Pin #	Signal	Pin #	Signal
1	EMU1	34	V <sub>SS</sub>	67	LAD7	100	RCA4
2	EMU0	35	HA16	68	LAD23	101	RCA3
3	EMU2	36	HA15	69	V <sub>SS</sub>	102	RCA2
4	GI	37	HA14	70	V <sub>SS</sub>	103	RCA1
5	RESET	38	HA13	71	LAD8	104	RCA0
6	LINT2	39	HA12	72	LAD24	105	SF
7	LINT1	40	HA11	73	LAD9	106	TR/QE
8	CAMD	41	HA10	74	LAD25	107	VSYNC
9	BUSERR	42	HA9	75	LAD10	108	HSYNC
10	SIZE16	43	HA8	76	LAD26	109	CBLNK/VBLNK
11	PGMD	44	HA7	77	LAD11	110	CSYNC/HBLNK
12	LRDY	45	HA6	78	LAD27	111	V <sub>SS</sub>
13	V <sub>cc</sub>	46	HA5	79	V <sub>cc</sub>	112	V <sub>SS</sub>
14	VCLK	47	HBS3	80	LAD12	113	ALTCH
15	CLKIN	48	HBS2	81	LAD28	114	DDIN
16	HWRITE	49	HBS1	82	V <sub>SS</sub>	115	DDOUT
17	HREAD	50	HBS0	83	LAD13	116	WE
18	HCS	51	LAD0	84	LAD29	117	CAS3
19	HA31	52	LAD16	85	LAD14	118	CAS2
20	HA30	53	LAD1	86	LAD30	119	CAS1
21	HA29	54	LAD17	87	LAD15	120	CAS0
22	HA28	55	LAD2	88	LAD31	121	V <sub>CC</sub>
23	HA27	56	LAD18	89	SCLK	122	RAS
24	HA26	57	V <sub>SS</sub>	90	RCA12	123	V <sub>SS</sub>
25	HA25	58	LAD3	91	RCA11	124	R0
26	HA24	59	LAD19	92	RCA10	125	R1
27	HA23	60	V <sub>cc</sub>	93	RCA9	126	HOE
28	HA22	61	LAD4	94	RCA8	127	HDST
29	HA21	62	LAD20	95	RCA7	128	HRDY
30	HA20	63	LAD5	96	RCA6	129	HINT
31	HA19	64	LAD21	97	RCA5	130	EMU3
32	HA18	65	LAD6	98	V <sub>cc</sub>	131	LCLK1
33	HA17	66	LAD22	99	V <sub>SS</sub>	132	LCLK2

Table 2–3. Numerical List of TMS34020 Pin Assignments (132-Pin QFP)

Signal	Pin #	Signal	Pin #	Signal	Pin #	Signal	Pin #
ALTCH	113	HA21	29	LAD10	75	RCA2	102
BUSERR	9	HA22	28	LAD11	77	RCA3	101
CAMD	8	HA23	27	LAD12	80	RCA4	100
CAS0	120	HA24	26	LAD13	83	RCA5	97
CAS1	119	HA25	25	LAD14	85	RCA6	96
CAS2	118	HA26	24	LAD15	87	RCA7	95
CAS3	117	HA27	23	LAD16	52	RCA8	94
CBLNK/VBLNK	109	HA28	22	LAD17	54	RCA9	93
CLKIN	15	HA29	21	LAD18	56	RCA10	92
CSYNC/HBLNK	110	HA30	20	LAD19	59	RCA11	91
DDIN	114	HA31	19	LAD20	62	RCA12	90
DDOUT	115	HBS0	50	LAD21	64	RESET	5
EMU0	2	HBS1	49	LAD22	66	SCLK	89
EMU1	1	HBS2	48	LAD23	68	SF	105
EMU2	3	HBS3	47	LAD24	72	SIZE16	10
EMU3	130	HCS	18	LAD25	74	TR/QE	106
GI	4	HDST	127	LAD26	76	V <sub>cc</sub>	13
HA5	46	HOE	126	LAD27	78	V <sub>cc</sub>	60
HA6	45	HINT	129	LAD28	81	V <sub>cc</sub>	79
HA7	44	HRDY	128	LAD29	84	V <sub>cc</sub>	98
HA8	43	HREAD	17	LAD30	86	V <sub>cc</sub>	121
HA9	42	HSYNC	108	LAD31	88	VCLK	14
HA10	41	HWRITE	16	LCLK1	131	V <sub>SS</sub>	34
HA11	40	LAD0	51	LCLK2	132	V <sub>SS</sub>	57
HA12	39	LAD1	53	LINT1	7	V <sub>SS</sub>	69
HA13	38	LAD2	55	LINT2	6	V <sub>SS</sub>	70
HA14	37	LAD3	58	LRDY	12	V <sub>SS</sub>	82
HA15	36	LAD4	61	PGMD	11	V <sub>SS</sub>	99
HA16	35	LAD5	<b>63</b> .	R0	124	V <sub>SS</sub>	111
HA17	33	LAD6	65	R1	125	V <sub>SS</sub>	112
HA18	32	LAD7	67	RAS	122	V <sub>SS</sub>	123
HA19	31	LAD8	71	RCA0	104	VSYNC	107
HA20	30	LAD9	73	RCA1	103	WE	116

Table 2–4. Alphabetical List of TMS34020 Pin Assignments (132-Pin QFP)

## 2.2 The TMS34020's Major Interfaces

The TMS34020's pins are divided among several interfaces:

Name	Pins
Local-memory interface	39 pins
DRAM/VRAM control interface	22 pins
Multiprocessor interface	3 pins
Host interface	38 pins
Video interface	6 pins
Emulation interface	4 pins
System control	6 pins
Power and ground	26 pins (PGA) 14 pins (QFP)

Figure 2–3 associates the TMS34020's pins with its major interfaces.





## 2.3 Signal Descriptions

This section describes the TMS34020 signals. Table 2–5 associates the signals with the proper interfaces, provides brief descriptions, and references page numbers on which you can find detailed signal descriptions. Subsections 2.3.1 through 2.3.7 provide details concerning the individual groups.

Table 2–5.	TMS34020 Pin	Descriptions
------------	--------------	--------------

	Signal Name	I/O	Description	Refer to Page
	ALTCH	0	Address latch	2-11
ace	BUSFLT	I	Bus fault	2-11
terfa	DDIN	0	Data bus direction input enable	2-11
y Ini	DDOUT	0	Data bus direction output enable	2-11
lemor	LAD0—LAD31	I/O	Local address/data multiplexed sig- nals	2-11
al-M	LRDY	I	Local ready	2-11
Loc	PGMD	l	Page mode	2-11
—	SIZE16	1 -	Bus size	2-11
	CAMD	1	Column-address mode	2-12
trol	CAS0-CAS3	0	Column-address strobes	2-12
Con	RAS	0	Row-address strobe	2-12
RAM (	RCA0-RCA12	0	Thirteen multiplexed row-/column- address signals	2-12
NW	SF	0	Special function pin	2-12
RA	TR/QE	0	Transfer/output enable	2-12
	WE	0	Write enable	2-12
cessor	ចា	I	Bus-grant input	2-13
Multiproc	R0, R1	0	Bus-request code	2-13
	**			

## Table 2–5. TMS34020 Pin Descriptions (Continued)

	Signal Name	I/O	Description	Refer to Page
	HA5HA31	Ι	Host-address input signals	2-13
	HBS0—HBS3	I	Host byte selects	2-13
9	HCS	I	Host chip select	2-13
erfa	HDST	0	Host data-latch strobe	2-13
Inte	HINT	0	Host interrupt	2-13
lost	HOE	0	Host data-latch output enable	2-13
Ξ	HRDY	0	Host ready	2-13
	HREAD	Ι	Host read strobe	2-13
	HWRITE	1	Host write strobe	2-13
Ø	CBLNK/VBLNK	0	Composite blanking or vertical blank- ing	2-15
terface	CSYNC/HBLNK	I/O	Composite sync or horizontal blank- ing	2-15
u o	HSYNC	I/O	Horizontal sync	2-15
'ide	SCLK	I	Serial data clock	2-15
>	VCLK	1	Video clock	2-15
	VSYNC	I/O	Vertical sync	2-15
ation	EM0-EMU2	I	Emulation pins 0-2	Appendix A
Emul	EMU3	0	Emulation pin 3	Appendix A
	CLKIN	1	Clock input	2-16
Eel	LCLK1, LCLK2	0	Local output clocks	2-16
Son	LINT1, LINT2	1	Local interrupt requests	2-16
	RESET	1	System reset	2-16
ver troi	V <sub>cc</sub>	I	Nominal 5-volt power supply inputs (5 pins on QFP, 9 pins on PGA)	2-16
Po Po Po Po Po Po Po Po Po Po Po Po Po P	V <sub>SS</sub>	I	Electrical ground inputs (9 pins pins on QFP, 17 pins on PGA)	2-16

## 2.3.1 Local-Memory Interface Signals

The TMS34020 communicates with external memory and with external memory-mapped I/O devices through its local-memory interface. This interface's signals are also used in conjunction with the DRAM and VRAM interface.

Signal Name	I/O	Description
ALTCH	0	Address latch. The high-to-low transition of ALTCH can be used to capture the ad- dress and status present on the LAD bus. A transparent latch (such as a 74ALS373) will maintain the current address and status as long as ALTCH remains low.
BUSFLT	I	<b>Bus fault.</b> External logic asserts BUSFLT high to the TMS34020 to indicate that an error or fault has occurred on the current bus cycle. BUSFLT is also used with LRDY to generate bus-cycle retries so that the entire memory address is presented again on the LAD pins.
DDIN	0	<b>Data bus direction, input enable.</b> This active-high output is used to drive the active-high input enables on bidirectional transceivers (such as the 74ALS623). The transceivers buffer data input and output on the LAD0—LAD31 pins when the TMS34020 is interfaced to several memories.
DDOUT	0	<b>Data bus direction, output enable</b> . This active-low signal drives the active-low output enables on bidirectional transceivers (such as the 74ALS623). The transceivers buffer data input and output on the LAD0—LAD31 pins.
LAD0—LAD31	I/O	<b>Multiplexed local address/data bus.</b> At the beginning of a memory cycle, the word address is output on LAD4—LAD31, and the cycle status is output on LAD0—LAD3. After the address is presented, LAD0—LAD31 are used for transferring data within the TMS34020 system. LAD0 is the LSB and LAD31 is the MSB.
LRDY	I	<b>Local ready.</b> External circuitry drives this signal low to stop the TMS34020 from completing a local-memory cycle it has initiated. While LRDY remains low, the TMS34020 will wait, unless the TMS34020 is given a retry request (through the BUSFLT signal). Wait states are generated in increments of one full LCLK1 cycle. LRDY can be driven low to extend local-memory read and write cycles, VRAM serial-data-register transfer cycles, and DRAM-refresh cycles. During internal cycles, the TMS34020 ignores LRDY.
PGMD	I	<b>Page mode.</b> The memory decode logic asserts this signal low if the currently ad- dressed memory supports burst (page mode) accesses. Burst accesses occur as a series of CAS cycles for a single RAS cycle to memory.
SIZE16	ł	<b>Bus size.</b> The memory decode logic may pull this signal low if the currently ad- dressed memory or port supports only 16-bit transfers. SIZE16 can also be used to determine which 16 bits of the data bus are used for a data transfer.

Table 2–6 lists the bus cycle completion conditions controlled by LRDY and BUSFLT.

## Table 2–6. Bus-Cycle Completion Conditions

Completion Condition	BUSFLT	LRDY	
Wait	0	0	
Successful transfer	0	1	
Retry	1	0	
Bus fault	1	1	

## 2.3.2 DRAM and VRAM Control Signals

Signal Name	I/O	Description
CAMD	1	<b>Column-address mode.</b> This input dynamically shifts the column address on the RCA0—RCA12 bus to allow the mixing of DRAM and VRAM address matrices using the same multiplexed address RCA0—RCA12 signals.
CAS0—CAS3	0	<b>Column-address strobes</b> . The $\overline{CAS}$ outputs drive the $\overline{CAS}$ inputs of DRAMs and VRAMs. These signals strobe the column address on RCA0—RCA12 to the memory. The four $\overline{CAS}$ strobes provide byte write access to the memory.
RAS	.0	<b>Row-address strobe</b> . The $\overline{RAS}$ output drives the $\overline{RAS}$ inputs of DRAMs and VRAMs. The high-to-low transition on this signal strobes the row address on RCA0—RCA12 to memory.
RCA0—RCA12	0	<b>Multiplexed row-address/column-address signals.</b> At the beginning of a memory access cycle, the row address for DRAMs is present on RCA0—RCA12. The row address contains the most significant address bits for the memory. As the cycle progresses, the memory column address is placed on RCA0—RCA12. The addresses that are actually output during row and column times depend on the memory configuration (set by RCM0 and RCM1 in the CONFIG register) and the state of CAMD during the access. RCA0 is the LSB and RCA12 is the MSB.
SF	0	<b>Special-function pin.</b> This is the special-function signal to 1M VRAMs. This signal allows the use of block write, load write mask, load color mask, and write using write mask. This signal is also used to differentiate instructions and addresses for the coprocessor as part of the coprocessor interface.
TR/QE	0	<b>Transfer/output enable</b> . This signal drives the $\overline{TR}/\overline{QE}$ input of VRAMs. During a local-memory read cycle, $\overline{TR}/\overline{QE}$ functions as an active-low output enable to gate data from memory to LAD0—LAD31. During special VRAM function cycles, $\overline{TR}/\overline{QE}$ controls the type of cycle that is performed.
WE	0	<b>Write enable</b> . The active low $\overline{WE}$ output drives the $\overline{WE}$ inputs of DRAMs and VRAMs. $\overline{WE}$ can also be used as the active-low write enable to static memories and other devices connected to the TMS34020 local interface. During a local-memory read cycle, $\overline{WE}$ remains inactive high while $\overline{CAS}$ is strobed active low. During a local-memory write cycle, $\overline{WE}$ is strobed active low before $\overline{CAS}$ is. During VRAM serial-data-register transfer cycles, the state of $\overline{WE}$ at the falling edge of $\overline{RAS}$ controls the direction of the transfer.

## 2.3.3 Multiprocessor Interface Signals

The multiprocessor interface allows multiple TMS34020s to share the same local memory by providing a request/grant protocol for devices that want to access shared memory.

Signal Name	I/O	Description
GI	I	<b>Bus grant input</b> . External bus arbitration logic drives $\overline{GI}$ low to enable the TMS34020 to gain access to the local-memory bus. The TMS34020 must release the bus if $\overline{GI}$ is high so that another device can access the bus.
R1, R0	0	Bus request and control. These two signals indicate a request for use of the bus in a multiprocessor system; they are decoded as shown below.
		<ul> <li>R1 R0 Bus Request Type         <ul> <li>0 0 High-priority bus request</li> <li>0 1 Bus cycle termination</li> <li>1 0 Low-priority bus request</li> <li>1 1 No bus request pending</li> </ul> </li> <li>A high-priority bus request provides for VRAM serial-data-register transfer cycles, DRAM refresh (when 12 or more refresh cycles are pending), or a host-initiated access. The external arbitration logic should grant this request as soon as possible by asserting GI low.</li> </ul>
		A low-priority bus request is used to provide for CPU-requested access and DRAM refresh (when less than 12 refresh cycles are pending).
		Bus cycle termination status is provided so that the arbitration logic can determine that the device currently accessing the bus is completing an access and other devices may compete for the next bus cycle. A <i>no bus request pending</i> status is output when the currently active device does not require the bus on subsequent cycles.

## 2.3.4 Host Interface Signals

The host interface signals are used for communication between the TMS34020 and a host processor. Signals input on these pins are assumed to be asynchronous with respect to the local clocks (LCLK1 and LCLK2). Signals output on these pins are synchronized only when responses are dependent on memory cycles that must be generated by the TMS34020.

The host interface allows the TMS34020's memory to be mapped into a host processor's address space. The TMS34020 can act as a DRAM controller for a host processor. The address of the required access is input to the TMS34020, and data is transferred through external transceivers.

1/0	Description
I	<b>27 host-address input signals.</b> A host can access a long-word by placing the ad- dress on these lines. HA5—HA31 correspond to the LAD5—LAD31 signals that output the address to the local memory.
I	<b>4 host byte selects.</b> The byte selects identify which bytes within the long-word are being selected.
1	<b>Host chip select.</b> A host drives this signal low to latch the current host address pre- sent on HA5—HA31 and the host byte selects on HBS0—HBS3. This signal also enables host access cycles to the TMS34020 I/O registers or local memory. During the low-to-high transition of RESET, the level on the HCS input determines whether the TMS34020 is halted (HCS is high for host-present mode) or whether it begins executing its reset service routine (HCS is low for self-bootstrap mode).
0	<b>Host data strobe.</b> The rising edge of this signal latches data from the TMS34020 local address space to the external transceivers on host read accesses. It can be used in conjunction with HRDY to indicate that data is valid in the external transceivers.
0	<b>Host interrupt</b> . This signal allows the TMS34020 to interrupt a host by setting the INTOUT bit in the HSTCTLL I/O register. This signal can also be used to interrupt the host if a BUSFLT or RETRY occurs due to a host access cycle.
0	<b>Host-data output enable.</b> This signal enables data from the external transceivers to the TMS34020 local address space on host write cycles. HOE can be used in conjunction with HRDY to indicate data has been written to memory from the external transceivers.
0	<b>Host ready</b> . This signal is normally low and goes high to indicate that the TMS34020 is ready to complete a host-initiated read or write cycle. A host can use HRDY logically combined with HDST and HOE to determine when the local bus access cycles have completed.
1	Host read strobe. This signal is driven low during a read request from a host proc- essor. This notifies the TMS34020 that the host is requesting access to local memory or to the I/O registers. HREAD should not be asserted at the same time that HWRITE is asserted.
I	<b>Host write strobe</b> . This signal is driven low to indicate a write request by a host processor. This notifies the TMS34020 that a write request is pending. The rising edge of HWRITE is used to indicate that the data provided by the host in the external data transceivers can be written. HWRITE should not be asserted at the same time HREAD is asserted.
	I/O I I I O O O I I

## 2.3.5 Video Interface Signals

Signal Name	I/O	Description
CBLNK/VBLNK	0	<b>Composite blanking/vertical blanking.</b> You can program this signal to select one of two blanking functions:
		Composite blanking for blanking the display during both horizontal- and verti- cal- retrace periods in <i>composite-sync video mode</i> .
		Vertical blanking for blanking the display during vertical retrace in separate- sync video mode.
		Immediately following reset, this signal is configured as CBLNK output.
CSYNC/HBLNK	I/O	<b>Composite sync/horizontal blanking</b> . You can program this signal to select one of two functions:
		Composite sync (either input or output as set by a control bit in the DPYCTL register) in composite-sync video mode:
		As an <b>input</b> , <u>CSYNC</u> synchronizes the TMS34020 video-control registers to externally generated horizontal-sync pulses. The actual synchronization can be programmed to begin at any VCLK cycle; this allows for any external pipelining of signals. <u>CSYNC</u> extracts <u>HSYNC</u> and <u>VSYNC</u> from externally generated horizontal-sync pulses.
		As an output, CSYNC is the active-low composite-sync pulse generated by the TMS34020's on-chip video timers.
		Horizontal blank (output only) for blanking the display during horizontal retrace in separate-sync video mode.
		Immediately following reset, this signal is configured as a CSYNC input.
HSYNC	I/O	Horizontal sync. HSYNC is the horizontal-sync signal that controls external video circuitry. You can program this signal to be either an input or an output by modifying a control bit in the DPYCTL register.
		As an <b>output</b> , HSYNC is the active-low horizontal-sync signal generated by the TMS34020's on-chip video timers.
		As an <b>input</b> , HSYNC synchronizes the TMS34020 video-control registers to ex- ternally generated horizontal-sync pulses. The actual synchronization can be programmed to begin at any VCLK cycle; this allows for any external pipelining of signals.
		Immediately following reset, HSYNC is configured as an input.
SCLK	1	Serial data clock. This signal is the same as the signal that drives VRAM serial-da- ta registers. This allows the TMS34020 to track the VRAM serial-data-register count, providing serial-register-transfer midline-reload cycles. (SCLK may be asynchronous to VCLK; however, it typically has a frequency that is a multiple of the VCLK frequency.)
VCLK	1	Video clock. This clock is a derivative of the video system's dotclock and is used internally to drive the video timing logic.

Signal Name	I/O	Description
VSYNC	I/O	<b>Vertical sync.</b> VSYNC is the vertical-sync signal that controls external video cir- cuitry. You can program this signal to be either an input or an output by modifying a control bit in the DPYCTL register.
		As an output, VSYNC is the active-low vertical-sync signal generated by the TMS34020's on-chip video timers.
		As an input, VSYNC synchronizes the TMS34020 video-control registers to ex- ternally generated vertical-sync pulses. The actual synchronization can be pro- grammed to begin at any horizontal line; this allows for any external pipelining of signals.
		Immediately following reset, VSYNC is configured as an input.

## 2.3.6 System Control Signals

Signal Name	I/O	Description
CLKIN	l	<b>Clock input</b> . This system input clock is used to generate the LCLK1 and LCLK2 outputs, to which all processor functions in the TMS34020 are synchronous. A separate asynchronous input clock (VCLK) controls the video timing and video registers.
LCLK1, LCLK2	0	<b>Local output clocks</b> . These two clocks are 90 degrees out of phase with each other. They provide convenient synchronous control of external circuitry to the internal timing. All signals output from the TMS34020 (except the CRT timing signals) are synchronous to these clocks.
LINT1, LINT2	l	<b>Local interrupt requests.</b> Interrupts from external devices are transmitted to the TMS34020 on LINT1 and LINT2. Each local interrupt signal activates the request for one of two interrupt request levels. An external device generates an interrupt request by driving the appropriate interrupt request pin to its active-low state. These signals can be applied asynchronously to the TMS34020 as they are synchronized internally before use. The signal should remain low until it is recognized by the TMS34020.
RESET	I	System reset. RESET is normally high. During normal operation, RESET is driven low to reset the TMS34020. When RESET is asserted low, the TMS34020's internal registers are set to an initial known state, all output pins are driven to inactive levels, and all bidirectional pins are driven to a high-impedance state. The TMS34020's be- havior following reset depends on the level of the HCS input just before the low-to- high transition of RESET. If HCS is low, the TMS34020 begins executing the instruc- tions pointed to by the reset vector. If HCS is high, the TMS34020 is halted until a host processor writes a 0 to the HLT bit in the HSTCTLL register.

## 2.3.7 Power Signals

Signal Name	I/O	Description
V <sub>CC</sub>	l ···	Nominal 5-volt power supply inputs (5 pins for the QFP, 9 pins for the PGA)
V <sub>SS</sub>		Electrical ground inputs (9 pins for the QFP, 17 pins for the PGA)

**Note:** For proper TMS34020 operation, all these signals must be connected externally.

## **Memory Organization and Data Structures**

Much of the TMS34020's power derives from its flexible memory access. Several memory organization features are tailored specifically for graphics applications:

- A large memory space supports a variety of display resolutions.
- You can access memory locations with linear or XY addresses.
- The TMS34020 provides hardware support for several data structures:
  - Fields are configurable data structures. A field can begin and end at any bit address and can be 1 to 32 bits long.
  - As used by the TMS34020, bytes are a special type of field; byte length is fixed at 8 bits.
  - Pixels are configurable data structures; pixel length can be any power of 2 in the range of 1 to 32 bits.
  - **Pixel arrays** are 2-dimensional, rectangular blocks of pixels.

Additionally, the TMS34020 can be addressed in little-endian or big-endian mode, and provides a system stack. Unless explicitly stated otherwise, all discussions refer to little-endian addressing.

	Sec	tion	Page
Memory organization sections illustrate the TMS34020's memory map and general addressing schemes	3.1 3.2	Memory Map	. 3-2 . 3-3
Graphics-specific features include hardware-supported data structures and the ability to use XY addressing.	3.3 3.4 3.5 3.6	Fields Pixels XY Addressing Converting an XY Address to	. 3-5 . 3-10 . 3-14
	3.7	a Linear Address	. 3-15 . 3-18
Additional features include endian modes and stack operations.	3.8 3.9	Big-Endian and Little-Endian Addressing Stacks	3-20 . 3-26

## 3.1 Memory Map

Figure 3–1 illustrates the TMS34020 memory map.

#### Figure 3–1. TMS34020 Memory Map

bit 2 <sup>32</sup> – 1 (last bit in memory)					
FFFF FFE0h	,				
	34 words	Interrupt vectors and trap vectors			
FFFF FBC0h					
FFFF FBA0h	222 words	Reserved for interrupt vectors and extended trap vectors			
FFFF E000h					
FFFF DFE0h	32,512 words	General use and extended trap vectors			
FFF0 0000h					
FFEF FFE0h	2 <sup>25</sup> – 33,024 words (33,521,408 words)	General use			
C000 2000h		_			
C000 1FE0h	224 words	Reserved for I/O registers			
C000 0400h					
C000 03E0h	32 words	I/O registers			
C000 0000h					
BFFF FFE0h	$3 \times 2^{25} - 32$ K words (100,630,528 words)	General use			
0010 0000h	(//////////////////////////////////////				
000F FFE0h	32,768 words	General use and extended trap vectors			
0000 0000h L		bit 0			
(first bit in memory)					

Memory is logically organized as 4 gigabits, but is physically accessed 32 bits at a time. Figure 3–1 shows locations as long (32-bit) words, identified by 32-bit addresses. Word addresses range from 0000 0000h to FFFF FFE0h (bit address 0000 0000h is the rightmost bit in the word at the bottom of Figure 3–1, and bit address FFFF FFFFh is the leftmost bit in the word at the top.) Reading or writing to an address in the range C000 0000h to C000 03E0h accesses an internal I/O register. (An external memory cycle is also generated on accesses to these locations, allowing the I/O registers to be shadow mapped in external memory.) Reading or writing to any address outside this range accesses external memory (or a memory-mapped device).

As Figure 3–1 shows, memory is divided into several regions:

#### General use

Address ranges 0h—BFFF FFE0h and C000 2000h—FFFF DFE0h are for general use (executable code, data tables, etc.).

#### □ I/O registers

Addresses C000 0000h—C000 03E0h are reserved for the internal I/O registers. Chapter 4 discusses the I/O registers; it contains a map of this memory area that associates each I/O register with the appropriate address.

#### □ Interrupt, reset, and trap vectors

Addresses FFFF FBC0h—FFFF FFE0h are reserved for 34 interrupt, reset, and trap vectors. A vector is a 32-bit address that points to the starting location in memory of the appropriate interrupt, reset, or trap service routine. Chapter 6 contains more information about interrupts and traps.

## Reserved memory

Addresses C000 0400h—C000 1FE0h are reserved for future expansion of the I/O registers. Addresses FFFF E000h—FFFF FBA0h are reserved for future expansion of the interrupt vectors.

## 3.2 Memory Addressing

The TMS34020 is a bit-addressable machine with a 32-bit memory address. The total memory capacity is 4 gigabits (512 Mbytes). Memory is accessed as a continuously addressable string of bits; each 32-bit address points to an individual bit within memory. Bit addresses range from 0000 0000h to FFFF FFFFh.

Figure 3–2 illustrates the TMS34020's logical memory structure.





Figure 3–3 illustrates physical memory organization.





The TMS34020 communicates with memory over a 32-bit address/data bus (LAD0—LAD31) and always reads a complete long (32-bit) word from memory. Writes to memory may be 8-, 16-, 24-, or 32-bit values through the use of the TMS34020's CAS (byte) strobes.

A long-word accessed during a memory cycle always begins on a 32-bit boundary; thus, the 5 LSBs of the 32-bit starting address of the word are always 0s. Bits within a word are numbered from 0 to 31; bit 31 is the MSB and bit 0 is the LSB. A word is identified by the address of its LSB. The LSB of a memory word is depicted as the rightmost bit in the word.

The 4 LSBs of the 32-bit logical address in Figure 3–3 do not appear on the LAD bus. Bit 4 is output for use with 16-bit memory devices only. When the TMS34020 accesses a field that does not begin and end on long-word boundaries, these 5 LSBs are used internally to identify a bit boundary within an accessed long-word.

Internal logic automatically performs the bit alignment and masking necessary to extract a field from physical memory; this is completely transparent to software. Similarly, inserting a field into memory may require a series of read and write cycles, accompanied by internal masking and shifting of data to properly align the data structure within memory. The memory control logic performs these tasks automatically.

## 3.3 Fields

The TMS34020 supports 2 software-configurable field types, **field 0** and **field 1**. A field is defined by 2 parameters:

- Starting address. A field's starting address is the address of the field's LSB. A field can begin at an arbitrary bit address in memory. When a field is moved from memory to a general-purpose register, the field is right-justified within the register; that is, the field's LSB coincides with the register's rightmost bit (bit 0). The register bits to the left of the field are all 1s or all 0s, depending on the values of both the appropriate FE (field extension) status bit and the field's sign bit (MSB). If FE=1 the field is sign-extended; if FE=0, the field is zero-extended.
- ➡ Field size. Field size can range from 1 to 32 bits. The lengths of fields 0 and 1 are defined by two 5-bit fields in the status register, FS0 and FS1.

Figure 3–4 identifies the status bits that control the size and extension of field 0 and field 1. Table 3–1 shows how the field size is encoded in FS0 and FS1.

Figure 3–4. Status Bits That Control Field 0 and Field 1



5 FS Bits	Field Size						
00001	1	01001	9	10001	17	11001	25
00010	2	01010	10	10010	18	11010	26
00011	3	01011	11	10011	19	11011	27
00100	4	01100	12	10100	20	11100	28
00101	5	01101	13	10101	21	11101	29
00110	6	01110	14	10110	22	11110	30
00111	7.	01111	15	10111	23	. 11111	31
01000	8	10000	16	11000	24	00000	32

Table 3–1.	Decoding the	Field-Size Bits	in the Status	Register
------------	--------------	-----------------	---------------	----------

Figure 3–5 illustrates a field in memory. In this example, the field straddles the boundary between words n and n+1 in memory. Field extraction and insertion is performed by on-chip hardware:

**To move the field to a general-purpose register, the TMS34020 extracts the field from memory by reading word** n and word n+1 in separate cycles.

To move the field **from** a general-purpose register, the TMS34020 inserts the field into memory by reading and writing word n and reading and writing word n+1.

The memory operations necessary to insert or extract a field are performed automatically by special hardware and are transparent to software.

Figure 3–5. Field Storage in External Memory



In Figure 3–5, word *n* is pointed to by a 27-bit physical address, output by the TMS34020 to memory. This 27-bit address corresponds to bits 5–31 of the field's 32-bit logical address. The 5 LSBs of the logical address point to the beginning of the field within word *n*.

The number of memory cycles required to extract or insert a field depends on how the field is aligned within memory. Field manipulation is more rapid when fields are stored in memory so that they do not cross word boundaries. Figure 3–6 illustrates various cases of alignment and nonalignment of fields to word boundaries in memory. Given a field starting address and field length, the memory controller will recognize the specified field alignment as one of the 6 cases in Figure 3–6. Field extraction and field insertion are performed in a manner that requires the minimum number of memory cycles.

- Cases A1—A4. The field begins and ends on byte boundaries within a single word.
  - In Case A1, the field is 8 bits wide and the starting address is aligned to a byte boundary within a word.
  - In Case A2, the field is 16 bits wide and the starting address is aligned to the first, second, or third byte boundary within a word.
  - In Case A3, the field is 24 bits wide and is aligned to the first or second byte boundary in a word.
  - In Case A4, the field is 32 bits wide and is word-aligned.

For Cases A1—A4, a field extraction requires a single read cycle, and a field insertion requires a single write cycle.

Figure 3–6. Field Alignment in Memory



- □ Case B. The field does not straddle a word boundary and does not begin and end on byte boundaries (that is, either it is not aligned on a byte boundary, or it is aligned on a byte boundary but is not a multiple of 8 bits). A field extraction requires a single read cycle. A field insertion requires the following sequence of memory cycles:
  - Read word n
  - Write word *n*
- **Case C.** The field straddles the boundary between 2 words and begins and ends on byte boundaries. A field extraction requires the following sequence of memory cycles:
  - Read word n
  - Read word n+1

A field insertion requires the following sequence or memory cycles:

- Write word *n*
- Write word n+1
- □ Case D. The field straddles the boundary between 2 words. The field address is byte aligned, but the end of the field does not coincide with the end of a byte. A field extraction requires the following sequence of memory cycles:
  - Read word n
  - Read word n+1

A field insertion requires the following sequence of memory cycles:

- Write word *n*
- Read word n+1
- Write word n+1
- Case E. The field straddles the boundary between 2 words. The end of the field is byte aligned, but the start is not. A field extraction requires the following sequence of memory cycles:
  - Read word n
  - Read word n+1

A field insertion requires the following sequence of memory cycles:

- Read word n
- Write word n
- Write word n+1
- ❑ Case F. The field straddles the boundary between 2 words and neither the start nor the end of the field is aligned to a byte boundary. A field extraction requires the following sequence of memory cycles:
  - Read word n
  - Read word n+1

A field insertion requires the following sequence of memory cycles:

- Read word n
- Write word *n*
- Read word n+1
- Write word n+1

A field insertion modifies only the portion of a word that lies within a field. The TMS34020 memory controller must perform a read-modify-write operation when a field that does not begin and end on byte boundaries is written to memory. The memory controller uses these 2 parameters (address LSBs and

field size) to produce a mask that identifies the bits in the word corresponding to the field. Hardware uses the mask to perform the read-modify-write cycle. The TMS34020's local memory control logic automatically generates the mask and executes the read-modify-write operation; this is transparent to the software.

Figure 3–7 shows an example of inserting a 14-bit field stored in a register to logical address 0000 0007h.

Figure 3–7. Field Insertion

(a) Field to be inserted

31 30 29 28 27 26 25 24 23 22 21 20 19 18 17 16 15 14 13 12 11 10 9 8 6 X XXX F F F F F F F F F F F F F F

(b) Rotate to align to bit 7

31 30 29 28 27 26 25 24 23 22 21 20 19 18 17 16 15 14 13 12 11 10 9 7 8 6 XXXXXXXXXXXXFFFFFF F FF FF FF F F X X X X X Х Х

(c) Initial destination data

(d) Mask generated

31 30 29 28 27 26 25 24 23 22 21 20 19 18 17 16 15 14 13 12 11 10 9 8 7 6 5 0 0 0 0 0 0 0 1 1 1 1 1 1 1 1 1 1 0 0 0 0 0 0 0 0 0 0 1 1 1 1

(e) Field destination data

31	30	29	28	27	26	25	24	23	22	21	20	19	18	17	16	15	14	13	12	11	10	9	8	7	6	5	4	З	2	1	0
Α	Α	А	Α	Α	Α	Α	Α	Α	Α	Α	F	F	F	F	F	F	F	F	F	F	F	F	F	F	Α	Α	Α	Α	Α	Α	Α

- (a) The field to be inserted is shown right-justified in the designated general-purpose register.
- (b) The CPU has rotated the field to align it with the destination in memory.
- (c) The TMS34020 reads the original word from the destination in memory.
- (d) The mask is generated to designate the bits to be modified.
- (e) The field is inserted into the word from memory, and the result is written back to the destination address in memory.

In the more complex case in which a field straddles a word boundary in memory, the portion of the field lying within each word is inserted into that word using the methods described above.

## 3.4 Pixels

The term **pixel** has two meanings in the context of a TMS34020-based graphics system. Outside the TMS34020, a physical pixel is a picture element on a display surface. Inside the TMS34020, a logical pixel is a software configurable data structure supported by the TMS34020 instruction set. The logical pixel data structure in TMS34020 memory contains the information needed to specify the attributes of a picture element visible on a screen. The information for a horizontal line of pixels on a screen is usually stored in consecutive words in memory.

#### 3.4.1 Pixels in Memory

Within TMS34020 memory, the pixel data structure is defined by 2 parameters:

its starting address (the address of the pixel's LSB) and

the pixel size (the number of bits per pixel).

The PSIZE register defines the current pixel size. A pixel can be 1, 2, 4, 8, 16, or 32 bits long. The TMS34020 treats pixels as a special case of a field in which the field size is constrained to be a power of 2. Unlike other memory fields, pixels do not cross long-word boundaries within memory; they are aligned within memory so that a memory word contains an integral number of pixels. For example, a 2-bit pixel should begin at a bit address whose LSB is 0, a 4-bit pixel should begin at a bit address whose of 0.

When a pixel is moved from memory to a general-purpose register, the pixel is right-justified within the register. That is, the pixel's LSB coincides with the rightmost bit (bit 0) of the register. Register bits to the left of the pixel are loaded with 0s.

Figure 3–8 illustrates pixel storage in memory. The pixel is located within the word pointed to by the 27-bit physical address corresponding to bits 5–31 of the pixel's 32-bit logical address. The 5 LSBs of the logical address specify the displacement of the pixel within the word. When the pixel length is less than 32, each word contains 2 or more pixels.



*Figure 3–8. Pixel Storage in External Memory* 

On-chip hardware performs pixel extraction and insertion in a manner that requires the minimum number of memory cycles. (The operations are transparent to software.) Two memory cycles (a read followed by a write) are always required to insert a pixel of less than 8 bits. Inserting an 8-, 16-, or 32-bit pixel requires a single write cycle (unless plane masking is enabled). Extracting a pixel (1 to 32 bits) requires a single read cycle.

#### 3.4.2 Pixels on the Screen

Figure 3–9 illustrates the mapping of pixels from memory to a display screen. The screen-refresh function outputs pixels in the sequence of ascending pixel addresses. However, the electron beam sweeps from the left edge of the screen to the right edge during each horizontal scan interval, so pixels appear on the screen in the opposite order of their representation in memory. That is, the least significant pixel (in terms of bit address) appears on the left, and the most significant pixel appears on the right.

Figure 3–9. Mapping of Pixels to a Monitor Screen



The TMS34020 allows you to identify a pixel in terms of its XY coordinates on the screen or in terms of the address of the logical pixel in memory. These 2 methods are called **XY addressing** and **linear addressing**, respectively.

When you use XY addressing, you can select the origin to lie in either the upper left or lower left corner of the screen. The DPYST and DINC registers control the origin's position. Figure 3–10 (*a*) illustrates the default coordinate system in which the origin of the 2 coordinate axes is located in the upper left corner of the screen. In this system, DPYST contains the address of the pixel at the upper left of the screen, and DINC contains the display pitch. Figure 3–10 (*b*) shows the alternate coordinate system in which the origin is located in the lower left corner of the screen. In this case, DPYST contains the address of the pixel at the lower left of the screen, and DINC contains the 2 compliment of the pixel at the lower left of the screen, and DINC contains the 2 compliment of the display pitch.


Using the default screen origin, Figure 3–11 illustrates the mapping of pixels from the memory to the screen. In Figure 3–11, horizontal movement represents travel in the X direction on the screen. Vertical movement represents travel in the Y direction. The depth of the buffer represents the pixel size. The on-screen memory contains the pixels that appear on the screen.

In Figure 3–11, the display memory is shown in terms of a *screen format*, rather than the memory format used in the memory map in Figure 3–1 (page 3-2). The screen format places the lowest pixel address at the upper left corner of the memory map. This is the same relative orientation in which the pixels appear on the screen. Compare this to the memory format shown in Figure 3–1, which places the lowest bit address at the lower right corner of the memory map. This convention is frequently used in industry to represent the relative location of addresses in memory. In this user's guide, assume the standard format is used unless the screen format is explicitly indicated.

Figure 3–11. Display Memory Dimensions



Figure 3–12 illustrates the mapping of XY coordinates to on-screen memory. For simplicity, assume that the screen origin coincides with the upper left corner of the display memory. *p* represents the X extent of the display memory; *n* represents the Y extent. Each box represents a pixel within the memory; the number in the box represents the pixel's memory location, relative to the beginning of the on-screen memory.



# Figure 3–12. Display Memory Coordinates

# 3.4.3 Display Pitch

**Display pitch** is the difference in memory addresses between 2 pixels that are vertically adjacent on the screen (one is directly above the other). In Figure 3–12, the pitch is calculated as p times the pixel size, where p is the X extent of the display memory. The pixel size is constrained to be a power of 2, so the multiply can be replaced by a shift operation. **Array pitch** is the difference in memory addresses of 2 vertically adjacent pixels in the array. If the array occupies a rectangular area on the screen, the array pitch is the same as the display pitch.

During a pixel operation such as a PIXBLT, the source array pitch, destination array pitch, and (if it is a masked PIXBLT) mask array pitch are defined in separate, dedicated hardware registers. This eases the transfer of pixel arrays between on-screen and off-screen memory, which may have different pitches.

As an example, here's how you would calculate the display pitch if the pixel size = 4 bits and the X extent of the pixel display = 1024 pixels:

**display pitch** = (1024 pixels per line)  $\times$  (4 bits per pixel) = 4096 (which is 2<sup>12</sup>)

Note that the TMS34020 does not require the display pitch to be a power of 2, as was the case for the TMS34010.

# 3.5 XY Addressing

The TMS34020 allows you to define pixel addresses in terms of 2-dimensional XY coordinates that correspond to screen locations. This is referred to as **XY** addressing. XY addressing has several benefits:

- TMS34020 software can be easily ported from one display configuration to another. System-dependent details, such as the number of bits per pixel and the X extent of the display memory, are transparent to the software. However, these are used by the machine to automatically convert the XY coordinates to the address of a pixel in memory.
- XY addressing allows you to think in terms of the high-level concept of XY coordinates rather than in terms of the machine-level mapping of pixels into memory.
- XY addressing facilitates operations such as window checking.

The TMS34020 supports XY coordinates in the range (-32768,-32768) to (+32767,+32767).

The TMS34010 did not support signed XY values, as the TMS34020 does.

Figure 3–13 illustrates the XY addressing format.





In Figure 3–13, a 32-bit general-purpose register contains an XY address. The X and Y components are treated as separate16-bit signed integers. The X component is right-justified within the 16 LSBs of the register. The Y component is right-justified within the 16 MSBs of the register.



# 3.6 Converting an XY Address to a Linear Address

For all instructions that use XY addressing, the TMS34020 automatically converts a pixel's XY address to a 32-bit logical address (linear address). The TMS34020 uses four parameters to perform XY-to-linear conversion:

logical pixel size	defined in the PSIZE register
pitch conversion factor	defined in the CONVSP, CONVDP, or CONVMP reg- ister
actual pitch	defined in the SPTCH, DPTCH, or MPTCH register if the conversion involves a pitch that is not a power of 2 or a sum two of powers of 2
offset	specifies the XY origin, defined in the OFFSET regis- ter

The TMS34020 uses the following formula to calculate the physical address associated with the XY address:

address =  $[(Y \times display pitch) + (X \times pixel size)] + offset$ 

Because the pixel size must always be a power of 2, the multiplication of the X component is performed using a shift operation. The method of calculating the Y component depends on the pitch value.

If the pitch is	This is how the Y value is calculated
a power of 2	The TMS34020 performs a left shift. The amount that the component is shifted is contained in the lower half of the appropriate CONVxP register.
two powers of 2	The operation is performed by summing 2 shifts of the Y value. The number of bits to be shifted during the first and second shifts are contained in the lower and upper halves of the appropriate CONVxP register, respectively. This adds a cycle to each conversion.
an arbitrary pitch (not a power of 2 and not two powers of 2)	The TMS34020 must perform a full 16-bit×32-bit multi- ply. In this case, the appropriate xPTCH register is used directly as the multiplier of the Y value. This adds about 12 cycles to each conversion.

The TMS34020 must perform one or more XY-to-linear conversions for the following instructions:

CVDXYL	FILL XY	PIXBLT L, XY
CVMXYL	FLINE	PIXBLT XY, L
CVSXYL	LINE	PIXBLT XY, XY
CVXYL	PIXBLT B, XY	PIXTs
DRAV		

The TMS34020 uses the pitch conversion factors in the CONVSP, CONVDP, and CONVMP registers to calculate the Y component of an address.

- **CONVSP** (source pitch) is used if the XY address points to a source pixel or pixel array.
- **CONVDP** (destination pitch) is used if the XY address points to a destination pixel or pixel array.
- **CONVMP** (mask pitch) is used for calculating the correct value when using a binary mask array.

Before executing an instruction that uses XY addressing, use the SETCSP, SETCDP, or SETCMP instruction to load the value for the appropriate register.

The TMS34020 uses the PSIZE value to determine the displacement of the X component.

The OFFSET register contains the linear memory address of the pixel located at coordinates (0,0). The TMS34020 uses the OFFSET register when translating XY coordinates into linear addresses. (Note that OFFSET does not control which region of the display memory is output to refresh the video screen—it is a virtual screen origin.) This allows the coordinate axes of the XY address to be translated to an arbitrary position in memory. The OFFSET register supports the use of window-relative addressing in which the XY coordinates are specified relative to coordinate offsets in the display memory. The window's position and size can be specified arbitrarily. You can use the CVXYL instruction to convert a new XY offset to a linear address. CVXYL converts an XY address to a linear address for the purpose of absolute memory addressing, or for using special features available to instructions that use linear addressing. Figure 3–14 illustrates the XY-to- linear conversion process.

3	31 16	150
(a) Original XY address	Y	X
(b) Extract 16 LSBs [ and sign-extend	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	s x
(c) Shift X left by log <sub>2</sub> (pixel size)	\$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$ \$	X 000
(d) Extract 16 MSBs from original XY address	Y	0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0
(e) Shift Y left by 16 and sign-extend	\$\$\$\$\$\$\$\$\$\$\$\$\$\$\$	Y
(f) Multiply by the pitch	SSS Y	0 0 0 0 0 0 0 0 0 0 0 0 0
(g) Add together the adjusted X and Y components	X	+ Y 0 0 0
(h) Add offset to result to get final memory [ address	memory	address

Figure 3–14. Conversion from XY Coordinates to Memory Address

Key: S represents the sign bit.

The example in Figure 3–14 corresponds to a pixel size of 8 bits and a pitch of 8,192.

- (a) shows the original XY address.
- (b) extracts the X component.
- (c) shifts the X component left by log<sub>2</sub>(pixel size). The result represents the product of the X component and the pixel size.
- (d) extracts the Y component.
- (e) left rotates the Y component by 16, sign-extends the Y component.
- (f) multiplies the shifted Y component by the pitch. (This may be a shift, the sum of 2 shifts, or an actual multiply depending on the pitch value. In this example, Y is shifted 12 bits to the left.)
- (g) adds the results of step (c) and (f) to form the displacement in memory of the pixel at (X,Y) from the pixel at the origin.
- (h) adds the offset to the result of (g), producing the the final memory address.

# 3.7 Pixel Arrays

A rectangular area of the screen that is DX pixels wide and DY pixels high is an example of a data structure called a 2-dimensional **pixel array**. An array may contain many pixels, but the TMS34020 can manipulate an array as a single structure. The TMS34020's instruction set includes a powerful set of raster operations, called PIXBLTs (pixel-block transfers), that manipulate pixel arrays on the screen and elsewhere in memory.

Figure 3–15 shows a pixel array that occupies a rectangular area in display memory. The pixels in each row are packed together into adjacent cells in the display memory. Rows don't usually occupy adjacent areas of memory; they're separated from each other by a constant displacement (the **array pitch**). The array pitch is the difference in memory addresses between 2 vertically adjacent pixels. In Figure 3–15, the array pitch equals the display pitch. The product of the array width (DX) and the pixel size must be less than or equal to the pitch.

#### Figure 3–15. Pixel Array



**Key:** DX = pixels per row of array

DY = pixels per column of array

A pixel array is specified in terms of its *width*, *height*, *pitch*, and *starting address*. The starting address is usually the address of the first pixel to be moved during a PIXBLT. The default starting address is simply the base address in the array—that is, the address of the pixel with the lowest address in the array.

In Figure 3–15, the XY origin is located in its default position at the upper left corner of the screen. The default starting address is the address of the pixel located in the upper left corner of the array. When a PIXBLT operation moves the pixels from a source pixel array to a destination array, the pixels in each row are moved in sequence from left to right, and the rows are moved in sequence from top to bottom.

Certain PIXBLT operations allow the starting pixel to be specified as the pixel in one of the other three corners of the array. This feature is provided so that when the source and destination arrays overlap, the appropriate starting corner can be selected to insure that no data is lost by being overwritten during PIXBLT execution. The order in which pixels in the array are moved can be altered to be from right to left and from bottom to top, as appropriate, to accommodate the change in the starting corner.

The starting address of a pixel array can be specified in terms of either the XY coordinates of the starting pixel (XY address), or the memory address of the starting pixel (linear address):

- An array whose starting location is specified as an XY address is referred to as an XY array. In this format, the starting location of the array is identified by the XY coordinates of the first pixel in the array.
- A pixel array whose starting location is specified as a memory address is referred to as a linear array. In this format, the location of the array is identified by the memory address of the first pixel in the array.

The XY array format has 2 advantages. First, the starting location of the array is given in system-independent Cartesian coordinates, rather than as a system-dependent memory address. Second, using XY addressing allows you to take advantage of the TMS34020's window checking facilities (which allows it to automatically detect an attempt to write a pixel inside or outside a defined area).

The linear format's main advantage is that it allows PIXBLTs to execute more quickly because it eliminates the need to translate from XY to linear format before accessing memory.

The general rules governing array pitch are

- When an array is specified in XY format, the pitch can be any multiple of the pixel size. However, PIXBLT operations performed on XY-format arrays are most efficient if the pitch is a power of 2.
- ❑ When an array is specified in linear format, the pitch must be a multiple of the pixel size. For the special case of a PIXBLT B,XY or PIXBLT B,L instruction, the source pitch may be any value. (Note that this corresponds to a pitch that is a multiple of the pixel size where the pixel size is 1.) This feature supports efficient use of memory by allowing adjacent rows of the source array to be packed together with no intervening gaps.

PIXBLTs are useful for moving arrays from one area of the screen to another; they can also be used to move arrays to the screen from other parts of memory, and vice versa. The pitch for the off-screen pixel array can be specified independently of the pitch for the on-screen array. This allows you to store off-screen data efficiently, regardless of the display pitch. On-screen objects can be defined as XY arrays but may be more efficiently stored as linear arrays in off-screen memory. The PIXBLT instructions support the transfer of a linear array to an XY array, and vice versa. PIXBLT instructions can also be used to rapidly move blocks of nonpixel data (for example, ASCII characters) from one memory location to another.

# 3.8 Big-Endian and Little-Endian Addressing

The TMS34020 allows you to address fields within memory in one of two ways—in **little-endian** or **big-endian mode**.

#### Note:

Unless specifically stated otherwise, all illustrations and discussions in this user's guide refer to little-endian mode.



The TMS34010 uses little-endian addressing only.

## 3.8.1 Selecting Big-Endian or Little-Endian Mode

The value of BEN[[CONFIG]] determines which endian mode the TMS34020 will use for addressing.

Figure 3–16. How BEN [[ CONFIG]] Determines the Endian Mode



You can set BEN in one of two ways:

- By writing to CONFIG via the host interface. In this case, the TMS34020 should be halted.
- By resetting the TMS34020. At the end of the reset routine, the TMS34020 copies the 4 LSBs of the reset vector into the 4 LSBs of the CONFIG register.

Program code should not change bits 0—2 of the CONFIG register; this could cause unpredictable behavior. To ensure that code doesn't accidentally change these bits, you can set CBP[[CONFIG]] to write-protect the LSbyte of CONFIG.





You can set CBP in one of three ways:

- By writing to CONFIG via the host interface.
- By resetting the TMS34020. At the end of the reset routine, the TMS34020 copies the 4 LSBs of the reset vector into the 4 LSBs of the CONFIG register.
- ❑ By allowing TMS34020 program code to write to this bit. If you do this, be sure that you don't alter bits 0—2 of CONFIG.

The only way to clear CBP is by resetting the TMS34020. CBP will remain cleared only if bit 3 of the reset vector is also 0 (if it's 1, then a reset will write a 1 back to CBP).

#### 3.8.2 How the TMS34020 Accesses Memory in These Modes

The following descriptions uses several terms and conventions:

- □ The terms *least significant bit* (LSB) and *most significant bit* (MSB) define specific bits within a 32-bit long-word, or specific bits within a field. These terms refer to the *arithmetic significance* of these bits.
- □ The following illustrations show the MSB of a field or long-word on the left side. The bits at the ends of the word are numbered 0 or 31, implying that the bits within a long-word are numbered 0 through 31. The 0 and 31 are positioned so that the implied number associated with each bit is its appropriate bit address within the long-word. The manner in which the TMS34020 addresses these bits differs in the two modes.
- Note also that there are two frames of reference for data; illustrations show data in
  - memory or an I/O register or
  - a general-purpose register.

The TMS34020 is primarily a little-endian processor; its method for placing data in a general-purpose register reflects this.

# Little-endian mode

This is the TMS34020's default mode. Figure 3–18 shows the same 32-bit hexadecimal value (01234 ABCDh) in a register and in a long-word in memory. Note that this illustration shows the MSB (bit 31) on the left side.



(b) Data in a register

(a) Data in memory at address 0000 1000h

NISP	<b></b>						LЭ
1	2	3	4	Α	В	С	D
31 MSB		L					LS
1	2	3	4	Δ	в	С	n

Figure 3–19 shows how the TMS34020 uses little-endian mode to access an 8-bit field that starts at bit 4 within the long-word. Notice that for little-endian mode, the field's bit address is determined by counting in from the LS (little) end of the long-word.

Figure 3–19. Addressing a Field in a Long-Word (Little-Endian)



If you want to move this 8-bit data field into a general-purpose register, you might execute the following instructions:

SETF 8,0,0 MOVE @00001004,A0,0

Figure 3–20 shows how the TMS34020 places this data into A0.

Figure 3–20. Moving a Field into a General-Purpose Register (Little-Endian)

	31 0	
ranistar AO		
Tegister AU		

Note that the data is right-aligned within the register, so that the LSB of the field coincides with the register's LSB.

# Big-endian mode

Figure 3–21 shows the same 32-bit hexadecimal value (01234ABCDh) in a register and in a long-word in memory. This illustration shows the MSB on the left side. The register's MSB is bit 31; the long-word's MSB is bit 0.

Figure 3–21. How Data Is Represented in Big-Endian Mode

(b) Data in a register 31 0 **MSB** LSB 1 2 3 4 А В С D (a) Data in memory at 0 31 address 0000 1000h MSB LSB 2 1 3 4 В С D А

Figure 3–22 shows how the TMS34020 uses big-endian mode to access an 8-bit field that starts at bit 4 within the long-word. Notice that for big-endian mode, the field's bit address is determined by counting in from the MS (big) end of the long-word.

Figure 3–22. Addressing a Field in a Long-Word (Big-Endian)

	0 MSB	8-bit fi	eld				31 LSB	
address 0000 1000h	1	2	3	4	A   E	з С	D	
	<b>⊢</b> ⇒	Iraaa						
	(numbe	er of bit	s in fro	m the t	oig end	)		

If you want to move this 8-bit data field into a general-purpose register, you might execute the following instructions:

SETF 8,0,0 MOVE @00001004,A0,0

Figure 3–23 shows how the TMS34020 places this data into A0.

Figure 3–23. Moving a Field into a General-Purpose Register (Big-Endian)

	31 MSB		0 LSB
register A0	0 0 0	0 0 0	2 3

Note that the data is right-aligned within the register, so that the LSB of the field coincides with the register's LSB.

#### Differences between big- and little-endian modes -

- ❑ The TMS34020 accesses 32-bit-wide, 32-bit-aligned fields in the same manner for both modes. Differences between the two modes are apparent only when data is not 32 bits long-or when it is not aligned to a 32-bit boundary within memory.
- In both modes, data is right-aligned when it is moved into a generalpurpose register.
- In big-endian mode, bits within a field or long-word are renumbered, not reordered.

#### 3.8.3 Assembling Code for Big-Endian or Little-Endian Addressing

The TMS34020 assembler can produce object code for little-endian or bigendian mode. By default, the assembler produces little-endian code; if you want it to produce big-endian code, be sure to use the –e assembler option.

To assemble little-endian code	To assemble big-endian code		
gspa <i>filename</i>	gspa <b>e</b> filename		

(For more information about the assembler, refer to the *TMS340 Family Code-Generation Tools User's Guide.*)

If you compare the listing files of big- and little-endian code, you'll find no differences in the listed object code. Figure 3–24 shows a listing file with object code. For ease of reading, the assembler lists object code in the same manner for both little- and big-endian code.

#### Figure 3–24. Sample Listing File (Assembler Output) for Little-Endian and Big-Endian Code

					~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~		
	0001	00000000	aabbccdd		.long	0AABBCCDDh	
	0002	00000020	09e0		MOVI	11223344h, A	.0
		00000030	11223344				
	0003	00000050	0380		ABS	AO	
	0004	00000060	09C1		MOVI	6677h, Al	
		00000070	6677				
			<b>A</b>				
1.1			1				
			object code				
	****			,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	*****	******	****

Although the object code in the listing file looks the same for both modes, the assembler actually creates different object code for the two modes. If you're writing a loader, it's important to know how to load the object code into memory. Figure 3–25 demonstrates this.

# Figure 3–25. Loading Object Code into Memory

31			0	31			(
AA	BB	CC	DD	AA	BB	CC	DD
33	44	09	E0	09	E0	33	44
03	80	11	22	11	22	03	80
66	77	09	C1	09	C1	66	77

# 3.8.4 Wiring VRAMs to the LAD Bus

Figure 3–26 shows examples of how you might use TMS44C251 VRAMs in your system. A TMS44C251 has 4 bidirectional data pins, DQ0—DQ3; each data pin is connected to an LAD pin. Figure 3–26 (*a*) shows wiring for little-endian mode; as (*b*) shows, you must wire the VRAMs backwards for big-endian mode.

Figure 3–26. Connecting VRAMs to the LAD Bus

#### (a) Little-endian mode



#### (b) Big-endian mode



# 3.8.5 Big-Endian Effects on Instruction Timing

The instruction timings listed in this document are for little-endian code. Timings for big-endian code are essentially the same as timings for littleendian code; however, the setup for graphics instructions may consume extra machine states (instructions' inner loops consume no additional states). The effect on timing is slight.

# 3.9 Stacks

The TMS34020's system stack is implemented in local memory and managed in hardware. The stack is used to store return addresses and processor status information during interrupts, traps, and subroutine calls. The contents of general-purpose registers can be pushed onto the stack and popped off the stack. The system stack can also be used for dynamically allocated data storage.

The stack is accessed through a dedicated 32-bit internal register, called the stack pointer, or **SP**. The SP points to the top of the system stack; it can be accessed as register 15 in either of the general-purpose register files, A or B.

In addition to the system stack, you can define your own auxiliary stacks. The system stack always grows toward lower memory addresses; an auxiliary stack can be defined to grow toward either lower or higher addresses. The MOVE instructions, combined with the predecrement and postincrement addressing modes, facilitate pushing and popping of auxiliary stack data. You can use one or more general-purpose registers as auxiliary stack pointers and frame pointers. The indexed addressing modes can be used in conjunction with a frame pointer to access variables embedded within the stack.

#### 3.9.1 System Stack

Figure 3–27 shows the structure of the system stack, which grows in the direction of lower memory addresses.



# The SP points to the top of the stack; it contains the 32-bit address of the LSB (bit 0) of the value on top of the stack. The SP can contain any 32-bit address; however, stack operations execute more efficiently when the 5 LSBs of the SP

#### Figure 3–27. System Stack

Memory Organization and Data Structures

are 0s. This aligns the SP to long-word boundaries in memory, reducing the number of memory cycles needed to push or pop values.

Any instruction that manipulates general-purpose registers can also be used to manipulate the SP. The SP can be specified as the source or destination operand in any instruction that operates on the general-purpose registers. Instructions that manipulate the SP include:

Instructions That Push Values on the Stack	Instructions That Pop Values from the Stack
MMTM SP, register list	MMFM SP, register list
CALL <i>Rs</i>	RETI
CALLA absolute address	RETS
CALLR relative address	POPST
TRAP <i>number</i>	MOVE *SP+, <i>Rd</i>
PUSHST	
MOVE Rs*SP	

#### 3.9.1.1 Saving Registers on the System Stack

Register information can be stored on the stack during an interrupt or a subroutine call. This frees up the register for use by an interrupt routine or a subroutine and allows you to restore the original register values from the stack when the routine completes.

During an interrupt, the contents of the PC and ST are automatically saved on the stack; if you want to save values that are in general-purpose registers, you can use the MMTM and MMFM instructions. MMTM pushes multiple general-purpose registers onto the stack, and MMFM pops multiple general-purpose registers from the stack.

When the contents of a 32-bit register are pushed onto the stack, they are stored in the 32-bit word below the word whose address is contained in the SP. This is shown in Figure 3–28, which demonstrates the effects of the following instruction sequence:

MMTM SP,A0 ; Push register A0 onto stack MMFM SP,A1 ; Pop stack into register A1

- Figure 3–28 (a) shows the original state of the stack and registers.
- Figure 3–28 (b) illustrates the state after A0 is pushed onto the stack.
- Figure 3–28 (c) shows the results of popping the top of the stack into A1.





The TMS34020 performs 2 steps to push the contents of a 32-bit register onto the top of the stack:

- 1) Decrements the SP by 32.
- 2) Pushes the register contents onto the stack.

The TMS34020 performs 2 steps to pop the top of the stack into a 32-bit register:

- 1) Pops the 32 bits at the top of the stack into the register.
- 2) Increments the SP by 32.

#### 3.9.1.2 Saving Information on the System Stack During an Interrupt

During an interrupt, the TMS34020 pushes the PC and ST onto the stack; this allows the interrupted routine to resume execution when the interrupt processing is completed. An interrupt routine performs the following actions:

- 1) Decrements the SP by 32.
- 2) Pushes the PC onto the stack.
- 3) Decrements the SP again by 32.
- 4) Pushes the ST onto the stack.

During a return from an interrupt

- 1) Pops the 32 bits at the top of the stack into the ST.
- 2) Increments the SP by 32.
- 3) Pops the 32 bits at the top of the stack into the PC.
- 4) Increments the SP again by 32.

#### 3.9.1.3 Saving Information on the System Stack During a Subroutine Call

A subroutine call saves the state of the calling routine on the stack; this allows the routine to resume execution when the subroutine completes. A subroutine call performs the following actions:

- 1) Decrements the SP by 32.
- 2) Pushes the PC onto the stack.

During a return from a subroutine

- 3) Pops the 32 bits at the top of the stack into the PC.
- 4) Increments the SP by 32.

#### **3.9.2 Auxiliary Stacks**

Auxiliary stacks, which are typically used to contain dynamically allocated data storage, can be managed in software. You can use any A- or B-file register (except the SP) as the auxiliary stack pointer. For the purposes of discussion,

the symbol STK represents the auxiliary stack pointer. STK is a symbol that must be equated to one of the general-purpose registers; for example:

STK .set A0

STK can contain any 32-bit value; however, stack operations execute more efficiently when the 5 LSBs of the STK are 0s. This aligns the STK to long-word boundaries in memory, reducing the number of memory cycles needed to push or pop values.

As Figure 3–29 and Figure 3–30 show, an auxiliary stack can grow in either direction in memory. These figures represent memory as a string of continuously addressable bits.

Figure 3–29 shows a stack that grows toward lower memory addresses.

Figure 3–29. An Auxiliary Stack That Grows Toward Lower Addresses



- Figure 3–29 (a) shows the original stack.
- In Figure 3–29 (b), a field of arbitrary size is pushed onto the stack with this instruction:

MOVE Rs,\*-STK

(*Rs* and STK represent general-purpose registers and must be in the same register file.)

In Figure 3–29 (c), the field is popped off the stack with this instruction:

MOVE \*STK+,Rd

(*Rd* and STK represent general-purpose registers and must be in the same register file.)

Between instructions, STK always points to the lowest bit address in the stack—this corresponds to the very top of the stack. You can use the MMTM STK, *register list* instruction to save multiple registers on the stack in Figure 3–29. Later, you can restore the registers to their former values with an MMFM STK, *register list* instruction.

Figure 3–30 shows a stack that grows toward higher memory addresses:

Figure 3–30. An Auxiliary Stack That Grows Toward Higher Addresses



Figure 3–30 (a) shows the original stack.

□ In Figure 3–30 (*b*), a field of arbitrary size is pushed onto the stack using the following instruction:

MOVE Rs,\*STK+

In Figure 3–30 (c), the field is popped off the stack with this instruction:

MOVE \*-STK,Rd

Between instructions, the STK always points to one plus the highest bit address in the stack—this location is one bit beyond the very top of the stack.



# Chapter 4

# **TMS34020 Registers**

The TMS34020 has two on-chip general-purpose register files, file A and file B. Each register file contains fifteen 32-bit registers. The register files share a 32-bit hardware stack pointer (SP) that automatically manages the system stack during interrupts and subroutine calls. The TMS34020 also has 2 dedicated 32-bit registers—a program counter and a status register.

In addition to the CPU registers, the TMS34020 has 54 memory-mapped registers that are dedicated to I/O functions.

	Sect	ion	Page
Dedicated registers include the	4.1	The Status Register (ST)	4-2
status register, program counter, and stack pointer.	4.2	The Program Counter (PC)	4-4
	4.3	The Stack Pointer (SP)	4-5
Programmable/general-purpose	4.4	General-Purpose Registers	
registers include dual register files		(Register Files A and B)	4-6
and memory-mapped I/O registers.	4.5	I/O Registers	4-9
	4.6	Alphabetical Summary of I/O Registers	
		and B-File Registers	4-14

# 4.1 The Status Register (ST)

The status register (ST) is a special-purpose, 32-bit register that reflects the processor status. The ST also contains several parameters that define the characteristics of two programmable data types, fields 0 and 1. At reset, the TMS34020 initializes the ST to 0000 0010h. Figure 4–1 illustrates the status register. Table 4–1 lists the functions associated with the status bits.

Figure 4–1. Status Register

31	30	29	28	26	25	22	21	11	10—6	5	40
Ν	С	Z	V	BF	IX	SS	IE	FE1	FS1	FE0	FS0
Note	Note: Shaded portions are reserved.										

Table 4–1. Definitions of Bits in the Status Register

Bit	Field				
Number	Name	Function			
04	FS0	Field size 0: Length in bits of the first memory data field.			
		$FS0 = 00001_2 - 11111_2$ defines a field size of 1-31			
		FS0 = 00000 <sub>2</sub> defines a field size of 32			
5	FE0	Field extension 0: Determines whether the memory field is extended with 0s or with the sign bit when loaded into a 32-bit general-purpose register.			
		FE0 = 0 selects zero extension for field 0			
		FE0 = 1 selects sign extension for field 0			
6—10	FS1	Field size 1: Length in bits of the second memory data field.			
		FS1 = 000012-111112 defines a field size of 131			
		FS1 = 00000 <sub>2</sub> defines a field size of 32			
11	FE1	Field extension 1: Determines whether the memory field is extended with 0s or with the sign bit when loaded into a 32-bit general-purpose register.			
		FE1 = 0 selects zero extension for field 1			
		FE1 = 1 selects sign extension for field 1			
21	IE	Interrupt enable: Master interrupt enable/disable bit.			
		IE = 0 disables all maskable interrupts			
		IE = 1 enables all maskable interrupts			
22	SS	Single step: Setting the SS bit to 1 causes the TMS34020 to interrupt program execu- tion following execution of each instruction. This is useful for debugging purposes.			
25	IX	Interruptible instruction executing: When an interrupt occurs during instruction ex- ecution, the TMS34020 sets or clears the IX bit before saving the ST on the stack.			
		IX=0 indicates that an interrupt occurred at an instruction boundary			
		<b>IX = 1</b> indicates that an interrupt occurred in the middle of an interruptible instruction			

Bit	Field	
Number	Name	Function
26	BF	Bus fault: Set when a bus fault occurs on a local-memory cycle.
28	V	Overflow: Set according to instruction execution.
29	Z	Zero: Set according to instruction execution.
30	С	Carry: Set according to instruction execution.
31	N	Negative: Set according to instruction execution.
12—20 23—24	ARAMANAN ARAMANAN ARAMANAN ARAMANAN ARAMANAN ARAMANAN	<b>Reserved:</b> These bits are reserved; the TMS34020 does not use them. At reset, the TMS34020 clears these reserved bits to 0.
27		Note: To maintain compatibility, you should write only 0s to these bits.

Table 4–1. Definitions of Bits in the Status Register (continued)

All instructions can potentially change the status register; during instruction execution, the TMS34020 may set the V, Z, C, and N bits. If you want to directly affect the ST, you can use the following instructions.

**PUTST** writes the contents of a specified general-purpose register into the status register. Here's an example:

MOVI 00000010h, A0 PUTST A0

- **GETST** copies the contents of the ST into a specified general-purpose register.
- **SETC** sets the C bit without altering any other status bits.
- **CLRC** clears the C bit without altering any other status bits.
- **SETF** writes values to the FS0 and FE0 or FS1 and FE1 bits without altering any other status bits.
- **EXGF** exchanges the 6 LSBs of a specified general-purpose register with the FS0 and FE0 bits or with the FS1 and FE1 bits.
- **EINT** sets the IE bit.
- **DINT** clears the IE bit.
- **PUSHST** pushes the contents of the ST onto the stack.
- **POPST** pops the value at the top of the stack into the ST.

# 4.2 The Program Counter (PC)

The program counter (PC) is a special-purpose, 32-bit register that points to the next instruction word to be executed. Instructions are always aligned on 16-bit boundaries; thus, as Figure 4–2 shows, the PC's 4 LSBs always contain 0s.

#### Figure 4–2. Program Counter

1	31 4	3	2	1	0
	word address	0	0	0	0
•	-28 bits ►	-	-4 k	oits	

An instruction consists of one or more 16-bit instruction words. The first word contains the opcode for the instruction; additional words may contain immediate data, displacements, or absolute addresses. As the TMS34020 fetches each 16-bit instruction word, it increments the PC to point to the next instruction word.

The PC contents are replaced during a branch instruction, subroutine call instruction, return instruction, or interrupt. As Table 4–2 shows, instructions can be categorized according to their effects on the PC.

#### Table 4–2. How Instruction Execution Affects the PC

Instruction Type	Effect on PC
No branch	The PC is incremented at the end of the instruction, al- lowing execution to proceed sequentially to the next in- struction.
Absolute branch (TRAP, CALLA, JAcc)	The PC is loaded with an absolute address; the ad- dress' 4 LSBs are set to 0s.
Relative branch (CALLR, JR <i>cc</i> , DSJ <i>cc</i> )	The signed displacement (8 or 16 bits) is added to the PC's current contents. The signed displacement is treated as a word displacement; that is, it is shifted left 4 bit positions before it is added to the PC.
Indirect branch (JUMP, CALL)	The PC is loaded with the register contents. The 4 LSBs are set to 0s.

Two additional instructions provide you with direct control of the PC.

- **GETPC** copies the contents of the PC into a specified general-purpose register.
- **EXGPC** exchanges the contents of the PC with the contents of a specified general-purpose register.

# 4.3 The Stack Pointer (SP)

The stack pointer (SP) is a special-purpose, 32-bit register that contains the bit address of the top of the system stack. *The TMS34020 contains only a single SP*; however, this SP can be addressed as a member of *either* register file, as register A15 or register B15. Any instruction that uses a general-purpose register as an operand can also use the SP as an operand.

Figure 4–3 illustrates the stack pointer; Section 3.9, <u>Stacks</u>, (page 3-26) describes stack operation in detail.

Figure 4–3. The Stack-Pointer Register



The system stack grows toward smaller addresses. The stack pointer always points to the value at the top of the stack. Specifically, the SP contains the 32-bit address of the LSB of that value. Although the SP's 5 LSBs can have any arbitrary value, stack operations execute more efficiently when the 5 LSBs are 0. Clearing these bits to 0s aligns the stack pointer on a 32-bit word boundary; thus, only a single memory access (two cycles) is necessary to push or pop the contents of a 32-bit register.

During subroutine calls and interrupts, the PC and ST are pushed onto the stack. These are both 32-bit registers. (If the SP is not long-word aligned when instruction execution is interrupted, the TMS34020 aligns the stack before saving the PC and ST.) The MMTM and MMFM instructions push/pop multiple 32-bit registers from the A or B file.

For the most efficient execution, you should ensure that the SP is always aligned to a long word and that it is incremented or decremented in multiples of 32 bits.

# 4.4 General-Purpose Registers (Register Files A and B)

The TMS34020 has thirty 32-bit general-purpose registers, divided into register files A and B. The register files share a single stack pointer (SP). Figure 4–4 illustrates the register files (note the shared SP).





As Figure 4–4 shows, 15 of the general-purpose registers, A0—A14, form register file A. Register file B also consists of 15 general-purpose registers, B0—B14. Many of the TMS34020 instructions use these registers for storing and manipulating data.

The TMS34020's register files have several advantages:

- The general-purpose registers are dual-ported. This allows the TMS34020 to read from or write to two separate registers at the same time.
- Several instructions use general-purpose registers to contain source and destination operands; these are called **register-to-register instructions**.

Multiple internal data paths link the ALU to the general-purpose registers, allowing the TMS34020 to execute most register-to-register instructions in a single machine state. Single-state instructions include add, subtract, Boolean operations, and shifts (1 to 32 bits).

During a single-state instruction, the following actions may occur:

- The TMS34020 reads, in parallel, two 32-bit operands from the general-purpose registers.
- The ALU performs the specified operation.
- The 32-bit result is stored in the specified general-purpose register.

All register-to-register instructions (except MOVE *Rs*, *Rd*) require both registers to be in the same file. Instructions that can use registers A0—A14 and B0—B14 as operands can also use the SP as an operand.

#### Note:

For some graphics operations, the B-file registers have hardware-dedicated functions. When their special functions are used, the contents of the B-file registers are referred to as **implied graphics operands**. Several I/O registers also contain implied operands.

No hardware-dedicated functions are associated with the A-file registers; generally, instructions do not use the A-file registers as implied operands.

Table 4–3 (page 4-8) summarizes the names and functions associated with the B-file registers when they are used as implied operands.

B0 SADDR Source address. Address (linear or XY) of a source pixel array; usually the address of the array's upper left corner (the lowest pixel address in the array).   B1 SPTCH Source pitch. Difference in start addresses (linear) between adjacent rows of a source pixel array.   B2 DADDR Destination address. Address (linear or XY) of a destination pixel array; usually the address of the array's upper left corner (the lowest pixel address in the array).   B3 DPTCH Destination pitch. Difference in start addresses (linear) between adjacent rows of a destination pixel array.   B4 OFFSET Offset. Linear bit address. Corresponds to the XY origin (X=0, Y=0).   B5 WSTART Window start address. XY address of the upper left corner of the window (smallest X and Y coordinate values in the window).   B6 WEND Window end address. XY address of the lower right corner of the window (largest X and Y coordinate values in the window).   B7 DYDX Detta Y/detta X. The 16 LSBs of DYDX define the width (X dimension) of a pixel array. The 16 MSBs define the height (Y dimension) of a pixel array.   B8 COLOR0 Background pixel color. COLOR0 contains the background color for graphics operations.   B10 MADDR Mask address. Address of the upper left corner of a mask pixel array (lowest pixel adress in the array).   B11 MPTCH Mask address. Address of the upper left corner of a m	Register	Function	Description
B1 SPTCH Source pitch. Difference in start addresses (linear) between adjacent rows of a source pixel array.   B2 DADDR Destination address. Address (linear or XY) of a destination pixel array; usually the address of the array's upper left corner (the lowest pixel address in the array).   B3 DPTCH Destination pitch. Difference in start addresses (linear) between adjacent rows of a destination pixel array.   B4 OFFSET Offset. Linear bit address, corresponds to the XY origin (X=0, Y=0).   B5 WSTART Window start address. XY address of the upper left corner of the window (smallest X and Y coordinate values in the window).   B6 WEND Window end address. XY address of the lower right corner of the window (largest X and Y coordinate values in the window).   B7 DYDX Delta Y/delta X. The 16 LSBs of DYDX define the width (X dimension) of a pixel array. The 16 MSBs define the height (Y dimension) of a pixel array.   B8 COLOR0 Background pixel color. COLOR0 contains the background color for graphics operations.   B9 COLOR1 Foreground pixel color. COLOR1 contains the foreground color for graphics operations.   B10 MADDR Mask address. Address of the upper left corner of a mask pixel array (lowest pixel adress in incement. LINE & FLINE use INC1 to identify the amount by which a pixel adress is incremented in the diagonal direction.   B11 MPTCH Mask pitch. Dif	B0	SADDR	<b>Source address.</b> Address (linear or XY) of a source pixel array; usually the address of the array's upper left corner (the lowest pixel address in the array).
B2 DADDR Destination address. Address (linear or XY) of a destination pixel array; usually the address of the array's upper left corner (the lowest pixel address in the array).   B3 DPTCH Destination pixch. Difference in start addresses (linear) between adjacent rows of a destination pixel array.   B4 OFFSET Offset. Linear bit address, corresponds to the XY origin (X=0, Y=0).   B5 WSTART Window start address. XY address of the upper left corner of the window (smallest X and Y coordinate values in the window).   B6 WEND Window end address. XY address of the lower right corner of the window (largest X and Y coordinate values in the window).   B7 DYDX Delta Y/delta X. The 16 LSBs of DYDX define the width (X dimension) of a pixel array. The 16 MSBs define the height (Y dimension) of a pixel array.   B8 COLOR0 Background pixel color. COLOR0 contains the background color for graphics operations.   B9 COLOR1 Foreground pixel color. COLOR1 contains the foreground color for graphics operations.   B10 MADDR Mask address. Address of the upper left corner of a mask pixel array (lowest pixel array.)   B11 MPTCH Mask pitch. Difference in start addresses (linear) between adjacent rows of a mask array.   B12 INC1 Diagonal increment. LINE & FLINE use INC1 to identify the amount by which a pixel address is incremented in the diagonal direction.	B1	SPTCH	Source pitch. Difference in start addresses (linear) between adjacent rows of a source pixel array.
B3DPTCHDestination pitch. Difference in start addresses (linear) between adjacent rows of a destination pixel array.B4OFFSETOffset. Linear bit address, corresponds to the XY origin (X=0, Y=0).B5WSTARTWindow start address. XY address of the upper left corner of the window (smallest X and Y coordinate values in the window).B6WENDWindow end address. XY address of the lower right corner of the window (largest X and Y coordinate values in the window).B7DYDXDelta Y/delta X. The 16 LSBs of DYDX define the width (X dimension) of a pixel array. The 16 MSBs define the height (Y dimension) of a pixel array. 	B2	DADDR	<b>Destination address.</b> Address (linear or XY) of a destination pixel array; usually the address of the array's upper left corner (the lowest pixel address in the array).
B4OFFSETOffset. Linear bit address, corresponds to the XY origin (X=0, Y=0).B5WSTARTWindow start address. XY address of the upper left corner of the window (smallest X and Y coordinate values in the window).B6WENDWindow end address. XY address of the lower right corner of the window (largest X and Y coordinate values in the window).B7DYDXDelta Y/delta X. The 16 LSBs of DYDX define the width (X dimension) of a pixel array. The 16 MSBs define the height (Y dimension) of a pixel array.B8COLOR0Background pixel color. COLOR0 contains the background color for graphics oper- ations.B9COLOR1Foreground pixel color. COLOR1 contains the foreground color for graphics opera- tions.B10MADDRMask address. Address of the upper left corner of a mask pixel array (lowest pixel address in the array). COUNTB11MPTCHMask pitch. Difference in start addresses (linear) between adjacent rows of a mask array. INC1B12INC2Dominant increment. LINE & FLINE use INC1 to identify the amount by which a pixel address is incremented in the diagonal direction. TEMPB13PATTERNArray or line pattern. The 1s and 0s within PATTERN identify a pixel pattern for an array or a line.	B3	DPTCH	<b>Destination pitch.</b> Difference in start addresses (linear) between adjacent rows of a destination pixel array.
B5WSTARTWindow start address. XY address of the upper left corner of the window (smallest X and Y coordinate values in the window).B6WENDWindow end address. XY address of the lower right corner of the window (largest X and Y coordinate values in the window).B7DYDXDelta Y/delta X. The 16 LSBs of DYDX define the width (X dimension) of a pixel array. The 16 MSBs define the height (Y dimension) of a pixel array. The 16 MSBs define the height (Y dimension) of a pixel array.B8COLOR0Background pixel color. COLOR0 contains the background color for graphics operations.B9COLOR1Foreground pixel color. COLOR1 contains the foreground color for graphics operations.B10MADDRMask address. Address of the upper left corner of a mask pixel array (lowest pixel address in the array). COUNTCOUNTLoop counter. LINE & FLINE instructions use B10 to count the number of pixels drawn within the line. Temporary register.B11MPTCHMask pitch. Difference in start addresses (linear) between adjacent rows of a mask array. INC1B12INC2Dominant increment. LINE & FLINE use INC1 to identify the amount by which a pixel address is in cremented in the diagonal direction. Temporary register.B13PATTERNArray or line pattern. The 1s and 0s within PATTERN identify a pixel pattern for an array or a line.B14TEMPTemporary register.	B4	OFFSET	Offset. Linear bit address, corresponds to the XY origin (X=0, Y=0).
B6WENDWindow end address. XY address of the lower right corner of the window (largest X and Y coordinate values in the window).B7DYDXDelta Y/delta X. The 16 LSBs of DYDX define the width (X dimension) of a pixel array. The 16 MSBs define the height (Y dimension) of a pixel array.B8COLOR0Background pixel color. COLOR0 contains the background color for graphics oper- ations.B9COLOR1Foreground pixel color. COLOR1 contains the foreground color for graphics opera- ations.B10MADDRMask address. Address of the upper left corner of a mask pixel array (lowest pixel address in the array). COUNTCOUNTLoop counter. LINE & FLINE instructions use B10 to count the number of pixels drawn within the line. TEMPB11MPTCHMask pitch. Difference in start addresses (linear) between adjacent rows of a mask array. INC1B12INC2Dominant increment. LINE & FLINE use INC1 to identify the amount by which a pix- el address is incremented in the diagonal direction.B13PATTERNArray or line pattern. The 1s and 0s within PATTERN identify a pixel pattern for an array or a line.B14TEMPTemporary register.	B5	WSTART	Window start address. XY address of the upper left corner of the window (smallest X and Y coordinate values in the window).
B7DYDXDelta Y/delta X. The 16 LSBs of DYDX define the width (X dimension) of a pixel array. The 16 MSBs define the height (Y dimension) of a pixel array.B8COLOR0Background pixel color. COLOR0 contains the background color for graphics oper- ations.B9COLOR1Foreground pixel color. COLOR1 contains the foreground color for graphics opera- tions.B10MADDRMask address. Address of the upper left corner of a mask pixel array (lowest pixel address in the array). COUNTLoop counter. LINE & FLINE instructions use B10 to count the number of pixels drawn within the line. TEMPB11MPTCHMask pitch. Difference in start addresses (linear) between adjacent rows of a mask array. INC1Dagonal increment. LINE & FLINE use INC1 to identify the amount by which a pixel address is incremented in the diagonal direction.B12INC2Dominant increment. LINE & FLINE use INC2 to identify the amount by which a pixel al address is incremented in the dominant direction. TEMPB13PATTERNArray or line pattern. The 1s and 0s within PATTERN identify a pixel pattern for an array or a line.B14TEMPTemporary register.	B6	WEND	Window end address. XY address of the lower right corner of the window (largest X and Y coordinate values in the window).
B8COLOR0Background pixel color. COLOR0 contains the background color for graphics operations.B9COLOR1Foreground pixel color. COLOR1 contains the foreground color for graphics operations.B10MADDRMask address. Address of the upper left corner of a mask pixel array (lowest pixel address in the array). Loop counter. LINE & FLINE instructions use B10 to count the number of pixels drawn within the line. TEMPB11MPTCHMask pitch. Difference in start addresses (linear) between adjacent rows of a mask array. INC1B12INC2Dominant increment. LINE & FLINE use INC1 to identify the amount by which a pixel address is incremented in the diagonal direction. TEMPB13PATTERNArray or line pattern. The 1s and 0s within PATTERN identify a pixel pattern for an array or a line.B14TEMPTemporary register.	B7	DYDX	<b>Delta Y/delta X</b> . The 16 LSBs of DYDX define the width (X dimension) of a pixel array. The 16 MSBs define the height (Y dimension) of a pixel array.
B9COLOR1Foreground pixel color. COLOR1 contains the foreground color for graphics operations.B10MADDRMask address. Address of the upper left corner of a mask pixel array (lowest pixel address in the array). COUNTCOUNTLoop counter. LINE & FLINE instructions use B10 to count the number of pixels drawn within the line. TEMPB11MPTCHMask pitch. Difference in start addresses (linear) between adjacent rows of a mask array. INC1B12INC2Dominant increment. LINE & FLINE use INC1 to identify the amount by which a pixel address is incremented in the diagonal direction.B13PATTERNArray or line pattern. The 1s and 0s within PATTERN identify a pixel pattern for an array or a line.B14TEMPTemporary register.	B8	COLOR0	Background pixel color. COLOR0 contains the background color for graphics oper- ations.
B10MADDRMask address. Address of the upper left corner of a mask pixel array (lowest pixel address in the array). Loop counter. LINE & FLINE instructions use B10 to count the number of pixels drawn within the line. TEMPB11MPTCHMask pitch. Difference in start addresses (linear) between adjacent rows of a mask 	B9	COLOR1	Foreground pixel color. COLOR1 contains the foreground color for graphics opera- tions.
Address in the array).COUNTLoop counter. LINE & FLINE instructions use B10 to count the number of pixels drawn within the line. TEMPB11MPTCHMask pitch. Difference in start addresses (linear) between adjacent rows of a mask array. INC1B12INC2Diagonal increment. LINE & FLINE use INC1 to identify the amount by which a pixel 	B10	MADDR	Mask address. Address of the upper left corner of a mask pixel array (lowest pixel
TEMPTemporary register.B11MPTCHMask pitch. Difference in start addresses (linear) between adjacent rows of a mask array. INC1INC1Diagonal increment. LINE & FLINE use INC1 to identify the amount by which a pixel address is incremented in the diagonal direction.B12INC2Dominant increment. LINE & FLINE use INC2 to identify the amount by which a pix- el address is incremented in the dominant direction. TEMPB13PATTERNArray or line pattern. The 1s and 0s within PATTERN identify a pixel pattern for an array or a line.B14TEMPTemporary register.		COUNT	address in the array). Loop counter. LINE & FLINE instructions use B10 to count the number of pixels drawn within the line.
B11MPTCHMask pitch. Difference in start addresses (linear) between adjacent rows of a mask array.INC1Diagonal increment. LINE & FLINE use INC1 to identify the amount by which a pixel address is incremented in the diagonal direction.B12INC2Dominant increment. LINE & FLINE use INC2 to identify the amount by which a pix- el address is incremented in the dominant direction. TEMPB13PATTERNArray or line pattern. The 1s and 0s within PATTERN identify a pixel pattern for an 		TEMP	Temporary register.
INC1Diagonal increment. LINE & FLINE use INC1 to identify the amount by which a pixel address is incremented in the diagonal direction.B12INC2Dominant increment. LINE & FLINE use INC2 to identify the amount by which a pix- el address is incremented in the dominant direction. TEMPB13PATTERNArray or line pattern. The 1s and 0s within PATTERN identify a pixel pattern for an array or a line.B14TEMPTemporary register.	B11	MPTCH	Mask pitch. Difference in start addresses (linear) between adjacent rows of a mask
B12INC2Dominant increment. LINE & FLINE use INC2 to identify the amount by which a pixel address is incremented in the dominant direction. TEMPB13PATTERNArray or line pattern. The 1s and 0s within PATTERN identify a pixel pattern for an array or a line.B14TEMPTemporary register.		INC1	<b>Diagonal increment.</b> LINE & FLINE use INC1 to identify the amount by which a pixel address is incremented in the diagonal direction.
B13 PATTERN Array or line pattern. The 1s and 0s within PATTERN identify a pixel pattern for an array or a line.   B14 TEMP Temporary register.	B12	INC2	Dominant increment. LINE & FLINE use INC2 to identify the amount by which a pix-
B13 PATTERN Array or line pattern. The 1s and 0s within PATTERN identify a pixel pattern for an array or a line.   B14 TEMP Temporary register.		TEMP	el address is incremented in the dominant direction. Temporary register.
B14 TEMP Temporary register.	B13	PATTERN	Array or line pattern. The 1s and 0s within PATTERN identify a pixel pattern for an array or a line.
	B14	TEMP	Temporary register.

Table 4–3. Summary of B-File Registers' Implied-Operand Functions

**Note:** Some graphics instructions use the TEMP (temporary) registers to store temporary values and context information during instruction execution.

# 4.5 I/O Registers

The TMS34020 supports a set of I/O registers that control and monitor

- communications between the TMS34020 and a host processor,
- the TMS34020's interface to local memory,
- interrupts,
- video timing and screen refreshing, and
- **G** graphics-drawing operations.

The I/O registers reside in the TMS34020's on-chip memory, occupying addresses C000 0000h—C000 03FFh. Figure 4–5 shows this.

Figure 4–5. I/O Register Memory Map

	Most Significant Half	Least Significant Hal	f
HESYNC HEBLNK HSBLNK HTOTAL DPYSTRT CONTROL HSTADRL HSTCTLL	C000 0010 C000 0030 C000 0050 C000 0070 C000 0090 C000 00B0 C000 00D0 C000 00F0	C000 0000 C000 0020 C000 0040 C000 0060 C000 0080 C000 00A0 C000 00C0 C000 00E0	VESYNC VEBLNK VSBLNK VTOTAL DPYCTL DPYINT HSTDATA HSTADRH
INTENB CONVSP PSIZE PMASKH CONTROL DPYTAP HCOUNT REFADR	C000 0110 C000 0130 C000 0150 C000 0170 C000 0190 C000 01B0 C000 01D0 C000 01F0	C000 0100 C000 0120 C000 0140 C000 0160 C000 0180 C000 01A0 C000 01C0 C000 01E0	HSTCTLH INTPEND CONVDP PMASKL CONVMP CONFIG VCOUNT DPYADR
DPYSTH DPYNXH DINCH HESERR reserved reserved BSFLTST reserved	C000 0210 C000 0230 C000 0250 C000 0270 C000 0290 C000 02B0 C000 02D0 C000 02F0	C000 0200 C000 0220 C000 0240 C000 0260 C000 0280 C000 02A0 C000 02C0 C000 02E0	DPYSTL DPYNXL DINCL reserved reserved SCOUNT DPYMSK
SETHCNT BSFLTDH reserved IHOST1H IHOST2H IHOST3H IHOST4H	C000 0310 C000 0330 C000 0350 C000 0370 C000 0390 C000 03B0 C000 03D0 C000 03F0	C000 0300 C000 0320 C000 0340 C000 0360 C000 0380 C000 03A0 C000 03C0 C000 03E0	SETVCNT BSFLTDL reserved IHOST1L IHOST2L IHOST3L IHOST4L

The TMS34020 can access these registers directly; a host processor can access them through the TMS34020's host interface. I/O registers are accessed like any other memory location. Table 4–4 summarizes the I/O registers and their functions.

Register	Address	Description
BSFLTDL BSFLTDH	C000 0320h C000 0330h	Bus-fault data. When a bus fault occurs, the TMS34020 stores the current LAD data in the BSFLTD registers.
BSFLTST	C000 2D0h	<b>Bus-fault status.</b> When a bus fault occurs, the TMS34020's memory con- troller saves its current state into BSFLTST.
CONFIG	C000 01A0h	<b>System configuration.</b> Contains several parameters that enable VRAM register loads and control little-/big-endian addressing, row-column address configuration, and refresh rates.
CONTROL	C000 00B0h C000 0190h	Memory control. Controls transparency, window checking, PIXBLT direc- tion, and cache operation.
CONVDP	C000 0140h	<b>Destination pitch conversion factor.</b> Contains the XY-to-linear factor for converting a destination array address.
CONVMP	C000 0180h	Mask pitch conversion factor. Contains the XY-to-linear factor for con- verting a mask array address.
CONVSP	C000 0130h	<b>Source pitch conversion factor.</b> Contains the XY-to-linear factor for converting a source array address.
DINCL DINCH	C000 0240h C000 0250h	<b>Display increment.</b> Contains the difference in addresses between vertically adjacent pixels (the <i>display pitch</i> ).
DPYADR	C000 01E0h	Display address. Provides TMS34010 compatibility.
DPYCTL	C000 0080h	Display control. Controls video timing parameters.
DPYINT	C000 00A0h	<b>Display interrupt.</b> Identifies the next scan line at which a display interrupt will be requested.
DPYNXL DPYNXH	C000 0220h C000 0230h	Display next address. The DPYNX registers contain a 32-bit address.
DPYMSK	C000 02E0h	<b>Display mask.</b> When screen refreshes are enabled, DPYMSK defines which bits of the address in the DPYNX and DPYST registers correspond to the tap-point portion of the address output during screen-refresh cycles.
DPYSTL DPYSTH	C000 0200h C000 0210h	<b>Display start address.</b> Points to the pixel at the left of the 1 <sup>st</sup> line displayed on the screen.
DPYSTRT	C000 0090h	Display start address. Provides TMS34010 compatibility.
DPYTAP	C000 01B0h	Display tap point address. Provides TMS34010 compatibility.
HCOUNT	C000 01D0h	Horizontal count. Tracks the number of VCLKs per horizontal scan line.
HEBLNK	C000 0030h	Horizontal end blank. Defines the point at which the horizontal blanking interval ends.
HESERR	C000 0270h	Horizontal end serration. Defines the point at which the composite sync pulse ends during the serration region of vertical blanking.
HESYNC	C000 0010h	Horizontal end sync. Defines the point at which the horizontal sync pulse ends.
HSTADRH HSTADRL	C000 00E0h C000 00D0h	Host interface address. Provides TMS34010 compatibility.

# Table 4–4. Summary of I/O Registers

Register	Address	Description
HSTCTLH	C000 0100h	Host interface control, high word. Controls host-interface functions such as halt acknowledge, software reset, the nonmaskable interrupt, host autoincrements and prefetches, and halting TMS34020 execution.
HSTCTLL	C000 00F0h	Host interface control, low word. Controls host-interface functions such as messages, emulator control, and bus-fault and retry information.
HSTDATA	C000 00C0h	Host interface data. Provides TMS34010 compatibility.
HSBLNK	C000 0050h	Horizontal start blank. Defines the point at which the horizontal blanking interval begins.
HTOTAL	C000 0070h	Horizontal total. Defines the duration of each horizontal scan line (in terms of VCLK periods).
IHOST	C000 0380h through C000 03F0h	Internal host interface address. The host interface uses these 32-bit locations.
INTENB	C000 0110h	<b>Interrupt enable.</b> Assuming that the status IE bit = 1, setting specific bits to 1 enables external interrupts 1 & 2, the host interrupt, the display interrupt, or the window-violation interrupt.
INTPEND	C000 0120h	<b>Interrupt pending.</b> The values of specific bits indicate whether an exter- nal interrupt, host interrupt, display interrupt, or window-violation interrupt has been requested but not yet serviced.
PMASKL PMASKH	C000 0160h C000 0170h	<b>Plane mask.</b> The PMASK registers form a 32-bit value that selectively en- ables/disables individual planes in a multiple-bit-per-pixel display system.
PSIZE	C000 0150h	<b>Pixel size.</b> Defines the pixel size (in bits). Valid pixel sizes include 1, 2, 4, 8, 16, and 32.
REFADR	C000 01F0h	<b>Refresh pseudo-address.</b> Contains the address output during DRAM-refresh cycles.
SCOUNT	C000 02C0h	<b>Shift clock counter.</b> Incremented during the active display time so that it always contains the tap point of the bit most recently shifted out of the VRAM serial registers.
SETHCNT	C000 0310h	Set horizontal count. When external video is enabled, SETHCNT con- tains the value that is loaded into HCOUNT.
SETVCNT	C000 0300h	Set vertical count. When external video is enabled, SETVCNT contains the value that is loaded into VCOUNT.
VCOUNT	C000 01C0h	Vertical count. Counts the horizontal lines in the video display, increment- ing on the same clock edge that resets HCOUNT to 0.
VEBLNK	C000 0020h	Vertical end blank. Defines the time at which the vertical blanking interval ends.
VESYNC	C000 0000h	Vertical end sync. Defines the time at which the vertical sync pulse ends.
VSBLNK	C000 0040h	Vertical start blank. Defines the time at which the vertical blanking inter- val begins.
VTOTAL	C000 0060h	Vertical total. Defines the time at which the vertical sync pulse begins.

Table 4–4.	Summar	v of I/O	Reaisters	(continued)
14010 1 11	ounnun	, .,,.	riegiotore	loonanaoa

#### 4.5.1 CPU Control Registers

CONTROL	PSIZE	CONVSP
CONVDP	CONVMP	

These 5 registers provide CPU control. They allow you to select those TMS34020 characteristics that meet your specific system needs, such as the pitches for pixel transfers, window-checking modes, transparency modes, Boolean or arithmetic pixel-processing options, PIXBLT direction, and pixel size.

## 4.5.2 Host Communications Registers

HSTCTLH	HSTCTU	
110101211	11010111	

These registers provide a host processor with the ability to interrupt or halt the TMS34020, flush the instruction cache, communicate with an emulator, and select modes for accessing TMS34020 local memory.

#### 4.5.3 Local-Memory and DRAM/VRAM Interface Registers

CONFIG	PMASK	REFADR
BSFLTD	BSFLTST	

The memory controller manages the TMS34020's interface to the local memory, automatically performing the bit alignment and the masking necessary to access data located at arbitrary bit boundaries within memory.

#### 4.5.4 Interrupt Registers

	IN THE PARTY AND A LODA
INITENID	INITOENIO

These registers control and monitor interrupt requests to the TMS34020, including 2 externally generated interrupts and 3 internally generated interrupts, including

- External interrupts 1 and 2
- □ Window-violation interrupt
- Host interrupt
- Display interrupt

If the IE status bit (global interrupt enable) = 1, you can set a bit in the INTENB register to enable any of these interrupts. You can check bits in the INTPEND register to see if any of these interrupts are pending.

# 4.5.5 Video Timing and Screen-Refresh Registers

Twenty-eight registers are dedicated to video timing and screen-refresh functions. The TMS34020 can drive composite sync or separate sync displays. Parameters in the DPYCTL register allow you to select the direction (input/output) of the sync signals:

Composite Sync Mode	Separate Sync Mode		
Signal Direction	Signal Direction		
VSYNC I/O	VSYNC I/O		
HSYNC I/O	HSYNC I/O		
CSYNC I/O	HBLNK O		
CBLNK O	VBLNK O		

In composite mode, the TMS34020 can extract VSYNC and HSYNC from an external composite sync, or it can generate CSYNC from separate VSYNC and HSYNC inputs. Internally, you can set the TMS34020 to preset the horizontal and vertical counts upon receiving an external sync signal. This allows compensation for any combination of internal and external delays that occur in the video synchronization process.

- An external HSYNC loads HCOUNT from SETHCNT.
- An external VSYNC loads VCOUNT.
- An external CSYNC loads both HCOUNT and VCOUNT from SETHCNT and SETVCNT, respectively.

The TMS34020 directly supports multiport VRAMs by generating the serialregister transfer cycles that are necessary for refreshing a display. The memory locations that contain the display information, as well as the number of horizontal scan lines displayed between serial-register transfer cycles, are programmable.

# 4.5.6 Latency of Writes to I/O Registers

The TMS34020 has a high degree of internal parallelism; for example, it can fetch instructions and data while still executing the current instruction. Normally this is beneficial. This could cause problems, however, if the current instruction alters an I/O register and the next instruction uses that register as an implied operand. In this situation, the second instruction may not execute properly. This could occur, for example, if a PIXBLT followed a MOVE instruction that modified the CONTROL register.

You can easily avoid this situation by ensuring that the write to the I/O register completes before any subsequent instructions use the modified register value. To do this, follow the write to the register with an MWAIT instruction.

# 4.6 Alphabetical Summary of I/O Registers and B-File Registers

The remainder of this chapter contains an alphabetical reference of the I/O and B-file registers. Some I/O registers contain implied operands for graphics instructions; the B-file registers also contain implied graphics operands. Therefore, the B-file registers and I/O registers are summarized together in this section.

Here's an important point: Although you'll use both B-file and I/O registers as implied operands, you must access them differently. Because the I/O registers are memory mapped, they are accessed similarly to external memory locations.

The code segment below shows a sample implied-operand setup for a FILL L instruction. It shows that you must use a different MOVE instruction for loading an I/O register than you would use for loading a B-file register. Note that most programs refer to registers by their symbolic names (such as DADDR or PSIZE, assuming you've equated these names to the actual register name or location).

*	Set	up the	e B-fi	le re	giste	rs			
		MOVI	0050h	, в2			;	DADDR	
		MOVI	0100h	, вз			;	DPTCH	
		MOVI	00005	0008h	<b>,</b> B7		;	DYDX	
*	Set	up the	e I/O :	regis	ters				
		MOVK	4, AO						
		MOVE	A0, 0	0C000	0150h	, 0	;	PSIZE	
		CLR	A0						
		MOVE	A0, @	00000	0160h	, 1	;	PMASK	
		MOVE	A0, @	00000	00B0h	, 0	;	CONTRO	L
		MWAIT	; ;	wait	until	data	has 1	been wr	itten

FILL L

B-file registe	ır?		register number:		
I/O register	? 🗸		BSFLTD (32-bit address): BSFLTDL (16-bit address): BSFLTDH (16-bit address):	C000 0320h C000 0320h C000 0330h	
Format	31	F	3SFLTD—32 bits of bus-fault data		0

or

15 0 BSFLTDL—16 LSBs of bus-fault data 0 BSFLTDH—16 MSBs of bus-fault data 0

#### Note:

You can access the bus-fault registers separately or together by using different addresses and different field sizes.

- □ To access BSFLTD as a single 32-bit register, access the 32-bit field at address C000 0320h.
- To access BSFLTDL as a 16-bit register, access the 16-bit field at address C000 0320h.
- □ To access BSFLTDH as a 16-bit register, access the 16-bit field at address C000 0330h.

Description When a bus fault occurs, the TMS34020's memory controller uses BSFLTD to store the data on the LAD bus. When any CPU-initiated memory access returns a bus-fault completion code on the LRDY and BUSFLT pins, the memory controller

- Step 1: Saves the data currently stored on LAD0—LAD31 into BSFLTD.
- Step 2: Signals the CPU that a bus fault occurred. The CPU
  - pushes the current machine state onto the stack,
  - executes a bus fault interrupt routine to clear the cause of the bus fault, and
  - pops the machine state off the system stack to restore the CPU to its previous state.
- Step 3: Uses the data in the BSFLTD registers to restore LAD0—LAD31 and re-executes the memory access that caused the bus fault. If the faulted memory access was a read, the data saved in and restored
from the BSFLTD registers has no significance. However, the memory controller saves and restores the LAD data, regardless of whether the faulted memory access was a read or a write.

Do not write to these registers. When a bus fault occurs, the saved LAD data writes over any data in the BSFLTD registers. If necessary, you can read the contents of the BSFLTD registers during your bus fault interrupt routine.

#### Note:

Although BSFLTDL and BSFLTDH are I/O registers, they are not loaded by a memory write when a bus fault occurs. If external memory shadows these locations, the BSFLTD registers are not copied to external memory.

0

B-file register?		register number:	
I/O register?	$\boxed{\checkmark}$	address: CC	000 02D0h

#### Format

15 bus fault status

Description

When a bus-fault occurs, the TMS34020's memory controller saves its current state into the BSFLTST register. The status information tells the memory controller what type of access triggered the bus fault and marks the point within the access where execution can resume.

When any CPU-initiated memory access returns a bus-fault completion code on the LRDY and BUSFLT pins, the memory controller

- Step 1: Saves its current state into BSFLTST.
- Step 2: Signals to the main processor that a bus fault occurred.
- **Step 3:** After the CPU clears the cause of the bus fault and restores its internal state, the memory controller restores its pre-bus-fault state from the BSFLTST register and re-executes the memory access that caused the bus fault.

Usually, you should not write to the BSFLTST register. When a bus fault occurs, the saved memory controller state writes over any data in the BSFLTST register. If you do not want the TMS34020 to re-execute the faulted memory access, your bus-fault interrupt routine should write the value FFFF<sub>16</sub> to the BSFLTST register. This causes the memory controller to return from the bus fault in an idle state.

#### Note:

Although BSFLTST is an I/O register, it is not loaded by a memory write when a bus fault occurs. If external memory shadows this location, the BSFLTST register is not copied to external memory.

B-file register?	√	register number: B8 address:	)
Format	31	replicated pixel value	
Description	COLOR0 provides a ba in a binary source arra COLOR0 corresponds pixels within COLOR0 word.	ckground color, defining the replacement color for 0 bi by or in the PATTERN register. Pixel alignment with directly to alterable pixels within memory; individu are used as they align with pixels in the destination	ts in al on
	a binary pixel array into	a multiple-bits-per-pixel array.	
	Note: You must replicate the Execution of graphics i	e color information throughout all 32 bits of COLOR0 nstructions does not modify COLOR0.	). 
which instructions use this register?	Instruction	COLOR0's function	
-	FLINE, LINE	Replaces 0s in the PATTERN value	
	FPIXEQ, FPIXNE	Comparison value	
	PFILL XY	Replaces 0s in the PATTERN value	
	PIXBLT B, L	Background pixel color for color-expanded array	
	PIXBLT B, XY	Background pixel color for color-expanded array	
Example	This example is for 4-b COLOR0 register.	it pixels. A pixel value of 5 is replicated throughout the	ıe
	COLORO .set B8	3	
	MOVI 55555555 ; value	h, COLOR0 ; store uniform pixel in COLOR0	

0

(			D
B-file register?	$\checkmark$	register number: B9	
I/O register?		address:	ノ

#### Format

replicated pixel value

#### Description

COLOR1 provides a foreground color, defining the replacement color for 1 bits in a binary source array. Pixel alignment within COLOR1 corresponds directly to alterable pixels within memory; individual pixels within COLOR1 are used as they align with pixels in the destination word.

Binary PIXBLTs use color information in COLOR0 and COLOR1 to transform a binary pixel array into a multiple-bits-per-pixel array. Other graphics instructions use COLOR1 as the replacement color for an alterable destination pixel or for alterable pixels within a pixel block.

#### Note:

31

You must replicate the color information throughout all 32 bits of COLOR1.

Execution of graphics instructions does not modify COLOR1.

Which instructions use this reaister?	Instruction	COLOR1's function
	DRAV	Pixel color for pixel draw
	FILLs (both)	Pixel color for filled array
	FLINE, LINE	Replaces 1s in the PATTERN value
	PFILL XY	Replaces 1s in the PATTERN value
	PIXBLT B, L	Foreground pixel color for color-expanded array
	PIXBLT B, XY	Foreground pixel color for color-expanded array
	TFILL	Pixel color for drawing
	VLCOL	Color-fill data value for VRAM color registers

#### Example

This example is for 4-bit pixels. A pixel value of 3 is replicated through COLOR1 register.

COLOR1 В9 .set

> MOVI 33333333h, COLOR1 ; Store uniform pixel ; value in COLOR1

B-file register?	register number:
I/O register?	address: C000 01A0h

Format

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	n n n n n n n n n n n n n n n n n n n			RR			VEN					CBP	RC	М	BEN

Bits

Bits	Name	Function
0	BEN	Enables big-endian memory addressing
12	RCM	Configures RCA bus address
3	CBP	Enables configuration byte protect
8	VEN	Enables VRAM internal register load
12—10	RR	Selects refresh rate
4—7, 9, 13—15	Print and and and	Reserved; do not use

#### Description

CONFIG controls several system parameters: it selects the memory addressing configuration, informs the TMS34020 that the system contains VRAMs with color-latch and write-mask registers, and selects the DRAM-refresh rate.

#### Note:

Future pin-compatible TMS340x0 devices may use bit 4, providing you with the ability to extend the Q4 phase of certain memory subcycles. This will ease interfacing to DRAMs if the TMS340x0's LCLK frequency is increased above 10 MHz. To ensure compatibility with your existing TMS34020 system, set bit 4 to 1. Setting this bit will not affect the TMS34020.

Before almost any system activity can take place, you must select appropriate values for CONFIG's 3 LSBs. BEN and RCM affect memory addressing; until BEN and RCM have appropriate values, the TMS34020 can successfully access only 32-bit words at memory addresses that have row addresses of all 1s or all 0s.

If the TMS34020 is not powered up in host-present mode, it reads the reset vector from address FFFF FFE0h. Then, before fetching any instructions, the TMS34020 writes the 4 LSBs of the reset vector to the 4 LSBs of CONFIG; this defines the system's memory addressing configuration. You should program the BEN and RCM values into the 3 LSBs of the reset vector; program bit 3 of the reset vector to set the CBP bit. Because the reset vector's row address is all 1s, the TMS34020 can successfully read the



reset vector, regardless of the BEN and RCM values. The TMS34020 assumes that the reset vector is aligned to a 16-bit word, so the values in the reset vector's 4 LSBs do not affect the location from which the TMS34020 starts fetching instructions.

☐ If the TMS34020 is powered up in host-present mode, the host must set BEN and RCM before accessing the TMS34020's local memory.

BEN bit 0

#### **Big-endian memory addressing enable**

BEN	Effect
0	Selects little-endian addressing (default)
1	Selects big-endian addressing

The TMS34020 can use either little- or big-endian addressing conventions. Little-endian is the default (BEN=0). To use big-endian memory addressing, set BEN to 1. For more information about these addressing modes, refer to Section 3.8, <u>Big-Endian and Little-Endian Addressing</u>, on page 3-20.

### RCM0-RCM1

bits 1 & 2

#### **RCA bus configuration mode**

RCM1	RCM0	Base Array Size (CAMD=0)	Logical Address Bits Output on RCA0—RCA12 at Row-Address Time
0	0	64K× <i>n</i>	24 to 12
0	1	256K×n	25 to 13
1	0	1M× <i>n</i>	26 to 14
1	1	$4M \times n$	27 to 15

The RCM bits determine which bits of the logical address are output on RCA0—RCA12 at row-address time. Additionally, the CAMD pin allows the address output at column-address time to be modified on a cycle-by-cycle basis. These capabilities allow you to directly wire DRAMs and VRAMs of more than 1 of the above sizes to the RCA bus in the same system, without using external multiplexing logic. If CAMD is set high during a cycle, most of the bits in the column address are shifted left by 1 bit. However, the logical address bits output on RCA0, RCA11, and RCA12 are not determined by a shift, and vary according to the value of the RCM bits.

RCM1	RCM0	Logical Address Bits Output on RCA0—RCA12 at Column-Address Time with CAMD=1
0	0	23, 22, 13, 12, 11, 10, 9, 8, 7, 6, 5, S, S
0	1	26, 14, 13, 12, 11, 10, 9, 8, 7, 6, 5, S, S
1	0	15, 14, 13, 12, 11, 10, 9, 8, 7, 6, 5, S, S
1	1	28, 14, 13, 12, 11, 10, 9, 8, 7, 6, 5, S, 16
Kow	S is the	16 bit word calact

**Key: S** is the 16-bit word select.

For more information about RCA0—RCA12 address multiplexing, refer to Section 8.16.2 on page 8-53.

CBP bit 3

#### Configuration byte protect

CBP	Effect	
0	LSbyte of CONFIG is not write-protected	
1	Write-protects the LSbyte of CONFIG	

Setting CBP to 1 write-protects the LSbyte of CONFIG (bits 0 to 7). You can set CBP by writing to it or by placing a 1 in bit 3 of the reset vector. During a reset, the TMS34020 automatically copies the 4 LSBs of the reset vector to the 4 LSBs of CONFIG. To clear CBP, reset the TMS34020 with a hardware reset or write a 1 to RST[[HSTCTLH]].

The BEN and RCM bits are extremely important system configuration bits; accidentally writing to them could cause your system to malfunction. When CBP=1, bits 0 to 7 of CONFIG remain write-protected until a reset occurs. Reset is the only operation that can clear CBP. When CBP is set to 1, the byte is write-protected from the next machine state.

## VEN

#### VRAM internal register load enable

VEN	Effect
0	Enables VRAM write-mask load and write with mask
1	Disables VRAM write-mask load and write with mask

The TMS34020 instructions and memory interface support VRAMs with internal write-mask and color registers (such as the TMS44251). Use VEN to inform the TMS34020 that your system's VRAMs support these features.

The VEN bit does not enable or disable execution of VBLT or VFILL instructions. Don't use these instructions if your system's VRAMs do not support the block-write feature.

If VEN=1 and any bit in the PMASK is written, the TMS34020 automatically executes a special load-write-mask memory cycle to load the 1s complement of the 32-bit plane mask into the VRAMs' write masks. This cycle is performed in the next available memory cycle. No further CPU-initiated memory cycles are executed until after the write mask is loaded.

If the TMS34020 subsequently performs a VFILL, VBLT, or pixel write, the plane mask $\neq$  0, and VEN=1, the TMS34020 automatically generates special block-write-with-mask and write-with-mask cycles. This allows selected planes within each pixel to be written without the need for read-modify-write cycles.

R	R0	RR2

bits 10-12

**Refresh** rate

RR2	RR1	RR0	Refreshes scheduled every	
0	0	0	8 machine states	
0	0	1	16 machine states	
0	1	0	32 machine states	
0	1	1	64 machine states	
1	0	0	128 machine states	
1	0	1	256 machine states	
1	1	0	undefined	
1	1	1	DRAM refresh disabled	

The RR bits determine the frequency of DRAM refreshes. An internal counter schedules a DRAM-refresh request at the frequency determined by RR. Each time a DRAM refresh is scheduled, the TMS34020 increments another internal counter to track the number of pending refreshes. (*Pending DRAM refreshes* are refreshes that are requested but not yet performed.) Each time a DRAM refresh is performed, the refresh pending counter is decremented. Note that if a retry terminates a DRAM-refresh cycle, the pending count is not decremented, and the refresh is retried.

A maximum of 15 DRAM refreshes can be pending. If more refreshes are requested, the pending counter overflows and the 16 refreshes are lost (15 pending, plus the refresh that caused the overflow). However, 12 or more pending DRAM refreshes cause DRAM-refresh memory cycles to become one of the highest priority memory cycles, so losing the refreshes should never happen. Realistically, even 12 DRAM refreshes pending are unlikely, because 4 or more DRAM refreshes pending are a higher priority than CPU-initiated memory accesses.

Which instructions use this register?

Any write to the PMASK register while VEN=1 causes the TMS34020 to load the VRAM write-mask registers with the 1s complement of PMASK.



Format

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
CD			PPOP			PBV	PBH	٧	/	T 🕴		nan annannan Mariannan	I	ТМ	

Bits

Bits	Name	Function
0—2	TM	Selects a transparency mode
5	Т	Enables transparency
6—7	W	Selects a window-checking mode
8	PBH	Selects PIXBLT horizontal direction
9	PBV	Selects PIXBLT vertical direction
10—14	PPOP	Selects a pixel-processing operation
15	CD	Disables cache
3—4		Reserved; do not use

Description

The CONTROL register controls several aspects of CPU instruction execution and of the memory interface. You can access this register at two addresses: at address C000 00B0h for compatibility with the TMS34010, and at C000 0190h so that all I/O registers used as implied operands are in a contiguous block (C000 0130h to C000 0190h). If you write to one location, both are affected.

TM bits 0---2

#### Transparency mode select

TM2	TM1	TM0	Description
0	0	0	Transparency on result equal 0
0	0	1	Transparency on source equal COLOR0
1	0	0	Transparency on result equal 0
1	Ó	1	Transparency on destination equal COLOR0
0	1	0	
0	1	1	Deserved
1	1	0	neservea
1	1	. 1	

When transparency is enabled, the TM bits select the transparency mode. You can enable transparency by setting the T bit to 1.

#### **Pixel transparency enable**

T bit 5

Т	Description	
0	Disables transparency	
1	Enables transparency	

The T bit enables or disables pixel transparency. When transparency is enabled, the TMS34020 inhibits overwriting of a transparent pixel (as determined by the current transparency mode).

w bits 6&7

> рвн bit 8

#### Window checking

WO	Description
0	No pixel writes are inhibited, and no interrupt requests are generated
1	Generate interrupt request on attempt to write to pixel lying inside win- dow and inhibit all pixel writes
0	Generate interrupt request on attempt to write to pixel lying outside win- dow
1	Inhibit pixel writes outside window, but do not request interrupt
	<b>W0</b> 0 1 0

The W bits select the action the TMS34020 takes when a pixel operation would cause the TMS34020 to write a pixel to a location lying either inside or outside specified window limits. Window checking applies to attempts to write to pixel locations defined by XY addresses only. Window checking affects neither non-pixel data writes nor writes to pixel locations defined by linear memory addresses.

A request for a window violation interrupt can occur when  $W=01_2$  or  $W=10_2$ . WVP[INTPEND] is set to 1 to indicate that a window violation occurred. This in turn interrupts the TMS34020, if both WVE[INTENB] and IE[ST] equal 1.

## PIXBLT horizontal direction

PBH	Description
0	Increment in the X direction (move from left to right)
1	Decrement in the X direction (move from right to left)

The PBH bit determines the horizontal direction (increasing or decreasing X) of pixel processing for these instructions:

PIXBLT XY,XY	PIXBLT L, XY
PIXBLT XY, L	PIXBLT L, L
PIXBLT L, M, L	

PIXBLT	vertical	direction	
--------	----------	-----------	--

**Pixel processing operation** 

рвv bit 9

PBV	Description (assuming default screen origin)	
0	Increment in the Y direction (move from top to bottom)	
1	Decrement in the Y direction (move from bottom to top)	

The PBV bit determines the vertical direction (increasing or decreasing Y) of pixel processing for these instructions:

PIXBLT XY,XY	PIXBLT L, XY
PIXBLT XY, L	PIXBLT L, L
PIXBLT L, M, L	

PPOP

#### bits 10—14

The PPOP bits define the manner in which a source pixel is combined with a destination pixel during a pixel operation. The following 16 PPOP codes perform Boolean operations on pixels of 1, 2, 4, 8, 16, and 32 bits.

PPOP:	4	3	2	1	0	Operation	Description
	0	0	0	0	0	S→D	Source replaces destination
	0	0	0	0	1	S AND $D \rightarrow D$	AND source with destination
	0	0	0	1	0	S AND $\overline{D} \rightarrow D$	AND source with NOT(destination)
	0	0	0	1	1	0→D	Os replace destination
	0	0	1	0	0	S OR D→D	OR source with NOT(destination)
	0	0	1	0	1	S XNOR $D \rightarrow D$	XNOR source with destination
	0	0	1	1	0	D→D	Invert destination
	0	0	1	1	1	S NOR D→D	NOR source with destination
	0	1	0	0	0	S OR D→D	OR source with destination
	0	1	0	0	1	D→D	Do not change destination (note, however, that memory cycles still occur)
	0	1	0	1	0	S XOR D→D	XOR source with destination
	0	1	0	1	1	S AND D→D	AND NOT(source) with destination
	0	1	1	0	0	1 <i>→</i> D	1s replace destination
	0	1	1	0	1	S OR D→D	OR NOT(source) with destination
	0	1	1	1	0	S NAND D→D	NAND the source and destination
	0	1	1	1	1	S→D	NOT (source) replaces destination

These PPOP codes perform arithmetic operations on 2, 4, 8, 16, and 32-bit pixels (but not on 1-bit pixels).

PPOP:	4	3	2	1	0	Operation	Description
	1	0	0	0	0	S + D→D	Add source to destination
	1	0	0	0	1	S ADDS $D \rightarrow D$	Add source to destination with saturation
	1	0	0	1	0	D – S→D	Subtract source from destination
	1	0	0	1	1	D SUBS S→D	Subtract source from destination with satu- ration
	1	0	1	0	0	S MAX D→D	Replace destination with maximum of source and destination
	1	0	1	0	1	S MIN D→D	Replace destination with minimum of source and destination

**Note:** PPOP codes 10110<sub>2</sub> through 11111<sub>2</sub> are reserved.

C	υ	
bit	1	5

#### **Cache disable**

5			

CD	Description	
0	Enables instruction cache	
1	Disables instruction cache	

CD enables or disables the instruction cache. When the cache is disabled, cache contents (including data, P flags, SSA registers, etc.) are not disturbed, and all instructions are fetched from memory, not the cache. When the cache is re-enabled, its previous state (before it was disabled) is restored, and the instructions retained within the cache are once again available for execution.

Instruction	Bits used			
all instructions	CD			
DRAV	PPOP	T & TM	W	
FILL L	PPOP	T & TM		
FILL XY	PPOP	T & TM	W	
FLINE	PPOP	T & TM		
LINE	PPOP	T & TM	W	
PIXBLT B, L	PPOP	T & TM		
PIXBLT B, XY	PPOP	T & TM	W	
PIXBLT L, L	PPOP	T & TM		PBH & PBV
PIXBLT L, XY	PPOP	T & TM	W	PBH & PBV
PIXBLT XY, XY	PPOP	T & TM	W	PBH & PBV
PIXBLT XY, L	PPOP	T & TM		PBH & PBV
PIXT <i>Rs</i> , * <i>Rd</i>	PPOP	T & TM		
PIXT * <i>Rs</i> , * <i>Rd</i>	PPOP	T & TM		
PIXT <i>Rs</i> , * <i>Rd<b>.XY</b></i>	PPOP	T & TM	W	
TFILL	PPOP	T & TM	W	

# Which instructions use this register?

B-file register?	register number:
I/O register? $$	CONVDP address: C000 0140h
	CONVMP address: C000 0180h

Format		15	0
	CONVDP	conversion factor for XY destination address	
		15	0
	CONVSP	conversion factor for XY source address	
		15	0
	CONVMP	conversion factor for mask value	

Description

CONVDP, CONVSP, and CONVMP are 16-bit registers that contain control parameters used during execution of a pixel operation. The TMS34020 uses CONVDP and CONVSP with

- XY addressing,
- window clipping, and
- □ FILLs or PIXBLTs (except for PIXBLT L,L) that process pixels from the bottom of the array to the top (PBV=1).

The TMS34020 uses CONVMP for XY addressing (CVMXYL).

Each conversion factor register is associated with an appropriate pitch register; each CONVxP register associated with an instruction that loads the conversion factor into CONVxP according to the pitch value in xPTCH.

Conversion Factor Register (CONVxP)	Associated Pitch Register (xPTCH)	Associated Instruction
CONVDP	DPTCH	SETCDP
CONVSP	SPTCH	SETCSP
CONVMP	MPTCH	SETCMP

TMS34020 internal hardware uses the CONVDP and CONVSP values when converts an XY destination or source address, respectively, to a linear address.

- PIXBLT and FILL instructions with an XY destination use DPTCH and CONVDP to convert the XY coordinates to a linear address before the pixel transfer begins.
- PIXBLT instructions with an XY source address use the SPTCH and CON-VSP values to convert the XY coordinates to a linear memory address before beginning the pixel transfer.

If a PIXBLT or FILL requires preclipping of the destination array in the Y direction, the TMS34020 uses CONVDP to calculate the effect of the clipped starting Y coordinate on the destination array's starting linear address. For PIXBLTs, the starting source address is modified to accommodate the resulting changes to the starting destination address. When a PIXBLT instruction's starting Y coordinate lies in either of the 2 lower corners of the destination array (PBV=1), the TMS34020 uses CONVDP and CONVSP to calculate the linear addresses corresponding to the specified starting coordinates.

CONVxP contains 1 of 3 types of values, depending on the value in the associated pitch register:

If the xPTCH register =	then
a power of 2	The 5 LSBs of CONVxP contain the 1s complement of $log_2(xPTCH)$ . During XY-to-linear conversion, the product of the Y value and the pitch is calculated by shifting Y left by $log_2(xPTCH)$ .
two powers of 2	CONVxP contains 2 conversion values. The 5 LSBs of CONVxP should contain the 1s complement of $\log_2$ of the greater of the powers of 2, and the 5 LSBs of the upper byte contain the 1s complement of $\log_2$ of the lesser of the powers of 2. During conversion, the product of the Y value and the pitch is calculated by adding Y shifted left by each of the 2 conversion factors.
arbitrary pitch	The LSbyte of CONVxP contains 0s. The TMS34020 must multiply the address by xPTCH. This is a 16-by-32-bit signed multiply in which only the 32 LSBs of the result are retained.

Which instructions use these registers?	CONVDP	CONVSP	CONVMP
	CVXYL	CVSXYL	CVMXYL
	CVDXYL	PIXBLT L, XY	SETCMP
	DRAV	PIXBLT XY, L	
	FILL XY	PIXBLT XY, XY	
	FLINE, LINE	PIXT * <i>Rs.XY</i> , <i>Rd</i>	
	PIXBLT B, XY	PIXT *Rs.XY, *Rd.XY	
	PIXBLT L, XY	SETCSP	
	PIXBLT XY, L		
	PIXBLT XY, XY		
	PIXT Rs, *Rd.XY		
	PIXT *Rs.XY, *Rd.XY		
	SETCDP		
	TFILL		

For more information about array pitches and XY-to-linear conversion, refer to Section 12.12, <u>Converting an XY Address to a Linear Address</u>, on page 12-47.

B-file register?	√	register number: B2 address:
<i>Format</i> or	31 Y half of 31	16    15    0      an XY address    X half of an XY address    0      Integration of the second s
Description	DADDR contain FLINE. DADDR nation array. Wh TMS34020 auto ner of the desti adjust DADDR f Some instructio function (FILL X with window opt ing XY address destination array and the window are undefined. The TMS34020 address, depen If DADDR conta sponding linear or FILL, DADDI pletes, DADDR row in the array of the next point next word of pix	s the destination array address for PIXBLTs, FILLs, LINE, and usually points to the pixel with the lowest address in the desti- tion the selected starting corner is not the upper left corner, the matically adjusts DADDR to point to the selected starting cor- nation array. (For PIXBLT L,L, however, you must manually to point to the starting corner.) ns use DADDR with DYDX to perform a common rectangle Y, PFILL XY, PIXBLT B,XY, PIXBLT L,XY, and PIXBLT XY,XY, ion 1). In these cases, the TMS34020 sets DADDR to the start- of the rectangle that represents the intersection of the original y and the clipping window. No drawing is performed. If the array do not intersect, the V bit is not set and the contents of DADDR treats the address in DADDR as an XY address or a linear ding on the instruction you use. ains an XY address, the instruction converts it to the corre- address before beginning the pixel transfer. During a PIXBLT R is maintained in linear format. When the instruction com- points to the linear starting address of the row following the last (for LINE, FLINE, and VFILL, DADDR contains the address is on the line). If a PIXBLT is interrupted, DADDR points to the els to be read.
Which instructions use this register?	Instruction	DADDR's format and function
	BLMOVE	Linear; points to the beginning of the destination array
	CLIP	XY; points to the beginning of the destination array

CLIP	At, points to the beginning of the destination array
FILL L	Linear; points to the beginning of the destination array
FILL XY	XY; points to the beginning of the destination array
FLINE	Linear; starting point for the line
LINE	XY; starting point for the line
PFILL XY	XY; points to the beginning of the destination array
PIXBLT B, L	Linear; points to the beginning of the destination array
PIXBLT B, XY	XY; points to the beginning of the destination array

Instruction	DADDR's format and function
PIXBLT L, L	Linear with special requirements when PBH=1 or PBV=1; refer to PIXBLT L,L for a description of its unique require- ments
PIXBLT L, XY	XY; points to the beginning of the destination array
PIXBLT XY, L	Linear; points to the beginning of the destination array
PIXBLT XY, XY	XY; points to the beginning of the destination array
PIXBLT L,M,L	Linear; points to the beginning of the destination array
TFILL	XY; instruction uses this to hold temporary values
VBLT	Linear; points to the beginning of the destination array
VFILL	Linear; points to the beginning of the destination array

Example

DADDR .set B2

> [0008h, 0015h], DADDR ; Move XY value MOVI ; Move linear MOVI 00010AFCh, DADDR ; value 10AFCh

; 15h,8h into B2 ; into B2

4-31

B-file register?		register number:	
I/O register?	DINC ( DINCL ( DINCH (	32-bit address): 16-bit address): 16-bit address):	C000 0240h C000 0240h C000 0250h



Bits

Bits	Name	Function	
0-4	YZINC	Y-zoom increment value	
5—31	SRINC	Screen-refresh address increment value	· .

#### Note:

You can access the display increment registers separately or together by using different addresses and different field sizes.

- □ To access DINC as a single 32-bit register, access the 32-bit field at address C000 0240h.
- To access DINCL as a 16-bit register, access the 16-bit field at address C000 0240h.
- To access DINCH as a 16-bit register, access the 16-bit field at address C000 0250h.

Description

The DINC registers contain two increment values. One controls the TMS34020's Y-zoom feature; the other is used for screen refreshes.

Y	ZINC	

#### In a way way and sealess

bits 0-4

1-20	DOIII	ппс	rem	ent	value

	Y	ZIN	С		Zoom	
4	3	2	1	0	Factor	Description
0	0	0	0	0	1	No repetition of scan lines
1	0	0	0	0	2	Repeat scan line 2 times
0	1	0	0	0	4	Repeat scan line 4 times
0	0	1	0	0	8	Repeat scan line 8 times
0	0	0	1	0	16	Repeat scan line 16 times
0	0	0	0	1	32	Repeat scan line 32 times

If you want to change the value in YZINC when video is enabled (ENV[[DPYCTL]]=1), you should also clear YZCNT[[DPYNX]] to 0.

#### SRINC bits 5----31

#### **Display increment value**

The 27-bit SRINC value specifies the amount by which the address stored in SRNX[DPYNX] should be incremented following completion of each horizontal-blanking screen-refresh cycle. This value corresponds to the display pitch. If you are using the Y-zoom feature, the TMS34020 will not increment SRNX after each horizontal-blanking screen refresh. Instead, it will increment SRNX after every *n*th screen refresh if  $zoom \times n$  is selected.

For both interlaced and noninterlaced video, load SRINC with the display pitch. In interlaced video, the SRNX [DPYNX] registers are automatically incremented by 2xSRINC to account for the fact that in any field (odd or even), only alternate lines are displayed.

The bits of DPYNX and DPYST that correspond to the column and row addresses actually latched into the VRAMs vary from system to system. The bits of SRINC that contain the display pitch depend on the alignment of the address in SRNX [DPYNX] and SRST [DPYST]. SRINC is usually a power of 2 or a sum of two powers of 2, but can be any arbitrary value required.

B-file register?		register number: B3 address:
Format	31	0
		linear bit address
Description	DPTCH defines the linear diffe cent rows of a destination arra move from row to row through ue that is a multiple of the curr is most efficient when DPTCH	erence in the starting memory addresses of adja- ay. The TMS34020 uses the value in DPTCH to the destination array. DPTCH can have any val- rent pixel size. Note that XY-to-linear conversion H is a power of 2.
	If you're manually converting the SETCDP instruction. SET tination pitch conversion facto contents of CONVDP are the instructions; these instruction	an XY address to a linear address, you can use CDP uses the DPTCH value to calculate the des- or and loads the correct value into CONVDP. The en available for use by the CVXYL or CVDXYL as perform the conversion.
Which instructions use this register?	Instruction	DPTCH's format and valid values
	CVXYL	Linear; any value
	CVDXYL	Linear; any value
	DRAV	Linear; any value
	FILLs (both)	Linear; any value
	FLINE, LINE	Linear; any value
	PFILL XY	Linear; any value
	PIXBLTs (all)	Linear; any value
	PIXT <i>Rs</i> , * <i>Rd</i> .XY	Linear; any value
	PIXT * <i>Rs<b>.XY</b>, *<i>Rd.<b>XY</b></i></i>	Linear; any value
	SETCDP	Linear; any value
	TFILL	Linear; any value
	VBLT	Linear; any value
	VFILL	Linear; any value
Example	DPTCH .set B3	
	MOVI 00001000 MOVI 00010AFCI MOVI 00000180	h, DPTCH ; Power of 2 h, DPTCH ; Arbitrary value h, DPTCH ; 2 powers of 2 ; (128 + 256)

	register number:	
I/O register? √	address:	C000 01E0h
15		
	1/O register? √	I/O register?    I/O address:      15    no defined function for TMS34020

Description

DPYADR is a simple 16-bit read/write location that is included for compatibility with the TMS34010. The TMS34010 used DPYADR as the source of the row and column addresses output during screen-refresh cycles. The TMS34020 uses a different register for this purpose and assigns no function to DPYADR.



Format

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
ENV	NIL	INTERNET CONTRACTOR	SRE	CST				VCD	SSV			CVD	CSD	VSD	HSD

Bits

Bits	Name	Function
0	HSD	Selects the horizontal-sync direction
1	VSD	Selects the vertical-sync direction
2	CSD	Selects the composite-sync direction
3	CVD	Disables composite video
6	SSV	Enables split-serial-register midline reload
7	VCE	Enables video capture
11	CST	Enables CPU serial-register transfers
12	SRE	Enables screen refreshes
14	NIL	Enables noninterlaced video
15	ENV	Enables video
4—5 8—10		Reserved; do not use

#### Description

DPYCTL contains several parameters that control video timing.

HSD bit 0

#### Horizontal-sync direction

When HSD =	HSYNC is an	
0	Input	
1	Output	

The HSD bit controls the direction (input or output) of the HSYNC signal.

❑ When HSD=0, HSYNC is an input. The TMS34020's internal video timing logic synchronizes to external pulses applied to HSYNC. Whenever the TMS34020 detects the start of an external horizontal-sync pulse on HSYNC, it loads HCOUNT from SETHCNT. The internal horizontal- and composite-sync intervals begin if they were not already started.

❑ When HSD=1, HSYNC is an output and is controlled according to the values in the video timing registers.

#### Vertical-sync direction

٧S	D	
bit	1	

When VSD =	VSYNC is an	· · · · · · · · · · · · · · · · · · ·
0	Input	
1	Output	

VSD controls the direction (input or output) of the VSYNC signal.

- When VSD=0, VSYNC is an input. The TMS34020's internal video timing logic synchronizes to pulses that an external source applies to VSYNC. Whenever the TMS34020 detects the start of an external vertical-sync pulse input on VSYNC, the TMS34020 loads VCOUNT from SETVCNT. The internal vertical-sync interval begins if it was not already started by the internal video timing logic. Enabling noninterlaced video (NIL=1) also loads HCOUNT from SETHCNT. The internal horizontal- and composite-sync intervals begin if they were not already started by the internal video timing logic.
- When VSD=1, VSYNC is an output and is controlled according to the values in the video timing registers.

CVD	CSD	Status of CSYNC/HBLNK				
0	0	CSYNC is an input				
0	1	CSYNC is an output				
1	0	undefined				
1	1	HBLNK is an output				

CSD controls the direction (input or output) of the CSYNC/HBLNK pin when it is configured as CSYNC (CVD=0).

- If CVD=1, the pin is configured as HBLNK, and CSD must be 1. When CSD=1, CSYNC (CVD=0) or HBLNK (CVD=1) is an output and is controlled according to values in the video timing registers.
- If CSD=0, CSYNC is an input and the TMS34020's internal video timing logic synchronizes to external pulses applied to CSYNC. Whenever the TMS34020 detects the start of an external composite-sync pulse input on CSYNC, the TMS34020 loads HCOUNT from SETHCNT. The internal composite-sync interval begins if it was not already started by the internal video timing logic. Normally, the internal horizontal-sync interval also begins if it has not already started. However, in interlaced video (NIL=0), external composite-sync pulses occur every half horizontal scan line during the equalization and serration regions of vertical blanking, so the internal horizontal-sync interval is started by alternate external composite-sync pulses at these times. The first serration pulse input on CSYNC also loads VCOUNT from SETVCNT, and the internal vertical-sync interval begins if it was not already started by the internal video timing logic.



#### **Composite-sync direction**

bit 2

CVD bit 3

#### 

Comp	osne	viaeo	aisable
-			

	Status of CSYNC/HBLNK	Status of CBLNK/VBLNK
CVD=0	Selects CSYNC	Selects CBLNK
CVD=1	Selects HBLNK	Selects VBLNK

CVD controls the functions of the CSYNC/HBLNK and CBLNK/VBLNK pins. Because both composite and separate synchronization and blanking signals are internal, CVD simply selects which of these functions is visible at the pins.

SSV bit 6

#### Split-serial-register midline reload enable

SRE	SSV	Effect	
0	0	Disables split-serial-register midline reload	
0	1	Disables split-serial-register midline reload	
1	0	Disables split-serial-register midline reload	
1	1	Enables split-serial-register midline reload	

SSV determines whether or not the TMS34020 performs screen-refresh cycles for VRAMs with split-serial registers during the active display time. SSV works in conjunction with the screen-refresh enable bit (SRE). If SSV=1 and screen refreshes are enabled (SRE=1), the TMS34020 performs an ordinary screen-refresh (memory-to-register) cycle during horizontal blanking, which

- Step 1: reloads an entire row of VRAM memory into the VRAM serial registers.
- **Step 2:** updates the address in SRNX[DPYNX] for the next screen refresh, and
- Step 3: reloads the SCOUNT register with the tap point of the current screenrefresh address, using the mask in the DPYMSK register.

This is immediately followed by a split register-to-memory cycle, which

- **Step 4:** reloads the half serial registers that do not contain the current tap point so that they contain data for the next half line to be displayed,
- **Step 5:** and initializes the VRAM for split-serial-register operation.

After blanking ends, SCLK starts shifting data from the VRAMs and increments SCOUNT (which tracks the VRAM tap point). When SCOUNT overflows from a tap point of all 1s to all 0s, this indicates that the VRAMs have switched from one half serial register to the other. A split register-to-memory cycle (midlinereload cycle) is executed, performing Step 4.

#### Note:

You must provide an SCLK pulse to the VRAMs between these two screenrefresh cycles to ensure that the tap-point address is latched correctly.

#### Video capture enable

٧C	Ε
bit	7

	· · · · · · · · · · · · · · · · · · ·	
VCE	Effect	
0	Selects memory-to-register screen-refresh cycles	
1	Selects register-to-memory screen-refresh cycles	

VCE determines whether TMS34020 screen-refresh cycles are memory-toregister cycles or register-to-memory cycles. VCE affects only those memory cycles that are initiated by the TMS34020's video timing logic.

When VCE=1, screen-refresh cycles initiated by the video timing logic are performed as screen-capture cycles; data shifted into the VRAM serial registers is transferred to the specified row of VRAM ready for the next line.

Do not use midline reload in systems with video capture; clear SSV to 0. VRAMs support only the transfer of an *entire* serial register's contents into the specified memory row. If you used midline reload to condense the display memory into a contiguous region of VRAM, it would be necessary to transfer only some of the bits of the serial register into the memory array in order to not overwrite previously captured data. This is not possible.

сsт bit 11

#### **CPU serial-register transfer enable**

CST	Effect
0	Pixel-access cycles occur normally
1	Converts pixel-access cycles into VRAM serial-register-transfer cycles

CST converts an ordinary pixel access into a VRAM serial-register transfer cycle. Several of the TMS34020's graphic instructions treat data as pixels.

By default, CST=0 and accesses of pixel data are normal read and write cycles. When CST=1, however, pixel accesses are converted to serial-register-transfer cycles:

A pixel read cycle becomes a memory-to-register cycle.

A pixel write cycle becomes a register-to-memory cycle.

This register-transfer cycle is performed under explicit program control, as opposed to the screen-refresh cycles enabled by the SRE bit, which are automatically generated at regular intervals.

CST is useful for bulk initialization of an entire VRAM array. You can clear the entire screen to a specified background color in only 256 memory cycles for  $64K \times n$  VRAMs, or 512 memory cycles for  $256K \times n$  VRAMs (where *n* is the number of planes within the VRAM). (Note that the TMS4461 and TMS44251 have this capability, but not all VRAMs support this function.) The CST bit affects only *pixel* accesses; it does not affect instruction fetches or nonpixel accesses.

#### Screen-refresh enable

bit 12

-	-	-	-	-	-	•	-	 -	-	 •	 	-	•	
	-	-	-											

ENV	SRE	Effect	ENV	SRE	Effect
0	0	Disables screen refresh	1	0	Disables screen refresh, but tracks the address
0	1	Disables screen refresh	1	1	Enables screen refresh

SRE enables automatic screen refreshing. Screen refreshes are performed by means of the VRAM memory-to-register cycles, which the TMS34020 performs automatically during each horizontal-blanking interval. DPYST, DINC, and DPYCTL control generation of addresses output during these cycles. If ENV=1, the TMS34020 continues to generate the screen-refresh address internally, even if SRE=0. This allows an external source to insert images into the display; during each horizontal-blanking period, the TMS34020 continues to track the address of the image hidden beneath the external image. Thus, the TMS34020 can restart screen refreshes after inserting the image without adjusting the address to account for the undisplayed lines.

Changing SRE's value affects screen refreshes, starting with the next horizontal-blanking period—or, if SSV=1 and SCLK is running, starting with the next time the VRAMs change active half serial registers, whichever comes first. Normally, however, SCLK does not shift data to the screen when screen refreshes are disabled.

Noninterlaced video enable

NIL	Effect		
0	Selects interlaced video timing		
. 1	Selects noninterlaced video timing	 	

NIL selects between an interlaced or a noninterlaced display. The TMS34020 modifies its video timing output signals according to NIL's value. Chapter 9 describes the timing differences between interlaced and noninterlaced video.

#### Enable video

ENV	Effect	
0	Blanks the entire video screen	
1	Enables the video display	

ENV enables or disables the video display.

When ENV=0, the display remains blanked. The signal output at CBLNK/ VBLNK (and at CSYNC/HBLNK if CVD=1) is forced to remain at active low throughout the frame, inhibiting the display interrupt. (DIP[INTPEND] can't be set. If DIP is already set when ENV changes from 1 to 0, it remains set until you explicitly clear it.)

When ENV=1, the video display is enabled. The output signals are controlled according to the parameters in the video timing registers, and DIP is set when VCOUNT becomes equal to DPYINT.

NIL bit 14

ENV bit 15

SRE

0

B-file register?		register number:	
I/O register?	$\checkmark$	address: C000 00A0h	,
			٧

#### Format

scan line address (for display interrupt)

Description

15

DPYINT identifies the next scan line (in some cases, the next half scan line) at which a display interrupt will be requested. DPYINT helps to coordinate software activity with the refreshing of a selected horizontal scan line on the screen.

The video timing logic compares the contents of DPYINT to VCOUNT. This usually coincides with the start of the horizontal-blanking interval that marks the end of the line designated by the value in DPYINT. If interlaced video is enabled (NIL=0), then during the part of the vertical-blanking interval when VCOUNT is incremented every half line, DPYINT is compared to VCOUNT just before VCOUNT is incremented, at the end and in the center of each horizontal scan line. When VCOUNT=DPYINT, a display interrupt is requested and DIP[INTPEND] is set to 1.

For split-screen applications, you can load a new value into the SRNX[[DPYNX]] bits, immediately following detection of the 0-to-1 transition of DIP. The new SRNX value does not affect the line that follows the current horizontal-blanking interval, but affects the next line. A screen-refresh cycle will be scheduled to occur at the start of the same horizontal-blanking period in which DIP is set. At the end of the screen-refresh memory cycle, the screen-refresh address in SRNX is automatically incremented. Requests for screen-refresh cycles have a higher priority than CPU requests. Thus, if the CPU loads a new value into SRNX immediately after setting the DIP bit, SRNX will not actually be modified until after the screen-refresh cycle completes and the existing contents are incremented. This new address becomes the address used in the next screen refresh. SRNX can change only during the scan line under explicit program control. The display interrupt is disabled when ENV[[DPYCTL]] is 0.

B-file register?	register number:
I/O register?  √	<i>DPYNX</i> (32-bit address): C000 0220h <i>DPYNXL</i> (16-bit address): C000 0220h
	DPYNXH (16-bit address): C000 0230h



Bits

Bits	Name	Function	
0—4	YZCNT	Y-zoom count	
5—31	SRNX	Next screen-refresh address	

#### Note:

You can access the display next-address registers separately or together by using different addresses and different field sizes.

- To access DPYNX as a single 32-bit register, access the 32-bit field at address C000 0220h.
- To access DPYNXL as a 16-bit register, access the 16-bit field at address C000 0220h.
- To access DPYNXH as a 16-bit register, access the 16-bit field at address C000 0230h.

#### Description

The DPYNX registers contain two values. One is used for the Y-zoom feature; the other is an address that is output during a screen-refresh cycle.

#### YZCNT bits 0---4

#### Y-zoom increment value

The 5-bit YZCNT value determines when the SRNX address can be incremented by SRINC [[DINC]]. After every local-memory screen-refresh cycle, the TMS34020 increments YZCNT by the value of YZINC [[DINC]]. If YZCNT=0 before it is incremented, SRNX is incremented at the same time. If YZCNT $\neq$ 0, SRNX is not incremented, so the next scan line contains the same pixels as the current scan line. This allows the image on the screen to be magnified (or zoomed) in the Y direction. The value of YZINC determines how many times the scan line is output, and thus determines the zoom factor. YZCNT will equal 0 when

 $n \times YZINC modulo 32 = 0$ 

(*n* is the Y-zoom factor). This occurs once every *n* scan lines. During each vertical-blanking interval, YZCNT is reset to YZINC.

#### Next screen-refresh address

SRNX bits 5---31

The 27-bit SRNX value represents the long-word address that is output during a screen-refresh cycle. When YZCNT=0, the TMS34020 increments SRNX (by SRINC[[DINC]]) after each screen-refresh cycle.

SRNX consists of a row-address portion and a column-address portion, corresponding to the bits of the address connected to the VRAMs at row- and column-address times on RCA0—RCA12. The column- and row-address fields should be contiguous to one another within SRNX. However, you can choose where the two fields are placed within the 27 bits of SRNX, provided that all of the row-address bits are output on RCA0—RCA12 at row-address time, and all the column-address bits are output on RCA0—RCA12 at column-address time. Section 8.16.2 (page 8-53) details which bits of the logical address are output on RCA0—RCA12 at row- and column-address times.

B-file register? I/O register?	register number:   √   address: C000 02E0h
Format	150 display mask
Description	DPYMSK is used when midline-reload screen refreshes are enabled (SSV[DPYCTL]=1). DPYMSK defines which bits of the address in SRST[DPYST] and SRNX[DPYNX] correspond to the tap-point portion of the address output during screen-refresh cycles. DPYMSK is loaded with a field of contiguous 1s to indicate where the tap point is within SRST and SRNX. This information is then used to perform these functions:
	Isolate the tap point from the 27-bit, long-word screen-refresh address so that it can be loaded into SCOUNT (the counter register that tracks the VRAM tap point and schedules midline-reload split-serial-register screen-refresh cycles during the active portion of the display).
	Determine which bit of the 27-bit, long-word address should be increm- ented so that the address output during a midline-reload memory cycle is the address of the next half-row of VRAM.
	DPYMSK maps to the 16 LSBs of SRST and SRNX, which correspond to bits 5 to 20 of DPYST and DPYNX, respectively. There are two reasons for this skewed mapping:
	Bit 5 of DPYNX is the least significant address bit output during screen-re- fresh memory cycles.
	If the mapping were not skewed, a 32-bit DPYMSK register would be re- quired to determine which bits in DPYNXH and DPYSTH were part of the tap point.
Figure 4–6. How DI	PYMSK Maps to the Logical Screen-Refresh Address



The tap point's LSB does not have to be in bit 5 of the logical address. This allows for some bank selection bits at the least significant end of the address.

The bits from the logical address identified by DPYMSK to be the tap point are automatically shifted right by the number of 0s at the least significant end of DPYMSK before being loaded into SCOUNT.

The number of contiguous 1s in DPYMSK depends on the number of address bits needed to specify the tap-point address for the VRAMs. This field is either right-justified in DPYMSK, or it is preceded by several 0s if there are interleaved banks addressed with bits of the logical address less significant than the tap point. The remaining bits at the most significant end of DPYMSK should be 0s.

During a midline-reload screen-refresh cycle, the address output by the TMS34020 is that of the next half-row of VRAM to be displayed. This address has a 0 tap point and is stored in an inaccessible register. This address must then be incremented to point to the next half-row after that, as there may be multiple midline reloads on any one horizontal scan line. The address is incremented at the next bit position to the left of the leftmost 1 in DPYMSK. The most significant 1 in DPYMSK must not be any higher than bit 10 or lower than bit 7 of DPYMSK.

Figure 4–7. The Functions of the Different Fields of DPYMSK



Because the serial registers in the VRAMs are split into two parts, the tap point for each VRAM needs to address only enough bits for half a row. This means, for instance, that for a 256K×4 VRAM (the smallest available with the split-serial-register), the tap point mask should be 8 contiguous 1s despite the fact that a 256K×4 VRAM or DRAM requires 9 row- and column-address bits; when considering the split-serial register, the MSB of the column address is not part of the tap point, but selects between the upper and lower serial register halves. Similarly,  $1M \times n$  VRAMs should have a 9-bit tap point mask.





Bits

Bits	Name	Function
04	HURAN AND AND AND AND	Reserved; do not use
5—31	SRST	Screen-refresh start address

#### Note:

You can access the display start address registers separately or together by using different addresses and different field sizes.

- To access DPYST as a single 32-bit register, access the 32-bit field at address C000 0200h.
- To access DPYSTL as a 16-bit register, access the 16-bit field at address C000 0200h.
- To access DPYSTH as a 16-bit register, access the 16-bit field at address C000 0210h.

Description

The 27-bit SRST value represents the address that points to the pixel at the left of the first line displayed on the screen. This address is used in calculating the screen-refresh address output just before the start of each frame (or field in interlaced video). There are a number of cases to consider.

- In noninterlaced video, the address output at the beginning of each field is simply that contained in SRST, so SRST is copied into SRNX[DPYNX] at the beginning of each vertical-blanking period.
- In interlaced video,
  - At the beginning of the even field, the address of the first pixel displayed is that of the pixel half way across the first line. So,

SRINC[[DINC]]/2 is added to SRST before the address is loaded into SRNX[[DPYNX]] at the start of the vertical-blanking interval.

- At the beginning of the odd field, the address of the first pixel displayed is that of the first pixel on the second line. So, SRST is added to SRINC[[DINC]] before being loaded into SRNX[[DPYNX]] at the start of the vertical-blanking interval.
- The address output at the beginning of the second line of the even field in interlaced video is that of the first pixel on the third line of the display. Normally, the address is generated by adding SRINC[[DINC]] or, in interlaced video, 2×SRINC to the value in SRNX[[DPYNX]]. However, in this particular instance ,SRNX contains the address of the pixel half way across the first line of the display, and so would need 1.5×SRINC added to it to arrive at the correct address. Because of this, the value of SRNX generated after the first screen refresh in the even field is generated by adding 2×SRINC to SRST.

The address consists of a row-address portion and a column-address portion, corresponding to the bits of the address connected to the VRAMs at row- and column-address times, respectively, on the RCA0—RCA12 bus. The column-and row-address fields should be contiguous to one another within the 27 bits of SRNX. However, you can choose where the two fields are placed within the 32-bit register, provided that all of the row-address bits are output on RCA0—RCA12 at row-address time, and all the column- address bits are output on RCA0—RCA12 at column-address time. Section 8.16.2 (page 8-53) details which bits of the logical address are output on RCA0—RCA12 at row-and column-address times.

15



B-file register?	registe	er number:	
I/O register?	<u>र</u>	address: C0	00 0090h

#### Format

no defined function for TMS34020

Description

DPYSTRT is a simple 16-bit read/write location that is included for compatibility with the TMS34010. The TMS34010 used DPYSTRT to indicate the address of the first pixel to be displayed in each frame. The TMS34020 uses a different register for this purpose and assigns no function to DPYSTRT.

0

	į
S∎,	

B-file register?	register number:		
I/O register?  √	address:	C000 01B0	h

Format

no defined function for TMS34020

Description

15

DPYTAP is a simple 16-bit read/write location that is included for compatibility with the TMS34010. The TMS34010 used DPYTAP to indicate the VRAM tap point used during screen-refresh cycles. The TMS34020 uses a different register for this purpose and assigns no function to DPYTAP.

0

B-file register?	V	register number: B7 address:
Format	31 di	0 elta Ydelta X
Description	DYDX defines the PIXBLT and FILL DX value is the nu of rows of pixels i	• X and Y dimensions of the rectangular destination array for instructions. Both the X and Y dimensions are in pixels; the imber of pixels in the width of the array, and DY is the number n the array.
	When window clip are determined by DYDX. If either th as having a dime	oping is selected, the pixel block dimensions for the transfer y the relationships between WSTART, WEND, DADDR, and he X or Y dimension is 0, then the entire block is interpreted nsion of 0; no transfer is performed.
	The values for DY (up to 65,535, dep dow operations, the Y <sub>start</sub> ]), and WEN + 1) and DX $\leq$ (X <sub>e</sub> WEND, and DAD	and DX can range up to the coordinate extent of the display bending on the display pitch and pixel size selected). For win- ne relationship between DYDX, WSTART (at location [X <sub>start</sub> , ID (at location [X <sub>end</sub> , Y <sub>end</sub> ]) is such that DY $\leq$ (Y <sub>end</sub> – Y <sub>start</sub> end – X <sub>start</sub> + 1). The value in DYDX is used with WSTART, DR to preclip pixels, lines, and pixel arrays.
	Most graphics ins PIXBLT B,XY, PIX er, DYDX is used case, the instruct represented by th dow. No drawing is set, and the value	structions do not modify the contents of DYDX. For FILL XY, (BLT L,XY, and PIXBLT XY,XY, with window option 1, howev- with DADDR to perform a common rectangle function. In this ion sets DYDX to the dimensions of the common pixel block he intersection of the original destination array and the win- is performed. If there is no common rectangle, the V bit is not e of DYDX is indeterminate.
Which instructions use this register?	Instruction	DYDX's format and function
	FILL L	Array dimensions in XY format
	FILL XY	Array dimensions in XY format; special results when W=1 is selected, as previously noted
	FLINE, LINE	Dimensions of the rectangle described by the line to be drawn
	PIXBLT B, L	Array dimensions in XY format
	PIXBLT B, XY	Array dimensions in XY format; special results when pick is selected, as previously noted
	PIXBLT L, L	Array dimensions in XY format
	PIXBLT L, XY	Array dimensions in XY format; special results when pick is selected, as previously noted
	PIXBLT XY, L	Array dimensions in XY format

TMS34020 Registers

Instruction	DYDX's format and function
PIXBLT XY, XY	Array dimensions in XY format; special results when pick is selected, as previously noted
PIXBLT L, L	Array dimensions in XY format
PIXBLT L, XY	Array dimensions in XY format; special results when pick is selected, as previously noted
PFILL XY	XY; dimensions of the fill area
VBLT	XY; dimensions of the pixel block
VFILL	XY; dimensions of the fill area

#### Example

This example illustrates the relationship of DYDX to WSTART and WEND by setting DYDX to the width and height of the clipping window.

WSTART	.set	в5
WEND	.set	в6
DYDX	.set	B7

MOVE	WEND, DYDX	;	Put WEND into DYDX
SUBXY	WSTART, DYDX	;	Generate (WEND-WSTART)
ADDXYI	[1, 1], DYDX	;	Increment by 1 in each
		;	dimension

4-51
B-file register?	register number: address: C000 01D0h
Format	15 0 16-bit counter for horizontal scan lines
Description	HCOUNT is a 16-bit counter used for generating horizontal and composite video signals. HCOUNT is incremented on the falling edge of the VCLK, thus counting the number of VCLK periods per horizontal scan line.
	To generate horizontal sync and blanking signals, the value of HCOUNT is compared to the values of HESYNC, HEBLNK, HSBLNK, and HTOTAL.
	To generate composite serration and equalization pulses, HCOUNT is compared to the value of HESERR and half the value of HESYNC, respectively.
	HCOUNT is reset to 0 on the next VCLK falling edge after HCOUNT= HTOTAL, and the HSYNC output is driven low. If CSYNC/HBLNK is configured to CSYNC, this pin is also driven low.
	If interlaced composite video is enabled, HCOUNT is also reset to 0 on the next VCLK falling edge after HCOUNT=HTOTAL/2 during the equalization and serration regions of vertical blanking.
	In external horizontal or composite-sync video, HCOUNT is reloaded from the SETHCNT register on the rising edge of the video input clock. This is 4 VCLK cycles after the HSYNC or CSYNC input signals, respectively, are driven low.
	Two separate, asynchronous elements of the TMS34020 logic can access the HCOUNT register.
	The video timing control logic (which runs synchronously to the VCLK) increments, clears, and reloads HCOUNT (from SETHCNT) in generating the sync and blanking signals.
	The internal processor (which runs synchronously to LCLK1 and LCLK2) can access HCOUNT as an I/O register.
	No synchronization between these subsystems is provided. HCOUNT can be reliably read from or written to only while VCLK is held at the logic-high level. HCOUNT is typically not read from or written to except during chip test.

B-file register?		register number:
I/O register?	$\checkmark$	address: C000 0030h

end of horizontal-blanking interval

Description

15

HEBLNK is used for generating the HBLNK or CBLNK signals output to a video monitor. The 16-bit HEBLNK value is compared to HCOUNT and defines the point at which the horizontal-blanking interval ends.

For composite video, select the <u>CBLNK/VBLNK</u> pin as <u>CBLNK</u>. <u>CBLNK</u> outputs the logical-OR of the internal horizontal- and vertical-blanking signals; it is low if horizontal- or vertical-blanking is active internally.

Most video monitors require HEBLNK to contain a value that is less than HSBLNK but greater than HESYNC.

# **HESERR** Horizontal End Serration Register

15

B-file register?	register number:	
I/O register? √	address: C000 0270h	J

#### Format

end of composite sync during serration region

Description

HESERR is used for generating the composite-serration pulses output to the video monitor. You need to program this register only when the CSYNC/ HBLNK pin is selected as CSYNC. The 16-bit HESERR value defines the point at which the composite-sync pulse ends during the serration region of vertical blanking (this coincides with the vertical-sync region). When the value in HCOUNT=HESERR during this region, the signal output from the CSYNC pin is driven inactive high to signal the end of the serration pulse.

NTSC and similar composite-video standards require HESERR to contain the value (HTOTAL/2) – HESYNC – 1. (Serration pulses occur every half line, and in each cycle, the CSYNC signal is inactive high for the same duration as horizontal sync is active low.)

When external composite sync is enabled, load HESERR with a value that ensures that the HCOUNT does not become equal to HESERR before the external composite-sync signal goes inactive high, but before CSYNC goes active low again.

B-file register?	register number:
I/O register?	 address: C000 0010h

end of horizontal-sync pulse

Description

15

HESYNC is used for generating the horizontal- and composite-sync signals output to a video monitor. The 16-bit HESYNC value defines the point at which the horizontal-sync pulse ends. If the CSYNC/HBLNK pin is selected as CSYNC, HESYNC also determines the point at which the composite- sync pulse ends (except during the serration region of vertical blanking). When the value in HCOUNT=HESYNC, the signal output from the HSYNC and CSYNC pins is driven inactive high to signal the end of the horizontal-sync interval. During the equalization regions of vertical sync, the CSYNC pin is driven inactive high when HCOUNT=HESYNC/2, indicating the end of the composite-equalization interval.

Monitors typically require HESYNC to contain a value less than HEBLNK; however, the TMS34020 does not require this. The minimum value of HESYNC is 0.

When external horizontal or composite sync is enabled, you should load HESYNC with a value that ensures two things:

- that the value in HCOUNT does not reach HESYNC before the external horizontal or composite-sync signal goes inactive high
- that the value in HCOUNT reaches HESYNC before HSYNC or CSYNC goes active low again

For external composite sync, HCOUNT must not become equal to HESYNC/2 before the composite-sync equalization pulse goes inactive.

n



B-file register?	register number:
I/O register?	→ HSTADRH address: C000 00E0h
	HSTADRL address: C000 00D0h

#### Format

no defined function for TMS34020

Format

HSTADRH and HSTADRL are simple 16-bit read/write locations that can be used to store information passed between the host and the TMS34020.

	B-file register?		register number:
	I/O register?	$\checkmark$	address: C000 0100h
N			

#### 15 14 13 12 11 10 9 8 7 6 5 4 з HINC HP FW NM IM CF NMI RST HLT HLB HACK

Bits

Bits	Name	Function
4	HACK	Acknowledges TMS34020 halt
5—6	HLB0, HLB1	Selects the host byte order
7	RST	Resets the TMS34020
8	NMI	Enables the nonmaskable interrupt
9	NMIM	Selects the mode for the nonmaskable interrupt
11	HPFW	Enables host prefetches after writes
12	HINC	Enables host increment
14	CF	Flushes the cache
15	HLT	Halts TMS34020 processing
0—3 10, 13	100 100 100 100 100 100 100 100 100 100	Reserved; do not use

Description

HSTCTLH contains10 programmable bits for controlling host interface communications. Both the TMS34020 and a host processor can read from or write to HSTCTLH; typically, however, only the host alters HSTCTLH.

#### HACK bit 4

#### Halt acknowledge

 HACK
 Description

 0
 The TMS34020 is running

 1
 The TMS34020 is halted

The TMS34020 sets HACK when the TMS34020 is halted by setting HLT. (Both the host processor and the TMS34020 can write to HLT; in either case, the TMS34020 sets HACK.) By polling the value of HACK, the host can determine when the TMS30420 actually halts. The TMS34020 automatically clears HACK when HLT is cleared to release the TMS34020 from halt.

HLB0, HLB1

# Host last byte

bits 5&6

#### HLB1 HLB0 Last Byte Accessed HLB1 HLB0 Last Byte Accessed 1 0 0 0 byte 3 byte 1 1 1 1 0 byte 2 byte 0

The host processor sets the HLB bits to inform the TMS34020 which byte of a 32-bit word the host will access last. The TMS34020's host interface uses the HLB value to determine when to prefetch the next 32-bit word in memory. If prefetches are enabled, the TMS34020 prefetches the next location from memory after the host accesses the last byte. The HPFW bit determines whether this access must be a read or a write. The 2-bit HLB code allows for all host byte-ordering conventions.

For an 8-bit host, the value of both bits determines after which byte access of the appropriate type (read or write) the TMS34020 will prefetch the next 32-bit location. For a 16-bit host, the value of HLB1 alone is sufficient to determine after which 16-bit word access the TMS34020 will prefetch the next 32-bit location. For a 32-bit host, any combination of HLB1—HLB0 causes the TMS34020 to prefetch the next 32-bit location after each host access of the appropriate type.

#### Reset

RST bit 7

Effect	
Allows normal operation	
Executes reset	
	Effect Allows normal operation Executes reset

Setting RST to 1 has the same effect as asserting the RESET pin low—the TMS34020 executes a reset. However, when RST is set, *only* the TMS34020 is reset; typically, the RESET signal is connected to all devices in the system, and asserting it low affects the entire system.

While the TMS34020 is executing reset internally and the RESET pin is high, DRAM CAS-before-RAS refresh cycles are performed, thus preserving the contents of the DRAMs in the system. During a hardware reset, the TMS34020 uses the value of HCS just before the rising edge of RESET to determine whether to come up in self-bootstrap or host-present mode. When reset is caused by setting RST, however, there is no rising edge on the RESET pin. Because of this, the TMS34020 remembers which mode it was brought up in the last time a reset was caused by asserting RESET, and it configures itself in that mode.

It is not necessary to clear RST; reset clears it automatically.

NIMI bit 8

#### Nonmaskable interrupt, host to TMS34020

 NMI
 Effect

 0
 No NMI is requested

 1
 The host is requesting an NMI

The nonmaskable interrupt allows a host processor to redirect the CPU's execution flow to an NMI routine, regardless of the current state of the interrupt mask flags. The host writes a 1 to the NMI bit to send a nonmaskable interrupt

request to the TMS34020. The interrupt request cannot be disabled, and is always executed (unless the TMS34020 is reset before it can complete interrupt execution). The interrupt is initiated immediately when NMI is set (at the time the current instruction completes execution, or at the next interruptible point in an instruction). Once the interrupt is taken, internal logic automatically clears the NMI bit to 0.

You can use NMI to generate a soft reset after the host downloads new code into TMS34020 memory. After execution of a nonmaskable interrupt, screen-refresh and DRAM-refresh functions continue unaffected. The interrupt does not alter the contents of internal registers except for HSTCTLH (the NMI bit), although the NMI service routine may alter them.

NMIM bit 9

# Nonmaskable interrupt mode

NMI	NMIM	Effect	NMI	NMIM	Effect
0	0	No effect	1	0	NMI, save context on current stack
0	1	undefined	1	1	NMI, discard context

The NMI mode bit determines whether or not the context (PC and ST) of the interrupted program is saved when a nonmaskable interrupt occurs.

- When NMIM=0, the TMS34020 saves the context on the system stack before executing the NMI service routine. This is useful for applications (single-step instruction execution, for example) that must preserve the PC's status between consecutive nonmaskable interrupts. Note that saving the context may be of no benefit if either
  - control will never be returned to the interrupted program, or
  - the integrity of the stack pointer is suspect.
- When NMIM=1, the TMS34020 discards the context when it executes the NMI service routine. You can use a nonmaskable interrupt to simulate a hardware reset in software (using the NMI vector). The NMI does not reset the I/O registers; if you simulate a hardware reset with an NMI, the NMI service routine should reset the I/O registers.

HPFW bit 11

# Host prefetch-after-write enable

HINC	HPFW	Effect
0	0	Disables prefetching and incrementing of internal address
0	1	Disables prefetching and incrementing of internal address
1	0	Enables prefetching after last byte read and incrementing of in- ternal address after any last byte access
1	1	Enables prefetching after last byte write and incrementing of in- ternal address after any last byte writes

HPFW works with the HINC bit to enhance a host processor's access to blocks of TMS34020 memory. When host prefetches are enabled (HINC=1), the value

of HPFW determines whether prefetches are executed after a read to or a write from the last byte of a word (identified by the HLB bits).

- Selecting HINC=1 and HPFW=0 enhances the host processor's ability to read contiguous blocks of TMS34020 memory. This tells the TMS34020 to prepare for the host's next read request by prefetching the next 32-bit location in memory after completing the read of the current long word. If the host uses implicit addressing to access TMS34020 memory (that is, the host provides only the first address of a contiguous block of memory), the TMS34020 automatically generates subsequent addresses by incrementing the address after each access (regardless of whether the access is a read or a write).
- Selecting HINC=1 and HPFW=1 enhances the host processor's ability to modify contiguous blocks of TMS34020 memory. This tells the TMS34020 to prepare for the host's next read request by prefetching the next 32-bit location in memory after the write to the current word is complete. This provides an efficient read-modify-write mechanism. If the host uses implicit addressing to access a block of TMS34020 memory, the TMS34020 generates subsequent addresses by incrementing the address after each write.

If the host is not using implicit addressing, prefetching could yield an unwanted address; however, the TMS34020 has a built-in mechanism that compares the fetched address to the requested address. If the TMS34020 prefetches an unwanted location, it makes an additional access to the requested location. This ensures that the host always accesses the correct location.

# HINC

#### Host increment

HINC	Effect
0	Disables prefetching, incrementing, and comparison of addresses
1	Enables prefetching, incrementing, and comparison of addresses

Setting HINC enhances the TMS34020's host interface performance by providing these features.

Address comparison. The TMS34020 compares the address most recently read or prefetched by the host with the address currently requested by the host on read accesses. This allows prefetching while ensuring that the correct location is always accessed. If the host requests access to a location different from a prefetched location, the TMS34020 detects this and initiates another access to the explicitly requested location.

Address comparison is also useful if the host is not a 32-bit machine. In this case, the host must perform multiple reads to fully read a 32-bit word. The address comparison ensures that once data from the address is latched into the external host data transceivers, accesses to other bytes of the same word do not cause the data to be fetched repeatedly from the TMS34020's local memory.

- **Address prefetch.** The TMS34020 supports prefetching, providing the host with an efficient method for accessing contiguous blocks of TMS34020 memory.
- Address increment. A host can use implicit addressing, supplying only the first address in the block of words that it will read, write, or modify. The TMS34020 automatically increments the address after each last-byte host read/write (HPFW=0), or after each last-byte write (HPFW=1).

# C

CF bit 14

Cache flush	
-------------	--

CF	Effect
0	No effect
1	Flushes and disables the cache

Setting CF to 1 disables the instruction cache and flushes the cache contents. While CF=1, all 4 of the cache's P flags are forced to 0. The TMS34020 must fetch instructions one-at-a-time from local memory.

Normal cache operation resumes when CF is cleared to 0 (assuming that CD[[CONTROL]] also = 0). When the value of CF changes from 1 to 0, the cache begins operation in the same initial state as that which immediately follows reset.

Flushing the cache is useful when the host processor downloads new code to TMS34020 local memory. By setting CF to 1 and then to 0, the host forces the TMS34020 to load new instructions into the cache from memory rather than to continue executing the stale instructions already in the cache.



# Halt TMS34020 program execution

HLT	Effect	
0	Allows TMS34020 to run	
1	Halts TMS34020 instruction execution	

When HLT=1, the TMS34020 suspends instruction processing at the next instruction boundary. Once halted, the TMS34020 does not respond to interrupt requests, including NMI. Local-memory-refresh and video-timing functions continue unaffected while the TMS34020 is halted. When HLT is cleared to 0, the TMS34020 continues execution.

The state of HLT immediately after reset is determined by the state of the  $\overline{HCS}$ pin at the low-to-high transition of RESET:

- If HCS is low, HLT is set to 0 and the TMS34020 can begin executing its reset routine.
- If HCS is high, HLT is set to 1 and the TMS34020 is halted.

Both the host processor and the TMS34020 can write to the HLT bit; this means the TMS34020 can halt itself by setting HLT to 1.



15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
HB REN	HB FI	HR YI	EM IEN	EMG	EMR				М	SGOL	JT	INT IN	N	ASGIN	١

Bits

Bits	Name	Function
0—2	MSGIN	Buffers an input message
3	INTIN	Sends input interrupt from host to TMS34020
46	MSGOUT	Buffers an output message
7	INTOUT	Sends output interrupt from TMS34020 to host
10	EMR	Emulator handshake (request to host)
11	EMG	Emulator handshake (grant from host)
12	EMIEN	Enables emulator inhibit host port interrupt
13	HRYI	Indicates a retry on a host access
14	HBFI	Indicates a bus fault on a host access
15	HBREN	Enables host retry or bus-fault interrupt
89	per and and and and an	Reserved; do not use

#### Description

HSTCTLL controls host interface communications. Both the TMS34020 and the host can *read* all of HSTCTLL's bits, but these restrictions apply to *writes*:

	TMS34020	Host Processor
MSGOUT	Can modify	Can't modify
MSGIN	Can't modify	Can modify
INTIN	Can write a 0; writing a 1 has no effect	Can write a 1; writing a 0 has no effect
INTOUT	Can write a 1; writing a 0 has no effect	Can write a 0; writing a 1 has no effect

Message in-host to TMS34020

#### MSGIN

bits 0-2

The message-in bits buffer a 3-bit interrupt message to the TMS34020 from the host. The host can read from and write to MSGIN, but the TMS34020 can only read MSGIN. MSGIN typically contains a command or status code from the host, which the TMS34020 reads in response to a host-generated interrupt (INTIN=1). The code's meaning depends on your application.

#### INTIN bit 3

# Interrupt in—host to TMS34020

INTIN	Effec	t

INTIN	Effect	
0	No interrupt request to TMS34020	
1	Host requests a TMS34020 interrupt	

INTIN controls the host's message interrupt to the TMS34020. To generate this request, the host sets INTIN to 1. The TMS34020 can deactivate the request by clearing INTIN. The host cannot clear INTIN; similarly, the TMS34020 cannot set INTIN. The HIP [INTPEND] bit reflects the status of INTIN.

# MSGOUT

bits 4-6

# Message out—TMS34020 to host

MSGOUT buffers a 3-bit interrupt message to the host from the TMS34020. The TMS34020 can read from and write to MSGOUT, but the host can only read MSGOUT. MSGOUT typically contains a command or status code from the TMS34020, which the host reads in response to a TMS34020-generated interrupt (INTOUT=1). The code's meaning depends on your application.

# INTOUT

bit 7

# Interrupt out—TMS34020 to host

INTOUT	Effect	
0	No interrupt request to host	
1	TMS34020 requests a host interrupt	

The INTOUT bit controls the TMS34020's message interrupt to the host. The TMS34020 transmits an interrupt request to the host by driving  $\overline{\text{HINT}}$  active low. When INTOUT=1,  $\overline{\text{HINT}}$  is driven active low; when INTOUT=0,  $\overline{\text{HINT}}$  is driven inactive high. The TMS34020 activates the interrupt request by setting INTOUT to 1; the host deactivates the request by clearing INTOUT. The TMS34020 cannot clear INTOUT; similarly, the host cannot set INTOUT.



bits 10&11

# Emulator (or debugger) handshake-request to/grant from host

EMG	EMR	Interpretation
0	0	No request, no interrupt
0	1	Host request from EMU, interrupt if enabled
1	0	Host released by EMU, interrupt if enabled
1	1	Host grant to EMU, no interrupt

An in-circuit emulator or software debugger may use EMR and EMG for exchanging information and coordinating activity with a host processor. The precise meaning of these bits depends on your application, the emulator or debugger software, and the host processor.

If a debugger or an in-circuit emulator needs to start emulation activity with the TMS34020, the debugger or emulator may set EMR to make this request to the host. If EMIEN=1, setting EMR causes the host to be interrupted via HINT. The host then sets EMG to acknowledge this, causing HINT to return to its inac-

tive level. The emulator or debugger then clears EMR, signalling the end of the activity to the host. Again, if EMIEN=1, clearing EMR causes the host to be interrupted via HINT. The host then clears EMG, completing the transaction and causing HINT to return to its inactive level.

Only an emulator or debugger should modify EMR, and only the host should modify EMG. If you are not using this protocol, clear these bits to 0.

EMIEN

# Emulator inhibit host port interrupt enable

EMIEN	Effect
0	EMR XOR EMG causes no host interrupt via HINT
1	EMR XOR EMG causes an interrupt to the host via HINT

EMIEN controls whether the exclusive-OR of the EMR and EMG bits causes the HINT pin to be driven active low, thus interrupting the host.

HRYI bit 13

# Retry on host access interrupt

HRYI	Effect
0	Host access was not retried
1	Host access was retried (HINT was set active, if enabled)

The TMS34020's host interface sets HRYI if a host access returns a retry memory cycle completion code. The TMS34020 automatically attempts to retry the memory access. If enabled (HBREN=1), the TMS34020 interrupts the host via the HINT pin. The host must ensure that the appropriate action (if any) is taken to clear the cause of the retry, and then the host must clear HRYI.

HBE bit 14

# Bus fault on host access interrupt

HBFI	Effect
0	Host access was not faulted
1	Host access was faulted (HINT was set active, if enabled)

The TMS34020's host interface sets HBFI if a host access returns a bus-fault memory cycle completion code. The TMS34020 performs no further error recovery, but terminates the host request and drives HRDY high as if the cycle completed successfully. If enabled (HBREN=1), the TMS34020 interrupts the host via the HINT pin. The host must ensure that the appropriate action (if any) is taken to clear the cause of the bus fault, and then the host must clear HBFI.

HBREN bit 15

# Host bus-fault or retry interrupt enable

# HBREN Effect

0	No interrupt sent to the host via $\overline{\text{HINT}}$ if HRYI or HBFI is set
1	An interrupt is sent to the host via $\overline{\text{HINT}}$ if HRYI or HBFI is set

HBREN enables or inhibits the TMS34020 from interrupting the host when a retry or bus fault occurs on a host access.



B-file register?	register number:
I/O register? √	address: C000 00C0h

Format	150
	no defined function for TMS34020
Description	HSTDATA is a simple 16-bit read/write location that can be used to store infor- mation passed between the host and the TMS34020.

ſ	B-file register?			regist	er number:			
	I/O register?	$\boxed{}$			address:	C000	0050h	
V								

Format	15	0 start of horizontal blanking				
Description		BLNK is used for generating the HBLNK or CBLNK signal output to the vid- monitor. The 16-bit HSBLNK value is compared to HCOUNT and defines point at which the horizontal-blanking interval begins.				
	For composite video, select the CBLNK/VBLNK pin as CBLNK. CBLNK outputs the logical-OR of the internal horizontal- and vertical-blanking signals; it is low if either horizontal- or vertical-blanking is active internally.					
	Se	veral internal events coincide with the start of horizontal blanking:				
	G	A request for a screen-refresh memory cycle is sent to the TMS34020's memory controller.				
	Q	If a display interrupt is programmed to occur at a particular horizontal scan line, the actual interrupt request is generated at this point.				
	Mo gre	nitors typically require that HSBLNK contain a value less than HTOTAL, but ater than HEBLNK.				

Ω

B-file register?	register number:
I/O register?	address: C000 0070h

Format

total horizontal scan lines

Description

15

HTOTAL is used for generating the horizontal- and composite-sync signals output to the video monitor. The 16-bit HTOTAL value is compared to HCOUNT and defines the duration of each horizontal scan line on the screen in terms of VCLK periods.

HTOTAL is compared with the horizontal count in HCOUNT to determine the point at which the horizontal- and composite-sync pulses begin. Usually, HCOUNT counts from 0 to the value in HTOTAL. When HCOUNT = HTOTAL, the HSYNC output is driven active low on the next falling edge of VCLK, and HCOUNT is reset to 0 on the same clock edge. If the CSYNC/HBLNK pin is selected as CSYNC, then CSYNC is also driven active low.

In addition, for interlaced composite video, HCOUNT is reset to 0 when HCOUNT=HTOTAL/2 during the equalization and serration regions. This condition triggers the equalization and serration pulses on the CSYNC pin (which occur every half horizontal scan line). During this time, the beginning of horizontal-sync pulses on the HSYNC pin are caused by alternating occurrences of HCOUNT=HTOTAL/2.

In interlaced video, HTOTAL should contain an odd number (LSB=1) to achieve equal spacing between lines. Equalization and serration pulses are then evenly separated by half a scan line. The total number of VCLKs per horizontal scan line is calculated as HTOTAL + 1.

When external horizontal or composite video is enabled, HTOTAL should contain a value not less than HCOUNT's value at the point at which the external sync pulse is expected. If you use SETHCNT to exactly align the internal video timing with the external sync, set HTOTAL to exactly match the number of VCLKs between external syncs.

HTOTAL should contain a 16-bit value greater than HSBLNK but less than or equal to 65,535 (FFFF<sub>16</sub>).



31	C
IHOST 1	
IHOST 2	
IHOST 3	
IHOST 4	

#### Description

The TMS34020's host interface uses these 32-bit locations for storing the addresses used in making host-requested reads, writes, and prefetches to TMS34020 local memory. These locations are included in the I/O register memory space for purposes of chip test only. You cannot write to these locations. The data read from these locations is generally not 0.

B-file register?		register number:
I/O register?	$[ \downarrow ]$	address: C000 0110h

15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0 WVE DIE HIE X2E X1E

Bits

Bits	Name	Function					
1	X1E	Enables external interrupt 1					
2	X2E	Enables external interrupt 2					
9	HIE	Enables the host interrupt					
10	DIE	Enables the display interrupt					
11	WVE	Enables the window violation interrupt					
0 3—8 12—15	H <sup>ar</sup> indan dan dan dan Dan dan dan dan dan dan Dan dan dan dan dan dan dan dan dan dan d	Reserved; do not use					

Description

INTENB contains an interrupt mask that selectively enables 3 internally generated and 2 externally generated interrupt requests.

- **X1 & X2** *external interrupts 1 and 2.* Generated by active-low signals on the LINT1 and LINT2 input pins, respectively.
- **HI** *host interrupt.* Generated when the host processor sets INTIN[[HSTCTLL]] to 1.
- **DI** *display interrupt*. Generated when the vertical count in the VCOUNT register reaches the value of DPYINT.
- **WV** *window-violation interrupt*. Caused by an attempt to write a pixel inside or outside the current window limits (depending on the selected window-checking mode).

The status register contains a global interrupt enable bit (IE). The INTENB register contains individual interrupt enable bits associated with each of the 5 interrupts described above. Interrupts are enabled through a 2-step process:

Step 1: Set the IE bit to 1.

**Step 2:** Set the appropriate bits in INTENB to 1.

Setting IE to 0 disables all of these interrupts, regardless of the value in INTENB. When IE=1, each interrupt is enabled according to the appropriate value in INTENB (1 enables the interrupt, 0 disables it).

B-file register?	regis	ster number:
I/O register?		address: C000 0120h

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
				WVP	DIP	HIP							X2P	X1P	

Bits

Bits	Name	Function				
1	X1P	Identifies a pending external interrupt 1				
2	X2P	Identifies a pending external interrupt 2				
9	HIP	Identifies a pending host interrupt				
10	DIP	Identifies a pending display interrupt				
11	WVP	Identifies a pending window violation interrupt				
0, 3—8 12—15		Reserved; do not use				

Description

INTPEND indicates which interrupt requests are currently pending (for a description of these interrupts, refer to the discussion of the INTENB register). The individual pending bits in the INTPEND register reflect the status of interrupt requests. The interrupt is requested if the corresponding pending bit is 1; there is no request if the pending bit is 0. INTPEND reflects the status of each interrupt request, regardless of whether the interrupt is enabled or not; this allows the TMS34020 to poll interrupts.

X1P and X2P are read-only bits that reflect the input levels on LINT1 and LINT2; they are not affected when INTPEND is written to. LINT1 and LINT2 are asynchronous inputs, but signals to these pins are synchronized internally so that you can always reliably read X1P and X2P. If an external interrupt is disabled, the TMS34020 ignores the interrupt request, even if the corresponding pending flag is set. The TMS34020 takes the interrupt only if the external request is maintained at the request pin, until the interrupt is again enabled.

DIP and WVP reflect the status of interrupt requests generated internally. These 2 bits are implemented as latches. Once set, DIP or WVP remain set until a 0 is written to the appropriate bit (or until the TMS34020 is reset). Writing a 1 to either of these bits has no effect at any time. While an internal interrupt is disabled, the interrupt request is ignored, even if the corresponding pending flag is set. If the interrupt is then enabled while the interrupt-pending flag is set (because of a prior interrupt request), the TMS34020 takes the interrupt.

HIP is a read-only bit that always displays the current contents of INTIN. Writing to the INTPEND register does not affect HIP. A host interrupt request is generated when the host processor writes a 1 to INTIN. The TMS34020 clears the interrupt request by writing a 0 to INTIN[[HSTCTLL]].

B-file register?	√			regis	ter number: address	B10	)
Format	31		li	near bit add	ress		
Description	MADDR contains the mask array address for PIXBLT L,M,L. MADDR usually points to the mask bit with the lowest address in the mask array. When the selected starting corner is not the upper left corner, you must manually adjust MADDR to point to the mask array's starting corner. MADDR always contains a linear address. When the PIXBLT L,M,L completes,					lly he ist	
	MADDF array. If to be rea	l points to PIXBLT   ad.	o the starting loo L,M,L is interrup	cation of the oted, MADI	e row that fol OR points to	lows the last row in tl the next word of pixe	he els
Which instructions	Instruct	tion	MADDR's	s format			يتقسي
	SETC	MP	Linear; a	any value			
	PIXBL	T L, M, L	Linear; a	any value			
Example	MADDR	.set	B10				
		MOVI	00010AFCh,	MADDR	; Mo ; va ; in	ve linear lue 10AFCh to B10	

B-file register?	√	register number: B11 address:
Format	31	0 linear bit address
Description	MPTCH defines cent rows of the in MPTCH to mc any value.	the linear difference in the starting memory addresses of adja- mask array for PIXBLT L,M,L. The TMS34020 uses the value ove from row to row through the mask array. MPTCH can have
	If you're manual the SETCMP in mask pitch conv can then use CV	ly converting an XY address to a linear address, you can use struction; SETCMP uses the MPTCH value to calculate the version factor and loads the correct value into CONVMP. You /MXYL to perform the conversion.
Which instructions use this register?	Instruction	MPTCH's format
	CVMXYL	Linear; any value
	FLINE, LINE	Linear; any value
	FPIXEQ, FPIX	NE Linear; any value
	PIXBLT L, M, L	Linear; any value
	SETCMP	Linear; any value
Example	MPTCH .set	B11
	MOVI MOVI MOVI	00000100h, MPTCH ; Power of 2 000A03D0h, MPTCH ; Arbitrary value 00000220h, MPTCH ; 2 powers of 2 ; (512 + 32)

B-file register?	<u>√</u>	register number: B4 address:
Format	31	0 linear bit address
Description	OFFSET contains the lin space for instructions usir linear address of the XY of the memory base for perfe	ear address of the first pixel in the XY coordinate ng XY addressing. This address corresponds to the origin (X=0,Y=0). The TMS34020 uses this value as orming XY-to-linear address conversions.
	OFFSET always contains position in the TMS34020 versions, transparency, pi contain a pixel-aligned va operand do not modify the	a <i>linear</i> address. The offset address may be at any linear address space. For proper XY address con- xel processing, and plane masking, OFFSET should alue. Instructions that use OFFSET as an implied e register's contents.
Which instructions use this register?	Instruction	OFFSET's function
<b>g</b>	CVXYL	Linear address of XY origin
	DRAV	Linear address of XY origin
	FILL XY	Linear address of XY origin
	LINE	Linear address of XY origin
	PFILL XY	Linear address of XY origin
	PIXBLT B, XY	Linear address of XY origin
	PIXBLT L, XY	Linear address of XY origin
	PIXBLT XY, L	Linear address of XY origin
	PIXBLT XY, XY	Linear address of XY origin
	PIXBLT L, M, L	Linear address of XY origin
	PIXT Rs, *Rd.XY	Linear address of XY origin
	PIXT * <i>Rs.</i> XY, <i>Rd</i>	Linear address of XY origin
	PIXT * <i>Rs.</i> XY, * <i>Rd</i> .XY	Linear address of XY origin
	TFILL	Linear address of XY origin
Example	OFFSET .set B4	

B-file register? [ I/O register?	√	register number: B13 address:
Format	31	oreplicated pixel value
Description	PATTERN uses the tern. COLOR0 def provides the repla	e information in COLOR0 and COLOR1 to define a pixel pat- ines the replacement color for 0 bits in the pattern; COLOR1 cement color for 0 bits in the pattern.
	Note:	
	If the PATTERN throughout all 32	value is less than 32 bits, you must replicate the pattern bits of the PATTERN register.
Which instructions use this register?	Instruction	PATTERN's function
<b>9</b>	FLINE, LINE	Line pattern
	PFILL XY	Array pattern. If PATTERN contains all 1s, PFILL uses COLOR1 to produce a solid fill. If PATTERN contains all 0s, PFILL uses the COLOR0 value to produce a solid fill.

B-file register?	register number:
I/O register?  √	PMASK (32-bit address): C000 0160h PMASKL (16-bit address): C000 0160h PMASKH (16-bit address): C000 0170h

or

31	0
PMASK—32-bit plane mask	
15	0
PMASKL—16 LSBs of 32-bit plane mask	
15	0
PMASKH—16 MSBs of 32-bit plane mask	

# Note:

You can access the plane mask registers separately or together by using different addresses and different field sizes.

- □ To access PMASK as a single 32-bit register, access the 32-bit field at address C000 0160h.
- To access PMASKL as a 16-bit register, access the 16-bit field at address C000 0160h.
- To access PMASKH as a 16-bit register, access the 16-bit field at address C000 0170h.

#### Description

The PMASK registers selectively enable or disable various planes in the bitmap of a multiple-bit-per-pixel display system. Together, PMASKL and PMASKH (referred to as PMASK) contain a 32-bit value that determines which bits of each pixel can be modified during execution of a graphics instruction. The PMASK registers enable you to identify which bits in each pixel are protected (mask bit=1) or not protected (mask bit=0) from modification.

- During a pixel write, 0s in the plane mask identify destination bit positions that can be modified. The 1s in the plane mask represent bit positions within the destination that are protected from modification.
- During a pixel read, 0s in the plane mask identify readable bits within a pixel; bits corresponding to 1s in the mask are always read as 0s.

Display memory organization can be described in terms of bit planes. If the pixel size is 4 bits, for example, and the bits in each pixel are numbered from 0 to 3, the display memory is composed of 4 bit planes, numbered 0 to 3. Plane 0 contains all the bits numbered 0 from all the pixels, plane 1 contains all the bits numbered 1 from all the pixels, and so on. A 4-bit mask is constructed so that bit 0 of the mask enables or disables writes to the bits in plane 0, mask bit 1 enables or disables writes to plane 1, etc. The plane-mask value for a 4-bit pixel is a 4-bit value; the plane mask for an 8-bit pixel is an 8-bit value, etc. You must replicate the plane mask throughout the 32 bits of the PMASK registers. For example, when PSIZE=8, you must load the PMASK registers with 4 identical copies of the 8-bit plane-mask value, as Figure 4–8 shows. In general, all 32 bits of the registers are used, and a mask for a pixel size of less than 32 bits must be duplicated *n* times (where *n* is 32 divided by the pixel size).

# Figure 4–8. Replicating the Mask Value for an 8-Bit Pixel

31 24	23 16	15 8	7 0
8-bit mask	8-bit mask	8-bit mask	8-bit mask
			L

The individual bits of the PMASK registers are associated with corresponding bits of the 32-bit LAD bus (data are multiplexed over the same LAD0—LAD31 pins as the address). PMASK register bit 0 corresponds to bit 0 of the data bus (the bit transferred on LAD0), PMASK bit 1 is associated with bit 1 of the data bus, etc. In general, if PMASK bit *n* is a 0, the mask enables bit *n* of the data bus; if PMASK bit *n* is a 1, the mask disables bit *n*.

You can effectively disable plane masking by loading all 0s into the PMASK registers; this allows all bits of each pixel to be modified. This is the default state of the PMASK registers following reset.

If your system's VRAMs can store a copy of the plane mask internally (the TMS44251 can do this), then you should set VEN[[CONFIG]]. The TMS34020 automatically detects when the PMASK registers are modified. It subsequently performs a special load-write-mask memory cycle to copy the 1s complement of the PMASK contents into the VRAMs' internal write-mask. (The PMASK is inverted because the meaning of the bits in a VRAM's write mask is opposite to the meaning of the bits in the PMASK.) The TMS34020 can use the VRAM copy of the plane mask to perform plane-masked writes without performing read-modify-write cycles.

Instruction	PMASK's function
DRAV	Plane-mask value for graphics operations
FILLs (both)	Plane-mask value for graphics operations
FLINE, LINE	Plane-mask value for graphics operations
FPIXEQ, FPIXNE	Plane-mask value for graphics operations
PIXBLTs (all)	Plane-mask value for graphics operations
PIXT <i>Rs</i> , * <i>Rd</i> and * <i>Rs</i> , * <i>Rd</i> .XY	Plane-mask value for graphics operations
TFILL	Plane-mask value for graphics operations
VBLT	Plane-mask value for graphics operations
VFILL	Plane-mask value for graphics operations

# Which instructions use this register?

B-file register? [		register number:
I/O register?	1	address: C000 0150h
Format	15	pixel size
Description	PSIZE defines the pixel si value 4; if the pixel size is register can be written to bits; any other value of PS	ze in bits. If the pixel size is 4, load PSIZE with the 8, load PSIZE with 8, etc. All 16 bits of the PSIZE or read. Legal pixel sizes are 1, 2, 4, 8, 16, and 32 SIZE is undefined.
	PSIZE = 0001h PSIZE = 0002h PSIZE = 0004h PSIZE = 0008h PSIZE = 0010h PSIZE = 0020h	Pixel size =1bit per pixelPixel size =2bits per pixelPixel size =4bits per pixelPixel size =8bits per pixelPixel size =16bits per pixelPixel size =32bits per pixel
Which instructions use this register?	Instruction	PSIZE's function
•	CVXYL	X shift amount for XY-to-linear conversion
	CVDXYL	X shift amount for XY-to-linear conversion
	CVMXYL	X shift amount for XY-to-linear conversion
	CVSXYL	X shift amount for XY-to-linear conversion
	DRAV	Pixel size for graphics operations
	FILLs (both)	Pixel size for graphics operations
	FLINE, LINE	Pixel size for graphics operations
	FPIXEQ, FPIXNE	Pixel size for graphics operations
	PIXBLTs (all)	Pixel size for graphics operations
	PIXT <i>Rs</i> , * <i>Rd</i>	Pixel size for graphics operations
	PIXT * <i>Rs</i> , * <i>Rd</i>	Pixel size for graphics operations
	PIXT <i>Rs</i> , * <i>Rd</i> .XY	Pixel size for graphics operations
	PIXT * <i>Rs.<b>XY</b>, *Rd</i>	Pixel size for graphics operations
	PIXT * <i>Rs<b>.XY</b>, *Rd<b>.XY</b></i>	Pixel size for graphics operations
	RPIX	Field size for replication
	TFILL	Pixel size for graphics operations
	VBLT	Pixel size for graphics operations
	VFILL	Pixel size for graphics operations

B-file register?		register number:
I/O register?	$\checkmark$	address: C000 01F0h
		<u> </u>

Format

REFADR

Description

REFADR contains the address output during DRAM-refresh cycles. DRAMs require periodic refreshing to retain their data. The TMS34020 automatically generates CAS-before-RAS cycles that refresh the DRAMs at regular intervals. You can select the interval between refresh cycles by loading an appropriate value into RR0—RR2[CONFIG]. This determines how often, if at all, DRAM refreshes should be performed.

REFADR forms a contiguous binary counter. Each time a DRAM refresh is performed, the address in REFADR is output on both LAD16—LAD31 and RCA0—RCA12. RCM0—RCM1[CONFIG]] determines which bits of the logical address appear on RCA0—RCA12 at row-address time. During a DRAM-refresh cycle, the address is valid on both LAD16—LAD31 and RCA0—RCA12 throughout the memory cycle. This memory cycle is 3 machine states long, allowing plenty of time for the external decode logic to detect the refresh (from the status code output on LAD3—LAD0) and then to enable the appropriate banks of memory for refresh. The refresh pseudo-address is incremented after each DRAM-refresh cycle that completes normally (that is, does not return the retry completion code on the LRDY and BUSFLT pins). If a refresh cycle does return a retry condition, the refresh cycle is automatically rescheduled and the same address is output.

You can use the refresh pseudo-address to determine which banks of memory will be refreshed. Or, you can use it as the refresh address required by DRAMs that support  $\overline{RAS}$ -only refresh. The TMS34020 does not directly support  $\overline{RAS}$ -only refresh; if you use  $\overline{RAS}$ -only refresh, you must use external hardware to prevent activation of the  $\overline{CAS}$  strobes.

Reset clears the REFADR register to 0; no refreshes are performed while the RESET pin is held active low. However, if the RESET pin is held high while the TMS34020 is still executing reset internally, DRAM refreshes are performed. After RESET is taken high, no CPU-initiated memory cycles occur until 8 DRAM-refresh cycles are completed. This ensures that the DRAMs and VRAMs in the system are initialized correctly.

B-file register?	√	register number: B0 address:
Format	31 Y half of	an XY address X half of an XY address
or		linear bit address
Description	SADDR contains to the pixel with starting corner is SADDR to point allows you to ha however, you m	the source array address for PIXBLTs. SADDR usually points the lowest address in the source array. When the selected not the upper left corner, the TMS34020 automatically adjusts to the selected starting corner of the source array. This feature ndle overlapping arrays. (For PIXBLT L,L and PIXBLT L,M,L, ust manually adjust SADDR to point to the starting corner.)
	The TMS34020 address, depend	treats the address in SADDR as an XY address or a linear ding on the instruction using it.
	During PIXBLT of PIXBLT complet follows the last r the next word of	operations, SADDR is maintained in linear format. When the tes, SADDR points to the starting location of the row that ow in the array. If a PIXBLT is interrupted, SADDR points to pixels to be read.
Which instructions	Instruction	SADDR's format and function
	BLMOVE	Linear; points to the beginning of the source array
	FLINE, LINE	Decision variable $d = 2b - a$ , used for the line draw
	PIXBLT B, L PIXBLT B, XY	Linear; points to the beginning of the binary source array
	PIXBLT L, L PIXBLT L,M,L	Linear with special requirements when PBH = 1 or PBV=1; re- fer to the PIXBLT L,L discussion for a description of its unique requirements
	PIXBLT L, XY	Linear; points to the beginning of the source array
	PIXBLT XY, L PIXBLT XY,XY	XY; points to the beginning of the source array
	TFILL	XY; points to the first pixel in the line
	VBLT	Linear; points to the beginning of the source array
Example	SADDR .set	во
	MOVI	[08h, 015h], SADDR ; Move XY value ; 15h,8h into B0 0000AAAAh, SADDR ; Move linear value ; AAAAh into B0

B-file register?	√ 	register number: address: C000 02C0h
Format	15	SCOUNT
Description	Du ma in I SC SC ser bit reg flov the	ring horizontal-blanking screen-refresh cycles, the video timing logic auto atically loads SCOUNT with the VRAM tap point (determined using the value DPYMSK). The tap point is automatically right-justified before it is loaded o SCOUNT. SCOUNT is incremented by the rising edge of a pulse on the LK pin. You should connect SCLK to the VRAM's serial clock signal so tha COUNT is incremented each time a bit of data is shifted out of the VRAM's rial register. This means that SCOUNT always contains the tap point of the most recently shifted. When the VRAMs shift the last bit out of one half serial gister and start shifting bits from the other half serial register, SCOUNT over ws from all 1s to 0 and schedules a midline-reload screen refresh to transfe e next half-row of VRAM into the half serial register not being shifted out.
	Ho bla	Id the SCLK pin at the inactive-low level throughout horizontal and vertica anking, when the VRAM serial registers are not shifting data.
	Tw acc	o separate asynchronous elements of the TMS34020 internal logic car cess SCOUNT:
	Q	The midline-reload timing control logic, which runs synchronously to SCLK, increments SCOUNT during the active regions of the screen (when neither vertical nor horizontal blanking is active).
		The internal processor, which runs synchronously to the local clocks LCLK1 and LCLK2, can access SCOUNT as an I/O register and can load it with the VRAM tap point during horizontal blanking.
	No reli is r	synchronization between these subsystems is provided. SCOUNT can be iably read or written only while SCLK is held at the logic-low level. SCOUN not typically read or written except during chip test.

B-file register?		register number:
I/O register?	1	address: C000 0310h

programmable value for HCOUNT

Description

15

If external horizontal or composite video is enabled (by clearing the HSD or CSD bit, respectively, in DPYCTL), the video timing logic loads the value of SETHCNT into HCOUNT when

the logic detects an external horizontal-sync pulse on HSYNC, or

the logic detects an external composite-sync pulse on CSYNC.

Setting HCOUNT to a programmable value rather than clearing it counteracts delays inherent in the synchronization of external sync pulses and other external system delays.

It takes 4 VCLK cycles from the time an external sync is detected at the appropriate sync input pin until its effects propagate to the video output pins. If SETHCNT contains the value 4, then HCOUNT is set to 4 at the beginning of the fifth VCLK cycle (that is, 4 whole VCLK cycles) after the external sync pulse is detected at the pins. This has the same effect as if HCOUNT were cleared to 0 in the same VCLK cycle that the external sync signal went active low. If the HTOTAL value matches the parameters of the external video source, then HCOUNT=HTOTAL coincides with the beginning of the next external sync pulse. This condition causes HCOUNT to be loaded with 0, and the internal horizontal- and composite-sync pulses begin. As a result, any sync pins configured as outputs go active low on the same VCLK cycle as the external sync signal, and internally generated video signals are synchronous to and aligned with the external video signals.

This is especially useful if the TMS34020 is performing sync conversion. By programming all the video timing registers to match the parameters of the external video source, an external composite sync can be used to generate horizontal- and vertical-sync outputs, or external horizontal and composite syncs can be used to generate a composite-sync output.

Programming SETHCNT to a value greater than 4 causes the sync pins configured as outputs to change *SETHCNT*-4 VCLK cycles before the external sync (or syncs) changes. This is useful if the system contains external clock skews. Similarly, programming SETHCNT to a value less than 4 causes the sync pins configured as outputs to change after the external sync (or syncs). Do not set SETHCNT to a value less than 0.

B-file register?	register number: √ address: C000 0300h
Format	15 0 programmable value for VCOUNT
Description	If external vertical or composite video is enabled (by clearing the VSD or CSD bit, respectively, in DPYCTL), the video timing logic loads the value of SETVCNT into VCOUNT when
	$\Box$ the logic detects an external vertical-sync pulse on $\overline{\text{VSYNC}}$ , or
	the logic detects the first serration pulse in the external composite-sync signal on CSYNC.
	SETVCNT provides symmetry with the SETHCNT register. If you are using interlaced video, program SETVCNT to 0. If you do not, the TMS34020 will not be able to distinguish between external odd and even fields and may not correctly synchronize to the external source.
	For noninterlaced video, you can set SETVCNT to nonzero values. This causes the internal vertical-blanking signal (visible externally on $\overline{CBLNK}$ / $\overline{VBLNK}$ ) and the internal vertical-sync signal (visible externally on $\overline{VSYNC}$ if it is an output) to start and end an integral number of scan lines ahead of the external vertical sync and blanking signals. Programming SETVCNT to <i>n</i> causes the internal signals to be <i>n</i> scan lines in advance of the external signals.

B-file register?	$\checkmark$	register number: B1
I/O register?		address:

#### Format

linear bit address

# Description

SPTCH defines the linear difference between the starting addresses of adjacent rows of a source array. The TMS34020 uses the value in SPTCH to move from row to row through the source array. SPTCH can have any value that is a multiple of the current pixel size. Note that XY-to-linear conversion is most efficient when SPTCH is a power of 2.

If you're manually converting an XY address to a linear address, you can use the SETCSP instruction; SETCSP uses the SPTCH value to calculate the source pitch conversion factor and loads the correct value into CONVSP. You can then use CVSXYL to perform the conversion.

Instruction	SPTCH's format	
CVSXYL	Linear; any value	
PIXBLTs (all)	Linear; any value	
PIXT *Rs.XY, Rd	Linear; any value	
PIXT * <i>Rs<b>.XY</b>, *<i>Rd<b>.XY</b></i></i>	Linear; any value	
SETCSP	Linear; any value	

# Example

SPTCH .set B1

MOVI	00001000h,	SPTCH	;	Power of 2
MOVI	00000900h,	SPTCH	;	2 powers of 2
MOVI	00010AFCh,	SPTCH	;	Arbitrary value

B-file register?	√ 	register number: address: C000 01C0h
Format	15	0 16-bit video counter
Description	VC ver clo on and lac pul	COUNT counts the horizontal lines in the display, and is used for generating tical and composite sync and blanking signals. VCOUNT increments on the ck edge that resets HCOUNT; in interlaced video, VCOUNT also increments the clock edge after HCOUNT=HTOTAL/2. This causes $\overrightarrow{VSYNC}$ 's falling drising edges to coincide with the falling edge of $\overrightarrow{CSYNC}$ or $\overrightarrow{HSYNC}$ (in intered video, some $\overrightarrow{VSYNC}$ transitions occur halfway between two $\overrightarrow{HSYNC}$ ses). If $\overrightarrow{CBLNK/VBLNK}$ is selected as $\overrightarrow{VBLNK}$ , this also applies to $\overrightarrow{VBLNK}$ .
	To VC inte full reg eo) fall syr del	generate vertical sync and blanking signals, the video timing logic compares COUNT to the values of VEBLNK, VSBLNK, VTOTAL, and VESYNC/2. In erlaced composite video (when CSYNC/HBLNK is selected as CSYNC), the value of VESYNC is used to determine the end of the second equalization gion. When HCOUNT=HTOTAL (or HCOUNT= HTOTAL/2 in interlaced vider and VCOUNT=VTOTAL simultaneously, VCOUNT is reset to 0 on the next ing edge and VSYNC is driven active low. In external vertical- or composite-inc video, VCOUNT is reloaded from SETVCNT when a falling edge is sected on VSYNC, or when the first serration pulse is detected on CSYNC.
	Wh of p vid equ ery beg	then $\overrightarrow{CSYNC}/\overrightarrow{HBLNK}$ is selected as $\overrightarrow{CSYNC}$ , VCOUNT determines the type bulse output on this pin. If VESYNC <vcount <math="">\leq VSBLNK, then <math>\overrightarrow{CSYNC}</math> outputs ordinary horizontal-sync pulses that coincide with <math>\overrightarrow{HSYNC}</math>. In interlaced eo, if VSBLNK<vcount <math="">\leq VTOTAL or VESYNC/2<vcount <math="">\leq VESYNC, pulses are generated on <math>\overrightarrow{CSYNC}</math>. Equalization pulses appear evaluation pulses are generated on <math>\overrightarrow{CSYNC}</math>. Equalization pulses appear evaluation pulses on <math>\overrightarrow{HSYNC}</math>. Every other pulse gins coincident with a pulse on <math>\overrightarrow{HSYNC}</math>.</vcount></vcount></vcount>
	lf V the dei so	TOTAL VESYNC/2, CSYNC generates serration pulses. HESERR defines length of these pulses. In noninterlaced video, they always begin coincint with a pulse on HESYNC. In interlaced video, they occur every half line, every other pulse begins coincident with a pulse on HSYNC.</td
	A c coc	lisplay interrupt is generated when VCOUNT=DPYINT. You can use this to ordinate software activity with the refreshing of selected lines on the screen.
	Tw acc	o separate, asynchronous elements of the TMS34020 internal logic can cess VCOUNT:
	<b>G</b>	The video timing control logic (which runs synchronously to VCLK) incre- ments and clears or reloads VCOUNT while generating sync and blanking signals.

The internal processor (which runs synchronously to LCLK1 and LCLK2) can access VCOUNT as an I/O register.

The TMS34020 provides no synchronization between these subsystems. VCOUNT can be reliably read or written only while VCLK is held at the logic-low level. VCOUNT is not typically read or written except during chip test.

B-file register?	regi	ster number:
I/O register?		address: C000 0020h

#### Format

end of vertical blanking

Description

VEBLNK is compared to VCOUNT to determine when the vertical-blanking interval ends. The vertical-blanking interval ends when VCOUNT=VEBLNK and either of these conditions is satisfied:

HCOUNT=HTOTAL or

HCOUNT=HTOTAL/2 in the interlaced even field.

In separate sync, the vertical-blanking signal is output on the <u>CBLNK/VBLNK</u> pin, which is selected as <u>VBLNK</u>. In composite sync, <u>CBLNK/VBLNK</u> is selected as <u>CBLNK</u>; in this case, <u>CBLNK</u> outputs the logical-OR of the internal horizontal- and vertical-blanking signals. <u>CBLNK</u> is low if either horizontal- or vertical-blanking is active internally.

Monitors typically require VEBLNK to contain a value less than VSBLNK and greater than VESYNC/2.

n

B-file register?	register number:	<u>)</u>
I/O register? √	address: C000	) 0000h

end of vertical-sync pulse

Description

VESYNC is compared to VCOUNT to determine when the vertical-sync pulse ends and, in interlaced composite video, when the second equalization region ends. The sync pulse ends when VCOUNT= VESYNC/2 and either of these conditions is satisfied:

HCOUNT=HTOTAL or

15

HCOUNT=HTOTAL/2 in interlaced video.

The  $\overline{\text{VSYNC}}$  output is driven inactive high to signal the end of the vertical-sync interval.

In interlaced video, the second serration region ends if both of these conditions are satisfied:

VCOUNT=VESYNC and

HCOUNT=HTOTAL/2.

Monitors typically require VESYNC/2 to contain a value less than VEBLNK; VESYNC's minimum value is 0.

In external vertical or composite video, you should load VESYNC with a value such that the internal vertical-sync pulse ends before or at the same time as the end of the external vertical-sync pulse or the end of the external serration region. If the external vertical-sync pulse or the external serration region is still active when the internal vertical-sync pulse ends, it causes VCOUNT to be reloaded from SETVCNT, and the internal vertical-sync interval starts again.
15

B-file register?	register number:	
I/O register? √	address:	C000 0040h

Format

start of vertical blanking

Description

VSBLNK is compared to VCOUNT to determine when the vertical-blanking interval starts. Vertical blanking starts when VCOUNT=VSBLNK and either of these conditions is satisfied:

- HCOUNT=HTOTAL or
- HCOUNT=HTOTAL/2 in the interlaced odd field.

Additionally, vertical blanking will start if the video timing logic detects an external composite-sync pulse (when CSYNC is an input) or a horizontal-sync pulse (when HSYNC is an input).

In separate sync, the vertical-blanking signal is output on the <u>CBLNK/VBLNK</u> pin, which is selected as <u>VBLNK</u>. In composite sync, <u>CBLNK/VBLNK</u> is selected as <u>CBLNK</u>; in this case, <u>CBLNK</u> outputs the logical-OR of the internal horizontal- and vertical-blanking signals. <u>CBLNK</u> is low if either horizontal- or vertical-blanking is active internally.

Monitors typically require VSBLNK to contain a value less than VTOTAL and greater than VEBLNK.

0

B-file register?	register number:	
I/O register?	address:	C000 0060h

Format	15 0 total of vertical-sync pulses
Description	VTOTAL defines the time at which the vertical-sync pulse begins. The video timing logic compares VTOTAL to VCOUNT to determine when to start the vertical-sync pulse. The vertical-sync pulse starts and VCOUNT is reset to 0 when VCOUNT=VTOTAL and either of these conditions is satisfied:
	HCOUNT=HTOTAL or
	HCOUNT=HTOTAL/2 in the interlaced video.
	The internal vertical-sync pulse begins (if it was not already caused to do so by either of the conditions above) when
	the video timing logic detects an external vertical-sync pulse (when $\overrightarrow{\text{VSYNC}}$ is an input) or
	the video timing logic detects the first external composite serration pulse (when CSYNC is an input).
	When this happens, VCOUNT is reloaded from the SETVCNT register.
	The $\overline{\text{VSYNC}}$ output is driven low to signal the start of the vertical-sync interval. The falling and rising edges of $\overline{\text{VSYNC}}$ coincide with the falling edge of $\overline{\text{CSYNC}}$ or $\overline{\text{HSYNC}}$ (in interlaced video, some transitions of $\overline{\text{VSYNC}}$ occur halfway between 2 HSYNC pulses).
	Set VTOTAL to a value greater than VSBLNK. VTOTAL's maximum value is 65,535.

B-file register?	√register number: B6 address:
Format	<u>31 16 15 0</u>
	A rian of Window's ending address
Description	
Description	WEND defines the XY address of the most significant pixel within the rectangu- lar destination clipping window. WEND must be valid for instructions that use an XY destination address and a nonzero window option. The most significan pixel is the pixel with the highest address within the window. For a screen with the origin in the top left corner of the pixel array, this address corresponds to the pixel in the lower right corner.
	The X and Y portions of the address are signed values; WEND can be at any position in any quadrant of the XY address space. It describes an inclusive pixel; that is, the pixel at the XY location in WEND is included in the window. The value in WEND is used with WSTART, DADDR, and DYDX to preclip pixels lines, and pixel arrays. WEND is not modified by instruction execution.
Which instructions use this register?	Instruction WEND's function
	CPW XY address of most significant window corner
	DRAV XY address of most significant window corner
	FILL XY XY address of most significant window corner
	FLINE, LINE XY address of most significant window corner
	PIXBLT B, XY XY address of most significant window corner
	PIXBLT L, XY XY address of most significant window corner
	PIXBLT XY, XY XY address of most significant window corner
	PIXT Rs, Rd.XY XY address of most significant window corner
	PIXT Rs.XY, Rd.XY XY address of most significant window corner
	TFILL XY address of most significant window corner
Example	WEND .set B6 MOVI [040h, 0100h], WEND ; XY value (256,64)
	; stored in WEND

<u> </u>		
B-file register?	$\checkmark$	register number: B7
I/O register?		address:

Format

31	16	15	0
Y half of window's starting	address	X half of window's star	ting address
L sign of Y coordinate		L sign of X coordinate	

**Description** WSTART defines the XY address of the least significant pixel within the rectangular destination clipping window. WSTART must be valid for instructions that use an XY destination address and a nonzero window option. The least significant pixel is the pixel with the lowest address in the array. For a screen with the origin in the top left corner of the pixel array, this address corresponds to the pixel in the upper left corner.

The X and Y portions of the address are signed values; WSTART can be at any position in any quadrant of the XY address space. It describes an inclusive pixel; that is, the pixel at the XY location in WSTART is included in the window. The value in WSTART is used with WEND, DADDR, and DYDX to preclip pixels, lines, and pixel arrays. WSTART is not modified by instruction execution.

Which instructions use this reaister?	Instruction	WSTART's function
	CPW	XY address of least significant window corner
	DRAV	XY address of least significant window corner
	FILL XY	XY address of least significant window corner
	FLINE, LINE	XY address of least significant window corner
	PIXBLT B, XY	XY address of least significant window corner
	PIXBLT L, XY	XY address of least significant window corner
	PIXBLT XY, XY	XY address of least significant window corner
	PIXT <i>Rs</i> , <i>Rd</i> .XY	XY address of least significant window corner
	PIXT <i>Rs</i> .XY, <i>Rd</i> .XY	XY address of least significant window corner
	TFILL	XY address of least significant window corner
Example	WSTART .set B5	

MOVI [040h, 0100h], WSTART ; XY value (256,64)
; stored in WSTART



# **Instruction Cache and Internal Parallelism**

Most program execution time is spent on repeatedly executing a few main procedures or loops. Program execution can be speeded up by placing these often-used code segments into a fast memory. The TMS34020 uses a 512-byte, on-chip **instruction cache** for this purpose.

To further enhance execution speed, the TMS34020 can access several areas of memory, including the cache, in parallel. Although the TMS34020 stores code and data in a single memory space, the TMS34020's internal parallelism provides benefits that are often found in processors that use separate code and data spaces.

This chapter includes the following topics:

	Sect	ion	Page
Cache information describes the	5.1	Cache Architecture	5-2
architecture and operation of the	5.2	Cache Replacement Algorithm	5-4
instruction cache.	5.3	Cache Operation	5-5
	5.4	Performance with Cache Enabled	
		vs. Cache Disabled	5-9
Internal parallelism describes how the TMS34020's ability to simultaneously access various memory areas improves system performance.	5.5	Internal Parallelism	5-10

# 5.1 Cache Architecture

Figure 5–1 illustrates cache organization.

Figure 5–1. TMS34020 Instruction Cache



Only instruction words (memory words that the PC points to) can be accessed from the cache. This includes

- Opcodes
- Immediate operands
- Displacements
- Absolute addresses

Instructions and data can reside in the same area of memory; therefore, data may occasionally be copied into the instruction cache along with instruction words. However, the TMS34020 cannot access data from the cache; all reads and writes of data in memory bypass the cache.

The instruction cache contains 512 bytes of RAM for storing up to 128 long (32-bit) words (this is equivalent to 256 6-bit instruction words). Each instruction word in cache is aligned on an even long-word boundary.

As Figure 5–1 shows, the cache is divided into four 64-word segments. Each cache segment may contain up to 64 instruction words of a 64-word segment from memory. This memory segment is a block of 64 contiguous instruction words, beginning at an even 64-word boundary in memory.

Each cache segment is further divided into 8 subsegments; each subsegment contains 4 long-words (up to 8 instruction words). Dividing each segment into subsegments reduces the number of word fetches required from memory when fewer than 64 words of a memory segment are used.

Each of the 4 cache segments is associated with a segment start address (**SSA**) register. Figure 5–2 shows how a long-word is partitioned into the components used by the cache-control algorithm.



Figure 5–2. Segment Start Address

The 22 bits of the SSA register correspond to the 22 MSBs of the segment's memory address. These 22 MSBs are common to all 8 subsegments within a segment. The next three bits (bits 6—8) identify one of the 8 subsegments. Bits 4, 5, and 6 identify one of the 8 instruction words within a subsegment. The 4 LSBs are always 0s because instructions are aligned on word boundaries.

# 5.2 Cache Replacement Algorithm

When the TMS34020 requests an instruction word from a segment that is not in the cache, the contents of one of the 4 cache-resident segments must be discarded to make room for the segment that contains the requested word. A modified form of the least-recently-used (**LRU**) replacement algorithm is used to select the segment to be discarded.

The LRU segment manager (part of the cache control logic) maintains an LRU stack to track use of the 4 segments. The LRU stack contains a queue of segment numbers, 0 through 3. Each time a segment is accessed, its segment number is moved to the top of the stack, pushing the other segment numbers down as necessary to make room at the top. Thus, the number at the top of the LRU stack identifies the most-recently-used segment and the number at the bottom identifies the least-recently-used segment.

When a new segment must be loaded into cache, the least-recently-used segment is discarded. The 8 P flags (described in Section 5.3) of the selected segment are set to 0s, and the segment's SSA register is loaded with the new segment address. After the requested subsegment is loaded from memory, its P flag is set to 1, and the requested instruction fetch is allowed to complete.

Following reset, all P flags in the cache are set to 0.

# 5.3 Cache Operation

When the TMS34020 requests an instruction word, it checks to see if the cache contains the word. First, the processor compares the 22 MSBs of the instruction address to the 4 SSA registers. If the TMS34020 fins a match, it searches for the appropriate subsegment. A present (P) flag, associated with each subsegment, indicates the presence of a particular subsegment within a cache segment:

- P=1 indicates that the requested instruction word is in cache. This is called a cache hit.
- □ If there is no match, or if there is a match and P=0, the instruction word is not in cache. This is called a **cache miss**.

# 5.3.1 Cache Hits

A **cache hit** means that the cache contains the requested instruction word. In this case, the TMS34020 performs the following actions:

- 1) Performs a short (one machine state) access cycle to read the instruction word from cache.
- 2) Moves the segment number to the top of the LRU stack, pushing the other three segment numbers toward the bottom of the stack (assuming that this segment was not the most recently used segment).

Because of pipelining, instruction fetches from the cache overlap completion of preceding instructions. Thus, the overhead due to instruction fetches is effectively 0.

## 5.3.2 Cache Misses

A **cache miss** means that the cache does not contain the instruction word. There are two types of cache miss—subsegment miss and segment miss.

#### **G** Subsegment miss

The 22 MSBs of the instruction-word address match one of the 4 SSA registers' 22 MSBs; that is, the appropriate segment is in the cache. However, the P flag for the requested subsegment is not set. The TMS34020 performs these actions:

- 1) Reads into cache the 8-instruction-word subsegment that contains the requested instruction word.
- 2) Moves the segment number to the top of the LRU stack, pushing the other three segment numbers toward the bottom of the stack (assuming that this segment was not the most recently used segment).

- 3) Sets the subsegment's P flag to 1.
- 4) Reads the instruction word from the cache.

#### Segment miss

The instruction word address does not match any of the SSA registers. The TMS34020 performs the following actions:

- 1) Selects the least-recently-used segment for replacement; clears the P flags of all 8 subsegments.
- Loads the SSA register for the selected segment with the 22 MSBs of the address of the requested instruction word.
- 3) Reads into cache the 8-instruction-word subsegment in memory that contains the requested instruction word. This word is placed in the appropriate subsegment of the least-recently-used segment. The TMS34020 sets the subsegment's P flag to 1.
- 4) Adjusts the LRU stack by moving the number of the new segment from the bottom (indicating that it is least recently used) to the top (indicating that it is most recently used). This pushes the other three segment numbers in the stack down one position.
- 5) Reads the instruction word from the cache.

## 5.3.3 Fetching Data into the Cache Following a Cache Miss

Following either type of cache miss, the TMS34020 loads 4 long-words into a cache subsegment. The order in which the TMS34020 fetches these long-words is determined by the position, within the 4 long-words, of the opcode or immediate data that caused the cache miss.



Consider code that starts at address 0 and continues to a high address (such as 010000h). Assume that this code contains no loops or immediate data. When the TMS34020 begins to execute this code, it jumps to the first opcode (at address 0) and finds that the opcode is not in cache. So, it fills the first subsegment with the 8 opcodes that are in the first 4 long-words in memory. The TMS34020 reads the data in this order:

1<sup>st</sup> read:32 bits at address 020h (opcodes 2&3)2<sup>nd</sup> read:32 bits at address 040h (ops 4&5)3<sup>rd</sup> read:32 bits at address 060h (opcodes 6&7)4<sup>th</sup> read:32 bits at address 000h (ops 0&1)

Note that the TMS34020 does not read the words in the expected cyclic order—it reads the long-word with the first opcode *last*, not first. This is the general case.



As another example, consider	the following code se	gment.	
0001 0000000 5600		clr	a0
0002 00000010 5684		clr	a4
0003 00000020 56a5		clr	a5
0004 00000030 56c6		clr	a6
0005 00000040 c000		jruc	next_subseg+48
00000050 0005			
0006 00000060 0300		nop	
0007 00000070 0300		nop	
0008 00000080	next_subseg:		
0009 00000080 0300		nop	
0010 00000090 0300		nop	
0011 000000a0 0300		nop	
0012 000000b0 5673		clr	b3
0013 000000c0 5694		clr	b4
0014 000000d0 56b5		clr	b5
0015 000000e0 56d6		clr	b6
0016 000000f0 56f7		clr	b7
0017 00000100	loop:		
0018 00000100 c0ff		jruc	loop
This example jumps from the m ment. The first subsegment is lo until the TMS34020 encounters address 0B0h. This opcode is n	iddle of the first subs aded into cache as de the jump on line 5. A not in cache, so the T	egment escribed At this pc MS3402	to the middle of the second subseg- in Example 5–1. The code executes bint, control passes to the opcode at 20 loads the next subsegment. The

address 0B0h. This opcode is not in cache, so the TMS34020 loads the next subsegment. The 4 long-words are loaded in cyclic order; the long-word containing the opcode at address 0B0h is read last. The order of memory accesses is

1 <sup>st</sup> read:	32 bits at address 0C0h	2 <sup>nd</sup> read:	32 bits at address 0E0h
3 <sup>rd</sup> read:	32 bits at address 080h	4 <sup>th</sup> read:	32 bits at address 0A0h

Even though the code jumps over the long-word at address 080h, this word is loaded into cache.

# Example 5–3. Code with Immediate Data

Some instructions have immediate data; for example,

```
movi 0ABCDABCDh,A0
```

move @OFFFFFF20h,@OEEEEEE00,0

- ; (32 bits of ; immediate data)
- ; (64 bits of
- ; immediate data)

Immediate data follows the opcode in the object code. If an opcode with immediate data is near the end of a subsegment, the TMS34020 may encounter a cache miss when it attempts to access the immediate data. The next subsegment is loaded with 4 long-words, in cyclic order, so that the long-word containing the immediate data that caused the first cache miss is loaded in last.

## 5.3.4 Self-Modifying Code

Avoid using self-modifying code; it can cause unpredictable results. When a program modifies its own instructions, only the copy of the instruction that resides in external memory is affected. Copies of the instructions that reside in cache are not modified, and the TMS34020 doesn't attempt to detect this.

#### 5.3.5 Flushing the Cache

Flushing the cache sets it to an initial state, identical to the state of the cache following reset: The cache is empty and all 32 P flags are cleared to 0.

You can flush the cache by setting the CF (cache flush) bit in the HSTCTLH register to 1. The CF bit retains the last value loaded until a new value is loaded or until the TMS34020 is reset. The contents of the cache remain flushed as long as the CF bit equals 1. All instruction fetches bypass the cache and are accessed directly from memory.

Unless the cache is disabled, normal cache operation resumes when the CF bit is cleared to 0.

One use for flushing the cache is to facilitate downloading new code from a host processor to TMS34020 local memory. The host typically halts the TMS34020 during downloading by writing a 1 to HLT [[HSTCTLH]]. Before allowing the TMS34020 to execute downloaded code, the host should flush the cache to purge it of stale instructions.

For performance reasons, CF[[CONTROL]] should not remain set to 1 for long periods. While CF=1, only 1 word is fetched at a time.

## 5.3.6 Disabling the Cache

Disabling the cache facilitates program debugging and emulation. The cache is disabled by setting CD[[CONTROL]] to 1. While the cache is disabled, the TMS34020 bypasses the cache and fetches all instructions from external memory.

Setting CD to1 is similar to setting CF to 1. However, when CD=1 and CF=0, data already in the cache is protected from change. When the CD bit is cleared to 0, the prior state of the cache (before CD was set to 1) is restored. The instructions in the cache are once again available for execution. If the cache contents become invalid while CD=1, they can be flushed by setting CF to 1.

For faster execution in some time-critical applications, you may wish to manipulate the CD bit to preserve code in the cache . For example, if an inner loop just exceeds 512 bytes, most of the loop, but not all of it, can fit in the cache. During execution of the few instructions that are not in the cache, you can set the CD bit to 1 to prevent the TMS34020 from replacing the code in the cache. In this instance, the loop's execution speed is improved by eliminating the thrashing of cache contents. Use this technique carefully; in some cases, it can negatively affect execution speed.

# 5.4 Performance with Cache Enabled vs. Cache Disabled

When the instruction cache is disabled, the TMS34020 fetches instruction words from external memory. Assuming no wait states are necessary, each instruction fetch from external memory adds 3 machine cycles to the access time. This is considerably slower than a program that uses the cache efficiently (when each word in cache is used several times before it is replaced).

A less efficient use of cache occurs when words in cache are used only once before they are replaced. This produces a cache miss every eighth word (even in this case, operation is usually much better than operation when the cache is disabled). With the cache enabled, the time penalty due to cache misses in this case is .75 machine states per single-word instruction (compare this to 3 states when the cache is disabled), which is calculated as follows:

- □ 5 machine cycles are required to load 4 long-words (8 instruction words) into cache from memory (in page mode).
- 1 additional machine state is required to begin processing the instruction.
- Dividing the total of 6 machine states by 8 instruction words yields an average of .75 machine states per instruction word.

Performance with the cache enabled is nearly always better than performance with the cache disabled. There are two exceptions:

- □ If the code contains many jumps, only a portion of each subsegment may be executed before control is transferred to another subsegment.
- If an inner loop is larger than the cache, only a portion of the instructions in the inner loop can be contained in the cache at any time. In this situation, you can improve performance by manipulating the CD bit as described in Section 5.3.6.

While the cache is disabled, the TMS34020's internal memory controller fetches each instruction word from memory only as it is requested by the CPU. This differs from operation with the cache enabled, in which case a cache miss causes the entire 8-instruction-word subsegment containing the requested instruction word to be loaded into the cache at once.

# 5.5 Internal Parallelism

Figure 5–3 illustrates the internal data paths associated with TMS34020 processor functions. The TMS34020 uses a single, logical memory space for storing both data and instructions. However, internal parallelism provides the TMS34020 with the benefits found in architectures that use separate data and instruction storage (sometimes referred to as *Harvard architectures*). The ability to fetch instructions from cache in parallel with accessing data from memory greatly enhances execution speed. Hardware parallelism allows the TMS34020 to access these three storage areas simultaneously:

- Instruction cache
- Dual-ported, general-purpose register files A and B
- External memory

### Figure 5–3. Internal Data Paths



The TMS34020 can access each storage area independently of the other two. This allows the TMS34020 to perform the following actions in parallel during a pair of machine states:

- □ 1 external memory cycle
- 2 instruction fetches from cache
- 4 reads and 2 writes to the general-purpose register files

The need to schedule conflicting internal operations can limit the TMS34020's ability to perform these actions in parallel. For example, an instruction that requires the memory controller to perform a read must finish executing before the next instruction can be executed. Figure 5–4 illustrates an example of internal parallelism.

# Figure 5–4. Parallel Operation of Cache, Execution Unit, and Memory Interface

(a) Code

•							
A L1:	MOVE	*B1+,	B10, 0 ;	Get D	ELTAX		
В	ADD	B10, E	38 ;	Adjus	t pixel	point	er
С	PIXT	*B1, *	<b8 ;<="" td=""><td>Draw</td><td>next pi</td><td>xel</td><td></td></b8>	Draw	next pi	xel	
D	ADD	B0, B1	. ;	Add f	ield si	ze	
E	DSJS	B11, I	.1 ;	Loop	N times		

(b) Result



Figure 5–4 (a) shows the inner loop of a graphics routine; Figure 5–4 (b) represents execution of the code in (a). Figure 5–4 (b) shows three activities occurring in parallel:

- Instructions are fetched from cache.
- Instructions are executed through the general-purpose registers and the ALU.

The local memory interface controller performs memory accesses.

The memory controller accesses pixels while the ALU fetches instructions from cache. The memory controller completes a write cycle while execution begins on the next instruction.



,

# Interrupts, Traps, and Reset

The TMS34020 supports 10 interrupts, including reset, and up to 65,536 software traps. These interrupts and traps use a set of 32-bit vector addresses that point to appropriate service routines. The TMS34020 also supports bus-fault conditions and single-step execution through these vectors.

You'll find these topics on the following pages:

	Sect	ion	Page
Basic information includes a	6.1	Related Signals	6-2
summary of related signals and registers, information about	6.2	Related Registers	6-2
	6.3	Enabling and Disabling Interrupts	6-6
of interrupt priorities, and a map of	6.4	Interrupt Priorities and Vector Addresses	6-7
the vector addresses.	6.5	Interrupt Processing	6-9
	6.6	Interrupting Instruction Execution	6-13
Specific information describes the	6.7	External Interrupts 1 and 2	6-15
various types of interrupts and an	6.8	Internal Interrupts	6-16
application for the single-step interrupt.	6.9	The Bus-Fault Interrupt	6-19
	6.10	Interrupting a Host Processor	6-21
	6.11	Traps	6-21
	6.12	Reset	6-22
	6.13	An Application for Interrupts:	
		Debugging Code	6-28

# 6.1 Related Signals

Several of the TMS34020's pins request interrupts. Chapter 2 describes the interrupt signals in detail; they are summarized below for your convenience.

Signals	Descriptions	I/O
BUSFLT	is a bus-fault signal that tells the local-memory controller that an error (or <i>fault</i> ) occurred on the current bus cycle. BUSFLT operates in conjunc- tion with the LRDY signal; if both BUSFLT and LRDY are sampled high during a local-memory cycle, a bus-fault interrupt is generated.	I
HINT	is the interrupt signal that allows the TMS34020 to send an interrupt request to a host processor. This interrupt is activated by setting bits in the HSTCTLL register.	0
LINT1, LINT2	are level-sensitive, active-low inputs. They allow external devices to interrupt the TMS34020.	I
RESET	is the system reset signal. During normal opera- tion, RESET is driven low to reset the TMS34020.	I

## 6.2 Related Registers

Several of the TMS34020's I/O registers provide you with control over interrupts. (Chapter 4 provides detailed descriptions of all the I/O registers.) Some registers contain bits that you must set to enable certain interrupts; others contain bits that the TMS34020 or another device sets to identify an interrupt request.

These registers control interrupt functions:

- The **status register** contains a bit that globally controls interrupts; it also reflects the status of certain interrupts.
- The **INTENB** register is the interrupt-enable register.
- The **INTPEND** register is the interrupt-pending register.
- □ The **HSTCTLL** register is a host-interface register that provides control over general TMS34020-to-host and host-to-TMS34020 interrupts.
- The **HSTCTLH** register is a host-interface register that provides control over the nonmaskable interrupt, halt, and reset.

#### Note:

You can access I/O registers in the same manner as any other TMS34020 memory locations. You can access the status register with the GETST and PUTST instructions.

The remainder of this section describes these registers and tells you which bits are associated with the interrupts. In the pictures of the registers, shaded areas indicate bits that have no interrupt functions.



Setting IE (global interrupt enable) to 1 allows you to globally enable the interrupts that are controlled by the INTENB register. If IE = 0, then interrupts are globally disabled; in this case, the values in the INTENB register have no effect.

- Setting SS (single-step enable) to 1 causes a special interrupt to be generated bit 22 after each instruction is executed. This allows you to single-step through a program.
- The TMS34020 sets IX (interruptible instruction executing) when it takes an interrupt at an interruptible point in an instruction. The TMS34020 uses IX to ensure that instruction execution resumes correctly after returning from the interrupt.
- The TMS34020 sets BF (bus fault) when it takes a bus-fault interrupt. The bit 26 TMS34020 uses this bit to ensure that instruction execution resumes correctly after returning from a bus fault.

INTENB register				ac	Idress	: C000	0110h
<u>15</u>	11	10	9		2	1	0
	WVE	DIE	HIE		X2E	X1E	

The INTENB register allows you to selectively enable or disable interrupts (when IE=1).

Setting WVE (window-violation interrupt enable) to 1 enables the windowbit 11 violation interrupt.

DIE bit 10 HIE bit 9 X2E bit 2 X1E bit 1 Setting DIE (display interrupt enable) to 1 enables the display interrupt.

Setting HIE (host interrupt enable) to 1 enables the host interrupt.

Setting X2E (external interrupt 2 enable) to 1 enables external interrupt 2.

Setting X1E (external interrupt 1 enable) to 1 enables external interrupt 1.

INTPEND register 15	 WVP [ [	<u>10 (</u> DIP H	9 IP	In and a state of the state of	n Allan and an and a star	Bullion Block Block Block	n Malika da ana ana ang ang ang ang ang ang ang an	Hall Market Market	addres	s: C000 0 1 X1P	0 0
	The IN interru	ITPEN pt that	D regis	ster ider equeste	ntifies ed but	pendin has no	ng interru	pts. A <sub>/</sub> n serv	pending	n interrup	ot is an
wvp bit 11	When violatic	the W	VP (wi rrupt is	ndow-v pendir	iolatio g.	on interr	rupt penc	ling) b	it equal	s 1, a wi	ndow-
DIP bit 10	When pendin	the DI g.	P (disp	olay inte	errupt	pendin	ng) bit eq	uals 1	, a disp	lay inter	rupt is
HIP bit 9	When	the HIF	P (host	interrup	otpen	ding) bi	it equals <sup>-</sup>	1, a ho	st interr	upt is pe	nding.
x2E bit 2	When t 2 is pe	the X2 nding.	P (exte	ernal inte	errupt	2 pend	ling) bit e	quals 1	l, an ex	ternal int	errupt
bit 1	When t 1 is pe	he X1 nding.	P (exte	ernal inte	errupt	1 pend	ling) bit e	quals 1	l, an ext	ternal int	errupt
HSTCTLH register 15			3 8	87					address	s: C000 0	100h
				MI RS	T phillippin						

Note: For a complete illustration and description of HSTCTLH, see Chapter 4.

The primary function of these bits is to allow the host processor to interrupt the TMS34020; however, the TMS34020 is also able to write to these bits and may therefore set them itself.



A host processor can halt the TMS34020's on-chip processor by setting the HLT (halt TMS34020 program execution) bit.

bit 15

HSTCTL	_ registe	٢			add	ress: CO	00 00F0h
15	14	13		7	64	3	20
HBREN	HBFI	HRYI		INTOUT	MSGOUT	INTIN	MSGIN
Note: For a complete illustration and description of HSTCTLL, see Chapter 4.							

- MSGIN The MSGIN (message in) field buffers a 3-bit interrupt message from the host Bits 0–2 processor to the TMS34020.
- The host processor can set the INTIN (input interrupt) bit to 1 to generate an interrupt request to the TMS34020.
- MSCOUT The MSGOUT (message out) field buffers a 3-bit interrupt message from the Bits 4--6 TMS34020 to the host.
- The TMS34020 can set the INTOUT (output interrupt) field to send an interrupt Bit 7 request to the host processor.
- The TMS34020 sets the HRYI (host retry interrupt) bit if a retry occurs on a host access. If HBREN=1, setting HRYI sends an interrupt request (via HINT) to the host processor.
- HBFIThe TMS34020 sets the HBFI (host bus-fault interrupt) bit if a bus fault occursBit 14on a host access. If HBREN=1, setting HBFI sends an interrupt request (via<br/>HINT) to the host processor.
- HBREN When the HBREN (host bus-fault/retry interrupt enable) bit is set, setting HRYI Bit 15 or HBFI causes an interrupt request to be sent to the host processor.

# 6.3 Enabling and Disabling Interrupts

The TMS34020 supports 10 interrupts; 6 of the interrupts must be enabled before the TMS34020 can recognize them. These interrupts include

- Single-step interrupt
- External interrupt one
- External interrupt two
- Host interrupt
- Display interrupt
- Window-violation interrupt

Note that only these 6 interrupts can be enabled or disabled; reset, bus fault, NMI, and ILLOP cannot be disabled.

# Enabling an interrupt -

To enable the single-step interrupt, set the SS status bit to 1. The single-step interrupt operates independently of the IE status bit.

To enable any of the other interrupts listed above (excluding the single-step interrupt), follow these steps:

- Step 1: Set the IE status bit to 1 (you can do this by executing an EINT instruction).
- Step 2: Set the appropriate bit in the INTENB register to 1:

To enable this interrupt:	Set this INTENB bit:
external interrupt 1	X1E (bit 1)
external interrupt 2	X2E (bit 2)
host interrupt	HIE (bit 9)
display interrupt	DIE (bit 10)
window-violation interrupt	WVE (bit 11)

# Disabling an interrupt ·

To disable the single-step interrupt, clear the SS status bit to 0. The IE status bit does not affect this interrupt.

To disable any of these other interrupts (excluding the single-step interrupt), you can do one of two things:

- If the IE bit = 1, clear the appropriate bit in the INTENB register.
- Clear the IE bit to 0 (You can do this by executing a DINT instruction). This disables all five of these interrupts, regardless of the values in the INTENB register.

# 6.4 Interrupt Priorities and Vector Addresses

Table 6–1 lists the TMS34020 interrupts by priority. Figure 6–1 shows the interrupts' vector addresses.

Interrupt	Priority	Source	Description
RESET	1	external/ internal	<b>Device reset</b> . Taken when the <b>RESET</b> input signal is asserted low or when the <b>RST[HSTCTLH]</b> bit is set.
BF	2	external	<b>Bus-fault interrupt</b> . External logic generates a bus-fault interrupt by asserting the BUSFLT signal high; LRDY must also be high.
NMI	3	internal	Nonmaskable interrupt. Setting NMI[HSTCTLH]]generates a nonmask- able interrupt.
ні	4	internal	Host interrupt. The host generates this interrupt by setting INTIN [[HSTCTLL]].
DI	5	internal	<b>Display interrupt</b> . The TMS34020's video timing hardware generates the display interrupt.
WV	6	internal	<b>Window-violation interrupt</b> . The TMS34020's CPU generates a win- dow-violation interrupt when a pixel lies either inside a window (windowing mode 1) or outside a window (windowing mode 2).
INT1	7	external	External interrupt 1. Asserting LINT1 low generates this interrupt.
INT2	8	external	External interrupt 2. Asserting LINT2 low generates this interrupt.
SS	9	internal	<b>Single-step interrupt</b> . When the SS status bit is set, this interrupt is gener- ated after each instruction execution.
ILLOP	10	internal	<b>Illegal-opcode interrupt</b> . The TMS34020 generates this interrupt when it encounters an illegal opcode.

Table 6–1. Interrupt Priorities

**Notes:** 1) In order for the TMS34020 to recognize interrupts with priorities 4 through 8, you must set the IE status bit.

Because only the host can set INTIN [HSTCTLL]], HI could be considered an external interrupt. However, HI is not generated directly from an input pin; in keeping with the other interrupts, it is listed as internal.

As Table 6–1 shows,  $\overrightarrow{\text{RESET}}$  has the highest priority. If 2 interrupts are requested at the same time, the highest priority interrupt is serviced first (assuming it is enabled). The bus-fault condition is considered to be in the interrupt priority chain, although it behaves differently from other interrupts.

## Figure 6–1. Vector Address Map

Trap Number	Address	Name	Description
-32768	000F FFE0h	Trap 32,768	
-1	0000 0000h	Trap1	Application defined
	FFFF FFE0h	Reset	Reset
1	FFFF FFC0h	INT1	External interrupt 1
2	FFFF FFA0h	INT2	External interrupt 2
3	FFFF FF80h	Trap 3	
4	FFFF FF60h	Trap 4	Reserved for future
5	FFFF FF40h	Trap 5	hardware or on-chip
6	FFFF FF20h	Trap 6	interrupts
7	FFFF FF00h	Trap 7	
8	FFFF FEE0h	NMI	Nonmaskable interrupt
9	FFFF FEC0h	HI	Host interrupt
10	FFFF FEA0h	DI	Display interrupt
11	FFFF FE80h	WV	Window-violation interrupt
12	FFFF FE60h	Trap 12	_
13	FFFF FE40h	Trap 13	Reserved for future
14	FFFF FE20h	Trap 14	interrupts
15	FFFF FE00h	Trap 15	
16	FFFF FDE0h	Trap 16	
			Application defined
29	FFFF FC40h	Trap 29	
30	FFFF FC20h	ILLOP	Illegal-opcode interrupt
31	FFFF FC00h	Trap 31	Application defined
32	FFFF FBE0h	SS	Single-step/Emulator
33	FFFF FBC0h	BF	Bus fault
34	FFFF FBA0h	Trap 34	
32767	FFF0 0000h	Trap 32,767	Application defined

- **Notes:** 1) Traps –1 through –32,768 use the memory at the bottom of the address space for vector addresses. Traps 0 through 32,767 use the memory at the top of the address space.
  - 2) Traps 0 through 31 may be accessed by either a TRAP or TRAPL instruction.
  - 3) Traps –1 through –32,768 and 32 through 32,767 are accessed only by TRAPL.
  - 4) Traps 3 through 7 and 12 through 15 are reserved for future interrupts.

# 6.5 Interrupt Processing

When an interrupt has been requested but has not yet been processed, it is called a *pending* interrupt. If a pending interrupt is enabled (and no interrupt with a higher priority is also pending), the TMS34020 accepts the interrupt at the end of the current instruction cycle (or at the next interruptible point within instruction execution). Figure 6–2 lists the actions that the TMS34020 takes when an interrupt occurs.

Figure 6–2. Actions Performed When the TMS34020 Takes an Interrupt

- 1) If necessary, the TMS34020 pushes onto the stack any temporary registers that the current instruction is using. This allows the instruction to resume execution correctly upon return from the interrupt.
  - a) If the interrupt is taken part-way through an interruptible instruction, the TMS34020 pushes twenty-four 32-bit words. (If the SP is not word aligned when the interrupt is taken, the TMS34020 may push another long word on as padding.)
  - b) If the interrupt is caused by a bus fault, the TMS34020 pushes thirty-one 32-bit words.
- 2) The TMS34020 pushes the PC onto the stack.
- 3) If necessary, the TMS34020 modifies the ST so that instruction execution can resume correctly after returning from the interrupt.
  - a) If the interrupt is taken part-way through an interruptible instruction, the TMS34020 sets the IX bit.
  - b) If the interrupt is caused by a bus fault, the TMS34020 sets the BF bit.
- 4) The TMS34020 pushes the ST onto the stack.
- 5) The TMS34020 modifies the contents of the ST as follows:



- 6) The TMS34020 fetches the interrupt vector from external memory and places it into the PC.
- 7) The TMS34020 begins executing the instruction pointed to by the new PC value.

When the first instruction of the service routine begins execution, the new status register contents imply the following conditions:

- All interrupts (except BF, NMI, and reset) are disabled.
- Field 0 is 16 bits long and zero-extended.
- Field 1 is 32 bits long and zero-extended.
- Gingle-stepping is disabled.

If a graphics instruction is interrupted, the TMS34020 *does not save* the B-file registers used as implied operands (it doesn't push them onto the stack). If your interrupt service routine needs to use these registers, the routine should first push them onto the stack, then pop them from the stack before returning.

You may usually want an interrupt service routine to complete before allowing any more interrupts. However, if you want to be able to interrupt a service routine, the routine should

**Step 1:** Set the IE status bit to 1.

Step 2: Set the appropriate bits in the INTENB register.

The service routine can also load new field sizes, if required.

If you want to single-step through the interrupt service routine, you can do so by setting the SS status bit.

# 6.5.1 Returning from an Interrupt Service Routine

Interrupt service routines should not assume anything about the state of the stack except that the stack contains the PC, the ST, and possibly some extra words (as outlined in Figure 6–2, item 1). The interrupt service routine *must* return using a RETI or RETM instruction. Only these instructions pop the PC, ST, and any extra words from the stack to their correct internal locations, thereby enabling instruction execution to proceed from the point at which the interrupt occurred. Note that RETS cannot be used, because it pops only the PC.

Figure 6–3. Actions Performed When the TMS34020 Executes a RETI or RETM Instruction

- 1) The TMS34020 pops the value of the ST from the stack.
- 2) The TMS34020 pops the value of the PC from the stack.
- 3) If necessary (as indicated by the IX and BF bits of the restored ST value), any extra words that were pushed onto the stack are popped from it.
- 4) If the restored IE bit is 1, the TMS34020 takes one of the following actions, depending on the instruction used:
  - RETI does not alter the restored value of IE. If another interrupt is pending, it is taken as soon as RETI completes, before the TMS34020 can resume execution of the interrupted program.
  - RETM masks IE during the last machine state of the return. This has the effect of not enabling interrupts until one machine state after RETM completes, which means that even if another interrupt is pending, the TMS34020 resumes execution of the interrupted program. The interrupt is then taken at the next interruptible point within the program. RETM is used primarily with the single-step interrupt (see Section 6.13, page 6-28).

The RETI and RETM instructions perform the actions described in Figure 6–3. Provided the interrupt routine has not changed any of the values on the stack, this restores the CPU to its state immediately prior to taking the interrupt. Under no circumstances should you change the value of any of the additional words that may have been pushed onto the stack.

If the cause of an interrupt remains when the TMS34020 completes execution of RETI or RETM, the interrupt is taken again. When necessary, the interrupt service routine should take the appropriate steps to clear the cause of the interrupt. Consideration is given to this in Sections 6.7 to 6.9, which discuss each of the interrupts in detail.

## 6.5.2 Interrupt Latency

The delay between when an interrupt request is made and when the TMS34020 begins servicing the interrupt depends primarily on what activity the TMS34020 is performing at the time. The delay can be broken down into a number of smaller delays, which fall into four categories. These are listed below.

- Delay 1: Interrupt request recognition. This is the period between the time the interrupt is requested and the time the interrupt is recognized. This is one machine state for interrupts generated synchronous to the TMS34020 (such as HI, NMI, and WV), and one to two machine states for interrupts generated asynchronously to the TMS34020 (such as INT1, INT2, and DI).
- Delay 2: CPU response time. This is the time required for an instruction that was already executing when the interrupt was recognized either to complete or to reach the next interruptible point. This depends on the instruction. The instruction timings in Chapter 15 provides details of the delay possible for each instruction.
- Delay 3: Interrupt context switch. This is the time required to push the PC, ST, and any extra words required onto the stack (as detailed in Figure 6–2), and read the appropriate interrupt vector from memory.
- Delay 4: Local memory traffic. Any other memory controller activity that occurs while the CPU is completing the current instruction or performing the context switch has an adverse effect on Delay 3, and affects Delay 2 if the executing instruction (such as a PIXBLT or FILL) performs many local-memory cycles. You should determine what percentage of local-memory bus bandwidth is taken up with screen refresh, DRAM refresh, and host local-memory cycles, then increase the delay produced by Delay 2 + Delay 3 by this amount.

Table 6–2 summarizes these delays and gives some best- and worst-case figures. Because the local-memory interface is typically the limiting factor in the calculations, two worst-case conditions are described; one with 32-bit page-mode memory, and the other with 16-bit non-page-mode memory. These figures are intended only to give you some idea of the delay involved in servicing an interrupt. Obviously, other factors not discussed here could further delay the interrupt: for instance, if another interrupt, which maintained IE=0 through-out its service routine (disabling other interrupts), was being serviced at the time the interrupt being considered here occurred. Inserting wait states into the local-memory cycles also increase the delays. The precise effect of this is difficult to estimate, because not all the delays are determined by local-memory interface performance.

#### *Table 6–2. Sources of Interrupt Delay*

		Latency (in states)				
Delay	<sup>у</sup> Туре	Minimum	Maximum A	Maximum B		
Inter	rupt recognition	1	2	2		
CPU response time (note 3)		0	32	80		
Inter	rupt context switch	12				
	Interruptible instruction		48	129		
<b>G</b>	Bus-fault interrupt		55	157		
Loca	al memory traffic	0				
	Per screen refresh		2	2		
	Per DRAM refresh		3	3		
	Per host access		2	4		

Notes: 1) Maximum A assumes 32-bit wide memory, which supports page mode.

- 2) Maximum B assumes 16-bit wide memory, which does not support page mode, with no wait states.
- This is for the worst-case instruction (PIXBLT XY,XY). Other instructions are less. See Chapter 13, Assembly-Language Instruction Set, for more details.
- 4) If both the host and the CPU request accesses as frequently as they can, the memory controller priorities are arranged so that they each receive alternative local-memory cycles.

# 6.6 Interrupting Instruction Execution

When an instruction is interrupted, instruction execution is suspended until the interrupt routine completes. At this point, the instruction resumes an finishes executing.

While an instruction is executing, the state of the instruction is stored in inaccessible internal registers. When an interrupt occurs, the contents of these registers must be moved to memory before the TMS34020 begins executing the interrupt routine, so that the instruction state can be restored when the interrupt routine completes.

The following events take place when an instruction is interrupted.

- **Step 1:** The IX[[ST]] (interruptible instruction executing) bit is set to 1. This indicates that the interrupt occurred while an instruction was executing.
- **Step 2:** The contents of any internal temporary registers are pushed onto the stack.
- Step 3: The PC and ST are also pushed onto the stack.
- **Step 4:** Control branches to your TRAP routine, which should use the MMTM instruction to stack any register values that need to be preserved for later use outside the trap routine.
- Step 5: At the end of the TRAP routine, you should use the MMFM instruction to restore the stacked register file values and execute a RETI instruction. (RETI marks the end of the TRAP routine.) Executing RETI returns control to the interrupted program, popping the ST and PC from the stack. When the IX bit is detected, the internal register values are also popped from the stack, and the interrupted instruction resumes execution. RETI clears IX.

Note that the graphics instructions described in Chapter 12 may take several thousand machine cycles to execute, depending on the size of the lines and arrays involved. These instructions check for interrupts at regular intervals, preventing delays to high-priority interrupts from becoming prohibitively long.

#### Note:

- 1) IX is not set to 1 when a PIXBLT or FILL instruction is *aborted* as a result of a window violation. In this case, returning from an interrupt routine causes the TMS34020 to execute the instruction that *follows* the interrupted instruction.
- 2) If a graphics instruction is executing when a *bus fault* occurs, the steps described on page 6-13 take place. In addition, BF[[ST]] is set to 1. In this case, executing RETI at the end of the bus-fault TRAP routine also clears BF.
- 3) If the SP is not long-word aligned when an instruction is interrupted, then the TMS34020 aligns the stack to expedite the interrupt sequence. RETI always restores the SP to its original alignment.
- 4) The FPIXEQ and FPIXNE instructions are exceptions. These two instructions follow the same basic operation when interrupted, but they do not preserve the contents of temporary registers (skipping step 2, listed on page 6-13). Instead, these instructions reset their operands so that they can resume execution when control returns from the interrupt.

# 6.7 External Interrupts 1 and 2

The TMS34020 supports two general-purpose interrupts that allow external devices to interrupt the TMS34020. This is achieved by driving  $\overline{\text{LINT1}}$  or  $\overline{\text{LINT2}}$  low. The interrupts generated by requests to  $\overline{\text{LINT1}}$  and  $\overline{\text{LINT2}}$  are called INT1 and INT2. Table 6–3 lists these interrupts, the signals that generate them, and the interrupt trap vectors.

#### Table 6–3. External Interrupt Vectors

Name	Input Pin	Vector Address	
INT1	LINT1	FFFF FFC0h	
INT2	LINT2	FFFF FFA0h	

Each signal is dedicated to an individual interrupt, allowing 2 separate and distinct external-interrupt requests. INT1 has a higher priority than INT2; if  $\overline{\text{LINT1}}$  and  $\overline{\text{LINT2}}$  become active at the same time and both external interrupts are enabled, INT1 is serviced first.

X1P[INTPEND] and X2P[INTPEND] reflect the current state of the  $\overline{\text{LINT1}}$  and  $\overline{\text{LINT2}}$  inputs. A bit equals 1 if the corresponding request is active, 0 if it is not. You can poll these bits to detect transitions at the interrupt inputs.

Once an external devices request an interrupt, the device should continue to drive the interrupt signal low until the TMS34020 has started to execute the interrupt service routine.

- If the device permits the interrupt pin to go inactive high before the routine recognizes the interrupt, the request may be missed.
- □ If the active level is maintained after the interrupt service routine completes, the interrupt is taken again.

How you ensure that the interrupt pin is held active until after the beginning of the service routine depends on the application. However, two possibilities are

- The interrupt service routine writes to an external location to cause the appropriate interrupt pin to be deactivated.
- External hardware decodes the vector fetch for the interrupt from the status code and vector address output on the LAD bus.

The TMS34020 assumes that signals input to  $\overline{\text{LINT1}}$  and  $\overline{\text{LINT2}}$  are asynchronous to the TMS34020 local clocks; the TMS34020 synchronizes the signals before it processes them. The TMS34020 samples the state of the  $\overline{\text{LINT1}}$  and  $\overline{\text{LINT2}}$  inputs at each high-to-low transition of LCLK1 and updates the X1P and X2P bits accordingly. The delay from the transition at the input to the corresponding change in the X1P or X2P bit is from 1 to 2 states, depending on the transition's phase relationship to LCLK1.

# 6.8 Internal Interrupts

Table 6–4 summarizes the internal interrupts.

Table 6–4. Interrupts That are Associated with Internal Events

Name	Function	Trap #	Vector Location
NMI	Nonmaskable interrupt	8	FFFF FEE0h
HI	Host interrupt	9	FFFF FEC0h
DI	Display interrupt	10	FFFF FEA0h
WV	Window-violation interrupt	11	FFFF FE80h
SS	Single-step interrupt	32	FFFF FBE0h
ILLOP	Illegal-opcode interrupt	30	FFFF FC20h

If more than one interrupt is pending, the interrupts are serviced according to the priorities listed in Table 6–1 (page 6-7).

#### 6.8.1 The Nonmaskable Interrupt (NMI)

The nonmaskable interrupt occurs when a 1 is written to NMI[[HSTCTLH]] (this is normally done by a host processor). This interrupt cannot be disabled and always occurs as soon as possible following the request. The NMI is delayed only for completion of an instruction already in progress, or until the next interruptible point of an interruptible instruction (such as a PIXBLT) is reached.

NMIM[[HSTCTLH]] (NMI mode bit) determines whether context information is saved on the stack when a nonmaskable interrupt occurs:

- If NMIM = 0, the PC and ST are pushed on the stack before servicing the interrupt.
- If NMIM = 1, nothing is saved on the stack before servicing the interrupt.

The TMS34020 automatically clears the NMI bit when it takes the interrupt, so there is no need for the interrupt service routine to do this.

## 6.8.2 The Host Interrupt (HI)

The host interrupt occurs when a 1 is written to INTIN[[HSTCTLL]], providing that HI is enabled (HIE=1). Only the host processor can do this; the CPU cannot write a 1 to INTIN. The host interrupt is serviced as soon as possible following the request.

The MSGIN[[HSTCTLL]] bits provide a mechanism for specifying the action taken by the host interrupt; by checking the value of MSGIN at the beginning of the interrupt routine, you can branch into one of up to eight different procedures. The host can write to MSGIN when it sets INTIN. Only the host can write to these bits; a write by the CPU has no effect.

Before returning from the interrupt, the service routine must clear INTIN. This is the only way INTIN can be cleared, because the host cannot write a 0 to INTIN. Clearing HIP[INTPEND] does not clear the interrupt.

### 6.8.3 The Display Interrupt (DI)

The display interrupt coordinates processing activity with display refreshes. The display-interrupt request becomes active when a particular display line, specified in the DPYINT register, has been output to the monitor screen. At the start of each horizontal blanking period, the VCOUNT register is compared to the DPYINT register. When the vertical-count value in VCOUNT has reached the value in DPYINT, the TMS34020 generates a display interrupt request. If enabled, the interrupt is taken.

Before returning, the display interrupt service routine should clear DIP[[INTPEND]] so that the interrupt isn't taken again.

## 6.8.4 Window-Violation Interrupt (WV)

The window-violation interrupt may occur when the TMS34020 is executing a graphics operation and windowing option 1 or 2 is selected. W[[CONTROL]] defines the window-checking option. WVP[[INTPEND]] is set if

W=1 and an attempt is made to write a pixel inside the specified window, or

**W=2** and an attempt is made to write a pixel *outside* the specified window.

Before returning, the window-violation interrupt service routine should clear WVP[[INTPEND]] so that the interrupt is not taken again.

#### 6.8.5 The Single-Step Interrupt

The single-step interrupt provides a mechanism for executing instructions one at a time. This is useful when developing and debugging new programs. While SS=1, an interrupt is generated after each instruction is executed. Unlike other interrupts, the single-step interrupt is not taken at interruptible points within an instruction, only on instruction boundaries.

The single-step interrupt service routine should use the RETM instruction to return. This allows the next instruction to be executed before the interrupt is taken again. If RETI is used, the single-step interrupt is taken again as soon as the RETI completes, and the program is not executed at all; only the interrupt service routine is executed.

Because SS is contained in ST, and ST is saved on the stack before servicing the interrupt routine, and then restored afterwards, the only methods for clearing the SS bit are

- To modify the value of the ST on the stack during the single-step service routine, so that when the ST is restored by RETM, SS is cleared.
- **u** To single-step a PUTST instruction within the program which clears SS.

#### Note:

All interrupts clear the SS bit so that interrupt service routines execute normally (the instructions in the routine aren't single-stepped). If you want to singlestep through an interrupt service routine, one of the routine's first instructions should set the SS bit.

#### 6.8.6 Illegal-Opcode Interrupts

The TMS34020 recognizes several reserved opcodes as illegal. If the TMS34020 encounters one of these opcodes, it traps to vector number 30 (located at memory address FFFF FC20h). An illegal opcode is similar in effect to a TRAP 30 instruction. The illegal-opcode interrupt cannot be disabled.

For testing purposes, opcodes 0000h and FFFFh are reserved as illegal opcodes on all TMS340 family devices. Other currently illegal opcodes may be used for special functions on future TMS340 devices.

A typical cause of an illegal-opcode interrupt is that the program being executed is corrupted (perhaps because insufficient stack space was allocated). If you wish to resume execution of the program, the interrupt service routine should take whatever steps are necessary to remove the illegal opcode from the program, and then flush the cache before resuming. If the cache is not flushed, the illegal opcode is executed again, because it is still present in the cache.

# 6.9 The Bus-Fault Interrupt

The bus-fault interrupt provides a mechanism by which the TMS34020 can be interrupted by its local memory system. This allows correction of an error that occurred during an access to a particular location. The precise locations (or groups of locations) within the local-memory cycle that can generate a bus fault depend entirely on the application. Here are some examples of the use of the bus-fault mechanism:

- It can indicate that the TMS34020 is attempting to access invalid areas of memory.
- Let can indicate that the TMS34020 is attempting to access protected devices.
- It can indicate that the TMS34020 is attempting to access an area of memory implemented as virtual memory space that is not currently mapped into the physical memory.

A bus fault may be generated as one of the options for ending a local-memory cycle. Asserting the BUSFLT and LRDY pins high at the rising edge of LCLK2 during the data subcycle causes a bus fault to occur. This is discussed in detail in Section 8.6, <u>Ending a Local-Memory Cycle</u> (page 8-12). Bus faults can be generated on virtually all types of local-memory cycle. However, a bus-fault interrupt is generated only if the local-memory cycle was initiated by the CPU.

## 6.9.1 Activity During a Bus-Fault Interrupt

Unlike any other interrupt, a bus fault does not occur at an instruction boundary or at an interruptible point within an instruction. By definition, it occurs *during* an instruction and must be serviced before program execution can continue. However, the pipelining of data between the CPU and the memory controller means that when a bus fault occurs on a memory cycle, the CPU may already be requesting the *next* memory operation. If this is the case, the CPU will have discarded the information relevant to the bus-faulted memory cycle. Because of this, the CPU cannot stack information relating to the bus-faulted access, and so the memory controller must save its own state at the time of the bus-faulted access. This means that the sequence of events when a bus fault occurs is slightly different from other interrupts:

- 1) The memory controller saves the value of the LAD bus from the bus-faulted cycle in the 32-bit BSFLTD register.
- 2) The memory controller saves its own state in BSFLTST.
- 3) The memory controller generates a bus-fault interrupt to the CPU.
- 4) The CPU responds to this interrupt *immediately* (within one machine state) and proceeds to push its state onto the stack as outlined in Figure 6–2.
The bus-fault service routine should start from the address stored at the bus-fault vector (address FFFF FBC0h). The interrupt service routine should take the action necessary to clear the cause of the bus fault. Upon returning from the bus-fault routine, the following actions occur:

- 1) The CPU restores its state from the stack as outlined in Figure 6–3.
- 2) The CPU signals the memory controller to resume normal execution.
- 3) The memory controller restores its state to that of the bus-faulted access from the BSFLTST register.
- 4) The memory controller restarts the faulted access. If the access was a write, the data stored in the BSFLTD register is driven out on the LAD bus during the cycle.

If you do not want to restart the memory cycle that was originally bus-faulted, your interrupt service routine should write FFFFh to the BSFLTST register. This causes the memory controller to restore its inert state (no local-memory accesses pending). Do not, under any circumstances, modify the BSFLTST register to any other value; doing so causes unreliable operation.

#### 6.9.2 Bus Fault System Considerations

Because the memory controller saves the state of the bus-faulted memory access in the BSFLTD and BSFLTST I/O registers, and not on a stack, you should ensure that bus faults cannot be generated when accessing the system stack. If a bus fault occurs while the CPU is pushing its state onto the stack before servicing the bus-fault interrupt (or popping its state off the stack after returning from the bus-fault interrupt), the state of the original bus-faulted memory cycle is lost.

If it is possible for a bus fault to occur while you are executing the bus-fault service routine, you should ensure that one of the first operations performed by the service routine is to push the BSFLTD and BSFLTST registers onto the stack. These should be restored before returning from the bus fault. Care should be taken to ensure that any other interrupts which could occur before the service routine has stacked these registers (such as an NMI) cannot access locations that could generate a bus fault.

# 6.10 Interrupting a Host Processor

The TMS34020 has an output pin dedicated to interrupting a host processor. The TMS34020's CPU can interrupt the host by writing a 1 to INTOUT [[HSTCTLL]]. When this occurs, the HINT pin is asserted active low. It remains at this level until INTOUT is cleared. Only the host can clear INTOUT; only the TMS34020 can set it.

The MSGOUT [[HSTCTLL]] bits provide a mechanism for specifying the action taken by the host when it takes the interrupt; by reading the value of MSGOUT at the beginning of the interrupt routine, the host can branch into one of up to eight different procedures. The CPU can write to MSGOUT when it sets INTOUT. Only the CPU can write to these bits; a write by the host has no effect.

In addition, provided that HBREN [[HSTCTLL]] (host bus-fault/retry interrupt enable) is set, HINT is driven low whenever a retry or bus fault occurs during a host-initiated local-memory cycle. When one of these situations arises, the TMS34020's memory controller sets either HRYI [[HSTCTLL]] (host retry interrupt) or HBFI [[HSTCTLL]] (host bus-fault interrupt). HINT remains active until the host clears the appropriate bit (HRYI or HBFI). Clearing HBREN also causes HINT to be driven inactive.

# 6.11 Traps

The TMS34020 supports 65,536 software traps, numbered –32,768 through 32,767. Software traps behave similarly to interrupts, except that they are initiated when the TMS34020 executes a TRAP or TRAPL instruction. The TRAP instruction provides access to traps 0—31 by using a single 16-bit instruction word. The TRAPL instruction allows access to all 65,536 traps. Unlike an interrupt, a software trap cannot be disabled.

When the TMS34020 executes a TRAP or TRAPL instruction, it performs the same sequence of actions that it performs for interrupts. All traps except trap 0 push the status register and the PC onto the stack. Trap 0 is similar to a hardware reset because it does not push the status register or PC onto the stack; it differs from a hardware reset because it does not set the TMS34020's internal registers to a known initial state. Trap 8 is similar to an NMI interrupt, except that NMIM[[HSTCTLH]] does not affect instruction execution. The status register and PC are stacked unconditionally when trap 8 is executed.

A 32-bit vector address is associated with each software trap. Here's how you determine the vector address for a trap number n:

If n = 0 through 32,767, subtract 32*n* from FFFF FFE0h.

If n = -1 through -32,767, subtract 32 from -32n.

Figure 6–1 on page 6-8 shows the vector addresses for the software traps.

## 6.12 Reset

Reset is the highest priority interrupt; it puts the TMS34020 into a known initial state. There are two ways to invoke reset.

- Assert an active-low level on the RESET pin. At power-up, RESET should always be asserted for a minimum of forty local clock periods after power levels have become stable. At other times, you may reset the TMS34020 by asserting RESET for a minimum of four local clock periods.
- ❑ Write a 1 to RST [HSTCTLH]. This achieves the same result as asserting the RESET pin, but without affecting any other devices in the system to which the RESET pin may be wired. Reset should not be invoked in this way at power-up.

Unlike other interrupts and software traps, reset does not save the previous ST and PC values on the stack. The value of the stack pointer just before a reset may be invalid. Saving these values on the stack could corrupt valid memory locations.

#### 6.12.1 Activity During Reset

When reset is initiated, the TMS34020 takes 34 local clock periods to completely initialize itself (40 cycles must be allowed at power-up because the TMS34020 may be in an illegal state not achievable during normal operation). Most of this time is spent clearing the I/O registers, which are cleared at the rate of two 16-bit registers per machine state. While the RESET pin is asserted, the local-memory control signals are in the states shown in Table 6–5.

Table 6–5.	Initial State of	Output Pins while	RESET and GI are Low

Outputs to Higl	s Driven n Level	Outputs Driven to Low Level	Bidirectional Pins Driven to High-Impedance
RAS CAS3—CAS0 WE TR/QE DDOUT ALTCH	HINT R1 HOE HDST RCA12—RCA0 SF	HRDY CBLNK/VBLNK DDIN Ro	VSYNC HSYNC CSYNC/HBLNK LAD31—LAD0

Note: When GI is high, all GI-controlled pins are driven to high impedance. GI-controlled pins include RAS, CAS0—CAS3, WE, TR/QE, DDOUT, DDIN, ALTCH, HOE, HDST, RCA0—RCA12, LAD0—LAD31, and SF.

The specifications for certain DRAM and VRAM devices require that at power-up the RAS signal be driven inactive-high for 1 millisecond after power becomes stable. As long as RESET is maintained active, the TMS34020 drives its RAS and CAS signals inactive-high. In general, holding RESET low for *t* microseconds ensures that RAS remains high initially for  $t-(10-t_q)$  microseconds, where  $t_q$  is one quarter of the local clock period.

At times other than power-up, the TMS34020 may be in the process of resetting itself while RESET is high. This is the case if reset was initiated by setting the RST bit or by asserting RESET for less than 34 LCLK periods. If RESET is high and the TMS34020 is internally resetting itself, the memory controller performs consecutive DRAM-refresh cycles. This ensures that the DRAM contents are maintained while the TMS34020 is reset.

The value of the REFADR register is output as a pseudo-address during each DRAM-refresh cycle. REFADR is incremented after each DRAM refresh cycle. However, if the DRAM refreshes start before REFADR is cleared, the address output reverts to zero when this occurs, and then starts incrementing again.

## 6.12.2 Initial State Following Reset

The TMS34020 completes its reset procedure when RESET is deactivated, or thirty-four local clock periods after the high-to-low transition of the RESET pin, which ever occurs last. Immediately following reset, the TMS34020 is in the following state:

#### Registers

- All I/O registers are cleared to 0000h. The only possible exceptions to this are HLT[[HSTCTLH]] (see Section 6.12.4), REFADR (which will have incremented if DRAM refreshes were performed during reset), and SCOUNT (if SCLK is oscillating during reset).
- The general-purpose register files A and B are uninitialized.
- The ST is set to 0000 0010h.
- The PC is uninitialized.

#### **Cache**

- The cache SSA (segment start address) registers are uninitialized.
- The cache LRU (least recently used) stack is set to the sequence 0, 1, 2, 3. This indicates that segment 0 is the most recently used, and segment 3 is the least recently used.
- All cache P (present) flags are cleared. This indicates that the cache is empty.
- The DRAM refresh-pending counter is set to 9.

#### 6.12.3 Activity Following Reset

Immediately following reset, the memory controller begins normal operation. At this time, the refresh-pending counter is set to 9. Four or more DRAM refreshes take priority over any CPU memory request, so the memory controller performs DRAM refreshes continuously until the pending counter counts down to 3. At this point, any pending CPU memory request (in this case the reset-vector fetch) is performed. Because reset sets the TMS34020 to increment the refresh-pending counter every 8 machine states, additional DRAM refreshes are requested before the 9 counts down to 3. This results in a total of 9 consecutive DRAM refreshes, which occur before any CPU-initiated memory requests are performed. The remaining refreshes are performed using the normal memory controller priority scheme.

This fulfills most DRAM/VRAM requirements that the DRAM/VRAM's RAS pins (after being held inactive for 1 ms) are cycled a minimum of 8 times after power-up before any memory accesses are made. This ensures that the DRAMs/VRAMs are initialized for correct operation.

Note, however, that if a host requests access before the 8 DRAM refreshes complete, the host request are performed. Thus, at power-up, the host should not make any requests to DRAM memory until the 8 initialization cycles have had time to complete.

If at other times, reset is initiated by setting RST [HSTCTLH] or by asserting RESET in a manner that maintains the data in the DRAMs. There is no need for the host to delay making a memory request, because the DRAMs will already be initialized.

If you initiate reset by asserting RESET, the memory will be maintained if RESET is not held low for longer than the maximum refresh interval less the time taken for the TMS34020 to refresh the memories.

After reset completes and the 8 DRAM-refresh cycles are performed, the TMS34020 either

begins executing instructions (self-bootstrap mode), or

halts until the host clears HLT [[HSTCTLH]].

The level on the HCS pin just before RESET's low-to-high transition selects between these two modes. The TMS34020 remembers this information, so that if reset is initiated via RST[[HSTCTLH]], the CPU is configured in the mode indicated at the most recent rising edge of RESET.

#### 6.12.3.1 Self-Bootstrap Mode

In self-bootstrap mode, the TMS34020 begins executing instructions immediately following reset. This mode is typically used in a system in which the reset vector and reset service routine are contained in nonvolatile memory (such as a bootstrap ROM). This type of system does not necessarily require a host processor, and the TMS34020 may be responsible for performing host-processor functions for the system.

The TMS34020 is configured in self-bootstrap mode when the  $\overline{\text{HCS}}$  pin is low just before  $\overline{\text{RESET}}$ 's low-to-high transition. The low level on  $\overline{\text{HCS}}$  does not alter the HLT bit, which was cleared to 0 during reset. Immediately following the end of reset and the 9 DRAM-refresh cycles, the TMS34020 fetches the level-0 vector address (from address FFFF FFE0h) and begins executing the reset interrupt routine.

Transitions of the HCS and RESET signals are assumed to be asynchronous with respect to the TMS34020 local clock. HCS and RESET are internally synchronized to the local clock by being held in latches for at least 1 clock period before the TMS34020 uses them. The delay through the synchronizer latch is from 1 to 2 local clock periods, depending on the phase of the signal transitions relative to the clock. TMS34020 on-chip logic delays the HCS low-to-high transition to ensure that it is detected **after** RESET's low-to-high transition. The level of the delayed HCS signal at this time determines the value of the HLT bit. In systems without a host processor, this allows HCS and RESET to be wired together without the need for any external logic to delay the transition on the HCS pin.

#### 6.12.3.2 Host-Present Mode

Host-present mode assumes that a host processor is connected to the TMS34020's host-interface pins. In this mode, the TMS34020 local memory can be composed entirely of RAM. Following reset, the host processor may download the initial program code, interrupt vectors, etc., before allowing the TMS34020 to begin executing instructions.

Here's how the TMS34020 is configured in host-present mode. The HCS input is sampled on the trailing edge of RESET. If HCS is inactive high, internal logic forces the HLT bit to a 1. In this fashion, the TMS34020 is automatically halted following reset, and does not begin executing its reset service routine until the host processor clears HLT to 0. In the meantime, the host processor can load the memory and I/O registers with the appropriate initial values before the TMS34020 begins executing instructions. This may, for example, include writing the reset vector and reset service routine into the TMS34020's memory.

No additional external logic is required to force  $\overline{\text{HCS}}$  high before  $\overline{\text{RESET}}$ 's low-to-high transition. External decode logic is typically used to drive the  $\overline{\text{HCS}}$  input active low only when the TMS34020 is addressed by the host processor. Assuming that the host processor is not actively chip-selecting the TMS34020 at the end of reset,  $\overline{\text{HCS}}$  is high.

#### 6.12.4 System Configuration Following Reset

Before any memory locations can be accessed, or instructions executed, the TMS34020 must be configured in the correct addressing mode. Two different aspects must be considered:

- Little-endian or big-endian addressing. This determines which bit within a word is addressed as the least significant (see Chapter 3 more details).
- The base array size of the DRAMs and VRAMs in the system. This determines how logical address bits are mapped to the RCA bus to form the row and column addresses used by DRAMs and VRAMs in the system (see Section 8.16.2, page 8-51, for details).

The mode bits that determine these configurations are BEN[[CONFIG]] (big-endian enable) and RCM0-RCM1[[CONFIG]] (RCA mode).

The TMS34020 provides a mechanism for ensuring that these bits are set correctly following reset.

Reset normally clears BEN and RCM0—RCM1, configuring the TMS34020 to little-endian operation with a DRAM base array size of  $64K \times n$ . Before these bits are set correctly, the only memory locations that can be reliably accessed are those with a row address of all 1s or all 0s and are long-word-aligned, 32-bit words (locations with a bit address of zero, and a field size of 32). The reset vector fulfills these characteristics. This is the first location accessed after reset completes (assuming that the TMS34020 is configured in self-bootstrap mode).

The reset vector contains the address of the first instruction to be executed. The reset vector's 4 LSBs are not required to specify this address because instructions must be aligned to 16-bit word boundaries in memory. The TMS34020 takes advantage of this fact. You should write the values you choose for BEN and RCM0—RCM1 in bits 0—3 of the reset vector. When the reset vector is fetched, the TMS34020 automatically copies these bits into CONFIG and then sets the 4 LSBs of the PC to zero.

Bit 3 of CONFIG is the CBP (configuration byte protect) bit. When a 1 is written to this bit, CONFIG's LSbyte is write-protected until the next time the TMS34020 is reset. You can write the value of this bit to bit 3 of the reset vector; it is also copied into CONFIG with the BEN and RCM0—RCM1 bits. By setting bit 3 of the reset vector to 1, you can ensure that this is the only time BEN and RCM0—RCM1 are modified.

If the TMS34020 completes reset in host-present mode, the host *must* write to BEN and RCM0—RCM1 *directly* before accessing the TMS34020's local memory (unless the system configuration requires these bits to remain 0). After the host writes to these bits, it must not access the local memory during the machine state immediately following the write, because of the latency required to set the bits. There must be a single machine state between the write to CONFIG and an access to local memory. If the host is capable of requesting an access *immediately* following the write to CONFIG, you should ensure that the host makes a dummy request (such as reading CONFIG) before attempting to write to memory.

If the TMS34020's I/O registers are shadowed in external memory, the initial write to CONFIG to set BEN, RCM0—RCM1, and CBP (either by the TMS34020 or the host) may not be duplicated in external memory because the logical address bitmapping may be incorrect.

# 6.12.5 RESET and Multiprocessor Synchronization

You can use RESET to synchronize multiple TMS34020s that share a local memory. (Systems that use the TMS34020's multiprocessor interface to control memory access must synchronize the processors.) Synchronization is achieved by taking RESET high within a specific interval relative to CLKIN. TMS34020s to be synchronized should use the same CLKIN and RESET inputs; use CLKIN to clock the RESET. All of the local-memory and bus-control signals should be connected in parallel (without buffers) between the processors. After power-up, the processors may not all be executing the same machine state quarter cycle at the same time, because each machine state consists of 4 CLKIN cycles. When the low-to-high transition of RESET occurs, the TMS34020 could be in any one of the 4 quarter cycles. The TMS34020 stretches the first quarter phase (signified by LCLK1 high and LCLK2 low), until the eleventh CLKIN cycle after the rising edge of RESET. This ensures that all TMS34020s in the system are exactly synchronized. Section 11.3.2, page 11-3, discusses this in more detail.

The setup and hold times of RESET relative to CLKIN's low-to-high transition (specified in the *TMS34020 Data Sheet*) must be met only to guarantee that a RESET transition is detected at a particular clock edge. In a system with a single TMS34020, the RESET input signal can be asynchronous.

#### 6.12.6 State of VCLK During Reset

In many systems, the VCLK pin continues to be clocked during reset. However, a system in which VCLK is not clocked during reset should maintain VCLK at the logic-high level. This is necessary to ensure that the video counters are reset properly. (In fact, VCLK should be held at the logic-high level when it is not being clocked, regardless of whether the device is being reset.) While VCLK is low, storage nodes in the VCOUNT and HCOUNT registers rely on their internal capacitance to maintain their state. If VCLK remains low for a sufficiently long period, charge leakage may cause bit errors in these registers.

# 6.13 An Application for Interrupts: Debugging Code

The single-step interrupt causes the TMS34020 to interrupt execution after each instruction; this can be especially useful if you're developing and debugging new programs. This section describes this application.

## 6.13.1 How a Debugger Works

A debugger typically runs as code on the TMS34020 and code on a host system. If you decide to inspect a TMS34020 register or memory location, one of two situations may arise:

- If you're inspecting an I/O register or a local-memory location, the hostsystem portion of the debugger can perform a host read to directly access the information.
- □ If you're inspecting an internal register (general-purpose, ST, or PC), the host-system portion of the debugger must send a request to the TMS34020 portion of the debugger. The TMS34020 can put the required information into an agreed place in memory where the host software can access it through the host interface.

## 6.13.2 Using a Debugger

A debugger is usually used for loading and running development software under controlled conditions. Such conditions allow you to stop the software at any point and examine or change the state of the TMS34020 before continuing program execution.

If you want to examine the state of the TMS34020 after each instruction, you can do one of two things:

Insert breakpoints into your code; that is, insert instructions that force the TMS34020 to take a software interrupt. Any of these instructions can act as a breakpoint:

TRAP EMU TRAPL any illegal opcode

Inserting breakpoints requires the debugger to replace opcodes in the development software. This can be difficult because the debugger must be able to spot potential branches within the code and place breakpoints at all the addresses that could be branched to.

Use single-step mode. Single-step mode requires no changes to the development software and is much simpler to manage.

## 6.13.3 Entering Single-Step Mode

In single-step mode, the TMS34020 executes one instruction at a time. This happens regardless of how many machine cycles the instruction consumes and regardless of the amount of immediate data required. After executing an instruction, the TMS34020 saves the PC and ST values on the stack and

passes program control to the single-step routine (the routine's address is at vector address FFFF FBE0h). The single-step routine usually clears the single-step bit is cleared so the routine itself is not single-stepped. The routine then returns control to the debugger so that you to examine the state of the TMS34020.

When you are ready to continue program execution, the TMS34020 portion of the debugger should execute a RETM instruction, marking the end of the single-step routine in much the same way that RETI marks the end of other interrupt code.

RETM tells the TMS34020 to pop the PC and ST values and resume execution of the development code. By popping the ST, the single-step bit is restored to 1; development code resumes execution in single-step mode. After executing the next instruction, the TMS34020 again stacks the PC and ST and passes control to the single-step routine.

## 6.13.4 Clearing the Single-Step Bit

Note that a single-step routine may alter the ST value on the stack. Clearing the single-step bit in the stacked ST value switches off single-step mode. The RETM that terminates the routine pops the new ST value and returns to the interrupted code. The code now executes normally because the SS bit is cleared.

# 6.13.5 A Few Things to Keep in Mind

# Instructions that change the ST

Instructions that directly change the contents of the status register could cause problems if the development code does not deal correctly with the changes to the ST. For example, assume the following code is being single-stepped:

```
clr a0
putst a0
movi 012345678h,a9
```

After executing CLR A0, the TMS34020 takes the single-step trap. When the next RETM executes, control passes back to the development code. When the TMS34020 executes PUTST A0, it puts the value 0h (from A0) into the ST, thus clearing the single-step bit. From this point, single-step mode is no longer enabled; the debugger loses control of the development code.

To avoid these problems, the debugger should check the next instruction that the stacked PC points to, before executing it, to see if executing the instruction would affect the SS bit.

Only the PUTST and POPST instructions can directly alter the SS bit. If the debugger finds that the next instruction is PUTST, the debugger can alter the

register value that will be put into the ST so that SS is set to 1. Similarly, if the next instruction is a POPST, the debugger can alter the stack value that will be put into the ST so that SS is set to 1.

#### Interrupts during execution of development code

When the TMS34020 takes an interrupt, it stacks the PC and ST<sup>1</sup>, then clears the ST to 00000010h. The TMS34020 then reads the appropriate trap vector and branches to an interrupt routine.

If single-step mode is enabled when an interrupt occurs, the SS bit is cleared (because the interrupt sets the ST to 0000010h); thus, the interrupt code will not execute in single-step mode. Interrupts that may cause this include

reset	nonmaskable interrupt
host interrupt	display interrupt
window-violation interrupt	external interrupts 1 & 2

Executing one of the following instructions may also cause this:

TRAP	TRAPL
TRAP 0	TRAPL 0
EMU (with a single-step code)	any illegal opcode

As an example, assume that the TMS34020 is single-stepping through this development code:

clr a0 xor a7 movi 012345678h,a9

Assume that an interrupt occurs while the TMS34020 is executing the XOR instruction. The following events occur while the TMS34020 is executing this development code:

- 1) The CLR instruction executes.
- 2) A single-step trap returns control to the debugger (single-step routine).
- The RETM instruction at the end of the single-step routine returns control to the development software.
- 4) The XOR executes.
- 5) The interrupt occurs. The TMS34020 stacks the current PC and ST values, sets the ST to 0000010h, and passes control to the interrupt routine. The interrupt routine executes normally (it does not execute in single-step mode).

<sup>&</sup>lt;sup>1</sup> The following interrupts do not stack the PC or ST: reset, NMI (when NMIM = 1), TRAP 0, and TRAPL 0.

- The RETI that terminates the interrupt routine restores the old PC and ST values (with SS set).
- 7) Control immediately passes to the single-step routine.
- 8) The RETM instruction that terminates the single-step routine returns control to the development code.
- 9) The MOVI executes.
- 10) The development code continues execution, taking the single-step routine after each instruction.
- 11) An RETM instruction terminates the single-step routine.

Here's another look at this sequence:

Cod	e that's	single-stepped	Cod	e that's not single-stepped
1)	clr	a0	2)	ovecuto single stop reutine
			2) 3)	RETM
4)	xor	a7	5)	execute interrupt routine
			6)	RETI
			7) 8)	RETM
9)	movi	012345678h,a9	10)	execute sinale-step routine
			11)	RETM

Notice that the flow of development code and single-step routines is unbroken by the interrupt. The non-single-step interrupt is essentially transparent to the development code.

## Single-stepping interrupts -

If you want to single-step through an interrupt, then you should place a breakpoint at the start of all relevant interrupt routines. This breakpoint should be

- an EMU instruction (with the associated single-step code),
- a trap instruction, or
- any illegal opcode.

When the development code single-steps to the interrupt point, it takes the interrupt. The first instruction in the interrupt routine causes yet another interrupt, this time set up by the debugger to take the actions necessary to ensure that SS is set when control returns to the initial interrupt routine. As an example, this sequence of events might be

- 1) Single-step through the development code .
- 2) Encounter interrupt#1 (this is a non-single-step interrupt such as the display interrupt). The TMS34020 then automatically
  - a) Stacks PC#1 and ST#1.
  - b) Sets ST to 00000010h (the SS bit is now cleared).
  - c) Branches to the first instruction of the interrupt routine (IR#1).
- 3) The debugger should have changed IR#1 so that its first instruction causes interrupt#2 (that is, IR#1's first instruction could be an illegal opcode). Now, the TMS34020 automatically
  - a) Stacks PC#2 and ST#2.
  - b) Sets ST to 00000010h (the SS bit is still clear).
  - c) Branches to the first instruction of the illegal-opcode trap routine (IR#2).
- 4) IR#2 is installed by the debugger and could change the stacked value ST#2, setting ST#2's SS bit to 1. Finally, IR#2 should execute a RETM, after which the TMS34020 automatically pops the new ST#2 and PC#2 values before returning to IR#1. In IR#1, the TMS34020 passes control to the single-step trap routine before executing the instruction following the breakpoint.
- 5) The RETM that terminates the single-step causes control to pass back to IR#1, where the instruction following the breakpoint can execute. This and subsequent instructions in IR#1 execute in single-step mode.
- 6) Finally, at the end of IR#1, the terminating RETI executes, causing the TMS34020 to
  - a) Pop ST#1 (which already has the SS bit set) and PC#1, and
  - b) Resume single-stepping through the development code.
- Differences between RETI and RETM. The main difference between these two instructions is that RETM allows the TMS34020 to execute one instruction before returning to single-step instruction.

# **Communicating with a Host Processor**

A host processor can communicate with a TMS34020 through the TMS34020's **host interface**. The host interface allows you to map the TMS34020's local memory into a host's memory address space so that you can transfer data, commands, and status information between the host processor and the TMS34020's local memory.

This chapter describes host-interface operation.

	Sect	ion	Page
Basic information includes a	7.1	Related Signals	7-2
review of TMS34020 signals and	7.2	Related Registers	7-3
registers that affect the host interface: a system block	7.3	A Basic Block Diagram	
diagram; and general information		for the Host Interface	7-6
about how a host processor	7.4	Basic Communication: How a Host Processor	
communicates with the TMS34020.		Reads from and Writes to Local Memory	/-/
Advanced information describes	7.5	Features That Improve Performance of	
specific TMS34020⇔host		the Host Interface	7-10
communication features. For	7.6	Completing Host Accesses	7-16
of features that improve the	7.7	Timing Examples	7-18
efficiency of block accesses.	7.8	Halting TMS34020 Execution and	
-		Downloading New Code	7-32
	7.9	Host Interface Data Throughput (Bandwidth) .	7-34
	7.10	Delays to Host Accesses	7-37
System-specific information for	7.11	Systems with Multiple TMS34020s	7-40
systems using 16-bit memory,	7.12	Systems with 16-Bit Memory Devices	7-42
big-endian addressing.	7.13	Systems with Big-Endian Addressing	7-44

# 7.1 Related Signals

Many of the TMS34020's pins are devoted to host-interface functions; the TMS34020 provides a 27-bit address bus, 4 byte-select signals, and 7 control lines. Chapter 2 describes the host-interface signals in detail; the signals are summarized below for your convenience.

Signals	Descriptions	1/0
HA5—HA31	is the 27-bit host address input bus. The host uses this bus to identify the address of a word in local memory that it requires access to.	I
HBS0—HBS3	are the 4 host byte-select inputs. The host uses the HBS signals in conjunction with HA5—HA31 to access specific bytes in a selected word.	l
HCS	is the host chip-select signal. Driving this signal active low allows the host processor to access the TMS34020's local memory or I/O registers.	
HDST	is the host data-strobe signal, driven low by the TMS34020 during host read accesses. Data is strobed into the transceivers on HDST's rising (trailing) edge.	0
HINT	is the interrupt signal that allows the TMS34020 to send interrupt requests to the host processor.	0
HOE	is the host output-enable signal. The TMS34020 drives this signal active low to allow the transceivers to output data onto the TMS34020's local address/data bus (LAD0—LAD31) during host writes.	0
HRDY	is the host ready signal that informs the host processor when the TMS34020 is ready to complete a host-ini- tiated access cycle.	0
HREAD	is the host read signal. Driving this signal active low allows the host processor to read the contents of a selected location in the TMS34020's local memory or I/O registers. The TMS34020 latches the requested data into the transceivers.	1
HWRITE	is the host write signal. Driving this signal active low allows the host processor to write the contents of the transceivers to the selected location in the TMS34020's local memory or I/O registers.	1

# 7.2 Related Registers

Several of the TMS34020's I/O registers provide you with control over various aspects of the host interface. (Chapter 4 provides detailed descriptions of all the I/O registers.)

#### Note:

The *TMS34010* interface accessed certain I/O registers over its HAD bus. The TMS34020 does not access I/O registers this way. You must access an I/O register by its defined memory address within the TMS34020's local-memory space. For details, refer to Section 7.4 on page 7-7.

Five I/O registers are associated with the host interface:

- **HSTCTLL** and **HSTCTLH** control host-interface functions.
- □ HSTADRH, HSTADRL, and HSTDATA provide compatibility with the TMS34010.

The remainder of this section describes these registers, identifying the bits that are associated with host-interface functions. In the pictures of the registers, shaded areas identify bits that have no host-interface functions.

HSTCI	HSTCTLH register address: C000 0100h								
15	14	12	11	9	8	7	6, 5	4	0
HLT	CF	HINC	HPFW		NMI	RST	HLB	HACK	
Note: For a complete illustration and description of HSTCTLH, see Chapter 4.									

HACK	
bit 4	
HLB	
bit 5 &	6
BST	

When the TMS34020 is halted, it sets the HACK (halt acknowledge) bit.

The HLB (host last byte) bits form a 2-bit code. This code informs the TMS34020 which byte of a word the host accesses last.



bit 8

Setting the RST (reset) bit causes the TMS34020 to execute a reset.

The host processor can set the NMI (nonmaskable interrupt) bit to interrupt TMS34020 execution, causing the TMS34020 to execute an NMI routine.

When the NMIM (NMI mode) bit equals 0 and an NMI occurs, the TMS34020 bit 9 saves the context of the interrupted program. If NMIM=1, the TMS34020 does not save the context when an NMI occurs.

HPFW bit 10 When HINC=1, the HPFW (host prefetch after write) bit controls whether the TMS34020 prefetches (prereads) the next word in memory after a host read or write.

HINC	Setting the HINC (host increment) bit enhances the host's ability to access
bit 12	blocks of TMS34020 memory; when HINC=1, the TMS34020 increments a
	host-supplied address and prefetches the contents from this new location.

CF Setting the CF (cache flush) bit flushes the contents of the TMS34020 instrucbit 14 tion cache. The host processor can force the TMS34020 to execute new, downloaded code by flushing old instructions out of the cache.



The host processor can halt the TMS34020's CPU by setting the HLT (halt TMS34020 program execution) bit.

HSTCTLL register address: C000 00F0									00 00F0h
15	14	13	12	11	10	7	64	3	2—0
HBREN	HBFI	HRYI	EMIEN	EMG	EMR		MSGOUT	INTIN	MSGIN
Note: For a complete illustration and description of HSTCTLL, see Chapter 4.									

The MSGIN (message in) bits buffer a 3-bit interrupt message from the host bits 0–2 processor to the TMS34020.

The host processor can set the INTIN (input interrupt) bit to 1 to generate an bit 3 interrupt request to the TMS34020.

MSCOUT The MSGOUT (message out) bits buffer a 3-bit interrupt message from the bits 4–6 TMS34020 to the host.



The TMS34020 can set the INTOUT (output interrupt) bit to send an interrupt request to the host processor.



bits 10 & 11

An in-circuit emulator can use the EMR (emulator request) and EMG (emulator grant) bits as a mechanism for changing information and coordinating activity with the host processor. The information that these bits provide depends on the application, the host processor, and the software executed by the emulator. The emulator can set EMR to signal the host that an activity is beginning. The host can then set the EMG bit to acknowledge this. When the activity ends, the emulator signals the host by clearing EMR; the host then clears EMG. Only the emulator should modify EMR, and only the host should modify EMG. If you are not using this protocol with your emulator, these bits should always = 0.

EMIEN

If you are using an in-circuit emulator, setting the EMIEN (emulator host interrupt enable) causes the exclusive-OR of EMR and EMG to interrupt the host by asserting a low level on the HINT pin.

#### Note:

If you are not using an in-circuit emulator, clear the EMG, EMR, and EMIEN bits to 0.

The TMS34020 sets the HRYI (host retry interrupt) bit if a retry occurs on a host bit 13 access. If HBREN=1, setting HRYI sends an interrupt request (via HINT) to the host processor.

HBFIThe TMS34020 sets the HBFI (host bus-fault interrupt) bit if a bus fault occursbit 14on a host access. If HBREN=1, setting HBFI sends an interrupt request (via<br/>HINT) to the host processor.

When the HBREN (host bus-fault/retry interrupt enable) bit is set, setting HRYI bit 15 or HBFI causes an interrupt request to be sent to the host processor.



The TMS34010's host interface used the HSTADRH, HSTADRL, and HSTDA-TA registers for transferring address and data information. **The TMS34020's host interface does not use these registers**. The TMS34020 maintains these registers in the I/O register map to provide compatibility with TMS34010 software that uses these registers for passing information to the host interface. Because the TMS34020's host interface does not use HSTADRH, HSTADRL, or HSTDATA, any additional characteristics (such as autoincrementing) present for the TMS34010 are not implemented for the TMS34020. For more information about the TMS34020's treatment of these registers, refer to Section 1.6, Compatibility Between the TMS34020 and the TMS34010, on page 1-16.

## 7.3 A Basic Block Diagram for the Host Interface

In order for the TMS34020 to share data with a host processor, you must place bidirectional latching transceivers (such as the 74ALS652) between the TMS34020's local address/data bus and the host's data bus.

Figure 7–1. Block Diagram with a Host System, a TMS34020, and External Transceivers



#### Note:

Throughout this chapter, all references to *transceivers* refer to the external, bidirectional, latching, bus transceivers shown in Figure 7–1.

# 7.4 Basic Communication: How a Host Processor Reads from and Writes to TMS34020 Local Memory

The host processor initiates reads and writes through the TMS34020's host interface. The host controls the read and write cycles through a 2-step process:

## Step 1: Provide address and byte information.

- □ Identify a long (32-bit) word address on the HA5—HA31 bus.
- Activate the appropriate HBS signal(s) to identify a specific byte (or bytes) within the selected word. A host processor accesses data in groups of 1 to 4 bytes. The host byte-select signals tell the TMS34020 two things:
  - How many bytes to access in the selected word and
  - Which bytes to access.

Figure 7–2 (page 7-8) shows several examples of how HBS0– HBS3 determine which bytes of a long-word are accessed.

## Step 2: Set the HCS, HREAD, and HWRITE control signals.

- □ If the host wants to **read from** an address, it asserts both HCS and HREAD active low.
- ☐ If the host wants to write to an address, it asserts both HCS and HWRITE active low.

The host can assert or deassert the  $\overline{\text{HCS}}$ ,  $\overline{\text{HREAD}}$ , and  $\overline{\text{HWRITE}}$  control signals in any order. The last control signal that becomes active begins the access; the first control signal that becomes inactive ends the access. *The control signal that begins or ends an access is referred to as the strobe for the access.* 

The address and byte selection information becomes valid on the falling edge of  $\overline{\text{HCS}}$ .

## Note:

HREAD and HWRITE should not be active low simultaneously if HCS is also active low or if all 4 HBS signals are active. Having both signals low under these conditions could cause unpredictable host-interface behavior.

## Figure 7–2. How a Host Processor Uses the Host Byte-Select Signals to Access Data in TMS34020 Memory



Note: All other combinations of HBS0—HBS3 are valid as well.

## 7.4.1 How a Host Processor Requests a Read Cycle

When a host processor wants to read data from a TMS34020 memory location, the host must

- 1) Provide address and byte-select information over HA5—HA31 and HBS0—HBS3, then
- 2) Assert HCS and HREAD (in either order).

The host must keep  $\overline{\text{HREAD}}$  and  $\overline{\text{HCS}}$  active until the HRDY signal becomes active; when HRDY becomes active, the host can terminate the access and read the data from the transceivers.

The host must provide the transceivers with an output-enable signal that tells the transceivers to transfer their information to the host's data bus. No matter what size bus the host processor has, the transceivers must be capable of handling 32-bit information (for example, you could use four 8-bit transceivers). Regardless of how many bytes the host requests, the TMS34020 latches the entire 32 bits of data into the transceivers; the host's controls must enable the appropriate transceivers to select the desired bytes.

## 7.4.2 How a Host Processor Requests a Write Cycle

When a host processor wants to write data to a TMS34020 memory location, the host must

- 1) Provide address and byte-select information over HA5—HA31 and HBS0—HBS3, **then**
- 2) Assert HCS and HWRITE (in either order).

The host must keep HWRITE and HCS active until the HRDY signal becomes active. When HRDY becomes active, the host can terminate the request and latch the data-to-be-written into the transceivers.

The TMS34020 uses a byte-write feature that ensures that the host modifies only the selected bytes. The host must correctly align the data in the transceivers. The TMS34020 ignores the transceiver contents of unselected bytes.

# 7.4.3 Local-Memory Faults and Retries

It is possible, in some applications, that the host may request access to locations in the TMS34020's local memory, which may cause a retry or bus fault. During the local-memory access initiated in response to a host request, a retry or bus-fault completion code could occur on the LRDY or BUSFLT pins, respectively.

- □ If a host access causes the local memory to generate a retry, the TMS34020 automatically reschedules the access. The HRDY signal is not asserted until the TMS34020 successfully completes the access. In addition, the TMS34020 sets HRYI [[HSTCTLH]] to indicate to the host that the retry occurred. No further host accesses can be made until the retried access completes successfully.
- □ If a host access causes the local memory to generate a bus fault, the TMS34020 sets HBFI[[HSTCTLH]] to indicate to the host that the bus fault occurred; the TMS34020 takes no further action. HRDY is asserted as if the access terminated normally. It is then the host's responsibility to take the appropriate action (if any) to clear the cause of the bus fault. (The appropriate action depends on your application.)

Once HRYI or HBFI is set, it remains set until the host or the TMS34020 explicitly clears it.

Setting HBREN [[HSTCTLH]] to 1 causes the TMS34020 to assert the  $\overline{HINT}$  pin active low if either HRYI or HBFI is set. This allows a retry or bus fault on a host access to directly interrupt the host.

# 7.5 Features That Improve Performance of the Host Interface

This section describes several features that increase the performance and efficiency of the host interface.

## 7.5.1 Prefetching Data from the TMS34020's Local Memory

Prefetching (or prereading) information is an optional feature that can speed up the transfer of information. Instead of waiting for the host's next request, the TMS34020 fetches data from the next consecutive long-word address. The new data is fetched as soon as the current access completes and is placed in the transceivers in anticipation of the host's next request. Prefetching is beneficial for both reads and read(modify)writes:

- After **reads**, prefetching enables efficient reads of contiguous memory locations.
- After writes, prefetching enables efficient read(modify)writes of contiguous memory locations.

You can enable the prefetching feature by setting HINC [[HSTCTLH]] to 1. Once this is done, you must choose to prefetch information after reads or after writes. You control this by setting or clearing HPFW [[HSTCTLH]]. Figure 7–3 shows how different combinations of HINC and HPFW values affect prefetching.

Figure 7–3. How the Values of HINC [[HSTCTLH]] and HPFW [[HSTCTLH]] Affect Prefetching



The TMS34020 has an address-comparison feature that ensures the host interface always accesses the correct location. Figure 7–4 shows how you can enable the address-comparison feature.





- Prefetching enabled. On each read access, the host interface compares the address provided by the host with the address of the prefetched data. If the two addresses differ, the host interface automatically reads the new location requested by the host. This ensures that the host always reads from the address it requests, even when the TMS34020 prefetches data from a different location. This allows the host to access one contiguous block of words and then access another contiguous block, discontiguous from the first block, without disabling prefetching.
- Prefetching disabled. The TMS34020 performs no address comparison. Each time the host makes a read request, the TMS34020 explicitly copies the data from the requested location into the transceivers.

The size of a host's data bus can affect prefetching. If the host does not have a 32-bit data bus, then the host needs more than one access to fully read or write a 32-bit word. In this case, the TMS34020 must know which byte of the word the host will access last. This information is conveyed by the value of the HLB[[HSTCTLH]] bits. Figure 7–5 shows how this value identifies which byte the host will access last.

Figure 7–5. How the Value of HLE [[ HSTCTLH]] Affects Prefetching



The HLB bits allow the host interface to accommodate all possible byte-ordering conventions and several different data-bus sizes.

- A host processor with an 8-bit data bus can select any of the 4 bytes as the last byte. For example, assume that such a host processor is reading a contiguous block of TMS34020 local memory; assume also that the host sets HLB to 00₂ (designating byte 3 as the last byte that is prefetched). The host interface does not prefetch the next word until after the host reads byte 3 of the current word. All 32 bits of data are placed into the transceivers when the first byte is read; no memory accesses are required to read the subsequent 3 bytes because the transceivers already have the data.
- □ For a 16-bit host bus, only the value of HLB1 (bit 6) is needed to determine which 16-bit word the host accesses last.
- □ A 32-bit host need not program the HLB bits.

Although a non-32-bit host must make multiple read requests, the TMS34020 reads the location only once—on the first access (or after the designated last byte of the previous access). From this single read, the TMS34020 stores all 32 bits of data from the requested long-word address into the transceivers. Subsequent host accesses to read the remaining bytes within the word do not generate local-memory accesses because the requested data is already in the transceivers. Only after the designated last byte is accessed is a local-memory access initiated; this copies the data from the *next* 32-bit location, preparing for the next host access. The TMS34020's single-read feature minimizes the extra time required by a non-32-bit host to read an entire long-word.

#### Note:

The TMS34020 ensures that the host always reads the current contents of a memory location. Each read that follows a host-initiated write requests a localmemory cycle to explicitly copy the location's contents into the transceivers. This happens even if a previous read copied this address into the transceivers. Thus, while using prefetch-after-write (HINC=1, HPFW=1) with a non-32-bit host, each write causes a prefetch, regardless of whether or not the write was to a designated last byte. If the write was not to the designated last byte, the same location is prefetched again. Only after the last byte is written is the address incremented and the next location prefetched.

If prefetches and address-comparison are disabled, each host read request initiates a local-memory read cycle. This cycle copies the contents of the requested address into the transceivers, even if the data from that location is already stored in the transceivers.

## 7.5.2 Autoincrementing (Implicit Addressing)

The TMS34020's autoincrementing feature allows a host processor to address a contiguous block of local memory by specifying the address of only the first word in a block. The TMS34020 automatically generates all subsequent addresses and byte selects. A host can access contiguous blocks of words 16 bits or 32 bits at a time.

To use autoincrementing, the host asserts  $\overline{\text{HCS}}$  active low at the beginning of the first access, providing the address and byte selects just as it does for a regular access. The host must maintain  $\overline{\text{HCS}}$  active low throughout; the TMS34020 enables autoincrementing when it detects that  $\overline{\text{HCS}}$  remains low between the end of the previous access and the beginning of the current access. The host then uses  $\overline{\text{HREAD}}$  or  $\overline{\text{HWRITE}}$  as a strobe to request access to subsequent addresses.

When autoincrementing is detected, the host interface uses the value of HBS0—HBS3 to determine whether the host is accessing the block 16 bits or 32 bits at time. If all 4 byte selects are active, the TMS34020 assumes that the host needs only 1 access to completely read or write a 32-bit location. If only

2 byte selects are active (HBS0 and HBS1, or HBS2 and HBS3), the TMS34020 assumes that the host requires 2 accesses to completely read or write a 32-bit location. Figure 7–6 shows valid byte-select combinations.

Legal Host Byte-Select Combinations for Autoincrementing Figure 7–6.



Note: All other combinations of HBS0-HBS3 are invalid for autoincrementing

The TMS34020 latches the value of HBS0—HBS3 at the falling edge of HCS. If the next access will be made to the subsequent 16-bit or 32-bit location, the TMS34020 automatically generates the byte selects for the access by swapping its internal copy of the byte selects after each access. Thus, if the latched value of HBS0-HBS3 is 00112, then the internal copy is swapped to 11002 at the beginning of the next access. This in turn is swapped back to 00112 at the beginning of the following access, and so on. Similarly, if the value of HBS0—HBS3 latched at the falling edge of  $\overline{HCS}$  is 1100<sub>2</sub>, it is swapped to 0011<sub>2</sub> at the beginning of the next access. In this way, the internal copy of the byte selects alternates between selecting one 16-bit word and the other. Obviously, if HBS0—HBS3 is 1111<sub>2</sub>, swapping produces the same value.

Just as for regular accesses, the TMS34020 uses the HLB code to determine when the host has completely accessed a 32-bit word. It does this by comparing the internal copy of the byte selects with the HLB bits. The TMS34020 increments the address only after the host accesses the designated last 16-bit word.

Figure 7–7 shows how HINC [HSTCTLH] and HPFW [HSTCTLH] control how the address is incremented and when the byte selects are swapped.





4

The following list describes the byte-swapping operation for different cases.

#### Autoincrementing for reads *and* writes.

HINC=1 HPFW=0 Host can read & write contiguous locations

The TMS34020 swaps the internal copy of the byte selects after each access. After each access to the designated last 16-bit word, the TMS34020 swaps the the byte selects, then increments the address. If the access is a read, the next 32-bit location is prefetched into the transceivers. In this mode of operation, all autoincrementing accesses to a contiguous block of addresses should be of the same type—either reads or writes, but not a mixture of the two.

#### Autoincrementing after writes *only*.

HINC=1 HPFW=1 Host can read-modify-write contiguous locations

The TMS34020 swaps the internal copy of the byte selects only after writes. After each write to the designated last 16-bit word, the address is incremented, and the next 32-bit location is prefetched into the transceivers. No modifications to the address or the byte selects are made after reads, so the host writes back to the same location. In this mode of operation, the host should perform read and write requests alternately; the first access can be either a read or a write. Note that in 16-bit autoincrement mode, a location is read only once. Because the sequence of reads and writes is predefined, there is no need to reread the 32-bit location after the first write.

#### No autoincrementing.

HINC=0 HPFW=don't care	Host can't autoincrement	
------------------------	--------------------------	--

It is not anticipated that this mode of operation will be very useful. However, for completeness, the response to autoincrement accesses while HINC=0 is described below.

- For writes, the TMS34020 accesses the same 32-bit address on each host request. If 16-bit autoincrementing is selected, the byte selects are still swapped, so successive accesses oscillate between the two 16-bit words within the specified 32-bit location.
- For reads, the TMS34020 copies all 32 bits of data from the specified location into the transceivers during the first access. On subsequent reads in autoincrement mode, no new data is transferred into the transceivers.

In this mode, all accesses should be of the same type—either reads or writes, but not a mixture of the two.

If the host wants to access an address that is not contiguous with the block, it must deassert  $\overline{\text{HCS}}$  and provide a new address just as it would for a regular access.

When autoincrementing, the actual local-memory cycles performed by the TMS34020 in response to host requests do not differ from those performed when not autoincrementing. Autoincrement accesses offer no throughput advantage compared to similar non-autoincrement accesses.

## 7.5.3 The TMS34020's Default Memory Cycle

When no other memory requests are pending, the TMS34020's memory controller executes a state similar to the address subcycle of a host-initiated local-memory cycle. This is known as the **host-default state**. The only difference between the host-default state and the address subcycle is that ALTCH and RAS do not go active. If a host request is synchronized while a host-default state is being executed, the host-default state is converted into a regular host access by internal control logic, which causes ALTCH and RAS to go active. This action reduces the cycle time required for host-initiated memory cycles by one LCLK cycle because it allows the first half of the host access to occur in parallel with synchronization of the request.

The address output on LAD0—LAD31 and RCA0—RCA12 during a hostdefault cycle is the contents of the internal register used for storing the address presented by the host. If the host-default cycle is converted into a real host-initiated memory cycle, then the synchronization delays inherent in the host-interface logic ensure that the address output is the correct one for the access. However, at other times (when the host-default memory cycle is not converted into a real host access) the address output on LAD0—LAD31 and RCA0—RCA12 is not valid. Because ALTCH and RAS do not go active in this case, this should not matter.

# 7.6 Completing Host Accesses

When the TMS34020 is ready to complete a host-initiated access, it asserts HRDY active high. *The host must not deassert its request strobe until HRDY becomes active.* By default, the TMS34020 maintains HRDY in an inactive-low state. After the host terminates a request, HRDY becomes inactive low again.

Various conditions govern when HRDY can become active-high. These are described in the following sections.

# 7.6.1 Activating HRDY for Host Reads

When prefetches are disabled and the host requests a read to any address, or when prefetches are enabled and the host requests a read to a location other than the one currently copied into the transceivers, the TMS34020 fetches the data and places it in the transceivers.

The leading edge of this type of host request triggers the TMS34020 to access the requested address. During the last machine state of the memory access, HRDY becomes active. This happens while the data is being transferred into the transceivers from the local memory and HDST is active low. The data is not guaranteed valid in the transceivers until HDST becomes high. The host can deassert the read-request strobe as soon as HRDY goes active, but must not read the data from the transceivers while HDST is low. HRDY is asserted before the data is valid so that it is possible for the host to make requests sufficiently close together that the maximum throughput of the TMS34020's local-memory interface can be used. This is discussed in more detail in Section 7.9.2.

## 7.6.2 Activating HRDY for Host Writes

When the host requests a write, the TMS34020 asserts HRDY when it is ready for the host to deassert its write-request strobe. This trailing edge of the host write-request triggers the local-memory cycle required to perform the write, which therefore occurs after the host terminates its request. The data for the write must be latched into the transceivers and valid before HOE becomes active low. This could be as few as 1.25 LCLK cycles after the host write-request strobe is deasserted.

If the host requests another write, the actions taken depend on whether any access previously initiated by the host is still pending:

- ➡ If no incomplete memory accesses are pending or in progress from the previous host write, HRDY immediately becomes active and the host can latch new data into the transceivers.
- If the previous host write is not finished, HRDY remains low until the last machine state of the memory access that writes the contents of the transceivers to the requested memory location. HRDY becomes active while

HOE is still active low, enabling the contents of the transceivers onto LAD0—LAD31. The host can deassert the write-request strobe as soon as HRDY goes active, but must not latch new data into the transceivers while HOE is low. HRDY is asserted while the previous data is still being written so that it is possible for the host to make requests sufficiently close together that the maximum throughput of the TMS34020's local-memory interface can be used. This is discussed in more detail in Section 7.9.2.

## 7.6.3 Activating HRDY for Host Reads and Writes after Prefetches

When prefetching is enabled (HINC=1), the trailing edge of the host's readrequest strobe (if HPFW=0) or write-request strobe (if HPFW=1) on an access to the designated last byte of the current location automatically initiates a prefetch of the next location. If the host makes a request while a local-memory cycle to prefetch data into the transceivers is in progress, HRDY cannot become active high until that memory cycle is almost complete. It is likely, if you are trying to use the maximum throughput of the TMS34020's local-memory interface, that the prefetch memory cycle will not have completed when the host makes its next read or write request.

Here are the rules that govern when HRDY becomes active.

- ❑ Writes. If a prefetch is pending or in progress when the host requests an access, HRDY remains inactive until the last machine state of the prefetch.
- Reads. The address to be read is compared with the address of the location that is currently copied into the transceivers. The action taken depends on whether the addresses match and whether the prefetch has completed:
  - Addresses match, prefetch complete. HRDY immediately becomes active. Because the requested data is already in the transceivers and HDST is already inactive-high, the data is valid immediately.
  - Addresses match, prefetch in progress. HRDY becomes active in the last machine state of the prefetch while HDST is still low. The data is valid in the transceivers as soon as HDST becomes high.
  - Addresses do not match, prefetch complete. The host initiates another read to the required location before HRDY becomes active.
  - Addresses do not match, prefetch in progress. The host interface waits for the prefetch access to complete. Then it initiates another read to the required location before HRDY becomes active.

In the second, third, and fourth cases described above, HRDY becomes active while HDST is low. HRDY is asserted before the data is valid. This allows the host to make requests quickly enough to use the TMS34020's maximum local-memory bandwidth. This is discussed in more detail in Section 7.9.2 (page 7-35).

## 7.7 Timing Examples

This section contains timing diagrams that illustrate the host interface's response to various host requests. Each host-access request to a new 32-bit location in the TMS34020's local memory or I/O registers generates a local-memory cycle. The prefetch mechanisms described in Section 7.5.1 (page 7-10) also generate local-memory accesses.

Each diagram shows the host interface signals at the top. The buses marked *DATA (in)* and *DATA(out)* indicate the validity of the data in the transceivers for host writes and reads, respectively.

Below the host signals, a scale marks each quarter phase of a TMS34020 machine state (Q1—Q4). Below this scale are the local-memory buses and control signals. This illustrates the relationship between the host requests and the resulting local-memory cycles.

All the diagrams assume that:

- The local-memory interface is always immediately available to the host.
- There are no wait states (unless otherwise noted).

With the exception of HRDY, no timing relationships are implied between the host-interface signals above the scale and the local-memory signals below it. However, in order to achieve optimum performance, the synchronization time between the host interface and the local-memory interface should be minimized. The local-memory access can be initiated in the next machine state only if the host request strobe that initiates the local-memory access occurs before the fourth quarter phase (Q4) of the current machine state. If the strobe occurs after the beginning of Q4 of the current machine state, the earliest that the access can begin is two machine states later. The timing diagrams illustrate this synchronization (Section 7.9, page 7-34, discusses this in more detail).

Some of the timing diagrams show  $\overline{\text{HCS}}$  being asserted (or deasserted) before  $\overline{\text{HREAD}}$  or  $\overline{\text{HWRITE}}$ ; some show  $\overline{\text{HREAD}}$  or  $\overline{\text{HWRITE}}$  being asserted (or deasserted) before  $\overline{\text{HCS}}$ . This emphasizes that any assertion order is allowed and implies no relationship between the assertion order required for the request and the nature of the request. The signal that begins (or ends) an access and that initializes a local-memory cycle for the host is referred to as the **host request strobe** for the access.



Figure 7–8. Single Host Read Cycle; HCS Used as Strobe

- Notes: 1) Prefetch after read is disabled (HPFW = 1 or HINC = 0).
  - This figure refers to host reads of TMS34020 local memory and does not include host reads of TMS34020 I/O registers.

Figure 7–8 shows a simple read cycle; the host is accessing a location in local memory. The leading edge (high-to-low transition) of the host request strobe initializes the local-memory cycle.



Figure 7–9. Single Host Read from I/O Registers; HREAD Used as Strobe

#### Note: Prefetch after read is disabled (HPFW = 1 or HINC = 0).

Figure 7–9 shows a simple read cycle; the host is accessing an I/O register. The leading edge of the strobe initializes the local-memory cycle. Comparing this figure to the previous figure, you'll see that to the host, there's little difference between accessing an I/O register and accessing the rest of the local memory. However, some of the local-memory strobes are different; CAS0—CAS3, DDIN, and TR/QE are not activated because the data is transferred from within the TMS34020, not from the local memory.



Figure 7–10. Single Host Read with One Wait State; HCS Used as Strobe

- Notes: 1) Prefetch after read is disabled (HPFW = 1 or HINC = 0).
  - 2) This figure refers to host reads of TMS34020 local memory and does not include host reads of TMS34020 I/O registers.

Figure 7–10 is identical to Figure 7–8 except that the memory cycle includes one wait state. Both figures show a simple read cycle in which a host processor is accessing a location in the TMS34020's local memory. The leading edge of the host request strobe initializes the local-memory cycle.



Figure 7–11. Host Read Back-to-Back with Prefetch of Next Word; HCS Used as Strobe

Notes: 1) Prefetch after read is enabled (HINC = 1 and HPFW = 0).

2) This figure refers to host reads of TMS34020 local memory and does not include host reads of TMS34020 I/O registers.

Figure 7–11 shows 2 back-to-back read cycles; the second location is prefetched after the first location is read. The trailing edge (low-to-high transition) of the host request strobe initiates the prefetch local-memory cycle.



Figure 7–12. Back-to-Back Host Read Cycles with Implicit Addressing; HREAD as Strobe

Notes: 1) Prefetch after read is enabled (HINC = 1 and HPFW = 0).

2) This figure refers to host reads of local memory and does not include host reads of I/O registers.

Figure 7–12 shows 2 back-to-back read cycles; the second location is prefetched after the first location is read. The trailing edge of the host request strobe initiates the prefetch. Just as in Figure 7–11, the prefetch relaxes the timing of the request strobe. Figure 7–12 also illustrates implicit addressing. Although not explicitly shown, the value on HBS0—HBS3 must be 1111<sub>2</sub>, 0011<sub>2</sub>, or 1100<sub>2</sub>. The initial address is the one supplied at the falling edge of HCS.


Figure 7–13. Successive Reads to Same 32-Bit Location; HCS and HREAD Strobed Together

- Notes: 1) Prefetch after read or write is enabled (HINC = 1 and HPFW = don't care).
  - This figure refers to host reads of TMS34020 local memory and does not include host reads of TMS34020 I/O registers.

Figure 7–13 shows successive reads to the different bytes at the same location. Notice how repeatedly accessing the same word carries little overhead because the TMS34020 accesses the data only once. The HINC bit must equal 1 (enabling prefetching) to support the necessary address comparison.





- Notes: 1) Prefetch after write is disabled (HPFW = 1 or HINC = 0).
  - This figure refers to host writes to TMS34020 local memory and does not include host writes to TMS34020 I/O registers.

Figure 7–14 shows a simple write cycle; the host is writing to a location in local memory. The trailing edge of the host request strobe initializes the local-memory cycle.



Figure 7–15. Single Host Write Cycle to I/O Registers; HWRITE Used as Strobe

Note: Prefetch after write is disabled (HPFW = 0 or HINC = 0).

Figure 7–15 shows a simple write cycle; the host is writing to an I/O register. The trailing edge of the host request strobe initializes the local-memory cycle. Comparing Figure 7–15 to Figure 7–14, you'll see that to the host, there's little difference between writing to an I/O register appears and accessing any other memory location. However, data is latched into the TMS34020 on LAD31–LAD0.





- Notes: 1) Prefetch after write is disabled (HPFW = 0 or HINC = 0).
  - This figure refers to host writes to TMS34020 local memory and does not include host writes to TMS34020 I/O registers.

Figure 7–16 shows a simple write cycle; the host is writing to a location in local memory. The trailing edge of the host request strobe initializes the local-memory cycle. Figure 7–16 is identical to Figure 7–14 except that the memory cycle includes one wait state.



Figure 7–17. Back-to-Back Host Write Cycles; HCS Used as Strobe

Notes: 1) Prefetch after write is disabled (HPFW = 0 or HINC = 0).

2) This figure refers to host writes to local memory and does not include host writes to I/O registers.

Figure 7–17 shows 2 consecutive writes. The trailing edge of the host request strobe initializes the local-memory cycle. On the second write, notice that HRDY remains low until just before the data in the transceivers (from the first write) is written to memory.



Figure 7–18. Back-to-Back Host Write Cycles with Implicit Addressing; HWRITE as Strobe

Notes: 1) Prefetch after read is enabled (HINC = 1 and HPFW = 0).

2) This figure refers to host writes to local memory and does not include host writes to I/O registers.

Figure 7–18 shows 2 consecutive writes. The trailing edge of the host request strobe initializes the local-memory cycle. Note that on the second write, HRDY remains low until just before the data in the transceivers (from the first write) is written to memory. Figure 7–18 also illustrates implicit addressing. Although not explicitly shown, the value on HBS0—HBS3 must be  $1111_{2}$ ,  $0011_{2}$ , or  $1100_{2}$ . The initial address is the one supplied at the falling edge of HCS.



Figure 7–19. Host Write Cycle Back-to-Back with Prefetch of Next Word; HCS Used as Strobe

**Notes:** 1) Prefetch after write is enabled (HINC = 1 and HPFW = 1).

2) This figure refers to host writes to local memory and does not include host writes to I/O registers.

Figure 7–19 shows a write cycle followed by a prefetch of the next location. The TMS34020 automatically initiates the prefetch after it completes the write. Notice that HRDY remains low until just before the prefetched data is latched into the transceivers.



Figure 7–20. Host Write Cycle Back-to-Back with Prefetch of Next Word and Implicit Addressing; HREAD and HWRITE Used as Strobes

**Notes:** 1) Prefetch after write is enabled (HINC = 1 and HPFW = 1).

2) This figure refers to host writes to local memory and does not include host writes to I/O registers.

Figure 7–20 shows a write cycle followed by a prefetch of the next location. Notice that HRDY remains low until just before the prefetched data is latched into the transceivers. Figure 7–20 also illustrates implicit addressing. Although not explicitly shown, the value on HBS0—HBS3 must be 1111<sub>2</sub>, 0011<sub>2</sub>, or 1100<sub>2</sub>.

# 7.8 Halting TMS34020 Execution and Downloading New Code

A host processor can halt TMS34020 execution by setting HLT[[HSTCTLH]]. While halted, the TMS34020 cannot access its local memory. This removes the major source of delay to host-initiated memory cycles and reduces (by half) the time taken for the host to access local memory.

Halting the TMS34020 involves several bits in the HSTCTLH register. The picture below identifies the pertinent bits:

HSTC	TLH re	gister			address: C000 (	)100h	
15	14	12	11	9	4	0	
HLT	CF	HINC	HPFW				
Note:	Note: For a complete illustration and description of HSTCTLH, see Chapter 4.						

The TMS34020 may not immediately recognize the halt (reasons for this are described in Section 7.10.2, <u>Halt Latency</u>, on page 7-39). If the host is transferring large amounts of data, however, the increased throughput gained by halting the TMS34020 more than compensates for any delay in waiting for the TMS34020 to recognize the halt.

In order for a host processor to determine when the TMS34020 actually halts, it can

- Wait for a period equal to the TMS34020's maximum halt latency, or
- Repeatedly read the value of the HACK bit until HACK=1. When the TMS34020 halts, it sets HACK. When the TMS34020 is released from halt, it automatically clears HACK.

The TMS34020 recognizes a halt request on an instruction boundary only. Unlike an interrupt, a halt is not recognized during an interruptible instruction such as a PIXBLT. A PIXBLT of a very large area of memory could take a significant time to complete. If the host needs a prompter response to the halt request—that is, if it cannot wait for the current instruction to complete—the host should set the NMI bit when it sets HLT. The NMI (nonmaskable interrupt) is taken at the first interruptible point, which may be midway through an interruptible instruction. As soon as the NMI is taken, the halt is recognized (when both NMI and HLT are set, the TMS34020 halts before executing the first instruction in the NMI service routine).

One of the most useful applications for halting the TMS34020 is to provide a host processor with an efficient method for downloading new code to TMS34020 local memory. The host can download code by following these steps:

- Step 1: Set the NMI, HLT, and CF bits to 1. This interrupts and halts the TMS34020 and flushes the cache.
- **Step 2:** Download the code through the host interface. You can use the autoincrementing (implicit addressing) feature by setting HINC to 1, clearing HPFW to 0, and providing the starting address of the code's destination.
- Step 3: Write the starting address of the new code to the NMI vector so that the new code begins execution at the vectored address. (The NMI vector address is FFFF FEE0h.)
- **Step 4:** Set the NMI bit to 1 to initiate a nonmaskable interrupt (the bit was cleared automatically if the previous nonmaskable interrupt was taken before the halt). At the same time, set the NMIM bit to 1 if the host does not need the current context to be stored on the stack or if the nonmaskable interrupt was taken in the first step; otherwise, clear NMIM to 0.
- Step 5: Restart the TMS34020 by writing a 0 to HLT; at the same time, clear the CF bit.

# 7.9 Host-Interface Data Throughput (Bandwidth)

The host-interface **bandwidth** is the number of bits per second that can be transferred through the host interface during a block data transfer to or from TMS34020 local memory.

## 7.9.1 Achieving Maximum Bandwidth

The maximum host-interface bandwidth approaches the bandwidth of the TMS34020's local-memory interface. For a 40MHz TMS34020 with no memory wait states, and allowing for the fact that successive host-initiated local-memory cycles cannot be performed using page mode, the local-memory interface can perform one memory cycle every 200 nanoseconds. This is a bandwidth of 160 megabits per second.

Memory requests from other sources (such as screen refreshes, DRAM refreshes, or the TMS34020's CPU) tend to reduce the rate of data transfer. However, refreshes of either type occur infrequently (and therefore have little effect), and halting the TMS34020 prevents the CPU from making memory requests.

If the CPU is not halted, it could reduce the bandwidth through the host interface by up to 50%. If both host and CPU memory requests are pending, the host request is performed first, but the CPU request is performed before the next host request. Thus, if both the host and the CPU continue to make memory requests, host and CPU memory cycles occur alternately. Table 7–1 summarizes the different host interface bandwidths that can be expected.

## Table 7–1. Host Interface Estimated Maximum Bandwidth

Assumptions		Approximate Throughput		
	TMS34020 is halted			
Q	40 MHz TMS34020	32 bits per transfer	= 160 Mbits per second	
	No wait states			
	TMS34020 is running			
	40 MHz TMS34020	400 pa par transfer	= 80 Mbits per second	
	No wait states	400 fis per transier		
	TMS34020 is halted	00 kilo sostassofa		
	40 MHz TMS34020	32 bits per transfer	Mbits per second	
	N wait states	(200+/v) 100 ns per trai	ISIEI	

# 7.9.2 Timing Considerations for Optimizing Host-Interface Bandwidth

To use the local memory's full bandwidth, the host must be able to initiate a memory cycle every 200 nanoseconds. However, because there is an inherent synchronization and recognition delay between when the host request strobe occurs and when the local-memory cycle it initiates can begin, the request strobe that initiates the next memory cycle must occur before the current memory cycle has completed.

To enable the host to do this, HRDY is asserted active-high *early* during host-initiated memory cycles, shortly after LCLK2's low-to-high transition. The host request strobe that initiates the next memory cycle must occur before LCLK2's high-to-low transition in the current machine state, so that the memory cycle can begin in the next machine state. If the host request strobe occurs after this, it is not recognized until LCLK2's next high-to-low transition, so the memory cycle it initiates cannot begin until the machine state after that. This is illustrated in Figure 7–21.











Note: This illustration uses the hypothetical signal STROBE, which is the logical-OR of HCS with either HREAD or HWRITE; it is low when HCS and either HREAD or HWRITE are low. Host read cycles are initiated by STROBE's high-to-low transition (the leading edge of the host request). Write and prefetch cycles are initiated by STROBE's low-to-high transition (the trailing edge of the host request). The diagrams illustrate both possibilities. In both cases, the arrow (←──) identifies the earliest point at which the host-initiated local-memory cycle can begin. Obviously, this may be delayed if another local-memory cycle is being performed at this time.

For write requests or prefetch-initiating read requests, a low-to-high transition of the host request strobe is all that is necessary to initiate the next local-memory cycle. However, for host read requests when prefetching is disabled, the strobe's high-to-low transition enables the next local-memory cycle for the host. Because of this, it is not possible to achieve maximum bandwidth when performing multiple host read requests where prefetches are disabled. There is not enough time, after HRDY goes active shortly following LCLK2's low-to-high transition, to deassert the host request strobes and then reassert them before LCLK2's high-to-low transition. In most applications, this should not matter because consecutive host accesses are typically to contiguous blocks where prefetching can be used.

Although HRDY is asserted just after LCLK2's low-to-high transition, the host cannot access the transceivers until both HDST and HOE are high. One of these signals is always low when HRDY goes high during a host-initiated cycle. This means that the host must gate HRDY with HDST and HOE to determine when it can access the transceivers.

The manner in which HRDY is gated with HDST and  $\overline{HOE}$  depends primarily on the rate at which the host can consecutively access the TMS34020:

- Host can make requests fast enough to use the maximum available throughput. As described above, the host request strobe that initiates the next memory cycle must occur before the end of the current memory cycle, before the host can access the transceivers. This requires additional data buffering (or pipelining) into and out of the transceivers. HRDY will have returned to its inactive-low state before HDST or HOE goes high at the end of the current memory cycle. Because of this, the rising edge of HRDY must be used to set an edge-triggered latch, so that the output of the latch becomes a 1. This must be ANDed with HDST and HOE, so that when HDST is high and the HRDY latch is set, the transceivers can be read from; when HOE is high and the HRDY latch is set, the transceivers can be written to. A mechanism for clearing the latch must be provided, although this depends on the precise requirements of the application. However, the latch must be cleared before the next low-to-high transition of HRDY, which can occur while either HDST or HOE is low. One possibility would be to use an edge-cleared latch cleared on the falling edge or either HDST or HOE.
- Host cannot make requests fast enough to use the maximum available throughput. There is no need for the host request strobe that initiates the next local-memory access to occur until after the current host-initiated memory cycle completes. This can be achieved by ANDing HRDY with HDST and HOE, so that the host does not respond to HRDY until HRDY, HDST, and HOE are all high.

## 7.10 Delays to Host Accesses

The longest (worst-case) delay that can occur on a host access is the accumulation of a number of smaller delays from a variety of sources. Worst-case delay depends on the application; all of these delays do not exist for all systems. It is unlikely that all possible delay sources exist for a single application.

- □ In some applications, you must determine not only the effective bandwidth of the host interface but also the **data transfer delays** that can occur under worst-case conditions. These conditions rarely occur, but the fact that they can occur makes them worthy of consideration. These types of delays affect the time it takes for the host to access local memory; see Section 7.10.1.
- It is sometimes advantageous for a host processor to halt TMS34020 execution by setting the HLT bit. However, there can be a delay from the time that a host write request sets HLT until the time that the TMS34020 actually halts. This delay is referred to as halt latency; see Section 7.10.2.

#### 7.10.1 Worst-Case Delay

Worst-case delays are unusual; they occur too infrequently to significantly affect overall performance. The following paragraphs list all possible sources of delay. Some of these sources may produce different delays, depending on the application. Each delay is characterized by two parameters:

- *n* is the **number of wait states** per TMS34020 memory cycle.
- is the **local clock period** (nominally 100 nS for a 40-MHz TMS34020).

Table 7–2 summarizes the various delay sources.

Table	7–2.	Sources	of I	Delay
-------	------	---------	------	-------

	Delay (in local clock periods, t )			
Operation	Minimum	Maximum		
Request synchronization	0.25	2.25		
Screen refresh	0			
with midline reload		2(2 + n)		
without midline reload		2 + <i>n</i>		
DRAM refresh	0			
15 pending		4(3+n)		
12 or fewer pending		3 + <i>n</i>		
Bus-master arbitration	0	system dependent		
Previous host cycle	0			
with prefetch after write		2(2+n)		
without prefetch after write		2 + <i>n</i>		
CPU cycle	0	2 + <i>n</i>		

To calculate your application's worst case, determine which delays apply, and sum these individual delay times. This figure is the worst-case delay for a host access. It does not include the time required to actually perform the host access. The following paragraphs describe the delay sources.

- Delay 1: Host request synchronization. This is the time required to internally synchronize a host request to the TMS34020's local clock. Typically, host signals are not synchronous to the TMS34020's local clocks. Before using a signal from the host, the TMS34020 must pass it through a synchronizing latch (part of the TMS34020's on-chip host-interface logic). The delay through the synchronizer is 1.25 to 2.25 local clock cycles. However, the host-default cycle performed by the local-memory interface when no other requests are pending allows this synchronization to occur in parallel with the address subcycle. Thus, synchronization delay can be as small as 0.25 local clock cycles.
- Delay 2: Screen-refresh cycles. Any screen refresh has a higher priority than a host request. Thus, if a screen refresh and a host access are requested simultaneously, the TMS34020 performs the screen refresh. The longest delay that can be caused by screen refreshes occurs during horizontal blanking in split-serial register VRAM mode; this delay takes 4 machine states plus wait states. If this mode is not used, the longest delay is 2 machine states plus wait states (see Chapter 9, Video Timing and Screen Refresh).
- Delay 3: DRAM-refresh cycles. The TMS34020 can have up to 15 DRAM refreshes pending. If 12 or more DRAM refreshes are pending, they have higher priority than host requests. In this situation, if a DRAM refresh and a host access are requested simultaneously, the TMS34020 performs the DRAM refresh first. Thus, the TMS34020 may perform up to 4 DRAM refresh cycles before servicing a host request. In most systems, however, no more than 12 refreshes will ever be pending. The only way more than 12 DRAM refreshes could be pending is if a memory cycle is wait-stated for an extremely long time, or if the GI pin is deasserted for long enough. If 12 or fewer DRAM refreshes are pending, a host access can be delayed by one DRAM refresh only.
- ❑ Delay 4: Bus-master arbitration. This occurs only in systems where multiple TMS34020s and/or other devices share local memory and arbitrate for bus control through the multiprocessor interface (see Chapter 11). Typically, the host is connected to the only TMS34020 in the system that can make high-priority bus-master requests. Host requests generate the high-priority request code on R0 and R1. The time taken to reclaim bus ownership after a high-priority request code is output is entirely application dependent.
- Delay 5: Previous host cycle. The worst form of this delay occurs when the previous host request was a write with prefetch, and the host then

attempts to read from or write to a location other than the prefetched location. In this case, the TMS34020 performs the write and the prefetch before performing the current request. This is a delay of 4 machine states plus wait states. If prefetch-after-write is not enabled, the worst-case delay is 2 machine states plus wait states.

- Delay 6: CPU cycle. The TMS34020 CPU might be using the localmemory interface when the host requests a write. This causes a delay before the memory controller can perform the requested write. Note that a CPU cycle cannot delay a host access if
  - The TMS34020 encounters a bus-master arbitration delay.
  - The TMS34020 is halted.

In any system, worst-case delays occur too infrequently to affect overall performance. For example, DRAM-refresh cycles typically consume approximately 2% of the available memory bandwidth, and screen-refresh cycles (using VRAMs) consume approximately 1.5%. Do not use the worst-case delay to determine host-interface throughput.

The worst-case delay assumes that

- A screen-refresh cycle is generated within the TMS34020 on the same clock edge at which the host request arrives (after synchronization).
- □ 12 or more DRAM refreshes are requested during the next 1 + n clock edges. This is the number pending when the host and screen-refresh requests were made, plus additional requests (if any) made in the ensuing 1 + n machine states while the screen refresh is executed.
- ❑ An equivalent delay occurs when 12 or more DRAM-refresh requests and the host request occur on the same clock edge and a screen refresh is requested on a later clock edge before the host access begins. Host, DRAM-refresh, and screen-refresh requests all assert the high-priority request code through the multiprocessor interface (see Chapter 11).

## 7.10.2 Halt Latency

Section 7.8 (page 7-32) describes a method for halting the TMS34020 by setting the HLT bit. If you use this halt function, you may find it useful to know how long it can take the TMS34020 to enter the halt condition. This time, called **halt latency**, *always* includes

	1 or 2 machine states	caused by the host-request synchronization (see delay 1 on page 7-38)
+	2 machine states	for the host to actually write to the HLT bit
<u>+</u>	1 machine state	for the CPU to halt after recognizing the request
	4 or 5 machine states	is the minimum amount of time required to halt the TMS34020

In addition to the delays listed in Section 7.10.1, the TMS34020 may be executing an instruction when it receives the halt request. In this case, the TMS34020 does not halt until it completes the instruction execution. This could take several machine states (depending on the instruction), and applies to noninterruptible instructions as well. The TMS34020 recognizes a halt request only on an instruction boundary.

Section 7.10.1 shows the worst-case delay for accessing local memory (including I/O registers). Delays 2 through 6 can affect the time taken to halt the TMS34020, but their effect on halt latency is usually much less than the worst-case delay. In fact, they usually do not affect halt latency at all; even a single DRAM refresh is unlikely to delay halt recognition.

# 7.11 Systems with Multiple TMS34020s

Multiple TMS34020s running within a single system can share the same local memory. This can improve system performance significantly; however, it may complicate host-interface communications. For a general description of multiple TMS34020 systems, see Chapter 11, *Multiprocessing and System Architecture*.

If a multi-TMS34020 system contains ROM, the TMS34020s will probably be initialized from the ROM at power-up. However, if the system contains no ROM, the host processor must be able to download code to all of the TMS34020s. To do this, a host typically halts the TMS34020s; as an alternative, the host can prevent the processors from fetching code by setting their  $\overline{GI}$  pins inactive, which prevents them from using the local-memory bus.

If all the TMS34020s are halted at power-up, the host must be able to restart them. To restart a halted TMS34020, the host must write a 0 to that TMS34020's HLT bit. The host can access the I/O registers of only those TMS34020s to which it is directly attached; one TMS34020 cannot access another's I/O registers. This means that the only method for clearing the HLT bit of a TMS34020 to which the host is not connected is to assert its RESET pin. *Only those TMS34020s to which the host has direct access can be powered-up halted.* 

The host can access multiple TMS34020s by having a different  $\overline{\text{HCS}}$  signal for each one. Because host requests are considered high priority, your multiprocessor arbitration scheme must prioritize simultaneous high-priority bus requests from different TMS34020s. To accomplish this, you can use one of two possible transceiver configurations:

❑ All TMS34020s share one set of transceivers. Wire together all the HDST pins, and wire together all the HOE pins. All data is transferred through the one set of transceivers to whichever TMS34020 is being accessed. This solution requires less hardware to implement. However, be very cautious when using prefetch modes; the location prefetched by one TMS34020 will no longer be stored in the transceivers if another

TMS34020 performed a host read. This could result in the host reading incorrect data. There are a number of solutions to this problem; here are two:

- Do not use the prefetch modes at all, or
- Ensure that the each time the host requests an access from a different TMS34020, the first read request performed is to a dummy location (this ensures that any previously fetched data is flushed out of the transceivers).
- Each TMS34020 uses its own set of transceivers. Connect the HDST and HOE pins from each TMS34020 to their respective sets of transceivers. Each set of transceivers transfers data between the host and one TMS34020. Although this implementation requires more hardware, it does allow you to use the prefetch modes. It also allows each TMS34020 to have a different host, if you desire.

When a TMS34020 loses bus mastership, the HDST and HOE pins are set to high impedance just like the local-memory control signals. However, unlike the local-memory signals, HDST and HOE both have internal pull-up resistors. This allows either of the above configurations to be implemented without requiring any additional external control circuitry. If HDST and HOE are not driven by another TMS34020 (the second case described above), the resistors hold them at the logic-high level. However, the resistors are sufficiently large that they do not prevent another TMS34020 from subsequently driving HDST and HOE (the first case described above).

If you want to connect the host to only one TMS34020, but want the host to download code to the other TMS34020s (to which it is not directly attached), you can use indirect-access methods. These other methods take advantage of the fact that even if not halted, a TMS34020 cannot access memory while its GI pin is held inactive-high. By explicitly controlling the multiprocessor-arbitration logic, the host can prevent TMS34020s to which it is not attached from fetching and executing new code until after the host downloads code through the TMS34020 to which it is attached.

If you adopt this procedure, a device can inform the TMS34020 of where to execute code from by asserting an external interrupt (via LINT1 or LINT2) while  $\overline{GI}$  is held in its inactive state, then writing the address of the first word of code into the appropriate interrupt vector before asserting  $\overline{GI}$  active low. This may be most useful at power-up (when the TMS34020 fetches code from the address specified in the reset vector). At other times, the TMS34020 may not start executing the new code immediately, because it may not have been at an instruction boundary or an interruptible point within an interruptible instruction when  $\overline{GI}$  was deasserted.

In addition, the host can access other TMS34020s by using interrupt routines and/or flags (semaphores) to access information at a known location in memory, accessible to the host through the TMS34020 to which it is attached. The TMS34020's SWAPF instruction is helpful for this type of operation.

# 7.12 Systems with 16-Bit Memory Devices

If some or all of the TMS34020's local memory is made up of 16-bit memory devices, the TMS34020's memory controller uses its dynamic bus-sizing capability to access this memory.

If the host wants to access 16-bit local memory, you must add transceivers to route the data to and from the correct halves of the TMS34020's local address/ data bus (LAD0—LAD31).

If the host uses a 32-bit bus and at least part of the TMS34020's local memory is 16 bits wide, then you'll need six 8-bit transceivers:

- The 4 regular transceivers for transferring data between the host and the 32-bit local memory.
- 2 additional transceivers to multiplex data between whichever half of the local bus the 16-bit memory is attached to and the opposite half of the host's data bus.

For example, if the 16-bit memory is configured on LAD0—LAD15, then the extra transceivers are required to multiplex data between LAD0—LAD15 and bits 16—31 of the host's data bus (when accessing the second of the two 16-bit words contained in the 32-bit location specified by HA5—HA31).

If 16-bit memory is configured to both halves of the TMS34020's local-memory address/data bus (LAD0—LAD15 and LAD16—LAD31), then you need eight 8-bit transceivers because it is necessary to swap both halves of the local address/data and host data buses.

Even if the host is configured to a 16-bit bus and all of the TMS34020's local memory is 16 bits wide, four 8-bit transceivers are still required because the I/O registers are 32 bits wide. Only in the unlikely situation in which the host would never access any I/O registers could you use two 8-bit transceivers to connect the host's 16-bit data bus to the appropriate half of the TMS34020's local address/data bus.

Data should be routed according to the SIZE16 signal. For more details about dynamic bus sizing, refer to Section 8.6, <u>Dynamic Bus Sizing</u>, on page 8-12.

If the host accesses 16-bit-wide memory, the TMS34020's memory controller automatically performs both cycles necessary to access the full 32 bits of the long-word address present on HA5—HA31, regardless of which HBS byte strobes are active.

Figure 7–22 shows the transceiver combinations required for the various system configurations.

Figure 7–22. Host-to-TMS34020 Transceiver Wiring with 16-Bit Memory

(a) Without 16-bit memory



(c) With 16-bit memory connected to LAD16—LAD31



(b) With16-bit memory connected to LAD0—LAD15



(d) With 16-bit memory connected to both LAD0—LAD15 and LAD16—LAD31



# 7.13 Systems with Big-Endian Addressing

If the your system uses the TMS34020 in big-endian mode, then the bytes addressed by HBS0—HBS3 are swapped. See Chapter 3, *Memory Organiza-tion and Hardware-Supported Data Structures*, for more detail on the differences between big-endian and little-endian operation. Figure 7–23 shows the byte addressing for both.

#### Figure 7–23. Big-Endian and Little-Endian Byte Addressing Modes

	Byte 3	Byte 2	Byte 1	Byte 0
Bits:	31—24	23—16	15—8	7—0
Byte Strobes:	HBS3	HBS2	HBS1	HBS0

(a) Little-endian byte addressing

(b) Big-endian byte addressing

	Byte 3	Byte 2	Byte 1	Byte 0
Bits:	7—0	15—8	23—16	31—24
Byte Strobes:	HBS3	HBS2	HBS1	HBS0

For example, in order to access bits 7—0 of a word in the TMS34020's local memory in big-endian mode, the host must assert HBS3 instead of HBS0.

# Local-Memory and DRAM/VRAM Interfaces

This chapter describes the local-memory interface and the DRAM/VRAM interface.

	Sect	ion	Page
Basic information includes a review of TMS34020 signals and registers that affect the local memory and DRAM//RAM		Related Signals	8-2
		Related Registers	8-4
		Priorities of Memory Bus Requests	8-6
interfaces.	8.4	General Form of a Local-Memory Cycle	8-8
	8.5	Local-Memory Cycle Status Codes	8-10
The <b>local-memory interface</b> consists of a multiplexed	8.6 8.7	Ending a Local-Memory Cycle Performing Local-Memory Cycles	8-12
address/data bus and associated		in Page Mode	8-15
cycle, the TMS34020 outputs the	8.8	Local-Memory Read and Write Cycles	8-18
address and status on the LAD bus and then transfers the data over the LAD bus.	8.9	Accessing 16-Bit or 32-Bit Memory Devices (Dynamic Bus Sizing)	8-25
The DRAM/VRAM interface	8.10	VRAM Serial-Register Transfers	8-29
adds DHAM/VHAM control to the	8.11	VRAM Write-Mask Local-Memory Cycles	8-34
TMS34020 interfaces directly to	8.12	VRAM Block-Write Local-Memory Cycles	8-37
DRAMs and VRAMs, providing address multiplexing support for 64K×n, 256K×n, 1M×n, and 4M×n devices The DRAM//RAM	8.13	DRAM Refresh Local-Memory Cycles	8-44
	8.14	Local-Memory Cycles with Wait States	8-46
	8.15	The Host-Default Local-Memory Cycle	8-49
interface also provides	8.16	Addressing Mechanisms	8-50
programmable DRAM refresh.	8.17	Double-Buffered Display Example	8-57

# 8.1 Related Signals

Many of the TMS34020's pins provide access to and control over the local-memory and DRAM/VRAM interfaces. Chapter 2 describes these signals in detail; they are summarized below for your convenience.

Signals	Descriptions	I/O
ALTCH	is the address latch signal. An external latch (such as a 74ALS373) can use a high-to-low transition on ALTCH to maintain the current address and status from LAD0—LAD31.	0
BUSFLT	is the bus-fault signal. If external logic detects an error or fault in the current cycle, the logic asserts BUSFLT high. BUSFLT is also used in conjunction with LRDY to generate bus-retry cycles.	I
CAMD	is the column-address mode input that allows mixing of DRAM address matrices using the same multiplexed RCA0—RCA12 address signals.	1
CAS0—CAS3	are the column-address strobe signals that drive the $\overline{CAS}$ inputs of DRAMs and VRAMs. They also provide access to the individual bytes in each long-word in memory.	0
CLKIN	is the clock input signal. The TMS34020's processor functions are synchronous to CLKIN.	1
DDIN	is the local data-bus-direction input-enable signal, driven high to enable data transceivers (such as a 74ALS623) on LAD0—LAD31 to input data to the TMS34020.	0
<b>DDOUT</b> is the local data-bus-direction output-enable signal. I is driven low to enable data transceivers (such as a 74ALS623) on LAD0—LAD31 to output data from the TMS34020.		0
LAD0-LAD31	form the local address/data bus.	I/O
LCLK1, LCLK2	are the local output clocks. These signals are available to the system for synchronous control of external logic.	0
LINT1, LINT2	are the TMS34020's local interrupt-request signals. External devices can use LINT1 and LINT2 to interrupt TMS34020 operation.	I
LRDY	is the local ready signal. External circuitry drives LRDY low to inhibit the TMS34020 from completing a local-memory cycle.	I

Signals	Descriptions	I/O
PGMD	is the page-mode signal. External memory decode logic asserts PGMD low if the currently addressed memory supports burst (page mode) accesses.	I
RAS	is the row-address strobe signal that drives the $\overline{\text{RAS}}$ inputs of DRAMs and VRAMs.	0
RCA0-RCA12	form the multiplexed row/column address bus, used for the row and column addresses for DRAMs and VRAMs.	0
SF	is the special-function signal that connects to the SF pin of 1M VRAMs.	0
SIZE16	is the bus-size signal. External memory decode logic asserts SIZE16 low if the currently addressed memory supports 16-bit accesses only. SIZE16 is also used to determine to which half of the LAD bus the accesses are made.	<b>1</b> -
TR/QE	is the transfer/output-enable signal that connects to the $\overline{\text{TR}/\text{QE}}$ pins of VRAMs.	0
WE	is the write-enable signal that drives the $\overline{\text{WE}}$ inputs of DRAMs and VRAMs.	0

# 8.2 Related Registers

The local-memory interface registers summarized below are a subset of the I/O registers discussed in Chapter 4.

Four registers are associated with the local-memory and DRAM/VRAM interfaces:

- CONFIG contains several bits that control individual aspects of the local-memory interface.
- **DPYCTL** controls aspects of the DRAM/VRAM interface.
- The **PMASK** and **REFADR** registers contain 16-bit values that are necessary for proper operation of the local-memory or DRAM/VRAM interface.

CONFIG register			ado	dress: C000	01A0h
15	12—10	8	3	21	0
	RR		CBP	RCM	BEN
Note: For a complete illustration and description of CONFIG, see Chapter 4.					

BEN Setting the BEN (big-endian enable) bit to 1 tells the TMS34020 to access bit 0 memory in big-endian mode. Clearing BEN to 0 tells the TMS34020 to access memory in little-endian mode. Little-endian is the default.



The RCM (RCA bus mode configuration) bits determine the basic row- and column-address mode for the RCA bus.



Setting the CBP (configuration byte protect) to 1 write-protects bits 0—7 of the CONFIG register. This prevents modification of the selected configuration.

VEN bit 8

Setting the VEN (VRAM enable) bit to 1 informs the TMS34020 that the system contains VRAMs with special features such as write-mask and color-latch registers.

The RR (DRAM refresh rate) bits control the frequency of DRAM refresh bits10—12 cycles.



PMASKL register	address: C000 0160h
PMASKH register	address: C000 0170h
REFADR register	address: C000 01F0h
15	0

PMASKL, PMASKH	work together to form a 32-bit plane mask. The 0 and 1 values in the plane mask selectively enable or disable various planes in a multiple-bit-per-pixel bitmapped display.
REFADR	is the refresh-address register. It contains a pseudo-address that is output on each refresh cycle.

# 8.3 Priorities of Memory Bus Requests

The TMS34020's local-memory controller assigns priorities to local-memory requests from various on- and off-chip sources. The memory controller can generate VRAM serial-data-register transfer cycles, refresh cycles, host accesses to memory, CPU cycles, and emulator accesses, and it can release the local bus for other processors. Table 8–1 shows the order in which the TMS34020 responds to these access requests; priority 1 is the highest priority (the TMS34020 responds to it first).

Table 8–1. Priorities for Memory Cycle Requests

Priori	y Memory Cycle Request
1	Loss of bus grant, indicated by GI (grant input) high
2	Any video-generated VRAM serial-data-register transfer cycle
3	12 or more DRAM-refresh cycles are pending
4	Emulator (if emulator priority requested)
5	Host access
6	4 or more DRAM-refresh cycles are pending
7	TMS34020 CPU access (single cycle, page mode, or CPU-initiated serial-data-register transfer cycles)
8	Less than 4 DRAM-refresh cycles are due
9	Host-default cycle
Note:	If enabled the TMS34020 waits for a host request when no other cycle is

requested.

Here are more detailed descriptions of these memory cycle requests.

- The highest priority is assigned to the loss of bus grant, which takes priority over all other types of memory requests. However, it is recommended that the bus grant is not removed when a high-priority memory cycle code is output on the R0 and R1 signals. Requests of priorities 2—5 generate this code.
- 2) Any video-generated, VRAM serial-data-register transfer requests have the second highest priority. This is the highest priority of any internally generated request.
- 3) DRAM-refresh requests are programmed to occur at regular intervals. The TMS34020 has an internal counter that tracks how many refreshes are due. The TMS34020 increments this counter by 1 when the refresh counter requests a refresh; the TMS34020 decrements the counter by 1 when a refresh takes place. When 12 refreshes are due, refreshing becomes a higher priority to ensure that a refresh cycle occurs.

- 4) An in-circuit emulator (such as the Texas Instruments TMS34020 Emulator) can raise the priority of the emulator's and the CPU's local-memory requests above the priority of the host's local-memory requests. If you do not do this, the TMS34020 treats emulator requests with the same priority as regular CPU requests.
- 5) The host interface has the next highest priority. If a CPU-initiated cycle is in progress, it is interrupted on the next 32-bit boundary. (If the CPU is inserting a field, then both the read and write cycles complete. Any operation required as a result of dynamic bus sizing also completes.)
- 6) If the DRAM-refresh counter indicates that 4 or more cycles are pending and no page-mode accesses are in progress, then refresh cycles assume a higher priority than CPU-initiated requests. This prevents the refresh counter from progressing too far without interrupting page-mode blocks.
- 7) CPU requests are serviced if no higher priority requests are present. Once a CPU cycle begins, request with priorities 1 through 5 can interrupt a series of CPU-initiated page-mode accesses. The interrupted sequence will restart as with any ordinary access.
- 8) If the refresh counter is not 0 and nothing else is happening, then DRAM refreshes occur.
- 9) If no definite request is pending, then the TMS34020 performs a hostdefault cycle. This is a special type of idle cycle. (For more information, see Section 8.15 on page 8-49.)

Even if host requests are being made at the full bandwidth rate, there will always be one cycle after a host access when the host interface request to the memory controller is not active. If no other accesses are pending, the host-default state will enable host accesses to use the full bus bandwidth. However, if another type of access is pending, this other access will be executed in preference to the host default.

Thus, if both the host interface and the CPU are requesting access at the maximum rate possible, the memory controller services the requests alternately. This prevents the host from completely dominating the local-memory interface; the host cannot prevent servicing of other requests.

# 8.4 General Form of a Local-Memory Cycle

Most local-memory cycles initiated by the TMS34020's memory controller are at least 2 machine states in duration. (An exception to this is page-mode cycles, discussed in Section 8.7, page 8-15.) Each machine state is one LCLK period long and begins at the rising edge of LCLK1. As Figure 8–1 shows, each local-memory cycle has two parts:

- the address/status subcycle and
- the data subcycle.

Figure 8–1 also shows the signals required to latch the addresses that are on the LAD and RCA buses. Page-mode cycles are an extension of this form.





- **Notes:** 1) The data subcycle is an integral number of LCLK cycles long. The diagram shows the minimum length of 1 LCLK cycle.
  - 2) This illustration does not apply to page-mode memory cycles.
  - 3) For complete timing diagrams, refer to Sections 8.8-8.13 (pages 8-18-8-44).

## 8.4.1 The Address/Status Subcycle

The address/status subcycle coincides with the first machine state of the local-memory cycle. The address, status code, and row address for the access are output at this time. This portion of the cycle is often called the **row-address time** because the row address is output during this subcycle.

Section 8.5 (page 8-10) discusses the status codes output during this subcycle.

The LAD bus outputs an address that can be latched (using either  $\overline{\text{ALTCH}}$  or  $\overline{\text{RAS}}$ ) and used to decode the TMS34020's address space. This address points to the currently accessed long-word. The RCA bus outputs a subset of this address to serve as the *row address*. The RCA bus is usually connected

directly to DRAMs and VRAMs, which require the row and column parts of the address to be time-multiplexed over their address input pins. The address can also be latched using either ALTCH or RAS. Section 8.16 (page 8-50) provides a detailed discussion of the addressing modes.

LAD0—LAD3 convey a status code that indicates the type of local-memory cycle being performed. This information can be latched, along with the address on LAD. Section 8.5 (page 8-10) contains a detailed description of the status codes.

## 8.4.2 The Data Subcycle

This subcycle is at least one machine state long and immediately follows the address/status subcycle. The column address for the cycle is output at this time, and data is transferred between the TMS34020 and the local memory (or vice versa). This portion of the cycle is often called the **column-address time**.

The LAD bus transfers data between the TMS34020 and local memory. Data is either driven out (for a write cycle) or latched in (for a read cycle). The RCA bus outputs a column address that can be latched using any of the CAS signals. The RCA bus is usually connected directly to DRAMs and VRAMs, which require the row and column parts of the address to be time-multiplexed over their input pins. Section 8.16 (page 8-50) discusses available addressing modes.

If a 1-LCLK data subcycle is not sufficient for a data transfer to complete, the data subcycle can be extended, as required, by inserting wait states. Section 8.6 (page 8-12) deals with this in more detail.

# 8.5 Local-Memory Cycle Status Codes

In addition to the address output over the LAD lines, the TMS34020 outputs a status code on LAD0—LAD3. This status code identifies the type of local-memory cycle being performed. The code can be latched by the falling edge of ALTCH, along with the address output on the remaining LAD pins. Table 8–2 lists the status codes and their meanings.

# Table 8–2. LAD-Bus Status Codes

LAD PIN		Ν	_		
3	2	1	0	Bus Status	Туре
0	0	0	0	Coprocessor cycle	miscella-
0	0	0	1	Emulator operation	neous
0	0	1	0	Host cycle	(00xx)
0	0	1	1	DRAM refresh	· · /
0	1	0	0	Video-generated VRAM serial-register transfer	VRAM
0	1	0	1	CPU-generated VRAM serial-register transfer	(01xx)
0	1	1	0	Write-mask load	
0	1	1	1	Color-register load	
1	0	0	0	Data access	CPU
1	0	0	1	Cache fill	(1xxx)
1	0	1	0	Instruction fetch	(
1	0	1	1	Interrupt-vector fetch	
1	1	0	0	Bus-locked operation	
1	1	0	1	Pixel operation	
1	1	1	0	Block write	
1	1	1	1	Reserved	

Here are more detailed descriptions of these status codes.

- □ Coprocessor cycle. Output for any memory cycle that transfers instruction information or data between a coprocessor and either the TMS34020 or the local memory. These memory cycles are beyond the scope of this chapter but are discussed in detail in Chapter 10, *Communicating with a Coprocessor*.
- **Emulator operation.** Output during a special memory cycle that only an in-circuit emulator can initiate. This cycle is intended for loading external mapping registers implemented within the emulator using SRAM.
- Host cycle. Output on any memory cycle that is generated in response to the host interface.
- **DRAM refresh.** Output when the memory controller performs a DRAM-refresh cycle.

- ❑ Video-generated VRAM serial-register transfer. Output when the TMS34020 memory controller performs a midline-reload cycle, or a serialregister-transfer cycle initiated by horizontal blanking. (VCE[[DPYCTL]] controls the transfer direction.)
- CPU-generated VRAM serial-register transfer. Output when the TMS34020 memory controller performs a serial-register-transfer cycle due to a pixel operation instruction with CST[[DPYCTL]] = 1. A pixel-read operation generates a read transfer (VRAM memory to serial register). A pixel write generates a write transfer (serial register to VRAM memory).
- Write-mask load. Generated during the special 1-Mbit VRAM cycle to load the VRAM's on-chip write-mask register with the 1s complement of the TMS34020's PMASK registers.
- Color-register load. Generated during the special 1-Mbit VRAM cycle initiated by the VLCOL instruction, which loads the VRAM's on-chip color register.
- Data access. Output when the current memory access (read or write) involves transferring data to or from the TMS34020's CPU.
- **Cache fill.** Output when the TMS34020 is reading a subsegment (4 long-words) into the on-chip instruction cache.
- □ Instruction fetch. Output when the instruction cache is disabled (CD[[CONTROL]] = 1) and the TMS34020 is reading instructions or immediate data from memory.
- □ Interrupt-vector fetch. Output any time a TRAP, interrupt, or reset occurs. The code indicates that the TMS34020 is reading the appropriate vector address from memory.
- **Bus-locked operation.** Executed when the TMS34020 executes a SWAPF instruction (indicating a locked read-modify-write operation).
- **Pixel operation.** Output when the TMS34020 performs any pixel operation and CST[[DPYCTL]] = 0.
- Block write. Generated during one of the special 1-Mbit VRAM block-write cycles performed when executing the VBLT or VFILL instruction.

# 8.6 Ending a Local-Memory Cycle

A data subcycle begins in the machine state immediately following the address/status subcycle. The data subcycle lasts for a minimum of 1 machine state. During this subcycle, the TMS34020 samples the LRDY and BUSFLT pins to determine when and how the local-memory cycle will end. The TMS34020 also samples the PGMD and SIZE16 pins at this time. (Section 8.7, page 8-15, discusses PGMD; Section 8.9, page 8-25, discusses SIZE16.)

Immediately following the address subcycle, the TMS34020 samples the LRDY and BUSFLT pins on LCLK2's low-to-high transition in *each* machine cycle of the data subcycle. The pins are *not* sampled in subsequent page-mode data subcycles (see Section 8.7). External hardware must decode the address output during the address/status subcycle and determine the appropriate levels for LRDY and BUSFLT. Table 8–3 lists the logical combinations of these signals.

 Table 8–3.
 Bus Cycle Completion Conditions

Completion Condition	BUSFLT	LRDY	
Insert a wait state	0	0	
Successful transfer	0	1	
Retry	1	0	
Bus fault	1	1	

#### 8.6.1 Extending a Local-Memory Cycle with Wait States

BUSFLT low	LRDY low

Under these conditions, the TMS34020 inserts a wait state to extend the data subcycle. All local-memory control signals activated during the cycle remain active for an additional LCLK cycle. The TMS34020 samples LRDY and BUSFLT again on LCLK2's next rising edge. If the TMS34020 detects a wait condition again, it extends the data subcycle for another LCLK cycle, and so on.

After the first data subcycle completes successfully, subsequent accesses may occur as page-mode cycles (see Section 8.7 on page 8-15). During page-mode cycles, the TMS34020 does not sample LRDY and BUSFLT. This means that wait states can be inserted during the first data subcycle, but not during subsequent page-mode cycles. This can be useful, for example, when using external memory management hardware that must translate an address; although the memory management hardware may allow for high-speed page-mode accesses, it may not be able to quickly translate the address during the first access.

## 8.6.2 Completing a Successful Local-Memory Cycle

BUSFLT low	LRDY high
Under these conditions, all the cycle are driven inactive on LCL cycle. A successful transfer sh	local-memory control signals active during this K2's next high-to-low transition, completing the ould be indicated only if
for <b>read cycles</b> , valid data	will be available on the LAD bus at the next fall-

for write cycles, the devices being written to will be able to latch the data on the LAD bus at the next falling edge of LCLK2.

If these criteria cannot be met, the TMS34020 must extend the data subcycle by inserting wait states.

## 8.6.3 Retrying a Local-Memory Cycle

BUSFLT high

LRDY low

These conditions terminate the local-memory cycle (in the same way as a successful transfer) and subsequently restart the cycle from the beginning. The original address and status code are output again during the address/ status subcycle of the retried access.

When a retry occurs, the cycle may not restart immediately because the memory controller will first perform any higher priority requests that might be pending when the retry occurs.

Some memory operations require multiple local-memory cycles. (For example, modifying a field within a word requires a read and a write. If a field crosses a word boundary, 2 reads and writes are required. Also, accessing a 16-bit memory device requires twice the number of local-memory cycles.) Except for bus-locked local-memory cycles, a retry restarts only the memory cycle on which the retry occurred. If a retry occurs on the write portion of a read-modify-write sequence, it restarts from the write, not the read. Similarly, if a retry occurs on the second of the 2 accesses required to read or write a location implemented as 16-bit memory, the first of the 2 accesses is not performed again; the address output on the LAD and RCA buses during the restarted access will have the S bit set to 1.

The SWAPF instruction initiates bus-locked local-memory cycles used for exchanging semaphores. The read and write portions of a bus-locked operation must occur one immediately after the other. Because of this, if a retry occurs on the write part of a bus-locked operation, it is restarted from the read cycle.

If a retry occurs during a host-initiated local-memory cycle, the cycle restarts normally. However, the memory controller also sets HRYI [[HSTCTLL]], indicating that a retry occurred. If HBREN [[HSTCTLL]] is also set, the HINT pin is driven low to interrupt the host. Once set by the memory controller, HRYI remains set until the host clears the bit.

## 8.6.4 Bus Faulting a Local-Memory Cycle

BUSFLT high	LRDY high
<u> </u>	

These conditions inform the TMS34020 that the access was unsuccessful. The local-memory cycle is terminated (in the same way as a successful transfer); subsequent actions depend on the type of access that caused the bus fault.

☐ If the CPU initiated the access, the memory controller first saves its current state in the BSFLTD and BSFLTST registers, then signals a bus-fault interrupt to the CPU. (This applies to all memory cycles except those with host, DRAM refresh, or video-initiated VRAM serial-register transfer status codes.) The CPU saves on the stack all the data necessary to allow it to correctly restart execution after returning from the bus fault, then reads the bus-fault vector address from FFFFBC0h.

The bus-fault routine should clear the cause of the bus fault. Upon returning from the bus-fault routine, the CPU pops all the data it pushed before taking the interrupt. The memory controller restores the state of the faulted memory cycle from BSFLTD and BSFLTST and restarts the cycle (in the same way as for a retry). Section 6.9 (page 6-19) discusses the bus-fault interrupt in detail.

- If the host initiated the access, the memory controller terminates the local-memory cycle (in the same way as for a successful transfer) and sets HBFI[HSTCTLL] to indicate that a bus fault occurred. No other action is taken. If HBREN[HSTCTLL] is also set, the HINT pin is driven low (while HBFI is set), to interrupt the host. The host must then clear the source of the bus fault before attempting to restart the access. Once set by the memory controller, HBFI remains set until the host clears it.
- □ If the access was a DRAM-refresh or video-initiated screen-refresh cycle, no action is taken. The local-memory cycle is terminated in the same way as a successful transfer.

# 8.7 Performing Local-Memory Cycles in Page Mode

Many DRAMs and VRAMs can operate in **page mode**. This mode can greatly reduce the time required to access a memory location when performing multiple memory cycles, provided that the row address for each access is the same. The TMS34020 directly supports page-mode accesses and uses page mode whenever possible.

Figure 8–2 shows the general form of page-mode accesses. It also shows the signals required to latch the addresses present on the LAD and RCA buses. The first access of any sequence always consists of an address/status subcycle and a data subcycle, as outlined in Section 8.4 (page 8-8). However, if multiple accesses are to be made using page mode, the ALTCH and RAS signals do not go inactive at the end of the first data subcycle. They remain active until the end of the last access in the sequence, thereby ensuring that the original row address stays latched in the DRAMs throughout the access. After the first word is accessed, all subsequent accesses are made using single machine-state page-mode cycles, during which the column address for the access is output and data is transferred.





- **Notes:** 1) Except for the first data subcycle (which may be extended by inserting wait states), each page-mode data cycle is only 1-LCLK cycle long.
  - 2) n is an integer in the range of 0—63.
  - 3) For complete timing diagrams for specific local-memory cycles, refer to Sections 8.8—8.13 (pages 8-18—8-44).

## 8.7.1 Selecting Page-Mode Operation

Each time the TMS3020 performs a standard memory access, it has the opportunity to perform a page-mode access. The TMS34020 samples the PGMD pin to determine whether the DRAM/VRAM supports page mode. If subsequent memory accesses are required, these are (whenever possible) performed in page mode unless the local memory indicates that page-mode accesses are not supported at the current address.
External hardware must decode the address output during the address/status subcycle and determine the appropriate level for the PGMD pin, which the TMS34020 samples at the same time as LRDY and BUSFLT. Sampling a 0 on the PGMD pin indicates that page-mode accesses are supported. If wait states are inserted into the initial memory cycle, PGMD must be kept at a valid level on LCLK2's low-to-high transition during each machine state of the data subcycle.

The TMS34020 supports a page size of sixty-four 32-bit words. All memory locations for which logical address bits 10 through 31 are the same are considered to be within the same page. If, while performing a sequence of page-mode accesses, the TMS34020 wishes to access a location that is not on the same page as the previous access (that is, the value on bits 10 through 31 differ), the page-mode sequence ends. The access is made using an standard local-memory cycle, with an address/status subcycle and a data subcycle, so that a new row address can be output. LRDY, BUSFLT, and PGMD can be resampled.

Once  $\overrightarrow{PGMD}$  is detected low and a page-mode sequence begins, the TMS34020 does not sample LRDY, BUSFLT, and  $\overrightarrow{PGMD}$  again. This means that

- All 64 locations within the page must be capable of supporting page-mode operation.
- All locations within a page must support 0 wait-state accesses.
- ❑ No bus faults or retries can be generated within the page, once a page-mode sequence has started. If you are using bus faults to implement a virtual memory system, this means that the entire page must be mapped into the physical address space.

### 8.7.2 How the TMS34020 Uses Page Mode

The TMS34020 uses page mode any time it must make multiple accesses to contiguous addresses. Page-mode cycles can originate from two sources: the CPU and the memory controller. Assuming that the addressed memory supports page mode, a page-mode cycle occurs:

- Any time the CPU wishes to read or write multiple consecutive words in memory. This can occur during almost all graphics instructions (PIXBLT, FILL, VBLT, VFILL, etc.) and the multiple move instructions (MMTM, MMFM, and BLMOVE). It can also occur anytime a cache fill is performed, or if a context switch is made before servicing or when returning from an interrupt.
- Any time the memory controller must perform multiple memory accesses to complete a single memory operation. This occurs when the CPU wishes

to write a non-byte-aligned field within a word (in which case a read-modify-write must be performed), or if the CPU accesses a field that straddles a word boundary. It can also occur when dynamic bus sizing is used to access a 16-bit memory (because 2 memory accesses are required to completely read or write one word). This occurs on both host-and CPU-initiated accesses. Section 8.9 (page 8-25) discusses dynamic bus sizing.

#### Note:

The TMS34020's I/O registers cannot be accessed using page mode.

To allow instructions that move data from one area of memory to another to use page-mode, the TMS34020 contains a temporary register file capable of storing eight 32-bit words. This allows up to 9 words to be read from the source using page mode, and then written to the destination. If the original destination data is not to be combined with the source data in any way, these writes can also be performed using page mode.

# 8.8 Local-Memory Read and Write Cycles

This section describes all the basic, local-memory, read and write cycles. Figure 8–3 (page 8-19) illustrates general timing of the local-memory interface for reads and writes. Use these figures to observe signal relationships for the various cycles (do not use these figures to determine specific signal timings).

All of the TMS34020's local-memory output signals are synchronous to local clocks LCLK1 and LCLK2. The 90° phase shift between the two local clocks is used to divide each machine cycle into quarter cycles or **Q phases**. All local-memory output signals start transitions at the beginning of a Q phase.

All subsequent timing diagrams in this chapter indicate the following:

- GI is sampled low at the beginning of each memory cycle. After the memory cycle, GI may be at either level because the TMS34020 may or may not retain control of the local-memory bus after the cycle completes. This is shown by the dual level for DDOUT during Q1 after memory cycle completion. If bus control is regained, DDOUT is driven to high impedance at this time; otherwise, it remains high. Chapter 11, Multiprocessing and System Architecture, discusses GI in detail.
- R0 and R1 are shown in all the possible states that could occur during a particular cycle. For instance, in Figure 8–3, they initially indicate either a no-request or low-priority code (either may be appropriate if no memory cycle was performed in the previous machine state), or an access-termination code (identifying the end of the previous access). During the memory access, the low-priority code is output, followed by the access-termination code during the access' final machine state. Chapter 11 also discusses R0 and R1.
- ❑ Wherever appropriate, a page-mode sequence (consisting of a standard memory cycle followed by one page-mode cycle) is shown. Obviously, the number of page-mode cycles is typically greater than 1.

Note also that

- CAMD is sampled on LCLK1's low-to-high transition, immediately following the address/status subcycle. This selects the logical address bits that are output on the RCA bus at column-address time. (See Section 8.16, page 8-50.)
- ❑ LRDY, BUSFLT, and PGMD are sampled on LCLK2's low-to-high transition in each machine state of the sequence's first data subcycle (as described in Sections 8.6 and 8.7, pages 8-12 and 8-15). SIZE16 is sampled at this time in each data subcycle (during page-mode cycles as well). This is discussed in Section 8.9 (page 8-25).



LRDY is shown being sampled high on LCLK2's high-to-low transition during an address/status subcycle. This is to maintain compatibility with future pin-compatible TMS340x0 devices. Current devices ignore the state of LRDY. However, future devices may not; the value of LRDY sampled at this time may be used to prolong the address/status subcycle. This eases interfacing to DRAMs if the TMS34020's LCLK frequency is increased above 10MHz. Maintaining LRDY high at this time in your current designs ensures that the address/ status subcycle timing shown in the diagrams will not be affected when you use future TMS3402x devices.

## Figure 8–3. General Timing of the Local-Memory Read and Write Cycles



# 8.8.1 Local-Memory Read Cycle Timing (with Page Mode)

A local-memory read cycle (with page mode) transfers data and instructions to the TMS34020 from memory that supports page-mode accesses, as Figure 8–4 shows. The status code on LAD0—LAD4 identifies a data transfer, cache fill, instruction fetch, interrupt-vector fetch, or pixel operation.

DDOUT goes low during the address subcycle to enable the address through external bus transceivers to the memory. DDOUT returns high after the address is latched, indicating that a memory read cycle is about to take place. The LAD bus goes into a high-impedance state following the address subcycle, allowing data from the addressed memory to be placed on the bus.

External decode logic asserts PGMD low at the beginning of Q2 after RAS low (the first data subcycle), indicating that the memory supports page-mode operation. External logic also asserts LRDY high and BUSFLT low at this time, indicating that the cycle may continue without a retry, bus fault, or wait-state insertion. SIZE16 must be valid at the beginning of Q2 during all data transfer cycles, indicating whether the access is to a 32-bit or a 16-bit memory device. In Figure 8–4, the external logic asserts SIZE16 high to allow access to 32-bit memory.

The TMS34020 outputs DDIN high during the data transfer cycle, allowing the external bus transceivers to enable data to the processor. Data from the accessed memory must be valid at the beginning of Q4 in the data transfer cycle (with some data setup time). The data from the memory should be disabled with the low-to-high transition of ALTCH, RAS, CAS, or TR/QE, or with DDIN's high-to-low transition. These transitions occur well before the time at which the LAD bus turns on to output the address for the next cycle.

Note that the TMS34020 does not sample LRDY, PGMD, and BUSFLT during subsequent page-mode cycles when accessing 32-bit memory. The TMS34020 does not sample SIZE16 during the page-mode cycle, because the memory has responded that it supports 32-bit accesses. Section 8.9 (page 8-25) discusses dynamic bus sizing and SIZE16 in detail.

### 8.8.2 Local-Memory Write-Cycle Timing (with Page Mode)

As Figure 8–5 (page 8-22) shows, this page-mode write cycle transfers data from the TMS34020 to memory that supports page-mode accesses. LAD0—LAD31 output the data during the data transfer subcycle. The status code output on LAD0—LAD3 identifies a data transfer or a pixel operation.

DDOUT remains low after the initial address output on LAD (during Q4 after RAS goes low). This indicates that a memory write cycle is about to take place, allowing the data to be output through external data transceivers. WE is output low before the CAS signals go low to implement an early write to the DRAMs and VRAMs. Because data is valid both before and after the CAS signals are asserted low, external devices can latch the data on either the high-to-low or low-to-high edge of a signal that is the logical-OR of CAS and WE.



Figure 8–4. Local-Memory Read-Cycle Timing (with Page Mode)

Key: LAD (TMS34020): The TMS34020 outputs this to the LAD bus. LAD (memory): Memory outputs this to the LAD bus.



Figure 8–5. Local-Memory Write-Cycle Timing (with Page Mode)

Note that when accessing a 32-bit memory, the TMS34020 does not sample LRDY, PGMD, and BUSFLT during subsequent page-mode cycles. The TMS34020 does not sample SIZE16 during the page-mode cycle, because the memory has responded that it supports 32-bit accesses. Section 8.9 (page 8-25) discusses dynamic bus sizing and SIZE16 in detail.

# 8.8.3 Local-Memory Read/Write or Read-Modify-Write Cycle Timing

The TMS34020 performs a read-modify-write cycle when inserting a field that crosses byte boundaries. This is not the same as the read-modify-write cycle specified for some DRAMs, because the  $\overline{CAS}$  signals do not remain active low between the read and write. The read-modify-write operation consists of:

**Step 1:** A word is read from memory.

- **Step 2:** The portion of that word corresponding to the field being inserted is loaded with the new value.
- Step 3: The modified word is written back to memory.

If the accessed memory supports page-mode accesses, this operation occurs as a read followed by a page-mode write. Figure 8–6 shows the timing for the cycle. The status code output on LAD0—LAD3 identifies a data transfer, pixel operation, or bus-locked operation. Note that the column address presented during the write data transfer cycle is the same as that present during the read data transfer cycle.



Figure 8–6. Local-Memory Read/Write or Read-Modify-Write-Cycle Timing

If a bus fault or retry aborts this cycle, the cycle restarts from the aborted access. If the cycle is interrupted by a higher priority access between the read and the write, the memory controller restarts it from the write (unless the access was bus locked, in which case it restarts from the read).

Note that when accessing 32-bit memory, the TMS34020 does not sample LRDY, PGMD, and BUSFLT during subsequent page-mode cycles. The TMS34020 does not sample SIZE16 during the page-mode cycle, because the memory has responded that it supports 32-bit accesses. Section 8.9 (page 8-25) discusses dynamic bus sizing and SIZE16 in detail.

### 8.8.4 Host-Initiated Local-Memory Read and Write Cycles

When the host requests access to the TMS34020's local memory, the host interface schedules a host request to the memory controller. The memory controller then performs the appropriate local-memory cycle. The host can request a read or a write; there are two types of each request.

- Host read from ....
  - Iocal memory. This cycle transfers data from the local memory to the bidirectional data transceivers required to interface the LAD bus to the host's data bus.
  - TMS34020 I/O register. This cycle transfers data from one of the TMS34020's I/O registers to the bidirectional data transceivers required to interface the LAD bus to the host's data bus.

### Host write to ....

- Iocal memory. This cycle transfers data from the bidirectional data transceivers to the local memory.
- **TMS34020 I/O register.** This cycle transfers data from the bidirectional data transceivers to one of the TMS34020's I/O registers.

Chapter 7, *Communicating with a Host Processor*, describes these cycles in detail. Cycle timings for the four host local-memory cycles are in Figure 7–8 (page 7-19), Figure 7–9 (page 7-20), Figure 7–15 (page 7-26), and Figure 7–16 (page 7-27).

Host-initiated cycles use page mode only when accessing 16-bit memory.

# 8.9 Accessing 16-Bit or 32-Bit Memory Devices (Dynamic Bus Sizing)

The TMS34020's dynamic bus sizing capability allows the local memory to be organized as 32 bits wide, 16 bits wide, or a combination of the two. This can be useful for minimizing system cost if only a small amount of a particular memory type (such as ROM or EPROM) is required.

Initially, when the TMS34020 accesses a new location, the 16-bit word-select bit (S) is 0. (S is output on LAD4 during the address/status subcycle and on RCA0 during the data subcycle.) S=0 means that the address output is aligned to a long-word. If the memory is 16 bits wide, only 16 bits of data are transferred during this access. Another memory access with S=1 must be performed to transfer the remaining 16 bits. The SIZE16 pin provides a mechanism for doing this.

The TMS34020 samples SIZE16 on LCLK2's low-to-high transition during the data subcycle (at the same time as LRDY, BUSFLT, and PGMD). SIZE16 conveys two different but related pieces of information to the TMS34020, depending on the value of S output during the memory cycle.

- During the initial access to a location (when S=0), SIZE16 determines whether the memory is 32 or 16 bits wide. Asserting SIZE16 low selects 16-bit wide memory.
- If SIZE16 was asserted low during the initial access to a location, the TMS34020 performs a second access (with S=1). During this memory cycle, SIZE16 is used to determine which half of the LAD bus the memory is attached to (LAD0—LAD15 or LAD16—LAD31).

Table 8–4 summarizes the interpretation of the different combinations of SIZE16 and S.

### Table 8–4. Interpretation of SIZE16

SIZE16 (S=0)	SIZE16 (S=1)	Interpretation	
 1		32-bit access	
0	0	16-bit access via LAD0—LAD15	
0	1	16-bit access via LAD16—LAD31	

**Note:** If SIZE16 is sampled as a 1 when S=0, no access is made with S=1.

Determining which level to assert for the  $\overline{SIZE16}$  pin depends on which half of the LAD bus the memories are wired to:

- ❑ When data is transferred over LAD0—LAD15, the external decode logic should simply assert SIZE16 low during all accesses to 16-bit memory.
- ❑ When data will be transferred over LAD16—LAD31, the external decode logic can use the S bit's value to determine which level to assert on SIZE16 during accesses to 16-bit memory.

### 8.9.1 Data Transfer Using Dynamic Bus Sizing

With dynamic bus sizing, the initial access (with S=0) is no different from a regular 32-bit access. The TMS34020 asserts the appropriate  $\overline{CAS}$  strobes for the access and transfers data over the LAD bus in the normal way. During the second access (with S=1), the local-memory control signals behave differently:

- The 2 least significant CAS strobes are swapped with the most significant CAS strobes; the levels initially output on CAS0 and CAS1 (when S=0) are output on CAS2 and CAS3, respectively, and vice versa.
- During a local-memory write, the LAD bus halves are swapped; the data initially output on LAD0—LAD15 (when S=0) is output on LAD16—LAD31, and vice versa.
- During a local-memory read, the data latched from LAD0—LAD15 is swapped onto bits 16—31 of the internal data bus, and vice versa.

This mechanism allows 16-bit memories to be wired directly to one half of the LAD bus and to the corresponding pair of  $\overline{CAS}$  strobes, without the need for any external multiplexing logic.

Figure 8–7 shows a read operation from 16-bit memory. All CAS strobes are active during both read cycles. Figure 8–8 shows a write operation to a 16-bit memory device. Note how the CAS strobes have been swapped during the second access. Both diagrams show page-mode operation (discussed in Section 8.9.2).

#### Note:

During local-memory read cycles to 16-bit memory, the TMS34020 must merge the 16 bits of valid data latched during the second (S=1) cycle with the 16 bits of valid data latched during the initial (S=0) cycle, forming the full 32 bits for the access. During the initial cycle, all 32 bits of data on the LAD bus are latched, but only half is valid. During the second cycle, the level sampled on  $\overline{SIZE16}$  determines which half of this data is overwritten by the second 16 bits of valid data. Because the halves of the LAD bus can be swapped only during the second access, memories connected to LAD16— LAD31 must actually transfer the 16 MSBs of data during the initial cycle, and the 16 LSBs during the second cycle. This can be achieved by inverting the S bit before connecting it to the memory's least significant address pin.

When performing a read-modify-write operation to 16-bit memory, the TMS34020 performs the 2 read cycles first, followed by the 2 write cycles.

If the local-memory system indicates that the location being accessed is arranged as 16-bit memory, the TMS34020 always accesses both 16-bit words, even if it was only attempting to insert a field in the first of the two 16-bit words.



*Figure 8–7.* Dynamic Bus Sizing for a Read Cycle (Connection to LAD0—LAD15, Indicated by SIZE16 Low During 2<sup>nd</sup> Data Cycle)





### 8.9.2 Page Mode and Dynamic Bus Sizing

Whenever possible, the TMS34020 uses page mode for accessing 16-bit memory. The TMS34020 samples SIZE16 at the same time as LRDY, BUSERR, and PGMD (on LCLK2's low-to-high transition immediately follow-ing the address/status subcycle). If the TMS34020 samples PGMD low and

SIZE16 high (indicating that page-mode accesses can be made to 32-bit memory), it does not sample SIZE16 again during the page-mode sequence that may follow. If, however, the TMS34020 samples SIZE16 low (indicating 16-bit memory), it samples the signal during all subsequent page-mode accesses to determine which half of the LAD bus the memory is connected to.

- If the memory is connected to LAD0—LAD15, SIZE16 is asserted low during all subsequent page-mode cycles.
- If the memory is connected to LAD16—LAD31, the value of S output on the RCA bus can be used to determine the level of SIZE16 required for each page-mode cycle, as Table 8–4 (page 8-25) illustrates.

Page mode greatly increases the rate of access to 16-bit memory. For example, reading or writing a long-word in 16-bit memory requires 2 standard memory cycles without page mode (at least 4 machine states), or a standard memory cycle followed by a page-mode cycle if page mode is supported (3 machine states). If subsequent words are accessed using page mode, only 2 machine states are required for each long-word. Similarly, a read-modify-write operation takes only 5 machine states in page mode, compared to 8 without.

## 8.9.3 Bus-Locked Operation and Dynamic Bus Sizing

The SWAPF instruction initiates the bus-locked read-modify-write operation. This operation is used for passing semaphores to and from a memory location that both the host and the TMS34020 can access. It is uninterruptible; if the cycle *is* interrupted between the read and the write, it starts again from the read. This ensures that the write always immediately follows the read. However, if 2 reads and 2 writes are required, uninterruptibility cannot be guaranteed, and because of this, the bus-locked local-memory cycle does not sample the SIZE16 pin and does not support 16-bit memory.

A single read and write are performed. Each cycle outputs only S=0. Any semaphores contained in 16-bit memory locations that require S=1 in order to access them are not accessed during a bus-locked operation.

# 8.10 VRAM Serial-Register Transfers

The TMS34020 provides control for data transfers between the serial registers and the memory array of VRAMs. This is achieved by the appropriate timing of the SF, CAS, TR/QE, and WE pins of the VRAMs at the falling edges of RAS and CAS. The cycles include

- Memory-to-serial-register cycle (VRAM read transfer)
- Memory-to-split-serial-register cycle (VRAM split-serial-register-read transfer)
- Serial-register-to-memory cycle (VRAM write, pseudo-write transfers)
- Serial-register-to-memory cycle (VRAM alternate-write transfer)

## 8.10.1 Memory-to-Serial-Data-Register Cycle (VRAM Read Transfer)

- Performed when
   executing a pixel-read instruction and
   CST[DPYCTL] =1
  - *or* video timing logic requests a horizontal-blank-reload cycle and
    - VCE[[DPYCTL]] = 0

Indicated by

TR/QE and SF low and
 CAS and WE high when RAS goes low

Status code

identifies a memory-to-register transfer

- 0100<sub>2</sub> (initiated by the video timing logic) or
- 0101<sub>2</sub> (initiated by the CPU)

Although the TMS34020 ignores PGMD and SIZE16 during this cycle, you should hold them at a valid level (as the figures show them).

As Figure 8–9 shows, this cycle causes TR/QE to make its low-to-high transition at the beginning of Q3.

Figure 8–9. Memory-to-Serial-Data-Register Cycle (VRAM Read Transfer)



# 8.10.2 Memory-to-Split-Serial-Data-Register Cycle (VRAM Split-Register Midline-Reload Transfer)

Performed when	<ul> <li>SCOUNT overflows from all 1s to all 0s,</li> <li>VCE[[DPYCTL]] is cleared to 0, and</li> <li>SSV[[DPYCTL]] and SRE[[DPYCTL]] are set to 1</li> </ul>
Indicated by	<ul> <li>TR/QE low and</li> <li>CAS, SF, and WE high when RAS goes low</li> </ul>
Status code	01002 (video-initiated VRAM-memory-to-register cycle)

Although the TMS34020 ignores PGMD and SIZE16 during this cycle, they should be held at valid levels.

Figure 8–10. Memory-to-Split-Serial-Data-Register Cycle (VRAM Split-Register Midline-Reload Transfer)



### 8.10.3 Serial-Data-Register-to-Memory Cycle (VRAM Write Transfer and Pseudo-Write Transfer)

Performed when	<ul> <li>video timing logic requests a horizontal-blank reload and</li> <li>VCE[DPYCTL] and SRE[DPYCTL] are set to 1</li> </ul>
Indicated by	<ul> <li>TR/QE, SF, and WE low and</li> <li>CAS high when RAS goes low</li> </ul>
Status code	01002 (video-initiated VRAM-memory-to-register cycle)

The level of the VRAM SOE pin at the falling edge of RAS selects between write transfer (SOE low) and pseudo-write transfer (SOE high) cycles. Section 9.13, <u>Video RAM Control</u> (page 9-42), describes application of this cycle.





Although the TMS34020 ignores PGMD and SIZE16 during this cycle, they should be held at valid levels as shown. Note that external logic must provide control of the VRAM SOE pin.

## 8.10.4 Serial-Data-Register-to-Memory Cycle (VRAM Alternate-Write Transfer)

 Performed when
 pixel-write instruction is executed and

 CST[[DPYCTL]] is set to 1

 Indicated by
 CAS, WE, and SF high when RAS goes low and

 TR/QE low when RAS goes low

Status code 0101<sub>2</sub> (CPU-initiated VRAM-memory-to-register cycle)

This cycle does not require use of the VRAM's SOE pin and does not affect the status of the serial I/O pins. Although the TMS34020 ignores PGMD and SIZE16 during this cycle, they should be held at valid levels, as shown.





### Note:

VRAMs without an SF pin do not directly support the VRAM alternate-write transfer. However, the only difference between this cycle and the VRAM write transfer (see Section 8.10.3) is the state of SF at the falling edge of  $\overline{RAS}$ , so both cycles appear the same to the VRAMs. In this case, the external hardware used to control the VRAM  $\overline{SOE}$  pin must operate for both types of cycle.

# 8.11 VRAM Write-Mask Local-Memory Cycles

Some VRAMs (such as the TMS44251) contain special logic to enhance the performance of writing certain types of data to the memory. The TMS34020 provides direct support for the VRAM's on-chip write-mask and block-write capability.

A write-mask register within the VRAM allows each 1-bit plane within the memory array to be selectively writable or write protected. This is equivalent to the TMS34020's plane mask (described in Section 12.10, page 12-39). When plane masking is used, the memory controller must normally perform a read-modify-write operation, so that only the required bits of each pixel are modified. However, when a copy of the plane mask is stored within the VRAM, the memory controller need only perform a special write cycle that uses the VRAM's write mask. The plane masking is then carried out *within the VRAM*, as data is written. Obviously, the ability to perform a write instead of a read-modify-write significantly increases the rate at which data within the memory can be modified when using plane masking.

Writing a 1 to VEN[[CONFIG]] indicates that the VRAMs in the system contain a write-mask register. The TMS34020 provides the following local-memory cycles to enable the VRAM's write mask to be used:

- Load-write-mask cycle
- Write cycle (with mask)
- Block-write cycle (with mask)

Section 8.12 (page 8-37) discusses block-write cycles, including block-write with mask.

### 8.11.1 Load-Write-Mask Cycle

Performed when P

PMASKL or PMASKH is written to and
 VEN[CONFIG] is set

Indicated by

- CAS, WE, TR/QE, and SF high at the falling edge of RAS and
- SF low at the falling edge of CAS

Status code

0110<sub>2</sub> (write-mask load)

After the plane mask is copied into the TMS34020's PMASK registers, the 1s complement of PMASK is written to a special register on the VRAM that is used in subsequent cycles requiring a write mask. (The 1s complement of PMASK is output on the load-write-mask cycle because the TMS34020 *masks* the bits that are set high in the PMASK registers and the VRAM write mask *enables* the bits that are set high in the write-mask register.)

Although the TMS34020 ignores CAMD, PGMD, and SIZE16 during this cycle, they should be at a valid level at the time they would normally be sampled.



Figure 8–13. Load-Write-Mask Cycle

# 8.11.2 Write Cycle (with Mask)



The data on LAD is written to memory just as a normal DRAM write, except that data in the VRAMs' write mask enables the data bits that are written to memory.

Q4 Q1 Q2 Q3 Q4 Q1 Q2 Q3 Q4 Q1 Q2 A3 Q4 Q1 Q2 Q3 Q4 Q1

Figure 8–14. Write Cycle (with Mask)

ଗ 🕅



# 8.12 VRAM Block-Write Local-Memory Cycles

Some VRAMs (like the TMS44251) contain special logic to enhance the performance of writing certain types of data to the VRAM. The TMS34020 directly supports the VRAM's on-chip write-mask and block-write capabilities.

The block-write feature provides a method for writing to up to 128 bits of data during a single TMS34020 local-memory cycle (when writing a specific data pattern). The value latched on each of the VRAM's data pins enables/disables a write of a multi-bit value into the memory. This multi-bit value (typically 4 or 8 bits) is stored in a color register within each VRAM. Block-write may be used to implement several functions; however, these are typically variations of two applications:

- Fast fills. A fast fill uses block write to replicate the same pixel value or pattern in an area of memory. The pixel value or pattern is stored in the VRAM's color register.
- Data expansion. An example of data expansion is a bitmap stored in compressed form (with 1s representing the presence of a pixel and 0s representing the absence of a pixel). Typically, this sort of bitmap expansion is applied to character fonts, which may be stored in compressed form to save memory; data expansion allows the characters to be displayed in color (multi-bit pixels), but stored in black and white (1-bit pixels).

# 8.12.1 VRAM Support of Block-Write Cycles

During a block-write cycle, VRAMs ignore the 2 LSBs of the column address. This means that the 4 locations specified by the remainder of the column address are all selected. Which of these 4 locations is written to is determined by the value on the VRAMs' data pins at the falling edge of  $\overline{CAS}$ ; a 1 on a data pin enables a write, a 0 prevents a write. Each data pin is associated with the appropriate location; the least significant data pin controls writes to the least significant location, the most significant data pin controls writes to the most significant location. If all 4 data pins are 1s, all 4 locations are written to.

When a write to a particular location is enabled, the value stored in the VRAM color register is copied into that location. If required, the VRAM write mask (described in Section 8.11.1, page 8-34) can be used to selectively enable or disable the write of each of the bits within the color register.

### 8.12.2 TMS34020 Support of VRAM Block-Write Cycles

The TMS34020 provides the following local-memory cycles to enable the VRAM block write to be used:

- Load-color-register cycle
- Block-write cycle (without mask)
- Block-write cycle (with mask)

All of these cycles are initiated by specific instructions. Refer to the VLCOL, VBLT, and VFILL instructions in Chapter 13, and Sections 12.5.4 (page 12-14,

VRAM Block Mode PIXBLT) and 12.5.7 (page 12-16, VRAM Block Mode Fill) for more information on restrictions and use of these cycles.

#### 8.12.3 Load-Color-Register Cycle

Performed when

executing a VLCOL instruction

Indicated by

- CAS, WE, TR/QE, and SF high at the falling edge of RAS and
- SF high at the falling edge of CAS

Status code

0111<sub>2</sub> (color-latch load)

The data in COLOR1 is output on the LAD bus and is written to a special VRAM register which is used during subsequent block-write cycles. The VRAMs ignore the address output during this cycle, so the TMS34020 outputs all 0s on LAD0—LAD31.

Although the TMS34020 ignores CAMD, PGMD, and SIZE16 during this cycle, these signals should be at a valid level when they would normally be sampled.





Local-Memory and DRAM/VRAM Interfaces

## 8.12.4 Block-Write Cycle (Without Mask)

Performed when	<ul> <li>executing a VBLT or VFILL instruction and</li> <li>PMASK = 0</li> </ul>
Indicated by	<ul> <li>CAS, WE, and TR/QE high at the falling edge of RAS,</li> <li>SF low at the falling edge of RAS, and</li> <li>SF high at the falling edge of CAS</li> </ul>
Status code	1110 <sub>2</sub> (block write)

The data stored in the VRAM's color register is written to the memory locations enabled by the appropriate data bits output on the LAD bus. This cycle allows up to a total of 128 bits to be written at once.

Although 16-bit transfers are allowed on this cycle, external multiplexing logic is required to map the data correctly to the appropriate memories.

Figure 8–16. Block-Write Cycle (Without Mask)



### 8.12.5 Block-Write Cycle (with Mask)

Performed when executing a VBLT or VFILL instruction and

- PMASK ≠ 0
- CAS, TR/QE, and SF high at the falling edge of RAS,
  - WE low at the falling edge of RAS, and
- SF high at the falling edge of CAS

Status code

Indicated by

1110<sub>2</sub> (block write)

The data stored in the VRAM's color register is written to the memory locations enabled by the appropriate data bits output on the LAD bus. The value in the VRAM's write mask determines which bits of the VRAM's color register are written. This cycle allows up to a total of 128 bits to be written at once. Although this cycle allows16-bit transfers, external multiplexing logic would be required to correctly map the data to the appropriate memories.

Figure 8–17. Block-Write Cycle (with Mask)



# 8.12.6 Data Mapping During Block-Write Cycles

Instructions that use block write (VBLT and VFILL) operate on data that is essentially a 1-bit-per-pixel representation of multi-bit pixels in the VRAM. However, because of the data expansion performed inside the VRAMs, the 1-bit data must be remapped (reordered) before being output on the LAD bus.

For example, consider a system that uses 4 bits per pixel and eight 4-bit VRAMs (such as 1-Mbit VRAMs constructed as 256Kx4). For reference, number the VRAMs from 0 to 7, with VRAM 0 connected to the 4 LSBs of the LAD bus, and VRAM 7 connected to the 4 MSBs of the LAD bus. Because the pixel size is 4, each regular memory access to a long-word accesses 8 pixels at a time. Because there are 8 VRAMs, pixels with adjacent addresses are not physically stored in the same VRAM. VRAM 0 holds pixels 0, 8, 16, 24, etc. In general, the VRAM containing pixel *n* also contains pixels *n*+8, *n*+16, *n*+24, etc. Table 8–5 shows this.

Table 8–5. Connections of 4-Bit VRAMs to the TMS34020 LAD Bus for 4 Bits per Pixel

	VRAM 7					'RA	M	6	VRAM 5				VRAM 4				VRAM 3				VRAM 2				۷	RA	M	1	VRAM 0			
VRAM Bit #	3	2	1	0	3	2	1	0	3	2	1	0	3	2	1	0	3	2	1	0	3	2	1	0	3	2	1	0	3	2	1	0
LAD Bus Connection	3 1	3 0	2 9	2 8	2 7	2 6	2 5	2 4	2 3	2 2	2 1	2 0	1 9	1 8	1 7	1 6	1 5	1 4	1 3	1 2	1 1	1 0	9	8	7	6	5	4	3	2	1	0
D0 Pixel #		7	7			6			5				4			3			2			1				0						
D1 Pixel #		1	5			1	4			1	3		12			11			10			9				8						
D2 Pixel #		23			22					21				20			19			18			17				16					
D3 Pixel #		3	1			30				29			28			27			26			25				24						

Note: Dn Pixel # refers to the pixels controlled by VRAM data pin Dn during a block-write cycle (n is 0, 1, 2, or 3).

Normally, when the TMS34020 addresses long-word 0, the 8 VRAMs access pixels 0—7; when the TMS34020 addresses long-word 1, the VRAMs access pixels 8—15. Now consider a block-write cycle; the VRAMs ignore the 2 LSBs of the address, selecting 4 contiguous 4-bit locations within each VRAM for access. The 4 data lines act as write enables; a 1 on a data line enables a write of the color register to the corresponding 4-bit location in the VRAM. The data from the TMS34020 is organized so that each bit enables (or disables) the writing of a pixel, allowing 32 pixels to be written for each long-word of data passed to the VRAMs. As Table 8–5 shows, the data on LAD0—3, for example, controls pixels 0, 8, 16, and 24 in the VRAM. It is evident that the order of the bits within the CPU is not the same as the order of the bits expected by the VRAM.

If you want to write pixel 8 only, then the D1 data input of VRAM 0 should be a 1 (that is, the long-word on the LAD bus must be 0000 0002h). However, because the TMS34020 operates on 32 contiguous bits that represent 32 contiguous pixel locations in the memory, enabling pixel 8 is represented by setting bit 8 (so the long-word in the CPU is 0000 0100h).

The TMS34020 logic performs the remapping necessary to reorder the bits within each 32-bit word, so the bits are written out over the LAD bus in the order expected by the VRAMs. Table 8–6 shows the remapping for 4 bits per pixel.

Internal Data Bit	LAD Pin Number	VRAM Connection	Internal Data Bit	LAD Pin Number	VRAM Connection
0	0	VRAM 0, data pin 0	16	2	VRAM 0, data pin 2
1	4	VRAM 1, data pin 0	17	6	VRAM 1, data pin 2
2	8	VRAM 2, data pin 0	18	10	VRAM 2, data pin 2
3	12	VRAM 3, data pin 0	19	14	VRAM 3, data pin 2
4	16	VRAM 4, data pin 0	20	18	VRAM 4, data pin 2
5	20	VRAM 5, data pin 0	21	22	VRAM 5, data pin 2
6	24	VRAM 6, data pin 0	22	26	VRAM 6, data pin 2
7	28	VRAM 7, data pin 0	23	30	VRAM 7, data pin 2
8	1	VRAM 0, data pin 1	24	3	VRAM 0, data pin 3
9	5	VRAM 1, data pin 1	25	7	VRAM 1, data pin 3
10	9	VRAM 2, data pin 1	26	11	VRAM 2, data pin 3
11	13	VRAM 3, data pin 1	27	15	VRAM 3, data pin 3
12	17	VRAM 4, data pin 1	28	19	VRAM 4, data pin 3
13	21	VRAM 5, data pin 1	29	23	VRAM 5, data pin 3
14	25	VRAM 6, data pin 1	30	27	VRAM 6, data pin 3
15	29	VRAM 7, data pin 2	31	31	VRAM 7, data pin 3

 Table 8–6.
 Data Remapping for Block Write at 4 Bits per Pixel

Due to the organization of the VRAM, which accesses 4 bits at a time, the data expansion of the block-write feature is possible only when PSIZE  $\geq$ 4.

Table 8–7.	Block-Write Data Expa	ansion
------------	-----------------------	--------

Bits per pixel	Pixels per 32-bit write	Pixels per block write	
1	32		
2	16	—	
4	8	32	
8	4	16	
16	2	8	
32	1	4	

The TMS34020 supports the data bus remapping required for 4, 8, 16, and 32 bits per pixel. The TMS34020 uses the value contained within the PSIZE register to determine the correct data remapping to be used. Table 8–8 and Table 8–9 show the data connections and required remapping for 8 bits per pixel. Because two 4-bit VRAMs are needed to hold an 8-bit pixel, the VRAM data lines must be controlled in pairs; thus, the 32-bit bus can write only 16 pixels at a time.

	RAM 7 RAM 6								RAM 5 RAM 4								RAM 3				RAM 2				RA	<b>M</b> 1	I	F	)			
VRAM Bit #	3	2	1	0	3	2	1	0	3	2	1	0	3	2	1	0	3	2	1	0	3	2	1	0	3	2	1	0	3	2	1	0
LAD Bus Connection	3 1	3 0	2 9	2 8	2 7	2 6	2 5	2 4	2 3	2 2	2 1	2 0	1 9	1 8	1 7	1 6	1 5	1 4	1 3	1 2	1 1	1 0	9	8	7	6	5	4	3	2	1	0
D0 Pixel # D1 Pixel #		3 7								2 6								1 5							0 4							
D2 Pixel # D3 Pixel #		11 15								10 14								9 13							8 12							

Table 8–8. Connections of 4-Bit VRAMs to the TMS34020 LAD Bus for 8 Bits per Pixel

Note: Dn Pixel # refers to the pixels controlled by VRAM data pin Dn during a block-write cycle (n is 0, 1, 2, or 3).

The VBLT and VFILL instructions always operate on 32 1-bit pixels at a time. As Table 8–7 shows, the number of pixels that can be written in each block-write cycle is less than 32 when the pixel size is 8 or more. The instructions automatically compensate for this by generating the appropriate number of block-write cycles to ensure that all the data is written. Similarly, 4 block-write cycles are required at 16 bits per pixel, and 8 at 32 bits per pixel.

Table 8–9. Data Remapping for Block Write at 8 Bits per Pixel

Internal Data Bit	Internal Data Bit	LAD Pin Number	VRAM Connection
0	16	0 and 4	VRAM 0, data 0 and VRAM 1, data 0
1	17	8 and 12	VRAM 2, data 0 and VRAM 1, data 0
2	18	16 and 20	VRAM 4, data 0 and VRAM 1, data 0
3	19	24 and 28	VRAM 6, data 0 and VRAM 1, data 0
4	20	1 and 5	VRAM 0, data 1 and VRAM 1, data 1
5	21	9 and 13	VRAM 2, data 1 and VRAM 1, data 1
6	22	17 and 21	VRAM 4, data 1 and VRAM 1, data 1
7	23	25 and 29	VRAM 6, data 1 and VRAM 1, data 1
8	24	2 and 6	VRAM 0, data 2 and VRAM 1, data 2
9	25	10 and 14	VRAM 2, data 2 and VRAM 1, data 2
10	26	18 and 22	VRAM 4, data 2 and VRAM 1, data 2
11	27	26 and 30	VRAM 6, data 2 and VRAM 1, data 2
12	28	3 and 7	VRAM 0, data 3 and VRAM 1, data 3
13	29	11 and 15	VRAM 2, data 3 and VRAM 1, data 3
14	30	19 and 23	VRAM 4, data 3 and VRAM 1, data 3
15	31	27 and 31	VRAM 6, data 3 and VRAM 1, data 3

**Note:** Data bits 0 & 16 use the same LAD pin numbering and VRAM connection information, as do data bits 1 & 17, 2 & 18, etc.

# 8.13 DRAM-Refresh Local-Memory Cycles

The TMS34020 supports DRAM and VRAM memory refresh with CASbefore-RAS refresh cycles. Each time the internal refresh counter counts the number of machine states indicated by RR[[CONFIG]], a DRAM-refresh request is scheduled. DRAM-refresh cycles are performed according to the local-memory cycle priorities outlined in Section 8.3 (page 8-6). The TMS34020 can keep track of up to 15 pending DRAM-refresh requests (although it is unlikely that this many refreshes could ever be scheduled but not performed).

The refresh cycle has been implemented on the TMS34020 to use 3 machine cycles (12 quarter cycles), so the refresh status may be used to modify the generation of the RAS and CAS signals to the DRAMs.

The refresh pseudo-address output to RCA0—RCA12 and LAD0—LAD31 comes from the 16-bit REFADR register, which is incremented after each refresh cycle. The 16 bits of the address are placed on LAD16—LAD31 and LAD0—LAD3 contain the refresh status code. All other LAD bus lines are 0s. The logical addresses on RCA0—RCA12 corresponding to LAD16—LAD31 also output the address from REFADR. The LAD and RCA buses are held constant through the refresh cycle.

Note that CAMD, PGMD, and SIZE16 are not sampled during a refresh cycle, although PGMD and SIZE16 must be held at a valid level as indicated in Figure 8–18. Once the refresh cycle has begun, GI is not sampled until it completes. LRDY and BUSFLT are not sampled until LCLK2's low-to-high transition after RAS has gone low.

If a refresh cycle is aborted because of a retry, then the count of refreshes pending is not decremented and the same pseudo-address is reissued when the refresh is restarted.

Figure 8–18. Refresh Cycle Timing



# 8.14 Local-Memory Cycles with Wait States

All other local-memory cycle timing diagrams throughout this chapter assume that LRDY is sampled high during the cycle and that there are no wait states. A slower memory that requires a longer cycle time may pull the LRDY pin low. The TMS34020 samples the LRDY input on the low-to-high transition of LCLK2 after RAS goes low, as indicated in the illustrations. If LRDY is low, the TMS34020 inserts an additional state, called a **wait state**, into the cycle by maintaining all of the signals output to memory at the same level. Wait states continue to be inserted until LRDY is sampled at a high level.

### 8.14.1 Adding Wait States in Read and Write Cycles

Figure 8–19 shows an example of a local-memory read cycle extended by 1 wait state. The first time LRDY is sampled, the TMS34020 detects a low level, causing the cycle to be extended by 1 wait state. When LRDY is sampled one LCLK period later, the TMS34020 detects a high level, permitting the cycle to complete.

All local-memory read and write cycles (including the special write-mask and block-write cycles for VRAMs) are extended in the same way as Figure 8–19. VRAM serial-register transfers are extended slightly differently.

#### Note:

PGMD, SIZE16, and BUSFLT must be held at a valid level at the beginning of each Q2 until LRDY is sampled high. The levels of PGMD and SIZE16 at the time LRDY is sampled high will affect the subsequent cycle (allowing page mode or dynamic bus sizing).



Figure 8–19. Local-Memory Read Cycle with 1 Wait State



### 8.14.2 Adding Wait States in VRAM Serial-Register Transfers

Figure 8–20 shows an example of a memory-to-serial-data-register transfer cycle extended by 1 wait state. The wait state is inserted in exactly the same way as for other local-memory cycles. During a serial register transfer,  $\overline{TR}/\overline{QE}$  normally makes a low-to-high transition 1/4 cycle earlier than the other local-memory control signals. When a wait state is inserted,  $\overline{TR}/\overline{QE}$  is not kept low during the wait state. All serial-register transfers exhibit this behavior.





# 8.15 The Host-Default Local-Memory Cycle

When no other local-memory requests are pending, the memory controller executes a special idle cycle known as the **host-default cycle**. Th memory controller repeats the host-default cycle until another memory request occurs.

The host-default cycle is similar to the address subcycle of a host-initiated access, except that ALTCH and RAS are not activated. Figure 8–21 shows a host-default cycle. During this cycle, the host access status code is output on LAD0—LAD3. The address output on the remainder of the LAD bus and the RCA bus comes from the internal register used to store the address provided by the host for a host access.





The address and status code output are normally of no significance, because ALTCH and RAS are not activated. The reason they are output at all is because

the host-default cycle provides a mechanism for reducing the time taken to respond to a host request when the local-memory interface is idle.

If the TMS34020 synchronizes a host request while performing the hostdefault cycle, internal logic converts the cycle into an ordinary host access (by asserting ALTCH and RAS at the appropriate times). This is similar to any other host access. If this happens, the address output at the beginning of the hostdefault cycle will be the valid address for the access.

# 8.16 Addressing Mechanisms

The TMS34020 uses a 32-bit logical address that points to a bit in memory. The logical address bits are numbered from 0 to 31, where bit 0 is the LSB and bit 31 is the MSB. The TMS34020 uses logical address bits 0—4 to specify the bit address for a particular access but does not output these bits. Thus, each of the TMS34020's memory accesses are initially aligned to a 32-bit boundary.

A 16-bit word address pointer (S) is output in place of logical address bit 4. This bit is a 1 only when you are using the dynamic bus sizing capability of the TMS34020 to access the second of the two 16-bit memory locations needed to contain all the data specified by the logical long-word (32-bit) address. Section 8.9 (page 8-25) describes dynamic bus sizing in detail.

During each local-memory cycle, the TMS34020 outputs the long-word address and 16-bit word select in two different formats: nonmultiplexed and multiplexed.

### 8.16.1 Nonmultiplexed Addressing

During an address/status subcycle, the following information is output on the LAD bus.

- □ full logical long-word address on LAD5—LAD31
- □ 16-bit word select (S) on LAD4
- local-memory cycle status code on LAD0—LAD3

Figure 8–22 shows this. The high-to-low transition of the address latch signal (ALTCH) can be used to hold this information in external latches (such as the 74ALS573) during the entire memory cycle for memory devices or decode logic that require a static address.

For accesses to 32-bit memories, the S bit on LAD4 is always 0. S=1 only if the TMS34020 accesses a 16-bit memory that does not support page mode. In this case, separate local-memory cycles (consisting of both address/status and data subcycles) must be performed for each 16-bit location within the long-word being accessed. S=1 for the second of these accesses. Sections 8.7 and 8.9 (pages 8-15 and 8-25) discuss page mode and dynamic bus sizing, respectively.

Figure 8–22. Logical Address Output on LAD



### 8.16.2 Multiplexed Addressing

The TMS34020 provides direct support for DRAMs and VRAMs, which expect an address' row and column parts to be provided sequentially, time-multiplexed over the same wires. The 13-bit RCA bus is used for this purpose. For the sake of brevity, this section uses the word *DRAM* to refer to DRAM, VRAM, or any other device that expects multiplexed row and column addresses.

During an address/status subcycle, a subset of the logical long-word address output on the LAD bus is output simultaneously on the RCA bus. This serves as the row address for the DRAMs. During a data subcycle, a different subset of the logical address is output on the RCA bus. This serves as the column address for the DRAMs. The 16-bit word select (S) is also output on the RCA bus during the data subcycle.

The RCA bus is designed so that DRAMs from  $64K \times n$  to  $16M \times n$  can be connected directly, without external multiplexing hardware. (*n* represents the number of 1-bit memories of the base size integrated onto a single chip. For example, 1-Mbit VRAMs are implemented as four 256K memories within a single chip; 256K×4.) To do this, the following requirements must be met:

- ❑ All the row address bits for the DRAM must be available during the address/status subcycle, and then all the column address bits must be available during the data subcycle on exactly the same RCA pins.
- All the logical address bits that comprise the row and column addresses must form a continuous field. That is, all the logical address bits between the lowest and the highest connected to the DRAMs must also be connected. If this is not the case, the memory will not be fully decoded.
- Logical address bits that are connected to the DRAMs at row-address time should not be connected to the DRAMs at column-address time.

Because the different DRAM sizes require different numbers of address lines (from 8 for  $64K \times n$  up to 12 for  $16M \times n$ ), this can be achieved only by providing a number of programmable choices for the subsets of the logical address output at row-address time and at column-address time. Two mechanisms are provided to do this:

RCM[CONFIG] determines the address subset output at row-address time. The RCM bits are loaded from bits 1 and 2 of the reset vector (address FFFF FFE0h; Section 6.12.4, page 6-26, discusses this).
The address subset output at column-address time is determined dynamically on a cycle-by-cycle basis by the CAMD pin. This allows DRAMs of different array sizes to coexist within the same system, still without the need for external multiplexing hardware.

RCM[CONFIG] determines the DRAM base array size for the application. This is the DRAM array size supported when CAMD=0. It is determined by the displacement between the logical address bit output on any given RCA pin at row-address time and the logical address bit output on the same pin at column-address time. For instance, when the base array size= $64K \times n$ , if the logical address bit output at row-address time on a given RCA pin is bit *m*, then the logical address bit output on the same pin at column-address time is bit *m*-8. For 256K×*n* base array size, it is bit *m*-9, and so on. Base DRAM array sizes supported are  $64K \times n$ , 256K×*n*, 1M×*n*, and 4M×*n*.

The TMS34020 samples the data on the CAMD pin on LCLK1's low-to-high transition at the end of the address/status subcycle. This allows you to decode the full logical address output on the LAD bus at this time (or the row address output on the RCA bus) with external logic to determine what the array size of the memory being addressed is, and thus what the column-address mode should be for the particular access. Table 8–10 shows all the possible DRAM array sizes that are supported by the various combinations of the RCM bits and the CAMD pin.

|--|

R	СМ	Base Array	Additional Arrays
1	0	(CAMD=0)	Supported by CAMD=1
0 0 1 1	0 1 0 1	64K×n 256K×n 1M×n 4M×n	256K× $n$ , 1M× $n$ 1M× $n$ , 4M× $n$ 4M× $n$ 16M× $n$ (32 bits wide only)

**Note:**  $16M \times n$  memory can be addressed only as 32-bit-wide memory; dynamic bus sizing cannot be used, because there is no way to incorporate the S bit into the address connected to DRAMs of this size.

- □ When CAMD=0, the logical address bits output on the RCA bus at column-address time allow DRAMs of the base array size to be connected.
- ❑ When CAMD=1, a different logical address mapping is generated at column-address time. For most of the RCA pins, the logical address bits output at column-address time when CAMD=0 are output 1 RCA pin higher when CAMD=1. For example, the logical address bit output on RCA6 when CAMD=0 is output on RCA7 when CAMD=1. This increases the displacement between the logical address bit output on a given RCA pin at row-address time and the logical address bit output on the same pin at column time by 1, thus allowing DRAMs with an array size 1 larger than the base array size specified by the RCM bits to be directly connected.

For some base array modes, RCA4, RCA11, and RCA12 may output logical address bits that are not determined by this CAMD=1 mapping:

- **64K**×*n* The logical address bits output on RCA11 and RCA12 are not contiguous with the logical address bits output on RCA1—RCA10. This allows for  $1M \times n$  DRAMs to be connected without the same logical address bit appearing in both row and column addresses.
- **256K**×*n* The logical address bit output on RCA12 is discontiguous with the logical address bits output on RCA1—RCA11. This allows  $4M \times n$  DRAMs to be connected without the same logical address bit appearing in both the row and column addresses.
- **4M**×*n* The logical address bits output on RCA12 and RCA4 allow  $16M \times n$  DRAMs to be connected without the same logical address bit appearing in both the row and column addresses.

### All except 4M×n

The S bit is always mapped to RCA4.

Table 8–11 lists the actual logical address bits output on the RCA bus for each of the base DRAM array sizes, with both states of CAMD.

R	CM	Base	Address		RCA Bus												
1	0	Array	Time	CAMD	12	11	10	9	8	7	6	5	4	3	2	1	0
		64K×n	Row		24	23	22	21	20	19	18	17	16	15	14	13	12
0	0	Δ=8	Column	0	16	15	14	13	12	11	10	9	8	7	6	5	S
			Column	1	23	22	13	12	11	10	9	8	7	6	5	S	S
		256K×n	Row	—	25	24	23	22	21	20	19	18	17	16	15	14	13
0	1	∆=9	Column	0	16	15	14	13	12	11	10	9	8	7	6	5	S
	-		Column	1	26	14	13	12	11	10	9	8	7	6	5	S	S
		1M× <i>n</i>	Row	—	26	25	24	23	22	21	20	19	18	17	16	15	14
1	0	∆=10	Column	0	16	15	14	13	12	11	10	9	8	7	6	5	S
			Column	1	15	14	13	12	11	10	9	8	7	6	5	S	S
		$4M \times n$	Row	—	27	26	25	24	23	22	21	20	19	18	17	16	15
1	1	Δ=11	Column	0	16	15	14	13	12	11	10	9	8	7	6	5	S
			Column	1	28	14	13	12	11	10	9	8	7	6	5	S	16

Table 8–11. Logical Addresses Output on the RCA Bus

Key:  $\Delta$  is the displacement between the logical address bits output at row-address time and column-address time on a given pin, for CAMD=0.

*n* is the number of 1-bit memories for each of the base array sizes, integrated within a single chip.

**S** is the 16-bit word select.

During page-mode memory cycles (described in detail in Section 8.7, page 8-15), the data subcycle is repeated. Each time access to a new location is required, the address output on the RCA bus at this time is incremented or decremented.

### 8.16.3 Display Memory Requirements for Multiplexed Addressing

The TMS34020's RCA bus makes provision for many different multiplexed addressing schemes. However, in a graphics based system, there are some additional considerations that should be taken into account regarding the display memory.

When addressing the display memory (the memory area that stores the graphics image output to the screen), the logical address bits connected to the VRAMs must be sequential. That is, the least significant logical address bit is the LSB of the address presented to the VRAMs, the next least significant logical address bit is the next LSB of the address presented to the VRAMs, and so on up to the most significant logical address bit, which is the MSB of the address presented to the VRAMs.

If this is not the case, pixels with sequential addresses are not stored in sequential locations within the VRAM. This means pixels do not appear on the screen in the order they appear in the address map. Because of this requirement, the display memory should not be implemented with

- $\square$  1M×*n* VRAMs when the base DRAM array size is 64K×*n*
- 4M×n VRAMs when the base DRAM array size is 256K×n
- **16**M×n VRAMs when the base DRAM array size is 4M×n

When using DRAMs to store data that is not to be output to the screen, the order in which the data is actually stored in the memory is unimportant; thus, these considerations do not apply.

### 8.16.4 Example Connections for Multiplexed Addressing

Table 8–12 provides example wiring configurations for DRAMs of different array sizes. This table assumes that any bank selecting is done with high-order logical-address bits so that the banks are not interleaved. This means that the DRAMs are wired to the lowest RCA pins possible.

In all modes except  $4M \times n$  with CAMD=1, connections are shown for memory arranged as 32 bits wide and 16 bits wide. The latter requires the use of the TMS34020's dynamic bus sizing capability. You can't connect16-bit-wide  $16M \times n$  memory; if the S bit is connected to one of the memory's address pins, logical address bit 15 cannot be connected. This means that the data within the DRAM appears at 2 logical addresses (one for each of the values of logical address bit 15).

Base		DRAM	RCA Bus Wiring												
Array	CAMD	Array	12	11	10	9	8	7	6	5	4	3	2	1	0
64K×n	0	64K (16) 64K (32)					a7	a7 a6	a6 a5	a5 a4	a4 a3	a3 a2	a2 a1	a1 a0	a0
	1	256K (16) 256K (32) 1M (16) 1M (32)	a9	a9	a8 a8	a8 a7 a8 a7	a7 a6 a7 a6	a6 a5 a6 a5	a5 a4 a5 a4	a4 a3 a4 a3	a3 a2 a3 a2	a2 a1 a2 a1	a1 a0 a1 a0	a0 a0	
256K×n	0	256K (16) 256K (32)				a8	a8 a7	a7 a6	a6 a5	a5 a4	a4 a3	a3 a2	a2 a1	a1 a9	a0
	1	1M (16) 1M (32) 4M (32)	a10	a9 a9	a9 a8 a8	a8 a7 a7	a7 a6 a6	a6 a5 a5	a5 a4 a4	a4 a3 a3	a3 a2 a2	a2 a1 a1	a1 a0 a0	a0	
1 M× <i>n</i>	0	1M (16) 1M (32)			a9	a9 a8	a8 a7	a7 a6	a6 a5	a5 a4	a4 a3	a3 a2	a2 a1	a1 a0	a0
	1	4M (16) 4M (32)	a10	a10 a9	a9 a8	a8 a7	a7 a6	a6 a5	a5 a4	a4 a3	a3 a2	a2 a1	a1 a0	a0	
4M× <i>n</i>	0	4M (16) 4M (32)		a10	a10 a9	a9 a8	a8 a7	a7 a6	a6 a5	a5 a4	a4 a3	a3 a2	a2 a1	a1 a0	a0
	1	16M (32)	a11	a10	a9	a8	a7	a6	a5	a4	a3	a2	a1		a0

Table 8–12. Example Connections to the RCA Bus

Note: a0-a11 are the DRAM address pins (a0 is the least significant).

# 8.16.5 Memory Organization and Bank Selecting

System memory is typically partitioned into several banks. Each bank contains the number of memory devices that can be accessed in a single memory cycle. Thus, the number of memory devices per bank is determined by dividing the data-bus width by the memory-device width. The TMS34020's data bus can access 16 or 32 bits per cycle (depending on dynamic bus sizing). Therefore, a bank composed of  $1M \times 1$  RAMs contains 16 or 32 RAM devices. Using  $256K \times 4$  RAMs requires either 4 or 8 RAM devices.

Logical address bits not used to form the row and column addresses of the DRAMs can be used to select between banks. These bits can be decoded with external logic, so that only the RAS or CAS strobe to the appropriate memory bank is enabled. This can be achieved in two ways:

□ The DRAM's address pins can be connected to the RCA bus' lower end. Use higher order logical address bits to select between different banks; prevent activation of the unselected banks' RAS pins. These can be decoded from either the LAD or RCA bus during the address/status subcycle. In this case, the long-words within a given bank are contiguous. For instance, if each bank contains *m* words, then one bank may contain words 0 to *m*−1, the next bank words *m* to 2*m*−1, and so on. ❑ The DRAM's address pins can be connected to the RCA bus' upper end. Use lower order logical address bits to select between different banks; prevent activation of the unselected banks' CAS pins. These can be decoded from the RCA bus during the data subcycle. In this case, the long-words within each bank are interleaved—if there are 2 banks, all odd words are in one bank, and all even words in another; each bank contains alternating words. If there are 4 banks, each bank contains every fourth word, etc.

Although you can use either form of bank selecting (or a combination of the two), be aware that interleaved banks do not easily support the use of the special block-write features available on some VRAMs and supported by the VBLT and VFILL instructions. This is because the VRAM block write allows 4 long-words to be written to simultaneously. If you use interleaved banking, these 4 words will not all be contained within the same memory bank. However, because the VRAMs ignore the 2 LSBs of the column address when a block-write cycle is performed, it is not possible to write less than 4 long-words at a time to a given bank.

In a typical graphics system, the local memory is divided into two parts:

- display memory
- *system memory* (additional DRAMs needed to store programs and data)

A high-order address bit is typically used to select between the two. Within each part, other address bits can be used to select particular banks.

The decode logic must be capable of more than just selecting a particular bank of the display memory or system memory during a memory read or write cycle. It must also be able to enable all DRAMs and VRAMs during a DRAM-refresh cycle and to enable the appropriate VRAMs during a serial-register transfer cycle. The decode logic must then distinguish DRAM-refresh and serial-register transfer cycles from memory-access cycles. This is done by decoding the 4 LSBs of the address output on the LAD bus to determine the current bus status.

### 8.16.6 Display Memory Hardware Requirements

The minimum number of memory bits required to implement the display memory is the product of the total number of pixels (on-screen and off-screen areas combined) and the number of bits per pixel. The minimum number of VRAMs required to contain the display memory is calculated as follows:

```
number of VRAMS = (pixels per line) × (lines per frame) × (bits per pixel)
number of bits per VRAM
```

This calculation yields the minimum number of VRAMs needed; however, some applications may require additional VRAMs. For example, the TMS34020 supports XY addressing most efficiently when the number of pixels per line of the display memory is a power of 2. Achieving this may require more than the minimum number of VRAMs needed to contain the display.

# 8.17 Double-Buffered Display Example (2×1280×1024)

As an example, consider a display system that implements a 1280×1024 double-buffered screen at 8 bits per pixel, with an additional 4 Mbytes of system memory.

The TMS34020's 32-bit data path allows four 8-bit pixels to be accessed simultaneously. This also means that 4 pixels are clocked out of the VRAMs' serial registers simultaneously. If the display is refreshed 60 times per second, pixels must be displayed at a rate of one every 12.7 ns. The VRAMs' serial registers must therefore be clocked once every 50.8 ns (that is, at a frequency of 19.6 MHz). This is well within the operating range of VRAMs.

The VRAM address decoding scheme shown in Figure 8–23 provides for these configuration requirements. Three logical address bits (23, 24, and 25) are used as bank-select bits. Logical address bits 5—13 are used as the 9-bit column address, and bits 14—22 are used as the 9-bit row address. The base VRAM array size should be set (via RCM[CONFIG]) to 256K×n. Referring to Table 8–11 (page 8-53), the row and column addresses are multiplexed out on the same 9 pins, RCA1—RCA10. The total number of address bits used to address external memory is 26, for a total address reach of 8 megabytes. The remaining 6 address bits output by the TMS34020 are not used for this example, but could be used for more system memory or any peripherals required for the system.





- **BS0 & BS1** (bank-select bits 0 and 1) select between the 2 buffers in the display memory.
- BS2

(bank-select bit 2) selects between the display memory and the system memory. BS2=0 selects the display memory and BS2=1 selects the system memory. If the system memory is implemented with 1Mx1 DRAMs, the decode logic should assert CAMD=1 when BS2=1 to configure the RCA bus correctly. Figure 8–24 shows the address decode configuration for the system memory. Logical address bits 5—14 are used as the 10-bit column address, and bits 15—24 are used as the 10-bit row address. Referring to Table 8–11, the row and column addresses are multiplexed out over the same 10 pins, RCA2—RCA11.

### Figure 8–24. DRAM Address Decode for Example System



The amount of VRAM required to implement the display memory in this system depends on whether midline reload is used to pack the data for each scan line into contiguous locations within the VRAM. The serial register within each VRAM is 512 bits long. Between them, the 32 serial registers hold 2048 pixels at any one time. However, each scan line requires only 1280 pixels. If midline reload is not used, each scan line must begin with a new row of VRAM, and 768 pixels from each row of memory cannot be displayed.

### 8.17.1 Display Memory Implementation Using Midline Reload

Midline reload allows all 2048 pixels in each row of VRAM to be displayed, because the VRAM row being clocked out can change in the middle of a scan line. Sections 9.13 to 9.15 (pages 9-42—9-51) discuss midline reload in detail. It is enabled by setting SSV[[DPYCTL]] to a 1. For this example, DPYMSK should be loaded with FFh (to indicate no bank-selects at the least significant end of the address, and a VRAM half serial register of 256 bits).

Figure 8–25 shows the structure of the display memory when midline reload is used. Note that only 3 banks are required to implement both display buffers. The total amount of display memory required is 24 Mbytes. BS0 and BS1 select between the banks. Display buffer 1 is spread across banks 0 and 1; display buffer 2 is spread across banks 1 and 2.

Because each buffer is spread across 2 banks, the bank from which pixels are being shifted must change part way down the screen. The changeover between banks must occur during a horizontal blanking period, when the VRAMs are not clocking pixels out of their serial registers. This means that the last pixel in a bank must also be the last pixel on a scan line. Each bank contains enough pixels for 4/5 of one display buffer (819.2 lines). The address of the first pixel in each display buffer must be such that the last pixel in banks 0 and 1 is also the last pixel on a line.



Figure 8–25. Example Display Memory Dimensions (with Midline Reload)

### 8.17.2 Display Memory Implementation Without Midline Reload

When midline reload is not used, all the pixel data for any single scan line must be contained within a single row of VRAM. Each VRAM row contains 2048 pixels, but only 1280 of these are required for a scan line. This means that 40% of the display memory is not output to the display (unless you wish to be able to pan the display around within the display buffer).

Figure 8–26 shows the structure of the display memory when midline reload is not used. In this case, 4 banks are required to implement both display buffers because each display buffer requires 1 row of VRAM for each of its 1024 lines. The total amount of display memory required is 32 Mbytes. BS0 and BS1 select between the banks. BS1 selects between the display buffers.

Unfortunately, the 12.8 Mbytes of memory not used for the display is not addressable as a contiguous block. This makes it very difficult to use for program storage, etc.



Figure 8–26. Example Display Memory Dimensions (Without Midline Reload)

**Note:** Each display buffer can contain 2048 pixels per line, but only 1280 are output to the VRAMs. The undisplayed 2/5 of all rows constitutes a total of 12.8 Mbytes of undisplayed memory.

# **Chapter 9**

# **Video Timing and Screen Refresh**

The TMS34020's video interface provides these features:

- Separate or composite sync and blanking
- Synchronization to internal or external signals
- Interlaced or noninterlaced video
- A variety of screen resolutions

Additionally, the TMS34020 directly supports the use of VRAMs by generating the memory-to-register cycles needed to refresh a screen. The TMS34020 can also control video capture by performing register-to-memory (instead of memory-to-register) cycles.

This chapter includes the following topics:

	Sectio	n	Page
<b>Basic information</b> will help you use the rest of this chapter. Besides reviews of signals and registers, these Sections provide details of horizontal and vertical	9.1 9.2 9.3	Related Signals Related Registers Relationship Between Horizontal and Vertical Timing Signals	. 9-2 9-4 9-9
timing.	9.4 9.5	Vertical Video Timing (Internal)	9-11 9-13
<i>Special features</i> of the TMS34020's variety of video modes are covered in detail.	9.6 9.7 9.8 9.9 9.10	Composite Video Timing Noninterlaced Video Timing Interlaced Video Timing External Synchronization Modes Screen Sizes and Dot Rate	9-15 9-18 9-21 9-29 9-36
<b>Examples</b> elaborate on the video discussion.	9.11 9.12	Display Interrupts and Applications Video Timing Programming Examples	9-37 9-38
The chapter finishes with a discussion of <b>VRAM topics</b> .	9.13 9.14 9.15	Video RAM Control         Scheduling Screen-Refresh Cycles         Generating Screen-Refresh Addresses	9-42 9-50 9-51

# 9.1 Related Signals

The TMS34020's video timing logic is driven by an external video clock; the TMS34020 generates sync and blanking signals on chip. These signals control the horizontal and vertical sweep rates of the screen and synchronize the screen display to data output by the VRAMs. The video-timing and screen-re-fresh signals are summarized below for your convenience; for detailed descriptions, refer to Chapter 2. Note that these signals depend on information that is provided in the DPYCTL register.

Signals	Descript	tions	I/O					
CBLNK/ VBLNK	can be cal-blank	either a composite-blanking signal or a verti- king signal, depending on the value of CVD:						
	CVD=0	selects $\overline{\text{CBLNK}}$ , which is the composite-blanking signal that turns off a CRT's electron beam during both horizontal and vertical retrace intervals. You can also use $\overline{\text{CBLNK}}$ to control stopping and starting of the VRAM serial registers.	0					
	CVD=1	selects $\overline{VBLNK}$ , which is the vertical-blanking signal that turns off a CRT's electron beam during vertical retrace intervals. You can also use $\overline{VBLNK}$ with $\overline{HBLNK}$ to control stopping and starting of the VRAM serial registers.	0					
	Both signals are always outputs.							
<u>CSYNC</u> / HBLNK	can be either a composite-sync signal or a horizontal-blank- ing signal, depending on the value of CVD:							
	CVD=0	selects $\overline{\text{CSYNC}}$ , which is the composite-sync signal that controls external video circuitry. $\overline{\text{CSYNC}}$ can be an input or an output, depending on the value of CSD:						
		CSD=0 selects external composite sync; CSYNC is an input	I					
		CSD=1 CSYNC is an output	0					
	CVD=1	selects HBLNK, which is the horizontal-blanking signal that turns off a CRT's electron beam during horizontal retrace intervals. HBLNK is always an output, regardless of the value of CSD (however, to ensure correct operation, you should always set CSD to 1 when CVD=1).	0					
	HBLNK	is always an output, but CSYNC can be an input or						

an output.

Video Timing and Screen Refresh

Signals	Descriptions	I/O						
HSYNC	is the horizontal-sync signal that controls external video cir- cuitry. HSYNC can be an input or an output, depending on the value of HSD:							
	HSD=0 selects external horizontal sync; HSYNC is an in- put	I						
	HSD=1 HSYNC is an output	0						
SCLK	is the VRAM serial-register clock. When using midline reload, this input is used to track the VRAM tap point. SCLK should be low during horizontal and vertical blanking. If you are not using midline reload, SCLK should be low.							
VCLK	is the video clock signal that drives the TMS34020's internal video timing logic. VCLK is derived from the dot clock of the external video system.							
VSYNC	is the vertical-sync signal that controls external video circuit- ry. VSYNC can be an input or an output, depending on the value of VSD:							
	<i>VSD=0</i> selects external vertical sync; <u>VSYNC</u> is an input	I						
	VSD=1 VSYNC is an output	~						
		0						

Holding VCLK low for long periods may cause video counter errors. When VCLK is not being clocked for long periods, hold it at the logic-high level. While VCLK is low, the storage nodes within the device rely on their internal capacitance to maintain state information; if VCLK is held low for a sufficiently long time, charge leakage may cause bit errors.

### 9.2 Related Registers

The video timing and screen-refresh registers are a subset of the I/O registers described in Chapter 4. These registers are divided into 3 groups:

- ❑ Horizontal timing registers control the timing of the HSYNC signal, the HBLNK signal, and the horizontal components of the CSYNC and CBLNK composite signals.
- Vertical timing registers control the timing of the VSYNC signal, the VBLNK signal, and the vertical components of the CSYNC and CBLNK composite signals.
- Screen-refresh registers control the addresses generated during localmemory screen-refresh (memory-to-register) cycles.

HCOUNT register	address: C000 01D0h
HESYNC register	address: C000 0010h
HESERR register	address: C000 0270h
HEBLNK register	address: C000 0030h
HSBLNK register	address: C000 0050h
HTOTAL register	address: C000 0070h
SETHCNT register	address: C000 0310h
15	0

- **HCOUNT** tracks the number of VCLK periods that occur per horizontal scan line.
- **HESYNC** defines the point in a horizontal scan line where horizontal sync ends and the HSYNC signal is driven inactive.
- **HESERR** is used in composite mode only. It identifies the point in the horizontal scan line where serration ends. Serration may occur on the CSYNC pin during the vertical-sync portion of the display, and is similar in form to horizontal sync, though of different duration.
- **HEBLNK** defines the endpoint of the horizontal-blanking period; this is the point when the HBLNK signal is driven inactive.
- **HSBLNK** defines the startpoint of horizontal blanking, when HBLNK is driven active.
- **HTOTAL** defines the number of VCLK periods allowed per horizontal line. It also specifies when the horizontal-sync period begins and HSYNC is activated.
- **SETHCNT** is used in external video modes to reload HCOUNT when an external composite or horizontal sync begins. This allows you to start the horizontal count from an arbitrary value, counteracting synchronization delays and/or external signal skew.

VCOUNT register	address: C000 01C0h
VESYNC register	address: C000 0000h
VEBLNK register	address: C000 0020h
VSBLNK register	address: C000 0040h
VTOTAL register	address: C000 0060h
SETVCNT register	address: C000 0300h
15	0

- **VCOUNT** counts the horizontal scan lines in the display.
- **VESYNC** defines the number of horizontal scan lines where vertical sync ends and the VSYNC signal is driven inactive.
- **VEBLNK** defines the endpoint of the vertical-blanking period; this is the point when the VBLNK signal is driven inactive.
- VSBLNK defines the startpoint of vertical blanking, when VBLNK is driven active.
- **VTOTAL** defines the number of horizontal scan lines allowed for the entire display. It also specifies when the vertical-sync period begins and VSYNC is activated.
- **SETVCNT** is used in external noninterlaced video modes to reload VCOUNT when the first external composite serration pulse or vertical sync begins. This allows you to start the vertical count from an arbitrary value, counteracting external signal skew. For external interlaced video this register must be programmed to 0 so that the video timing logic can distinguish the external odd and even fields and synchronize accordingly.

DPYCTL register address: C000 0080h										
15	14	12	11	7	6		3	2	1	0
ENV	NIL	BRE SRE	CST		E SSV		CVD	CSD	VCD	HSD
Note: For a complete illustration and description of DPYCTL, see Chapter 4.										

The DPYCTL register does not affect display *timing*, but it contains several bits that control various display functions:

HSD bit 0 The HSD (horizontal-sync direction) bit determines whether  $\overline{\text{HSYNC}}$  is an input or an output:

- HSD=0 selects HSYNC as an input.
- HSD=1 selects HSYNC as an output.

vsp bit 1	The VSD (vertical-sync direction) bit determines whether VSYNC is an input or an output:
	VSD=0 selects VSYNC as an input.
б	VSD=1 selects $\overline{\text{VSYNC}}$ as an output.
CSD bit 2	The CSD (composite-sync direction) bit helps to determine whether CSYNC/HBLNK is an input or an output:CSD=0selects CSYNC/HBLNK as an output.CSD=1selects CSYNC/HBLNK as an input, provided that CVD=0 (if CVD=1, this pin is always an output).
CVD bit 3	The CVD (composite-video disable) bit determines the functions of CSYNC/HBLNK and CBLNK/VBLNK:CVD=0selects CSYNC and CBLNK.CVD=1disables composite video and selects HBLNK and VBLNK.
ssv bit 6	Setting the SSV (split-serial-register midline-reload enable) bit to 1 enables the TMS34020 to perform screen refreshes for VRAMs with split serial registers.
VCE bit 7	The VCE (video-capture enable) bit selects the type of screen-refresh memorycycle:VCE=0vCE=1selects memory-to-register screen-refresh cycles.VCE=1selects register-to-memory screen-refresh cycles.
CST bit 11	Setting the CST (CPU serial-register transfer) bit to 1 tells the TMS34020 to convert ordinary pixel accesses into VRAM serial-register transfers.
bit 12	Setting the SRE (screen-refresh enable) bit to 1 enables automatic screen re- freshes when video is enabled (ENV=1).
bit 14	The NIL (noninterlaced video) bit chooses interlaced or noninterlaced video.NIL=0enables interlaced video.NIL=1enables noninterlaced video.
ENV bit 15	The ENV (enable video) bit <b>must be 1</b> to allow the video timing logic to operate. When ENV=0, the TMS34020's blanking outputs are permanently at the active-low level, and all display control is disabled.
bit 13	The TMS34010 used bit 13 as the DXV (disable external video) bit. The TMS34020 does not need this bit; the TMS34020 relies entirely on HSD, VSD, and CSD to individually select which video timing signals are external. The TMS34020 ignores bit 13.
ORG bit 10	The TMS34010 used bit 10 as the ORG (origin) bit to determine whether the display origin was in the top left or bottom left corner of the screen, and there- fore whether to increment or decrement the screen-refresh address during hor- izontal blanking. The TMS34020 does not need this bit; the TMS34020 allows you to load SRINC [DINC] with the 2s complement of the display pitch if you wish to decrement the screen-refresh address. The TMS34020 ignores bit 10.

DPYST registers	addres	s: C000	) 0200h
31	5	4	0
SRST			
<b>Note:</b> For a complete illustration and description of DPYST see Chapter 4			

**DPYSTL** and **DPYSTH** are two 16-bit registers that work together to form a single 32-bit register. Throughout this chapter, this register pair is referred to as **DPYST**. DPYST contains the 27-bit SRST value, which represents the longword address of the first pixel to be displayed on the screen.

DPYNX registers	address: C000 0220h
31	5 4 0
SRNX	YZCNT
Note: For a complete illustration and description of DPYNX,	see Chapter 4.

**DPYNXL** and **DPYNXL** are two 16-bit registers that work together to form a single 32-bit register. Throughout this chapter, this register pair is referred to as **DPYNX**. The DPYNX registers contain two values.

SRNX bits 5----31

The 27-bit SRNX value represents the long-word address of the next-screen refresh. This is the address of the first pixel on the next scan line of the display.

VZCNT The 5-bit YZCNT value is used to determine whether or not SRNX can be bits 0—4 incremented between screen refreshes. This allows the image on the screen to be magnified (zoomed) in the Y direction by repeating scan lines multiple times.

DINC registers	addres	s: C000	) 0240h
31	5	4	0
SRINC		YZ	NC

**Note:** For a complete illustration and description of DINC, see Chapter 4.

**DINCL** and **DINCH** are two 16-bit registers that work together to form a single 32-bit register. Throughout this chapter, this register pair is referred to as **DINC**. The DINC registers contain two values.

SRINC bits 5-31 The 27-bit SRINC value is the amount by which the address in SRNX is incremented between screen refreshes; this value is equal to the display pitch.

YZINC bits 0—4 The 5-bit YZINC value is used to determine the number of screen refreshes that must occur between each increment of SRNX. YZINC is added to YZCNT

between every screen refresh, but SRNX is incremented only when YZCNT=0. This allows the screen image to be magnified (zoomed) in the Y direction by repeating scan lines multiple times.

DPYMSK register	address: C000 02E0h
15	0

**DPYMSK** identifies the portion of the screen-refresh address that represents the column or tap-point address for midline reload. DPYMSK's LSB maps to the LSB (bit 5) of SRST[[DPYST]] and SRNX[[DPYNX]]. Allowing for this 5-bit offset, DPYMSK must be loaded with a string of contiguous 1s at the position of the column address in the logical address.



The TMS34020's screen-refresh registers differ from those of the TMS34010, although the registers are similar in function.

- SRST[DPYST] is equivalent to the SRSTRT field of the TMS34010's DPYSTRT register.
- SRNX[DPYNX] is equivalent to DPYADR.
- SRINC[[DINC]] is equivalent to the DUDATE field in the TMS34010's DPYCTL register.
- The TMS34020 does not need the TAPPNT register.

The TMS34020 supports the DPYSTRT, DPYADR, and TAPPNT registers for compatibility purposes only. They are simple read/write registers; they have no additional functionality and have no effect on the TMS34020's screen-refresh mechanisms.

There is no equivalent of the LCSTRT field in the TMS34010's DPYSTRT register. Screen-refresh cycles are generated during the horizontal-blanking interval at the beginning of every unblanked line of the display.

# 9.3 Relationship Between Horizontal and Vertical Timing Signals

Figure 9–1 illustrates the relationship between the horizontal and vertical timing signals for constructing a 2-dimensional raster display pattern.

- The **horizontal** sync and blanking signals span a single horizontal scan line within the frame and are repeated for each line.
- The **vertical** sync and blanking signals span an entire frame (1 complete pass of the display).





Figure 9–2, a simplified version of Figure 9–1, illustrates several terms and phrases used throughout this chapter.







horizontal front is the interval between the beginning of horizontal blanking and the beginning of porch horizontal sync.

horizontal back

vertical back porch

porch



is the interval between the end of horizontal sync and the end of horizontal blanking. vertical front porch

is the interval between the beginning of vertical blanking and the beginning of vertical sync.

sync and the end of vertical blanking.

is the interval between the end of vertical

# 9.4 Horizontal Video Timing (Internal)

This discussion applies to video timing signals that the TMS34020 generates internally (it does not apply to external signals). Horizontal timing signals are the same for interlaced and noninterlaced video displays. Figure 9–3 shows how the HESYNC, HEBLNK, HSBLNK, and HTOTAL registers control horizontal signal timing.





- All horizontal timing parameters are multiples of VCLK, which are counted by HCOUNT.
- ❑ The time between the start of 2 successive HSYNC pulses is specified by HTOTAL. Because HCOUNT counts 0 as its first value, the value in HTOTAL represents 1 less than the number of VCLK periods per horizontal scan line. For example, to specify a horizontal interval that is some even number 2n of VCLK periods in duration, the HTOTAL register must be set to an odd value, 2n-1.
- Similarly, the value in HESYNC represents the duration of the HSYNC pulse, minus 1, and the values in HSBLNK and HEBLNK represent the startpoints and endpoints of the horizontal-blanking interval.

Figure 9–4 shows a simplified diagram of the internal logic that generates the horizontal timing signals. HCOUNT is incremented once every VCLK period (on the high-to-low transition) until it equals the value in HTOTAL. During the VCLK period following HCOUNT = HTOTAL, HCOUNT is reset to 0, and begins counting again.

The limits of the horizontal-sync pulse are defined by the values in HESYNC and HTOTAL. HSYNC is driven active low after HCOUNT = HTOTAL; it is then driven inactive high after HCOUNT = HESYNC. After HCOUNT becomes equal to HTOTAL or HESYNC, a 2-VCLK delay occurs before the appropriate transition takes place at the HSYNC pin.

The HBLNK signal is driven active low after HCOUNT = HSBLNK; it is then driven inactive high after HCOUNT = HEBLNK. After HCOUNT becomes

equal to HSBLNK or HEBLNK, there is a 2-VCLK delay before the appropriate transition takes place at the HBLNK or CBLNK pin (depending on the value of CSD[[DPYCTL]]).





When HCOUNT= HSBLNK (shortly before the end of the horizontal scan), horizontal blanking begins. At this time, the video timing logic automatically schedules a screen-refresh memory cycle, preparing for the next displayed line.

#### Note:

For interlaced video, HSBLNK, HTOTAL/2, and HESYNC/2 must be separated from each other by a time of at least 2 VCLK periods, plus 2 LCLK periods, in order that the correct screen-refresh requests are detected by the TMS34020's memory controller.



Video Timing and Screen Refresh

# 9.5 Vertical Video Timing (Internal)

This discussion applies to video timing signals that the TMS34020 generates internally (it does not apply to external signals). Some additional vertical timing characteristics that are unique to interlaced or noninterlaced video are described in Sections 9.7 (<u>Noninterlaced Video Timing</u>, page 9-18) and 9.8 (<u>Interlaced Video Timing</u>, page 9-21).

Figure 9–6 shows how the VESYNC, VEBLNK, VSBLNK, and VTOTAL registers control vertical signal timing.

Figure 9–6. Vertical Timing for Noninterlaced Display



For most of the display, VCOUNT increments once every horizontal scan line (when HCOUNT = HTOTAL). For interlaced displays, it also increments when HCOUNT = HTOTAL/2 during the equalization and serration regions of vertical blanking because these pulses occur twice per line. For this reason, the values programmed into the vertical video timing registers do not necessarily have a 1-to-1 relationship with the number of horizontal scan lines. Sections 9.7 and 9.8 discuss programming these registers.

VTOTAL defines the time between the start of two successive  $\overline{VSYNC}$  pulses. VESYNC represents the duration of the  $\overline{VSYNC}$  pulse.  $\overline{VSBLNK}$  and  $\overline{VEBLNK}$  represent the startpoints and endpoints of the vertical-blanking interval.

Figure 9–7 shows a simplified schematic of the internal logic that generates the vertical timing signals. VCOUNT is incremented once every scan line (at the beginning of each  $\overrightarrow{\text{HSYNC}}$  pulse, when HCOUNT = HTOTAL) and also at the midpoint of each scan line (when HCOUNT = HTOTAL/2) during the equalization and serration portions of an interlaced display, until it equals the value in VTOTAL. After VCOUNT = VTOTAL, VCOUNT is reset to 0, and begins counting again.

The limits of the vertical-sync pulse are defined by the values in VESYNC and VTOTAL.  $\overrightarrow{VSYNC}$  is driven active low after VCOUNT = VTOTAL; it is then driven inactive high after VCOUNT = VESYNC. Transitions on the  $\overrightarrow{VSYNC}$  pin coincide with transitions on the  $\overrightarrow{HSYNC}$  or  $\overrightarrow{CSYNC}$  pins.

The  $\overline{VBLNK}$  signal is driven active low after VCOUNT = VSBLNK; it is then driven inactive high after VCOUNT = VEBLNK. Transitions on the  $\overline{CBLNK}$ /  $\overline{VBLNK}$  pin coincide with transitions on the  $\overline{HSYNC}$  or  $\overline{CSYNC}$  pins.





# 9.6 Composite Video Timing

This discussion applies to video timing signals that the TMS34020 generates internally (it does not apply to external signals).

Composite video signals combine the horizontal and vertical video timing into a single blanking signal, CBLNK, and a single sync signal, CSYNC.

- CBLNK is simply the logical-OR (negative logic) of the HBLNK and VBLNK signals. Thus, when either HBLNK or VBLNK is active low, CBLNK is active low.
- CSYNC is essentially the same as the HSYNC signal, except during portions of the vertical-blanking interval. During these portions, special sync pulses (known as *serration* and *equalization* pulses) are generated. These pulses allow a monitor to detect the vertical-sync interval, while at the same time ensuring that it remains in horizontal sync.

Some video monitors are designed to use composite rather than separate sync and blanking signals. Video systems used in television and broadcasting use composite video signals.

### 9.6.1 Theory Behind Serration and Equalization Pulses

Monitors pass the composite-sync signal through integrating (low pass) filters to extract the vertical-sync information from it. Serration pulses occur during the vertical-sync interval and are used by the monitor to determine when the vertical retrace occurs.

In noninterlaced video mode, serration pulses begin at the end of each scan line at the same time as the horizontal-sync pulses. In interlaced video mode, they also begin at the midpoint of each horizontal scan line. This is because in the even field, the vertical-sync interval begins and ends midway through a scan line. If the pulses occurred only once per scan line, the vertical-sync pulse extracted by the monitor would not start or end at the right time in the even field.

In interlaced video, equalization pulses occur in the regions immediately before and after the vertical-sync interval. They also begin at the end and the midpoint of each scan line, and because of this each pulse is only half the width of a regular horizontal-sync pulse. They are necessary to ensure that the monitor extracts a vertical-sync interval of exactly the same width in both the even and odd fields, because the vertical-sync interval begins at the midpoint of a scan line in the even field.

Because vertical sync always begins and ends at the end of a scan line in noninterlaced video mode, equalization pulses are not required. Figure 9–8 shows the regions of vertical blanking in which the serration and equalization pulses occur. Each region contains a number of scan lines specified by the vertical video timing registers. Outside these regions,  $\overrightarrow{\text{CSYNC}}$  is identical to  $\overrightarrow{\text{HSYNC}}$ .

Figure 9–8. Regions of Vertical Blanking Where Equalization and Serration Pulses Occur on CSYNC



- ① The first equalization region coincides with the vertical front porch. In broadcast-quality composite-video standards such as NTSC and PAL, this region has the same duration as the vertical-sync pulse. However, you can reprogram the duration by varying the duration of the vertical front porch.
- ② The **serration region** coincides with vertical sync and immediately follows the first equalization region.
- ③ The **second equalization region** immediately follows the serration region. Hardware defines this region to have duration equal to that of vertical sync.

### 9.6.2 Serration Pulses on CSYNC

Serration pulses are produced during the serration region (see Figure 9–8). Serration pulses begin at the end of each horizontal scan line (coinciding with the beginning of horizontal sync) after the condition HCOUNT = HTOTAL is reached. If the display is interlaced, they also begin at the midpoint of each scan line, after the condition HCOUNT = HTOTAL/2 is reached. The duration of these pulses is determined by the value of the HESERR register. This value is 1 less than the number of VCLK periods in the serration pulse, in the same way that the value in HESYNC is 1 less than the number of VCLK periods in the number of VCLK periods in the horizontal-sync pulse.

Serration pulses are generally longer in duration than regular horizontal-sync pulses. This is shown in Figure 9–9. Broadcast-quality composite-video standards such as NTSC and PAL require that serration pulses are of such a duration that CSYNC is inactive high for a period equal to the active-low time of HSYNC. This can be achieved by programming HESERR to HTOTAL minus the number of VCLK periods in the horizontal-sync pulse.





### 9.6.3 Equalization Pulses on CSYNC

If the display is interlaced, equalization pulses are produced during the equalization regions (see Figure 9–8, page 9-16). These pulses begin at both the end and the midpoint of each horizontal scan line, after HCOUNT = HTOTAL, and after HCOUNT = HTOTAL/2. Each pulse ends after HCOUNT = HE-SYNC/2. The beginning of every other pulse coincides with horizontal sync (see Figure 9–10).





Broadcast-quality composite-video standards such as NTSC and PAL require that equalization pulses are exactly half the duration of horizontal-sync pulses. To achieve this, set HESYNC to an odd value. HCOUNT starts counting from 0, so an odd value in HESYNC means that horizontal sync is an even number of VCLK cycles in length, and therefore exactly divisible by 2. Similarly, set HTOTAL to an odd value so that HTOTAL/2 is exactly the midpoint of each scan line.

If the display is noninterlaced, no special pulses are produced during the equalization regions. The CSYNC output continues to appear like HSYNC.

# 9.7 Noninterlaced Video Timing

You can select noninterlaced scan mode by setting NIL [DPYCTL] to 1.



### 9.7.1 Activity in Noninterlaced Mode

In noninterlaced mode, each video frame consists of a single vertical field. Figure 9–11 shows the path traced by the electron beam on the screen.

Figure 9–11. Electron Beam Pattern for Noninterlaced Video



**Monitor Screen** 

- (a) shows the vertical retrace, which is an integral number of horizontal scan lines in duration.
- (b) shows the active portion of the frame. Solid lines represent lines that are displayed; dashed lines are blanked. Each line on the display is separated by the horizontal sweep time, *H*, where

*H* = (*HTOTAL* + 1) x (*VCLK* period)

Figure 9–12 illustrates the video timing signals that generate the display shown in Figure 9–11. The display line numbers and regions from Figure 9–11 are shown for reference. For completeness, all 6 possible output signals are shown, although you can choose only 4 in either of the 2 allowed combinations:

Combina	tion 1 (CVD=0):	Combination 2 (CVD=1):	
HSYN	ō	HSYNC	
VSYN	C	VSYNC	
CSYN	C	HBLNK	
CBLN	K	VBLNK	

The example in Figure 9–12 uses the following register values:

VSBLNK = 8	VTOTAL = 9
VESYNC = 3	VEBLNK = 2

Actual applications use much larger values; these values are for illustration purposes only. Note that the VESYNC is not selected as you might expect (you would expect it to be 1). Section 9.7.2 describes the reason for this, along with formal equations for selecting proper values for all the vertical timing registers.

Each horizontal scan line is preceded by a horizontal retrace, triggered by the horizontal-sync pulse. The horizontal scan pattern repeats until VCOUNT = VTOTAL. Then, VCOUNT is reset to 0, and the vertical retrace (triggered by vertical-sync pulse) returns the beam to the top of the screen.

VCOUNT is incremented each time HCOUNT is reset to 0 (HCOUNT is reset to 0 at the end of each scan line). The  $\overrightarrow{VSYNC}$  output goes active when VCOUNT = VTOTAL, coinciding with the beginning of the horizontal-sync pulse. The  $\overrightarrow{VSYNC}$  signal goes inactive after VCOUNT = VESYNC; this transition also coincides with the beginning of the horizontal-sync pulse.

Horizontal blanking is typically active during vertical-blanking transitions, which coincide with the leading edges of horizontal-sync pulses.





 $(\overline{b})$  Active portion of the frame

### 9.7.2 Programming the Vertical Timing Registers for Noninterlaced Video

For noninterlaced video, VCOUNT increments only after HCOUNT = HTOTAL, because there are no equalization or serration pulses with a period of half a scan line. Thus, with one exception, all the vertical timing registers are programmed in terms of the number of integral horizontal scan lines. VTOTAL is 1 less than the number of lines in the frame because VCOUNT counts 0 as its first value. VSBLNK and VEBLNK contain the line number, minus 1, where vertical blanking starts and ends, respectively. For reasons associated with composite interlaced video, the VESYNC register detects the end of vertical sync at **half** the value it is programmed to, and so should be programmed to **twice** the number of lines in the vertical-sync interval, minus 1. Figure 9–13 summarizes this.

### Figure 9–13. Programming the Video Timing Registers for Noninterlaced Video

<ul> <li>(the number of VCLKs in horizontal sync) – 1</li> </ul>
<ul> <li>(the number of VCLKs from the start of horizontal sync to the end of horizontal blanking) - 1</li> </ul>
<ul> <li>(the number of VCLKs from the start of horizontal sync to the start of horizontal blanking) – 1</li> </ul>
= (the number of VCLKs in the line) – 1
= (the number of VCLKs in horizontal serration) – 1
= (twice the number of lines in vertical sync) - 1
<ul> <li>(the number of lines from the start of vertical sync to the end of vertical blanking) - 1</li> </ul>
<ul> <li>(the number of lines from the start of vertical sync to the start of vertical blanking – 1)</li> </ul>
= (the number of lines in the frame) $-1$

Note: You don't have to program HESERR if you're not using composite video.

# 9.8 Interlaced Video Timing

You can select interlaced scan mode by clearing NIL [[DPYCTL]] to 0.



In interlaced mode, each video frame consists of two vertical fields of horizontal scan lines. The display consists of alternate lines from the two fields. This doubles the display resolution while only slightly increasing the frequency with which data is supplied to the screen.

The TMS34020 can produce interlaced video signals compatible with NTSC (RS-170), PAL, SECAM, and similar broadcast-quality video standards. The TMS34020 also supports higher resolution standards such as RS-330 and RS-343.

Note:

For simplicity, the illustrations in this section show American (NTSC) waveforms. European waveforms differ slightly; Section 9.8.3 (page 9-27) describes these differences.

### 9.8.1 Activity in Interlaced Mode

Figure 9–14 shows the path traced by the electron beam on the screen for a typical interlaced-video display. In interlaced mode, 2 separate vertical scans are performed for each frame—1 for the even lines (even field) and 1 for the odd lines (odd field). The even field is scanned first.

- (a) shows the vertical retrace at the beginning of the even field, which coincides with the vertical-sync pulse. In this example, the vertical retrace is an integral number of horizontal scan lines in duration but, in fact, can be programmed to any number of *half* horizontal scan lines.
- (b) shows the active portion of the even field. Solid lines represent displayed lines; dashed lines are blanked.
- (c) & (d) show the vertical retrace and the active portion of the display, respectively, for the odd field.
- (e) shows how the 2 fields form the complete display. Note that active scan lines 0—7 are partially blanked.



### Figure 9–14. Electron Beam Pattern for a Typical Interlaced Display

Figure 9–15 illustrates the video timing signals that generate a display similar to the one shown in Figure 9–14. However, Figure 9–15 has more blanked lines, to properly illustrate the composite-sync signal. The display line numbers and regions from Figure 9–14 are shown for reference. For completeness, all 6 possible output signals are shown, although you can choose only 4 in either of the 2 allowed combinations:

Combination 1 (CVD=0):	Combination 2 (CVD=1):
HSYNC	HSYNC
VSYNC	VSYNC
CSYNC	HBLNK
CBLNK	VBLNK

The example in Figure 9–15 uses the following register values:

VSBLNK = 13	VTOTAL = 17
VESYNC = 7	VEBLNK = 9

VCOUNT increments every half horizontal scan line for parts of the display; you must allow for this when programming the vertical timing registers. The formal programming equations and their derivations are given in Section 9.8.2.





**Note:** This figure is continued on the next page.

- Vertical blanking begins at the end of a horizontal scan line near the end of the even field, and in the center of a horizontal scan line (after HCOUNT = HTOTAL/2) near the end of the odd field.
- Similarly, vertical blanking **ends** in the center of a horizontal scan line (after HCOUNT = HTOTAL/2) at the beginning of the even field, and ends at the end of a horizontal scan line at the beginning of the odd field.

The vertical-blanking interval is of the same duration in both the odd and even fields. In this way, the beam is positioned so that the horizontal scan lines in the odd field fall between the horizontal scan lines of the even field. Because each field contains an odd half line (not an integral number of lines), the total number of lines in the entire frame must be **odd**.



*Figure 9–15.* Interlaced Video Timing Waveform Example (Continued)

In Figure 9–15, VCOUNT is incremented twice per horizontal scan line during serration and equalization periods. This is the frequency at which equalization and serration pulses occur. Because VCOUNT increments every half line between the beginning of the first equalization period to the end of the second equalization period, you can specify the number of equalization and serration pulses required without being restricted to an integral number of lines.

Outside of the equalization and serration regions in Figure 9–15, VCOUNT increments only once per horizontal scan line, except there is an extra increment at the end of vertical blanking in the even field. This is necessary to ensure that the two fields are of exactly the same duration.

### 9.8.2 Programming the Vertical Timing Registers for Interlaced Video

As Figure 9–15 shows, VCOUNT is incremented twice per scan line during serration and equalization regions, and once per scan line outside of these regions. VTOTAL is not merely the total number of scan lines minus one, as it is for noninterlaced video. VTOTAL must account for those regions where VCOUNT is incremented twice per line. This applies to all interlaced video, independent of the value of CVD[[DPYCTL]]. VCOUNT always increments on half horizontal scan lines during equalization and serration regions of

interlaced displays, even if these pulses are not visible at the pins. For this reason, you must comply with the programming procedure and equations (described in this section and shown in Figure 9–17) for all interlaced displays.

To program the vertical video timing registers for interlaced video, divide vertical blanking into two parts for calculation purposes:

- **VFP** *Vertical front porch.* This is also the first equalization region (the region between the beginning of vertical blanking and the beginning of vertical sync).
- VSS Vertical sync to setup. This is the region between the beginning of vertical sync and the end of vertical blanking. Note that this can also be thought of as comprising the serration region, the second equalization region plus the remainder of vertical blanking, or the vertical-sync region plus the vertical back porch.

Figure 9–8 (page 9-16) first described these regions; Figure 9–16 shows them again, illustrating the two compound regions used for calculation.

Figure 9–16. The Two Regions of Vertical Blanking Used for Programming Calculations



Key: VFP: First equalization region VSS: Vertical sync to setup

Both VFP and VSS should be determined as a number of horizontal scan lines. The number of horizontal scan lines in vertical sync is also used in the calculations. If any one of these values is not an integral number of lines, they should **not** be rounded down. Rounding down should take place only at the end of each calculation.

After determining the duration of both VFP and VSS, you can consider the values that you must program into the registers:

VESYNC detects the end of vertical sync at half its stored value. This is because the second equalization region is of equal duration to vertical sync, and the end of this region is detected at the full value of VESYNC. In addition, because VCOUNT increments every half horizontal scan line during equalization and serration, VESYNC should be programmed to 4 times the number of lines in vertical sync, less 1 because VCOUNT starts from 0:

VESYNC = (4 x the number of lines in vertical sync) – 1

❑ VTOTAL contains the maximum VCOUNT value achieved for every field (odd and even). Because the counter increments every half line during equalization and serration, this is not simply the number of lines in the field, as is the case with noninterlaced video. Allowance must be made for the extra increments. VTOTAL is therefore 1/2 the total number of lines in the entire frame, plus 1 for the odd half line at the beginning or end of the field, plus the number of lines throughout which VCOUNT is incrementing every half line (to account for the half line increments), minus 1 because VCOUNT starts from 0. The plus 1 and the minus 1 cancel, and VCOUNT increments every half scan line during VFP, vertical sync, and the second equalization region (equal in duration to vertical sync). This reduces to:

VTOTAL = (the number of lines in the frame / 2) + (2 x the number of lines in vertical sync) + VFP

❑ VEBLNK schedules the end of vertical blanking, which occurs VSS scan lines after VCOUNT is set to 0 at the beginning of vertical sync. However, VCOUNT increments every half scan line during vertical sync and the second equalization region (equal in duration to vertical sync). Allowing for the fact that VCOUNT starts from 0, this gives rise to the following expression:

VEBLNK = VSS + (2 x the number of lines in vertical sync) – 1

❑ VSBLNK schedules the start of vertical blanking, which occurs VFP scan lines before the start of vertical sync (when VCOUNT = VTOTAL). Because VCOUNT increments twice every scan line during VFP, the value of VSBLNK is:

$$VSBLNK = VTOTAL - (VFP \times 2)$$

In this case, there is no need to subtract 1 to account for VCOUNT starting from 0, because this is already accounted for in determining VTOTAL.

Figure 9–17. Programming the Video Timing Registers for Interlaced Video

HESYNC	= (the number of VCLKs in horizontal sync) - 1
HEBLNK	<ul> <li>(the number of VCLKs from the start of horizontal sync to the end of horizontal blanking) – 1</li> </ul>
HSBLNK	<ul> <li>(the number of VCLKs from the start of horizontal sync to the start of horizontal blanking) – 1</li> </ul>
HTOTAL	= (the number of VCLKs in the line) – 1
HESERR	= (the number of VCLKs in horizontal serration) – 1
VESYNC	= 4 x (he number of lines in vertical sync) – 1
VTOTAL	<ul> <li>(the number of lines in the frame) / 2 + (twice the number of lines in vertical sync) + VFP</li> </ul>
VEBLNK	= VSS + (twice the number of lines in vertical sync) - 1
VSBLNK	= VTOTAL – (VFP x 2)

Note: You don't have to program HESERR if you're not using composite video.

### 9.8.3 American and European Video Standards

- RS-170 is the NTSC broadcasting standard used for all *American* television.
- **PAL** is the *British* standard.
- SECAM is the *French* standard.

The programming procedures and equations described in Section 9.8.2 are the same for these and similar standards. However, as far as the production of sync and blanking signals are concerned, there is a fundamental difference between the American and European standards. For the American standard, the vertical retrace (specified by vertical sync) is an integral number of horizontal scan lines. For the European standards, however, the vertical retrace is not a whole number of lines; it contains an odd half line. As a result, the composite-sync waveform is different for American and European standards at the end of the second equalization region.

The diagrams and discussions in this chapter apply to the American format. However, the TMS34020 automatically adjusts to the European format by detecting whether or not VESYNC is programmed to indicate an even or odd number of half lines in vertical sync (if vertical sync is not an integral number of horizontal scan lines, it contains an odd number of half lines). Figure 9–18 shows how the composite-sync waveforms differ for the two types of standard.






# (b) PAL



# 9.9 External Synchronization Modes

External synchronization modes enable the TMS34020 to use horizontal-, vertical-, and composite-sync signals from an external source. This allows you to superimpose or mix TMS34020-generated graphics with images from external sources.

Depending on the video pin configuration (separate or composite sync, selectable via CVD[[DPYCTL]]), either 2 or 3 of the 4 video pins are configured as sync pins. External synchronization mode is selected by configuring the appropriate sync pins as inputs by way of the CSD, VSD, or HSD bits in DPYCTL. Each pin is configured as an input when its associated bit is a 0. In separate-sync mode, the CSD bit must be a 1 because the CSYNC/HBLNK pin is configured as horizontal blanking and, as such, is an output only.

When all the sync pins are configured as outputs, the horizontal and vertical counters are cleared to 0s after HCOUNT = HTOTAL and after VCOUNT = VTOTAL, respectively. This also initiates the corresponding sync pulse. However, when one or more of the sync pins is configured as an input, this behavior is altered somewhat; a high-to-low transition on an input sync waveform (the beginning of an external sync pulse) causes one or both of the video counters to be loaded from SETHCNT and SETVCNT. By loading the video counters with a programmable value instead of 0, you can compensate for delays inherent in the process of synchronizing the external sync signals to VCLK within the TMS34020, for the time required for the TMS34020 to respond to the external sync signals, and also for external signal skews. HCOUNT and VCOUNT are reloaded as follows:

- HCOUNT is reloaded by a falling edge on either the HSYNC or the CSYNC pin. In noninterlaced video mode, a falling edge on VSYNC also reloads the horizontal counter.
- VCOUNT is reloaded by a falling edge on the VSYNC pin or the first serration pulse on the CSYNC pin (which occurs at the beginning of vertical sync). For this reason, the serration pulse width must be at least 2 VCLK periods greater than the internal horizontal-sync pulse width. In programming terms, this means HESERR ≥ HESYNC + 2. If this is not the case, the TMS34020 will not be able to detect the beginning of vertical sync from the composite input waveform. The HESERR and HESYNC registers **must** be programmed to accurately match the parameters of the external video source.

By causing VCOUNT and HCOUNT to follow the external synchronization signals in this way, the blanking intervals and screen-refresh cycles are also forced to follow the external video source, and the TMS34020 is therefore synchronized to the external video.

While the sync pins are independently configurable as inputs or outputs, not all of the possible combinations of inputs are useful. Typically, you would use one of the following 3 combinations of inputs:

## **External vertical-sync signal only.**

VSYNC: input	HSYNC: output	CSYNC/HBLNK: output
	Used with noninterlac	ced video

HCOUNT and VCOUNT are loaded simultaneously at the beginning of the external vertical-sync pulse, thus completely synchronizing the TMS34020 to the external source. Because HCOUNT is synchronized only once per frame, it is important that it be programmed to the correct value for the external source. For interlaced video, vertical sync coincides with a horizontal-sync pulse in every other frame only. Also, vertical sync alone is not sufficient to determine the field parity (odd/even) of the external video source, and therefore is insufficient to determine the field in which the horizontal and vertical sync coincide. For these reasons, external vertical sync has no effect on the horizontal timing registers in interlaced video.

## External vertical- and horizontal-sync signals.

VSYNC: input	HSYNC: input	CSYNC/HBLNK: output
Used in I	ooth composite- and se interlaced or noninterl	parate-sync modes, aced video

The beginning of the external vertical-sync signal reloads VCOUNT; the beginning of the external horizontal-sync signal reloads HCOUNT.

#### **External composite-sync signal only.**

VSYNC: output	HSYNC: output	CSYNC: input
	Used with interlaced	video only

The TMS34020 must be configured in composite video mode (CVD=0) so that the CSYNC/HBLNK pin is selected as CSYNC. The beginning of each external composite-sync signal reloads HCOUNT, and the beginning of the first serration pulse reloads VCOUNT. For noninterlaced video, the TMS34020 does not recognize equalization or serration pulses, and so cannot detect the beginning of vertical sync from a composite signal. For this reason, external composite sync has no effect on the vertical timing registers in noninterlaced mode.

To avoid any potential signal conflicts, all three sync pins are configured as inputs when the TMS34020 is reset. However, sync pins not actually being driven by external signals should subsequently be configured as outputs.

# 9.9.1 Odd and Even Field Alignment in Interlaced Mode

In interlaced mode, the TMS34020 synchronizes with the same field parity as the external source, *provided that* VTOTAL is programmed to exactly match the external source. This is necessary because of the way that the video timing logic adjusts to the correct field.

If the internal video timing has field parity opposite to the external source, the horizontal-sync pulse generated by the internal video timing logic is initially displaced from those of the external system by half a scan line. However, as soon as an external horizontal-sync pulse occurs, HCOUNT is reset, and VCOUNT is incremented. This realigns the internally generated horizontal-sync pulses with those of the external system. Because this occurs while the internal video timing logic is midway through a scan line (where VCOUNT would not normally be incremented), it causes the internal field to be shortened by half a scan line. The TMS34020 takes advantage of this at the end of the field to ensure that it becomes aligned to the same field as the external source.

After reaching VTOTAL, the internal video timing logic starts a new field with the opposite field parity. However, because the previous field was shortened by a half a scan line, the external vertical sync does not occur for another half scan line. When it does occur, it causes the video logic to resynchronize to the external source and start yet another new field, and the field parity changes again. Field parity changes twice, whereas the external source changes field parity only once. Thus, the internal video timing logic has the same field parity as the external source due to the extra (half scan line long) field.

If the internal and external vertical-sync pulses are separated by less than the duration of the horizontal-sync pulse (if CVD=1) or the composite-serration pulse (if CVD=0), the internal field parity does not change twice. This allows for normal differences between when VCOUNT = VTOTAL and detection of the beginning of the external vertical-sync interval. This will be the case when the internal and external field parities match correctly, *provided that* VTOTAL matches the external source. If this is not the case, one of the following situations applies.

- □ If VTOTAL gives an internal field duration less than that of the external source, the internal video timing logic changes field parity twice per field: once when VCOUNT = VTOTAL, and once at the beginning of the external vertical-sync interval.
- If VTOTAL gives an internal field duration greater than that of the external source, the internal video timing logic changes field parity only once per field: At the beginning of the external vertical-sync interval, even if the internal video timing logic does not have the same field parity as the external source.

# 9.9.2 Synchronizing External Syncs to VCLK

The TMS34020 synchronizes input signals to VCLK before passing them to the internal video logic. This means that inputs can be asynchronous to VCLK. The delay from the high-to-low transition of an external sync input to an occurrence of transition-induced changes at the video output pins will be from 4 to 5 VCLK periods, depending on the phase relationship between the transition and VCLK. If you do not require the TMS34020 to follow the external video source to an accuracy of 1 VCLK period (if, for example, the TMS34020 uses the external synchronization simply to perform screen refreshes or display interrupts), asynchronous operation may be acceptable.

If you use any of the video output signals (either blanking or sync) to control any other part of the system, then the TMS34020 may need to synchronize precisely to the external video source. In this case, you can present the external input sync signals synchronous to VCLK. If the setup and hold times for a valid level on one of the external video input pins with respect to the low-to-high transition of VCLK (described in the *TMS34020 Data Sheet*, Appendix A) are met, then the timing relationship between input signals and output signals is a constant, integral number of VCLK cycles. However, the exact number of cycles will appear to vary according to the value programmed in the SETHCNT register.

# 9.9.3 Loading the Video Counters

When the external video input signals are presented synchronous to VCLK, there is a 4-VCLK delay between the VCLK cycle when a falling edge is sampled on an external sync pin, and when the changes induced by this edge are visible at the video output pins. If you require the TMS34020 to accurately follow the external source, the effect of this delay can be eliminated by using the SETHCNT register.

SETHCNT is loaded into HCOUNT at the beginning of the fifth VCLK period after the appropriate external sync signal is detected at the input pin. This is shown in Figure 9–19 (a). By programming SETHCNT to 4, the TMS34020's video timing registers are aligned exactly to the external source. Loading HCOUNT to four 4-VCLK cycles, after the transition on the external sync pin, is equivalent to loading HCOUNT to 0 as soon as the transition on the external sync pin is detected.

Once the internal video timing logic is synchronized to the external system (by the first high-to-low transition of an external sync pin detected by the TMS34020), the video timing outputs are aligned with the external system on subsequent scan lines, provided that HCOUNT is programmed to match the external system. This is shown in Figure 9–19 (*b*); the internal HTOTAL occurs simultaneously with the end of the external system's scan line, causing HCOUNT to be reset to 0, activating the appropriate output sync signals at the same time as the input sync from the external system.







- **Note:** *S* is the value contained in SETHCNT. 4-VCLK cycles after the falling edge on the input SYNC pin occurs, HCOUNT is loaded with *S*, and the transition on the output SYNC is caused to occur.
- (b) Subsequent synchronization to external sync



**Note:** If HTOTAL matches the external system and *S*=4, HTOTAL causes the output sync pin to transition in the same VCLK cycle as the input sync.

Programming SETHCNT to values greater than 4 causes the TMS34020's video outputs to be in advance of the external video source. This can be useful for eliminating additional signal skews in the system.

There is a similar register for loading the vertical counter, SETVCNT. This must be loaded with **0** for interlaced video mode; otherwise, the TMS34020 will not be able to synchronize to the external field parity, for the reasons discussed in Section 9.9.1. If desired, you may load it with nonzero values in noninterlaced video mode; this displaces the internal video from the external video by a number of scan lines.

# 9.9.4 Synchronization Conversion

When you load SETHCNT with 4, the TMS34020's video output signals are synchronized with the input signals, provided that the input signals meet the setup and hold times for synchronous operation (see the *TMS34020 Data Sheet*). Because of this feature, the TMS34020 can be used to convert one form of synchronization waveforms into another.

- □ If the VSYNC and HSYNC pins are selected as inputs and the video timing registers accurately match the parameters of the external source, the CSYNC output is an exactly synchronized composite waveform with (in interlaced mode) the correct equalization and serration pulses.
- If the CSYNC pin is selected as an input and the video timing registers accurately match the parameters of the external source, the HSYNC and VSYNC outputs are exactly synchronized separate waveforms equivalent to the input.

# 9.9.5 Programming Flexibility and Limitations

For asynchronous operation in which the TMS34020 is not required to exactly follow the external source, you can program HTOTAL to a very large value (such as FFFFh). This prevents the condition HCOUNT = HTOTAL from occurring, and HCOUNT is reset only when a high-to-low transition on an external sync occurs. When using external synchronization modes there are, depending on the application, a number of different approaches that can be taken in programming the video timing registers.

For noninterlaced video, the same is also true for VTOTAL; however, for interlaced video, VTOTAL **must** be programmed to the correct value for the TMS34020 to be able to resolve the field parity of the external source.

Anytime SETHCNT is not 0 (typically for synchronous operation), HTOTAL should be loaded to match the external source. If this is not adhered to, then the TMS34020 will not be able to start horizontal- or composite-sync pulses coincident with the external-sync pulse, because the condition HCOUNT = HTOTAL will not occur. If you are not concerned about starting horizontal- or composite-sync pulses exactly aligned, but still wish to use SETHCNT, HTOTAL can be programmed to a large value as detailed above. If you take this approach, there are some pitfalls to avoid: Because HCOUNT is reloaded with the value in SETHCNT but HTOTAL is never reached, the counter never counts through those values from 0 up to SETHCNT – 1. You must ensure that the values of HESYNC, HESYNC/2, HESERR, HSBLNK, HEBLNK, or HTOTAL/2 do not coincide with any of these values; if they do coincide, the output waveforms could be corrupted. Again, for interlaced video, VTOTAL must be accurately programmed.

To allow the TMS34020 to detect the beginning of vertical sync from a composite-sync input, HESERR must be equal to at least HESYNC + 2. The TMS34020 samples the state of the composite input waveform when HCOUNT = HESYNC to determine whether serration has begun; if this is not adhered to, the TMS34020 may mistake a regular horizontal-sync pulse for a serration pulse.



For the TMS34010, it was important that HTOTAL and VTOTAL contained values large enough not to cause HCOUNT and VCOUNT to be cleared before the leading edges of the external sync pulses could clear them. This is not the case for the TMS34020. Instead, it is more important to program the registers accurately; it is essential to program VTOTAL correctly for interlaced field parity alignment, and HTOTAL correctly for synchronization conversion.

# 9.9.6 External Synchronization Pulse Widths

The external synchronization pulses input to the TMS34020 should have the following dimensions:

- Minimum width of HSYNC and VSYNC: Not less than 1 VCLK cycle.
- Minimum width of CSYNC: Same as for HSYNC/VSYNC except the serration pulses input during the vertical-sync portion of the signal must be at least 2 VCLK periods longer than the ordinary horizontal-sync pulse.
- Maximum width, all pins: Not greater than the internally generated pulse (HESYNC + 1, etc.).
- As for internal interlaced video, HSBLNK, HTOTAL/2, and HESYNC/2 must be separated from each other by a time of at least 2 VCLK periods plus 2 LCLK periods, ensuring that the memory controller can detect the correct requests.

## 9.10 Screen Sizes and Dot Rate

The TMS34020's 512-Mbyte address reach supports very high-resolution displays. For example, a large TMS34020-based system could use the lower half of the address space for display memory and use the upper half for storing programs and data. The 256-Mbyte display memory in this example could support the following display sizes:

- **a** 8,192 by 8,192 pixels at 32 bits per pixel
- □ 16,384 by 8,192 pixels at 16 bits per pixel
- □ 16,384 by 16,384 pixels at 8 bits per pixel
- 32,768 by 16,384 pixels at 4 bits per pixel
- 32,768 by 32,768 pixels at 2 bits per pixel
- G5,536 by 32,768 pixels at 1 bits per pixel

At most, all of the TMS34020's memory space below that allocated to the I/O registers could be allocated to the display.

The video timing registers also support high-resolution displays. The 16-bit vertical counter register, VCOUNT, directly supports screen lengths of up to 65,536 lines. The 16-bit horizontal counter register, HCOUNT, does not directly limit horizontal resolution. Each horizontal line can be up to 65,536 VCLK (video clock) periods in length. The VCLK period, however, is an arbitrary number of dot-clock periods long, depending on the external divide-down logic that derives VCLK from the dot clock. Therefore, the number of pixels per line supported by the TMS34020's horizontal timing registers is limited only by the amount of video memory present.

A typical screen must be refreshed 60 times per second for a noninterlaced display, or 30 times per second for an interlaced display. For a noninterlaced display, the dot period (or time to refresh 1 pixel) is estimated as

For an interlaced display, the dot period is estimated as

dot period = <u>
DBR × (1/30 second)</u> (pixels/line) (lines/frame)

*DBR* is the **display-to-blanking ratio**, equal to the unblanked fraction of each frame. This is typically about 0.8, although this factor varies from monitor to monitor. During each dot period, the complete information for 1 pixel must be obtained from the display memory, or frame buffer. Thus, the rate at which video data must be supplied from the display memory (usually the limiting factor for large systems) is a function of pixel size as well as screen dimensions.

# 9.11 Display Interrupts and Applications

You can program the TMS34020 to interrupt the CPU when a specified line is displayed on the screen. This is called the **display interrupt**. Enabling the display interrupt is a 2-step process:

- Step 1: Set DIE [INTENB] to 1.
- Step 2: Load the DPYINT register with the number of the desired horizontal scan line. When VCOUNT = DPYINT, the interrupt request is generated to coincide with the start of horizontal blanking at the end of the specified line.

DIP[INTPEND] is set every time the interrupt request is generated. You can poll the display interrupt by disabling the interrupt (setting DIE to a 0) and checking the value of DIP. Writing a 0 to DIP clears the interrupt request.

The display interrupt has several applications:

- ❑ Coordinating modifications to a bitmap with displaying the bitmap's contents. For example, while the bottom half of the screen is being displayed, the TMS34020 can modify the bitmap used for the top half of the display, and vice versa.
- Maintaining a cursor on the monitor screen. The cursor image resides in the on-screen memory only during the time the electron beam is scanning the lines containing the cursor, and remains free from flicker even during periods when the TMS34020 is busy drawing to the screen. The technique is to load the DPYINT register with the VCOUNT value of the scan line just above where the top of the cursor is to appear. When the display interrupt occurs, the interrupt service routine performs the following tasks:
  - Sets DPYINT to the scan line just below the cursor,
  - Saves the portion of the screen where the cursor is to appear, and
  - PIXBLTs the cursor onto the screen.

The cursor then remains on the screen until the electron beam next scans the lines it is on. In the mean time, as soon as the electron beam reaches the bottom of the cursor, a second display interrupt occurs. This causes the original screen to be restored in preparation for the next frame, and the TMS34020 can resume drawing to the screen.

□ Split-screen applications. By modifying the contents of SRNX[DPYNX] part way through a frame, you can display different parts of the bitmap in different horizontal bands of the screen. Sections 9.15.2 and 9.15.3 (beginning on page 9-53) discuss the use of SRNX as part of the screenrefresh mechanism. No special steps are necessary to ensure that loading a new value to SRNX does not interfere with the ongoing screen- refresh cycle; the display interrupt is requested at the beginning of the horizontalblanking interval coincident with a screen-refresh request. However, the TMS34020 cannot respond to the interrupt request until the screen refresh and subsequent updating of SRNX is complete (this is true whether the interrupt was taken or the TMS34020 polls the DIP bit for the 0-to-1 transition). After DIP is set to 1, SRNX can be loaded with a new value to achieve the split screen any time before the next screen-refresh cycle.

In interlaced mode, VCOUNT increments every half scan line during the equalization and serration regions of vertical blanking. When this is occurring, DPYINT is compared with VCOUNT twice per scan line; the display interrupt may occur at the start of horizontal blanking and also when HCOUNT = HTOTAL/2 (in the center of each line).

# 9.12 Video Timing Programming Examples

This section illustrates procedures for determining the values to be programmed into the TMS34020's video timing logic.

# 9.12.1 Noninterlaced 1024 × 768 Display

This example assumes that the NIL, CVD, and CSD bits in DPYCTL are set to 1, so that the TMS34020 is in noninterlaced video mode with separate horizontal- and vertical-blanking outputs. The values of other control bits in DPYCTL depend on other system specifications not covered in this example (such as external synchronization modes).

# Specifications<sup>-</sup>

The monitor in this example uses these parameters; you could easily modify the example to fit the specifications of another monitor.

Horizontal (refer to Figure 9–3 on page 9-11):

Scan line duration (LD):	20.625 μS
Sync duration (HS):	1 μS
Back porch (HBP):	2.875 μS
Front porch (HFP):	0.75 μS

**Vertical** (refer to Figure 9–6 on page 9-13):

16.665 mS
83 µS
660 µS
82 µS

Screen dimensions: 1024 pixels by 768 lines

## Procedure-

Figure 9–13 (page 9-20) lists the timing register programming equations for noninterlaced video. Before you can calculate proper values for the video timing registers, you must determine the active display time and the VCLK period:

Use the horizontal information to calculate the **active display time**:

active time (AT) =  $LD - HS - HBP - HFP = 16.000 \,\mu S$ 

At 1024 pixels per line, this is 15.625 nS per pixel.

Determine the VCLK period. The minimum VCLK period VCLK is 50 nS, which means that the pixel dot clock must be divided by at least 4 to generate VCLK. Assuming this to be the case,

VCLK period  $(T_{vclk}) = 62.5 nS$  and VCLK frequency = 16 MHz

Step	Calculate	Formula	Yields a register value of
1	Number of VCLK cycles per scan line	$\frac{LD}{T_{vclk}} = 330$	HTOTAL = 329 <sub>10</sub> (149h)
2	Number of VCLK cycles in horizontal sync	$\frac{HS}{T_{vclk}} = 16$	HESYNC = 15 <sub>10</sub> (0Fh)
3	Number of VCLK cycles in the horizontal back porch	$\frac{HBP}{T_{vclk}} = 46$	
4	Number of VCLK cycles between the beginning of horizontal sync and the end of horizontal blanking	$\frac{HS + HBP}{T_{vclk}} = 62$	HEBLNK = 61 <sub>10</sub> (3Dh)
5	Number of VCLK cycles in the horizontal front porch	$\frac{HFP}{T_{vc/k}} = 12$	
6	Number of VCLK cycles between the beginning of horizontal sync and the beginning of horizon- tal blanking	$\frac{LD - HBP}{T_{vclk}} = 218$	HSBLNK = 217 <sub>10</sub> (0D9h)
7	Number of lines per frame	$\frac{FD}{LD} = 808$	VTOTAL = 807 <sub>10</sub> (327h)
8	Number of lines in vertical sync	$\frac{VS}{LD} = 4$	VESYNC = 7 <sub>10</sub> (7h)
9	Number of lines in the vertical back porch	$\frac{VBP}{LD} = 32$	
10	Number of lines between the beginning of verti- cal sync and the end of vertical blanking	$\frac{VS + VBP}{LD} = 36$	VEBLNK = 35 <sub>10</sub> (23h)
11	Number of lines in the vertical front porch	$\frac{VFP}{LD} = 4$	
12	Number of lines between the beginning of verti- cal sync and the beginning of vertical blanking	FD – VFP <i>= 804</i>	VSBLNK = 803 <sub>10</sub> (323h)

## 9.12.2 Composite Interlaced NTSC Display Example

This example assumes that the NIL and CVD bits in DPYCTL are set to 0 to configure the TMS34020 for interlaced video, with CSYNC/HBLNK and CBLNK/VBLNK configured as CSYNC and CBLNK, respectively. The values of other control bits in DPYCTL depend on other system specifications not covered in this example (such as external synchronization modes).

## Specifications<sup>-</sup>

This example demonstrates how to program the TMS34020's video timing registers for fully NTSC-compatible composite video (that is, broadcast-standard TV). The required specifications are

Horizontal (refer to Figure 9–3 on page 9-11):

Scan line duration (LD):	63.556 μS
Sync duration (HS):	4.7 (±0.1) μS
Sync to setup (HSS) <sup>†</sup> :	9.4 (±0.1) μS
Front porch (HFP):	1.5 (±0.1) μS
Subcarrier period (SP):	(2 / 455) × LD

❑ Vertical (refer to Figure 9–6 on page 9-13):

Frame duration (FD):	525 lines
Sync duration (VS):	3 lines
Sync to setup (VSS) <sup>†</sup> :	17 lines
Front porch (VFP):	3 lines

- Screen dimensions: A typical TV display has 376 pixels per line. This yields screen dimensions of 376 pixels × 525 lines.
  - <sup>†</sup> Sync to setup is an NTSC-specified parameter, and is equal to sync plus the back porch.

#### Procedure

Figure 9–17 (page 9-26) lists the timing register programming equations for interlaced video. Before you can calculate proper values for the video timing registers, you must determine the minimum VCLK period and frequency.

In calculating the **minimum VCLK frequency**, the most important consideration for NTSC is that the NTSC subcarrier waveform (which transmits color information) and VCLK must be harmonically related. The number of subcarrier cycles per scan line is

There must be an integral number of VCLKs per line, which means that VCLK must be at least twice the frequency of the subcarrier, resulting in 455 VCLKs per line. However, for composite video, the equalization and serration pulses that occur in vertical blanking occur at exactly half scan line intervals, and this

means that the number of VCLKs per line must also be even. This results in VCLK being 4 times the frequency of the subcarrier, with 910 VCLKs per line. Therefore,

VCLK period ( $T_{vclk}$ ) = 69.48 nS and VCLK frequency = 14.318 MHz

Step	Calculate	Formula	Yields a register value of
1	Number of VCLK cycles per scan line	$\frac{LD}{T_{vc/k}} = 910$	HTOTAL = 909 <sub>10</sub> (148h)
2	Number of VCLK cycles in horizontal sync	$\frac{HS}{T_{vclk}} = 67.29$	HESYNC = 67 <sub>10</sub> (42h)
		This is not an inte derived is still with μS. Remember th ber of VCLKs long exactly half that d	eger, but if 68 is chosen, the value of HS hin the $\pm 0.1 \ \mu$ S tolerance specified: 4.749 at horizontal sync should be an even numg to ensure that the equalization pulses are uration.
3	Number of VCLK cycles in horizontal serration	$\frac{\frac{LD}{2} - HS}{T_{vclk}} = 387.$	HESERR = 387 <sub>10</sub> (183h) 7
		388 gives the inac is specified idention tolerance	ctive time between serration pulses, which cal to HS, as 4.679 $\mu$ S, which is within the
4	Number of VCLK cycles between the beginning of horizontal sync and the end of horizontal blanking	$\frac{HSS}{T_{vclk}} = 134.58$	HEBLNK = 134 <sub>10</sub> (86h)
		135 is within the te	olerance specified: HSS=9.428 $\mu$ S
5	Number of VCLK cycles in the hori- zontal front porch	$\frac{HFP}{T_{vclk}} = 21.47$	
		22 is within the to	lerance specified: HFP=1.536 $\mu$ S
6	Number of VCLK cycles between the beginning of horizontal sync and the beginning of horizontal blanking	$\frac{LD - HFP}{T_{vclk}} = 88$	HSBLNK = 887 <sub>10</sub> (337h) 8

Because the vertical timing parameters are already specified in lines, it is necessary only to work through the formulas given in Figure 9–17 to calculate the vertical timing register programming:

Step	Calculate	Formula	Yields a register value of
7	VESYNC	$VS \times 4 - 1$	VESYNC = 11 <sub>10</sub> (0Bh)
8	VTOTAL	$\frac{FD}{2}$ + (2 × VS) + VFP = 271.5	VESYNC = 271 <sub>10</sub> (10Fh)
9	VSBLNK	VTOTAL – VFP $\times 2$	VSBLNK = 265 <sub>10</sub> (109h)
10	VEBLNK	VSS + 2 × VS – 1	VEBLNK = 22 <sub>10</sub> (16h)

# 9.13 Video RAM Control

**Display memory** is the area of memory that holds the graphics image output to the video monitor. The display memory is typically implemented with VRAMs (video RAMs). The TMS34020 automatically schedules VRAM memory-to-register cycles, called **screen-refresh cycles**, needed to refresh a video screen. A screen-refresh cycle typically affects all VRAMs in a system. During a screen-refresh cycle, a selected row of the display memory is transferred to the internal serial register of each VRAM. The data is then shifted out to refresh the display.

The TMS34020 supports 2 distinct types of screen refresh:

- Screen refreshes that occur during horizontal blanking and
- Screen refreshes that occur during the active display time.

The addresses output during both types of screen-refresh cycles are generated by a separate mechanism discussed in Section 9.15 (page 9-51); this section also discusses other VRAM-control options programmed via the DPYCTL register.

#### 9.13.1 Screen Refreshes During Horizontal Blanking

The video timing logic schedules a screen-refresh cycle at the beginning of each horizontal-blanking interval. During blanking, the VRAM serial registers should not be shifting data (SCLK should be off). During the vertical-blanking interval, no screen refreshes take place until the horizontal-blanking interval at the beginning of the first displayed line of the new frame. Figure 9–20 shows the local-memory screen-refresh memory cycle.



Figure 9–20. Local-Memory Memory-to-Register Transfer Cycle

# 9.13.2 Screen Refreshes During the Active Display Time (Midline Reload)

The TMS34020 contains dedicated circuitry that operates synchronously to the VRAM's shift clock (SCLK). This enables the TMS34020 to perform screen refreshes during the active display time in systems with VRAMs that have split serial registers (such as the TMS44C251, a 1-Mbit,  $256K \times n$  VRAM). These occur without interrupting the flow of data to the screen. If enabled, mid-line-reload screen-refresh cycles occur as well as (not instead of) horizon-tal-blanking screen-refresh cycles.

The SCOUNT register is loaded automatically during the horizontal-blanking screen refresh with the column address (or tap-point) portion of the logical address output to the VRAMs. When blanking ends and SCLK starts again, SCOUNT is incremented each time the VRAM serial registers shift out a bit of data. In this way, SCOUNT always contains the column address of the bit of

data currently being shifted out. When SCOUNT increments from all 1s (the column address of the last bit to be shifted out of one-half of the serial register) to all 0s (the column address of the first bit to be shifted out of the other half of the serial register), a midline-reload screen-refresh cycle is scheduled.

You must set SSV[DPYCTL] to a 1 to allow the TMS34020 to perform split-serial-register screen refreshes when they are scheduled by SCOUNT. The DPYMSK register must also be programmed to ensure that SCOUNT is loaded with the correct portion of the full logical address (Section 9.15, page 9-51, describes this). Figure 9–21 shows the split-serial-register VRAM mid-line-reload screen-refresh memory cycle. The generation of the addresses output during the register-to-memory cycles described in this section is also covered in Section 9.15.





When midline reload is enabled, an extra screen refresh is scheduled during horizontal blanking. As for any horizontal-blanking period, an ordinary memory-to-register cycle loads the specified row of VRAM into each of the VRAM serial registers. This loads **both** halves of the split serial register. However, if the tap point is more than halfway along the row, then the idle half of each serial register (loaded from the lower half of the VRAM row) will not contain data from the next row of VRAM to be displayed. To ensure that the idle half contains the **next** half row, and not the **previous**, a split-serial-register screen refresh with the address of the **next** half-row is generated immediately after the ordinary horizontal-blanking screen refresh. This is the same type of cycle used for the midline-reload screen-refresh cycles, and loads only **half** of each of the VRAM serial registers. Figure 9–22 shows these memory cycles.





Note: During the second screen-refresh cycle, the MSB of the column address identifies which half of the serial register is reloaded. Remaining bits of the column address = 0 and represent the tap point. The row address output now addresses the memory row containing the next half row to be transferred to the split serial reg.

#### Note:

Normally, SCLK should be low during blanking. However, VRAMs require an SCLK pulse to latch the tap-point address into an internal counter. As Figure 9–22 shows, 2 tap-point addresses are provided during horizontal blanking. To ensure correct operation, you must provide a single SCLK pulse to the VRAMs between  $\overline{TR}/\overline{QE}$ 's low-to-high transition at the end of the first screen-refresh cycle, and  $\overline{RAS}$ 's high-to low transition at the beginning of the second screen-refresh cycle. (This is in accordance with the timing specifications for VRAMs.) Failure to do this causes the TMS34020 to ignore the tap-point address provided during the screen-refresh cycle.

#### 9.13.3 Why Use Midline Reload?

If serial-register transfers during horizontal blanking are the only type of transfers available in a system, the display memory must be arranged in a way that allows the VRAM serial registers to be loaded with enough information for an entire scan line during each horizontal-blanking interval.Each row of VRAM must contain at least as many bits as the number of pixels on one screen line. Depending on the number of pixels, this means that there can potentially be a significant amount of the display memory that cannot be used.

If, however, the VRAM serial registers can be reloaded during the active display time as well as during blanking, there is no need for any of the display memory to be wasted. Also, because each row of the display memory no longer needs to contain enough information for a whole screen line. The number of memory rows required may allow the display memory to be contained in fewer banks. Section 8.17 (Double-Buffered Display Example, page 8-57) illustrates how using midline reload can reduce the amount of VRAM required for display memory.

Although midline reload can reduce the amount of VRAM required, there is an inherent trade-off: The screen pitch may not be a power of 2. If XY addressing is being used for graphics instructions (as it typically is), this increases the time taken to convert an XY address to a linear address:

- If the screen pitch is a power of 2, each conversion takes 3 machine cycles.
- □ If the screen pitch is the sum of two powers of 2, each conversion takes 4 machine cycles.
- If the screen pitch is an arbitrary value, each conversion takes 15 cycles.

In addition to the CVXYL instruction, which performs one conversion, graphics instructions that have XY operands (either explicit or implicit) automatically perform XY-to-linear conversions. The following instructions are affected:

- DRAV performs one conversion.
- FILL XY and PFILL XY perform one conversion.

- LINE performs one conversion.
- PIXBLT instructions perform one conversion for each XY operand.
- PIXT instructions perform one conversion for each XY operand.

Chapter 12, *Graphics Instructions and Operations*, provides precise details of how the execution time for each instruction is affected by using a screen pitch that is not a power of 2.

#### 9.13.4 VRAM Bulk Initialization

VRAMs may be rapidly loaded with an initial value using a special feature that converts pixel accesses into register transfers. This rapid loading method is referred to as **bulk initialization**. When CST [DPYCTL] is set to 1, the TMS34020 converts all reads and writes of pixel data into register-transfer cycles. When CST = 0, pixel accesses are performed in the normal way.

When CST = 1, the TMS34020 can initiate register-transfer cycles under explicit program control. By performing a series of such cycles, some or all of the display memory can be set to an initial background color or pattern very rapidly (in a small fraction of a frame time). First, the VRAM serial registers are loaded with an initial value. The video memory is then set to the initial color or pattern one row at a time by writing the serial register contents to the memory.

The row address output during the register-transfer cycle determines which row of memory is involved in the transfer. The direction of the transfer is determined by whether the cycle is a read or write. A write cycle (such as a PIXT from a general-purpose register to memory) is converted to a VRAM register-to-memory cycle. Similarly, a read cycle (such as a PIXT from memory to a general-purpose register) is converted to a VRAM memory-to-register cycle.

The value of the CST bit affects only pixel transfers. Other data accesses and instruction fetches are unaffected.

Before bulk initialization, the VRAM serial registers must be loaded with the solid color or pattern with which the display memory will be loaded. This can be accomplished by

- serially shifting bits into the serial registers, or
- Ioading a row of display memory with the color or pattern using a series of normal pixel writes (CST = 0), then loading the contents of this row into the VRAM serial registers by means of a PIXT memory-to-register instruction (CST = 1).

To further speed bulk initialization, you can make a series of transfers more rapidly by using a single FILL instruction in place of a series of PIXTs. Select the fill region so that each pixel write cycle generates a new row address. The fill region should be 1, 2, 3, or 4 bytes wide. Plane masking and transparency

are disabled, and the pixel-processing replace option is selected. This ensures that the TMS34020 need not perform any read-modify-write operations, and each row is addressed only once during the course of the fill operation.

The number of bits of the display memory that are altered by a single register-to-memory transfer cycle is calculated by multiplying the number of VRAM devices by the number of serial register bits in each device.

#### 9.13.5 Video Capture

If VCE[[DPYCTL]] is set to 1, all screen-refresh cycles are executed as register-to-memory cycles, rather than memory-to-register cycles. Figure 9–23 shows a memory-to-register cycle. This feature is useful if the VRAMs are used to capture video data by shifting data bits *into* them, rather than out.





**Do not** use the video-capture feature with midline reloads. The registerto-memory cycle overwrites an entire VRAM row from the VRAM serial register. In the case of the first midline reload that follows the end of horizontal blanking, not all the bits in the serial register would need transferring—only those shifted in since the end of blanking.

#### 9.13.6 Disabling Screen Refreshes

In order for the TMS34020 to perform screen-refresh memory cycles initiated by the video-timing logic or the midline-reload logic, SRE[[DPYCTL]] (screen refresh enable) must be set to a 1. While it is 0, the TMS34020 still increments the screen-refresh address during horizontal blanking, but no actual screenrefresh memory cycles occur.

This can be useful for inserting a video image from another source onto the image generated by the TMS34020. If a section of the screen image is to be provided by another source, SRE is set to 0 during the line immediately preceding the inserted image. The TMS34020 then stops refreshing the screen, allowing another device to perform this function.

To revert to the TMS34020-generated image, set SRE to 1 again during the last line of the inserted image. The TMS34020 then restarts screen refreshes, beginning in the horizontal-blanking interval immediately following the last line of the inserted image. Because the screen-refresh address is incremented every line, regardless of the value of SRE, the first line displayed will be the same line that would have been displayed at this time if the screen refreshes had never been switched off.

If midline-reload screen refreshes are also used, the SSV bit should be set to 1 at the same time as the SRE bit. However, you should note that if the TMS34020's SCLK input is clocking during the last line of the inserted image after the SSV bit has been set to 1, a midline-reload screen refresh could occur before the end of the line. Holding the SCLK input low while the inserted image is being output to the screen eliminates this problem.

# 9.14 Scheduling Screen-Refresh Cycles

The horizontal-blanking period should be long enough to ensure that the TMS34020 has completed the serial-register transfer by the time blanking ends and data starts shifting out to the screen again. The delay from the start of horizontal blanking until the start of the screen-refresh memory cycle is called the **screen-refresh latency** and is determined by the TMS34020's memory controller. It consists of 2 elements:

- The time necessary to synchronize and recognize the screen-refresh request from the video timing logic, and
- The time to complete the memory cycle currently in progress when the request is recognized.

The synchronization and recognition time is, at worst, 3.5 machine states and, at best, 2.5 machine states. Table 9–1 shows the maximum and minimum screen-refresh latency.

#### Table 9–1. Screen-Refresh Latency

	٨	<i>l</i> easurement	Value
		Minimum screen-refresh latency	2.5T
		Maximum screen-refresh latency:	
		DRAM refreshes enabled	(5.5 + N)T
		DRAM refreshes <i>disabled</i>	(4.5 + N)T
Key:	T	local clock period	

N maximum number of wait states per memory cycle

Page mode bursts are terminated by the screen-refresh request and restart after the screen-refresh cycle from the next address in the sequence. In this way, they do not delay the start of the screen-refresh cycle.

A new memory access (in which the full address is output on the LAD bus) that starts, or is in a wait state, in the machine state just before the screen refresh is recognized cannot be aborted until the first word of data has been transferred. This, therefore, delays the start of the screen refresh by 1+N machine states. However, a DRAM refresh (which takes three machine states minimum) will delay the start of the screen refresh by 2+N machine states.

The length of time taken for the screen refresh to complete, once started, depends on the number of wait states used. The minimum time is 2 machine states. For systems using split-serial-register VRAM midline reloads, the horizontal-blanking screen-refresh cycle is actually 2 distinct memory cycles (see Section 8.13.2). In this case, the minimum duration of the horizontal-blanking screen refresh is 4 machine states, and both of the memory cycles can have wait states inserted. The maximum possible time from the beginning of horizontal blanking to the end of the screen-refresh cycle defines the minimum horizontal-blanking duration. Table 9–2 summarizes this.

## Table 9–2. Minimum Horizontal-Blanking Duration

Measurement	Value	
Minimum horizontal-blanking duration ( <i>without</i> split-serial-register VRAM reloads):		
DRAM refreshes enabled	(7.5 + N)T	
DRAM refreshes disabled	(6.5 + N)T	
Minimum horizontal-blanking duration ( <i>with</i> split-serial-register VRAM reloads)		
DRAM refreshes enabled	(9.5 + 2N)T	
DRAM refreshes disabled	(8.5 + 2N)T	
Key: T local clock period		

N maximum number of wait states per memory cycle

# 9.15 Generating Screen-Refresh Addresses

This section describes methods for programming the TMS34020 to generate the addresses for all types of screen refreshes.

The address output during a screen-refresh cycle identifies the first of a series of pixels to be output to the monitor. In the case of a horizontal-blanking screen refresh, this is the first pixel on the next scan line. For a midline reload, it is the first pixel of the next row of VRAM. The TMS34020 outputs a 28-bit logical address on LAD31—LAD4. This address can be broken down into several fields:

- the 16-bit word select on bit 4,
- the column-address and row-address fields, and
- a bank-select field or fields (optional). The bank-select field(s), if required, can be at either end of the address (less significant than the column address, or more significant than the row address, or both).

The precise positioning of all these fields (except for the 16-bit word-select bit), must be determined by system requirements. Figure 9–24 shows this.

# Figure 9–24. Screen-Refresh Address Fields

- (a) Bank select at the most significant end of the address
- (b) Bank select at the least significant end of the address

bank	row	column	S	status
		•		
31			4	3
row	col	umn	bank S	status

- Notes: 1) The screen refresh status code is output on LAD3—LAD0.
  - 2) The S bit is always 0 unless dynamic bus sizing is employed to access 16-bit wide VRAMs, and is always 0 during screen-refresh cycles.
  - 3) The address may not extend all the way to address bit 31 as shown. In a typical system, the full address range is unlikely to be required.

The row address output during the screen-refresh cycle specifies the row in memory to be loaded into the serial register internal to the VRAM. The column address determines which bit of that row is shifted out of the VRAM serial register first.

## 9.15.1 Horizontal-Blanking Screen-Refresh Addresses

The portion of the display memory actually output to the monitor is referred to as **on-screen memory**. To generate the appropriate address for each horizon-tal-blanking screen refresh, it is necessary to know

The starting location of the on-screen memory.

Typically, the starting location of the on-screen memory is the address of the pixel appearing in the top left-hand corner of the display. Load this address into SRST[DPYST].

The difference in 32-bit memory addresses between 2 vertically adjacent pixels on the screen; this is called the **screen pitch**.

Screen pitch is the difference between the memory addresses of the first pixels of 2 consecutive scan lines. For systems not using midline reload, this is also the width of the display memory because each row of VRAM must contain all the information for a whole scan line. Load the screen pitch into SRINC[[DINC]].

In some systems, the screen may be refreshed starting from the bottom of the screen. If this is the case, load SRINC with the 2s complement of the screen pitch so that the screen-refresh address is decremented between each horizontal-blanking screen refresh.

- □ The vertical magnification (Y-zoom) of the display. This is the number of times each scan line in the display memory will be displayed on the screen. Typically, each line is displayed only once; the remainder of this discussion assumes this is the case. For more information about the Y-zoom feature, refer to Section 9.15.5 on page 9-56.
- Whether or not the display is **interlaced** (NIL [DPYCTL] = 0).

The TMS34020 automatically calculates the address required for each screen refresh and stores the address in SRNX[[DPYNX]].

The DPYMSK register is required for midline-reload screen refreshes (see Section 9.15.4 on page 9-55). However, if you are not using midline reload, there is no need to program DPYMSK.

# 9.15.2 Screen-Refresh Addressing Sequence for Noninterlaced Displays

If NIL[[DPYCTL]] = 1, the TMS34020 generates screen-refresh addresses in the following sequence:

- 1) At the beginning of the vertical-blanking period, the address specified by SRST[[DPYST]] is loaded into SRNX[[DPYNX]].
- 2) In the horizontal-blanking period just before the first line of the frame, the addresses contained in SRNX is output to the VRAMs during the screenrefresh cycle. SRINC [[DINC]] is then added to this address, and the result is loaded back into SRNX.
- 3) During the next horizontal-blanking period, the contents of SRNX are output during the screen-refresh cycle. SRINC is then added to this, and the result is loaded back into SRNX. This process then repeats during all subsequent horizontal-blanking periods until the end of the frame.

Figure 9–25 illustrates this sequence in the form of a flow chart.

# 9.15.3 Screen-Refresh Addressing Sequence for Interlaced Displays

If NIL[DPYCTL] = 0, the TMS34020 generates screen-refresh addresses in 2 sequences—one for the odd field and one for the even field. These sequences are described below, and illustrated in the form of a flow chart in Figure 9–25.

- At the beginning of the vertical-blanking interval between the odd and even fields, the address specified in SRST[[DPYST]] is added to half SRINC[[DINC]], and the result is loaded into SRNX[[DPYNX]]. This is because vertical blanking ends halfway across the first line of the even field, and the first pixel displayed is half a line away from the on-screen memory start address (refer back to Figure 9–14).
- 2) In the last horizontal-blanking period before the first line of the even field, the address contained in SRNX is output to the VRAMs during the screen-refresh cycle. Then, *twice* SRINC is added to SRST, and the result is loaded into SRNX. SRNX then points to the first pixel on the third scan line because alternate lines are scanned in interlaced mode.
- 3) During the next horizontal-blanking period, the contents of SRNX are output during the screen-refresh cycle. Twice SRINC is then added to this, and the result loaded back into SRNX. This process then repeats during all subsequent horizontal-blanking periods until the end of the field.
- 4) At the beginning of the vertical-blanking interval between the even and odd fields, the address specified in SRST[DPYST] is added to SRINC, and the result is loaded into SRNX. SRNX then points to the first pixel on the second scan line (the first line scanned in the odd field).

- 5) In the horizontal-blanking period just before the first line of the odd field, the address contained in SRINC is output to the VRAMs during the screen-refresh cycle. Then, *twice* SRINC is added to SRNX, and the result is loaded back into SRNX. SRNX then points to the first pixel on the fourth scan line.
- 6) During the next horizontal-blanking period, the contents of SRNX are output during the screen-refresh cycle. Twice SRINC is then added to this, and the result is loaded back into SRNX. This process then repeats during all subsequent horizontal-blanking periods until the end of the field.

Figure 9–25. Screen-Refresh Address Generation Flow



## 9.15.4 Midline-Reload Screen-Refresh Addresses

If midline-reload screen refreshes are enabled, the column address part of the logical address must be identified, so that

- SCOUNT can be loaded with the VRAM tap-point to enable it to correctly schedule the midline-reload screen-refresh cycles.
- The correct row address can be isolated and output during the midline-reload screen refreshes.

DPYMSK stores a mask of contiguous 1s , which correspond to the column address portion of SRST [DPYST] or SRNX [DPYNX] .

This mask determines which bits of the logical address are loaded into SCOUNT. SCOUNT is loaded with the tap-point portion of the address loaded into SRNX whenever SRNX is loaded during horizontal blanking.

256K×4 (1 Mbit) VRAMs have 9 column address bits and 9 row address bits (512 rows and 512 columns), but because they also have split serial registers, each row of VRAM is conceptually split into 2; each midline-reload screen-refresh cycle loads only half a VRAM row into the relevant half serial register. Thus, the address must be incremented at the most significant column-address bit, rather than the least significant row bit (to select between the 2 halves of the VRAM). Therefore, when considering midline reload and DPYMSK, the most significant column-address bit of the VRAM is not considered as being part of the tap point, but as the least significant half-row address bit.

When a midline-reload screen refresh occurs, the address output to the VRAMs is the address of the first pixel in the next half-row of VRAM. Therefore, the column portion of the address is 0.

The mask stored in DPYMSK is also used to isolate the half-row address from the full logical address (all bits more significant than that mapped to by the most significant 1 in DPYMSK are considered to be the half-row address). A non user-accessible register is loaded with the row address portion of the address loaded into SRNX whenever SRNX is updated during horizontal-blanking screen refreshes. This address is then incremented to point to the next VRAM half-row at the bit position mapped to the rightmost 0 at the most significant end of DPYMSK.

If a midline-reload screen refresh occurs, this address is output to the VRAMs and then incremented to point to the next half row. All address bits less significant than the half-row address are masked by DPYMSK, so that they are output as 0s. This corresponds to a tap-point address of 0, thus pointing to the first pixel in the next half row. This procedure then repeats every time a midline-reload screen refresh is scheduled on the current scan line.

There is a 5-bit offset between DPYMSK and the logical address. Thus, if the column address is 8 bits long, starts at bit 7 of the logical address, and ends

at bit 14, then DPYMSK should be loaded with a contiguous field of 1s between bits 2 and 9 (inclusive). All other bits of DPYMSK should be 0. Figure 9–26 shows the mapping.





Note: Bit 0 of DPYMSK maps to bit 5 of DPYST, and so on.

Bit 13 is the lowest address bit that can be incremented to change the half-row address for midline-reload screen refreshes. The most significant 1 in DPYMSK must be at bit 7 or higher of DPYMSK.

Bit 16 is the highest address bit that can be incremented for midline-reload screen refreshes. The most significant 1 in DPYMSK must not be any higher than bit 10 of DPYMSK.

# 9.15.5 Display Magnification and Y-Zoom

It is often desirable to magnify part of an image on the display. To do this, you can make the pixels appear larger on the display by increasing their physical X and Y dimensions.

- The X dimensions of pixels can be increased by reducing the rate at which they are fed to the monitor. Fewer pixels are displayed on a scan line, so each pixel is wider.
- The Y dimension of pixels can be increased by repeating a scan line multiple times. If the information for one scan line is repeated on consecutive lines, the pixels appear taller.

The X dimension must be controlled by external hardware that shifts serial pixel data to the monitor. The Y dimension, however, can be controlled by the TMS34020, which provides direct support for changing the Y dimension.

To repeat a scan line multiple times, simply add 0 to SRNX (instead of adding SRINC). Control for this is provided via YZINC [[DINC]] and YZCNT [[DPYNX]]. After each horizontal-blanking screen-refresh cycle, YZINC is added to YZCNT. As Table 9–3 shows, YZINC is normally a power of 2 between 0 and 16. This means that periodically (when YZCNT + YZINC = 32), YZINC overflows to 0 because it is only a 5-bit field.

If YZCNT=0 during the horizontal-blanking screen-refresh cycle, SRNX is incremented in the normal way (as described in Sections 9.15.2 and 9.15.3, and summarized in Figure 9–25).

If YZCNT≠0 during the horizontal-blanking screen-refresh cycle, 0 is added to SRNX or SRST instead of SRINC, 2×SRINC, or SRINC/2 (as outlined in Figure 9–25) after the horizontal-blanking screen refresh cycle. During each vertical-blanking interval, YZCNT is reset to YZINC.

## Table 9–3. Y-Zoom Control

YZINC		Zoom				
4	3	2	1	0	Factor	Description
0	0	0	0	0	0	No repetitions of scan lines
1	0	0	0	0	2	Repeat scan line 2 times
0	1	0	0	0	4	Repeat scan line 4 times
0	0	1	0	0	8	Repeat scan line 8 times
0	0	0	1	0	16	Repeat scan line 16 times
0	0	0	0	1	32	Repeat scan line 32 times

Clearing YZINC to 0 causes no Y-zoom because YZCNT is always 0. Conversely, when YZINC=1, YZCNT equals 0 only once every 32 horizontal-blanking intervals.

If YZINC is set to an odd value (such as 31), YZCNT will never be 0, and the scan line will be repeated indefinitely. This could be useful if you wish to clear the screen temporarily while transferring a new image to the display memory. A particular line could be transferred to the entire screen without the need for bulk erasing the VRAMs of the display memory.

To ensure that YZCNT +  $n \times$ YZINC (where *n* is the zoom factor) always overflows to 0, YZCNT should always be cleared to 0 when YZINC is changed. This is the only time it should be necessary for you to write to YZCNT.

## 9.15.6 Panning the Display

By changing the value of the column-address portion of SRST[DPYST] between frames, you can pan the on-screen image horizontally across the display memory. By changing the value of the row-address portion of SRST between frames, you can pan the on-screen image vertically up and down the display memory.

SRST is loaded into SRNX[DPYNX] at the beginning of vertical blanking. To affect the position of the screen image in the display memory for the next frame, SRST should be changed before the beginning of vertical blanking at the end of the current frame.



ł

# **Communicating with a Coprocessor**

The TMS34020 can communicate with a coprocessor through the **coprocessor interface**, which consists of special instructions and bus cycles. A coprocessor extends TMS34020 capabilities, providing hardware and software enhancements without incurring significant control overhead.

A coprocessor differs from a typical peripheral device, which is usually mapped to reside within the TMS34020's address space. When using peripherals, you must write code to provide the communications protocol between the TMS34020 and the peripheral device. However, the TMS34020's coprocessor interface extends the base architecture of the TMS34020, adding instructions that implement the communication protocol in hardware. Thus, the TMS34020 can use the coprocessor's data types and registers without treating the coprocessor as a separate device.

This chapter provides the following details about the coprocessor interface:

	Sectio	n	Page
Basic information includes a	10.1 F	Related Signals	10-2
review of related TMS34020	10.2 0	Overview of the Coprocessor Interface	10-3
signals and an overview of the	10.3 F	Format of Commands Passed	
coprocessor interface.	te	o a Coprocessor	10-5
Advanced information discusses	10.4 L	_ocal-Memory Coprocessor Cycles	10-8
memory cycles.	10.5 0	Coprocessor Aborts and Status Checks	10-17
	10.6 S	System Configuration	10-18

#### Note:

Before reading this chapter, you should be familiar with the TMS34020's localmemory interface. If you are not, please read Chapter 8.

# **10.1 Related Signals**

The coprocessor interface uses a subset of the local-memory interface signals. Although these are primarily local-memory signals, their functions may differ when used for the coprocessor interface. Chapter 2 describes these signals in detail; they are summarized below for your convenience.

Signals	Descriptions	I/O
ALTCH	is the address latch signal. ALTCH's high-to-low transi- tion latches the current coprocessor instruction (and status code) or address. During data transfers from the coprocessor to TMS34020 memory, a high level on ALTCH enables data to the LAD bus.	0
BUSFLT	is the bus-fault signal.If external logic detects an error or fault in the current cycle, it asserts BUSFLT high. BUSFLT is used with LRDY to generate bus-retry cycles. A coprocessor must monitor this signal to determine the status of cycle termination.	I
CAS0—CAS3	are the column-address strobe signals. They can be connected to a coprocessor to control data transfers.	0
LAD0—LAD31	form the multiplexed local address/data bus. The coprocessor interface uses this bus to transfer coprocessor instructions, data, and the memory addresses that are used for the transfers.	I/O
LCLK1, LCLK2	are the local output clocks. These signals drive the coprocessor logic, providing control signals that are synchronous to the TMS34020.	0
LINT1, LINT2	are the TMS34020's local interrupt requests that a coprocessor can use to interrupt the TMS34020.	1
LRDY	is the local ready signal. External circuitry (which may be controlled by a coprocessor) can drive LRDY low to prevent the TMS34020 from completing a local- memory cycle. LRDY is used with BUSFLT to indicate retries and bus faults. A coprocessor must monitor this signal to determine the status of cycle termination.	ł
SF	is the special function signal that coprocessors monitor to distinguish between instruction cycles and address cycles.	0
WE	is the write-enable signal that identifies the direction of a data transfer.	0

# **10.2 Overview of the Coprocessor Interface**

The coprocessor interface provides both software and hardware features that help the TMS34020 communicate with a coprocessor. These features take the form of extensions to the TMS34020's instruction set and local-memory interface. Special coprocessor instructions may be used, which, in turn, invoke special local-memory interface cycles to pass commands and data between the TMS34020, local memory, and a coprocessor.

The instructions fall into two categories:

Generalized instructions for use with any coprocessor.

The TMS34020 supports several instructions that allow you to pass commands and data (when appropriate) to a coprocessor, in order that the coprocessor can execute the command. You pass commands and data as parameters to the TMS34020 instruction. Available coprocessor commands depend upon your coprocessor's requirements and upon the operation you wish it to perform. Table 10–1 lists these TMS34020 instructions; Chapter 13 describes them in detail.

Specialized instructions for use with the TMS34082 floating-point processor.

These are special cases of the generalized instructions, where the TMS34082 command is explicitly coded into the instruction opcode. Chapter 14 describes the TMS34082 instruction set.

#### Note:

The term *instruction* refers to an assembly-language instruction that the TMS34020 executes. The terms *command* and *coprocessor command* refer to assembly-language instructions that a coprocessor supports and executes.

Table 10–1. TMS34020 General Coprocessor Instructions

· Mnemor	nic I	Description
CEXEC	;	Execute coprocessor internal operation
CMOV	CG	Move, coprocessor to TMS34020
CMOV	СМ	Move, coprocessor to local memory
CMOV	CS	Move, coprocessor to status register
CMOV	GC	Move, TMS34020 to coprocessor
CMOVI	MC	Move, local memory to coprocessor

The coprocessor instructions can invoke five types of local-memory coprocessor cycles. Each cycle passes a command to the coprocessor; four of these cycles also transfer data. With a single instruction, the TMS34020 can pass a command to a coprocessor, and any associated data transferred to or from the coprocessor.

- The **coprocessor internal command** cycle passes a command to a coprocessor.
- The TMS34020 to coprocessor transfer cycle passes a command to a coprocessor, then performs one of the following data transfers between the TMS34020 and the coprocessor:
  - Move one 32-bit parameter
  - Move two 32-bit parameters
  - Move one 64-bit parameter
- □ The **coprocessor to TMS34020 transfer** cycle passes a command to a coprocessor, then performs one of the following data transfers between the coprocessor and the TMS34020:
  - Move one 32-bit parameter
  - Move two 32-bit parameters
  - Move one 64-bit parameter
  - Move 4 bits to the N, C, Z, and V bits of the TMS34020's status register
- The memory to coprocessor transfer cycle passes a command to a coprocessor, then performs one of the following data transfers of up to 32 32-bit words between the local memory and the coprocessor:
  - Move the number of 32-bit words specified in the coprocessor instruction using postincrement
  - Move the number of 32-bit words specified in the coprocessor instruction using predecrement
  - Move the number of 32-bit words specified in a register using postincrement
- □ The **coprocessor to memory transfer** cycle passes a command to a coprocessor, then performs one of the following data transfers of up to 32 32-bit words between the coprocessor and the local memory:
  - Move the number of 32-bit words specified in the coprocessor instruction using postincrement
  - Move the number of 32-bit words specified in the coprocessor instruction using predecrement

# **10.3 Format of Commands Passed to a Coprocessor**

The TMS34020 passes a command to a coprocessor over the LAD bus. The command replaces the address usually output during a local-memory cycle's address/status subcycle. In addition to the actual coprocessor command, some auxiliary control bits and the coprocessor operation status code are also output at this time, as Figure 10–1 shows.

Figure 10–1. Coprocessor Instruction Format



The information for the ID, the command, and the size bit are provided as arguments to the coprocessor instruction. The precise form of this information is determined by your processor's requirements and the operation you wish it to perform.

# 10.3.1 Coprocessor ID

The ID bits allow you to address multiple coprocessors within a system. Each coprocessor should respond to commands only when the ID output matches the coprocessor's assigned ID. If you do not supply an ID argument with the coprocessor instruction, the default ID is 000<sub>2</sub>. You can change the default ID with the .coproc assembler directive (refer to the *TMS340 Family Code Generation Tools User's Guide* for information about .coproc).

Table 10–2 lists suggested ID code assignments. Using these IDs promotes compatibility across different hardware configurations.

- **000<sub>2</sub>—011<sub>2</sub>** Use the first 4 coprocessor ID numbers to identify up to 4 TMS34082 floating-point processors.
- 100<sub>2</sub> Use as a broadcast ID to which all coprocessors should respond.
- **101**<sub>2</sub> & **110**<sub>2</sub> Reserve for compatibility with future devices.
- 111<sub>2</sub> Use for addressing your own coprocessor (in a manner that conforms to the TMS34020 coprocessor interface). This allows the TMS34020 to provide control through custom instructions.
| ID               | Description                                   | ID               | Description                                      |
|------------------|-----------------------------------------------|------------------|--------------------------------------------------|
| 000 <sub>2</sub> | TMS34082 <sub>0</sub> is to execute a command | 100 <sub>2</sub> | All coprocessors are to<br>execute command       |
| 001 <sub>2</sub> | TMS34082 <sub>1</sub> is to execute a command | 101 <sub>2</sub> | Reserved for future devices                      |
| 010 <sub>2</sub> | TMS34082 <sub>2</sub> is to execute a command | 110 <sub>2</sub> | Reserved for future devices                      |
| 011 <sub>2</sub> | TMS34082 <sub>3</sub> is to execute a command | 111 <sub>2</sub> | User-defined coprocessor to<br>execute a command |

Table 10–2. Suggested Coprocessor ID Assignments

#### Note:

You can adopt other assignments as required; however, all coprocessors should respond to the broadcast ID (100<sub>2</sub>).

#### 10.3.2 Coprocessor Command

The identified coprocessor uses the 21-bit coprocessor command to determine what operation it should perform. The command itself is specific to the coprocessor that must execute the command.

Some coprocessors may not require all 21 bits to fully specify a command. If this is the case, other information (such as data, status, or other coprocessor control) may also be coded into this area.

At least part of the command should inform the coprocessor of what type of data transfer is to follow the command (if any) and the number of 32-bit words to be transferred.

#### 10.3.3 Coprocessor Parameter Size (size)

During direct data transfers between the TMS34020 and a coprocessor, data is transferred during the data subcycle (immediately following the command). Section 10.4.2 (page 10-8) discusses direct and indirect data transfers in more detail. You can transfer 0, 1, or 2 32-bit words. The size bit identifies the size of the parameters passed to or from the coprocessor:

**size=0** Transfer 1 or 2 32-bit parameters.

**size=1** Transfer 1 64-bit parameter.

If no data is transferred following the command, size=0.

In either case, the data can be an integer or floating-point value. You can use size in conjunction with other bits of the coprocessor command to specify which of these formats is appropriate for the data.

## **10.3.4 Coprocessor Parameter Index (I)**

During direct data transfers between the TMS34020 and a coprocessor, data is transferred during a data subcycle (immediately following the command). Section 10.4.2 (page 10-8) discusses direct and indirect data transfers in more detail. You can transfer 0, 1, or  $2^{-5}$  32-bit words. The I bit identifies which of the words are transferred:

- **I=0** The first (and possibly only) 32-bit word are transferred immediately following the command.
- I=1 The second 32-bit word are transferred immediately following the reissued command.

If no data is transferred following the command, I=0.

Initially, I is output as a 0. It is output as a 1 only when two 32-bit words are transferred. If the second transfer cannot be made using page mode. In this case, a complete memory cycle (consisting of address/status and data subcycles) must be generated for the second 32-bit word to be transferred. During the address/status subcycle for this transfer, I is 1. The rest of the command is the same at this time.

The second transfer cannot be performed using page mode if

- The coprocessor does not support page-mode operation and asserts PGMD high during the first data transfers.
- A high-priority local-memory request (such as a VRAM serial-register transfer) is requested while the first 32-bit word is being transferred; it is performed before the second 32-bit word is transferred.

#### 10.3.5 16-Bit Word Select (S)

The S bit output on LAD4 at the beginning of the coprocessor cycle is always 0 (low). The S bit is 1 during cycles in which the TMS34020 accesses 16-bit memory devices; however, the TMS34020 does not support coprocessor accesses to 16-bit-wide memory devices. TMS34020-controlled coprocessor cycles are restricted to memory accesses of 32-bit memory devices.

## 10.3.6 Coprocessor Status Code (BCST)

This code is 0000<sub>2</sub>. It is output during the address/status subcycle of all local-memory coprocessor cycles.

## **10.4 Local-Memory Coprocessor Cycles**

This section describes the local-memory coprocessor cycles for interfacing to a coprocessor. A coprocessor must be able to recognize these cycles in order to properly receive and transmit data through the coprocessor interface. During the address/status subcycle of these cycles, the TMS34020 outputs the coprocessor operation status code (0000<sub>2</sub>) on LAD0—LAD3.

In these examples, the TMS34020 controls the local-memory interface and provides all controls necessary for transferring commands and data to and from a coprocessor. If you want the coprocessor to control the local-memory interface through the TMS34020's multiprocessor interface, refer to Chapter 11.

#### 10.4.1 Passing Commands to a Coprocessor

As Section 10.3 describes, the TMS34020 passes a command to a coprocessor over the LAD bus during the address/status subcycle. The coprocessor must distinguish a command from an ordinary memory address by

- □ the status code of 0000<sub>2</sub> output on LAD0—LAD3 and
- a high level on the SF pin.

The command should be latched on ALTCH's high-to-low transition when the these conditions occur. Because a coprocessor command is output on the LAD bus in essentially the same manner as a normal memory address, parts of the command are also output on the RCA bus in the same manner as the row and column portions of a regular address.

## 10.4.2 Transferring Data to or from a Coprocessor

After passing a command to a coprocessor, data can be transferred using one of these methods:

- Direct transfer between the TMS34020 and a coprocessor. This transfers data (via LAD) in the data subcycle, immediately following the command.
- Indirect transfer between local memory and a coprocessor. The TMS34020 provides the addresses and control signals for the memory. No data is transferred immediately after the command; the TMS34020 generates another memory cycle sequence, beginning with an address/status cycle in which the memory address for the transfer is output.

#### Note:

All local-memory coprocessor cycles are implemented as 32-bit data transfer operations. The TMS34020 does not support data transfers between 16-bit memory and a coprocessor (using the TMS34020's dynamic bus-sizing feature). The TMS34020 ignores the state of the SIZE16 pin during local-memory coprocessor cycles. However, SIZE16 should still be asserted at a valid level (high or low) at the time it is sampled by the TMS34020.

## 10.4.3 Data Transfer Sequences to or from a Coprocessor

When more than one 32-bit word of data is to be transferred, the TMS34020 uses page mode unless either the coprocessor or the local memory indicates that they do not support page mode (this is accomplished by asserting the PGMD pin high at the appropriate time).

If page mode is not supported, the TMS34020 outputs an address/status subcycle before each data transfer. During direct data transfers, the command is output again (with I=1 to indicate that the second 32-bit word is about to be transferred).

Even if page mode is supported, high-priority local-memory requests (such as a VRAM serial-register transfer) can interrupt a page-mode sequence; the cycle restarts from the next word after the high-priority request is serviced. The sequence restarts with an address/status subcycle, outputting either the address of the appropriate location in memory (during an indirect transfer), or the command with I=1 (during a direct transfer).

Because of this, a coprocessor must be able to tolerate an interruption of a data-transfer sequence, and must be informed in advance of the number of words of data to be transferred. This information could be included as part of the coprocessor command that precedes the data transfer.

#### 10.4.4 Ending a Local-Memory Coprocessor Cycle

During local-memory coprocessor cycles, the TMS34020 samples the LRDY and BUSFLT pins (just as it samples them for other local-memory cycles). This is how the TMS34020 determines when and how the memory cycle ends. For a detailed discussion of how the TMS34020 responds to these inputs, see Section 8.6, Ending a Local-Memory Cycle (page 8-12).

- Inserting wait states. You can extend a coprocessor data transfer, as required, by inserting wait states. If the coprocessor is not ready to perform a data transfer when the local-memory cycle is initiated, it can control LRDY and BUSFLT to insert wait states.
- Retrying local-memory coprocessor cycles. A coprocessor data transfer may be retried. If this occurs, the memory cycle is performed again in the same manner as a regular access. Any data on the LAD bus during a retried memory cycle should be considered invalid and should not be latched or used by the coprocessor.
- Bus faults on local-memory coprocessor cycles. A coprocessor data transfer may be bus faulted. As the CPU initiates local-memory coprocessor cycles, the bus-fault interrupt is taken when a bus fault occurs. Any data on the LAD bus during a bus-faulted memory cycle should be considered invalid and should not be latched or used by the coprocessor. After the bus-fault interrupt is serviced, the memory cycle is performed again in the same manner as for a retry.

## 10.4.5 Coprocessor Command Cycle

Performed when Passing a command to a coprocessor (see Figure 10–2). No other data is transferred.

Indicated by

- DDIN is low and
- RAS, TR/QE, and SF are all high while ALTCH is low

Caused by

CEXEC instruction

Status code 00002

The command typically causes the coprocessor to execute some internal function. However, the coprocessor command could also be used to pass values to the coprocessor.





**Note:** Although the coprocessor command cycle never requires the use of pagemode cycles, you should still assert PGMD at a valid level (high or low) at the time it is sampled by the TMS34020.

## 10.4.6 Transferring Values from TMS34020 Registers to a Coprocessor

Performed when	Passing one or two parameters from TMS34020 registers to a coprocessor. Figure 10–3 shows two words being transferred using page mode.
Indicated by	$\overline{\text{RAS}}$ and SF are high while $\overline{\text{ALTCH}}$ is low
Caused by	CMOVGC instruction
Status code	0000 <sub>2</sub>





One or two 32-bit words may be transferred. If two words are transferred, the size bit determines whether the coprocessor should treat the words as two

32-bit parameters or one 64-bit parameter. If two words are transferred but page mode cannot be used, the I bit equals 1 when the command is reissued before transferring the second word.

A coprocessor should latch the data from the LAD bus on the low-to-high transition of CAS.

#### **10.4.7 Transferring Values from a Coprocessor to TMS34020 Registers**

Performed when	Passing one or two parameters from a coprocessor to TMS34020 registers. Figure 10–4 shows two words being transferred using page mode.				
Indicated by	<ul> <li>RAS, TR/QE, and SF are high while ALTCH is low</li> <li>DDIN is low</li> </ul>				
Caused by	<ul> <li>CMOVCG, which transfers one or two 32-bit words</li> <li>CMOVCS, which transfers one 32-bit word. The</li> </ul>				

TMS34020 ignores the 28 LSBs of this word; however, the 4 bits of data output on LAD28—LAD31 are transferred to the N, C, Z and V status bits.

#### Status code 00002

One or two 32-bit words may be transferred. If two words are transferred, the value of the size bit (output as part of the command during the address/status subcycle) depends on whether the TMS34020 treats the words as two 32-bit parameters or one 64-bit parameter. If two words are transferred but page mode cannot be used, the I bit equals 1 when the command is reissued before transferring the second word.

During this type of cycle, a coprocessor should not assert data onto the LAD bus until the high-to-low transition of  $\overline{CAS}$ .



Figure 10–4. Transferring from a Coprocessor to a TMS34020 Register



## **10.4.8 Transferring Values from Local Memory to a Coprocessor**

Performed when Passing up to thirty-two 32-bit words from the local memory to a coprocessor. Figure 10–5 shows two words being transferred using page mode.



Status code 0000<sub>2</sub>





operands 1 & 2 Data to the coprocessor

Transferring data from local memory to a coprocessor requires explicit addressing of the memory. Because of this, no data is transferred immediately after the command. After the command cycle completes, the data-transfer sequence begins with another address/status subcycle. During this subcycle, the address of the memory location to be transferred to the coprocessor is output.

This data-transfer sequence is identical to an ordinary data read from memory, except that the status code is  $0000_2$ . A coprocessor should latch the data from the LAD bus on the low-to-high transition of CAS.

If page-mode operation is not supported (by either the coprocessor or the local memory), a complete memory cycle is generated for each word transferred.

A page-mode sequence may also be interrupted by a high-priority memory request. In this case, the data transfer resumes with the next word in memory following the high-priority request. The address of that location is output, along with the coprocessor operation status code; the command is not reissued.

#### 10.4.9 Transferring Values from a Coprocessor to Local Memory

This cycle is used to pass up to thirty-two 32-bit words from a coprocessor to local memory. Figure 10–6 shows 2 words being transferred using page mode.

This cycle requires the memory to be explicitly addressed. No data is transferred immediately following the command. After the command cycle completes, the data transfer sequence begins with another address/status subcycle. During this subcycle, the address of the location in memory to be written to by the coprocessor is output. In this case, however, data is not transferred immediately after the address/status subcycle of the data-transfer sequence. An extra machine state is inserted to separate the address/status and data subcycles of the transfer cycle. This **spacer** subcycle is inserted to ensure that the TMS34020 can set its LAD bus drivers to the high-impedance state before the coprocessor starts to drive data onto the LAD bus. This is necessary because the coprocessor must assert data onto the LAD bus following the high-to-low transition of  $\overline{WE}$  during this type of cycle, so that the LAD bus contains valid data for the memories on the high-to-low transition of  $\overline{CAS}$ .

Note that whereas data is asserted onto the LAD bus after the falling edge of WE during this type of cycle, it **must not** be asserted onto the bus before the falling edge of CAS during a direct coprocessor-to-TMS34020 transfer (see Section 10.4.6). To enable the coprocessor to distinguish between the two and assert data onto the LAD bus at the appropriate time, ALTCH is driven high at the end of the spacer subcycle.

This data-transfer sequence is identical to an ordinary data write to memory, except that the status code is 0000<sub>2</sub>, the spacer subcycle is inserted, and ALTCH is high while data is transferred.

If page-mode operation is not supported (by either the coprocessor or the local memory), a complete memory cycle (including spacer) is generated for each word transferred.

A page-mode sequence may also be interrupted by a high-priority memory request. In this case, the data transfer resumes from the next word in memory following the high-priority request. The address of that location is output, along with the coprocessor operation status code; the command is not reissued.

The CMOVCM instruction causes this type of cycle.





Key: LAD (TMS34020) LAD (memory) operands 1 & 2 Output to the LAD bus by the TMS34020 Output to the LAD bus to the memory Data to the coprocessor

## **10.5 Coprocessor Aborts and Status Checks**

All coprocessors should recognize at least two coprocessor command cycles and provide the appropriate response, even if busy. These cycles would be recognized as:

- abort The abort should terminate all coprocessor activity, restoring the coprocessor to a known state so that it is available for further commands from the TMS34020. The abort allows the TMS34020 to regain control of the coprocessor in the event of a system fault that may involve the coprocessor.
- **status check** The response to the status check command should be to provide information regarding the coprocessor's operating condition. The MSB output in response to the status check should be set high if the coprocessor is busy. Other bits in the response word may be used for other information concerning the coprocessor status. The TMS34020 can use the status check command to determine if a coprocessor is currently busy when multiple tasks are competing for the coprocessor. Thus, the TMS34020 does not have to enter an extended wait state to obtain access to the coprocessor, but may continue with another task not requiring the coprocessor.

The actual encoding of the coprocessor command field for the abort and status check cycles depends on the requirements of the specific coprocessor.

## **10.6 System Configuration**

Some applications execute the same software on two systems—one with a coprocessor and one without. The TMS34020 does not support this directly. However, you can accomplish this in a number of ways by building configuration information into the system hardware, such as a wire jumper or switch, and setting it one way or the other. depending on whether a coprocessor is present. A number of possible approaches can then be taken.

- Depending on the state of the jumper of switch, different values can be returned based on the value at a system-defined memory location. This would allow the software to configure itself to execute code that either did or did not use the coprocessor instructions.
- If the jumper or switch indicated that a coprocessor was not present in the system, external logic could be used to detect memory cycles that output the coprocessor operation status code and to generate a bus fault accordingly. The bus-fault service routine could then determine what type of operation the coprocessor instruction was trying to perform and emulate it in software.

By using more switches or jumpers, you could easily extend these mechanisms for use in systems in which the number of coprocessors varies.

# **Chapter 11**

# **Multiprocessing and System Architecture**

The TMS34020 can share its local-memory resources with other processors by means of a 3-wire interface. This **multiprocessor interface** allows multiple TMS34020s (as well as other processors) to share the same local-memory space. In order to accomplish this, these processors must support a conventional hold/hold-acknowledge protocol. This chapter includes these topics:

	Sect	ion	Page
Basic information includes a	11.1	Related Signals	11-2
review of related signals and an	11.2	Overview	11-2
overview of the multiprocessor interface.	11.3	Basic Multiprocessor System Configuration	11-3
Advanced information discusses	11.4	Protocols for Communicating in	
specific communications protocols		a Multiprocessor System	11-5
and provides several examples.	11.5	Arbitration Logic Requirements	11-13
	11.6	Multiprocessor Arbitration Examples	11-15
	11.7	Initializing Multiple TMS34020s	11-14
	11.8	Configuration with a Host Processor	11-20

#### Advantages of a 3-wire interface

The TMS34020's 3-wire multiprocessor interface provides more flexibility than the conventional 2-wire hold/hold-acknowledge interface supported by many other processors. The  $\overline{R}0$  and  $\overline{R}1$  request pins provide information about how urgently a TMS34020 requires access to the local-memory bus. They continue to provide this information, regardless of whether the TMS34020 is currently controlling the bus or not. This allows external arbitration logic to allocate control of the local-memory bus to the processor most in need.

This chapter uses the following terms:

- Bus master refers to the device that currently controls the local-memory interface. If the bus-master TMS34020 generates a high-priority request code, the arbitration logic can ensure that the TMS34020 maintains control of the bus.
- Bus requestor refers to a device that does not currently control the localmemory interface, but is requesting control. If a bus-requestor TMS34020 generates a high-priority request code, the arbitration logic can take the appropriate steps to ensure that the requestor obtains control of the local-memory bus as soon as possible.

## 11.1 Related Signals

Chapter 2 describes the multiprocessor-interface signals in detail; these signals are summarized below for your convenience.

Signals	Descriptions	I/O
GI	is the grant-input signal. When active low, $\overline{GI}$ informs the TMS34020 that it is the bus master, which means that it must drive the local-memory bus and that it controls the next memory cycle. When driven inactive high, $\overline{GI}$ informs the TMS34020 that it should terminate the current memory access as soon as possible and relinquish bus mastership.	I
R0, R1	form the bus-request code output by the TMS34020. External arbitration logic uses this encoded informa- tion when determining which processor should receive an active-low signal on its GI pin.	0

#### 11.2 Overview

The multiprocessor interface supports a general protocol that can be used with external arbitration logic to form a system of multiple processors sharing a common local-memory space. The TMS34020 provides a mechanism for synchronizing multiple TMS34020s to the same local clock (LCLK) phase. The interface signals are timed to allow multiple, synchronized TMS3020s to share the local-memory bus without wasting memory cycles when passing control from one TMS34020 to another.

When multiple TMS34020s share local memory, performance should increase because of the large internal program cache that can allow long time intervals during program execution where no external memory accesses are required. This memory bandwidth can then be used by another TMS34020.

In theory, there is no limit to the number of devices that can be configured together. However, it is doubtful that performance would increase significantly if a system included more than three TMS34020s. It is likely that the performance difference between one processor and two processors will be greater than the performance difference between two processors and three processors, and so on. Once the local-memory bandwidth is completely used, no performance benefit arises from adding extra processors.

## 11.3 Basic Multiprocessor System Configuration

This section contains general information about connecting and synchronizing multiple processors within a single system.

## **11.3.1 Connecting Multiple Processors Together**

All the local-memory control pins, address pins, and data pins should be wired in parallel between the processors. (This includes LAD0—LAD31, RCA0—RCA12, SF, ALTCH, RAS, CAS0—CAS3, WE, TR/QE, DDIN, and DDOUT.) If your system contains a non-TMS34020 device, wire the equivalent signals common. If you wish, you can wire the local-memory input pins (CAMD, LRDY, BUSERR, PGMD, and SIZE16) common. How you choose to wire HDST and HOE depends on your system architecture and on how you wish to connect a host processor. Section 11.7 describes this in detail.

All TMS34020s in the system should share the same RESET and CLKIN inputs. *Do not* wire the local clock outputs (LCLK1 and LCLK2) together. Use one TMS34020's outputs as the local clocks for the entire system.

## 11.3.2 Synchronizing Multiple TMS34020s at Reset

To allow multiple TMS34020s to share control of the local-memory interface without losing any memory bandwidth, the TMS34020s must be synchronized to the same local clock phase. The TMS34020 achieves this with special internal logic that detects the rising edge of the RESET pin at the end of reset. In a typical single-TMS34020 system, RESET is not required to be synchronous to CLKIN. However, to allow synchronization of multiple TMS34020s in a system, the rising edge of RESET must meet the setup and hold requirements with respect to CLKIN. This ensures that all TMS34020s respond to RESET on the same quarter phase (refer to the *TMS34020 Data Sheet* for details).

Figure 11–1 shows the four possible conditions for the state of the TMS34020 at the time RESET goes high. Within 10 CLKIN cycles after RESET goes high, all TMS34020s will be synchronized to the same quarter phase through the extension of the Q1 phase.



Figure 11–1. Synchronization of Multiple TMS34020s

Note: This figure does not imply timing dependences of LCLK1 and LCLK2 relative to CLKIN or RESET.

Kev: Case 1 Extension of Q1 for 1 quarter phase.

Case 2 Extension of Q1 for 2 quarter phases.

Case 3 Extension of Q1 for 3 quarter phases.

Case 4 Extension of Q1 for 4 quarter phases.

## 11.4 Protocols for Communicating in a Multiprocessor System

This section provides details of how the TMS34020 requests, is granted, and relinquishes control of the local-memory bus and control signals.

#### 11.4.1 How a Processor Requests Control of the Local-Memory Bus

The multiprocessor interface provides encoded information on  $\overline{R}0$  and  $\overline{R}1$ . These codes indicate whether or not the TMS34020 wishes to use the local memory and, if so, how urgently. Table 11–1 describes this coding.

(Q2) of the next LCLK cycle. (DDOUT is set to high impedance during Q1.)

Table 11–1. Bus Request Codes for the Multiprocessor Interface

R0	R1	Code and Description
0	0	<b>High-priority request.</b> The TMS34020 is requesting or using the bus to perform a high-priority memory access. High-priority memory requests are issued for accesses that must be performed urgently. They include
		VRAM serial-register transfer initiated by the video control logic.
		12 or more DRAM refreshes pending.
		Accesses initiated by a host processor.
0	1	<b>Low-priority request.</b> The TMS34020 is requesting or using the bus to perform a low-priority memory access. A low-priority request is an access requested by the TMS34020's CPU or a DRAM refresh (when less than 12 refreshes are pending).
1	0	Access termination. The TMS34020 will terminate the current memory access in this machine state. When performing multiple accesses using page mode, the access-termination code is output during the last data access of the sequence only.
1	1	<b>No request.</b> The TMS34020 does not require access to the local memory in the next machine state. This code is output during host-default states when the next machine state will be another host-default state.
		Section 8.5, <u>Local-Memory Cycle Status Codes</u> (page 8-10), details the differ- ent types of local-memory cycles and their relative priorities; the request code depends on the cycle's priority.
		<b>G</b> Cycles with priorities <b>2</b> through <b>5</b> generate a <b>high-priority</b> request code.
		Cycles with priorities 6 through 8 generate a low-priority request code.
		The host-default cycle can generate either a no-request or low-priority request code.
11.4.2	How	a Processor Releases Control of the Local-Memory Bus
		If external bus-arbitration logic drives the bus-master TMS34020's $\overline{\text{GI}}$ pin inac- tive high, the TMS34020 releases the local-memory bus as soon as possible:
		No-request or access-terminate code. If the TMS34020 is currently out- putting either of these codes, it relinquishes control of the bus and sets all its local bus outputs to high impedance during the second quarter phase

- Low-priority or high-priority request code. If the TMS34020 is currently outputting either of these codes, it terminates the current memory access as soon as possible. This occurs after the current access completes.
  - If the TMS34020 is partially through a sequence of page-mode accesses, the sequence is broken. When bus mastership is regained, the TMS34020 continues from the next address in the sequence.
  - If the TMS34020 is partially through a memory operation that requires multiple memory accesses (such as a read-modify-write or an access to 16-bit memory), the current access completes. When bus mastership is regained, the operation resumes with the next memory access in the sequence.

The current access is allowed to complete in the manner determined by the LRDY and BUSFLT pins. A detailed discussion of the function of these pins is given in Section 8.6, Ending a Local-Memory Cycle (page 8-12).

During the last machine state of any memory access, the TMS34020 issues the access-termination code. If  $\overline{GI}$  is driven inactive at this time, the TMS34020 *will* relinquish control of the local-memory bus at the beginning of the next machine state.

## 11.4.3 Passing Control of the Local-Memory Bus

When bus control is passed from one TMS34020 to another (or to/from another type of device), it is necessary to avoid conflicting driven outputs. However, it is also desirable that the shared local-memory bus and control signals are not left undriven; floating signals are prone to noise and can make the system unreliable. You could use external pull-up resistors to hold the undriven signals at a good voltage level, but this is undesirable because the resulting circuit board is larger and more expensive to produce. The TMS34020's multiprocessor interface allows signals to be driven to the same level by both processors involved in the exchange, avoiding all of these problems. When different TMS34020s pass control of the local-memory bus, the following steps happen automatically:

- Step 1: The TMS34020 that becomes bus master drives the local bus and control pins to their inactive levels from the beginning of the LCLK cycle after its GI pin is asserted.
- Step 2: The TMS34020 that releases control of the bus sets its local bus and control pins to high impedance during Q2 of the LCLK cycle after its GI pin is driven inactive. This provides one quarter phase of overlap where both TMS34020s are driving the signals inactive, and therefore no pull-up resistors are necessary. The only exception to this is DDOUT, which is not driven inactive by the releasing device during Q1, as the acquiring device may drive it active low during Q2.

Depending on inter-TMS34020 clock skew, this may result in a *short* period when DDOUT is undriven. This is preferable to having the pin driven to conflicting levels, however. DDOUT will never be undriven for more than a few nanoseconds, and so external resistors should not be necessary.

If your system contains a non-TMS34020 processor, you must ensure that this processor drives *all* the local-memory control and bus pins from the beginning of the memory cycle when it assumes control of the bus and that it stops driving them at the end of the last memory cycle when it releases control of the bus.

Undriven signals may produce unpredictable or unreliable system operation. Signals driven to different levels simultaneously by more than one device may damage the driving devices as well as cause unreliable system operation.

## **11.4.4 Functional Timing Examples**

The following diagrams show various examples of the TMS34020 obtaining and releasing control of the local-memory bus and control signals. The TMS34020 samples the  $\overline{GI}$  pin on the rising edge of LCLK1 (at the end of Q4).  $\overline{R0}$  and  $\overline{R1}$  change on the falling edge of LCLK1 (at the beginning of the Q3 phase of the machine state). Refer to the *TMS34020 Data Sheet* for the precise timing relationships of these pins.

Figure 11–2 shows the TMS34020 outputting the no-request code and releasing the bus during Q2 of the LCLK cycle, immediately after  $\overline{GI}$  is driven high.





Figure 11–3 is identical to Figure 11–2 except that the TMS34020 is outputting the access-termination code at the end of a low-priority write cycle. Note that  $\overline{GI}$  may be sampled at either level while the low-priority request code is being output. The TMS34020 cannot relinquish control of the local-memory interface until it completes the memory access (at which time it issues the access-termination code).

Figure 11–3. Releasing Control of the Local-Memory Bus and Control Signals (GI Driven High at the End of a Write Cycle)



Figure 11–4 shows the TMS34020 outputting the access-termination code at the end of a (high priority) host read cycle. Note that  $\overline{GI}$  may be sampled at either level while the high-priority request code is being output. The TMS34020 cannot relinquish control of the local-memory interface until it completes the memory access (at which time it issues the access-termination code).





Figure 11–5 shows  $\overline{GI}$  being driven inactive just after the TMS34020 has started a sequence of page mode accesses. The TMS34020 breaks the sequence and issues the access-termination code. If  $\overline{GI}$  had remained active, this code would not have been generated until the sequence completed.

#### Figure 11–5. Releasing Control of the Local-Memory Interface (GI Driven High During a Page-Mode Sequence)



In Figure 11–6, the TMS34020 regains bus mastership as soon as its  $\overline{GI}$  pin is driven active low.  $\overline{R0}$  and  $\overline{R1}$  could be outputting any of the codes with the exception of the access-termination code.

Figure 11–6. Regaining Control of the Local-Memory Bus and Control Signals



## **11.5 Arbitration Logic Requirements**

A multiprocessor system's external arbitration logic must generate the  $\overline{GI}$  outputs to each processor for the next cycle. To do this, it uses each processor's  $\overline{R0}$  and  $\overline{R1}$  outputs in conjunction with the current value of each processors'  $\overline{GI}$  input. The arbitration logic must **never** assert the  $\overline{GI}$  pins of more than one TMS34020 simultaneously, though there may be times when none of the processors has an active  $\overline{GI}$ .

## 11.5.1 Passing Control of the Local-Memory Bus

When passing control of the local-memory bus from one processor to another, the arbitration logic must not assert the acquiring processor's  $\overline{GI}$  pin until the current master processor indicates that it will relinquish control of the local-memory bus at the end of the current machine state. This is the case when the current bus master outputs either the no-request or access-termination code. These are easily identified by looking for an inactive high level on  $\overline{R0}$ .

Figure 11–7 shows a typical example for two TMS34020s.

- TMS34020#1 starts off as the bus master. It finishes using the bus during the fifth machine state (S5), but continues to drive the bus inactive until after TMS34020#2 requests the bus in the eighth machine state (S8).
- As TMS34020#1 outputs the no-request code during S8, the GI pins of both TMS34020s can be changed simultaneously. TMS34020#2 then becomes bus master in S9. However, a high-priority request (in this case, for a screen-refresh cycle) made by TMS34020#1 during S11 causes TMS34020#2 to relinquish control of the bus at the end of S12. Note that because TMS34020#2 was busy when TMS34020#1 first generated the high-priority request, its GI is not asserted until TMS34020#2 outputs the access-termination code.
- TMS34020#2 continues to output a low-priority request and resumes control of the bus (at the beginning of S15) as soon as TMS34020#1 finishes and asserts the access-termination code.

Note that although TMS34020 #2 is making a request, TMS34020 #1's grant-in is not removed, because it is performing a high-priority request. Also note that no bus cycles are lost.  $\overline{R}0$  and  $\overline{R}1$  change sufficiently early for the  $\overline{GI}$  signals to change and allow back-to-back local-memory cycles from different TMS34020s.



Figure 11–7. Passing Control of the Local-Memory Between Two TMS34020s



## 11.5.2 Wait States, Retries, and High-Priority Bus Requests

As previously indicated, if the TMS34020 is performing a memory cycle extended with wait states when  $\overline{GI}$  is driven inactive, the TMS34020 will not relinquish control of the local-memory interface until the access completes. This could consume many machine states. (For instance, the TMS34020 could be accessing a slow peripheral or possibly a location in the host's memory via the system bus.)

This could cause a problem if one of the TMS34020s has a limited time in which to perform high-priority accesses. For example, if the horizontal-blanking interval is very short, the screen-refresh cycle scheduled at this time must be serviced promptly (before the end of the blanking interval).

To circumvent this sort of problem, the arbitration logic can control LRDY and BUSFLT to cause a retry of the slow, wait-stated access. If a retry code is detected (LRDY low, BUSFLT high), the memory cycle will terminate (issuing the access-termination code) and subsequently restart. Once the access-termination code is issued, however, the arbitration logic can assign control of the local-memory interface to another TMS34020.

If you wish to do this, you should ensure that the accessed memory location can respond correctly to a retry. For example, a peripheral that increments an internal address pointer when a cycle completes would need to be able to distinguish a successful transfer from a retry, preventing the internal pointer from being incremented after the retried access.

## **11.6 Multiprocessor Arbitration Examples**

This section provides two examples of arbitration schemes; these examples should help you to understand the multiprocessor interface.

- The first example shows a typical scheme for two TMS34020s.
- The second example shows a scheme for a TMS34020 and another (non-TMS34020) device that uses a hold/hold-acknowledge protocol.

## 11.6.1 Arbitration Scheme for Two TMS34020s

This example describes a protocol between two TMS34020s, *TMS34020<sub>A</sub>* and *TMS34020<sub>B</sub>*. Assume that only TMS34020<sub>A</sub> can make high-priority requests; only these requests can cause TMS34020<sub>B</sub>'s  $\overline{GI}$  signal ( $\overline{GI}_B$ ) to be driven inactive when TMS34020<sub>B</sub> is not outputting the no-request or access-termination code. In this way, only high-priority requests interrupt page-mode bursts.

Table 11–2 shows how the arbitration logic allocates active or inactive levels for the next machine state on the  $\overline{GI}$  pins of the two TMS34020s ( $\overline{GI}_A$  and  $\overline{GI}_B$ , respectively). The values of the  $\overline{GI}$  pins for the next machine state are sampled on the rising edge of LCLK1, at the beginning of the next machine state.

The decision is based on the present state of the bus request outputs from the two TMS34020s ( $\overline{R}0_A$ ,  $\overline{R}1_A$ ,  $\overline{R}0_B$ , and  $\overline{R}1_B$ ) and the present state of the  $\overline{GI}$  outputs ( $p\overline{GI}_A$  and  $p\overline{GI}_B$ ). The bus-request pins become valid shortly after the falling edge of LCLK1, at the beginning of the Q3 phase of the TMS34020s' machine states. Assume that the  $p\overline{GI}_A$  and  $p\overline{GI}_B$  values are latched externally on the rising edge of LCLK1.

The top and left side of Table 11–2 show the possible combinations of  $\overline{R}0_A$ ,  $\overline{R}1_A$ ,  $p\overline{GI}_A$ , and  $\overline{R}0_B$ ,  $\overline{R}1_B$ ,  $p\overline{GI}_B$ . The resulting grid shows which  $\overline{GI}$  (if any) is asserted for the next machine state, given the current state.

Table 11–2. Arbitration Scheme for Two TMS3402020s

				TMS34020 <sub>A</sub>								
				pGIA	1	1	1	1	0	0	0	0
				ROA	1	0	0	1	1	0	0	1
	p <del>GI</del> B	R0 <sub>B</sub>	R1 <sub>B</sub>	R1 <sub>A</sub>	1	1	0	0	1	1	0	0
اھ	1	1	1		_			_	Α	Α	Α	Α
12	1	0	1				Х		В	А	Α	В
34	1	1	0				—	—				
MS	0	1	1		В	Α	Α		_	_		
	0	0	1	-	в	В	Х					
	0	1	0		В	Α	Α	—				—

**<u>Key:</u>** A Bus allocated to TMS34020<sub>A</sub> (assert  $\overline{GI}_A$ )

- **B** Bus allocated to TMS34020<sub>B</sub> (assert  $\overline{GI}_B$ )
- Illegal combination
- **X** No allocation (wait for cycle termination)

Table 11–2 can be summarized as follows:

- Active  $\overline{GI}$  is made to TMS34020<sub>A</sub> ( $\overline{GI}_A$ ) if
  - Active GI was already allocated to TMS34020<sub>A</sub> and there is no request from TMS34020<sub>B</sub>.
  - Active GI was already allocated to TMS34020<sub>A</sub> and both devices are requesting access (high or low priority).
  - TMS34020<sub>A</sub> is requesting an access (high or low priority) and TMS34020<sub>B</sub> is not requesting access or is terminating an access.
- Active  $\overline{\text{GI}}$  is made to TMS34020<sub>B</sub> ( $\overline{\text{GI}}_{\text{B}}$ ) if
  - Active GI was already allocated to TMS34020<sub>B</sub> and there is no request from TMS34020<sub>A</sub>.
  - Active GI was already allocated to TMS34020<sub>B</sub> and both devices are requesting a low-priority access.
  - TMS34020<sub>B</sub> is requesting a low-priority access and TMS34020<sub>A</sub> is not requesting access or is terminating an access.
- ❑ Active GI is made to *neither* TMS34020<sub>A</sub> or TMS34020<sub>B</sub> (GI<sub>X</sub>) if TMS34020<sub>A</sub> requests a high-priority access but TMS34020<sub>B</sub> has not yet terminated its access.

The following equations express these conditions.

$$\overline{GI}_{A} = (p\overline{GI}_{A} \cdot R0_{B} \cdot R1_{B}) + (p\overline{GI}_{A} \cdot \overline{R}0_{A}) + (p\overline{GI}_{B} \cdot R0_{B} \cdot \overline{R}0_{A})$$

$$\overline{GI}_{B} = (p\overline{GI}_{B} \cdot R0_{A} \cdot R1_{A}) + (p\overline{GI}_{B} \cdot \overline{R}0_{B} \cdot R1_{A}) + (p\overline{GI}_{A} \cdot \overline{R}0_{B} \cdot R0_{A})$$

$$\overline{GI}_{X} = (pGI_{A} \cdot \overline{R}0_{A} \cdot \overline{R}1_{A} \cdot \overline{R}0_{B})$$

You could easily expand this protocol to include additional TMS34020s by adding more slave devices, each with a successively lower priority.

#### 11.6.2 Arbitration Scheme for One TMS34020 and a Hold Device

If a multiprocessor system contains a non-TMS34020 device that supports a hold/hold-acknowledge protocol (known as a **hold device**), you can use a modified version of the arbitration scheme described in Section 11.6.1. In the modified version, the hold device replaces the slave TMS34020; however, the non-TMS34020 processor's priority is raised so that hold has a higher priority than a low-priority request from the bus-master TMS34020.

Table 11–3 shows which  $\overline{GI}$  is activated for a given set of inputs in this situation. The single TMS34020's signals have the suffix T.  $\overline{R}0_B$  and  $\overline{R}1_B$  are replaced by a single signal, HOLD. Bus mastership is allocated to the hold device through the  $\overline{GI}_H$  signal.

Table 11–3.	Arbitration Scheme for	One TMS3402020	and a Hold Device

			TMS34020								
			pGIT	1	1	1	1	0	0	0	0
			R0 <sub>T</sub>	1	0	0	1	1	0	0	1
e	pGIH	Hold	R1 <sub>T</sub>	1	1	0	0	1	1	0	0
evi	1	0			_			Т	Т	Т	Т
	1	1			Х	Х		Н	Х	Т	Н
ŏ	0	0		Т	Т	Т	_	_	<u> </u>	—	
ן בן	0	1		Н	Н	Х	—	_		—	

Key: T Bus allocated to TMS34020 (assert GIT)

- H Bus allocated to the hold device (assert GIH)
- Illegal combination
- X No allocation (wait for cycle termination)

Table 11–3 can be summarized as follows:

Active  $\overline{GI}$  is made to the TMS34020 ( $\overline{GI}_T$ ) if

- Active GI was already allocated to the TMS34020 and there is no request from the hold device.
- Active GI was already allocated to the TMS34020 and the TMS34020 is requesting a high-priority access.
- The TMS34020 is requesting an access (high or low priority) and the hold device is not requesting an access.

- Active  $\overline{GI}$  is made to the hold device  $(\overline{GI}_{H})$  if
  - Active GI was already allocated to the hold device and the TMS34020 is not requesting an access.
  - Active GI was already allocated to the hold device and the devices are requesting hold and low-priority accesses, respectively.
  - The hold device is requesting an access, and the TMS34020 is not making an access or is terminating an access.
- Active  $\overline{GI}$  is made to *neither* device ( $\overline{GI}_X$ ) if
  - The TMS34020 is requesting a high-priority access but the hold device has not yet terminated its access.
  - The hold device is requesting an access but the TMS34020 has not yet terminated its low-priority access.

The following equations express these conditions:

- $\overline{GI}_{T} = (p\overline{GI}_{T} \cdot \overline{HOLD}) + (p\overline{GI}_{T} \cdot \overline{R}0_{T} \cdot \overline{R}1_{T}) + (p\overline{GI}_{H} \cdot \overline{HOLD})$
- $\overline{\text{GI}}_{\text{H}} = (p\overline{\text{GI}}_{\text{H}} \cdot \text{HOLD} \cdot \text{R1}_{\text{T}}) + (p\overline{\text{GI}}_{\text{T}} \cdot \text{HOLD} \cdot \text{R0}_{\text{T}})$
- $\overline{\text{GI}}_{X} = (\text{pGI}_{T} \cdot \overline{\text{R}}_{T} \cdot \overline{\text{R}}_{T} \cdot \text{HOLD}) + (\text{p}\overline{\text{GI}}_{H} \cdot \overline{\text{R}}_{T} \cdot \overline{\text{R}}_{T} \cdot \text{HOLD})$

Note that  $\overline{GI}_{H}$  is equivalent to hold-acknowledge in this scheme. HOLD is effectively treated like an  $\overline{R}0$ . If back-to-back memory cycles between master and hold device are required, then HOLD going inactive should be an early warning that the hold device has finished with the bus. If this is not possible, then there will be a dead cycle when neither processor is using the bus when passing control.

Again, this protocol can be easily expanded to include more processors: more hold devices, more TMS34020s, or more of both.

## 11.7 Initializing Multiple TMS34020s

In a single-TMS34020 system, power-up initialization of the device can be performed in one of two ways.

- **Method 1:** Store an initialization routine in ROM and bring the TMS34020 out of reset in the self-bootstrap mode. The TMS34020 will immediately start executing the code from ROM.
- **Method 2:** Bring the TMS34020 out of reset in the host-present mode. In this situation, the TMS34020 is halted so that a host can download code to local memory before clearing the HLT[[HSTCTLH]] bit and allowing the TMS34020 to execute the code.

In a multiple-TMS34020 system, initialization is more complex. You can use method 1, method 2, or a combination of both, with some modification to the single-TMS34020 approach. Here are three alternatives:

- ❑ When operating in self-bootstrap mode, all the TMS34020s will fetch the same code from the same reset vector unless you provide additional mapping to point each processor to a different code. A mapper at the reset vector address can, when read, output a different address, depending on which GI pin is active at the time. In this way, each processor reads a different value from the reset vector and therefore fetches code from different locations.
- Alternatively, you can prevent each TMS34020 from fetching the reset vector and executing code by using the multiprocessor interface to inhibit assertion of each TMS34020's GI pin. If each TMS34020 becomes bus master in a predefined order at power-up, each can modify the reset vector after loading its own code so that the vector points to the next processor's code.
- You can use a host processor to download code to all the TMS34020s. If you do this, be careful when using host-present mode; the host cannot access the I/O registers of a TMS34020 to which it is not directly attached. This means that if a TMS34020 is powered up halted in host-present mode but is not directly connected to a host, there is no way to clear the HLT bit except through RESET. The next section outlines the ways in which a multiple-TMS34020 system can be connected to a host.

## 11.8 Configuration with a Host Processor

If your multiple-TMS34020 system contains a host processor, there are a number of ways to configure the system. The configuration you choose depends on the type of accesses the host needs to make and on the cost of the system. For a general description of host communications, refer to Chapter 7; Section 7.11, <u>Systems with Multiple TMS34020s</u> (page 7-40), discusses the implications of the different configurations in detail.

The HOE and HDST pins support several different configurations.

- Host connected to one TMS34020 only. In this case, the HOE and HDST pins of the host-connected TMS34020 should be wired to the external bidirectional data transceivers. There is no need to wire the HOE and HDST pins of the other TMS34020s in parallel.
- Host connected to more than one TMS34020 via one set of shared data transceivers. In this case, the HOE pins from all TMS34020s sharing the data transceivers should be wired together, as should the HDST pins.
- ❑ Host connected to more than one TMS34020 via individual sets of data transceivers for each TMS34020. In this case, each TMS34020 is connected to the host via its own set of data transceivers. The HDST and HOE pins should be connected between each TMS34020 and its associated transceivers only.

When bus mastership is regained, HDST and HOE are set to high impedance just like the local-memory control signals. However, unlike the local-memory control signals, they both are attached to internal pull-up resistors. This allows any of the above configurations to be implemented without requiring additional external control circuitry; if HDST and HOE are not driven by another TMS34020 (the first and third cases described above) they are held in the logichigh state by the resistors. Yet, the resistors are sufficiently large that they will not prevent another TMS34020 from driving HDST and HOE subsequently (the second case described above).

# **Graphics Instructions and Operations**

.....

The TMS34020 is especially useful in graphics systems because it has a large set of graphics-specific instructions. In addition, the TMS34020 supports a variety of special operations (such as windowing, transparency, and options for combining pixels) that you can apply to graphics instructions. This chapter contains four types of information:

	Sectio	on Page
<b>Overviews</b> summarize the instructions and operations.	12.1 12.2	An Overview of Graphics Instructions 12-2 An Overview of Graphics Operations 12-3
Instruction-specific information describes the special capabilities of the various types of graphics instructions.	12.3 12.4 12.5 12.6	Single-Pixel Instructions12-6Line Instructions12-7Array Instructions12-8Auxiliary Graphics Instructions12-17
<b>Operation-specific information</b> describes the graphics operations and how they work for the different types of graphics instructions.	12.7 12.8 12.9 12.10	Windowing12-19Pixel Processing12-27Transparency12-36Plane Masking12-39
<b>Related topics</b> discuss the ways that graphics instructions work.	12.11 12.12	Setting Up the Implied Operands for Graphics Instructions 12-43 Converting an XY Address to a Linear Address 12-47

## 12.1 An Overview of Graphics Instructions

The TMS34020 instruction set supports several fundamental graphics drawing instructions. Table 12–1 lists these instructions by categories.

Table 12–1. Summary of Graphics Instructions

	Instructio	ns	Refer	to					
jle-Pixel ructions	These ins DRAV	AV (draw-and-advance) draws a single pixel at a specified address and then incre- ments that address. This is useful for implementing incremental algorithms that draw circles, ellipses, arcs, and other curves.							
Sing	PIXT	(pixel transfer) transfers individual pixels from one location to another. There are several different versions of the PIXT instruction that use different address- ing modes.							
SC	These ins	tructions draw single lines or set up the implied operands for drawing a line.	nade	12-7					
ruction	LINE	LINE implements the inner loop of Bresenham's algorithm for drawing lines. For this instruction, a line's endpoints are identified by XY addresses.							
ne Inst	FLINE	is a faster implementation of the LINE instruction. For this instruction, a line's endpoints are identified by linear addresses.							
Ē	LINIT	initializes implied operands required by the LINE and FLINE instructions.							
suc	These ins	nade	12-8						
structio	PIXBLT	(pixel block transfer) copies a 2-dimensional block of pixels from one area in memory to another. There are several forms of the PIXBLT instruction.	puge	12 0					
۸ Ins	FILL	sets all the pixels in a pixel array to a known value.							
rray	PFILL	fills a block with the pattern from the PATTERN register.							
el-A	VFILL	special fill that takes advantage of VRAM features.							
Pix	VBLT	special block transfer that takes advantage of VRAM features.							
	The TMS: base set c	34020 supports several auxiliary instructions that extend the functions of the of graphics instructions	page	9 12-17					
Su	FPIXEQ	searches for a pixel that is equal to the COLOR0 pixel.							
:tio	FPIXNE	searches for a pixel that is not equal to COLOR0.							
atrue		FPIXEQ and FPIXNE are useful for seed fills and data compression.							
' Ins	EXGPS	exchanges value in PSIZE with the value in a general-purpose register.							
iary	GETPS	moves the value in PSIZE into a general-purpose register.							
lixu	RPIX	replicates a pixel value within a general-purpose register.							
A	CLIP	clips an array to fit within specified window dimensions.							
	CPW	compares a point to the window limits in WSTART and WEND.							
	TFILL	fills a trapezoidal block by filling a series of horizontal lines.							

## 12.2 An Overview of Graphics Operations

Table 12–2 summarizes the operations that you can apply to graphics instructions. These operations extend the functionality of the graphics instructions, providing you with explicit control over where a pixel is drawn and how it is combined with other pixels. Note that these operations affect only pixel transfers they do not affect data transfers by nongraphics instructions (such as MOVE).

Table 12–2. Summary of Graphics Operations

	Description	Refer to
бu	The TMS34020's special window-checking hardware compares a pixel's XY address to the XY addresses within a defined rectangular window. The TMS34020 supports four window-checking modes:	page 12-19
scki	No window checking—you can ignore window-checking information.	
ow Che	Window hit detection draws no pixels, but generates an interrupt if a pixel lies inside a window. This supports selection of screen objects that a cursor is pointing to.	
Wind	Window miss detection draws pixels that lie within a window and generates an inter- rupt if a pixel lies outside the window. This is useful for line clipping.	
	Window clipping mode draws only pixels that lie within a window; it generates no inter- rupts.	
Processing	Most of the graphics instructions move pixels to locations in memory; some instructions combine source pixels with pixels that are already at the specified destination address. The pixel-processing options provide you with control over how the source and destination pixels are combined.	page 12-27
Pixel	You can choose from 16 Boolean and 6 arithmetic pixel-processing options. The default option is simply to replace the destination pixels with the source pixels.	
cy	Sometimes you may want a background or an existing object to show through an object that you're drawing to the screen. The TMS34020 supports three modes for choosing which of the pixels in an object are transparent:	page 12-36
sparen	Transparency on result = 0 inhibits the pixel write if zero will be written to the destination address.	
Trans	Transparency on source = COLOR0 inhibits the pixel write if the source pixel is the same as the COLOR0 pixel.	
	Transparency on destination = COLOR0 inhibits the pixel write if the original destina- tion pixel is the same as the COLOR0 pixel.	
Plane Masking	The plane mask is a register that you can load with a mask value. The 1s and 0s in the mask tell the TMS34020 which bits in a source or destination pixel can be modified. Transparency and plane masking are similar but work at different levels. Transparency works at the pixel level, preventing an instruction from writing entire pixels. Plane masking, however, works at the bit level, preventing an instruction from writing individual bits within a pixel.	page 12-39
XY Addressing

Table TZ = Z. Summary of Graphics Operations (Communed)	Table 12–2.	Summar	v of Graphics	Operations	(Continued)
---------------------------------------------------------	-------------	--------	---------------	------------	-------------

Description	Refer to
XY addressing is not an operation in the sense of transparency or pixe ever, using XY addresses instead of linear addresses can simplify Many of the graphics instructions allow you to choose whether you addresses.	l processing; how- / program design. J use XY or linear
Because XY addressing allows you to specify an address in terms of nates, XY addressing is especially useful for manipulating 2-dimen- and for writing pixel data to a screen. For certain instructions, the TM cally converts XY addresses to linear addresses for internal use.	f Cartesian coordi- sional pixel arrays IS34020 automati-

Figure 12–1 shows how these operations interact when the TMS34020 transfers a source pixel to a destination location, combining the source pixel data with the destination pixel data. This is a general illustration; certain operations are not performed if they are not enabled or selected.

As Figure 12–1 shows, source and destination pixels are first read from memory and modified by the plane mask. Pixel processing is then performed on the modified pixel values. The plane mask is applied to the result. Bits that are 1s in the PMASK are read as 0s from the source array (however, these same bits are protected during a pixel write), Transparency is applied, depending on the mode chosen.



Figure 12-1. Graphics Operations Interaction

- Notes: 1) You can enable only 1 of the 3 transparency modes at any time.
  - 2) If you're using the pixel-processing replace option, the TMS34020 does not read the destination pixel *unless* you've enabled transparency-on-destination or plane masking.

# **12.3 Single-Pixel Instructions**

### **Definition**

A single pixel is defined by

- its **address**, which can be a linear address (this is an absolute memory address) or an XY address. Linear addresses must be aligned on pixel boundaries.
- its **size** (in bits), which is defined in the PSIZE register.

### Instructions<sup>-</sup>

Two types of instructions draw single pixels.

The **PIXT instructions** transfer individual pixels from a source location to a destination location. The PIXTs support three combinations of source-to-destination transfers.

source	destination
A- or B-file register	memory
memory	A- or B-file register
memory	memory

The DRAV instruction draws a pixel that is identified by an XY address. After drawing the pixel, DRAV advances the address by adding an XY increment to the original XY address. The value written to the destination pixel is the value in the COLOR1 register.

DRAV is especially useful for drawing circles, ellipses, arcs, and other curves. Selecting an appropriate XY increment allows an incremental algorithm to plot each pixel on a curve and then determine where the next pixel will be drawn. The next pixel is usually one of the 8 pixels immediately adjacent to the pixel just drawn.

You can also use DRAV for drawing straight lines; however, you can draw lines more efficiently with LINE or FLINE.

### Graphics operations

You can use the following graphics operations with single-pixel instructions.

window checking	You can use this when the destination pixel has an XY address.
transparency	Use any of the transparency modes.
plane masking	You can set the plane mask.
pixel processing	Use any pixel-processing option, but remember that the arith- metic options are valid only when the pixel size is at least 2 bits.

# **12.4 Line Instructions**

# **Definition** •

A line is defined by these implied operands:

- address of one of the line's endpoints
- magnitude of the major and minor axes of the line
- signs of the X and Y increments
- initial value of a decision variable
- initial value of a count variable that monitors drawing progress
- pixel values (in the COLOR0 and COLOR1 registers)
- **pixel size (in the PSIZE register)**
- **pattern** (in the PATTERN register)

## Instructions

- The LINE instruction implements the inner loop of Bresenham's linedrawing algorithm. The endpoint that you provide for LINE must be an XY address. When drawing a line, you might typically
  - Step 1: Determine the XY coordinates of the line's endpoints.
  - Step 2: Calculate the implied operands listed above. For the most efficient operation, clip the line to the dimensions of the display window before you calculate the implied operands.
  - Step 3: Draw the line with the LINE instruction.
- The FLINE instruction is a faster implementation of Bresenham's linedrawing routine. FLINE operates similarly to LINE, but the endpoint that you provide for FLINE must be a linear address. Because you cannot use XY addressing with FLINE, FLINE does not support window checking.
- ❑ The LINIT instruction initializes the implied operands that the LINE and FLINE instructions use. LINIT provides you with a quick, simple method for performing the initialization; LINIT uses a line's XY endpoints to set up the necessary B-file registers. LINIT also sets the status bits so that you can easily determine a line's location in relation to the current window, and you can determine whether a line can be trivially rejected.

### **Graphics operations**

You can use the following graphics operations with the LINE and FLINE instructions.

window checking	You can use window checking with LINE; you can't use window checking with FLINE—FLINE doesn't support XY addressing.
transparency	Use any of the transparency modes.
plane masking	You can set the plane mask.
pixel processing	Use any pixel-processing option, but remember that the arithme- tic options are valid only when the pixel size is at least 2 bits.

# **12.5 Pixel-Array Instructions**

### **Definition** -

	A 2	2-dimensional pixel array is defined by these implied operands:
		<b>Starting pixel address</b> . This is the address of the pixel with the lowest address in the array (in a display, this is usually the pixel at the top left of the array). Two registers supply starting array addresses; these addresses can be linear addresses or XY addresses.
		SADDR (B0) identifies the starting address of a <i>source array</i> .
		DADDR (B2) identifies the starting address of a <i>destination array</i> .
		<b>Array width</b> (DX) and <b>array height</b> (DY) in pixels (defined in the DYDX register). The 16 LSBs of DYDX identify the array width; the 16 MSBs identify the height. Source and destination arrays use the same DYDX value.
	C.	Array pitch (the difference between the linear addresses of any two verti- cally adjacent pixels in an array). Two registers supply array pitch values.
		SPTCH (B1) identifies the pitch of a <i>source array</i> .
		DPTCH (B3) identifies the pitch of a <i>destination array.</i>
	Q	The <b>pixel size</b> is defined in the PSIZE register. Only one pixel size is valid at a time, defining the size of all pixels in the array.
Instructions		
	Th	e TMS34020 supports two types of array instructions.

□ The PIXBLT instructions are a powerful set of raster operations that combine the pixels in a source array with the pixels in a destination array. The TMS34020 copies each pixel in the source array to the corresponding location in the destination array. The source pixel can simply replace the destination pixel, or the two can be combined using any pixel-processing option. The source and destination arrays must have the same width and height, but they can have different pitches.

The TMS34020 supports several PIXBLT instructions, with variations in the types of addresses that define the source and destination arrays.

source array	destination array	instructions	
binary	linear	PIXBLT B, L	VBLT
binary	XY	PIXBLT B, XY	
linear	linear	PIXBLT L,L	PIXBLT L,M,L
linear	XY	PIXBLT L,XY	
XY	linear	PIXBLT XY,L	
XY	XY	PIXBLT XY,XY	

Graphics Instructions and Operations

□ The **FILL instructions** fill a destination pixel array with the pixel value in the COLOR1 register. You can think of a FILL operation as a special type of PIXBLT that does not use a source array.

The TMS34020 supports two FILL instructions and two special-purpose FILLs, PFILL and VFILL. PFILL is a pattern-fill instruction, and VFILL takes advantage of VRAM block-mode accesses.

destination array	instructions	
linear	FILL L	VFILL
XY	FILL XY	PFILL XY

## Graphics operations ·

You can use the following graphics operations with pixel-array instructions.

window checking	You is io	can use window checking whenever the destination array lentified by an XY address.
transparency		You can use transparency with any array instruction except VBLT or VFILL
		The binary PIXBLTs do not apply transparency to unexpanded source data.
plane masking	<b>D</b>	You can use the plane mask with any array instruction. Note that for VBLT and VFILL, the masking comes from the VRAM write mask and not from the PMASK register.
	<b>D</b>	The binary PIXBLTs don't apply the plane mask to the un- expanded source data.
pixel processing		Use any of the pixel-processing options with the basic PIXBLTs.
		You cannot use pixel processing with VBLT or VFILL.

## 12.5.1 PIXBLTs with XY and Linear Addressing

Some PIXBLTs allow you to specify the source and destination arrays' start addresses in XY or linear format. You can use the following combinations of XY and linear arrays.

Source Array	Destination Array	Instruction	
Linear	$\rightarrow$ Linear	PIXBLT L, L	ر
Linear	$\rightarrow$ XY	PIXBLT L, XY	
XY	🛶 Linear	PIXBLT XY, L	
XY	→ XY	PIXBLT XY, XY	

## 12.5.1.1 Packed Linear Arrays

XY addressing provides a convenient method for manipulating on-screen objects. XY addressing supports both automatic window clipping and conversion of Cartesian coordinates to memory addresses. When you store pixel arrays off screen, however, you may use memory more efficiently if you use linear addressing. For example, you can pack successive rows of a linear pixel array into memory without gaps between the end of one row and the beginning of the next. The pitch of a linear array is set to the product of the array width (in pixels) and the pixel size (in bits). Figure 12–2 shows how storing an array in packed linear format can save memory space.

Figure 12–2. How XY and Linear Arrays are Stored in Off-Screen Memory



To store an XY array in packed linear form, use the PIXBLT XY, L instruction; to move a packed linear array onto a screen, use PIXBLT L,XY.

## 12.5.1.2 Selecting the Starting Corner for a PIXBLT

A PIXBLT's default starting address is the lowest pixel address in the source array. As Figure 12–3 shows, however, some PIXBLTs can start at any of the array's four corners. If the source and destination arrays overlap, it may be necessary to change a PIXBLT's starting corner.

Table 12–3 lists the PIXBLTs that can use this corner-adjust feature; note that for some PIXBLTs, the TMS34020 automatically adjusts the corner—for others, you must manually adjust the corner.

TADIO 12-0. L'INDETS THAL VAIT SLATLINUT ATTY VUITE	Table 12-3.	PIXBLTs	That Can	Start from	Any Corne
-----------------------------------------------------	-------------	---------	----------	------------	-----------

Automatic adjust	Manual adjust	
PIXBLT L, XY	PIXBLT L, L	
PIXBLT XY, L	PIXBLT L, M, L	
PIXBLT XY, XY		

By default, a PIXBLT first processes the pixel with the lowest address. The PIXBLT then processes the remaining pixels from left to right within each row, beginning at the top row and moving toward the bottom row. The pixel at the lower right corner of the array is processed last.

When you manually adjust the PIXBLT's starting corner, you must control the sequence in which pixels are moved when the arrays overlap, so that destination pixels do not write over source pixels that haven't been transferred yet.

Figure 12–3. Possible Starting Corners



As Figure 12–3 shows, the PBV[[CONTROL]] and PBH[[CONTROL]] bits determine the starting corner for the PIXBLT. PBH controls horizontal movement; PBV controls vertical movement.

- **PBH=0** Pixels are processed from **left to right** (increasing X direction).
- **PBH=1** Pixels are processed from right to left (decreasing X direction).
- **PBV=0** Rows are processed from top to bottom (increasing Y direction).

**PBV=1** Rows are processed from **bottom to top** (decreasing Y direction).

A PIXBLT processes all the pixels in one row before moving to the next row.

You supply the arrays' default starting addresses in the SADDR and DADDR registers. The starting corner adjustment is automatic or manual, depending on the types of addresses that define the arrays.

When one or both arrays are XY arrays, the TMS34020 automatically calculates the actual starting address identified by PBH and PBV from the default starting address and the array size. As the PIXBLT executes, it automatically adjusts SADDR and DADDR to the address of the corner selected by PBH and PBV. ❑ When you use an alternate starting corner with the PIXBLT L,L or PIXBLT L,M,L instructions, you must adjust the addresses in SADDR and DADDR to the corner selected by PBH and PBV.

## 12.5.2 Binary (Color Expanding) PIXBLTs

A **binary array** is a 2-dimensional array of 1-bit pixels. You can use a binary array for storing an object's shape information separately from attributes such as color and intensity. The shape is stored in compressed form as a bitmap of 1s and 0s. The color information is added as a PIXBLT draws the shape on the screen: the 1s in the bitmap are expanded to the COLOR1 value, and the 0s are expanded to the COLOR0 value.

The TMS34020 supports two instructions that allow you to expand a binary array to a linear or an XY array. These PIXBLTs are called **binary PIXBLTs** or **color-expanding PIXBLTs**.

Source Array		Destination Array	Instruction				
Binary Binary	$\stackrel{\rightarrow}{\rightarrow}$	Linear XY	PIXBLT B, L PIXBLT B, XY				

Binary PIXBLTs provide several benefits.

- You can easily change foreground and background colors by changing the values in the COLOR0 and COLOR1 registers; this doesn't change the source array.
- When the destination pixel size is greater than 1 bit, some applications (such as storing fonts off-screen in a binary format) require less memory than you would need in a nonbinary format.
- □ Font loading also takes less time—binary PIXBLTs execute quickly because there is less source data to read.

#### Note:

Binary PIXBLTs expand the source data before performing any pixel processing, transparency, or plane masking. They do not use the plane mask when reading the source data.

Binary PIXBLTs are especially useful in applications that use bitmapped text. Using the XY and linear PIXBLTs to draw and store bitmapped text is cumbersome, requiring you to store a complete copy of each font in off-screen memory. This could consume large amounts of memory. If, for example, you needed each font in several colors, you would need multiple copies of each font (one copy for each color). Binary PIXBLTs provide a better solution for this application.Typically, text characters appear as a foreground color drawn on top of a background color. Each text character can be stored in a binary bitmap where 1s represent a foreground color (COLOR1) and 0s represent the background color (COLOR0). This way, you don't need a full bitmap representation of each character (where each foreground and background pixel would require several bits of storage). Figure 12–4 shows a sample character that is stored in a binary array and then expanded using COLOR0 and COLOR1.

Figure 12–4. An Example of the Color-Expand Operation

	c c	OL OL	OR OR	10 11	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0		0	0	0	0	0	0	0	0	T T	<u>)</u> 1	0	0	0	0	0	0	0	] ]	
1	rec	Bi	nar es (	<b>y I</b> 25 l	No	ote	9: F	'ix	el	si	ze	is	. 4	b		•							re	E	xţ	bai es	nd 1	lec 00	b b	ts										 _
	1 0 0 1	1 0 0 1	1 1 1 1 1	1 0 0 1	1 0 0 1											1 C C 1	1 0 0 0 0		1 0 0 0 0 0 1	1 C C 1	1   0   0   1		5 · 5 ( 5 ( 5 (	1 0 0 1	1 1 1 1	1 1 1 1 1	0 0 0 0 0	1 1 1 1	1 C C 1	1 ) () ) () ) () () ()			1 5 5 7 1	1 0 0 1	1 0 0 0 1	0 0 0 0	1 0 0 1			

Using binary PIXBLTs, you might follow these steps to write text to the display:

- **Step 1:** Load the text font into memory as one table of binary bitmaps in packed, linear form. (Array pitch = DX × pixel size; for binary arrays, the pixel size = 1, so the pitch for a binary array = DX.)
- **Step 2:** Let COLOR0 contain the background color and COLOR1 contain the foreground color.
- **Step 3:** Use a binary PIXBLT, which will copy the character from memory and expand it onto the display bitmap.

As each character is drawn to the screen, the source array is set to point to the appropriate letter in the binary table. The destination array is defined as the next free space on the screen.

## Note:

Set the PSIZE register to the size of the *expanded* destination pixels. Binary PIXBLTs always treat source pixels as 1-bit pixels.

The expand function is also useful in applications that generate shapes or patterns dynamically. During the first stage of this process, a compressed image is constructed in an off-screen buffer area at 1 bit per pixel. The image is built up of geometric objects such as rectangles, circles, or polygons. Patterns can also be added. When complete, the compressed image is color-expanded onto the screen. This method defers the application of color until the final stage.

### 12.5.3 Masked PIXBLT

The masked PIXBLT (PIXBLT L,M,L) is an extension of the PIXBLT L,L instruction. The masked PIXBLT uses a **mask array**, a 1-bit-per-pixel array that determines on a pixel-by-pixel basis whether the particular destination pixel is overwritten. In effect, the mask array controls which pixels are forced to be treated as transparent, but this occurs independently of transparency.

The TMS34020 expands the bits in the mask array, then combines them with the plane mask and transparency detection. This combination provides bit-bybit control of which destination bits are protected from overwriting.

The masked PIXBLT uses two B-file registers:

- MADDR (B10) identifies the mask array's starting address.
- MPTCH (B11) identifies the mask array's pitch.

The masked PIXBLT is useful for manipulating nonrectangular display windows; you could use the mask array to define the shape of the window. You can also use the masked PIXBLT for manipulating patterned objects; load the pattern into the source array and load the object's shape into the mask.

## 12.5.4 VRAM Block-Mode PIXBLT (VBLT)

The TMS34020 supports a PIXBLT that takes advantage of VRAM block writes. The VBLT instruction is similar to the PIXBLT B,L instruction because it expands a binary source array to a linear destination array; however, there are several important differences.

- VBLT works only when the entire destination array lies in VRAM that supports block writes.
- As VBLT reads the source array, its actions depend on the value of the current source pixel.
  - **pixel value = 0** VBLT does nothing; it does not alter the corresponding destination pixel. VBLT can check an entire line in a source array; if all the values in a line are 0, VBLT doesn't write the line.
  - **pixel value = 1** VBLT tells the appropriate VRAM to copy the contents of the VRAM color register into the destination.
- ❑ VBLT uses the contents of the VRAM color register to represent 1s in the source array. Before executing VBLT, use the VLCOL instruction to copy the contents of the COLOR1 register into the VRAM color register.

- The VRAM color registers are a minimum of 4 bits wide; thus, VBLT supports pixel sizes of 4,8,16, and 32 bits. (VBLT doesn't support 1- and 2-bit pixels.)
- The source and destination array addresses in SADDR and DADDR must contain linear addresses.
- The destination array pitch must be an integer multiple of 128 (80<sub>16</sub>) bits.
- VBLT executes approximately twice as quickly as PIXBLT B,L. The TMS34020 normally writes 32 bits of data during each memory write cycle. VRAM block mode enables the TMS34020 to write 128 bits during each memory write cycle.
- ❑ VBLT does not support pixel processing, transparency, XY addressing, or window checking.
- ❑ VBLT supports plane masking if VEN[[CONFIG]] is set to 1. VBLT uses the VRAM write mask to perform plane masking.

The sequence of events required to use a VBLT might be

- **Step 1:** Load the required COLOR1 value in B9.
- **Step 2:** Execute VLCOL, copying COLOR1 into the VRAM color registers.
- **Step 3:** Set up the remaining implied operands (SADDR, DADDR, etc.).
- Step 4: Execute VBLT.

#### Note:

For more information about the VRAM block-write, color—mask, and writemask features, refer to Sections 8.11 (<u>VRAM Write-Mask Local-Memory</u> <u>Cycles</u>, page 8-34) and 8.12 (<u>VRAM Block-Write Local-Memory Cycles</u>, page 8-37).

### 12.5.5 FILLs

The FILL instruction fills a destination pixel array with the value in the COLOR1 register. A FILL is similar to a PIXBLT that uses no source array. The source pixel value used in pixel processing is the value in the COLOR1 register. The TMS34020 supports two basic FILL instructions; the destination array can be specified in either XY or linear format.

# 12.5.6 Horizontal Pattern Fill (PFILL)

The PFILL instruction is an extension of the basic FILL instruction. Instead of filling with solid COLOR1, PFILL expands the contents of the PATTERN register (B13). PFILL replaces each 0 in the pattern with a COLOR0 pixel; PFILL replaces each 1 with a COLOR1 pixel. PFILL uses this pattern of pixels to fill each horizontal line of the destination array. The pattern repeats at least once every 32 pixels (it may repeat more frequently, if you like).

For PFILL, the destination array address in DADDR should contain an XY address. PFILL can use window checking.

## 12.5.7 VRAM Block Mode-Fill (VFILL)

The VFILL instruction is an extension of the basic FILL instruction. VFILL takes advantage of VRAM block-mode memory accesses. VFILL's relationship to the FILL instruction is analogous to VBLT's relationship to the PIXBLT B,L instruction. Note that DADDR must contain a linear address; VFILL cannot use XY addressing or window checking.

Unlike VBLT, VFILL does not need to read data from memory. Thus, VFILL can use back-to-back block-mode writes. VFILL executes approximately four times faster than the corresponding FILL.

One useful application of VFILL is to use it in combination with VBLT to write text to a screen:

- Step 1: Load the background color into the COLOR1 register.
- **Step 2:** Execute VLCOL to copy the COLOR1 value into the VRAM color registers.
- **Step 3:** Set up VFILL's implied operands and execute VFILL to write the background color to the screen.
- Step 4: Load the foreground color value into the COLOR1 register.
- **Step 5:** Execute VLCOL to copy the COLOR1 value into the VRAM color registers.
- **Step 6:** Set up VBLT's implied operands and execute VBLT to write the text character to the screen.

#### Note:

For more information about the VRAM block-write, color—mask, and writemask features, refer to Sections 8.11 (<u>VRAM Write-Mask Local-Memory</u> <u>Cycles</u>, page 8-34) and 8.12 (<u>VRAM Block-Write Local-Memory Cycles</u>, page 8-37).

# **12.6 Auxiliary Graphics Instructions**

In addition to the single-pixel, line, and pixel-array instructions, the TMS34020 supports several auxiliary graphics instructions. These instructions do not draw pixels; they help you to obtain information or set up implied operands for the graphics drawing instructions. (An exception to this is TFILL.)

**FPIXEQ** and **FPIXNE** (find pixel value)

These instructions search through memory, comparing pixel values to the pixel value in the COLOR0 register.

- FPIXEQ stops searching when it finds a pixel that equals the COLOR0 value.
- FPIXNE stops searching when it finds a pixel that doesn't equal the COLOR0 value.

Usually, you will want to search through a pixel array in memory. You must supply the starting address for the block of memory that will be searched. The instructions can search forward or backward from the initial location.

These instructions are especially useful for seed fills and data compression.

GETPS and EXGPS (get and exchange pixel size)

The PSIZE register identifies the current pixel size. The GETPS and EXGPS instructions provide you with easy methods for getting and changing the PSIZE value.

- GETPS copies the contents of PSIZE into a general-purpose register.
- EXGPS exchanges the contents of the PSIZE register with the contents of a general-purpose register.
- **RPIX** (replicate pixel)

The RPIX instruction replicates a pixel value within a general-purpose register. RPIX uses the current pixel size to replicate the pixel value the correct number of times to produce a 32-bit value within the register. For example, if the current pixel size is 4, RPIX will replicate the pixel value 8 times to produce a 32-bit value; if the current pixel size is 16, RPIX will replicate the pixel value twice; etc.

**CLIP** (array clip)

The TMS34020 supports a CLIP instruction that adjusts an array specified by the DADDR register and the DYDX register to fit within a window. The adjusted values replace the original values.

## **TFILL** (trapezoidal fill)

The TFILL instruction fills a trapezoidal area by drawing a series of horizontal lines. Figure 12–5 shows a trapezoid that is drawn to the screen.

Figure 12–5. A Trapezoidal Fill



 $x_1$  and  $x_2$  define the length and position of a horizontal line. TFILL draws a line, adjusts  $x_1$  and  $x_2$  by a specified amount, and increments y to point to the next position where a line could be drawn. Executing TFILL once draws only one line; to draw a trapezoid, you must call TFILL an appropriate number of times.

# 12.7 Window Checking

The TMS34020 allows you to define a rectangular window and to determine if a pixel lies within the window. It does this by comparing the pixel's XY address to the window's starting and ending points.

Window checking allows you to select an area of the screen that can or cannot be affected by pixel writes. Window checking does not affect data writes by nongraphics instructions; it affects only pixel writes by the following instructions:

DRAV	PIXTs	LINE
FILL XY	PFILL XY	TFILL
PIXBLT L, XY	PIXBLT XY, XY	PIXBLT B, XY

### Note:

Window checking works with XY addresses only—you cannot define a window with linear addresses, nor can you compare a linear address to a window.

The TMS34020 supports four window-checking modes, selected by the value in W[[CONTROL]]. Note that the four modes affect single-pixel, line, and pixel array instructions in different ways.





# 12.7.1 Defining a Window

To define the window, you must

- Load the window's starting corner (minimum XY address) into WSTART.
- Load the window's ending corner (maximum XY address) into WEND.

Window start and end coordinates are *signed* 16-bit values.

Figure 12–7 illustrates a window in relation to display memory (a screen) and identifies the window's starting and ending points.





Figure 12–7 shows that a pixel with coordinates (X,Y) lies within the window if  $X_{start} \le X \le X_{end}$  and  $Y_{start} \le Y \le Y_{end}$ . If a pixel does not meet these conditions, it lies outside the window.

The window is empty when  $X_{start} > X_{end}$  or  $Y_{start} > Y_{end}$  (that is, the window contains no pixels). In this case, window checking detects all destination pixel addresses as lying outside the window.

Note that the condition  $X_{start} = X_{end}$  and  $Y_{start} = Y_{end}$  identifies a window that contains a single pixel.

## 12.7.2 Window-Violation Interrupt

A window-violation interrupt is requested by setting WVP[[INTPEND]] to 1; this happens when either of these situations occurs:

- □ W=1 and a pixel lies inside the window.
- □ W=2 and an attempt is made to write to a pixel outside the window.

The interrupt occurs if WVE[[INTENB]] = 1 and IE[[ST]] = 1. Even if the window-violation interrupt is disabled (IE=0 or WVE=0), you can detect a window violation by testing the value of WVP[[INTPEND]].

When a window-violation interrupt occurs, the current graphics instruction aborts. The registers that change during these instructions contain the intermediate values that existed at the time the violation was detected.

# 12.7.3 Window Checking for Single-Pixel Instructions

Table 12–4 describes the window-checking modes for single-pixel instructions.

Table 12-4. Window-Checking Modes for Single-Pixel Instructions

Mode	Effect
0	Ignore window-checking information
1	<i>Window hit.</i> The instruction draws no pixel, but it requests an interrupt if the pixel is <b>inside</b> the window. (This is a useful pick mode for mice.)
2	<i>Window miss.</i> The instruction draws a pixel only if the pixel lies <b>inside</b> the window; the instruction requests an interrupt if the pixel lies <b>outside</b> the window.
3	Window clip. The instruction draws a pixel only if the pixel lies <b>inside</b> the window; however, the instruction requests no interrupts.

The V[ST] (overflow) bit identifies the pixel's relationship to the window. When window checking is turned off (window mode 0), the V bit is unaffected. When window checking is enabled (mode 1, 2, or 3), here's how V is affected:

If the pixel is **outside** the window, V is set to 1.

If the pixel is **inside** the window, V is cleared to 0.

## 12.7.4 Window Checking for Pixel-Array Instructions

Table 12-5 describes the window-checking modes for pixel-array instructions.

Table 12–5. Window-Checking Modes for Pixel Array Instructions

Effect
Ignore window-checking information
Window hit. No pixels are drawn, but the specified destination array is clipped to lie within the window. If any pixel lies within the window, a window-violation interrupt is generated; the DADDR and DYDX registers are adjusted to be the starting address, width, and height that define the intersection of the destination array with the window. This window-checking function is useful for determining the intersection of any two rectangles on the screen.
Window miss. DADDR and DYDX are compared to the window dimensions. If any pixel in the array lies outside the window, the instruction aborts, and a window-violation interrupt is requested. Unlike window mode 1, however, DADDR and DYDX are not altered as a result of the window comparison (DADDR will change as pixels are drawn). If the entire array lies within the window, then the instruction proceeds.
<i>Window clip.</i> The instruction draws only those pixels that lie <b>inside</b> the window and requests no interrupts. No time is wasted attempting to draw pixels outside the window. DYDX is not changed.

The V[ST] (overflow) bit identifies the array's relationship to the window. When window checking is turned off (mode 0), V is unaffected.

## 12.7.4.1 Detecting a Window Hit (Window-Checking Mode 1)

You can use window-checking mode 1 to pick an object on the screen by moving the cursor to the object's position and selecting it. The actions taken depend on the array's location in relation to the window.

- □ If the array lies completely **outside**,
  - V is set to 1 and
  - no interrupt is requested.
- If any part of the array lies inside,
  - V is cleared to 0 and
  - an interrupt is requested.

To determine which object the cursor is pointing to, a program first sets the window to a small region surrounding the position of the cursor. The program then steps again through the same display list used to draw the current screen until one of the objects causes a window interrupt. The object causing the interrupt should be the object that the cursor is pointing to. If no object causes an interrupt, the pick window can be enlarged and the process repeated until the object is found. If two objects causes interrupts, the size of the pick window can be reduced until only one object causes an interrupt.

### 12.7.4.2 Detecting a Window Miss (Window-Checking Mode 2)

Window-checking mode 2 permits a PIXBLT or FILL instruction to be aborted if any pixel in the destination array lies outside the window. The actions taken depend on the array's location in relation to the window.

- □ If any part of the array lies outside,
  - V is set to 1,
  - the instruction aborts, and
  - an interrupt request is generated.
- □ If the array lies completely **inside**,
  - V is cleared to 0,
  - no interrupt request is generated, and
  - the entire destination array is written.

### 12.7.4.3 Window Clipping (Window-Checking Mode 3)

In this mode, the TMS34020 draws pixels that lie inside the window. The actions taken depend on the array's location in relation to the window.

- □ If the array lies completely **outside**,
  - V is set to 1 and
  - no interrupt is requested.

### If the array lies partly inside and partly outside,

- V is set to 1,
- no interrupt is requested, and
- the portion of the destination array lying within the window is written.

- □ If the array lies completely inside,
  - V is cleared to 0,
  - no interrupt is requested, and
  - the entire destination array is written.

When the instruction begins executing, the destination array is automatically preclipped to lie within the window before the first pixel is transferred. No execution time is lost attempting to write destination pixels that lie outside the window. In the case of a PIXBLT, the source array is also preclipped to fit the adjusted dimensions of the destination array before the transfer begins.

### 12.7.4.4 Clip Instruction for Preclipping a Pixel Array

The TMS34020 supports a CLIP instruction that adjusts the array specified by the DADDR register and the DYDX register to fit within a window. The adjusted values replace the original values. The status bits are set as follows:

- **Z** set to 1 if the array lies entirely outside the window, cleared to 0 if any part of the array lies within the window.
- V set to 1 if any part of the array lies outside the window, cleared to 0 if the array lies entirely within the window

The CLIP instruction is especially useful when there is a possibility that the array could overflow the maximum X and/or Y dimensions of the screen; such an array is called an **overflowing array**.

CLIP is the only instruction that can handle window overflows. If you think that executing a particular instruction could cause a window overflow, do not use a PIXBLT with an XY destination address. Instead, use CLIP, CVXYL, then a PIXBLT with a linear destination address.

## 12.7.5 Window Checking for the LINE Instruction

Table 12-6 describes window-checking modes for the LINE instructions.

Effect
Ignore window-checking information
Window hit. Points on the line are calculated, but no pixels are drawn. As soon as the line moves inside the window, V is cleared, the line-draw aborts, and the instruction requests an interrupt. (V is cleared to 0 if part of the line lies inside the window; otherwise, V is set to 1.)
Window miss. Points on the line are calculated and drawn. As soon as the line moves outside the window, V is set to 1, the line aborts, and the instruction requests an interrupt. (V is set to 1 if part of the line lies outside the window; otherwise, V is cleared.)
<i>Window clip.</i> All points on the line are calculated; points that lie inside the window are drawn. V is cleared to 0 if the last pixel on the line lies inside the window; otherwise, V is set to 1.

Table 12–6. Window-Checking Modes for the LINE Instruction

## 12.7.5.1 Line Clipping

The TMS34020 supports two methods for clipping a straight line to the boundaries of a rectangular window.

- In postclipping, each pixel on the line is compared to the window limits just before being drawn. If the pixel lies outside the window, the write is inhibited. W=3 window-checking mode is selected; window checking is automatically performed in parallel with execution of the LINE instruction, so no overhead is added to the time to draw a pixel. However, unless this form of clipping is used carefully, another type of overhead may become significant. For example, in a CAD environment, if only a small portion of a system diagram is displayed at once, a great deal of time could be spent performing calculations for points (or entire lines) that lie off screen.
- Preclipping determines, before drawing, which pixels in a line lie within the window. The algorithm draws only these pixels and makes no attempt to write pixels outside the window. A preclipped line may take less time to draw since no drawing calculations are performed for pixels lying outside the window. Preclipping is usually faster than postclipping, depending on the likelihood of a line lying outside the window.

The first step in preclipping a series of lines is to identify any lines that lie either entirely inside or outside the window. This is accomplished by using an outcode technique similar to that of the Cohen-Sutherland algorithm.

- Lines lying entirely outside the window are trivially rejected and consume no more processing time.
- Lines lying entirely inside the window are drawn.
- Any remaining lines cross one or more window boundaries and require intersection calculations to identify portions lying within a window.

The Cohen-Sutherland method for determining where a line lies in relation to a window uses **outcodes** that identify the location of the line's endpoints in relation to the window. Figure 12–8 illustrates the outcodes assigned to an endpoint that falls within certain window regions.



Figure 12–8. Outcodes for Line Endpoints

In this method, each region is assigned a 4-bit outcode. When an endpoint of a line falls within a particular region, it is assigned the outcode for that region. The outcodes determine the treatment of the lines.

- The outcode within the window is  $0000_2$ . If the two endpoints of a line both have outcodes  $0000_2$ , the line lies entirely within the window and is drawn.
- □ If the bitwise-AND of the outcodes of the two endpoints  $\neq$  0000<sub>2</sub>, the line lies entirely outside the window and is rejected.
- Lines that fall into neither of these categories may be partially visible within the window. Visible portions must be identified and drawn.

The TMS34020's CPW (compare point to window) instruction is a single-cycle instruction that compares an endpoint to all sides of the window.

**Midpoint subdivision** is an efficient method for finding the in-window portion of a line that crosses a window boundary. This method ensures that drawing calculations are performed only for pixels lying within the window. Figure 12–9 illustrates the midpoint-subdivision technique.

Figure 12–9. Using Midpoint Subdivision to Determine Which Portion of a Line Lies Within a Window



In Figure 12–9, line *AB* lies partially within the window. Here's how midpoint subdivision finds the portion that lies within the window.

- Step 1: Determine the coordinates of the line's midpoint at C.
- **Step 2:** Comparing the outcodes of *B* and *C* shows that segment *BC* lies entirely outside the window and can be trivially rejected.

- **Step 3:** Subdivide the part of segment *AC* that lies within the window.
- **Step 4:** Determine the coordinates of point *D*, which is the midpoint of *AC*. Point *D* lies within the window.
- **Step 5:** Now invoke the LINE instruction twice, once for segment *DC* and again for segment *DA*. Point *D* is the starting point for both cases. Window-checking mode 2 is used while drawing both segments, but the window-violation interrupt is disabled. When each line crosses the window boundary, the TMS34020 detects this and aborts the LINE instruction. Thus, the LINE instruction performs drawing calculations for only those portions of *DA* and *DC* that lie within the window.

#### 12.7.5.2 Using LINIT and FLINE for Preclipped Line Drawing

The TMS34020's LINIT instruction uses the XY coordinates of a line's endpoints ( $X_0$ ,  $Y_0$  and  $X_1$ ,  $Y_1$ ) to determine the operands required for the LINE and FLINE instructions. LINIT also sets up the status bits as follows.

Status bit set to	When	Notes
N = 1	$X_0 = X_1$	The line is vertical
C = 1	CPW $(X_0, Y_0)$ AND CPW $(X_1, Y_1) \neq 0$	The line lies entirely out- side the window
Z = 1	$Y_0 = Y_1$	The line is horizontal
V = 1	$(X_0, Y_0)$ or $(X_1, Y_1)$ lies outside the window	····

After you execute the LINIT instruction, you can examine the status bits to identify the following types of lines.

- a line that lines entirely outside a window (C=1)
- **a horizontal** line (Z = 1)
- $\Box$  a **vertical** line (N = 1)
- a point (Z=1 and N=1)
- a line that lies **partially within a window** (you can then use the midpoint subdivision and the LINE instruction)
- a line that lies **entirely inside a window** (use FLINE to draw the line)

# 12.8 Pixel Processing

**Pixel processing** (sometimes known as raster processing) controls how source pixels are combined with destination pixels. When the TMS34020 reads a pixel from its source location, the processor combines the pixel with the corresponding destination pixel using the selected pixel-processing option. The TMS34020 writes the result to the destination pixel location. The TMS34020 supports 16 Boolean and and 6 arithmetic pixel-processing options.

- Boolean operations are performed in bitwise fashion on pixels of 1, 2, 4, 8, 16, or 32 bits.
- Arithmetic operations treat operand pixels of 2, 4, 8, 16, or 32 bits as unsigned binary numbers. The arithmetic operations are especially useful for encoding color or intensity information when a display uses multiple bits per pixel. For example, the MAX and MIN operations allow two objects with antialiased edges to be smoothly merged into a single image.

The PPOP[[CONTROL]] selects the pixel-processing option. Figure 12–10 shows the PPOP bits in the CONTROL register.

## Figure 12–10. The PPOP [[ CONTROL]] Bits

99869			
50000		14—10	
		PPOP	
	Note: For a comple	ete illustration and	description of CONTROL, refer to Chapter 4.

Table 12-7 lists all of the TMS34020's pixel-processing options.

Table 12–7. Pixel-Processing Options

	Code O	peration	Code Operation
Boolean Operations	00000 \$ 00001 \$ 00010 \$ 00011 \$ 00100 \$ 00101 \$ 00110 \$	Source $\rightarrow$ Destination Source AND Destination $\rightarrow$ Destination Source AND ~Destination $\rightarrow$ Destination Os $\rightarrow$ Destination Source OR ~Destination $\rightarrow$ Destination Source XNOR Destination $\rightarrow$ Destination ~Destination $\rightarrow$ Destination Source NOR Destination $\rightarrow$ Destination	01000Source OR Destination $\rightarrow$ Destination01001Destination $\rightarrow$ Destination01010Source XOR Destination $\rightarrow$ Destination01011~Source AND Destination $\rightarrow$ Destination011001s $\rightarrow$ Destination01101~Source OR Destination $\rightarrow$ Destination01101Source NAND Destination $\rightarrow$ Destination01110Source $\rightarrow$ Destination01111~Source $\rightarrow$ Destination

Table 12–7.	Pixel-Processing (	Options (	(Continued)
-------------	--------------------	-----------	-------------

	Code	Operation	Code Operation
Arithmetic Operations	10000 10001 10010 10011	Source + Destination $\rightarrow$ Destination Source ADDS Destination $\rightarrow$ Destination Destination – Source $\rightarrow$ Destination Source SUBS Destination $\rightarrow$ Destination	10010 MAX(Source, Destination) → Destination 10101 MIN(Source, Destination) → Destination 10110—11111 Reserved

Pixel-processing options  $10000_2$  and  $10010_2$  correspond to standard 2s complement addition and subtraction. A result that overflows the specified pixel size causes the pixel value to wrap around within its 2-, 4-, 8-, 16-, or 32-bit range. Carry bits, however, are prevented from propagating to adjacent pixels.

## 12.8.1 Boolean Processing Examples

Figure 12–11 illustrates the effects of five commonly used Boolean operations when applied to 1-bit pixels. Black regions contain 0s, and white regions contain 1s. Figure 12–11 (a) and (b) show the original source and destination arrays. The source operand in (a) is the letter **A**. The destination in (b) is a calligraphic-style **X**.





- □ Replace destination with source. A simple replacement operation overwrites the pixels of the destination array with those of the source. Figure 12–11 (c) shows the letter A written over the center portion of a larger X using the replace operation. The rectangular region around the letter A obscures a portion of the X lying outside the A pattern. Other operations allow only those pixels corresponding to the A pattern within the rectangle to be replaced, permitting the background pattern to show through. These are the logical-OR and logical-AND-NOT (NOT source AND destination) operations. The replace-with-transparency operation performs similarly in color systems.
- ❑ Logical-OR of source with destination. Figure 12–11 (d) illustrates the use of the logical-OR operation during a PIXBLT. For a 1-bit-per-pixel display, the OR function leaves the destination pixels unaltered in locations corresponding to 0s in the source pixel array. Destination pixels in positions corresponding to 1s in the source are forced to 1s.
- ❑ Logical-AND of NOT-source with destination. Logically ANDing the negated source with the destination is complementary to the logical-OR operation. Destination pixels corresponding to 0s in the source array remain unaltered, but those corresponding to 1s in the source are forced to 0s. Figure 12–11 (*e*) shows an example of the AND-NOT PIXBLT operation (notice the negative image of the letter A). For comparison, Figure 12–11 (*f*) shows the result of simply ANDing the source and destination.
- Exclusive-OR of source with destination. The XOR operation is useful for making patterns stand out on a screen in instances where it is not known in advance whether the background will be 1s or 0s. At every point where the source array contains a pixel value of 1, the corresponding pixel of the destination array is flipped; that is, a 1 is converted to a 0, and vice versa. XOR is a reversible operation; by XORing the same source to the same destination twice, the original destination is restored. These properties make the XOR operation useful for placing and removing temporary objects such as cursors, and in rubberbanding lines. As Figure 12–11 (g) shows, however, the object may be difficult to see if both the source and destination arrays contain intricate shapes.

## 12.8.2 Multiple-Bit Pixel Operations

The Boolean operations described in Section 12.8.1 are sufficient for single-bit pixel operations. However, they may be inappropriate for multiple-bit pixel operations, especially when you are using color. For example, the result of a bitwise-OR operation on a black-and-white (1-bit-per-pixel) display is easily predicted: ORing black and white yields white. However, the result of this operation is less intuitive when applied to multiple-bit pixels. For example, in a population-density map, colors may be used to represent numeric values. If one color, such as red, represents one level of population density, and blue represents another, what happens when the two colors are bitwise-ORed? When pixels represent numeric values, numerical operations such as addition and subtraction yield more useful results.

Boolean operations are usually inadequate for merging antialiased objects into a single bitmapped image. Older graphics systems that are limited to Boolean operations on pixels are incapable of supporting many practical applications on multiple-bit-per-pixel images. For instance, where two antialiased lines cross, AND and OR operations yield chaotic pixel intensities that defeat the purpose of the antialiasing. However, merging the two lines by means of the TMS34020's MAX operation (for white on black) or MIN operation (for black on white) yields a smooth and aesthetically pleasing image.

### 12.8.2.1 Examples of Boolean and Arithmetic Operations

Figure 12–12 illustrates Boolean and arithmetic operations on multiple-bit pixels.

- Figure 12–12 (a) illustrates the source array, which contains a red letter A. The red pixels have the value 8 (1000<sub>2</sub>) and the black background pixels have the value 0 (0000<sub>2</sub>).
- Figure 12–12 (b) shows the destination array, which contains a yellow X. The yellow pixels have the value 12 (1100<sub>2</sub>) and the pixels in the blue background pixels have the value 2 (0010<sub>2</sub>).

Boolean operations can be applied to multiple-bit pixels by combining the corresponding bits of each pair of source and destination pixels on a bit-by-bit basis according to the specified Boolean operation. Figure 12-12 (c) through (g) shows the effects of combining the source and destination arrays using the replace, OR, AND-NOT, AND, and XOR PIXBLT operations. Compare these to Figure 12-11 (page 12-28).

Arithmetic operations treat 2-, 4-, 8-, 16-, and 32-bit pixels as unsigned binary numbers. An *n*-bit pixel represents a positive integer in the range 0 to 2n-1 (all 1s). Examples of arithmetic operations on source and destination pixels are shown in Figure 12–12 (*i*) through (*n*).

Figure 12–12. Examples of Boolean and Arithmetic Operations



(c)Source replaces destination



(f)Source AND destination



(i)Add



(I)Subtract with saturation



(d)Source OR destination



(g)Source XOR destination



(j)Subtract







>

(b)Destination

(e)Source AND destination



(h)Replace with transparency



## (k)Add with saturation





Figure 12–12 (i) and (j): Simple addition and subtraction. (i) shows the result of adding the source and destination arrays, using simple binary 2s complement addition. When the sum of the two pixels exceeds the maximum pixel value, the result overflows. Figure 12–12 (j) shows the result of subtracting the source array from the destination array. Underflow occurs for those pixels whose calculated difference is negative.

Simple addition and subtraction are complementary operations. They are reversible operations in the same sense as the XOR operation: by adding a source pixel to a destination pixel, and then subtracting the same source pixel, the original destination pixel is recovered.

□ Figure 12–12 (k) and (l): Add and subtract with saturate. The arithmetic add and subtract operations are binary 2s complement operations that allow overflow and underflow. An add-with-saturate operation stops the result at the maximum unsigned value without allowing the result to overflow. For example, with 4 bits per pixel, adding 0010<sub>2</sub> to 1110<sub>2</sub> produces 1111<sub>2</sub>. Similarly, a subtract-with-saturate operation stops the result at 0 without allowing it to underflow.

Figure 12–12 (*k*) and (*l*) illustrates addition with saturation and subtraction with saturation. In these examples, the pixel size is 4 bits. By dedicating a different color to each value, the effects of each PIXBLT operation become more visible.

An alternate method of encoding 4-bit pixels uses the 16 values 0 to 15 to represent increasing intensities of a single color component: red, green, or blue. The addition and subtraction operations now have obvious meaning: increasing or decreasing the intensity by specified amounts. At 12 bits per pixel, 4 bits of intensity can be dedicated to each of the three color components. Arithmetic operations are then performed on the corresponding components of each pair of source and destination pixels.

Figure 12–13 (page 12-33) presents examples in which the pixel values represent intensities of a gray from black to white.

□ Figure 12–12 (m): Maximum. MAX compares two pixel values and replaces the destination pixel with the larger value. In some respects, MAX is the arithmetic equivalent of the Boolean OR function (compare Figure 12–12 (m) with Figure 12–11 (b)). The use of MAX in gray-scale and color displays is similar to that of OR in simple black and white.

If the MSBs in each pixel are assigned to represent object priority (whether an object appears in front of or behind another object), the MAX operation can be used to replace only those pixels of the destination array whose priorities are lower than those of the corresponding pixels in the source array. This allows an object to be drawn to the screen so that it appears either in front of or behind other objects previously drawn. In Figure 12–12 (*m*), the red A has a numerical value that is greater than that of the blue background, but less than that of the X. The MAX function is also useful for smoothly combining two antialiased objects that overlap.

□ Figure 12–12 (n): Minimum. Figure 12–12 (n) illustrates the results of the MIN operation on the source and destination arrays. MIN compares two pixel values and replaces the destination pixel with the smaller value. MIN is similar to the Boolean AND function: MIN can be used with priority-encoded pixel values, similar to MAX, but the effect is reversed. In Figure 12–12 (n), the priorities of the two objects are reversed from that of the MAX example shown in Figure 12–12 (m). The MIN operation also has uses similar to those of MAX in smoothly combining antialiased objects that overlap.

#### 12.8.2.2 Operations on Pixel Intensity

Figure 12–13 illustrates the visual effects of various PIXBLT operations on two intersecting disks. In these examples, each pixel is a 4-bit value representing an intensity from 0 (black) to 15 (white). Before the PIXBLT operation, only a single disk resides on the screen, as Figure 12–13 (*a*) shows. The intensity of the disk is greatest at the center (where the value is 12), and gradually falls off as the distance from the center increases. Figure 12–13 (*b*) through (*f*) shows the effects of combining a second, identical disk with the first. Figure 12–13 (*b*) through (*e*) is produced using arithmetic operations; (*f*) is the result of a logical-OR of the source and destination.





The gradual change in intensity at the edge of the disk in Figure 12–13 (*a*) is similar to the result produced by antialiasing techniques that reduce jagged-edge effects. A text font might be stored in antialiased form, for example, to give the text a smoother appearance. When two characters from the font table are

PIXBLTed to adjacent positions on the screen, they may overlap slightly. The particular arithmetic or Boolean operation selected for the PIXBLT determines the way in which the antialiased edges of the characters are combined within regions of overlap.

Figure 12–13 (b): Replace with transparency on result. A second disk is PIXBLTed into a position near the first disk, using replace-with-transparency on result=0. Those pixels of the first disk that lie within the rectangular region containing the second disk, but are not part of the second disk, remain intact. Visually, the second disk (at the right) appears to lie in front of the original disk (at the left). However, assuming that the gradual change in intensity at the perimeter of the disks is done for the purpose of antialiasing, the sharp edge that results where the second disk covers the first defeats this purpose. In other applications, this sharp edge may be desirable; for example, it might be used to make a text character or a cursor stand out from the background. The replace-with-transparency-on-result operation also supports object priority by writing objects to the screen in ascending order of priority.

Figure 12–13 (c): Add with overflow and subtract with underflow. A second disk is PIXBLTed into an area overlapping the first disk, using an add-with-overflow operation. In this example, when 1 is added to an intensity of 15, the sum is truncated to 4 bits to produce the result 0. The effect of arithmetic overflow is visible at the intersection of the two disks as discontinuities in intensity.

This effect is useful for making objects stand out against a cluttered background. Add with overflow has an additional benefit: the object can be removed by subtracting (with underflow) the object image from the screen.

➡ Figure 12–13 (d): Add and subtract with saturation. The original disk is on the left. A second disk is PIXBLTed into a region overlapping the original disk, using an add-with-saturate operation. Whenever the sum of two pixels exceeds the maximum intensity value, which is 15 for this example, the sum is replaced with 15. The bright region that occurs where the two disks intersect is produced when the corresponding pixels of the two disks are added in this manner. Subtract-with-saturate is the complementary operation; when the difference of the two pixel values is negative, the sum is replaced by the minimum intensity value, 0.

The add-with-saturate operation shown in Figure 12–13 (*d*) approximates the effect of two light beams striking the same surface; the surface is brightest in the area in which the two beams overlap.

These operations can be used to achieve an effect similar to that of an airbrush in painting. Consider a display system that represents each pixel as 12 bits and dedicates 4 bits each to represent the intensities of the three color components, red, green, and blue. This method permits the intensity of each component to be directly manipulated. With each pass of the simulated airbrush over the same area of the screen, the color changes gradually toward the color of the paint in the airbrush. For example, assume that the paint is yellow (a mixture of red and green). Each time a pixel is touched by the airbrush, the intensity of the red and green components is increased by 1, and the intensity of the blue component is decreased by 1. With each sweep of the airbrush, the affected area of the screen turns more yellow until the red and green components reach the maximum intensity value (and are not allowed to overflow), and the blue component reaches 0 (and is not allowed to underflow).

Figure 12–13 (e): MAX and MIN operations. The original disk is on the left. A second disk is PIXBLTed into the rectangular region to its right using the MAX operation. In the region in which the disks overlap, each pair of corresponding pixels from the two disks is compared and the greater value is selected. This produces a relatively smooth blending of the two disks. Unlike add with saturate, the MAX function does not generate a hot spot where two objects intersect.

The visual effect achieved using the MAX operation is desirable in an application, for instance, in which white antialiased lines are constructed on top of each other over a black background. MAX also smooths out places in which the lines are overlapped by antialiased text. MAX is successful in maintaining two visually distinct antialiased objects, while the add-withsaturate tends to run them together.

MIN, which is complementary to MAX, can be used similarly to smooth the appearance of intersecting black antialiased lines and text on a white background.

The MAX and MIN operations are particularly useful in color applications in which the number of bits per color gun is small (8 bits or less). Other operators could also be used to smooth the transition between the two overlapping antialiased objects in Figure 12–13 *(e)*, but any additional accuracy attained by using a more complex smoothing function would probably be lost in truncating the result to the resolution of the integer used to represent the intensity at each point.

# 12.9 Transparency

**Transparency** is a graphics operation that allows you to decide which pixels are visible. If you want to draw an object on a background or on top of another object, you can make pixels in the current object transparent so that the background or object beneath shows through. In some cases, several pixels in a rectangular pixel array that contains an object may not be part of the object itself; you can use transparency to hide the pixels that aren't part of the object. This is useful for ensuring that only the object, and not the rectangle surrounding it, is drawn on the screen.

The CONTROL register controls transparency:

- T[CONTROL] (bit 5) enables transparency.
- □ TM[CONTROL] (bits 0—2) determines which transparency mode is selected when transparency is enabled.

Figure 12–14. Enabling Transparency and Selecting a Transparency Mode

(a) Setting the T bit to enable or disable transparency



(b) Setting the TM bits to select the transparency mode



Note: For a complete illustration and description of CONTROL, refer to Chapter 4.

Transparency works for any graphics instruction that writes pixels; transparency does not affect nonpixel data writes.

Example 12–1 through Example 12–3 illustrate PIXT \*Rs, \*Rd instructions that use the three transparency modes. In these PIXT examples, both the source and destination are read, regardless of whether or not the source is transparent. Note that PIXBLTs operate similarly, but they read in a long word full of pixels and then test for transparency; if transparency-on-source=COL-OR0 is enabled and all the pixels read are transparent, then no destination read (or write) occurs.

Example 12–1. Transparency on Result = 0 for PIXT \*Rs, \*Rd





Example 12–2. Transparency on Source = COLOR0 for PIXT \*Rs, \*Rd

Example 12-3. Transparency on Destination = COLOR0 for PIXT \*Rs, \*Rd



# 12.10 Plane Masking

The plane mask protects certain bits within pixels; graphics instructions do not modify mask-protected pixels.

The PMASK register contains the plane-mask value. Each bit in the plane mask corresponds to a bit position in a pixel. The 1s in the mask identify protected bits; the 0s identify modifiable bits. Pixel bits that are protected by the plane mask are always read as 0s during read cycles and are protected from alteration during write cycles. No status or control bit enables or disables plane masking, but you can effectively disable it by loading PMASK with all 0s (this is the default following reset).

The width of a quantity in the plane mask is the same as the pixel size. To maintain a consistent effect on all the pixels within a destination region, regardless of their position within the destination words, you should replicate the mask for a single pixel to fill the entire 32-bit PMASK register. For example, if the pixel size is 4 bits, replicate the 4-bit mask 8 times. These 8 copies of the mask are applied to the 8 pixels in a long word written to or read from memory. Similarly, replicate the value 16 times for a 2-bit pixel, twice for a 16-bit pixel, etc.

Figure 12–15. Replicating the Plane-Mask Value Through PMASK

_	31—28	2724	2320	1916	15—12	11—8	7—4	3—0	
	4-bit value	Replicate a 4-bit pixel 8 times							
	31_	_24	23_	_16	15	8	7-		
ſ	8- va	bit ue	8- va	bit ue	8- va	bit ue	8- va	bit ue	Replicate an 8-bit pixel 4 times
-									

You can use the RPIX instruction to replicate the plane-mask value in a general-purpose register and then move the value into PMASK.

The plane mask allows you to manipulate the bits within pixels as though the display memory were organized into bit planes (or color planes) that can be selectively protected from modification. The number of planes equals the number of bits per pixel. Consider an example in which the pixel size is 4 bits. The bits within each pixel are numbered 0—3 and belong to planes 0—3, respectively. All the bits numbered 0 in all the pixels form plane 0, all the bits numbered 1 in all the pixels form plane 1, and so on. The plane mask allows you to manipulate one or more planes independently of the others. If a display memory contains four planes, for example, you can dedicate three of the planes to 8-color graphics and use the fourth plane to overlay text in a single color. You can set the plane mask so that the text plane can be modified without affecting the graphics planes, and vice versa.

The plane mask affects only pixel accesses performed during execution of the PIXBLT, FILL, PIXT, DRAV, and LINE instructions. Data accesses by non-graphics instructions are not affected.
Instructions use the plane mask differently during reads and writes:

- Pixel read. The 0s in PMASK correspond to unprotected bits in the source pixel that are seen by the TMS34020 to contain the actual values read from memory. The 1s in PMASK correspond to protected bits in the source pixel that are seen as 0s by the TMS34020, regardless of the values read from memory.
- Pixel write. The 0s in PMASK specify those bits in the destination pixel in memory that may be altered. The 1s in PMASK specify protected bits in the destination pixel that cannot be altered.

When a pixel is transferred from a source to a destination location, plane masking is applied to the values read from the source and destination locations (before pixel processing). As the operands are read from memory, the bits protected by the plane mask are replaced with 0s before any pixel-processing operation is performed. Transparency detection can be applied to the masked data if the appropriate transparency mode is enabled. When the source and destination pixels have been combined to form the result, the plane mask is applied once more. If transparency-on-result=0 is enabled, it is applied to this result.

Source pixels that originate from registers are not affected by the plane mask and undergo pixel processing in unmodified form. Instructions that obtain their source pixels from registers include PIXT *Rs*,\**Rd* and PIXT *Rs*,\**Rd*.XY. Figure 12–16 shows how the TMS34020 applies the plane mask to pixel data during the read cycle of a PIXT \**Rs*.XY, *Rd* instruction.



(a) Original data in memory (4 pixels)



(b) Plane mask (PMASK)



(c) Data read into TMS34020 register



Notes: 1) This example assumes 8 bits per pixel.

2) The pixel moved into the TMS34020 register is right-justified. All register bits to the left of the pixel are zero-filled.

- (a) shows a 32-bit word that contains four 8-bit pixels.
- (b) shows the addressed pixel ANDed with the inverse of the plane mask.
- (c) contains the result, and shows that the bits within the data word that correspond to 1s in the mask are cleared to 0s.

After plane masking, the designated pixel is loaded into the 8 LSBs of the 32-bit destination register, and the 24 MSBs of the register are filled with 0s.

Figure 12–17 shows the transfer of 4 pixels during the course of a PIXBLT operation with transparency-on-result=0, pixel size = 8 bits, and the pixel-processing replace option. The inverse of PMASK is ANDed with the source data. Because the replace option is used, transparency is applied to the entire resulting pixel. In other words, the result controls the write in the manner described in the previous discussion of transparency. Since the 3 LSBs of the source pixel in bits 8—15 are 0s, and the rest of the pixel is masked off, the entire source pixel is interpreted as transparent. The memory interface logic generates an internal mask to govern which bits are modified during a write cycle. This mask contains 0s in the bits corresponding to the transparent pixel.

Figure 12-17. Write Cycle with Transparency on Result=0 and Plane Masking

(a) Original destination data in memory (4 pixels)

	2
1	11
A	
ŀ	
١	
A	
Α	
1	
٩.	
A	
A	2
A	24
	2
В	23
В	
В	
В	
}	
B	
В	
В	
E	1
3	6
A	15
. /	5
٩.	
A	
A	
ŀ	
4 ،	
A	
A	
A	8
1	7
В	
В	
В	
В	
В	
В	
В	
В	C
	)

(b) Source data in memory (to be moved)

YYYY YOO 0   ZZZZ ZZZZ   YYYY YOO 0   ZZZZ ZZZZ				
	YYYY Y 0 0 0	Z Z Z Z Z Z Z Z Z	YYYYY000	ZZZZ ZZZZ

(c) Plane mask (PMASK register)

	0.000			\$10000	19992	000000	22322		2000 (	22292	88888	00000	99999	80000	92000 1	000,0000		00000							88993 8	2000 Q	2000	*****			88888 1	22
1	1	1	1	1	0	0	0	1	1	1	1	1	0	0	0	1	1	1	1	1	0	0	0	1	1	1	1	1	· 0	0	0	100000

(d) Mask source data (SRC • PMASK)

0000	0000	0000	0 Z Z Z	0000	0000	0000	0 Z Z Z

(e) Transparency mask based on the source data (see note 2)

0000 0000	11111111	0000 0000	111111111

Figure 12–17. Write Cycle with Transparency on Result=0 and Plane Masking (Continued)

(f) Combined mask (PMASK • transparency mask)

0000 0000	00000111	0000 0000	00000111
08			

(g) Resulting memory data after write cycle

((combined mask • source data) + (combined mask • destination data))

AAAA AAAA BBBB BZZZ AAAA AAAA BBBB BZZZ

Notes: 1) This example assumes 8 bits per pixel.

- 2) Because this example uses the pixel-processing replace option, the source data is effectively the destination data.
- (a) shows the original data at the destination location.
- (b) shows the source data.
- (c) shows the source data is ANDed with the inverse of the plane mask.
- (d) shows the intermediate result produced by (c). This result is used to generate the transparency mask in (e), which is ANDed with the inverse of the plane mask in (c) to produce the composite mask shown in (f).
- (g) shows the result, produced by replacing with the source only those bits of the destination corresponding to 1s in the composite mask in (f).

# 12.11 Setting Up the Implied Operands for Graphics Instructions

The TMS34020 graphics instructions use the B-file registers and several of the I/O registers as **implied operands**. Just as the TMS34020 obtains information from regular instruction operands, the TMS34020 obtains information from these registers that it needs in order to properly execute the instruction. For example, when you use the ADD instruction, the TMS34020 expects you to supply some information: the values you would like to add. For the ADD instruction, you supply this information as operands to ADD in the same source statement. Some graphics instructions, such as the PIXT instructions, use operands in this same manner. However, they also use implied operands. Some of the information that the TMS34020 expects to find this information in the appropriate implied-operand registers. You must set up these implied operands *before* you can use a graphics instruction.

The code segment in Example 12–4 shows how you might set up the implied operands for a FILL XY instruction. Figure 12–18 illustrates the area and dimensions of the fill.

Figure 12–18. Filled Area for Example 12–4



#### Note:

Chapter 4 contains complete descriptions of all the B-file and I/O registers.

#### Example 12–4. Setting Up Implied Operands for a FILL Instruction

DADDR DPTCH OFFSET WSTART WEND DYDX COLOR1 CONTROL CONVDP PMASK PSIZE	.set .set .set .set .set .set .set .set	B2 ; B3 ; B4 ; B5 ; B7 ; B9 0C0000190h 0C0000140h 0C0000160h 0C0000150h	Equate with t regist addres map. 5 them 1	e the cheir cer fi sses i This a by nam	registers names locations in the le or with their n the I/O memory llows you to call e later.	
* Set un	the C	ONTROL reai	ster t	o cho	use the pixel *	
* proces	sing r	eplace opti	on. en	able t	ransparency. *	
* select	trans	parency-on-	destin	ation	=COLOR0, and *	
* select	. windo	w-checking	mode 2	. Don	t modify *	
* any ot	her bi	ts			- *	
-	MOVE ORI	@CONTROL, 000083E5h,	A1 A1	; ass	ume FSO = 32 bits	
	MOVE	A1, @CONTR	OL, 1	; ass	ume FS1 = 16 bits	
	MOVI	099999999h	, COLO	R1		
	MOVI	4*1024, DP	ТСН			
	SETCDE	>		; ini	tialize CONVDP	
	CLR	OFFSET				
	MOVI	[0,0], WST	ART			
	MOVI	[479,639],	WEND			
	MOVI	[5,8], DYD	Х			
	CLR	A1	; d	isable	PMASK	
	MOVE	A1, @PMASK				
	MOVK	4, A1		; 4-b	it pixels	
	MOVE	Al, @PSIZE				
	MOVI	[4,5], DAD	DR			
	FILL	XY		; exe	cute the instruction	n

**Note:** The purpose of this illustration is clarity, not efficiency.

Some of these registers may contain values that you'll use for more than one graphics instruction; if this is the case, it isn't necessary to explicitly load the register each time you use a graphics instruction. For example, if you always use the same pixel size, you need to load the PSIZE register only once. Some instructions, however, use the contents of certain registers as scratchpads and alter the register contents. In this case, you cannot assume that the register will contain the value that you originally supplied.

No single graphics instruction uses *all* of the implied operands. Table 12–8 identifies which implied operands each graphics instruction uses. Table 12–8 (*a*) lists the implied operands that are B-file registers; Table 12–8 (*b*) lists the implied operands that are I/O registers. A  $\sqrt{}$  symbol indicates that an instruction uses the register as an implied operand.

(a) B-file registers

	t (B0)	(B1)	ł (B2)	(B3)	T (B4)	(T (B5)	(B6)	B7)	10 (B8)	11 (B9)	3 (B10)	(B10)	l (B11)	11)	312)	RN (B13)	B14)
	ADDF	ртсн	ADDF	РТСН	FFSE	STAR	IEND	у ХДУ (	OLOF	OLOF	ADDF	INNO	PTCH	IC1 (B	IC2 (B	ATTEF	EMP (
BLMOVE	ທ √	S			0	3	3		<u>ပ</u>	<u>ပ</u>	2	о I	Σ	<u> </u>	<u> </u>	2	<u>н</u>
CLIP			$\overline{}$					$\overline{}$									
CVDXYL																	
CVMXYL						i											
CVSXYL																	
CVXYL					$\checkmark$												
DRAV																	
FILL L																	
FILL XY							$\checkmark$										
FLINE	$\checkmark$							$\checkmark$	$\checkmark$					$\checkmark$			
FPIXEQ													$\checkmark$				
FPIXNE													$\checkmark$				
LINE	$\checkmark$			$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$					$\checkmark$		$\checkmark$	
PFILL			$\checkmark$		$\checkmark$			$\checkmark$	$\checkmark$	$\checkmark$						$\checkmark$	
PIXBLT B, L	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$				$\checkmark$	$\checkmark$							
PIXBLT B, XY	$\checkmark$			$\checkmark$		$\checkmark$	$\checkmark$	$\checkmark$		$\checkmark$							
PIXBLT L, L	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$				$\checkmark$									
PIXBLT L, XY	$\checkmark$			$\checkmark$	$$	$\checkmark$	$\checkmark$	$\checkmark$									
PIXBLT XY, L	$\checkmark$			$\checkmark$	$\checkmark$												
PIXBLT XY, XY	$\checkmark$			$\checkmark$	$\checkmark$	$\checkmark$											
PIXBLT L, M, L	$\checkmark$	$\checkmark$			$\checkmark$												
PIXT Rs, *Rd.XY						$\checkmark$											
PIXT *Rs.XY, Rd																	
PIXT *Rs.XY, *Rd.XY		$\checkmark$			$\checkmark$												
SETCDP																	
SETCMP													$\checkmark$				
SETCSP																	
TFILL	$\checkmark$		$\checkmark$		$\checkmark$		$\overline{\mathbf{V}}$	$\checkmark$		$\checkmark$	$\checkmark$		$\checkmark$				
VBLT	$\checkmark$																
VFILL																	
VLCOL																	

			CON	TROL								
	PPOP field	T bit	TM field	W field	PBH bit	PBV bit	CONFIG (VEN bit)	CONVDP	CONVMP	CONVSP	PMASK	PSIZE
CVDXYL								$\checkmark$				$\checkmark$
CVMXYL									$\checkmark$			$\checkmark$
CVSXYL										$\checkmark$		$\checkmark$
CVXYL								$\checkmark$				$\checkmark$
DRAV	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$				$\checkmark$			$\checkmark$	$\checkmark$
FILL L	$\checkmark$	$\checkmark$	$\checkmark$								$\checkmark$	$\checkmark$
FILL XY	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$				$\checkmark$			$\checkmark$	$\checkmark$
FLINE	$\checkmark$	$\checkmark$	$\checkmark$					$\checkmark$			$\checkmark$	$\checkmark$
FPIXEQ											$\checkmark$	$\checkmark$
FPIXNE											$\checkmark$	$\checkmark$
LINE								$\checkmark$			$\checkmark$	$\checkmark$
PIXBLT B, L	$\checkmark$	$\checkmark$	$\checkmark$								$\checkmark$	$\checkmark$
PIXBLT B, XY	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$				$\checkmark$			$\checkmark$	√
PIXBLT L, L	$\checkmark$	$\checkmark$	$\checkmark$		$\checkmark$	√					$\checkmark$	
PIXBLT L, XY	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$		$\checkmark$		$\checkmark$	$\checkmark$	$\checkmark$
PIXBLT XY, L	$\checkmark$		$\checkmark$		√	$\checkmark$		$\checkmark$		√	$\checkmark$	
PIXBLT XY, XY	$\checkmark$	$\checkmark$		$\checkmark$	√	$\checkmark$		$\checkmark$		√	$\checkmark$	$\checkmark$
PIXBLT L, M, L	$\checkmark$	$\checkmark$			$\checkmark$	$\checkmark$					$\checkmark$	$\checkmark$
PIXT Rs, *Rd	$\checkmark$	$\checkmark$	$\checkmark$					ł			$\checkmark$	
PIXT * <i>Rs</i> , * <i>Rd</i>	$\checkmark$											
PIXT Rs, *Rd.XY	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$				$\checkmark$		$\checkmark$	$\checkmark$	
PIXT *Rs.XY, Rd												$\checkmark$
PIXT *Rs.XY, *Rd.XY								$\checkmark$		$\checkmark$		$\checkmark$
SETCDP								$\checkmark$				
SETCMP												
SETCSP										$\checkmark$		
TFILL	$\checkmark$	√	$\checkmark$	√				$\checkmark$		$\checkmark$	$\checkmark$	$\checkmark$
VBLT							$\checkmark$				$\checkmark$	$\checkmark$
VFILL							$\checkmark$			$\checkmark$	$\checkmark$	

Table 12–8.Summary of Implied Operands Used by the Graphics Instructions (continued)(b) I/O registers

## 12.12 Converting an XY Address to a Linear Address

The TMS34020 allows you to access a pixel by using its linear address or by using an XY address. It's usually easier and more natural to use XY addressing because screens are typically configured on an XY grid.

Figure 12–19 (*a*) shows how an XY address is represented as a 32-bit quantity. The 16 MSBs form the Y coordinate and the 16 LSBs form the X coordinate. Both coordinates are signed. Figure 12–19 (*b*) illustrates the relationship between an XY address and the position of a pixel in an array.

Figure 12–19. How an XY Address Is Represented



#### 12.12.1 Manual XY-to-Linear Conversion

Although the TMS34020 allows you to specify XY addresses, it usually converts them to linear addresses before it uses them. When it does this, it uses the following equation.

```
linear bit address = (Y \times array pitch) + (X \times pixel size) + offset
```

#### Note:

All legal pixel sizes (1, 2, 4, 8, 16, and 32) are powers of 2. Thus, the  $X \times pixel size$  portion of the XY-to-linear conversion equation can be reduced to a shift.

In some cases, you may find it necessary to manually convert an XY address to a linear address. You can convert an XY address to a linear address by using one of these instructions.

- CVSXYL converts the address using the conversion value in CONVSP.
- CVDXYL converts the address using the conversion value in CONVDP.
- CVMXYL converts the address using the conversion value in CONVMP.

In order to use any of these instructions, you must

- Load the XY address into a general-purpose register, and
- □ Load a conversion value into the appropriate CONVxP register; this conversion value is based on an array pitch. There are three CONVxP (conversion) registers; each is associated with a pitch register. Table 12–9 lists the conversion registers and associated pitch registers.

Table 12–9. TMS34020 Conversion (CONVxP) Registers

Array	Conversion R	egister	Pitch Regis	ster
source	CONVSP (	C000 0130h)	SPTCH	(B1)
destination	CONVDP (	C000 0140h)	DPTCH	(B3)
mask	CONVMP (	C000 0180h)	MPTCH	(B11)

The TMS34020 supports three instructions for loading the correct values into the CONVxP registers.

- SETCSP loads CONVSP according to the value in SPTCH.
- SETCDP loads CONVDP according to the value in DPTCH.
- SETCMP loads CONVMP according to the value in MPTCH.

The TMS34020 supports three categories of array pitch, allowing you to choose between conversion speed and flexibility in defining pitch values. These categories include

- □ Power of 2. When the array pitch is a power of 2, then the Y× array size operation can be performed as a shift. Reducing both multiplications to shifts decreases the linear-address conversion referred to by the CVxXYL instruction to 3 cycles. This is the most efficient calculation.
- ❑ Two powers of 2. If the pitch can be expressed as the sum of two powers of 2 (for example, 1280 = 1024 + 256), then the Y × array pitch operation can be calculated by summing two shifts of the Y value. The CVxXYL conversion time for this method is 4 cycles.
- ❑ Arbitrary pitch. A pitch can be any value. If the pitch is not a power of 2 or cannot be reduced to a sum of powers of 2, then the Y× array pitch operation must be calculated with a full 16-bit-by-32-bit multiplication. The CVxXYL conversion time for an arbitrary pitch consumes approximately 15 cycles.

Figure 12–20 shows how the TMS34020 determines the contents of a CONVxP register from the appropriate pitch register.





#### Note:

Be careful to set up the CONVxP registers correctly; *conversion value* 1 = 0 **always** causes the TMS34020 to perform a multiplication. This can be very time consuming.

## 12.12.2 The CONVxP Registers, Corner Adjusting, and Preclipping

If either a source or destination array is specified in XY format, then the contents of the CONVSP and CONVDP registers are used in instances in which the Y component of the starting address must be adjusted prior to the start of the PIXBLT. The Y component may require adjustment, either to preclip the array or to select a starting pixel in one of the lower two corners of the array. For this reason, the CONVSP and CONVDP registers are sometimes listed as implied operands for a PIXBLT instruction.

1



# **TMS34020 Assembly Language Instruction Set**

This section describes the TMS34020 assembly language instruction set (in alphabetical order). Related subjects, such as addressing modes, are presented first.

You'll find these topics on the following pages:

	Sect	ion	Page
These sections describe the	13.1	Addressing Modes and	
TMS34020 addressing modes and		Operand Formats	13-2
provide a table summary of the TMS34020 instructions.	13.2	Summary Table	13-9
These sections summarize	13.3	Move Instructions	13-19
categories of instructions	13.4	Arithmetic, Logical, and	
		Compare Instructions	13-24
	13.5	Program Control and	
		Context Switching Instructions	13-25
	13.6	Shift Instructions	13-28
	13.7	XY Instructions	13-29
	13.8	Instructions New to the TMS34020	13-30
The remainder of the chapter contains a page-by-page reference that describes the individual TMS34020 instructions.	13.9	Alphabetical Instruction Reference	13-31

## **13.1 Addressing Modes and Operand Formats**

The TMS34020 instruction set supports eight addressing modes. Most instructions have register-direct operands or a combination of register-direct and immediate operands; however, the move and graphics instructions use more complex combinations of operands. This section discusses the TMS34020 addressing modes and defines the symbols used in instruction syntax to indicate an addressing mode.

#### **13.1.1 Immediate Values and Constants**

An instruction syntax may use one of these symbols to indicate an immediate *source* operand:

- *IW* is a 16-bit (short) signed immediate value.
- *IL* is a 32-bit (long) signed immediate value.
- *K* is a 5-bit constant.

Instructions that have immediate source operands have register-direct destination operands. Many instructions that have an immediate value can use either a short or a long value.

Figure 13–1 illustrates a MOVI (move immediate) instruction whose first operand is a 32-bit immediate value. This is the syntax for this MOVI:

#### **MOVI** *IL*, *Rd* [, **L**]

The instruction in Figure 13–1 is as follows:

MOVI OFCOh, A2, L

Figure 13–1 shows the object code in memory (at word *n*) and the effect of the instruction on the CPU registers. The value 0FC0h is copied into register A2 as a zero-extended 32-bit value. (Note that this is a 2-word instruction; the next instruction to be executed is at words n + 2.)





## 13.1.2 Absolute Addresses

An instruction syntax may use one of these symbols to indicate an absolute operand:

@SAddress is a **source** address that contains the source data. @DAddress is a **destination** address.

Note that the @ character is entered as part of the operand (this distinguishes it from an immediate operand).

Figure 13–2 illustrates a MOVB (move byte) instruction that has an absolute operand (the first parameter is a 32-bit source address). This is the syntax for this MOVB:

MOVB @SAddress, Rd

The instruction in Figure 13–2 is as follows:

MOVB @RoutineA, A13

Figure 13–2 shows the object code in memory (at word n) and the effect of the instruction on the CPU registers. @RoutineA is the address of a byte; this MOVB instruction copies the byte at address RoutineA into register A13. (Note that this is a 3-word instruction; the next instruction to be executed is at word n + 3.)



Figure 13–2. An Example of Absolute Addressing

## 13.1.3 Register-Direct Operands

An instruction syntax may use one of these symbols to indicate a register-direct operand:

- *Rs* is a **source** register that contains the source data.
- *Rd* is a **destination** register that will contain the result.

When both operands of an instruction are register-direct operands, the registers *must be in the same file.* (The MOVE *Rs*, *Rd* instruction is an exception to this rule.)

Figure 13–3 illustrates a MOVE instruction that has two register-direct operands. This is the syntax for this MOVE:

#### MOVE Rs, Rd

The example shows this instruction:

MOVE A0, B1

Figure 13–3 shows the object code in memory (at word n) and the effect of the instruction on the CPU registers. The entire contents of register A0 are copied into register B1. (Note that this is a 1-word instruction; the next instruction to be executed is at word n + 1.)





## 13.1.4 Register-Indirect Operands

An instruction syntax may use one of these symbols to indicate a register-indirect operand:

\**Rs* is a register that contains the address of the **source** data.

\**Rd* is a register that contains the **destination** address.

Note that the \* character is entered as part of the operand (this distinguishes it from a register-direct operand).

Figure 13–4 illustrates a MOVE (move field) instruction that has two register-indirect operands. This is the syntax for this MOVE:

**MOVE** \**Rs*, \**Rd* [, *F*]

For more information about field moves, refer to Section 13.3.6 on page 13.3.6.

The example shows this instruction:

MOVE \*A4, \*A3

Figure 13–4 shows the object code in memory (at word *n*) and the effect of the instruction on the destination address. The contents of register A4 specify the address of data to be moved; the contents of register A3 specify the destination address. Assume that the field size for the move is 16 bits; the 16 bits of data at \*A4 are moved to the location at \*A3. (Note that this is a 1-word instruction: the next instruction to be executed is at word n + 1.)





#### 13.1.5 Register-Indirect with Offset

An instruction syntax may use one of these symbols to indicate a register-indirect operand that uses a signed offset:

- \**Rs(Offset)* is a **source** address formed by adding an offset to the contents of the source register.
- \**Rd(Offset)* is a **destination** address formed by adding an offset to the contents of the destination register.

The offset is used only to form an address — the contents of the register are not affected. Note that the \* character is entered as part of the operand.

Figure 13–5 illustrates a MOVE (move field) instruction; the first operand of this instruction is a register-direct operand; the second operand is a register-indirect operand with an offset. This is the syntax for this MOVE:

#### MOVE Rs, \*Rd(offset) [, F]

The example shows this instruction:

MOVE B5, \*B7(32)

Figure 13–5 shows the object code in memory (at word *n*) and the effect of the instruction on the destination location. The destination address is specified by adding the offset (32 bits) to the contents of register B7; this yields a destination location of 05020h. Assume that the field size for the move is 16 bits; the 16 LSBs in register B5 are copied into the destination location. (Note that this is a 2-word instruction; the next instruction is at word n + 2.)



*Figure 13–5.* An Example of Register-Indirect with Offset Addressing

TMS34020 Assembly Language Instruction Set

## 13.1.6 Register-Indirect with Postincrement

An instruction syntax may use one of these symbols to indicate a register-indirect operand that is postincremented:

\**Rs*+ is a register that contains the address of the **source** data.

\**Rd*+ is a register that contains the **destination** address.

Note that the \* and + characters are entered as part of the operand. After the operation is performed, the contents of the specified source or destination register are incremented by the field size used for the operation.

Figure 13–6 illustrates a MOVE (move field) instruction; both the source and the destination operands are postincremented register-indirect operands. This is the syntax for this MOVE:

**MOVE** \**Rs*+, \**Rd*+ [, *F*]

The example shows this instruction:

MOVE \*B4+, \*B14+

Figure 13–6 shows the object code in memory (at word n) and the effect of the instruction on the destination location and the CPU registers. The contents of register B4 are the address of the source data; the contents of register B14 specify the destination address. Assume that the field size for the move is 16 bits; the 16 bits of data at the source address are copied into the destination location. After the move, both registers are incremented by 16 bits (1 word). (Note that this is a 1-word instruction; the next instruction to be executed is at word n + 1.)



Figure 13–6. An Example of Register-Indirect with Postincrement Addressing

#### 13.1.7 Register-Indirect with Predecrement

An instruction syntax may use one of these symbols to indicate a register-indirect operand that is predecremented.

Before the operation is performed, the contents of the specified source or destination register are decremented by the field size used for the operation.

\*-Rs the decremented register contents are the address of the **source** data.

\*-Rd the decremented register contents specify the **destination** address.

Note that the \* and – characters are entered as part of the operand.

Figure 13–7 illustrates a MOVE (move field) instruction; the source operand is a register-direct operand, and the destination operand is a predecremented register-indirect operand. This the syntax for this MOVE:

MOVE Rs, \*-Rd [, F]

The example shows this instruction:

MOVE A4, \*-A3

Figure 13–7 shows the object code in memory (at word n) and the effect of the instruction on the destination location and the CPU registers. Assume that the field size for the move is 16 bits. Register A4 contains the source data. The contents of register A3, *minus the field size* (16 bits, or 1 word) form the destination address 5150h. The 16 LSBs in A4 are copied to address 5150h. (Note that this is a 1-word instruction; the next instruction to be executed is at word n + 1.)





TMS34020 Assembly Language Instruction Set

## 13.1.8 Register-Indirect in XY Mode

An instruction syntax may use one of these symbols to indicate that a register operand contains an XY address.

\**Rs*.XY is a register that contains the XY address of the **source** data. \*Rd.XY is a register that contains the XY destination address.

Note that the \* and .XY characters are entered as part of the operand. Here's an example that uses an indirect-XY destination operand:

A0, \*A6.XY PIXT

This instruction moves the pixel data at the least significant end of register A0 into the XY address specified by the contents of register A6.

## 13.2 Summary Table

Syntax and Description	Words	Machine States	16-Bit Instruction Word MSB LSB
ABS Rd Store absolute value	1	1	0000 0011 100R DDDD
ADD Rs, Rd Add registers	1	1	0100 000S SSSR DDDD
ADDC Rs, Rd Add registers with carry	1	1	0100 001S SSSR DDDD
ADDI <i>IW, Rd</i> Add immediate (16 bits)	2	2	0000 1011 000R DDDD
ADDI <i>IL, Rd</i> Add immediate (32 bits)	3	2, 3 †	0000 1011 001R DDDD
ADDK K, Rd Add constant (5 bits)	1	1	0001 00KK KKKR DDDD
ADDXY <i>Rs, Rd</i> Add registers in XY mode	1	1	1110 000S SSSR DDDD
ADDXYI IL, Rd Add immediate in XY mode	3	2, 3 †	0000 1100 000R DDDD
AND Rs, Rd AND registers	1	1	0101 000S SSSR DDDD
ANDI IL, Rd AND immediate (32 bits)	3	2, 3 †	0000 1011 100R DDDD

First value if immediate data is long-word aligned, second value if immediate data is not long-word aligned

First value if the SP is long-word aligned, second value if the SP is not long-word aligned See Section 15.1 page 15-2.

† ত

See Section 15.2 page 15-10.

		Machine	16-Bit Instruction Word
Syntax and Description	Words	States	MSB LSB
ANDN Rs, Rd AND register with complement (32 bits)	1	1	0101 001S SSSR DDDD
ANDNI IL, Rd AND not immediate (32 bits)	3	2, 3 †	0000 1011 100R DDDD
BLMOVE S, D Block move	1	complex instruction	0000 0000 1111 00SD
<b>BTST</b> <i>K</i> , <i>Rd</i> Test register bit, constant	1	1	0001 11KK KKKR DDDD
BTST <i>Rs, Rd</i> Test register bit, register	1	2	0100 101S SSSR DDDD
CALL Rs Call subroutine indirect	1	3+(1) 3+(4) ‡	0000 1001 001R DDDD
CALLA Address Call subroutine address	3	see instruction	0000 1101 0101 1111
CALLR Address Call subroutine relative	2	3+(1) 3+(4) ‡	0000 1101 0011 1111
<b>CEXEC</b> <i>size, instruction [, ID]</i> Coprocessor internal operation execution, long	3	2 (1) 3 (1) †	0000 0110 0000 0000
<b>CEXEC</b> <i>size, instruction [, ID]</i> Coprocessor internal operation execution, short	2	2(1)	1101 1000 OCCC CCCS
CLIP Clip an array to fit within a window	1	complex instruction	0000 1000 1111 0010
CLR Rd Clear register	1	1	0101 011R DDDR DDDD
CLRC Clear carry	1	1	0000 0011 0010 0000
<b>CMOVCG</b> <i>Rd</i> <sub>1</sub> [, <i>Rd</i> <sub>2</sub> [ <i>size</i> ]], <i>command</i> [, <i>ID</i> ] Move coprocessor to register	3	¶	0000 0110 011R DDDD
<b>CMOVCM</b> * <i>Rd+, transfers, size, command [, ID]</i> Coprocessor to memory indirect (postincrement)	3	5+[cnt–1] 6+[cnt+1]†	0000 0110 101R DDDD
<b>CMOVCM</b> –* <i>Rd, transfers, size, command [, ID]</i> Coprocessor to memory indirect (predecrement)	3	5+[cnt–1] 6+[cnt+1]†	0000 0110 110R DDDD
CMOVCS command [, ID] Move from coprocessor to status register	3	4, 5†	0000 0110 0110 0000
CMOVGC <i>Rs, command</i> [, <i>ID</i> ] One register to coprocessor	3	2(1) 3(1) †	0000 0110 001R SSSS
<b>CMOVGC</b> <i>Rs</i> <sub>1</sub> , <i>Rs</i> <sub>2</sub> , <i>size</i> , <i>command</i> [, <i>ID</i> ] Two registers to coprocessor	3	3(1) 4(1) †	0000 0110 010R SSSS

First value if immediate data is long-word aligned, second value if immediate data is not long-word aligned First value if the SP is long-word aligned, second value if the SP is not long-word aligned See Section 15.1 page 15-2. See Section 15.2 page 15-10.

† ত

Syntax and Description	Words	Machine States	16-Bit Instruction Word MSB LSB
<b>CMOVMC</b> * <i>Rs+</i> , transfers, size, command [, ID] Memory indirect (postincrement) to coprocessor, constant count	3	5+[cnt–1] 6+[cnt+1]†	0000 0110 100T TTTT
<b>CMOVMC</b> –* <i>Rs, transfers, size, command</i> [, <i>ID</i> ] Memory indirect (predec) to coprocessor, constant count	3	5+[cnt–1] 6+[cnt+1]†	0000 1000 001T TTTT
<b>CMOVMC</b> * <i>Rs+, Rd, size, command</i> [, <i>ID</i> ] Memory indirect (postinc) to coprocessor, register count	3	¶	0000 0110 111R SSSS
CMP Rs, Rd Compare registers	1	1	0100 100S SSSR DDDD
<b>CMPI</b> <i>IW, Rd</i> Compare immediate (16 bits)	2	2	0000 1011 010R DDDD
<b>CMPI</b> <i>IL, Rd</i> Compare immediate (32 bits)	3	2, 3 †	0000 1011 011R DDDD
CMPK Compare to set status bits	1	1	0011 01KK KKKR DDDD
<b>CMPXY</b> <i>Rs, Rd</i> Compare X and Y half of registers	1	1	1110 010S SSSR DDDD
<b>CPW</b> <i>Rs, Rd</i> Compare point to window	1	1	1110 011S SSSR DDDD
CVDXYL Rd Convert destination XY address to linear	1	¶	0000 1010 100R DDDD
CVMXYL Rd Convert mask address to linear	1	¶	0000 1010 011R DDDD
CVSXYL Rs, Rd Convert source XY address to linear	1	¶	1110 101S SSSR DDDD
CVXYL <i>Rs, Rd</i> Convert XY address to linear	1	¶	1110 100S SSSR DDDD
DEC Rd Decrement register	1	1	0001 0100 001R DDDD
DINT Disable interrupts	1	3	0000 0011 0110 0000
DIVS <i>Rs, Rd</i> Divide registers signed	1	¶	0101 100S SSSR DDDD
DIVU Rs, Rd Divide registers unsigned	1	¶	0101 101S SSSR DDDD

First value if immediate data is long-word aligned, second value if immediate data is not long-word aligned First value if the SP is long-word aligned, second value if the SP is not long-word aligned See Section 15.1 page 15-2. See Section 15.2 page 15-10. † ত

			40 Dit In struction Wand
Syntax and Description	Words	States	MSB LSB
<b>DRAV</b> <i>Rs, Rd</i> Draw and advance	1	¶	1111 011S SSSR DDDD
<b>DSJ</b> <i>Rd, Address</i> Decrement register and skip jump	2	2 no jump 3 if jump	0000 1101 100R DDDD
<b>DSJEQ</b> <i>Rd, Address</i> Conditionally decrement register and skip jump	2	2 no jump 3 if jump	0000 1101 101R DDDD
<b>DSJNE</b> <i>Rd, Address</i> Conditianally decrement register and skip jump	2	2 no jump 3 if jump	0000 1101 110R DDDD
<b>DSJS</b> <i>Rd, Address</i> Decrement register and skip jump, short	1	2 no jump 3 if jump	0011 1 D offset R DDDD
EINT Enable interrupts	1	3	0000 1101 0110 0000
EMU Initiate emulation	1	1	0000 0001 000 count
<b>EXGF</b> <i>Rd, F</i> Exchange field size	1	F0 1 F1 2	1101 01F1 000R DDDD
<b>EXGPC</b> <i>Rd</i> Exchange program counter with register	2	2	0000 0001 001R DDDD
EXGPS Rd Exchange pixel size	1	2(1)	0000 0010 101R DDDD
FILL L Fill array with processed pixels, linear	1	complex instruction	0000 1111 1100 0000
FILL XY Fill array with processed pixels, XY	1	complex instruction	0000 1111 1110 0000
FLINE {0   1} Fast line draw with linear addressing	1	¶	1101 1110 Z001 1010
FPIXEQ Find pixel equal to COLOR0	1	complex instruction	0000 1010 1011 1011
FPIXNE Find pixel not equal to COLOR0	1	complex instruction	0000 1010 1101 1011
GETPC Rd Get program counter into register	1	1	0000 0001 010R DDDD
GETPS Rd Get pixel size into register	1	2	0000 0010 110R DDDD
GETST Rd Get status register into register	1	1	0000 0001 100R DDDD

First value if immediate data is long-word aligned, second value if immediate data is not long-word aligned First value if the SP is long-word aligned, second value if the SP is not long-word aligned See Section 15.1 page 15-2. See Section 15.2 page 15-10.

† **#**¶§

Syntax and Description	Words	Machine States	16-Bit Instruction Word MSB LSB
IDLE Wait for interrupt	1	¶	0000 0000 0100 0000
INC Increment register	1	1	0001 0000 001R DDDD
JAcc Address Jump absolute conditional	3	3 no jump 4 if jump	1100 code 1000 0000
JR <i>cc Address</i> Jump relative conditional, long	2	1 no jump 2 if jump	1100 code 0000 0000
JRcc Address Jump relative conditional, short	1	2 no jump 3 if jump	1100 code xxxx xxxx
JUMP Rs Jump indirect	1	2	0000 0001 011R DDDD
LINE {0   1} Line draw	1	¶	1101 1111 Z001 1010
LINIT Line intialization	. 1	9	0000 1100 0101 0111
LMO Rs, Rd Leftmost one	1	1	0110 101S SSSR DDDD
<b>MMFM</b> <i>Rs, [, List]</i> Move multiple registers from memory	2	¶	0000 1001 101R DDDD
MMTM Rs, [, List] Move multiple registers to memory	2	¶	0000 1001 100R DDDD
MODS <i>Rs, Rd</i> Modulus signed	1	¶	0110 110S SSSR DDDD
MODU <i>Rs, Rd</i> Modulus unsigned	1	35 3 if Rs = 0	0110 111S SSSR DDDD
MOVB Rs, *Rd Move byte, register to indirect	1	ş	1000 110S SSSR DDDD
MOVB *Rs, Rd Move byte, indirect to register	1	ş	1000 111S SSSR DDDD
<b>MOVB</b> * <i>Rs(offset), Rd</i> Move byte, indirect with offset to register	2	ş	1010 111S SSSR DDDD
<b>MOVB</b> * <i>Rs(SOffset),</i> * <i>Rd(DOffset)</i> Move byte, indirect with offset to indirect with off- set	3	ş	1011 110S SSSR DDDD
MOVB Rs, @DAddress Move byte, register to absolute	3	ş	0000 0101 111R SSSS

First value if immediate data is long-word aligned, second value if immediate data is not long-word aligned First value if the SP is long-word aligned, second value if the SP is not long-word aligned See Section 15.1 page 15-2. See Section 15.2 page 15-10. ++¶§

Syntax and Description	Words	Machine States	16-Bit Instruction Word MSB LSB
MOVB @SAddress, Rd Move byte, absolute to register	3	ş	0000 0111 111R DDDD
MOVB @SAddress, @DAddress Move byte, absolute to absolute	5	§	0000 0011 0100 0000
MOVE Rs, Rd Move register to register	1	1	0100 11MS SSSR DDDD
MOVE Rs, *Rd [, F] Move field, register to indirect	1	§	1000 00FS SSSR DDDD
MOVE Rs, -*Rd [, F] Move field, register to indirect (predecrement)	1	ş	1010 00FS SSSR DDDD
<b>MOVE</b> <i>Rs, *Rd+ [, F]</i> Move field, register to indirect (postincrement)	1	ş	1001 00FS SSSR DDDD
MOVE *Rs, Rd [, F] Move field, indirect to register	1	ş	1000 01FS SSSR DDDD
MOVE –*Rs, Rd [, F] Move field, indirect (predecrement) to register	1	ş	1010 01FS SSSR DDDD
MOVE *Rs+, Rd [, F] Move field, indirect (postincrement) to register	1	Ş	1001 01FS SSSR DDDD
MOVE *Rs, *Rd [, F] Move field, indirect to indirect	1	ş	1000 10FS SSSR DDDD
MOVE -*Rs,-*Rd [, F] Move field, indirect (predecrement)to indirect (pre- decrement)	1	§	1010 10FS SSSR DDDD
<b>MOVE</b> * <i>Rs+,</i> * <i>Rd+</i> Move field, indirect (postincrement) to indirect (postincrement)	1	§	1001 10FS SSSR DDDD
MOVE Rs, *Rd(offset) [, F] Move field, register to indirect with offset	2	, §	1011 00FS SSSR DDDD
<b>MOVE</b> * <i>Rs(offset), Rd [, F]</i> Move field, indirect with offset to register	2	ş	1011 01FS SSSR DDDD
<b>MOVE</b> * <i>Rs(offset),</i> * <i>Rd+ [, F]</i> Move field, indirect with offset to indirect (postin- crement)	2	§	1101 00FS SSSR DDDD
<b>MOVE</b> * <i>Rs(SOffset),</i> * <i>Rd(DOffset) [, F]</i> Move field, indirect with offset to indirect with off- set	3	§	1011 10FS SSSR DDDD
MOVE Rs, @DAddress [, F] Move field, register to absolute	3	§	0000 01F1 100R SSSS

First value if immediate data is long-word aligned, second value if immediate data is not long-word aligned First value if the SP is long-word aligned, second value if the SP is not long-word aligned See Section 15.1 page 15-2. See Section 15.2 page 15-10. † ত

Syntax and Description	Words	Machine States	16-Bit Instruction Word MSB LSB
MOVE @SAddress, Rd [, F] Move field, absolute to register	3	ş	0000 01F1 101R DDDD
<b>MOVE</b> @SAddress, *Rd+ [, F] Move field, absolute to indirect (postinc)	3	ş	1101 01F0 000R DDDD
<b>MOVE</b> @SAddress,@DAddress [, F] Move field, absolute to absolute	5	ş	0000 01F1 1100 0000
MOVI <i>IW, Rd</i> Move immediate (16 bits)	2	2	0000 1001 110R DDDD
<b>MOVI</b> <i>IL, Rd</i> Move immediate (32 bits)	3	2, 3 †	0000 1001 111R DDDD
MOVK <i>K, Rd</i> Move constant (5 bits)	1	1	0001 10KK KKKR DDDD
MOVX Rs, Rd Move X half of register	1	1	1110 110S SSSR DDDD
MOVY <i>Rs, Rd</i> Move Y half of register	1	1	1110 111S SSSR DDDD
MPYS <i>Rs, Rd</i> Multiply registers (signed)	1	5+FS1/2	0101 110S SSSR DDDD
MPYU Rs, Rd Multiply registers (unsigned)	1	¶	0101 111S SSSR DDDD
MWAIT Memory wait	1	minimun of 2	0000 0000 1000 0000
NEG <i>Rd</i> Negate register	1	1	0000 0011 101R DDDD
NEGB Rd Negate register with borrow	1	1	0000 0011 110R DDDD
NOP No operation	1	1	0000 0011 0000 0000
NOT Rd Complement register	1	1	0000 0011 111R DDDD
OR Rs, Rd OR registers	1	1	0101 010S SSSR DDDD
<b>ORI</b> <i>IL, Rd</i> OR immediate (32 bits)	3	2,3†	0000 1011 101R DDDD
PFILL XY Pattern fill	1	complex instruction	0000 1010 0011 0111
<b>PIXBLT B, L</b> Pixel block transfer, binary to linear	1	complex instruction	0000 1111 1000 0000

First value if immediate data is long-word aligned, second value if immediate data is not long-word aligned First value if the SP is long-word aligned, second value if the SP is not long-word aligned See Section 15.1 page 15-2. See Section 15.2 page 15-10. † ত

Syntax and Description		Machine	16-Bit Instruction Word
	woras	States	MSB LSB
PIXELI B, XY Pixel block transfer and expand, binary to XY	1	complex instruction	0000 1111 1010 0000
<b>PIXBLT L, L</b> Pixel block transfer, linear to linear	1	complex instruction	0000 1111 0000 0000
<b>PIXBLT L, M, L</b> Pixel block transfer, linear to linear with mask	1	complex instruction	0000 1110 0001 0111
<b>PIXBLT L, XY</b> Pixel block transfer, linear to XY	1	complex instruction	0000 1111 0010 0000
<b>PIXBLT XY, L</b> Pixel block transfer, XY to linear	1	complex instruction	0000 1111 0100 0000
<b>PIXBLT XY, XY</b> Pixel block transfer, XY to XY	1	complex instruction	0000 1111 0110 0000
<b>PIXT</b> <i>Rs,*Rd</i> Pixel transfer, register to indirect		2 + P	1111 100S SSSR DDDD
<b>PIXT</b> <i>Rs, *Rd</i> <b>.XY</b> Pixel transfer, register to indirect XY		¶	1111 000S SSSR DDDD
<b>PIXT</b> * <i>Rs, Rd</i> Pixel transfer, indirect to register	1	3	1111 101S SSSR DDDD
<b>PIXT</b> * <i>Rs,</i> * <i>Rd</i> Pixel transfer, indirect to indirect	1	4 + <i>P</i>	1111 110S SSSR DDDD
<b>PIXT</b> * <i>Rs.</i> <b>XY</b> , <i>Rd</i> Pixel transfer, indirect XY to register	1	6 + <i>CS</i>	1111 001S SSSR DDDD
<b>PIXT</b> * <i>Rs.</i> <b>XY</b> , * <i>Rd</i> <b>.XY</b> Pixel transfer, indirect XY to indirect XY	1	¶	11.11 010S SSSR DDDD
POPST Pop status register from stack	1	6, 7‡	0000 0001 1100 0000
PUSHST Push status register onto stack	1	2(1) 2(2)‡	0000 0001 1110 0000
PUTST Rs Copy register into status	1	3	0000 0001 101R DDDD
RETI Return from interrupt	1	ſ	0000 1001 0100 0000
RETM Return from monitor	1	¶	0000 1000 0110 0000
RETS [N] Return from subroutine	1	¶	0000 1001 011N NNNN
REV Rd Return revision level	1	1	0000 0000 001R DDDD

First value if immediate data is long-word aligned, second value if immediate data is not long-word aligned First value if the SP is long-word aligned, second value if the SP is not long-word aligned See Section 15.1 page 15-2. See Section 15.2 page 15-10. †#¶§

Syntax and Description	Words	Machine States	16-Bit Instruction Word MSB LSB
<b>RL</b> <i>K, Rd</i> Rotate left, constant	1	1	0011 00KK KKKR DDDD
<b>RL</b> <i>Rs, Rd</i> Rotate left, register	1	1	0110 100S SSSR DDDD
RMO Rightmost one	1	1	0111 101S SSSR DDDD
RPIX Rd Replicate pixel	1	¶	0000 0010 100R DDDD
SETC Set carry	1	1	0000 1101 1110 DDDD
SETCDP Set CONVDP	1	¶	0000 0010 0111 0011
SETCMP Set CONVMP	1	¶	0000 0010 1111 1011
SETCSP Set CONVSP	1	¶	0000 0010 0101 0001
<b>SETF</b> <i>FS, FE, F</i> Set field parameters	1	1	0000 01F1 01FS SSSS
SEXT Rd, F Sign extend to long	1	2	0000 01F1 000R DDDD
<b>SLA</b> <i>K, Rd</i> Shift left arithmetic, constant	1	3	0010 00KK KKKR DDDD
<b>SLA</b> <i>Rs, Rd</i> Shift left arithmetic, register	1	3	0110 000S SSSR DDDD
SLL <i>K, Rd</i> Shift left logical, constant	1	1	0010 01KK KKKR DDDD
<b>SLL</b> <i>Rs, Rd</i> Shift left logical, register	1	1	0110 001S SSSR DDDD
<b>SRA</b> <i>K</i> , <i>Rd</i> Shift right arithmetic, constant	1	1	0010 10KK KKKR DDDD
<b>SRA</b> <i>Rs, Rd</i> Shift right arithmetic, register	1	1	0110 010S SSSR DDDD
SRL K, Rd Shift right logical, constant	1	1	0010 11KK KKKR DDDD
<b>SRL</b> <i>Rs, Rd</i> Shift right logical, register	1	1	0110 011S SSSR DDDD
SUB Rs, Rd Subtract registers	1	1	0100 010S SSSR DDDD

First value if immediate data is long-word aligned, second value if immediate data is not long-word aligned First value if the SP is long-word aligned, second value if the SP is not long-word aligned See Section 15.1 page 15-2. See Section 15.2 page 15-10. t#¶§

Syntax and Description	Words	Machine States	16-Bit Instruction Word MSB LSB
SUBB <i>Rs, Rd</i> Subtract registers with borrow	1	1	0100 011S SSSR DDDD
SUBI /W, Rd Subtract immediate (16 bits)	2	2	0000 1011 111R DDDD
SUBI <i>IL, Rd</i> Subtract immediate (32 bits)	3	2, 3 ‡	0000 1101 000R DDDD
SUBK K, Rd Subtract constant (5 bits)	1	1	0001 01KK KKKR DDDD
SUBXY Rs, Rd Subtract registers in XY mode	1	1	1110 001S SSSR DDDD
<b>SWAPF</b> <i>Rs, Rd,</i> 0 Swap field	1	¶	0111 111S SSSR DDDD
TFILL XY Trapezoid fill	1	complex instruction	0000 1110 1111 1010
TRAP N Software interrupt	1	ſ	0000 1001 000N NNNN
TRAPL Software interrupt, signed	2	¶	0000 1000 0000 1111
VBLT Linear VRAM pixel block transfer	1	complex instruction	0000 1000 0101 0111
<b>VFILL</b> Linear VRAM fast fill	1	complex instruction	0000 1010 0101 0111
VLCOL Latch COLOR1 into the VRAM color registers	1	2(1)	0000 1010 0000 0000
<b>XOR</b> <i>Rs, Rd</i> Exclusive OR registers	1	1	0101 011S SSSR DDDD
<b>XORI</b> <i>IL, Rd</i> Exclusive OR immediate value (32 bits)	3	2, 3 †	0000 1011 110R DDDD
ZEXT Rd, F Zero extend to long	1	1	0000 01F1 001R DDDD

First value if immediate data is long-word aligned, second value if immediate data is not long-word aligned First value if the SP is long-word aligned, second value if the SP is not long-word aligned See Section 15.1 page 15-2. See Section 15.2 page 15-10. † ত

## **13.3 Move Instructions Summary**

The TMS34020 supports a variety of move instructions, allowing you to move immediate values into registers, move data between registers, and move data between registers and memory. Table 13–1 summarizes the various types of move instructions.

Table 13–1. Summa	ry of MOVE Instructions
-------------------	-------------------------

Move Type	Mnemonic	Description
Register	MOVE	Move register to register
Constant	MOVK	Move constant (5 bits)
	MOVI	Move immediate (16 bits)
	MOVI	Move immediate (32 bits)
XY	MOVX	Move 16 LSBs of register (X half)
	MOVY	Move 16 MSBs of register (Y half)
Multiple register	MMFM	Move multiple registers from memory
	MMTM	Move multiple registers to memory
Byte	MOVB	Move byte (8 bits, 9 addressing modes)
Field	MOVE	Move field to/from memory/register (18 addressing modes)

#### 13.3.1 Register-to-Register Moves

The **MOVE** *Rs*, *Rd* instruction is a register-to-register move; it moves a full 32 bits of data between any two general-purpose registers. *This is the only MOVE instruction that allows you to move data between register files A and B.* 

#### 13.3.2 Value-to-Register Moves

The **MOVI** and **MOVK** instructions move immediate values into registers. MOVK moves a zero-extended value into a register; the value must be in the range of 1 to 32. The MOVI instruction has two forms; it can move a 16-bit or a 32-bit immediate value.

#### 13.3.3 XY Moves

The **MOVX** and **MOVY** instructions move values into the 16 LSBs or 16 MSBs, respectively, of a register.

#### 13.3.4 Multiple-Register Moves

The **MMTM** and **MMFM** instructions use register-direct operands. MMTM allows you to save several register values in memory; MMFM allows you to retrieve register values from memory. Both instructions have two types of operands:

- □ The *Rp* operand is a *register pointer*. For the MMTM instruction, Rp contains the memory address where MMTM stores the register values. For the MMFM instruction, Rp contains the memory address from which MMFM loads the stored register values.
- The register list operand is an optional list of registers. It specifies which registers are stored or retrieved and also indicates the storing or retrieval order.

Note that Rp and all the registers in the list *must be in the same register file.* If you are saving or restoring registers that are from both files, two MMTMs are required.

#### 13.3.5 Byte Moves

The **MOVB** instruction is a special form of the MOVE instruction; when you use MOVB, the field size is restricted to 8 bits. MOVB supports nine combinations of operand formats. There are three basic combinations:

- Register to memory (requires a field insertion),
- Memory to register (requires a field extraction), and
- Memory to memory (requires both field insertion and extraction).

Note that the MOVB instruction does not move data between registers.

The MOVB instruction allows a byte to begin on any bit boundary in memory. The byte's memory address points to the LSB of the byte. When a byte is moved into a register, the byte's LSB coincides with the register's LSB; the byte is sign-extended into the 24 MSBs of the register.

See Table 13–4, page 13-154, for a summary of the valid combinations of operand formats for the MOVB instruction.

Sequences of byte moves are more efficient if the byte addresses are aligned on even 8-bit boundaries.

## 13.3.6 Field Moves

The **MOVE** instruction supports eighteen combinations of operand formats. There are four basic combinations:

- Register to register
- Register to memory
- Memory to register
- Memory to memory

The MOVE instruction moves a *field*. A field is a configurable data structure that is identified by its starting address and its length. Field lengths can range from 1 to 32 bits. A field's memory address points to the LSB of the field; the field occupies contiguous bits. A field in a register is right-justified within the register; the field's LSB coincides with the register's LSB.

Note that all forms of the MOVE instruction have an optional **F** parameter. (MOVE Rs,Rd is an exception to this; it doesn't have an F parameter because it always moves 32 bits.) **F** selects the field size and field extension for the MOVE:

□ If F=0, FS0 and FE0 determine the field size and extension.

If F=1, FS1 and FE1 determine the field size and extension.

If you don't specify 0 or 1, 0 is used as the default. The selected field size determines the size of the field that is moved. A moved field is either sign-extended or zero-extended, depending on the value of the appropriate field extension bit. You can use the SETF instruction to set the field size and extension.

See Table 13–5, page 13-159, for a summary of the valid combinations of operand formats for the MOVE instruction.

#### 13.3.6.1 Register-to-Memory Moves

Figure 13–8 illustrates the register-to-memory move operation. In this type of move, the source register contains the right-justified field data (width is specified by the field size). The destination location is the bit position pointed to by the destination memory address. The address consists of two portions: one defines the starting word in which the field is to be written and the other defines an offset into that word, the bit address. Depending on the bit address within this word and the field size, the destination location may extend into one or two long words.



Figure 13–8. Register-to-Memory Moves

#### 13.3.6.2 Memory-to-Register Moves

Figure 13–9 shows the memory-to-register move operation. The source memory location is the bit position pointed to by the source address. The address consists of two portions: one defines the starting word from which the field is to be read and the other defines an offset into that word, the bit address. Depending on the bit address within this word and the field size, the source location may extend into two or more long words. After the move, the destination register LSBs contain the right-justified field data (width is specified by the field size). The registers's MSBs contain either all 1s or all 0s.

#### Figure 13–9. Memory-to-Register Moves



#### 13.3.6.3 Memory-to-Memory Moves

Figure 13–10 shows a memory-to-memory field move operation. The source memory location is the bit position pointed to by the source address. The destination location is the bit position pointed to by the destination memory address. Depending on the bit addresses within the respective words and the field size, either the source location or destination locations may extend into two or more long words. After the move, the destination location contains the field data from the source memory location.



Figure 13–10. Memory-to-Memory Moves

## 13.4 Arithmetic, Logical, and Compare Instructions

The TMS34020 supports a full range of arithmetic, logical, and compare instructions. Most of these instructions use register-direct operands; some use a combination of immediate and register-direct operands. Some instructions have several versions; each uses a different operand format. For example, the *ADD* instruction has several versions:

- The ADD instruction uses register-direct operands for both the source and destination operands.
- The **ADDI** instruction uses an immediate source with a destination register.
- The **ADDK** instruction uses a 5-bit constant as the source operand with a destination register.
- The ADDXY instruction is similar to the ADD instruction because both operands are register-direct operands; however, the registers contain XY values.

Some instructions that have immediate values as source operands (such as the ADDI instruction) have two forms: a *short form* and a *long form*. In the short form, the source operand is a *16-bit* immediate value, and the instruction occupies *two words*. In the long form, the source operand is a *32-bit* immediate value, and the instruction occupies *three words*. Each form of the instruction has an optional third operand: W for short and L for long. If you don't use the W or L operand, the assembler chooses the short or the long form, depending on the size of the source operand. Using W or L *forces* the assembler to use the short or long form, respectively. If you use W and the source value is greater than 16 bits, the assembler discards all but the 16 LSBs and issues a warning message. If you use L and the source value is less than 32 bits, the assembler sign-extends the value to 32 bits.

Some instructions that use immediate operands have only one version. In this case, the operand is long (32 bits).

#### Note:

When an instruction's source and destination operands are both registerdirect operands, the registers *must be in the same file*. (The MOVE *Rs*, *Rd* instruction is an exception to this rule.)

# 13.5 Program-Control and Context-Switching Instructions

The TMS34020 supports a variety of instructions that allow you to control program flow and to save and restore information by letting you do the following:

- **call and return from subroutines**
- enable or disable interrupts
- set software interrupts
- set, save, or restore status information
- use jump instructions to redirect program flow

Most of these instructions use register-direct or absolute operands; however, several of them have no operands.

## 13.5.1 Subroutine Calls and Returns

The TMS34020 allows you to call a subroutine in three ways:

- indirectly, by loading an address into a register
- directly, by using an absolute address
- relatively, by specifying an address that is an offset

The CALL instructions automatically save status information on the stack. The RETS instruction pops status information off of the stack and returns control to the program or routine that called the subroutine.

## 13.5.2 Interrupt Handling

The TMS34020's EINT and DINT instructions allow you to enable or disable hardware interrupts by providing control of the IE status bit. The TMS34020 also supports a TRAP instruction that provides you with control over 32 software interrupts. TRAPL allows 64K (-32,768 to +32,767) interrupts.

## 13.5.3 Setting, Saving, and Restoring Status Information

Although some instructions automatically save or restore status information, you will often want explicit control over these functions. The TMS34020 supports several instructions that allow you to save and restore PC and ST information. The TMS34020 also supports a SETF instruction that allows you to set field-0/field-1 information in the status register.

#### 13.5.4 Jump Instructions

The TMS34020 supports both conditional and unconditional jumps. The conditional jumps use absolute operands or a combination of register-direct and absolute operands.

- There are four DSJ instructions.
  - DSJ and DSJS decrement the contents of a register and jump to the specified address if the new contents of Rd do not equal 0. If Rd is decremented to 0, then execution continues with the next instruction.

DSJ provides a jump range of -32,768 to +32,767 words; DSJS provides a jump range of  $\pm 32$  words (excluding 0).
The operation of DSJEQ and DSJNE depends on the value of the Z (zero) status bit.

**DSJEQ** decrements the contents of Rd when Z=1 and jumps to the specified address if the new contents of Rd do not equal 0. If Rd is decremented to 0, then execution continues with the next instruction. If Z=0, DSJEQ skips the jump, and execution continues with the next instruction.

**DSJNE** decrements the contents of Rd when **Z=0** and jumps to the specified address if the new contents of Rd do not equal 0. If Rd is decremented to 0, then execution continues with the next instruction. If **Z=1**, DSJNE skips the jump, and execution continues with the next instruction.

The address specified for the DSJ instructions is relative; the assembler uses this address automatically to calculate a displacement, and then it inserts the displacement into the instruction.

DSJEQ and DSJNE provide a jump range of 32K words (excluding 0).

- The **JUMP** instruction is unconditional. The source register contains the address for the jump.
- The conditional jump instructions, **JA***cc* and **JR***cc*, use the condition codes listed in Table 13–2.

The JRcc instruction has a long and a short form. The short form supports a jump range of  $\pm 127$  words (excluding 0). The long form supports a jump range of  $\pm 32$ K words (excluding 0).

The 32-bit address specified for the JAcc instruction is absolute; the assembler inserts this address into words 2 and 3 of the instruction. The address specified for the JRcc instructions is relative; the assembler uses this address automatically to calculate a displacement, and then it inserts the displacement into the instruction. The short form has an 8-bit displacement that is inserted into bits 0—7 of the opcode; the opcode is 1 word long. The long form has 16-bit displacement; the opcode is 2 words long, and the displacement occupies the entire 16 bits of the second word.

Table 13–2 lists the condition codes used with the JR*cc* and JA*cc* instructions. (To use the codes, replace the *cc* with the appropriate mnemonic code; for example, JRUC, JALS, JRYGT, etc.) Before using one of these jump instructions, use the CMP, CMPI, or CMPXY instruction; the compare instructions set the condition codes for the jump by subtracting a source value from a destination value. The first mnemonics code column in Table 13–2 lists the codes that can be used for a jump following a CMP or CMPI. The second mnemonics code column list codes that can be used for a jump following a CMP or CMPI. The second mnemonics code column list codes that can be used for a jump following a CMP or CMPI. The second mnemonics code column list codes that can be used for a jump following a CMPXY (codes that are preceded with an *X* can be used with the result of the X comparison and codes that are preceded with a *Y* can be used with the result of the Y comparison).

	Mnem				
	Non XY	ХҮ	Result of Compare	Status Bits	Code
Unconditional Compares	UC		Unconditional	Don't care	0000
Unsigned Compares	LO (C) (B)	— YN	Dst lower than Src	С	1000
	LS	YLE	Dst lower or same as Src	C + Z	0010
	ні	YGT	Dst higher than Src	<u></u>	0011
	HS (NC) (NB)	YNN	Dst higher or same as Src	ō	1001
	EQ (Z)		Dst = Src	Z	1010
	NE (NZ)	 YNZ	Dst ≠ Src	Z	1011
Signed	LT	XLE	Dst < Src	$(N \cdot V) + (\overline{N} \ \overline{V})$	0100
Compares	LE	—	Dst ≤ Src	$(N \cdot \overline{V}) + (\overline{N} \cdot V) + Z$	0110
	GT	—	Dst > Src	$(N \cdot V \cdot \overline{Z}) + (\overline{N} \cdot \overline{V} \cdot \overline{Z})$	0111
	GE	XGT	Dst ≥ Src	$(N \cdot V) + (\overline{N} \ \overline{V})$	0101
	EQ (Z)	YZ	Dst = Src	Z	1010
	NE (NZ)	 YNZ	Dst ≠ Src	Z	1011
Compare to	Z	YZ	Result = zero	Z	1010
Zero	NZ (NE)	YNZ	Result ≠ zero	Z	1011
	P		Result is positive	$\overline{N} \cdot \overline{Z}$	0001
	N	xz	Result is negative	N	1110
	NN	XNZ	Result is nonnegative	N	1111
General	Z (EQ)	YZ	Result is 0	Z	1010
Arithmetic	NZ (NE)	YNZ	Result ≠ 0	Z	1011
	C (LO) (B)	YN	Carry set on result	C	1000
	NC (HS) (NB)	YNN	No carry on result	C	1001
	B (JALO) (C)	YN	Borrow set on result	С	1000
	NB (HS) (NC)	YNN	No borrow on result	<u></u>	1001
	V†	XN	Overflow on result	V	1100

Table 13–2. Condition Codes for JRcc and JAcc Instructions

Note: A mnemonic code in parentheses is an alternate code for the preceding code.

Logical AND

•

+ Logical OR

Logical NOT

### **13.6 Shift Instructions**

The TMS34020 supports several instructions that left-rotate, left-shift, or right-shift the contents of the destination register. These instructions use register-direct operands or a combination of register-direct and immediate operands; the shift amount is specified by the value of a 5-bit constant or by the value specified in the 5 LSBs of a source register. (Note that the **SRA** *Rs*, *Rd* and **SRL** *Rs*, *Rd* use the 2s complement of Rd's 5 LSBs.)

- The RL instruction left-rotates the contents of the destination register. (This rotation is a barrel shift.) The bits shifted out of the MSB are shifted into the LSB. The C (carry) bit is set to the final value shifted out of the MSB.
- The SLA instruction left-shifts the contents of the destination register. 0s are shifted into the LSBs. The MSBs are shifted out through the C (carry) bit so that the C bit is set to the final value shifted out of the MSB. If either the N (sign) bit or any of the bits shifted out differ from the original sign bit, the V (overflow) bit is set.
- The SLL instruction left-shifts the contents of the destination register. 0s are shifted into the LSBs. The MSBs are shifted out through the C (carry) bit so that the C bit is set to the final value shifted out of the MSB. The main difference between SLL and SLA is that SLL does not check to see if the sign bit changes.
- The SRA instruction right-shifts the contents of the destination register. The value of the sign bit is shifted into the MSBs; this sign-extends the value and preserves the original value of the sign bit. The LSBs are shifted out through the C (carry) bit so that the C bit is set to the final value shifted out of the LSB.
- The SRL instruction right-shifts the contents of the destination register. Os are shifted into the MSBs, beginning with bit 31. The LSBs are shifted out through the C (carry) bit so that the C bit is set to the final value shifted out of the LSB. The main difference between SRL and SRA is that SRL does not preserve the original value of the sign bit.

# **13.7 XY Instructions**

Table 13–3 summarizes the instructions that use XY addresses. This is useful for specifying pixel addresses on the screen. Many of the graphics instructions use XY addressing; the TMS34020 instruction set also supports several other instructions that allow you to manipulate XY addresses.

An XY address is a 32-bit address that is divided into two parts. Within a register, the 16 LSBs contain the X half of the address; the 16 MSBs contain the Y half. The two parts are treated as completely separate values; for example, when you use ADDXY, the X half does not propagate into the Y half.

Instruction Description		Instruction	Description
ADDXYI IL, Rd	Add IL to register in XY	MOVX Rs, Rd	Move X half of Rs to X half of Rd
ADDXY Rs, Rd	Add registers in XY	MOVY Rs, Rd	Move Y half of Rs to Y half of Rd
CPW Rs, Rd	Compare point to window	PFILL	Pattern fill
CMPXY Rs, Rd	Compare registers in XY mode	PIXBLT B, XY	Pixel block transfer (binary to XY)
CVDXYL Rd	Convert destination XY ad- dress to linear	PIXBLT L, XY	Pixel block transfer (linear to XY)
CVMXYL Rd	Convert mask to linear	PIXBLT XY, L	Pixel block transfer (XY to lin- ear)
CVSXYL Rs, Rd	Convert source XY address to linear	PIXBLT XY, XY	Pixel block transfer (XY to XY)
CVXYL Rs, Rd	Convert XY address to lin- ear address	PIXT Rs, *Rd.XY	Pixel transfer (register to indi- rect XY)
DRAV Rs, Rd	Draw and advance	PIXT *Rs.XY, Rd	Pixel transfer (indirect XY to register)
FILL XY	Fill array with processed pixels	PIXT *Rs.XY, *Rd.XY	Pixel transfer (indirect XY to in- direct XY)
LINE {0   1}	Line draw	SUBXY Rs, Rd	Subtract registers in XY mode
LINIT	Line initialization	TFILL	Trapezoid fill

Table 13–3. Summary for XY Instructions

- The **PIXBLT** and **FILL** instructions use XY source and/or destination addresses.
- The **PIXT** instructions use the contents of registers as XY addresses.
- The **LINE** instruction draws a line along points that are calculated as XY addresses.
- The **MOVX/MOVY** instructions move the X or Y half of a source register into the X or Y half of a destination register.
- The arithmetic and logical instructions add, subtract, or compare the X and Y halves of the registers separately.

# 13.8 Instructions New to the TMS34020

The following is a list of new instructions for the TMS34020; these instructions were not included with the TMS34010.

Instruction Name	Instruction Name
ADDXYI	IDLE
BLMOVE	LINIT
CEXEC, 2 versions	MWAIT
CLIP	PFILL XY
CMOVCG	PIXBLT L,M,L
COMVCM, 2 versions	RETM
CMOVCS	RMO
CMOVGC, 2 versions	RPIX
CMOVMC, 3 versions	SETCDP
СМРК	SETCMP
CVDXYL	SETCSP
CVMXYL	SWAPF
CVSXYL	TFILL
EXGPS	TRAPL
FPIXEQ	VBLT
FPIXNE	VFILL
FLINE	VLCOL
GETPS	

### **13.9 Alphabetical Instruction Reference**

The remainder of this section is an alphabetical reference of the TMS34020 assembly language instructions. Most instructions begin on a new page, and contains the following information:

- Syntax: Shows you how to enter an instruction. (The Preface describes the symbols used in instruction syntaxes.)
- Execution: Illustrates the effects of instruction execution on CPU registers and memory.
- **Instruction Words:** Shows the object code generated by an instruction.
- **Description:** Discusses the purpose of the instruction and any other general information related to the instruction.
- **Machine States:** Lists the instruction cache-enabled cycle timing.
- Status Bits: Lists the effects of instruction execution on the status bits (N, C, Z, and V).
- **Examples:** Show the effects of the instruction on memory and registers using various sets of data and initial conditions.

Several instructions discuss additional topics; for example, the conditional jump instructions list the conditions codes and mnemonics for various jumps, and the graphics instructions list the implied operands that they use.

Syntax	ABS Rd														
Execution	Rd	$ Rd  \rightarrow Rd$													
Instruction Words	15	14 13	12	11 1	0	9	8	7	6	5	4	3	2	1	0
	0	0 0	0	0	0	1	1	1	0	0	R		R	d	
Description	ABS into	ABS stores the absolute value of the contents of the destination register back into the destination register. This is accomplished by:													
		Subtracti	ing the o	conte	nts c	of the	e de	stina	tion	regist	ter da	ta fro	om 0	and	
	Ģ	Storing the result back into Rd if status bit N indicates that the result is positive.													
	If the result of the subtraction is negative, then the original contents of the desti- nation register are retained.														
Machine States	1														
Status Bits	N C Z V	Set to the of Rd are Unaffecto 1 if the o 1 if there 8000000	e sign of e negativ ed riginal d e is an o 0h (800	the rove (ur lata is overfic 0000	esuli niess s 0, <i>0</i> ow, <i>0</i> oh is	t of C s Rd ) oth ) oth s retu	) – F = 8 nerw nerw urne	ise ise; a ise; a	pical 0001 an o	ly, N= h), 1 c verflo	0 if th otherv w occ	ne ori wise curs i	ginal if Rd	cont cont	ents ains
Fxamples	Cod	<b>A</b>	Before			Δfto	r								
	000	<u>~</u>	A1			NC	<u> </u>		A1						
	ABS	A1	7FFFFF	FFh		1 x C	0 0		7F	FFFF	FFh				
	ABS	A1	FFFFF	FFh		0 x 0	0 0		00	00000	)1h				
	ABS	A1	800000	00h		1 x C	) 1		80	00000	)0h				
	ABS	A1	800000	01h		0 x 0	0 0		7F	FFFF	FFh				
	ABS	A1	000000	01h		1 x C	0 (		00	00000	01h				
	ABS	A1	000000	00h		0 x 1	0		00	00000	)0h				
	ABS	A1	FFFA00	)11h		0 x 0	0 (		00	05FFI	EFh				

Syntax	ADD Rs, Rd								
Execution	$Rs + Rd \rightarrow Rd$								
Instruction Words	15 14 13	12 11 10 9	8 7 6	5 4	3 2 1 0				
	0 1 0	0 0 0 0	Rs	R	Rd				
Description	ADD adds the contents of the source register to the contents of the destination register and stores the result in the destination register.								
	You can use the ADD instruction with the ADDC instruction to perform multiple-precision arithmetic.								
	Rs and Rd mus	t be in the same	register file.						
Machine States	1								
Status Bits	<ul> <li>N 1 if the result</li> <li>C 1 if there is</li> <li>Z 1 if the result</li> <li>V 1 if there is</li> </ul>	It is negative, 0 a carry, 0 otherw It is 0, 0 otherwis an overflow, 0 of	otherwise vise se therwise						
Examples	Code	<u>Before</u>		After					
		A1	A0	NCZV	A0				
	ADD A1,A0	FFFFFFFh	FFFFFFFh	1100	FFFFFFEh				
	ADD A1,A0	FFFFFFFh	0000001h	0110	00000000h				
	ADD A1,A0	FFFFFFFh	00000002h	0100	00000001h				
	ADD A1,A0	FFFFFFFh	80000000h	0101	7FFFFFFh				
	ADD A1,A0	FFFFFFFh	80000001h	1100	8000000h				
	ADD A1,A0	7FFFFFFh	80000001h	0110	00000000n				
	ADD A1,A0	/		1000	PPPPPPPPPP				
	AL,AU	00000002b	000000011	0000	000000000				
Examples	Code           ADD         A1,A0           ADD         A1,A0	Before           A1           FFFFFFFh           FFFFFFFh           FFFFFFFh           FFFFFFFh           FFFFFFFh           7FFFFFFFh           7FFFFFFFh           7FFFFFFFh           7FFFFFFFh           7FFFFFFFh           7FFFFFFFh           7FFFFFFFh           00000002h	A0 FFFFFFFh 00000001h 0000002h 80000000h 80000001h 80000000h 00000001h 00000001h	After NCZV 1100 0110 0100 0101 1100 0110 1000 1001 0000	A0 FFFFFFFh 00000000h 00000001h 7FFFFFFFh 80000000h FFFFFFFFh 80000000h 0000000h				

#### ADDC Add Registers with Carry

Syntax	ADDC Rs, Rd											
Execution	Rs + Rd + C -	→ R	d									
Instruction Words	15 14 13	12	11 10	9	87	6	5	4	3	2	1	0
	0 1 0	0	0 0	1	F	ls		R		Ro	1	
Description	ADDC adds the contents of the source register, the carry bit, and the contents of the destination register and then stores the result in the destination register. Note that the status bits are set on the final result.											
	Rs and Rd must be in the same register file.											
Machine States	1											
Status Bits	<ul> <li>N 1 if the result is negative, 0 otherwise</li> <li>C 1 if there is a carry, 0 otherwise</li> <li>Z 1 if the result is 0, 0 otherwise</li> <li>V 1 if there is an overflow, 0 otherwise</li> </ul>											
Examples	Code	Bef	ore				<u>Afte</u>	er				
		С	A1		A0		NCZ	ZV	Α	0		
	ADDC A1,A0	1	FFFFFFF	h	FFFFFF	FFh	11(	00	Fl	FFF	=Fh	
	ADDC A1,A0	1	FFFFFFF	h	0000000	1h	010	00	00	00000	01h	
	ADDC A1,A0	1	FFFFFFF	h	0000000	2h	010	00	00	00000	02h	
	ADDC A1,A0	1	FFFFFFF	h	8000000	0h	110	00	80	00000	00h	
	ADDC A1,A0	1	FFFFFFF	h	8000000	1h	11(	00	80	00000	01h	
	ADDC A1,A0	1	FFFFFFF	h	8000000	1h	010	00	80	00000	01h	
	ADDC A1,A0	1	FFFFFFF	h	8000000	0h	01	10	00	00000	00h	
	ADDC A1,A0	1	7FFFFFFF	n	0000000	1h	100	01	80	00000	01h	
	ADDC A1,A0	1	0000002h		0000000	2h	00	0 0	00	00000	05h	
	ADDC A1,A0	0	FFFFFFF	h	FFFFFF	FFh	11(	0 0	FI	FFFI	=FEh	
	ADDC A1,A0	0	FFFFFFF	h	0000000	1h	01	10	00	00000	00h	
	ADDC A1,A0	0	FFFFFFF	h	0000000	2h	010	00	00	00000	01h	
	ADDC A1,A0	0	FFFFFFF	h	8000000	0h	010	01	<b>7</b> F	FFFF	FFh	
	ADDC A1,A0	0	FFFFFFF	h	8000000	1h	11(	00	80	00000	00h	
	ADDC A1,A0	0	7FFFFFFF	h	8000000	1h	01	10	00	00000	00h	
	ADDC A1,A0	0	7FFFFFFF	h	8000000	0h	10	0 0	FI	FFFI	=FFh	
	ADDC A1,A0	0	7FFFFFFF	h	0000000	1h	100	01	80	00000	00h	
	ADDC A1,A0	0	00000002h		0000000	2h	000	0 0	00	00000	04h	

Syntax	ADDI IW, Rd [, W]									
Execution	Rd + 16-bit immediate value → Rd									
Instruction Words	15 14 13 12 11 10	987	6 5	4 3 2	1 0					
	0 0 0 0 1 0	1 1 0	0 0	R R	d					
	16-bit value									
Description	This ADDI instruction adds a sign-extended, 16-bit immediate value to the contents of the destination register and stores the result in the destination register. (The symbol <i>IW</i> in the syntax above represents a 16-bit, sign-extended immediate value.) The assembler uses the short (16 bit) add if the immediate value is previously defined and is in the range –32,768 to 32,767. You can force the assembler									
	to use the short form by following the register operand with a W:									
	ADDI IW,Rd,W									
	If you use the <b>W</b> parameter an bler discards all but the 16 LS	d the value is ou Bs and issues a	utside the le an appropri	egal range, the ate warning n	e assem- nessage.					
	You can use the ADDI instruct ple-precision arithmetic.	tion with the Al	ODC instru	ction to perfo	rm multi-					
Machine States	2									
Status Bits	<ul> <li>N 1 if the result is negative,</li> <li>C 1 if there is a carry, 0 oth</li> <li>Z 1 if the result is 0, 0 othe</li> <li>V 1 if there is an overflow, 0</li> </ul>	0 otherwise erwise rwise 0 otherwise								
Examples	Code	<u>Before</u> A0	<u>After</u> N C Z V	A0						
	ADDI 1,A0	FFFFFFFh	0110	00000000h						
	ADDI 2,A0	FFFFFFFh	0100	00000001h						
	ADDI 1,A0	7FFFFFFh	1001	80000000h						
	ADDI 2,A0	00000002h	0000	000000004h						
	ADDI $32/0/, AU$	FFFFFF0h	0110	000000000000000000000000000000000000000						

Syntax	ADDI IL, Rd [, L]							
Execution	Rd + 32-bit immediate value $\rightarrow$ Rd							
Instruction Words	15     14     13     12     11     10       0     0     0     0     1     0	9         8         7           1         1         0           16 LSBs of 32-bit v           16 MSBs of 32-bit v	6 5 4 0 1 R alue /alue	3 2 1 0 Rd				
Description	This ADDI instruction adds a 32-bit, signed immediate value to the contents of the destination register and stores the result in the destination register. (The symbol IL in the syntax above represents a 32-bit, signed immediate value.) The assembler uses the long (32 bit) ADDI if it cannot use the short form. You can force the assembler to use the long form by following the register operand with an L:							
	ADDI IL,Rd,L							
Machine States	2 if the immediate data is lon 3 if the immediate data is not	g-word aligned long-word alig	ned					
Status Bits	<ul> <li>N 1 if the result is negative</li> <li>C 1 if there is a carry, 0 oth</li> <li>Z 1 if the result is 0, 0 othe</li> <li>V 1 if there is an overflow,</li> </ul>	, 0 otherwise Ierwise rwise 0 otherwise						
Examples	Code ADDI OFFFFFFFh,AO ADDI 8000000h,AO ADDI 8000000h,AO ADDI 32768,AO ADDI 2,AO,L	<u>Before</u> A0 FFFFFFFFh FFFFFFFh 7FFFFFFFh 7FFFFFFFh FFFFFFFF	After NCZV 1100 0101 1000 1001 0100	<b>A0</b> FFFFFFFEh 7FFFFFFFh FFFFFFFFh 80007FFFh 00000001h				

¥

Syntax	ADDK K, Rd								
Execution	Rd + 5-bit constant → Rd								
Instruction Words	15 14 13 12 0 0 0 1	11 10 9 0 0	<u>876</u> К	5 5 4 R	3 2 1 Rd	0			
Description	ADDK adds a 5-b stores the result in represents a 5-bit	it constant to th the destination r constant.)	e contents egister. (Th	of the destine symbol K i	nation register in the syntax at	and pove			
	nal value of K=32, then K is converted to 0 in the opcode. The assembler issues an error if you try to add 0 to a register with this instruction.								
	You can use the A ple-precision arith	DDK instruction metic.	with the AD	DC instructio	on to perform n	nulti-			
Machine States	1								
Status Bits	<ul> <li>N 1 if the result</li> <li>C 1 if there is a</li> <li>Z 1 if the result</li> <li>V 1 if there is ar</li> </ul>	is negative, 0 oth carry, 0 otherwis is 0, 0 otherwise n overflow, 0 othe	nerwise e erwise						
Examples	Code           ADDK         1, A0           ADDK         2, A0           ADDK         1, A0           ADDK         1, A0           ADDK         32, A0	Before AO FFFFFFFh FFFFFFFh 7FFFFFFh 8000000h 8000000h	After NCZV 0110 0100 1001 1000 1000	A0 00000000h 00000001h 80000000h 80000000h 80000020h					
	ADDK 32,A0	00000002h	0000	00000022h					

Syntax	ADDXY Rs, R	d							
Execution	X half of Rs + X Y half of Rs + Y	( half of Rd → )  half of Rd → ) (	X half of Rd Y half of Rd						
Instruction Words	15 14 13	12 11 10	987	65	4	3	2 1	0	
	1 1 1	0 0 0	0 Rs		R		Rd		
Description	ADDXY adds the signed source X value to the signed destination X value, adds the signed source Y value to the signed destination Y value, and stores the result in the destination register. The source and destination registers are treated as if they contained separate X and Y values. Any carry out from the lower (X) half of the register does not propagate into the upper (Y) half.								
	If you only want be 0 (the metho	t to add the X h od for adding th	alves together e Y halves is s	; then or similar).	ne of	the Y	values r	nust	
	You can use this instruction to manipulate XY addresses in the register file; ADDXY is also useful for incremental figure drawing.								
	Rs and Rd mus	t be in the sam	e register file.						
Machine States	1								
Status Bits	<ul> <li>N 1 if resulting</li> <li>C The sign bit</li> <li>Z 1 if Y field is</li> <li>V The sign bit</li> </ul>	g X field is all 0 t of the Y half o s all 0s, <i>0</i> other t of the X half o	s, <i>0</i> otherwise If the result wise If the result						
Examples	Code	Before		After					
		A1	A0	A0		Ν	czv		
	ADDXY A1,A0	00000000h	00000000h	000000	00h	1	010		
	ADDXY A1,A0	00000000h	00000001h	000000	01h	C	010		
	ADDXY A1,A0	00000000h	00010000h	000100	00h	1	000		
	ADDXY A1,A0	00000000h	00010001h	000100	01h	0	000		
	ADDXY A1,A0	0000FFFFh	00000001h	000000	00h	1	010		
	ADDXY A1,A0	0000FFFFh	00010001h	000100	00h	1	000		
	ADDXY A1,A0	0000FFFFh	00000002h	000000	01h	0	010		
	ADDXY A1,A0	0000FFFFh	00010002h	000100	01h	0	000		
	ADDXY A1,A0	FFFF0000h	00010000h	000000	00h	1	010		
	ADDXY A1,A0	FFFF0000h	00010001h	000000	01h	0	010		
	ADDXY A1,A0	FFFF0000h	00020000h	000100	00h	1	000		
	ADDXY A1,A0	FFFF0000h	00020001h	000100	01h	0	000		
	ADDXY A1,A0	FFFFFFFh	00010001h	000000	00h	1	010		
	ADDXY A1,A0	FFFFFFFh	00010002h	000000	01h	0	010		
	ADDXY A1,A0	FFFFFFFh	00020001h	000100	00h	1	000		
	ADDXY A1.A0	FFFFFFFh	00020002h	000100	01h	0	000		

TMS34020 Assembly Language Instruction Set

Syntax	ADDXYI IL, Rd									
Execution	LSW (X half) of 32-bit immediate value + X half of Rd $\rightarrow$ X half of Rd MSW (Y half) of 32-bit immediate value + Y half of Rd $\rightarrow$ Y half of Rd									
Instruction Words	15 14 13 12 11 1	098	7 6 5 4	3 2 1 0						
	0 0 0 0 1	1 0 0	0 0 0 F	R Rd						
	16 LS	Bs of 32-bit imme	diate value (X half)							
	16 MS	16 MSBs of 32-bit immediate value (Y half)								
Description	ADDXYI adds the 32-bit immediate value to the destination register in XY mode and stores the result back into the destination register.									
Machine States	2 if the immediate data is long-word aligned 3 if the immediate data is not long-word aligned									
Status Bits	<ul> <li>N 1 if resulting X field is all 0s, 0 otherwise</li> <li>C The sign bit of the Y half of the result</li> <li>2 1 if Y field is all 0s, 0 otherwise</li> <li>V The sign bit of the X half of the result</li> </ul>									
Examples	<u>Code</u>	<u>Before</u>	<u>After</u>							
		A0	A0	NCZV						
	ADDXYI 000000000h,A0	00000000h	00000000h	1010						
	ADDXYI 000000000h,A0	00000001h	00000001h	0 0 1 0						
	ADDXYI 000000000h,A0	00010000h	00010000h	1 0 0 0						
	ADDXYI 000000000h,A0	00010001h	00010001h	0 0 0 0						
	ADDXYI 00000FFFFh,A0	00000001h	00000000h	1010						
	ADDXYI 00000FFFFh,A0	00010001h	000100001							
	ADDXYI 00000FFFFA,A0	0000000211	00000001h							
	ADDXYI OUCOUFFFFF,AO	0001000211	0001000m	1 0 1 0						
	ADDXII OFFFF000001,A0	000100001	00000001h	0 0 1 0						
	ADDXII OFFFF000001,A0	000100011	000000011	1 0 0 0						
	ADDXII OFFFF000001,A0	000200001h	00010001h	0 0 0 0						
	ADDXII OFFFFFFFF AO	000200011	000100011	1010						
	ADDAVI OFFFFFFFF AO	00010002h	000000016	0 0 1 0						
	ADDXYI OFFFFFFFF AO	00020001h	00010000h	1 0 0 0						
	ADDXYI OFFFFFFFFh,AO	00020002h	00010001h	0 0 0 0						

### AND AND Registers

Syntax	AND Rs, Rd							
Execution	$Rs AND Rd \rightarrow Rd$							
Instruction Words	15 14 13 0 1 0	12 11 10 1 0 0	9 8 7 6 0 Rs	5 4 R	3 2 1 0 Rd			
Description	AND bitwise-Al destination reg and Rd must b	NDs the conter ister and then e in the same r	ts of the source r stores the result register file.	egister with in the dest	the contents of the ination register. Rs			
Machine States	1							
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z 1 if the result is 0, 0 otherwise</li> <li>V Unaffected</li> </ul>							
Examples	Code AND A1,A0 AND A1,A0 AND A1,A0 AND A1,A0 AND A1,A0 AND A1,A0	Before A1 FFFFFFFh FFFFFFFh 0000000h AAAAAAA AAAAAAA 55555555 55555555	A0 FFFFFFFh 00000000h 555555555 AAAAAAAA 555555555 AAAAAAAAA	After NCZV xx0x xx1x xx1x xx1x xx1x xx0x xx0x	A0 FFFFFFFh 00000000h 00000000h 00000000h AAAAAAAA			

Syntax	ANDI IL, Rd										
Execution	32-bit immediate value Al	32-bit immediate value AND Rd $\rightarrow$ Rd									
Instruction Words	15     14     13     12     11       0     0     0     0     1	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         0       0       0       1       0       1       1       1       0       0       R       Rd         Is complement of 16 LSBs of <i>IL</i>									
Description	ANDI bitwise-ANDs the va the destination register ar (The symbol IL in the sym This is an alternate mner stores the 1s complement	alue of a 32-bit imp nd then stores the tax above represe nonic for ANDNI t of IL in the 2 ext	mediate va e result in tl ents a 32-b IL, Rd. No ension wor	lue with the contents of he destination register. it immediate value.) ote that the assembler ds.							
Machine States	2 if the immediate data is 3 if the immediate data is	long-word aligned not long-word alig	d gned								
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z 1 if the result is 0, 0 o</li> <li>V Unaffected</li> </ul>	therwise									
Examples	Code ANDI OFFFFFFFh,A0 ANDI OFFFFFFFh,A0 ANDI 0000000h,A0 ANDI 0AAAAAAAh,A0 ANDI 0AAAAAAAh,A0 ANDI 5555555h,A0 ANDI 5555555h,A0	Before           A0           FFFFFFh           00000000h           5555555h           AAAAAAAAh           5555555h           AAAAAAAAA	After NCZV × × 0× × × 1 × × × 1 × × × 1 × × × 0× × × 0× × × 0×	A0 FFFFFFFh 00000000h 00000000h 00000000h AAAAAAAA							

### ANDN AND Registers with Complement

ANDN Rs, Rd								
(NOT Rs) AND	$Rd \rightarrow Rd$							
15 14 13 0 1 0	12 11 10 1 0 0	9 8 7 1	' 6 5 Rs	4 R	32 Rd	1	0	
ANDN bitwise-A of Rd and then Rs and Rd mus same effect as	NDs the 1s co stores the res t be in the san CLR Rn.	mplement o ult in the de ne register f	of the conte estination r file. Note t	nts of F egister hat AND	Rs with the : N Rn, Rn	conte has	ents the	
1								
<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z 1 if the result</li> <li>V Unaffected</li> </ul>	ılt is 0, <i>0</i> other	wise						
Code ANDN A1, A0 ANDN A1, A0 ANDN A1, A0 ANDN A1, A0 ANDN A1, A0	Before A1 FFFFFFFh FFFFFFFh 00000000h AAAAAAAA AAAAAAAA 55555555	A0 FFFFFF 00000000 55555555 AAAAAA 55555555	At N FFh x Oh x Oh x 5h x AAh x 5h x	ter C Z V x 1 x x 1 x x 1 x x 0 x x 0 x x 1 x x 1 x	A0 0000000 0000000 555555 0000000 0000000	0h 0h 0h 5h 0h		
	(NOT Rs) AND 15 14 13 0 1 0 ANDN bitwise-A of Rd and then Rs and Rd mus same effect as 1 N Unaffected C Unaffected Z 1 if the resu V Unaffected Z 2 1 if the resu V Unaffected Code ANDN A1, A0 ANDN A1, A0	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	(NOT Rs) AND Rd $\rightarrow$ Rd15141312111098701010011ANDN bitwise-ANDs the 1s complement of of Rd and then stores the result in the defect of Rd and then stores the result in the defect as cLR Rn.1NUnaffectedCodeLR Rn.1NNUnaffectedCodeBefore A1ANDN A1, A0FFFFFFFhANDN A1, A0FFFFFFFhANDN A1, A0AAAAAAAAANDN A1, A0S555555hANDN A1, A0S555555hANDAAAAAAAA	(NOT Rs) AND Rd $\rightarrow$ Rd151413121110987650101001RsANDN bitwise-ANDs the 1s complement of the conter of Rd and then stores the result in the destination rRs and Rd must be in the same register file. Note the same effect as CLR Rn.1NUnaffectedCodeBeforeAfA1A0NANDN A1, A0FFFFFFFhFFFFFFhANDN A1, A0AAAAAAAAAS555555hANDN A1, A0S555555hS555555hANDN A1, A0S555555hAAAAAAAAA	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	$\begin{array}{c c c c c c c c c c c c c c c c c c c $	

Syntax	ANDNI IL, Rd											
Execution	(NOT 32-	IOT 32-bit immediate value) AND Rd $\rightarrow$ Rd										
Instruction Words	<u>15 14 13 12 11 10 9 8 7 6 5 4 3 2 1</u>										0	
		16 LSBs of <i>IL</i>										
				16	MSBs of II	-						
Description	ANDNI bit contents o register. ( value.) AN	wise-AND of the desti The symb IDI also u	os the 1s nation re ol IL in t ses this o	comple gister a he synt opcode	ement of and then tax abov	a 32- store /e rep	bit in s the prese	nme resi ents	diate ult in t a 32·	value he de bit ir	e with estina nmeo	n the ation diate
Machine States	2 if the immediate data is long-word aligned 3 if the immediate data is not long-word aligned											
Status Bits	N Unaffe C Unaffe Z 1 if the V Unaffe	ected ected e result is ected	0, <i>0</i> othe	erwise								
Examples	<u>Code</u>			<u>Befor</u> A0	<u>e</u>	<u>Af</u> N (	<u>ter</u> C Z V	,	A0			
	ANDNI	OFFFFF	FFh,A0	FFFFI	FFFh	x	x 1 x		00000	0000h	1	
	ANDNI	0FFFFFF	FFh,A0	00000	000h	x	x 1 x		00000	0000h	1	
	ANDNI	0000000	0h,A0	00000	000h	<b>x</b> :	x 1 x		00000	0000h	1	
	ANDNI	55555	555h	X	x 0 x		55555	5555h	555h			
	ANDNI	0АААААА	AAh,AO	AAAA	AAAAh	<b>x</b> :	x x 1 x 00			00000000h		
	ANDNI	5555555	5h,A0	55555	555h	<b>x</b> :	x 1 x		00000	0000h	1	
	ANDNI	ANDNI 55555555h,A0				X X 0 X AAAAAAA					۹h	

Syntax	BLMOVE S, D
Execution	<i>n</i> bits from address in SADDR $\rightarrow n$ bits at address in DADDR
Instruction Words	15     14     13     12     11     10     9     8     7     6     5     4     3     2     1     0       0     0     0     0     0     0     1     1     1     1     0     0     S     D
Description	BLMOVE transfers a specified number of bits of data starting from a specified source address to a specified destination address. BLMOVE is unusual in 2 respects:
	BLMOVE deals with a continuous block of memory (no pitch concept as in PIXBLT).
	BLMOVE moves a specified number of bits of data (not pixels).
	The values of <i>S</i> and <i>D</i> determine how a block is moved:
	<ul> <li>If S=1 and D=1, then</li> <li>B0, the source address, may be aligned on any bit boundary.</li> <li>B2, the destination address, may be aligned on any bit boundary.</li> <li>B7 should contain the number of bits to be moved.</li> </ul>
	As BLMOVE executes, B2 is incremented to reflect the current state of the move. At the end of the move, B0 is incremented by the initial value stored in B7 so that B0 points to the bit that would have been moved next, had the move continued. At the end of the move, B2 points to the bit after the last bit moved.
	<ul> <li>If S=0 and D=1, then</li> <li>B0, the source address, must be 32-bit long-word aligned.</li> <li>B2, the destination address, may be aligned on any bit boundary.</li> <li>B7 should contain the number of bits to be moved.</li> <li>As BLMOVE executes, B2 is incremented to reflect the current state of the move; B0 remains fixed. At the end of the move, B0 is incremented by the</li> </ul>
	initial value stored in B7, and B2 points to the bit after the last bit moved.
	<ul> <li>If S=1 and D=0, then</li> <li>B0, the source address, may be aligned on any bit boundary.</li> <li>B2, the destination address, must be 32-bit long-word aligned.</li> <li>B7 should contain the number of bits to be moved.</li> <li>As BLMOVE executes, B2 and B0 remain fixed. At the end of the move, B2 and B0 are incremented by the initial value stored in B7.</li> </ul>
	<ul> <li>If S=0 and D=0, then</li> <li>B0, the source address, must be 32-bit long-word aligned.</li> <li>B2, the destination address, must be 32-bit long-word aligned.</li> <li>B7 should contain the number of bits to be moved.</li> <li>As BLMOVE executes, B2 and B0 remain fixed. At the end of the move, B2 and B0 are incremented by the initial value stored in B7.</li> </ul>
13-44	TMS34020 Assembly Language Instruction Set

In all cases, the DYDX register initially contains a count of the number of bits to be moved. This count is decremented as BLMOVE executes, so DYDX reflects the number of bits remaining to be moved. At the end of the move, DYDX will contain 0.

If BLMOVE is interrupted, the PC is decremented to point back to the BLMOVE instruction, the PC and ST are pushed onto the stack, and the program control branches to the appropriate interrupt trap routine. At the end of the trap routine, BLMOVE restarts, so the trap routine must restore the B-file registers used by BLMOVE.

Implied Operands Register Name Format Description B0 SADDR Linear Source block address B2 DADDR Linear Destination block address B7 DYDX Integer Number of bits to move

*Machine States* complex instruction

Status Bits

- N Unaffected
- C Unaffected
- Z Unaffected
- V Unaffected

Examples

This example shows how the memcpy C runtime-support function could be implemented using this instruction. The memcpy function is invoked with 3 arguments on the C parameter stack (pointed to by A14):

- s1 destination address
- s2 source address
- n number of BYTES to move

Note that this function does not check for overlapping memory areas

	.glob]	l _memcpy	;	provide reference for external calls
STK	.set	A14	;	C-parameter stack pointer
SADDR	.set	в0	;	Source address register
DADDR	.set	B2	;	Destination address register
DYDX	.set	В7	;	Delta X/delta Y register
TEMP	.set	B14	;	Temporary register
_memcpy:				
	mmtm	SP, SADDR, DAD	DF	R,DYDX,TEMP ;save the required registers
	move	STK, TEMP		; copy C-stack to B-file register
	move	*-TEMP, DADDR	2,1	; pop s1 (FS 1 assumed to be 32)
	move	*-TEMP, SADDR	1,1	; pop s2
	move	*-TEMP, DYDX,	1	; pop n (byte count)
	move	TEMP, STK		; Update C stack
	sll	3, DYDX		; convert to a bit count
	blmove	e 1,1		; perform the block move
	mmfm	SP, SADDR, DAD	DF	R,DYDX,TEMP ; restore the required registers
	rets	2	;	return to caller (with C-calling convention)

Syntax	BTST constant, Rd									
Execution	Set status on valu	e of constant in	Rd							
Instruction Words	15 14 13 12	11 10 9	8 7 6 5	4	3	2	1	0		
			omplement of constant	n		<u>п</u>	u			
Description	BTST tests a bit ir This form of the BT that is tested. The ates to a number in the assembler issu 5 LSBs.	n the destination IST instruction u constant value n n the range 0 to 3 ues a warning an	register and sets s ses a 5-bit constan nust be an absolute 31; if the constant v d truncates the valu	tatus t to s e exp alue ue of	s bit 2 specif ressi is gre the o	Z aco y the on the eater cons	cordir e bit ir nat ev r than tant t	າgly. າ Rd valu- າ 31, o its		
	Note that the asse it into the opcode.	mbler 1s-comple	ements the constan	it val	ue be	efore	insei	rting		
Machine States	1									
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z 1 if the bit test</li> <li>V Unaffected</li> </ul>	ed is 0, <i>0</i> if the b	it tested is 1							
Examples	Code	<u>Before</u>	<u>After</u>							
		A0	NCZV							
	BTST 0,A0	55555555h	XX OX							
	BTST 15,A0	55555555h	XX 1X							
	BTST 31,A0	55555555	XX 1X							
	BTST 0,A0	AAAAAAAA	XX IX							
	BTST 15,A0		XXUX							
	BTST 31,AU	EEEEEEE								
	BIST U,AU	FFFFFFFF								
		FEFEFEED								
		000000000	x x 1 x							
	BTST 15.A0	00000000h	xx 1 x							
	BTST 31,A0	00000000h	x x 1 x							

Syntax	BTST Rs, Ra	1					
Execution	Set status on v	value of specifie	ed bit in Rd				
Instruction Words	15 14 13	12 11 10	9876	54	32	1	0
	0 1 0	0 1 0	1 Rs	R	F	d	
Description	BTST tests a l This form of th specify the bit sents the sour	bit in the destina the BTST instruct in Rd that is test ce register). No	ation register and s stion uses the 5 LS sted (the symbol R te that the 27 MSE	sets statu SBs of the Is in the s Ss of Rs a	s bit Z ac source r yntax abc re ignore	cordi egist ve re d.	ngly. er to pre-
	Rs and Rd mu	st be in the san	ne register file.				
Machine States	1						
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z 1 if the bit</li> <li>V Unaffected</li> </ul>	1 1 tested is 0, <i>0</i> if 1	the bit tested is 1				
Examples	<u>Code</u>	Before		<u>After</u>			
		A1	A0	NCZ	v		
	BTST A1,A0	00000000h	55555555h	XXU	X		
	BTST A1,A0	0000000Fn	555555555	X X I	x		
	BTST AL,AU	0000001Fn		X X I V V 1	X		
	BTST AL,AU	00000000n			x		
	BTST AL,AU	0000000FN			×		
	BTST AL,AU	COUCOUTEN			~		
	BTST A1,A0	0000000h	EEEEEEE		~		
	BTST AL,AU	000000000	FFFFFFF		^ Y		
	BISI AL,AU	00000001 Fh	FEFEFEED	× × 0	×		
	BIGI AI,AU BUSU AI AO	00000000	00000000h	x x 1	x		
	BTST AL, AO	0000000Eb	00000000h	x x 1	x		
	BTST A1,A0	0000001Fh	00000000h	x x 1	x		

Syntax	CALL Rs												
Execution	PC' → TOS Rs → PC SP $-32$ →	SP											
Instruction Words	15 14 13 0 0 0	12 11 0 1	10 0	9 0	8 1	7 0	6 0	5 1	4 R	3	2 R	1 s	0
Description	CALL pushe jumps to a su can use this SP, SP is dec nal value of S	s the add ubroutine instructior remented SP).	ress o whose for in <b>after</b>	f the addr dexe being	next i ess is d subr writte	instru cont routin en to t	ictior aine ie ca he P	n (PC d in t IIs. N C (th	C') or the s lote t te PC	nto th ource that v Con	ne sta e reg vhen tains	ack, ister. Rs is the c	then You s the origi-
	The TMS340 tions are alw	20 always ays word	setst aligne	he4l d.	SBso	ofthe	prog	jram	cour	nter to	00, s	o ins	truc-
	The stack po external men addresses. P before the re affects timing	pinter (SP mory. The C' is push turn addr g as indica	) point e stac ed ont ess is ited in	s to t k gro to the loade <b>Mac</b>	he toj ows ir stack ed on <b>hine</b> \$	p of t n the a, and to the <b>State</b>	he si dire the s e state <b>s</b> , be	tack; ectior SP is ck. S elow.	the n of prec Stack	staci decr decre poir	k is le reasi emen nter a	ocate ng li ted b alignr	ed in near by 32 nent
	Use the RET	S instruct	on (pa	ige 1	3-220)	) to re	eturn	from	n a si	Jbrou	itine.		
Machine States	3 + (1) if the 3 + (4) if the	SP is alig SP is not	ned aligne	ed									
Status Bits	<ul> <li>N Unaffector</li> <li>C Unaffector</li> <li>Z Unaffector</li> <li>V Unaffector</li> </ul>	ed ed ed ed											
Example	CALL A0												
	<u>Before</u> A0 01234560h	<b>PC</b> 0444221	Dh	<b>SP</b> F000	020h	<u>Af</u> P( 01	<u>ter</u> C 2345	60h	s F	<b>SP</b> =0000	)00h		
	Memory cont	tains the f	ollowir	ng va	lues a	fter i	nstru	ictior	n exe	ecutic	on:		
	<b>Address</b> F000010h F000020h	<b>Data</b> 2220h 0444h											

Syntax	CALLA Addre	ess								
Execution	$\begin{array}{l} PC' \rightarrow TOS \\ Address \rightarrow P \end{array}$	С								
Instruction Words	15 14 13	12 11 10	9 8 7	6 5	4 3	2	1	0		
	0 0 0	0 1 1		1 0	1 1	1	1	1		
			16 LSBs of Addres	88 						
Description	CALLA pushes jumps to the ad is a 32-bit absol enced calls (gr	CALLA pushes the address of the next instruction (PC') onto the stack, then umps to the address contained in the 2 extension words. The <i>Address</i> operand $3 \times 32$ -bit absolute address. This instruction is used for long or externally referenced calls (greater than $\pm 32$ K words).								
	The 4 LSBs of always word al	the program co igned.	ounter are alwa	ays set to	0, so in	istruc	tions	s are		
	The stack poin external memoraddress. PC' is before the returaffects timing a	ter (SP) points bry. The stack pushed onto the rn address is to is indicated in <b>N</b>	to the top of the grows in the stack, and the baded onto the <b>Machine State</b>	ne stack; f direction he SP is p stack. St <b>s</b> , below.	the stac of dec predecre tack poir	k is lo reasi ment nter a	ocate ng li ted b tignr	ed in near by 32 ment		
	Use the RETS	instruction (pag	e 13-220) to re	turn from	a subroi	utine.				
Machine States	3 if immediate 4 4 if SP is also 1 3 + (3) if immed 4 + (3) if SP is	data is long-wo ong-word align diate data is no also not long-w	rd aligned ed t long-word aliç ord aligned	gned						
Status Bits	<ul><li>N Unaffected</li><li>C Unaffected</li><li>Z Unaffected</li><li>V Unaffected</li></ul>									
Example	CALLA 012345	60h								
	Before PC 04442210h Memory contai	SP 0F000020h	After PC 01234560h	SP 0F00000	i0h executio	<b>.</b>				
	memory contai		values allel II	isti uotioff	executio					
	<b>Address</b> F000010h F000020h	<b>Data</b> 2240h 0444h								

Syntax	CALLR Address									
Execution	PC' $\rightarrow$ TOS PC' + (offset × 16)	→ PC								
Instruction Words	15 14 13 12	11 10 9	8 7	6 5	4	3	2	1	0	
	0 0 0 0	1 1 0		0 1	1	1	1	1	1	
Description	CALLR pushes the jumps to the subro tion address and th a specified module	e address of t utine at the ad ne signed word e or section.	he next ins dress spec d offset. Thi	truction ified by t s instruc	(PC') c he sum ction is	onto n of t used	the s he ne d for c	tack, xt ins alls w	then truc- /ithin	
	The <i>Address</i> oper 32,767) <b>of the PC</b> . assembler does n defined within a dif value for the opco- instruction word in tion.	The <i>Address</i> operand is a 32-bit address within $\pm$ 32K words (–32,768 to 32,767) <b>of the PC</b> . The address must be defined within the current section; the assembler does not accept an address value that is externally defined or defined within a different section than PC'. The assembler calculates the offser value for the opcode as (Address – PC')/16, where PC' is the address of the instruction word immediately following the second word of the CALLR instruction.							68 to a; the ed or offset of the truc-	
	The 4 LSBs of the program counter are always set to 0, so instructions are always word aligned.									
	The stack pointer (SP) points to the top of the stack; the stack is located in external memory. The stack grows in the direction of decreasing linear address. The PC is pushed on to the stack, and the SP is predecremented by 32 before the return address is loaded onto the stack. Stack pointer alignment affects timing as indicated in Machine States, below.								ed in near ed by ment	
	Use the RETS inst	ruction (page	13-217) to	return fr	om a s	ubrc	outine			
Machine States	3 + (1) if the SP is 3 + (4) if the SP is	aligned not aligned								
Status Bits	<ul><li>N Unaffected</li><li>C Unaffected</li><li>Z Unaffected</li><li>V Unaffected</li></ul>									
Examples	<u>Code</u>	Before		Af	<u>ter</u>					
	CALLR 0447FFF0h CALLR 04480000h	<b>PC</b> 04400000h 04400000h	SP 0F000020 0F000020	PC Dh 04 Dh 04	2 47FFF0 480000	)h )h	SP 0F00 0F00	10000	า า	
	Memory contains	the following	alues after	r instruct	tion ex	ecut	ion:			
	Address         Date           F000010h         00           F000020h         04	ata 020h 140h								

Syntax	CEX	CEXEC size, command [, ID] [ ,L]														
Execution	ID, d	comm	and	$\rightarrow$	Cop	oroce	essor	com	imanc	ł						
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	0	0	0	0	0	0	0	0
		8 LS	Bs of o	coproc	essoi	comr	nand		size	0	0	0	0	0	0	0
	сор	rocess	or ID				13	MSB	s of co	oroces	ssor c	omma	and			
Description	CEXEC sends a 21-bit <i>command</i> to a coprocessor. The coprocessor operate on the <i>command</i> without any transfer of data. The <i>size</i> operan value of 0 or 1 (0 is the default); the coprocessor interprets the size bit to mine the size of the values to be operated upon:						ssor eranc it to d	may I is a eter-								
		lf <i>size</i>	e = 0,	then	the	copr	oces	sor ι	ises 3	2-bi	t val	ues.				
		lf <i>size</i>	9 = 1,	then	the	copr	oces	sor ι	ises 6	i4-bi	t val	ues.				
	The supp	<i>ID</i> op oly thi	erano s ope	d is a rand	n opt , it de	ional efault	3-bi s the	t copi valu	roces: e spe	sor ic cified	lentif I in th	icati e coj	on co oroce	de; if essor	you direc	don't tive.
	The man	outpu d cyc	ut of t le (w	this ir hen s	nstru SF is	ction high	on t ı) is a	he L as fol	AD bu llows:	us at	ALT	CH I	ow di	uring	the	com-



For more information, refer to Section 10.3, <u>Formats of Commands Passed to</u> <u>a Coprocessor</u>, on page 10-5.

The TMS34020 assembler checks the coprocessor command mode bits to determine which form of CEXEC is used. If both mode bits (6 and 7 of the coprocessor command) are 0, a short CEXEC is generated, otherwise a long CEXEC is generated. You can force a long CEXEC by placing an **,L** at the end of the CEXEC operand.

Machine States	<ul><li>2 (1) if the immediate data is long-word aligned</li><li>3 (1) if the immediate data is not long-word aligned</li></ul>
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>
Example 1	This example sends coprocessor 2 the instruction <i>wxyz</i> and tells the coprocessor to use 32-bit values.

CEXEC 0,wxyz,2

*Example 2* This example sends the default coprocessor the instruction *qrst* and tells the coprocessor to use 64-bit values.

CEXEC 1,qrst

*Example 3* This example compares the single-precision (Size = 0, T = 1) contents (fpuop = 00010) of TMS34082 coprocessor 3 (ID = 3) registers RA3 (ra = 0011) and RB4 (rb = 0100). The result is stored in the coprocessor's RB0 register (rd = 10000).



Syntax	CEXEC size, command [, ID]
Execution	$ID, command \rightarrow Coprocessor command$
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         1       1       0       1       1       0       0       0       0       6 LSBs of coprocessor command       Size         coprocessor ID       13 MSBs of coprocessor command       13 MSBs of coprocessor command       1       1
Description	<ul> <li>The CEXEC instruction sends a 21-bit <i>command</i> to a coprocessor. This version of CEXEC outputs bits 14 and 15 as 0 when sending the instruction, so that 19 bits of the 21-bit <i>command</i> may be set to any combination of 1s and 0s necessary to represent a particular coprocessor operates on the <i>instruction</i> without only 19 may be specified). The coprocessor operates on the <i>instruction</i> without any transfer of data. The <i>size</i> operand is a value of 0 or 1; it determines the size of the values that the coprocessor uses:</li> <li>If <i>size</i> = 0, then the coprocessor uses <b>32-bit values</b>.</li> <li>If <i>size</i> = 1, then the coprocessor uses <b>64-bit values</b>.</li> <li>The <i>ID</i> operand is an optional 3-bit coprocessor identification code; if you don't supply this operand, it defaults to the value specified in the coprocessor directive.</li> <li>The assembler uses this version of CEXEC when bits 6 and 7 of the coprocessor is or instruction (corresponding to bits 14 and 15 on the LAD bus) are 0. You can force a long CEXEC by placing an ,L at the end of the CEXEC operand.</li> <li>The output of this instruction on the LAD bus at ALTCH low during the command cycle (when SF is high) is as follows:</li> </ul>
LAD #	



For more information, refer to Section 10.3, <u>Formats of Commands Passed to</u> <u>a Coprocessor</u>, on page 10-5.

#### Machine States 2 (1)

Status Bits

- N Unaffected
- C Unaffected
- Z Unaffected
- V Unaffected

*Example 1* This example sends coprocessor 2 the instruction *wxyz* and tells the coprocessor to use 32-bit values.

CEXEC 0,wxyz,2

*Example 2* This example sends coprocessor 0 the instruction *qrst* and tells the coprocessor to use 64-bit values.

CEXEC 1,qrst

Example 3

This example divides the double-precision (Size = 1, T = 1) contents (fpuop = 01001) of the default TMS34082 coprocessor registers RA2 (ra = 0010) by RB7 (rb = 0111). The result is stored in the coprocessor's RA8 register (rd = 01000).

DIVD\_RA2\_RB7 .set 04E813h CEXEC 1,DIVD\_RA2\_RB7 ; size = 1 ID = default

0		4	4			E	Ξ			8	3			1	I		3				=	4E813h
0	0	1	0	0	1	1	1	0	1	0	0	0	0	0	0	1	0	0	1	1		
	r	а			r	b				rd			m	d		fj	ouo	р		Т		

Syntax	CLIP		
Execution	CLIP destination array adjusted DYDX $\rightarrow$ DYDX adjusted DADDR $\rightarrow$ DAD	DR	
Instruction Words	15         14         13         12         11         10           0         0         0         0         1         0	9 8 7 0 0 1	6     5     4     3     2     1     0       1     1     1     0     0     1     0
Description	CLIP adjusts the pixel array, and the XY dimensions cont specified by WSTART and replace the original values i the window and no adjustme lies entirely outside the wind DYDX are left unchanged. CLIP is the only instruction t ing array is different from a more information, refer to su	specified by the ained in DYDX, I WEND. The a in DADDR and ent is necessary, dow, then the Z hat can deal wit n array that sin ubsection 12.7.4	e XY address contained in DADDR to fit within the rectangular window adjusted pointer and dimensions DYDX. If the array currently fits in the V bit will be set to 0. If the array bit is set to a 1, and DADDR and h overflowing arrays; an overflow- nply strays outside a window. For 4.4, <u>Clip Instruction for Preclipping</u>
Implied Operands	<u>a Pixel Array</u> , on page 12-2	3.	
Implied Operands	a Pixel Array, on page 12-2 Register Name	3. Format	Description
Implied Operands	<u>a Pixel Array</u> , on page 12-2 Register Name B2 DADDR	3. Format XY	Description Destination array address
Implied Operands	<u>a Pixel Array</u> , on page 12-2 Register Name B2 DADDR B5 WSTART	3. Format XY XY	Description Destination array address Window start corner
Implied Operands	<u>a Pixel Array</u> , on page 12-2 Register Name B2 DADDR B5 WSTART B6 WEND	3. Format XY XY XY XY	Description Destination array address Window start corner Window end corner
Implied Operands	a Pixel Array, on page 12-2RegisterNameB2DADDRB5WSTARTB6WENDB7DYDX	3. Format XY XY XY XY XY	Description Destination array address Window start corner Window end corner Array dimensions
Implied Operands Machine States	a Pixel Array, on page 12-2RegisterNameB2DADDRB5WSTARTB6WENDB7DYDXComplex Instruction	3. Format XY XY XY XY XY	Description Destination array address Window start corner Window end corner Array dimensions
Implied Operands Machine States Status Bits	a Pixel Array, on page 12-2RegisterNameB2DADDRB5WSTARTB6WENDB7DYDXComplex InstructionNUnaffectedCUnaffectedZZ = 1 if the array lies entirinside the windowV1 if any portion of the array	3. Format XY XY XY XY xY rely outside wind	Description         Destination array address         Window start corner         Window end corner         Array dimensions         dow, 0 if the array lies all or partially         the window, 0 otherwise

·

2000

This routine makes the following assumptions:

These B registers and I/O registers have been set up by the calling program:

B-file registers DPTCH, OFFSET, WSTART, WEND and COLOR1

I/O registers CONTROL, CONVDP, PSIZE, PMASK and CONFIG

- The system contains a global flag \_vfill\_ok which is cleared if the VFILL is not possible. Reasons for this may be:
  - DPTCH is not an integral multiple of 80h
  - PSIZE is 1 or 2
  - Pixel processing is not set to replace
  - Transparency is not set
  - The system does not contain VRAMs which support this feature

DADDR DYDX CONTROL	.set .set .set	B2 B7 0C00000B0h	;Destination address register ;Delta X/delta Y register ;Control register
	.globl .ref	_fill_rect ; provide _vfill_ok	e reference for external calls ; flag to enable VFILLs
fill r	ect:		
	mmtm	SP, B2, B7, B10, B11, B12	;save required registers
	move	A14,B10 ;move c-s	stack pointer into B-file
	move	*-B10,DYDX,1	;get width
	move	*-B10,B12,1	;get height
	sll	16,B12	
	movy	B12,DYDX	;concatenate width & height
	move	*-B10,DADDR,1	;get xleft
	move	*-B10,B12,1	;get ytop
	move	B10,A14	;restore c-stack pointer
	sll	16,B12	
	movy	B12, DADDR	;concatenate xleft & ytop
	move	e_vill_ok,A8,1	;get state of viill flag
	jrz	no_vtill	
	clib	• •	;clip to the window
	jrz	exit	; if outside the window, exit
	cvaxyı	DADDR	; convert to linear dest address
	VICOL	т	; load VRAM color latches
	VIIII irua		;perform linear lill
no wfil	]1.	exit	
110_viiii	 fill	xy •fill +be	rectangle using standard fill
ovit.	* * * * *		s rectangle using standard fiff
CAIL.	~~£~	CD D2 D7 D10 D11 D12	exectors required registers
		SF, DZ, DI, DIV, BII, BIZ	restore required registers
	rets	2	

Syntax	CLR Rd								
Execution	Rd XOR Rd	→ Rd							
Instruction Words	15 14 13 0 1 0	12 11 1	0 9 8	7 <u>65</u> Rd	4 R	3	2 R	1 d	
Description	CLR clears th itself. This is	ne destination an alternate r	register by XC	Ring the co KOR Rd,Rd	ntents (page	s of th e 13-2	ne reg 266 ).	jister	<sup>.</sup> with
Machine States	1								
Status Bits	<ul> <li>N Unaffecter</li> <li>C Unaffecter</li> <li>Z 1</li> <li>V Unaffecter</li> </ul>	ed ed ed							
Examples	<u>Code</u>	<u>Before</u> A0	<u>After</u> A0	NCZV					
	CLR A0	FFFFFFFFh	00000000h	xx1x					
	CLR A0	00000001h	00000000h	xx1x					
	CLR A0	80000000h	00000000h	xx1x					
	CLR A0	AAAAAAAAh	00000000h	xx1x					

## CLRC Clear Carry

Syntax	CLRC										
Execution	0 → C										
Instruction Words	15 14 0 0	13 12 0 0	11 10 0 0	9 8 1 1	3 7 0	6 0	5 4 1 0	3 0	2	1 0	0
Description	CLRC sets ter is unaf	s the C (car fected. Us	rry) bit in e SETC i	the stat instruct	us regis ion (pag	ter to 0 je 13-2	); the re 226) to	st of tl set th	ne sta e C b	atus ro pit.	egis-
	This instru a subrouti	uction is us ne without	seful for r using a	eturnin genera	g a true I-purpo	/false se reg	value ( ister.	n the	carry	y bit)	from
Machine States	1										
Status Bits	B Unaffe C 0 Z Unaffe V Unaffe	ected ected ected									
Examples	CLRC CLRC CLRC CLRC	<u>Before</u> ST F0000000 40000010 B000001F	<b>N</b> h 1 h 0 īh 1	<b>C Z V</b> 1 1 1 1 0 0 0 1 1	<u>After</u> ST B0000 00000 B0000	0000h 0010h 001Fh	N 1 ( 0 ( 1 (	<b>C Z V</b> 0 1 1 0 0 0 0 1 1			

Syntax	СМС	VCG	i Ro	d <sub>1</sub> [,	Rd <sub>2</sub>	[, siz	:e]], c	comn	nand	[, <i>ID</i> ]						
Execution	<i>ID, c</i> e Copr	omm oces	<i>and</i> sor	$\rightarrow$ $\rightarrow$	Cop Rd <sub>1</sub>	oroce , <i>Rd<sub>2</sub></i>	ssor	com	mand							
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	0	1	1	R		R	d <sub>1</sub>	
		8 LSI	3s of c	oproc	essor	comn	nand		size	0	0	R		R	d <sub>2</sub>	
	copr	ocesso	or ID				13	MSBs	s of cop	roces	sor co	omma	nd			
Description	CMC TMS I i	VCG 3402 f <i>size</i> gnore	i mov 0 des = 0, ed; bi	res or stinat then ts 0 t	ne or tion r the o the o	two egist copro of the	32-bi er or ocess seco	t valu regis sor m ond ii	ues fro sters: oves nstruo	om a one : ction	copr 32-bi worc	oces it valı 1 are	sor to ue inf set t	the to <i>Rc</i> o 0.	spec d <sub>1</sub> . <i>R</i> a	ified d <sub>2</sub> is
	li T c	f <i>size</i> he o opro	= 1, rder cess	then in wh or us	the c nich t ed.	opro he N	icess ISBs	or mo and	oves a LSBs	a 64- s are	bit va trans	alue i sferre	in to <i>l</i> ed de	Rd <sub>1</sub> a epend	and <i>I</i> ds or	<i>קd₂.</i> ו the
	The commond encycled encyclication instruction (01 bits of information define															

The *command* operand specifies an instruction (21 bits of information define the instruction) that the coprocessor should execute to define the source data for the move. The *ID* operand is an optional 3-bit coprocessor identification code; if you don't supply this operand, it defaults to the value specified in the coprocessor directive.

The output of this instruction on the LAD bus at ALTCH low during the command cycle (when SF is high) is as follows:



	<i>I</i> is the coprocessor parameter index bit. For more information, refer to Section 10.3, <u>Formats of Commands Passed to a Coprocessor</u> , on page 10-5.
Machine States	If size = 04 if the immediate data is long-word aligned5 if the immediate data is not long-word alignedIf size = 16 if the immediate data is not long-word aligned6 if the immediate data is not long-word aligned
Status Bits	<ul> <li>N 1 if the last 32-bit value read is negative, 0 otherwise</li> <li>C Unaffected</li> <li>Z 1 if the last 32-bit value read is 0, 0 otherwise</li> <li>V 0</li> </ul>

*Example 1* This example moves 32 bits, specified by coprocessor instruction *wxyz*, from coprocessor 2 to register A0.

CMOVCG A0, wxyz, 2

*Example 2* This example moves one 64-bit value, specified by coprocessor instruction *jklm*, from coprocessor 5 to registers A3 and B7.

CMOVCG A3, B7, 1, jklm, 5

Example 3

This example moves a double-precision floating-point value from TMS34082 coprocessor number 2. The 32 MSBs are stored in A0 and the 32 LSBs are stored in A1. This example assumes that the LOAD bit of the TMS34082 configuration register is set for MSBs transferred before LSBs.

The coprocessor command moves a double-precision (T = 1, size = 1) value from the coprocessor register (fpuop = 00111) RB2 (rd = 10010) to a TMS34020 register (md = 01).

]	мо	v_1	RB2	2_2	0			.s CM	et 0V(	CG	0 A	01 0,	241 A1,	FΗ ,1,	MC	v_	RB2	2_2	20,	2;	s	ize	=	1	ID	=	2
1	0		(	D		1	•	1			2	2			4	4			۱	F		=	0.	124	4Fh		
	0	0	0	0	0	0	0	0	1	0	0	1	0	0	1	0	0	1	1	1	1	]					
		r	a			rl	b				rd			m	d		fŗ	buo	р		т						

Syntax	CMOVCM	*Rd-	<b>⊦,</b> cou	nt, siz	:e,c	omma	and [,	ID]						
Execution	ID, commar	nd →	• Co	proce	ssor	com	mand							
	If size = 0, Repeat <i>count</i> times (1 $\leq$ <i>count</i> $\leq$ 32): Coprocessor $\rightarrow *Rd$ Rd + 32 $\rightarrow Rd$													
	If size = 1, F Coproce Rd + 32 Coproce Rd + 32	Repea essor ! → essor ! →	tt coui → Rd → Rd	nt time ∗Rd ∗Rd	es (1	≤ CO	unt ≤	16):						
Instruction Words	15 14 1	3 12	11	10	9	8	7	6	5	4	3	2	1	0
	0 0	0 0	0	1	1	0	1	0	1	R		R	d	
	8 LSBs	of cop	ocesso	or comr	nand		size	0	0		tr	ransfer	'S	
	coprocessor	ID			13	B MSBs	s of cop	proces	sor co	omma	nd			
Description	This version ing on the va execution, F each transfe address.The values that If size = In this c	of the lue of d cor er, the size are tra 0, the ase, c	e CMC transi itains e valu opera ansfer en cou count	)VCN fers)t the 32 Je in and is red: <i>unt</i> sp must	l inst from 2-bit <i>Rd</i> i a va ecific be a	ructio a cop addre s inc alue c es the	n mo roces ess of remen of 0 o e num e of 1	ves c sort the f nted r 1; i ber c $\rightarrow$ 3	one o o me irst lo by 3 t det of <b>32</b> 32.	r mory mory ocatic 32 to ermin -bit t	re va v. Bef on in poil nes t	ilues iore ir mem nt to the si	(dep istruction ory; the ize o to m	end- ction after next f the nake.

□ If size = 1, then *count* specifies the number of **64-bit transfers** to make. In this case, *count* must be a value of  $1 \rightarrow 16$ .

The value of *transfers* is set by the assembler, according to the values of *size* and *count*:

	If size = 0 and count = $1 \rightarrow 31$ , If size = 0 and count = 32,	then <i>transfers = count</i> then <i>transfers =</i> 0
C.	If size = 1 and count = $1 \rightarrow 15$ , If size = 1 and count = 16,	then <i>transfers</i> = 2x <i>count</i> then <i>transfers</i> = 0

The *command* operand specifies an instruction (21 bits of information define the instruction) that the coprocessor should execute to specify the source data for the move. The *ID* operand is an optional 3-bit coprocessor identification code; if you don't supply this operand, it defaults to the value specified in the coprocessor directive.
The output of this instruction on the LAD bus at ALTCH low during the command cycle (when SF is high) is as follows:

LAD # 31-29 28	
ID	coprocessor command size 0 0 0
	13 MSBs 8 LSBs
	For more information, refer to Section 10.3, Formats of Commands Passed to a Coprocessor, on page 10-5.
Machine States	5 + [transfers –1] if the immediate data is long-word aligned 6 + [transfers –1] if the immediate data is not long-word aligned (assumes that there are no 32-bit transfers)
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>
Example 1	This example moves thirty-two 32-bit values, specified by the coprocessor instruction <i>wxyz</i> , to memory from coprocessor 2 (register A0 contains the address of the first memory location).
Example 2	CMOVCM *A0+, 32, 0, wxyz, 2 This example moves eight 64-bit values, specified by the coprocessor instruc- tion <i>qrst</i> , to memory from the default coprocessor (register B7 contains the address of the first memory location).
	CMOVCM *B7+,8,1,qrst
Example 3	This example moves 5 (count = 5) 32-bit integers from coprocessor 1 (ID = 1) to the postincremented memory block pointed to by A3. The coprocessor command specifies a TMS34082 Move To Host Memory (fpuop = 00111, md = 10) of integer (size = 0, T = 0) quantities starting in coprocessor register RB3 (rd = 10011).
	The memory location pointed to originally by A3 will receive the first 32-bit integer transferred (from coprocessor register RB3). Memory location A3+10h will receive the second integer (from RB4). The third through fifth integers will be placed into A3+20h, A3+30h, and A3+40h (from RB5, RB6, and RB7 respectively).
	MOVE5_RB3 .set 00138Eh CMOVCM *A3+,5,0,MOVE5_RB3,1; count=5 size=0 ID=1
	0       0       1       3       8       E       =       0138Eh         0       0       0       0       0       1       0       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0       0       1       1       1       0<

Constant	~		,					(0)						
Syntax	CN	OVCM -*Rc	1, COL	int, si	ze, c	omma	and [,	[D]						
Execution	ID,	command –	• Co	oproc	essoi	. com	mand							
	lf s	ize= 0, Repea <i>Rd</i> – 32 → Coprocessor	t coui Rd →	nt tim *Rd	es (1 ,	≤ COl	unt ≤ \$	32):						
	ļf s	ze = 1, Repear $Rd - 32 \rightarrow$ Coprocessor $Rd - 32 \rightarrow$ Coprocessor	t cou Rd → Rd →	nt tim *Rd *Rd	nes (1 ,	≤ CO	unt ≤	16):						
Instruction Words	15	14 13 12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0 0 0	0	1	1	0	1	1	0	R		F	٦d	
		8 LSBs of copr	ocess	or instr	uction		size	0	0		t	ransfe	rs	
	со	processor ID			13	MSB	s of cop	roces	sor in	struct	ion	·		
Description	Thi ing exe bef ado valu	s version of the on the value of oution, <i>Rd</i> co ore each trans tress.The <i>size</i> ues that are tra	e CM trans ntain fer, th oper ansfe	OVCN s <i>fers</i> ) s the ne val and i rred:	V inst from 32-b ue in s a v	ructic a cop it add <i>Rd</i> is alue d	on mo roces dress decre of 0 o	ves o sort of the mer r 1; i	one c o me he fin ited t it det	or mo mory rst Io by 32 ermi	re va v. Be catio to p nes	alues fore in on in oint t the s	(dep nstru men o the ize o	end- ction nory; next of the
	G	If <i>size</i> = 0, the In this case, c	en co count	<i>unt</i> s mus	becifi t be a	es the a valu	e num le of 1	iber ( → 3	of <b>32</b> 32.	-bit 1	tran	sfers	to m	iake.
	G	If <i>size</i> = 1, th In this case, <i>c</i>	en cc count	<i>unt</i> s mus	pecifi t be a	es th a valu	e num le of 1	nber → 1	of <b>64</b> 16.	-bit 1	tran	sfers	to m	nake.
	The and	e value of <i>trans</i> I <i>count</i> :	fers	is set	by th	e ass	emble	er ac	cord	ing to	o the	valu	es of	size
	<b>.</b>	If <i>size</i> = 0 and If <i>size</i> = 0 and	l cou l cou	nt = 1 nt = 3	→ 3 32,	1,	th th	ien <i>ti</i> ien <i>ti</i>	ransf ransf	ers = ers =	= <i>col</i> = 0	ınt		
	G	If <i>size</i> = 1 and If <i>size</i> = 1 and	l cou l cou	nt = 1 nt = 1	→ 1 6,	5,	th th	ien ti ien ti	ransf ransf	ers = ers =	= 2x = 0	coun	t	

The *command* operand specifies an instruction (21 bits of information define the instruction) that the coprocessor should execute to specify the source data for the move. The *ID* operand is an optional 3-bit coprocessor identification code; if you don't supply this operand, it defaults to the value specified in the coprocessor directive.

The output of this instruction on the LAD bus at ALTCH low during the command cycle (when SF is high) is as follows:

LAD # 31—29 28 <b>~</b>			8 7 6	543		0								
	coprocessor command		size	0 0	0									
	13 MSBs	8 LSBs												
	For more information, refer <u>a Coprocessor</u> , on page 1	to Section 10.3, <u>Fc</u> 0-5.	ormats of Co	ommand	ls Pas	sed to								
Machine States	5 + [transfers –1] if the imn 6 + [transfers –1] if the imn	nediate data is long nediate data is not	g-word aligr long-word a	ned aligned										
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>												
Example 1	This example moves ninete tion <i>wxyz</i> , from coprocesso memory address).	en 32-bit values, sp or 3 to memory (re	pecified by o gister A3–3	coproce: 32 speci	ssor ir fies th	nstruc- ne first								
	CMOVCM -*A3,19,0,wxyz,	3												
Example 2	This example moves two 6 <i>qrst</i> , from the default coproc memory address).	64-bit values, speci cessor to memory (r	fied by cop register A3–	rocesso -32 spec	r instr ifies tl	uction ne first								
	CMOVCM -*A0,2,1,qrst													
Example 3	This example moves eight <i>wxyz</i> , from coprocessor 3 memory address).	64-bit values, spec 3 to memory (regi	ified by cop ister B7–32	rocesso 2 specifi	r instr ies th	uction e first								
	CMOVCM -*B7,8,1,wxyz,3													
Example 4	This example moves 5 (cou TMS34082 coprocessor to A1. The coprocessor comr 00111, md = 10) of integer of coprocessor register RA1 (	unt = 5) 32-bit quan the predecrement mand specifies a M quantities (size = 0, (rd = 00001).	itities (size ed memory /love to Ho T = 0), with	= 0) from block p st Memo n the sta	n the c ointec ory (fp rting s	default d to by uop = source								
	The first integer is transferre second integer number from the last from RA5.	ed from the default n RA2, the third fror	coprocessc n RA3, the f	or's RA1 ourth fro	regist m RA	er, the 4, and								

The memory location pointed to by A1–10h is the destination for the 32 bits of the first integer to be transferred. Memory location A1–20h is the destination for the second integer, with the remaining destination addresses as A1–30h, A1–40h, and A1–50h. After the transfer, A1 will point to the last integer transferred.

MOVE\_RA1\_20 .set 00018Eh CMOVCM \*-A1,5,0,MOVE\_RA1\_20;count = 5 size = 0 ;ID = default

0		(	)			(	)			1	I			8	3			E	=		= 0018Eh
0	0	0	0	0	0	0	0	0	0	0	0	1	1	0	0	0	1	1	1	0	
	ra	a			r	b		1		rd			m	d		fŗ	ouo	р		Т	

Syntax	CMOVCS command [, ID]											
Execution	D, <i>command</i> → Coprocessor command Coprocessor → ST											
Instruction Words	<u>15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0</u>											
	0 0 0 0 0 1 1 0 0 1 1 0 0 0 0 0 0											
	8 LSBs of coprocessor command         0         0         0         0         0         0         0         1											
	coprocessor ID 13 MSBs of coprocessor command											
Description	CMOVCS moves one value from a coprocessor to the TMS34020 status register. (Note that this is a special coding of the CMOVCG instruction.) The value is masked by the TMS34020 so that its 4 MSBs are placed in the status register N, C, Z, and V bits. The remaining bits from the coprocessor are ignored by the TMS34020.											
	The <i>command</i> operand specifies an instruction (21 bits of information define the instruction) that the conrocessor should execute to specify the source data											

the instruction) that the coprocessor should execute to specify the source data for the move. The *ID* operand is an optional 3-bit coprocessor identification code; if you don't supply this operand, it defaults to the value specified in the coprocessor directive.

The output of this instruction on the LAD bus at ALTCH low during the command cycle (when SF is high) is as follows:



	For more information, refer to Section 10.3, <u>Formats of Commands Passed to</u> <u>a Coprocessor</u> , on page 10-5.
Machine States	4 if the immediate data is long-word aligned 5 if the immediate data is not long-word aligned
Status Bits	<ul> <li>N Set to bit 31 from the coprocessor data.</li> <li>C Set to bit 30 from the coprocessor data.</li> <li>Z Set to bit 29 from the coprocessor data.</li> <li>V Set to bit 28 from the coprocessor data.</li> </ul>
Example 1	This example moves data from coprocessor 2 to the status register after issuing the coprocessor instruction <i>wxyz</i> .
	CMOVCS wxyz,2
Example 2	This example moves data from the default coprocessor to the status register after issuing the coprocessor instruction <i>qrst</i> .
	CMOVCS qrst

TMS34020 Assembly Language Instruction Set

Syntax	СМС	VGC	C Rs	s, co	mma	nd [,	ID]									
Execution	ID, c Rs	omm →	<i>and</i> Copr	→ oces	Cop sor	oroce	ssor	com	mano	ł						
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	З	2	1	0
	0	0	0	0	0	1	1	0	0	0	1	R		F	ls	
		8 LS	Bs of c	oproc	essor	comn	nand		0	0	0	0	0	0	0	0
	copr	ocess	or ID				13	MSBs	of co	proces	ssor c	omma	and			
Description	This regis bits c	versi ter to of info	on of a co ormat	the proc ion d	CMC esso lefine	VGC r. The	c inst e <i>cor</i> instru	ructio n <i>mar</i> uctior	on mo nd op n) tha	oves eran It the	the c d spe copr	onte ecifie oces	nts c s an sor s	f a T instri houl	MS3 uctio d exe	4020 n (21 ecute

bits of information define the instruction) that the coprocessor should execute to specify the destination for the move. The *ID* operand is an optional 3-bit coprocessor identification code; if you don't supply this operand, it defaults to the value specified in the coprocessor directive.

The output of this instruction on the LAD bus at ALTCH low during the command cycle (when SF is high) is as follows:



For more information, refer to Section 10.3, <u>Formats of Commands Passed to</u> <u>a Coprocessor</u>, on page 10-5.

Machine States	2 (1) if the immediate data is long-word aligned 3 (1) if the immediate data is not long-word aligned
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>
Example 1	This example moves 32 bits to coprocessor 2 (the destination is specified by the coprocessor instruction <i>wxyz</i> ) from register A0.
	CMOVGC A0,wxyz,2
Example 2	This example moves 32 bits to the default coprocessor (the destination is specified by the coprocessor instruction <i>qrst</i> ) from register B7.
	CMOVGC B7,qrst

Example 3

This example loads a TMS34082 register with a 32-bit integer from a TMS34020 register. The source register is B5 and the destination register is in the default coprocessor.

The coprocessor command loads (fpuop = 00110) the 32-bit integer (T = 0, size = 0), sent from the TMS34020 register (md = 01), into register RA3 (rd = 00011).

MOVE\_R3 .set 00034CH CMOVGC B5,MOVE\_RA3 ; ID = default

l	0		C	)			(	)			3	3			2	1			C	)		=	0034Ch
	0	0	0	0	0	0	0	0	0	0	0	1	1	0	1	0	0	1	1	0	0		
Ī		ra	a			r	b				rd			m	d		fŗ	ouo	р		Т		

Syntax	СМС	VGC	Rs	s <sub>1</sub> , R	s <sub>2</sub> , s	ize, c	comn	nand	[, <i>ID</i> ]								
Execution	ID, c Rs <sub>1</sub> ,	omm Rs <sub>2</sub>	and →	→ Cop	Cop proce	oroce essor	ssor	com	mand								
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0	
	0	0	0	0	0	1	1	0	0	1	0	R		R	s <sub>1</sub>		
	8 LSBs of coprocessor command size 0 0 R												R	Rs <sub>2</sub>			
	copr	ocess	or ID				13	MSBs	s of cop	roces	sor c	omma	ind				
Description	This regis sor ir	versio ters t nterpi	on of o a co rets ti	the C oproc he <i>si</i>	MO cesso <i>ze</i> bi	/GC or.Th t to d	instru e <i>siz</i> e leterr	uctior e ope nine	n mov erand how 1	es th is a v to mo	e cor /alue ove t	ntent of 0 he da	s of t or 1; ata:	wo Tl the c	MS34 copro	1020 .ces-	
		f <i>size</i>	= 0, t	then	the v	alues	s that	are r	nove	dare	two	sepa	rate	32-b	it val	ues.	
	Li I V	f <i>size</i> v <b>alue</b>	= 1, t	then	the v	alues	s that	are r	nove	dare	two	halve	es of	a sin	gle <b>6</b> 4	4-bit	

The *command* operand specifies an instruction (21 bits of information define the instruction) that the coprocessor should execute to specify the destination for the move. The *ID* operand is an optional 3-bit coprocessor identification code; if you don't supply this operand, it defaults to the value specified in the coprocessor directive.

The output of this instruction on the LAD bus at ALTCH low during the command cycle (when SF is high) is as follows:



*I* is the coprocessor parameter index bit. For more information, refer to Section 10.3, Formats of Commands Passed to a Coprocessor, on page 10-5.

Machine States3 (1) if the immediate data is long-word aligned4 (1) if the immediate data is not long-word aligned

Status Bits

- N Unaffected
- C Unaffected
- Z Unaffected
- V Unaffected

## *Example 1* This instruction moves two 32-bit values to coprocessor 2 (the destination is specified by the coprocessor instruction *wxyz*)from registers A0 and B3.

CMOVGC A0, B3, 0, wxyz, 2

Example 2This instruction moves one 64-bit value to the default coprocessor (the destination is specified by the coprocessor instruction *qrst*) from registers B7 and B8.

CMOVGC B7, B8, 1, qrst

Example 3

This example moves two 32-bit floating-point values (in registers B3 and B4) into a TMS34082 coprocessor number 0 register.

The coprocessor command loads (fpuop = 00110) the 32-bit floating point (T = 1, size = 0) data from the TMS34020 registers (md=01) into RA1 (rd = 00001) and RA2.

MOVEF\_RA1 .set 00014DH CMOVGC B3,B4,0,MOVEF\_RA1,0 ; size = 0 ID = 0

	0		(	)			0	)			1				4	ŀ			0	)		= 0014Dh
	0	0	0	0	0	0	0	0	0	0	0	0	1	0	1	0	0	1	1	0	1	- -
Ì		r	a			r	b				rd			m	d		fŗ	ouo	р		Т	

Syntax	<b>CMOVMC</b> * <i>Rs</i> +, count, size, command [, <i>ID</i> ]
Execution	ID, command $\rightarrow$ Coprocessor command
	If size= 0, Repeat <i>count</i> times (1 $\leq$ <i>count</i> $\leq$ 32): * <i>Rs</i> $\rightarrow$ Coprocessor <i>Rs</i> + 32 $\rightarrow$ <i>Rd</i>
	If size = 1, Repeat <i>count</i> times ( $1 \le count \le 16$ ): * $Rs \rightarrow Coprocessor$ $Rs + 32 \rightarrow Rd$ * $Rs \rightarrow Coprocessor$ $Rs + 32 \rightarrow Rd$
Instruction Words	
	0     0     0     0     1     1     0     1     0     0     transfers       8 L SBs of coprocessor command
	coprocessor ID     13 MSBs of coprocessor command
Description	This version of the CMOVMC instruction moves one or more values (depending on the value of <i>transfers</i> ) from memory pointed to by $Rs$ to the coprocessor designated by the coprocessor instruction. Before instruction execution, $Rs$ contains the 32-bit memory address of a 32-bit value; after each transfer, the value in $Rs$ is incremented by 32 bits to point to the next address. The <i>size</i> operand is a value of 0 or 1; it determines the size of the values that are transferred. If <i>size</i> = 0, then <i>count</i> specifies the number of <b>32-bit transfers</b> to make
	In this case, <i>count</i> must be a value of $1 \rightarrow 32$ .
	If <i>size</i> = 1, then <i>count</i> specifies the number of <b>64-bit transfers</b> to make. In this case, <i>count</i> must be a value of $1 \rightarrow 16$ .
	The value of <i>transfers</i> is set by the assembler according to the values of <i>size</i> and <i>count</i> :
	□If size = 0 and count = 1 $\rightarrow$ 31,then transfers = countIf size = 0 and count = 32,then transfers = 0
	□If size = 1 and count = 1 $\rightarrow$ 15,then transfers = 2x countIf size = 1 and count = 16,then transfers = 0
	The <i>command</i> operand specifies an instruction (21 bits of information define the instruction) that the coprocessor should execute to specify the destination for the move. The <i>ID</i> operand is an optional 3-bit coprocessor identification code; if you don't supply this operand, it defaults to the value specified in the coprocessor directive.
	The output of this instruction on the LAD bus at ALTCH low during the com- mand cycle (when SF is high) is as follows:



The first 64-bit single-precision number is transferred to coprocessor one register RB5, the second double number to RB6 and the third to RB7.

Svntax	СМС	) VMC	_*F	is co	nunt	size	com	mano	אר <i>וו</i>	וכ						
Execution			and	.0 00	Cor	v.20,	000 <i>m</i>			- - 1						
Execution	D, c	OIIIII	anu		Cot	noce	ssor	com	nano							
	lf siz F *	e = 0 R <i>s –</i> 3 R <i>s →</i>	, Rep 32 → • Co	beat Re opro	coun s cess	t time or	es (1	≤ C0I	unt ≤	: 32):						
	If siz F F F	e = 1 ?s – 3 ?Rs → ?s – 3 ?Rs →	, Rep 32 → • Co 32 → • Co	oeat Re opro Re opro	coun s cess s cess	t time or or	es (1	≤ COI	unt ≤	: 16):						
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	1	0	0	0	0	0	1		t	ransfers	;	
		8 LSE	3s of c	oproc	cessor	comm	hand		size	0	0	R		Rs		
	copr	ocesso	or ID				13	MSBs	ofco	proces	sor co	mmar	nd			
ing on the value of <i>transfers</i> ) from memory pointed to by <i>Rs</i> to the coprocessor designated by the coprocessor instruction. Before the first transfer, <i>Rs</i> i decremented to contain the 32-bit memory address of the first 32-bit value t transfer. The <i>size</i> operand is a value of 0 or 1; it determines the size of th values that are transferred:											ssor s is to the					
	⊑∎ ti Ii	f <i>size</i> n this	= 0, case	then e, <i>co</i>	ı coui unt ı	<i>nt</i> spe nust	ecifie be a	s the valu	e nun e of 1	nber o 1 → 3	of <b>32-</b> 2.	bit t	rans	sfers t	o ma	ake.
		f <i>size</i> n this	e = 1, case	ther e, <i>co</i>	n <i>cou</i> unt i	<i>nt</i> sp nust	ecifie be a	es the valu	e nur e of 1	nber o 1 → 1	of <b>64</b> 6.	-bit t	ran	sfers t	o ma	ake.
	The v and d	value count	of <i>tra</i>	ansfe	ers is	set b	y the	ass	embl	er ac	cordi	ng to	the	value	s of	size
	Car In In	f <i>size</i> f <i>size</i>	= 0 a = 0 a	and and a	couni	t = 1 t = 32	-→ 31 2,	,	ti ti	hen tr hen tr	ansfe ansfe	ers = ers =	<i>соц</i> 0	ınt		
	La li	f <i>size</i> f <i>size</i>	= 1 a = 1 a	and and a	couni	t = 1 t = 16	→ 15 },	5,	ti ti	hen tr hen tr	ansfe ansfe	ers = ers =	2x ( 0	count		
	The of the in for the code	comm nstruc ne mc ; if yo ocess	nand ition) ive. 1 u doi ior di	oper that The <i>I</i> n't su rectiv	rand the c /D op upply ve.	speci oproc eran this	fies a cesso d is a opera	an in: or she an op and,	struc ould otiona it del	tion (; execu al 3-b faults	21 bi ute to it cop to th	ts of spec proce e val	info cify I esso ue s	rmatio the des r ident specifie	n de stina tifica ed in	fine Ition Ition the

The output of this instruction on the LAD bus at ALTCH low during the command cycle (when SF is high) is as follows:

LAD #		
ID	coprocessor command	size 0 0 0
	13 MSBs 8	LSBs
	For more information, refer to Sectio <u>a Coprocessor</u> , on page 10-5.	n 10.3, <u>Formats of Commands Passed to</u>
Machine States	5 + [transfers –1] if the immediate d 6 + [transfers –1] if the immediate d	ata is long-word aligned ata is not long-word aligned
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>	
Example 1	This instruction moves thirty-two 32- tion is specified by the coprocessor A0 specifies the first memory addre	-bit values to coprocessor 2 (the destina instruction <i>wxyz</i> ) from memory (registe ss).
	CMOVMC -*A0,32,0,wxyz,2	
Example 2	This instruction moves eight 64-bit va nation is specified by the coprocesso B7 specifies the first memory addres	alues to the default coprocessor (the destion or instruction <i>wxyz</i> ) from memory (registents) ss).
	CMOVMC -*B7+,8,1,wxyz	
Example 3	This example demonstrates the TMS34082 using the typical C subro routine _COPROC_SUB requires tw quantities, all passed on the progr returned on the program stack. Assu and that the LOAD bit of the configur transferred before LSBs.	predecrement transfer of data to the utine calling convention. Assume that the vo 32-bit integer quantities and 3 double am stack.The result is a double value ume that field size 1 is 32 bits (FS1 = 32) ration register is set to 0 so the MSBs are
	_COPROC_SUB moves the values the CMOVMC opcode in predecrer values to the coprocessor. The com coprocessor registers (fpuop = 00110 (T = 1, size = 1) from the memory poi value of STK, to coprocessor register	from the stack to the coprocessor using ment mode. It first moves the 3 double mand MC_3_DOUB performs a move to 0) of 3 (count = 3) double precision values inted to (md = 10) by the predecremented ers starting at RA3 (rd = 00011).
	Next, the two 32-bit integer values a $MC_2$ _INT command. It performs a 00110) of 2 (count = 2) integer values to (md = 10) by the predecrementer starting at RB1 (rd = 10001).	are moved to the coprocessor using the move to coprocessor registers (fpuop = (T = 0, size = 0) from the memory pointed d value of STK to coprocessor registers

After manipulating the data in the coprocessor, the result is loaded from the coprocessor and placed onto the stack. This is performed by the CM\_1\_DOUB command of the CMOVCM opcode in predecrement mode. Predecrement mode is used to preserve the proper ordering of MSBs and LSBs on the stack. The stack pointer is kept pointing in the proper place by the 2 stack pointer adds. Interrupts are disabled when the stack pointer is not properly positioned. You may prefer to use other methods of ensuring program stack position integrity if interrupts are used for timing-critical functions.

The CM\_1\_DOUB command performs a move of coprocessor registers (fpuop = 00111) to TMS34020 memory (md = 10) of one (count = 1) double-precision value (T = 1, size = 1) from coprocessor register RB5 (rd = 10101) to the predecremented memory location pointed to by STK.



```
MOVE
       INT_2,*STK+,1
       INT_1,*STK+,1
MOVE
       DOUB 3 LO,*STK+,1
MOVE
       DOUB 3 HI,*STK+,1
MOVE
       DOUB_2_LO,*STK+,1
MOVE
       DOUB_2_HI,*STK+,1
MOVE
MOVE
       DOUB 1 LO,*STK+,1
       DOUB_1_HI,*STK+,1
MOVE
      _COPROC_SUB
CALLA
MOVE
       -*STK,A0,1
                       ;GET DOUBLE RESULT FROM STACK, HI
       -*STK,A1,1
MOVE
                      ;LO
.
COPROC SUB:
       SP,A0,A1
MMTM
CMOVCM -*STK, 3, 1, MC 3 DOUB
CMOVCM -*STK,2,0,MC_2_INT
                ;OTHER COPROCESSOR COMMANDS
.
.
.
DINT
ADDI
       40h,STK
                              ;MAKE ROOM FOR DOUBLE VALUE
CMOVMC -*STK, 1, 1, CM_1_DOUB
                              ; POINT STACK TO CORRECT TOP
ADDI
       40h,STK
EINT
MMFM
       SP,A0,A1
RETS
```

Syntax	CMOVMC *Rs+, Rd, size, command [, ID]															
Execution	$ID, command \rightarrow$ Coprocessor command															
	Repeat number of times specified by $Rd$ * $Rs \rightarrow$ Coprocessor $Rs + 32 \rightarrow Rs$															
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	1	1	1	R		R	d	
	8 LSBs of coprocessor command size 0 0 R Rs															
	copr	ocesso	or ID				13	MSBs	of cop	roces	sor co	mma	nd			

Description

This version of the CMOVMC instruction moves one or more 32-bit values (depending on the value in *Rd*) from memory to the specified coprocessor register. Before instruction execution, *Rs* contains the 32-bit memory address of a 32-bit value; after each transfer, the value in *Rs* is incremented by 32 bits to point to the next address. The *size* operand is a value of 0 or 1; it determines the size of the values that are transferred:

- □ If *size* = 0, then **32-bit** values are moved. In this case, *Rd* should specify the number of 32-bit values to move.
- ☐ If *size* = 1, then **64-bit values** are moved. In this case, *Rd* should specify **twice** the number of 64-bit values to move.

The value of *Rd (number of transfers)* is interpreted by the TMS34020 as follows:

- □ If  $Rd = 1 \rightarrow 31$ , then the number of 32-bit transfers = Rd
- If Rd = 0 then the number of 32-bit transfers = 32

The *command* operand specifies an instruction (21 bits of information define the instruction) that the coprocessor should execute to specify the destination for the move. The *ID* operand is an optional 3-bit coprocessor identification code; if you don't supply this operand, it defaults to the value specified with the .coproc directive.

The output of this instruction on the LAD bus at ALTCH low during the command cycle (when SF is high) is as follows:



For more information, refer to Section 10.3, <u>Formats of Commands Passed to</u> <u>a Coprocessor</u>, on page 10-5.

Machine States	5 + [register value –1] if the immediate data is long-word aligned 6 + [register value –1] if the immediate data is not long-word aligned
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>
Example 1	This instruction moves the number of 32-bit values specified by A2 to coprocessor 2 (the destination is specified by the coprocessor instruction $wxyz$ ) from memory (register A0 contains the first memory address).
	CMOVMC *A0+,A2,0,wxyz,2
Example 2	This instruction moves the number of 64-bit values specified by $B7 \div 2$ to the default coprocessor (the destination is specified by the coprocessor instruction <i>wxyz</i> ) from memory (register A3 contains the first memory address).
	CMOVMC *A3+, B7, 1, qrst
Example 3	This example moves 64-bit quantities (size = 1, T = 1) from the postincrem- ented memory block pointed to by A3 to TMS34082 coprocessor number 1 (ID = 1). Assume that the LOAD bit in the TMS34082 configuration register is set for MSBs transfer before LSBs transfer. Because register B4 specifies the number of 32-bit transfers to be performed, it should contain twice the number of 64-bit quantities to be transferred.
	The coprocessor command specifies a TMS34082 move to coprocessor regis- ters (fpuop = 00110) from TMS34020 memory (md = 10) of double quantities (T = 1, size = 1), with the starting destination of RB5 (rd = 10101).
	The memory location pointed to originally by A3 should contain the 32 MSBs of the first double number to be transferred. Memory location A3+10h should contain the 32 LSBs of the first double number. The remaining data should be in the MSBs before LSBs form.
	The two 32-bit halves of the first double number will be transferred to coprocessor one register RB5, the second double number to RB6, etc.
	MOVD_RB5 .set 00158Dh CMOVMC *A3+,B4,1,MOVD_RB5,1 ; count = 3 size = 1 : ID = 1
	0     0     1     5     8     D       0     0     0     0     0     0     0     0     0       ra     rb     rd     md     fpuop     T

Syntax	СМ	P R	s, Rd													
Execution	Set	Set status bits on the result of Rd – Rs														
Instruction Words	15 0	14   1	13 0	12 0	11 1	10 0	9 0	8	7	6 Rs	5	4 R	3	2 R	1 Id	0
Description	CM the ters or	CMP sets the status bits on the result of subtracting the contents of Rs from he contents of Rd. This is a nondestructive compare; the contents of the regisers are not affected. This instruction is often used in conjunction with the JAcc or JRcc conditional jump instructions.														from egis- JA <i>cc</i>
	Rs	and F	ld mu	st be	in th	e sai	me re	egist	er fil	е.						
Machine States	1															
Status Bits	N C Z V	<ul> <li>N 1 if the result is negative, 0 otherwise</li> <li>C 1 if a there is a borrow, 0 otherwise</li> <li>Z 1 if the result is 0, 0 otherwise</li> <li>V 1 if there is an overflow, 0 otherwise</li> </ul>														
Examples		Defe						A 64 -								

<u>Code</u>	<u>Before</u>		After	Jumps Taken
	A1	A0	NCZV	
CMP A1,A0	00000001h	00000001h	0010	UC,NN,NC,Z,NV,LS,GE,LE,HS
CMP A1,A0	00000001h	00000002h	0000	UC,NN,NC,NZ,NV,P,HI,GE,GT,HS
CMP A1,A0	0000001h	FFFFFFFh	1000	UC,N,NC,NZ,NV,P,HI,LT,LE,HS
CMP A1,A0	00000001h	80000000h	0001	UC,NN,NC,NZ,V,HI,LT,LE,HS
CMP A1,A0	FFFFFFFh	7FFFFFFh	1101	UC,N,C,NZ,V,LS,GE,GT,LO
CMP A1,A0	FFFFFFFh	8000000h	1100	UC,N,C,NZ,NV,LS,LT,LE,LO
CMP A1,A0	8000000h	7FFFFFFFh	1101	UC,N,C,NZ,V,LS,GE,GT,LO

Syntax	CMPI IW, Rd	[, <b>W</b> ]													
Execution	Set status bits	on the result of F	Rd – 16-bi	t immediate	e value										
Instruction Words	15     14     13       0     0     0	12 11 10 9 0 1 0 1 1s	8 7 1 1 0 complemen	6 5 1 0 t of <i>IW</i>	4 3 R	2 1 Rd	0								
<i>Description</i>	CMPI sets the s immediate valu in the syntax a nondestructive affected. This in ditional jump in Note that the as second instruct The assembler value is previou the assembler t	status bits on the e from the conte bove represents compare; the o istruction is often structions. sembler inserts ion word. uses the short f sly defined and is to use the short f	e result of nts of the c a 16-bit, contents c used in co the 1s con form of the s in the ran orm by foll	subtracting destination signed imm of the dest njunction w nplement o e CMPI ins ge –32,768 lowing the r	g a 16-bit, register. ( ination re vith the JA f the 16-b truction if to 32,761 register o	sign-exto The syml alue.) Th egister a cc or JRc hit value ir the imm 7. You car perand w	ended bol <i>IW</i> is is a re not c con- nto the ediate								
	CMPI IW, Rd, W The assembler truncates the upper bits and issues an appropriate warning message if the value is greater than 16 bits.														
Machine States	2														
Status Bits	<ul> <li>N 1 if the result</li> <li>C 1 if there is</li> <li>Z 1 if the result</li> <li>V 1 if there is</li> </ul>	ult is negative, 0 a borrow, 0 othe ult is 0, 0 otherwi an overflow, 0 o	otherwise erwise ise therwise												
Examples	Code CMPI 1,A0 CMPI 1,A0 CMPI 1,A0 CMPI 1,A0 CMPI 1,A0 CMPI -2,A0 CMPI -2,A0 CMPI -2,A0	Before           A0           00000002h           00000000h           FFFFFFFh           80000000h           FFFFFFFh           9000000h           FFFFFFFh           FFFFFFFh           FFFFFFFh           FFFFFFFFh           FFFFFFFFh           FFFFFFFFh           FFFFFFFFFh           FFFFFFFFh           FFFFFFFFh           FFFFFFFFh           FFFFFFFFh           FFFFFFFFFFFFFFFh	After         N C Z V         0       0       0         1       1       0       0         1       0       0       0         1       0       0       0       0         1       0       0       0       1         0       0       0       0       1         0       0       0       0       0         0       0       0       0       0         0       0       0       0       0         1       1       0       0	Jumps Tak UC,NN,NC UC,N,C,NZ UC,N,NC,N UC,NN,NC UC,NN,C,N UC,NN,NC UC,NN,NC UC,NN,NC	sen ,NZ,NV,LS, ,Z,NV,LS,LT ,IZ,NV,P,HI ,NZ,V,HI,L ,IZ,NV,P,LS ,NZ,NV,P,I ,Z,NV,LS,LT	HI,GE,GT, GE,LE,HS ,LE,LO ,LT,LE,HS T,LE,HS S,GE,GT,L LI,GE,GT,I GE,LE,HS ,LE,LO	HS O HS								

CMPI -1,A0 7FFFFFFh 1 1 0 1 UC,N,C,NZ,V,LS,GE,GT,LO

Syntax	CMPI IL, Rd [,	L]														
Execution	Set status bits of	on the result of	Rd – 32-bit	immediate	e value											
Instruction Words	15 14 13	12 11 10	987	65	4 3	2 1	0									
	0 0 0	0 1 0	1 1 0	1 1	R	Rd										
		1s con	plement of 16	SLSBs of IL												
		1s com	plement of 16	MSBs of IL												
Description	CMPI sets the s value from the c above represen compare; the co	tatus bits on the ontents of the d Its a 32-bit, sigr ontents of the d	result of sul estination re ned immedia estination r	otracting a egister. (Th ate value.) egister are	32-bit, sig ne <i>IL</i> symb This is a e not affeo	gned imm ool in the s nondestr cted.	ediate syntax uctive									
	Note that the as into the second of the value into	lote that the assembler inserts the 1s complement of the 16 LSBs of the value nto the second instruction word and inserts the 1s complement of the 16 MSBs of the value into the third instruction word.														
	The assembler form. You can fo operand with ar	The assembler uses this form of the CMPI instruction if it cannot use the short form. You can force the assembler to use the long form by following the register operand with an L:														
	CMPI IL, Rd,	CMPI IL, Rd, L														
	This instruction jump instructior	is often used in Is.	conjunctio	n with the .	JA <i>cc</i> or J	R <i>cc</i> cond	itional									
Machine States	2 if the immedia 3 if the immedia	ite data is long- ite data is not lo	word aligne	ed ligned												
Status Bits	<ul> <li>N 1 if the result</li> <li>C 1 if there is</li> <li>Z 1 if the result</li> <li>V 1 if there is</li> </ul>	It is negative, <i>C</i> a borrow, <i>0</i> oth It is 0, <i>0</i> otherw an overflow, 0	) otherwise erwise vise otherwise													
Examples																
	<u>Code</u>	Before	After NCZV	Jumps T	<u>aken</u>											
	CMPT 8000h. A0	00008001h	0000	UC.NN N		P.HI.GE G	THS									
	CMPI 8000h,A0	00008000h	0010	UC,NN.N	C,Z,NV.LS	GE,LE.H	s.,									
	CMPI 8000h,A0	00007FFFh	1100	UC,N.C.N	IZ,NV.LS.I	LT,LE.LO	-									
	CMPI 8000h.A0	FFFFFFFh	1000	UC,N,NC	,NZ,NV,P.I	HI,LT,LE,F	S									
	CMPI 8000h,A0	80007FFFh	0001	UC,NN,N	C,NZ,V,HI	,LT,LE,HS										
	CMPI OFFFF7FFFh,	10 00000000h	0100	UC,NN,C	,NZ,NV,P,I	LS,GE,GT	LO,									
	CMPI OFFFF7FFEh,	0 FFFF7FFFh	0000	UC,NN,N	C,NZ,NV,I	P,HI,GE,G	T,HS									

CMPI 0FFFF7FFEh, A0 FFFF7FFEh

CMPI OFFFF7FFEh, A0 FFFF7FFDh

CMPI 0FFFF7FFFh,A07FFF7FFFh

Syntax	CMPK consta	MPK constant, Rd											
Execution	Rd – constant	→ Status											
Instruction Words	15 14 13 0 0 1	12 11 1 1 0	09	8 7 cons	6 stant	5	4 R	3	2 R	1 d	0		
Description	CMPK subtract (Rd) and sets unsigned num	ts the 5-bit co status on f ber in the ra	onstant the res nge 1-	from th ult of th -32.	e conte ne subl	nts o tracti	fthe on. 7	desti Fhe c	natio const	n reg ant i	ister s an		
	The assemble constant=0 in the value 32 to try to compare	er automatica the opcode c o 0. Using th o 0 with a rec	ally bui orrespo is instri gister.	lds the onds to t uction, t	5 LSB: he valu he ass	s into ie 32; embl	the the a er iss	opco asser sues	ode. mbler an ei	Note conv ror if	that verts you		
	This instructio	n does not a	lter the	conten	ts of Re	d.							
Machine States	1												
Status Bits	<ul> <li>N 1 if the res</li> <li>C 1 if there i</li> <li>Z 1 if the res</li> <li>V 1 if there i</li> </ul>	sult is negati s a borrow, ( sult is 0, 0 ot s an overflov	ve, 0 of 0 other herwise v, 0 oth	herwise vise e erwise	9								
Examples	Code	Before A0	After N C Z	v	Jump	s Tak	en						
	cmpk 1,A0	0000002h	000	0	UC,NN	I,NC,	NZ,N	IV,P,H	II,GE,	GT,H	S		
	cmpk 2,A0	0000002h	001	0	UC,NN	1,NC,	Z,NV	,LS,G	E,LE	,HS			
	cmpk 32,A0	0000000h	1 1 0	D	UC,N,	C,NZ	,NV,L	S,LT,	LE,LC	)			
	cmpk 16,A0	FFFFFFFFh	100	0	UC,N,	NC,N	Z,NV	, P, HI,	LT,LE	,HS			

cmpk 1,A0 8000000h 0 0 0 1 UC,NN,NC,NZ,V,HI,LT,LE,HS

Syntax	CMPXY Rs, I	₽d			
Execution	Set status bits X half of R Y half of R	on the results d – X half of F d – Y half of F	s of Rs Rs		
Instruction Words	15 14 13	12 11 10	9 8 7	65	4 3 2 1 0
	1 1 1	0 0 1	0	Rs	R Rd
Description	CMPXY compa and sets the s nondestructive source and des no overflow de	ares the source status bits as a compare; the stination regis tection is prov	ce register to t if a subtraction e contents of ters are treate vided.	he destina on had bee the registe d as signee	tion register in XY mode en performed. This is a er are not affected. The d XY registers. Note that
	Rs and Rd mu	st be in the sa	ame register fi	le.	
Machine States	1				
Status Bits	<ul> <li>N 1 if source</li> <li>C Sign bit of</li> <li>Z 1 if source</li> <li>V Sign bit of</li> </ul>	X field = dest Y half of the r Y field = dest X half of the r	ination X field esult ination Y field esult	l, 0 otherwi I, 0 otherwi	se
Examples	Code	<u>Before</u> A1	A0	<u>After</u> N C Z V	<u>Jumps Taken</u>
	CMPXY A1,A0	00090009h	00010001h	0101	NN,C,NZ,V,LS,LT
	CMPXY A1,A0	00090009h	00090001h	0011	NN,NC,Z,V,LS,LT
	CMPXY A1,A0	00090009h	00010009h	1100	N,C,NZ,NV,LS,LT
	CMPXY A1,A0	00090009h	00090009h	1010	N,NC,Z,NV,LS,LT
	CMPXY A1,A0	00090009h	00000010h	0100	NN,C,NZ,NV,LS,GE
	CMPXY A1,A0	00090009h	00090010h	0010	NN,NC,Z,NV,LS,GE
	CMPXY A1,A0	00090009h	00100000h	0001	NN,NC,NZ,V,HI,LT
	CMPXY A1,A0	00090009h	00100009h	1000	N,NC,NZ,NV,HI,LT
	CMPXY A1,A0	00090009h	00100010h	0000	NN,NC,NZ,NV,HI,GE

Syntax (	CPW	Rs, Rd
----------	-----	--------

*Execution* Point code  $\rightarrow$  Rd

Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	1	1	1	0	0	1	1		F	<b>}s</b>		R		R	d	

Description CPW compares a point represented by an XY value in the source register to the window limits in the WSTART and WEND registers and places a 4-bit *outcode* in bits 5 through 8 of the destination register. The remaining 28 bits of the destination register are set to 0. The contents of the source register are treated as an XY address that consists of 16-bit signed X and Y values. WSTART and WEND are also treated as signed XY-format registers. WSTART and WEND can contain signed values. The location of the point with respect to the window is encoded as shown below; the outcode is loaded into the destination register.



The following list describes the contents of the destination register after CPW execution:

## **Bit Position: Contents:**

- 0-4 Os
- 5 1 if X half of WSTART > X half of Rs, 0 otherwise
- 6 1 if X half of Rs > X half of WEND, 0 otherwise
- 7 1 if Y half of WSTART > Y half of Rs, 0 otherwise
- 8 1 if Y half of Rs > Y half of WEND, 0 otherwise
- 9-31 Os

This instruction can also be used to trivially reject lines that do not intersect with a window. If the CPW codes for the 2 points defining the line are ANDed together and the result is nonzero, then the line must lie completely outside the window (and does not intersect it). A *0* result indicates that the line *may* intersect the window, and a more rigorous test must be applied. For more information, refer to Section 12.7, Window Checking, on page 12-19.

Rs and Rd must be in the same register file.

Implied Operands	Register	Name	Format	Description			
	B5	WSTART	XY	Window start. Defines inclusive starting corner of window (lesser value corner).			
	B6	WEND	XY	Window end. Defines inclusive ending con ner of window (greater value corner).	or-		
Machine States	1						
Status Bits	<ul> <li>N Unaffe</li> <li>C Unaffe</li> <li>Z Unaffe</li> <li>V 1 if poi</li> </ul>	cted cted cted nt lies outsid	de window,	<i>0</i> otherwise			
Examples	You must so instruction. ing a block	elect approp In this exam of 36 pixels	riate implied ple, the imp	d operand values before executing the CI olied operands are set up as follows, spec	⊇W ;ify-		
	WSTART = WEND = A CPW A1,A0	= 5,5 .,A					
	Before		Afte	er			
	A1	NCZY	√ <mark>A0</mark>	NCZV			
	00040004h	x x x i	0000	000A0h x x x 1			
	00040005h	x x x i	0000	00080h x x x 1			
	0004000Ah	x x x (	0000	00080h x x x 1			
	0004000Bh	x x x *	1 0000	000C0h x x x 1			
	00050004h	xxx	1 0000	00020h x x x 1			
	00050005h	x x x (	0000	00000h x x x 0			
	0005000Ah	x x x (	0000	00000h x x x 0			
	0005000Bh	x x x (	0000	00040h x x x <b>1</b>			
	000A0004h	xxx	0000	00020h x x x 1			
	000A0005h	ххх	1 0000	00000h x x x 0			
	000A000Ah	xxx	1 0000	00000h x x x 0			
	000A000Bh	x x x i	0000	00040h x x x 1			

00000120h

00000100h

00000100h

00000140h

000B0004h

000B0005h

000B000Ah

000B000Bh

X X X 0

 $x \times x 0$ 

x x x 0

 $x \times x 0$ 

 $X \times X 1$ 

 $x \times x 1$ 

 $X \times X 1$ 

 $x \times x 1$ 

Syntax	CVDXYL Rd									
Execution	(Y half of Rd $\times$ DPTCH) + (X half of F	Rd × PSIZE) + (A4 or B4) → Rd								
Instruction Words	15 14 13 12 11 10 9 8	7 6 5 4 3 2 1 0								
Description	CVDXYL converts the XY value contained in the destination register to a linear value, by using the destination pitch, and writes the result back to the destination register.									
	If Rd is an A file register, then A4 is us then B4 is used as the offset. CONVI the multiply of the Y half by the pitch. P tiply by the X half (done by shift).	ed as the offset; if Rd is a B file register, DP (based on DPTCH) is used to effect SIZE provides the pixel size used to mul-								
	Use the SETCDP instruction (page 13 mation, refer to Section 12.12, <u>Conver</u> on page 12-47.	-227) to set up CONVDP. For more infor- ting an XY Address to a Linear Address,								
Implied Operands	Address Name Descript	tion and Elements (Bits)								
	A4 or B4 OFFSET linear s	creen origin (0,0)								
	C0000140 CONVDP XY-to-li	near conversion (destination)								
	C0000150 PSIZE Pixel siz	ze (1,2,4,8,16,32)								
Machine States	pitch is a power of 2: 2 2 powers of 2: 3 arbitrary 14									
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>									
Examples	The examples assume the following of	definitions:								
	DADDR .set B2 ; De DPTCH .set B3 ; De OFFSET .set B4 ; X3 TEMP .set B14 ; Te	estination address register estination pitch register Y offset register emporary register								

Example 1 Screen dimensions of 640 by 480 by 4 bits-per-pixel; screen pitch of 4096 (the smallest power of 2 greater than  $640 \times 4$ ). ; the following code will typically be done once : at initialization time setf 16,0,0 ; set field size 0 to 16 4, TEMP movk exgps TEMP ; set the pixel size to 4 movi 01000h,DPTCH ; set up display pitch 0100h,OFFSET movi ; set up the screen offset setcdp ; set CONVDP to 0013h (LMO of DPTCH) mwait ; set up XY address in DADDR and convert to linear 00010001h,DADDR movi cvdxyl DADDR ; This sets DADDR to 1104h Example 2 Screen dimensions of 640 by 480 by 8 bits-per-pixel; screen pitch of 5120 (640  $\times$  8) which is a sum of 2 powers of 2. ; the following code will typically be done once : at initialization time ; set field size 0 to 16 setf 16,0,0 movk 8,TEMP ; pixel size of 8 exgps 01400h, DPTCH ; set up display pitch movi clr OFFSET ; set up the screen offset ; set CONVDP to 1513h setcdp ; (2 LMOs of DPTCH) mwait ; set up XY address in DADDR and convert to linear movi 00010001h, DADDR ; This sets DADDR to 1408h cvdxyl DADDR Example 3 Perform an XY to linear conversion of a nonscreen bitmap with an arbitrary pitch of 224. ; the following code will typically be done once at ; initialization time setf 16,0,0 ; set field size 0 to 16 movk 16, TEMP ; ; pixel size of 16 exqps ; set up the value of DPTCH, CONVDP, have user variables ; for offset and address movi 0E0h,DPTCH ; set CONVDP to 0000h (which setcdp ; indicates arbitrary pitch) mwait movi 0FF300000h,A4 ; offset points to an area ; of DRAM movi 00010001h,A7 ; this sets A7 to FF3000F0h cvdxyl A7

Syntax	CVMXYL Rd									
Execution	(Y half of Rd $\times$ MPTCH) +	(X half of Rd) $\rightarrow$ Rd								
Instruction Words	15     14     13     12     11     1       0     0     0     0     1     1	0 9 8 7 6 5 4 3 2 1 0 D 1 0 0 1 1 R Rd								
Description	CVMXYL converts the XY value, by using the mask pi ister.	value contained in the destination register to a linear tch, and writes the result back to the destination reg-								
	Note that <i>no</i> offset is add MPTCH) is used to effect	ed when you use CVMXYL. CONVMP (based on the multiply of the Y half by the pitch.								
	Use the SETCMP instruction (page 13-228) to set up CONVMP. For more mation, refer to Section 12.12, <u>Converting an XY Address to a Linear Add</u> on page 12-47.									
Implied Operands	Address Name	Description and Elements (Bits)								
	C0000150 PSIZE	Pixel size (1,2,4,8,16,32)								
	C0000180 CONVMP	XY-to-linear conversion (destination)								
Machine States	pitch is a power of 2: 2 powers of 2: arbitrary	2 3 14								
Status Bits	<ul><li>N Unaffected</li><li>C Unaffected</li><li>Z Unaffected</li><li>V Unaffected</li></ul>									
Examples	The examples assume the	oflowing definitions:								
	MADDR .set B10 MPTCH .set B11 TEMP .set B14	; Mask address register ; Mask pitch register ; Temporary register								
Example 1	Screen dimensions of 640 smallest power of 2 greate	by 480 by 4 bits-per-pixel; screen pitch of 4096 (the er than 640 × 4)								
	<pre>; the following code will typically be done once ; at initialization time setf 16,0,0 ; set field size 0 to 16 movi 01000h,MPTCH ; set up mask pitch setcmp ; set CONVMP to 0013h (LMO ; of MPTCH)</pre>									
	; set up XY address in movi 0001000 cvmxyl MADDR	MADDR and convert to linear 1h,MADDR ; This sets MADDR to 1001h								

Example 2	Screen dimensions of 640 by 480 by 8 bits-per-pixel; screen pitch of 5120 (640 $\times$ 8) which is a sum of 2 powers of 2.								
	<pre>; the following code will typically be done once ; at initialization time setf 16,0,0 ; set field size 0 to 16 movi 01400h,MPTCH ; set up mask pitch setcmp ; set CONVMP to 1513h ; (2 LMOs of MPTCH) mwait ; set up XY address in MADDR and convert to linear movi 00010001h,MADDR cvmxyl MADDR ; This sets MADDR to 1401h</pre>								
Example 3	Perform an XY to linear conversion of a nonscreen bitmap with an arbitrary pitch of 224.								
	<pre>; the following code will typically be done once ; at initialization time setf 16,0,0 ; set field size 0 to 16 movi 0E0h,MPTCH ; set up mask pitch setcmp ; set CONVMP to 0000h (which ; indicates arbitrary pitch) mwait ; set up XY address in MADDR and convert to linear movi 00010001h,MADDR cvmxyl MADDR ; This sets MADDR to E1h</pre>								

Syntax	CVSXYL Rs, Re	d									
Execution	(Y half of Rd × Si	PTCH) + (X hali	of Rd × PSIZE)	+ Rs →	Rd						
Instruction Words	15 14 13 12 1 1 1 0	11 10 9 1 0 1	8 7 6 Rs	54 R	3	2 F	1 Rd	0			
Description	CVSXYL converts value, by using th register. The sou process. CONVS half by the pitch. (done by shift).	CVSXYL converts the XY value contained in the destination register to a linear value, by using the source pitch, and writes the result back to the destination register. The source register contains the offset used in the conversion process. CONVSP (based on SPTCH) is used to effect the multiply of the Y half by the pitch. PSIZE provides the pixel size used to multiply by the X half (done by shift).									
	Use the SETCSP instruction (page 13-229) to set up CONVSP. For more infor- mation, refer to Section 12.12, <u>Converting an XY Address to a Linear Address</u> , on page 12-47.										
	This instruction allows the programmer to have independent source and desti- nation offsets. The source offset can be any register, and the destination offset is B4. As an example, assume B14 is the source offset. To synthesize PIXBLT XY,XY with independent offsets use CVSXYL B14,B0 to convert the source pointer to linear and follow with a PIXBLT L,XY.										
Implied Operands	Address	Name	Description and	Elements	s (Bit	ts)					
	C0000130	CONVSP	XY-to-linear conversion (source)								
	C0000150	PSIZE	Pixel size (1,2,4	,8,16,32)							
Machine States	pitch is a powe 2 powe arbitra	er of 2: 2 ers of 2: 3 ry 14									
Status Bits	<ul><li>N Unaffected</li><li>C Unaffected</li><li>Z Unaffected</li><li>V Unaffected</li></ul>										
Examples	CVSXYL A0, A1										
	Before A0	= 00000100									

## CVXYL Convert XY Address to Linear Address

Syntax	CVXYL Rs, Rd										
Execution	(Y half of Rs × DPTC	CH) + (X half o	f Rs × PSIZE) ·	+ offset → F	łd						
Instruction Words	15 14 13 12 1 1 1 0	11 10 9 1 0 0	8 7 6 Rs	543 R	2 1 Rd	_0					
Description	CVXXI converts an	XV address to	a linear addres	I		I					
	The source register contains an XY address. The signed X value occup the 16 LSBs of the register, and the signed Y value occupies the 16 MS The X value must be positive.										
	The XY address in the destination <u>Converting an X</u>	is converted ir n register. For Y Address to a	nto a 32-bit line more informati a Linear Addres	ar address, v ion, refer to <u>ss,</u> on page v	which is st Section 12 12-47.	ored 2.12,					
	The offset value (see execution) is in the OFFSET register. The CONVDP value is used to determine the shift amount for the Y value, while the PSIZE register determines the X shift amount.										
	Use the SETCDP ins	truction (page	13-227) to set (	up CONVDF	) • *						
	Rs and Rd must be i	n the same re	gister file.								
Implied Operands	Address Name	De	scription and E	lements (Bit	s)						
	B3 DPTC	H (linear) D	Destination pitch								
	B4 OFFS	ET (linear) S	creen origin (loc	ation 0,0)							
	C0000140h CONV	DP X	Y-to-linear conve	ersion (destin	ation pitch)						
	C0000150h PSIZE	Р	ixel size (1,2,4,8	,16,32)							
Machine States	pitch is a power o 2 powers arbitrary	f 2: 3 of 2: 4 14									
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>										

Examples	Code	<u>Before</u>	<u>After</u>			
		A0	OFFSET	PSIZE	CONVDP	A1
	CVXYL A0,A1	00400030h	00000000h	0010h	0014h	00020300h
	CVXYL A0,A1	00400030h	00000000h	0008h	0014h	00020180h
	CVXYL A0,A1	00400030h	00000000h	0004h	0014h	00020000h
	CVXYL A0,A1	00400030h	00008000h	0004h	0014h	00028000h
	CVXYL A0,A1	00400030h	0F000000h	0004h	0014h	0F020000h
	CVXYL A0,A1	00400030h	00000000h	0002h	0014h	00020060h
	CVXYL A0,A1	00400030h	00000000h	0001h	0014h	00020030h
	CVXYL A0,A1	00400030h	00000000h	0001h	0013h	00040030h
	CVXYL A0,A1	00400030h	00000000h	0001h	0015h	00010000h
		13h correspond	ds to DPTCH =	00001000	h	

\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*\*

CONVDP = 0014h corresponds to DPTCH = 00000800h CONVDP = 0015h corresponds to DPTCH = 00000400h

.....

## **DEC** Decrement Register

Syntax	DEC Rd								
Execution	$Rd - 1 \rightarrow Rd$								
Instruction Words	15 14 1 0 0	3 12 11 0 1 0	10 9 8 1 0 0	7 6 0 0	5 4 1 R	32	1 Rd	0	
Description	DEC subtra result in the SUBK 1, Rd You can use precision a	acts 1 from the destination re the DEC instr ithmetic.	e contents of the contents of the contents of the contents of the content of the	he destina struction i SUBB ina	ation regi is an alter struction	ster and mate mo to perforr	stores emonie m multi	the c for iple-	
Machine States	1								
Status Bits	<ul> <li>N 1 if the</li> <li>C 1 if then</li> <li>Z 1 if then</li> <li>V 1 if then</li> </ul>	result is nega re is a borrow, result is 0, <i>0</i> c re is an overflo	tive, 0 otherwi 0 otherwise otherwise ow, 0 otherwis	e					
Examples	CodeDECA1DECA1DECA1DECA1	Before A1 00000010h 00000001h 00000000h FFFFFFFh 8000000h	<u>After</u> A1 00000000F 00000000 FFFFFFF FFFFFFF 7FFFFFFF	N 0 h 0 Fh 1 Eh 1 Fh 0	<b>C Z V</b> 0 0 0 0 1 0 1 0 0 0 0 0 0 0 1				

Syntax	DINT													
Execution	$0 \rightarrow IE$													
Instruction Words	15 14 0 0	+ 13   0	12 1 0	11 10 0 0	9 1	8 1	7 0	6 1	5 1	4 0	3 0	2 0	1 0	0
Description	DINT di 0. All int rupt en status r The Ell	sables i errupts able ma egister IT instr	interru excep ask in is una uction	opts by c ot reset, the INT ffected. enable	learir NMI, ENB s inte	ng the bus f regi rrupt	e glot ault, s ster i s.	oal in and I s igr	terru ILLO norec	pt er P are I. Th	nable e disa le rei	bit (I abled main	E[[S ; the der c	[]) to inter- of the
Machine States	3													
Status Bits	<ul> <li>N Unit</li> <li>C Unit</li> <li>Z Unit</li> <li>V Unit</li> <li>IE 0</li> </ul>	affected affected affected affected	1 1 1											
Examples	Code DINT DINT	Before ST 000000 002000	2 010h 010h	<u>Afte</u> ST 000 000	e <b>r</b> 00010 00010	lh Ih								

Syntax	DIVS Rs, F	Rd											
Execution	lf Rd is an e If Rd is an o	ven-nur Rd : Ro Rs dd-num Rd Rs	mbered $\frac{d+1}{3} \rightarrow$ bered r Rd	regist Rd, egiste	er rema r	inde	r → F	Rd +	1				
Instruction Words	15 14 13 0 1 0	3 <u>12</u> 1 1	11 10 1 0	9 0	8	7 F	6 Is	5	4 R	3	2 R	1 Id	0
Description	DIVS perform the 32-bit sig dend or the whether Rd example, A8	ms a sig ned div most si is an od or B2)	gned 32 risor. Th ignificar Id regist :	-bit o e dest nt half er (fo	f 64-b inatio of a r exan	it div n reg 64-b nple,	ride. <sup>-</sup> gister it sig A1 o	The s cont ned or B3	sourc ains divid ) or a	ce reg a 32- end, n eve	gister bit si depe en re	r con gned endin giste	tains divi- g on r (for
	Rd Even	DIVS performs a signed divide of the 64-bit operand contained the 2 consecutive registers, starting at the specified destinat register, by the 32-bit contents of the source register. The specified even-numbered destination register, Rd, contains the MSBs of the dividend. The next consecutive register (which odd-numbered) contains the 32 LSBs of the dividend. The qu tient is stored in the destination register, and the remainde stored in the following register (Rd+1). The remainder is alwa the same sign as the dividend (in Rd:Rd+1). Avoid using A14 B14 as the destination register, since this overwrites the SP; assembler issues a warning in this case								ed in ation peci- e 32 ch is quo- der is ways 14 or 2; the			
	Rd Odd	DIVS performs a signed divide of the 32-bit operand contained in the destination register by the 32-bit value in the source register. The quotient is stored in the destination register; the remainder is not returned.											
	Rs and Rd n	nust be	in the s	ame	registe	ər file	э.						
Machine States	Rd Odd:	39 (no 41 (if 7 (if R	ormal ca result = Is = 0)	se) 8000	0000h	1)							
	Rd Even:	40 (no 41 (if 7 (if R	ormal ca result = ls = 0)	se) 8000	0000h	1)							

	IN	Ult Bs=0 or
		Rd is even and Rd $\geq$ Rs. or
		Quotient is nonnegative.
		1 if
		Result = 80000000h <b>or</b>
		Quotient is negative.
	С	Unaffected
	Ζ	0 if
		Rs = 0, <b>or</b>
		Rd is even and Rd ≥ Rs, <b>or</b>
		Result = 8000000h, <b>or</b>
		Quotient ≠ 0.
		1 if
		Quotient = 0.
	V	<b>1</b> if quotient overflows (cannot be represented by 32 bits), <b>0</b> otherwise. The following conditions cause an overflow and set the overflow flag: Divisor (Rs) is 0.
		Quotient cannot be contained within 32 bits.
Example 1	Thi reg	s example divides the contents of registers A0 and A1 by the contents of ister A2, stores the result in register A0, and stores the remainder in A1.

DIVS A2,A0

<u>Before</u>			<u>After</u>			
A0	A1	A2	A0	A1	A2	NCZV
12345678h	87654321h	87654321h	D95BC60Ah	5CA1DD7h	87654321h	1 x 0 0
EDCBA987h	789ABCDFh	87654321h	26A439F6h	EA35E229h	87654321h	0 x 0 0
EDCBA987h	789ABCDFh	789ABCDFh	D95BC60Ah	EA35E229h	789ABCDFh	1 x 0 0
12345678h	87654321h	789ABCDFh	26A439F6h	15CA1DD7h	789ABCDFh	0 x 0 0
12345678h	87654321h	00000000h	12345678h	87654321h	00000000h	0 x 0 1
00000000h	00000000h	00000000h	0000000h	00000000h	00000000h	0 x 0 1
00000000h	00000000h	87654321h	00000000h	00000000h	87654321h	0 x 1 0
87654321h	00000000h	87654321h	87654321h	00000000h	87654321h	0 x 0 1

Example 2

This example divides the contents of register A1 by the contents of of register A2 and stores the result in register A1. Note that the contents of register A2 are not affected by instruction execution.

DIVS A2,A1

<b>Before</b>			<u>After</u>								
A0	A1	A2	A0	A1	A2	NCZV					
00000000h	87654321h	12345678h	00000000h	FFFFFFAh	12345678h	1 x 0 0					
00000000h	87654321h	0EDCBA988h	00000000h	00000006h	EDCBA988h	0 x 0 0					
00000000h	789ABCDFh	0EDCBA988h	00000000h	FFFFFFFAh	EDCBA988h	1 x 0 0					
00000000h	789ABCDFh	12345678h	00000000h	00000006h	12345678h	0 x 0 0					
00000000h	87654321h	00000000h	00000000h	87654321h	00000000h	0 x 0 1					
00000000h	00000000h	00000000h	00000000h	00000000h	00000000h	0 x 0 1					
Syntax	DIVU Rs, F	Rd									
-------------------	-------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	---------------------------------------------------------------------------------------------------------------------------------	---------------------------------------------------------------------------------------------	----------------------------------------------------------------------------------	--------------------------------------------------------------------------------------------------	-------------------------------------------------------------------	--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	-----------------------------------------------------------	---------------------------------------------------------------	-----------------------------------------------------------	-----------------------------------------------------------
Execution	If Rd is an e	ven-numbered $\frac{\text{Rd}: \text{Rd} + 1}{\text{Rs}}$ dd-numbered $\frac{\text{Rd}}{\text{Rs}} \rightarrow \text{Rd}$	ed regisi → Rd, d registe	er rema er	inder →	Rd +	1				
Instruction Words	15 14 13	3 12 11	<u>10 9</u>	8	76	5	4	3	2	1	0
	0 1 0	D   1   1	0 1		Rs		R		R	d	
Description	<ul> <li>DIVU performs an unsigned 32-bit or 64-bit divide. The source register contains the 32-bit divisor. The destination register contains a 32-bit dividem or the most significant half of a 64-bit dividend, depending on whether Rd i an odd register (for example, A1 or B3) or an even register (for example, A or B2):</li> <li>Rd Even DIVU performs an unsigned divide of the 64-bit operantice.</li> </ul>								dend Rd is e, A8		
		tion registe specified ev 32 MSBs of odd-numbe tient is stor stored in the the destinat	r, by the ven-nun the divi red) col ed in th followi tion regi a warni	e 32-bi nberec dend. ntains e desi ng regi ster, si ng in t	it conten destina The next the 32 L tination r ister (Rd ince this his case	ts of t tion re conse SBs c egiste +1). A overw	the s egistrecution of the er, ar void vrites	ource er, Revere divice nd the using the S	e reg d, coi gister lend. e ren gA14 SP; th	ister. ntain r (wh The nainc or B ne as	The s the ich is quo- ler is 14 as sem-
	Rd Odd	DIVU perfe contained i source regi	orms a n the de ster. The	n uns estinat equoti	igned d tion registient is sto	ivide ster by pred in	of the official sectors and the official sectors are as the official sec	he 3 32-li destii	2-bit bit va natio	ope lue i n reg	rand n the ister;

Rs and Rd must be in the same register file.

the remainder is not returned.

Ma	chine	e States

Rd Odd:	37 (normal case)
· ·	7 (if Rs = 0)
Rd Even:	37 (normal case)
	5 (if $Rs = 0$ or $Rs \le Rd$ )

Status Bits	N C Z V	Unaffected Unaffected <b>0</b> if Rs = 0, or $Rd$ is even and $Rd \ge Rs$ , or Quotient $\ne 0$ . <b>1</b> if Quotient = 0. <b>1</b> if quotient overflows (cannot be represented by 32 bits), <b>0</b> otherwise. The following conditions cause an overflow and set the overflow flag:
Example 1	Thi reg in /	Quotient cannot be contained within 32 bits. s instruction divides the contents of registers A0 and A1 by the contents of ister A2, stores the unsigned result in register A0, and stores the remainder A1. Note that the contents of register A2 are not affected by instruction
	exe	ecution.

DIVU A2,A0

<u>Before</u>			<u>After</u>			
A0	A1	A2	A0	A1	A2	NCZ V
12345678h	87654321h	789ABCDFh	26A439F6h	15CA1DD7h	789ABCDFh	x x 0 0
12345678h	87654321h	00000000h	12345678h	87654321h	00000000h	x x 0 1
0000000h	00000000h	00000000h	00000000h	00000000h	00000000h	x x 0 1
0000000h	00000000h	87654321h	00000000h	00000000h	87654321h	x x 1 0
87654321h	00000000h	87654321h	87654321h	00000000h	87654321h	x x 0 1

**Example 2**This instruction divides the contents of register A1 by the contents of register<br/>A2 and stores the unsigned result in register A1. Note that the contents of regis-<br/>ter A2 are not affected by instruction execution.

DIVU A2,A1

<u>Before</u>			<u>After</u>			
A0	A1	A2	A0	A1	A2	NCZ V
0000000h	789ABCDFh	12345678h	00000000h	00000006h	12345678h	x x 0 0
0000000h	12345678h	00000000h	00000000h	12345678h	00000000h	x x 0 1
0000000h	00000000h	00000000h	00000000h	00000000h	00000000h	x x 0 1
0000000h	00000000h	87654321h	00000000h	00000000h	87654321h	x x 1 0
0000000h	87654321h	87654321h	00000000h	00000001h	87654321h	x x 0 0

Syntax	DRAV Rs, R	d						
Execution	COLOR1 pixe X half of Rs + Y half of Rs +	els → *Rd X of half Rd - Y of half Rd -	$\rightarrow$ X half of $\rightarrow$ Y half of	f Rd F Rd				
Instruction Words	15 14 13 1 1 1	12 11 10 1 0 1	9 8 1	7 6 5 Rs	5 4   R	32	1 Rd	0
Description	DRAV writes t by the XY ad address in the register: the X added to the Y does not prop	he pixel value Idress in the of e destination ro ( half of Rs is Y half of Rd. A bagate into the	in the COI destinatior egister is i added to t ny carry ou upper (Y)	LOR1 register register. For ncremented he X half of ut from the lo half.	er to the loollowing by the va Rd, and to wer (X) l	ocatior the wr alue in the Y h half of t	n point ite, the the sc alf of the reg	ed to e XY ource Rs is gister
	Use the SETC	DP instruction	n (page 13	-228) to set (	up CONV	/DP.		
	COLOR1 bits data used fro 32-bit pattern: file.	0—31 are outp m COLOR1 is s can be imple	out on data that whic mented. F	bus lines 0– h aligns to t Is and Rd m	–31, resp he destir ust be in	ectivel nation I the sa	y. The ocatio me reç	pixel n, so gister
Implied Operands	Register	Name	Format	Descri	ption			
Implied Operands	Register B3	Name DPTCH	<b>Format</b> Linear	<b>Descri</b> Destir	<b>ption</b> nation pito	:h		
Implied Operands	Register B3 B4	Name DPTCH OFFSET	Format Linear Linear	Descri Destir Scree	<b>ption</b> nation pito en origin (l	h ocation	0,0)	
Implied Operands	Register B3 B4 B5	Name DPTCH OFFSET WSTART	Format Linear Linear XY	Descri Destir Scree Windo	<b>ption</b> nation pito en origin (l ow starting	h ocation g corne	0,0) r	
Implied Operands	RegisterB3B4B5B6	Name DPTCH OFFSET WSTART WEND	Format Linear Linear XY XY	Descri Destir Scree Windo Windo	ption nation pitc en origin (I ow starting ow ending	ch ocation g corner	0,0) r	
Implied Operands	RegisterB3B4B5B6B9	Name DPTCH OFFSET WSTART WEND COLOR1	Format Linear Linear XY XY Pixel	Descri Destin Scree Windo Windo Pixel	ption nation pitc en origin (l ow starting ow ending color	ch ocation g corne g corner	0,0) r	
Implied Operands	RegisterB3B4B5B6B9	Name DPTCH OFFSET WSTART WEND COLOR1	Format Linear XY XY Pixel	Descri Destir Scree Windo Windo Pixel	ption nation pitc en origin (I ow starting ow ending color	ch ocation g corne g corner	0,0) r	
Implied Operands	RegisterB3B4B5B6B9Address	Name DPTCH OFFSET WSTART WEND COLOR1 Name	Format Linear XY XY Pixel Descript	Descri Destin Scree Windo Windo Pixel	ption nation pitc en origin (l ow starting ow ending color nents (Bit	contion g corner corner ts)	0,0) r	
Implied Operands	Register           B3           B4           B5           B6           B9           Address           C00000B0h	Name DPTCH OFFSET WSTART WEND COLOR1 Name CONTROL	Format Linear XY XY Pixel Descript PPOP F W Y T T	Descri Destin Scree Windo Windo Pixel Pixel-process Window-chec Transparency Selects 1 of 3	ption nation pitc on origin (I ow starting ow ending color nents (Bit ing operat king operat operation transpare	corner corner corner ts) tions (22 ation	0,0) r 2 optior ions	  ns)
Implied Operands	Register           B3           B4           B5           B6           B9           Address           C00000B0h           C0000140h	Name DPTCH OFFSET WSTART WEND COLOR1 Name CONTROL	Format Linear XY XY Pixel Descript PPOP F W T TM S XY-to-lir	Descri Destin Scree Windo Windo Pixel Pixel-process Window-chec Transparency Selects 1 of 3 mear conversion	ption nation pito en origin (I ow starting ow ending color nents (Bit ing operat king operat operation transpare on (destin	ts) tions (22 ation pinary option	0,0) r 2 optior ions tch)	
Implied Operands	Register           B3           B4           B5           B6           B9           Address           C00000B0h           C0000140h           C0000150h	Name DPTCH OFFSET WSTART WEND COLOR1 COLOR1 Name CONTROL	Format Linear XY XY Pixel Descript PPOP F W Y T TM S XY-to-lin Pixel siz	Descri Destin Scree Windo Windo Pixel Dixel-process Window-check Transparency Selects 1 of 3 mear conversion te (1,2,4,8,16)	ption nation pitc on origin (I ow starting ow ending color nents (Bit ing operat king operat operation transpare on (destin ,32)	ts) tions (22 ation ation pi	0,0) r 2 optior ions tch)	ns)
Implied Operands	Register           B3           B4           B5           B6           B9           Address           C00000B0h           C0000140h           C0000150h           C0000160h	Name DPTCH OFFSET WSTART WEND COLOR1 COLOR1 CONTROL CONTROL PSIZE PMASK (32 bits)	Format Linear Linear XY XY Pixel Descript PPOP F W N T 5 TM 5 XY-to-lir Pixel siz Plane m	Descri Destin Scree Windo Windo Pixel Pixel-processi Window-check Transparency Selects 1 of 3 near conversion te (1,2,4,8,16) mask — pixel f	ption nation pito on origin (l ow starting ow ending color nents (Bit ing operat king operat operation transpare on (destin ,32) format	conner g corner corner ts) tions (22 ation ation pi	0,0) r 2 optior ions tch)	ns)

Due to the pipelining of memory writes, the *last* I/O register that you write to may not, in some cases, contain the desired value by the time the DRAV instruction begins executing. To ensure that this register contains the correct value for execution, you may want to follow the write to that location with an MWAIT (page 13-178) instruction. Refer to Section 4.5.6 on page 4-13 for a description of the potential latency of writes to I/O registers.

Pixel Processing	Set PPOP[CON is applied to the default pixel-pro refer to Section	NTROLI to select a pixel-p pixel as it is moved to th pocessing operation is <i>rep</i> 12.8, <u>Pixel Processing</u> , o	processing operation. This operation he destination location. At reset, the lace (S $\rightarrow$ D). For more information, on page 12-27.
Window Checking	Select a window active window-o ters define the 2 X and Y values i refer to Section	w-checking mode by setti checking mode (W = 1, 2, c XY starting and ending co n both WSTART and WEN 12.7, <u>Window Checking</u> ,	ng W[CONTROL]]. If you select an or 3), the WSTART and WEND regis- orners of a rectangular window. The ID can signed. For more information, on page 12-19.
Transparency	You can enable Select 1 of 3 tran mation, refer to	transparency for this instrunsparency options by setti Section 12.9, <u>Transparer</u>	uction by setting T[[CONTROL]] to 1. ing TM[[CONTROL]]. For more infor- ncy, on page 12-36.
Plane Masking	The plane mask Section 12.10, J	k is enabled for this instru <u>Plane Masking</u> , on page 1	ction. For more information, refer to 12-39.
Shift Register Transfers	When this instruand write opera	uction is executed and CS tions become SRT reads	T bit is set, the normal memory read and writes.
Machine States	Refer to Section	on 15.1 on page 15-2.	
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V 1 if a windownot used.</li> </ul>	w violation occurs, 0 other	wise; unaffected if window clipping is
Examples	These DRAV ex	camples use the following	implied operand setup.
	Register File B DPTCH) OFFSET WSTART WEND COLOR1	= 200h = 00010000h = 00100000h = 003C0040h = FFFFFFFh	I/O Registers CONVDP = 0016h
	Assume that me tion:	emory contains the follow	ing values before instruction execu-
	<b>Address</b> 00018040h	<b>Data</b> 8888h	
	Note that the in	itial XY address in A0 is e	equivalent to linear address 18040h.

<u>Code</u>	<u>Before</u>					Afte	ST.		
	÷	A0	A1	PSIZE	PPOP	W	PMASK	A0	@18040h
DRAV	A1,A0	00400040h	00100010h	0001h	00000	00	0000h	00500050h	8889h
DRAV	A1,A0	00400020h	00100010h	0002h	00000	00	0000h	00500030h	888Bh
DRAV	A1,A0	00400010h	00100010h	0004h	00000	00	0000h	00500020h	888Fh
DRAV	A1,A0	00400008h	00100010h	0008h	00000	00	0000h	00500018h	88FFh
DRAV	A1,A0	00400004h	00100010h	0010h	00000	00	0000h	00500014h	FFFFh
DRAV	A1,A0	00400004h	0000FFFFh	0010h	01010	00	0000h	00400003h	0000h
DRAV	A1,A0	00400004h	FFFF0000h	0010h	10011	00	0000h	003F0004h	0000h
DRAV	A1,A0	00400004h	00010001h	0010h	00000	11	0000h	00410005h	0000h
DRAV	A1,A0	00400004h	00400004h	0010h	00000	00	00FFh	00800008h	FF00h

Syntax	DSJ Rd, Address	
Execution	Rd – 1 → Rd If Rd $\neq$ 0, then (offset × 16) + PC' → PC If Rd = 0, then go to next instruction	
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2 0 0 0 0 1 1 1 0 1 1 0 0 R R Rd	10
	offset	
Description	DSJ decrements the contents of the destination register by 1. Deper the decremented value of Rd, the TMS34020 either jumps or skips the	nding on ne jump:
	<b>Rd</b> $-1 \neq 0$ . The updated PC (used in the jump address calculation to the instruction word that immediately follows the second word of instruction. The signed word offset is converted to a bit offset by mu by 16. The new PC address is then obtained by adding the resulting offset (offset × 16) to the current PC.	1) points the DSJ Iltiplying g signed
	■ Rd –1 = 0. The TMS34020 skips the jump and continues program tion with the next sequential instruction.	n execu-
	The Address operand is a 32-bit address. The assembler calculates thas (Address – PC')/16, where PC' is the address of the instruction wor diately following the second word of the DSJEQ instruction; this resignment range of –32,768 to +32,767 words. (The offset is the second instruct of the opcode.)	ne offset rd imme- sults in a struction
	The DSJ instruction is useful for large loops involving a counter. For loops, the assembler automatically translates the DSJ mnemonic into instruction.	r shorter a DSJS
Machine States	2 if no jump 3 if jump	
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>	
Examples	Code         Before A5         After A5         Jump taken?           DSJ A5, LOOP         00000009h         00000008h         Yes           DSJ A5, LOOP         00000001h         0000000h         No           DSJ A5, LOOP         0000000h         FFFFFFFh         Yes	

Syntax	DSJEQ Rd, Address							
Execution	If Z = 1, then Rd – 1 $\rightarrow$ Rd If Rd $\neq$ 0, then PC' + (offset × 16) $\rightarrow$ PC If Rd = 0, then go to next instruction							
	If $Z = 0$ , then go to next instruction							
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         0       0       0       0       1       1       0       1       1       0       1       Rd         offset							
Description	The DSJEQ instruction evaluates the status Z bit. Depending on the value of that bit, the TMS34020 either skips the jump, or decrements Rd and then makes a decision to jump or skip the jump:							
	<b>Z</b> = 1. The TMS34020 decrements the contents of the destination register by 1.							
	Rd – 1 ≠ 0. The TMS34020 jumps relative to the current PC. The current PC points to the instruction word that immediately follows the second word of the DSJEQ instruction. The signed word offset is converted to a bit offset by multiplying by 16. The new PC address is then obtained by adding the resulting signed offset (offset × 16) to the address of the next instruction.							
	Rd – 1 = 0. The TMS34020 skips the jump and continues program execution at the next sequential instruction.							
	<b>Z</b> = 0. The TMS34020 skips the jump and continues program execution at the next sequential instruction.							
	The <i>Address</i> operand is a 32-bit address. The assembler calculates the offset as (Address – PC')/16, where PC' is the address of the instruction word immediately following the second word of the DSJ instruction; this results in a jump range of $-32,768$ to $+32,767$ words. (The offset is the second instruction word of the opcode.)							
	You can use this instruction after an explicit or implicit compare to 0. Additional information on these types of compares can be obtained in the CMP, CMPI, and MOVE-to-register instructions.							
Machine States	2 if no jump 3 if jump							
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>							

TMS34020 Assembly Language Instruction Set

Examples	Code	<b>Before</b>		<u>After</u>	
		A5	NCZV	A5	Jump taken?
	DSJEQ A5,LOOP	00000009h	x x 1 x	0000008h	Yes
	DSJEQ A5,LOOP	00000001h	x x 1 x	00000000h	No
	DSJEQ A5,LOOP	00000000h	x x 1 x	FFFFFFFFh	Yes
	DSJEQ A5,LOOP	00000009h	x x 0 x	00000009h	No
	DSJEQ A5,LOOP	00000001h	ххОх	00000001h	No
	DSJEQ A5,LOOP	00000000h	x x 0 x	00000000h	No

Syntax	DSJNE Rd, Address
Execution	If Z = 0, then Rd – 1 $\rightarrow$ Rd If Rd $\neq$ 0, then PC' + (offset × 16) $\rightarrow$ PC If Rd = 0, then go to next instruction If Z = 1, then go to next instruction
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         0       0       0       0       1       1       0       1       1       0       R       Rd         offset
Description	The DSJNE instruction evaluates the status Z bit. Depending on the value of that bit, the TMS34020 either skips the jump or decrements Rd and then makes a decision to jump or skip the jump:
	<b>Z</b> = <b>0.</b> The TMS34020 decrements the contents of Rd by 1.
	Rd – 1 ≠ 0. The TMS34020 jumps relative to the current PC. The current PC points to the instruction word that immediately follows the second word of the DSJNE instruction. The signed word offset is converted to a bit offset by multiplying by 16. The new PC address is then obtained by adding the resulting signed offset (offset × 16) to the address of the next instruction.
	■ Rd-1=0. The TMS34020 skips the jump and continues program exe- cution at the next sequential instruction.
	<b>Z</b> = 1. The TMS34020 skips the jump and continues program execution at the next sequential instruction.
	The Address operand is a 32-bit address. The assembler calculates the offset as (Address – PC')/16, where PC' is the address of the instruction word immediately following the second word of the DSJNE instruction; this results in a jump range of –32,768 to +32,767 words. (The offset is the second instruction word of the opcode.)
	You can use this instruction after an explicit or implicit compare to 0. Additional information on these types of compares can be obtained in the CMP, CMPI, and MOVE-to-register instructions.
Machine States	2 if no jump 3 if jump
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>

Examples	<u>Code</u>	<b>Before</b>		<u>After</u>	
		A5	NCZV	A5	Jumps taken?
	DSJNE A5,LOG	OP 0000009h	x x 1 x	0000009h	No
	DSJNE A5, LOG	OP 0000001h	x x 1 x	0000001h	No
	DSJNE A5,LO	OP 0000000h	x x 1 x	00000000h	No
	DSJNE A5,LOG	OP 0000009h	x x 0 x	0000008h	Yes
	DSJNE A5,LOG	OP 0000001h	x x 0 x	00000000h	No
	DSJNE A5,LOG	OP 0000000h	x x 0 x	FFFFFFFh	Yes

Syntax	DSJS Rd, Addres	SS		
Execution	$Rd - 1 \rightarrow Rd$ If $Rd \neq 0$ , then $PC$ If $Rd = 0$ , then go t	;' + (offset × 16) to next instructio	→ PC n	
Instruction Words	15 14 13 12 0 0 1 1	11 10 9   1   D	8 7 6 5 Offset	4 3 2 1 0 R Rd
Fields	D is a 1-bit directi D=0 forward D=1 backwa	on bit (from PC' l jump ard jump	to Address):	
Description	DSJS decrements the result, the TMS	the contents of t S34020 either ju	he destination re mps or skips the	gister by 1. Depending on jump:
	Rd – 1 ≠ 0. The tion word that is word offset is r range of –30 t	e TMS34020 jum mmediately follov nultiplied by 16 to o +32 words fror	ps relative to PC ws the DSJS instr convert it to a bi n PC'.	'. PC' points to the instruc- ruction. Internally, the 5-bit t offset. This allows a jump
	If direction resulting of	<b>n bit D = 0</b> . The offset to PC'.	new PC address	is obtained by adding the
	If direction the resulti	<b>n bit D = 1</b> . The ing offset from P	new PC address C'.	is obtained by subtracting
	<b>Rd – 1 = 0</b> . The tion at the next	e TMS34020 skij t sequential insti	os the jump and o ruction.	continues program execu-
	The <i>Address</i> opera as (Address – PC' PC'. (The offset is	and is a 32-bit ad )/16; this results encoded as par	dress. The asse in a jump range t of the instructic	mbler calculates the offset of –30 to +32 words from on word.)
	Note that the asser is the jump is not i	mbler also calcula n the range.	ates the value of <i>l</i>	Dand generates a warning
	This instruction is	useful for coding	g tight loops for c	ache-resident routines.
Machine States	2 if no jump 3 if jump			
Status Bits	<ul><li>N Unaffected</li><li>C Unaffected</li><li>Z Unaffected</li><li>V Unaffected</li></ul>			
Examples	<u>Code</u>	<u>Before</u> A5	<u>After</u> A5	Jump taken?
	DSJS A5,LOOP DSJS A5,LOOP DSJS A5,LOOP	00000009h 00000001h 00000000h	00000008h 00000000h FFFFFFFh	Yes No Yes

TMS34020 Assembly Language Instruction Set

Syntax	EINT						
Execution	$1 \rightarrow IE$						
Instruction Words	15 14 0 0	13 12 11 1 0 0 1 1	0 9 8 0 1	7 6 0 1	5 4 1 0	32 00	1 0 0 0
Description	EINT sets enabled. V ate bits in is unaffect	the global inter Vhen IE=1, indiv the INTENB inte ed.	rupt enable idual interru errupt mask	bit (IE) to pts are en register. T	o 1, allowi abled by s The rest of	ing intern setting the f the state	upts to be appropri- us register
	The DINT	instruction disat	pies interrup	DIS.			
Machine States	3						
Status Bits	N Unaffe C Unaffe Z Unaffe V Unaffe IE 1	ected ected ected ected					
Examples	Code EINT EINT	<u>Before</u> ST 00000010h 00200010h	<u>After</u> ST 00200010h 00200010h				

### EMU Initiate Emulation

Syntax	EMU count
Execution	Conditionally enter emulator mode
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         0       0       0       0       0       0       1       0       0       0       count
Description	This instruction is not for general use. It is supplied to proved support for emula- tion of the TMS34020.
Machine States	8 (or more if EMU mode is entered)
Status Bits	<ul> <li>N Indeterminate</li> <li>C Indeterminate</li> <li>Z Indeterminate</li> <li>V Indeterminate</li> </ul>

Syntax	<b>EXGF</b> <i>Rd</i> [, <i>F</i> ]
Execution	Rd $\rightarrow$ FS0, FE0 <b>or</b> Rd $\rightarrow$ FS1, FE1 FS0, FE0 $\rightarrow$ Rd <b>or</b> FS1, FE1 $\rightarrow$ Rd
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         1       1       0       1       0       1       F       1       0       0       R       Rd
Description	EXGF exchanges the 6 LSBs of the destination register with the selected 6 bits of field information (field size and field extension). Bit 5 of the 6-bit quantity in Rd is exchanged with the field-extension value. The upper 26 bits of Rd are cleared.



EXGF's *F* parameter is optional:

- F=0 selects FS0, FE0 to be exchanged
- F=1 selects FS1, FE1 to be exchanged

If you do not specify an F parameter, the default is 0.

For more information, refer to Section 4.1, The Status Register, on page 4-2.

Machine States	F0 1 F1 2				
Status Bits	<ul> <li>N Unaffect</li> <li>C Unaffect</li> <li>Z Unaffect</li> <li>V Unaffect</li> </ul>	ed ed ed ed			
Examples	Code EXGF A5,0 EXGF A5,1	<u>Before</u> A5 FFFFFFC0h FFFFFFC0h	<b>ST</b> F0000FFFh F0000FFFh	<u>After</u> A5 0000003Fh 0000003Fh	<b>ST</b> F0000FC0h F000003Fh

## **EXGPC** Exchange Program Counter

Syntax	EXGPC Rd										
Execution	$Bd \rightarrow PC P$	$C' \rightarrow Bd$									
Instruction Words	15     14     13       0     0     0	12 11 0 0	10 9 0 (	) 8 D 1	7	65 01	4 R	3	2 Ro	1	0
Description	EXGPC excha ter contents. A contains the a instruction.	inges the r fter this in address o	next pro struction f the ir	ogram c on has t ostructio	counter been ex on imn	value w xecuted, nediately	ith the the d / follo	e dest lestin wing	tinati atior the	on re n regi EXC	gis- ster àPC
	Note that the aligned).	TMS34020	0 sets t	he 4 LS	SBs of	the prog	gram o	count	ter to	o (v	/ord
	This instructio in a register (ra by repeating th tine address n	n provides ather than ne instruct nust be rel	s a <i>quic</i> on the s ion at th loaded	ck call c stack). T ne end c followir	capabil The retu of the s ng eacl	ity by sa urn from ubroutin h call-rei	ving t the ca le. No turn o	the re all is a te tha perai	eturn accor at the tion.	addr mplis sub	ress hed rou-
Machine States	2										
Status Bits	<ul><li>N Unaffected</li><li>C Unaffected</li><li>Z Unaffected</li><li>V Unaffected</li></ul>	t t t									
Examples	Code EXGPC A1 EXGPC A1	Before A1 00001C10 00001C50	)h )h	<b>PC</b> 000020 000020	980h 980h	<u>After</u> A1 000020 000020	)90h )90h	<b>PC</b> 000	; 001C 001C	;10h ;50h	

Syntax	EXGPS	Rd											
Execution	$Rd \rightarrow PC$	;, PC'-	$\rightarrow \text{Rd}$										
Instruction Words	15 14 0 0	13 12 0 0	11 0 (	10 9 ) 1	8	7	6 0 1	5 R	4	3	2 Rd	1	0
Description	EXGPS ex of PSIZE r or 32). The contents a tion, refer	change egister. e destina re writter to Sectio	s the co The pix ation re n with a on 12.6	ontents el size gister i 16-bit v , <u>Auxili</u>	of the is assu s loade write to ary Gr	e des umec ed wi o the l caphic	tinatic I to be ith the PSIZE <u>cs Ins</u>	n reg a leg pixe regi tructi	gister gal va I size ster. <u>ons</u> ,	with alue e, an For on p	n the (1, 2, id its more bage	cont 4, 8 prev infor 12-1	ents , 16, ious ma- 7.
Implied Operands	Address		Name		Desc	riptio	on and	l Ope	ratio	ns			
	C000015	0h	PSIZE		Pixe	l size	(1,2,4	,8,16	,32)				
Machine States	2 (1)												
Status Bits	N Unaffe C Unaffe Z Unaffe V Unaffe	ected ected ected ected											
Examples	EXGPS A0												
	Before After	A0 = A0 =	000000	101 108	PSIZE PSIZE	=	0008 0001						

\*\*\*\*\*

Syntax	FILL L										
Execution	COLOR1 pix	els $\rightarrow$ pixel a	rray (wit	h proces	sing)						
Instruction Words	15 14 13 0 0 0	12 11 10 0 1 1	9	8 7 1 1	6 5 1 0	5	4	3 0	2	1	0
Description	FILL process with a destination	es a set of sour ation pixel arra	ce pixel ay.	values (s	pecifie	edby	the (	COL	.OR	l reg	ister)
	This instruction defined in the combined with ations.	on operates COLOR1 reg th destination	on a 2-c jister. As pixels a	dimensio the FILL according	nal ar proce to the	ray ( eeds, e se	of pi , the lecte	xels sou ed g	irce    rce    rapl	ng p pixel lics d	oixels s are oper-
	Note that the or a register; t address of th	L parameter in he L is entered e pixel array a	n the inst d as part is a linea	truction s of the ins ar addres	yntax tructio ss.	does n and	s not d ide	repi ntifi	rese es th	nt a v ie sta	arting/
	The following and define bo	set of implied oth the source	l operan pixels a	ds gover nd the de	n the c estinat	opera tion a	ation array	of t	he ir	nstru	ction
Implied Operands	Register	Name	Format	Descri	ption						
	B2 †	DADDR	Linear	Pixel	array s	startin	ig ad	dres	s		
	B3	DPTCH	Linear	Pixel	array p	bitch					<u> </u>
	B7	DYDX	XY	Pixel	array c	limen	sion	s (ro	ws:c	olum	ns)
	B9	COLOR1	Pixel	Fill co	olor or 3	32-bit	t patte	ern			
	† Changed by	/ FILL during ex	ecution.								
	Address	Name	Descri	ption and	Opera	ation	s				
	C00000B0h	CONTROL	PPOP T TM	Pixel-pro Enables Selects	cessin transpa 1 of 3 tr	g ope arenc ransp	eratio cy ope aren	ns ( erati cy o	22 op on ption	otions	5)
	C0000150h	PSIZE	Pixel s	ize (1,2,4	,8,16,3	2)					
	C0000160h	PMASK (32 bits)	Plane	mask — p	oixel for	rmat					
	Due to the pip setting up the when you exe correct value an MWAIT (p a description	elining of mem I/O registers) cute the FILL for execution, age 13-178) ir of the potentia	nory write may not instruction you may instruction al latenc	es, the <i>las</i> , in some on. To ens want to fo n. Refer f y of write	st I/O re cases sure th ollow th to Sec es to I/0	egisto s, cor at thi he wr tion O reg	er tha ntain is reg rite to 4.5.6 giste	at yo the giste tha o tha o n rs.	ou wi desi er col at loc pag	rite (v ired v ntain atior e 4-1	when value is the i with 3 for

# **Destination Address** The contents of the DADDR, DPTCH, and DYDX registers define the location of the destination pixel array. For a detailed description, refer to Section 12.5, <u>Pixel Array Instructions</u>, on page 12-8.

Pixel Processing	Set PPOP[[CONTROL]] to select a pixel-processing operation. This operation is applied to the pixel as it is moved to the destination location. There are 16 Boolean and 6 arithmetic operations; the default operation at reset is <i>replace</i> ( $S \rightarrow D$ ). Note that the destination data is read through the plane mask and then processed. The 6 arithmetic operations do not operate with a pixel size of 1 bit per pixel. For more information, refer to Section 12.8, <u>Pixel Processing</u> , on page 12-27.
Window Checking	Window checking <b>cannot</b> be used with this instruction. The contents of the WSTART and WEND registers are ignored.
Corner Adjust	There is no corner adjust for this instruction. The direction of the FILL is fixed as increasing linear addresses.
Transparency	You can enable transparency for this instruction by setting T[[CONTROL]] to 1. You can also select 1 of 3 transparency options by setting TM[[CONTROL]]. For more information, refer to Section 12.9, <u>Transparency</u> , on page 12-36.
Interrupts	This instruction can be interrupted at a word or row boundary of the destination array. For more information, refer to Section 6.6, <u>Interrupting Graphics Operations</u> , on page 6-13.
Plane Masking	The plane mask is enabled for this instruction. For more information, refer to Section 12.10, <u>Plane Masking</u> , on page 12-39.
Shift Register Transfers	If CST[DPYTCTL] is set, each memory read or write initiated by the FILL generates a shift-register-transfer read or write cycle at the selected address. This operation can be used for bulk memory clears or transfers. (Not all VRAMs support this capability.) For more information, refer to subsection 9.13.4, <u>VRAM Bulk Initialization</u> , on page 9-47.
Machine States	complex instruction
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>
Examples	These FILL examples use the following implied operand setup.
	Register File B:         I/O Registers:           DADDR         = 00002010h         PSIZE = 0008h           DPTCH         = 00000080h         PSIZE = 0008h           DYDX         = 0002000Dh         COLOR1
	Assume that memory contains the following values before instruction execution.
	Linear Address Data 00002000h 1100h, 3322h, 5544h, 7766h, 9988h, BBAAh, DDCCh, FFEEh 00002080h 1100h, 3322h, 5544h, 7766h, 9988h, BBAAh, DDCCh, FFEEh

Example 1

This example uses the pixel-processing replace  $(S \rightarrow D)$  operation. Before instruction execution, PMASK = 0000h and CONTROL = 0000h (T=0, PPOP=00000).

After instruction execution, memory contains the following values:

Linear Data Address Data 00002000h 1100h, 3030h, 5F30h 1100h, 3030h, 3030h, 3030h, 3030h, 3030h, 3030h, FF30h

Example 2

This example uses the  $(\overline{S} \text{ and } D) \rightarrow D$  pixel-processing operation. Before instruction execution, PMASK = 0000h and CONTROL = 2C00h (T=0, PPOP=01010).

After instruction execution, memory contains the following values:

Linear Address Data 00002000h 1100h, 0302h, 4544h, 4746h, 8988h, 8B8Ah, CDCCh, FFCEh 00002080h 1100h, 0302h, 4544h, 4746h, 8988h, 8B8Ah, CDCCh, FFCEh

Example 3

This example uses plane masking — the 4 MSBs are masked. Before instruction execution, PMASK = 0F0F0h and CONTROL = 0000h (T=0, PPOP=00000).

After instruction execution, memory contains the following values:

Linear	_
Address	Data
00002000h	1100h, 3020h, 5040h, 7060h, 9080h, B0A0h, D0C0h, FFE0h
00002080h	1100h, 3020h, 5040h, 7060h, 9080h, B0A0h, D0C0h, FFE0h

Syntax	FILL XY									
Execution	COLOR1 pix	els $\rightarrow$ pixel	array (with	processing	g)					
Instruction Words	15 14 13	12 11 1	10 9 8	76	5	4	3	2	1	0
	0 0 0	0 1	1   1   1	1 1	1	0	0	0	0	0
Description	FILL process with a destina	es a set of sou ation pixel ar	urce pixel v ray.	alues (spec	ified b	ythe	CO	LOR	1 reg	ister)
	This instruction defined in the combined with tions.	on operates COLOR1 re th destination	on a 2-d egister. As n pixels ac	mensional the FILL pro cording to t	array oceeds the se	of p s, the lecte	oixel e so ed g	s us urce raph	ing p pixel ics o	oixels Is are pera-
	Note that the or a register; address of th	<b>XY</b> paramete it is entered e pixel array	er in the inst as part of as an XY :	ruction synt the instructi address.	ax doe ion an	es no d ide	ot rep entif	orese ies tl	ent ar ne sta	value arting
	The following and define bo	set of implie oth the sourc	ed operand e pixels an	s govern th d the destir	e ope nation	ratio arra	n of y.	the	instru	iction
Implied Operands	Register	Name	Format	Description	n					
	B2 †‡	DADDR	XY	Pixel array	startin	g ad	dres	s		
	B3	DPTCH	Linear	Pixel array	pitch					
	B4	OFFSET	Linear	Screen orig	gin (ad	dress	s of (	),0)		
	B5	WSTART	XY	Window sta	arting c	corne	r			
	B6	WEND	XY	Window en	iding c	orner				
	B7 †‡	DYDX	XY	Pixel array	dimen	sions	s (rov	ws:co	olumn	s)
	B9	COLOR1	Pixel	Fill color or	· 32-bit	patte	ərn			
	† Changed by	/ FILLXY durir	ng executior	l <b>.</b>						
	<sup>‡</sup> Used for co	mmon rectang	gle function	with window	hit ope	eratio	n (W	/=1).		
	Address	Name	Descrip	tion and Ele	ement	s (Bi	ts)			
	C00000B0h	CONTROL	. PPOP W T TM	Pixel-process Window-cheo Enables tran Selects 1 of 3	sing op cking o sparen 3 trans	perati pera ncy pare	ons tion ncy (	(22 a	ption:	s)
	C0000140h	CONVDP	XY-to-li	near convers	ion (de	estina	ation	pitch	1)	
	C0000150h	PSIZE	Pixel siz	e (1,2,4,8,16	5,32)					
	C0000160h	PMASK (32 bits)	Plane m	ask — pixel	format	t				

Due to the pipelining of memory writes, the *last* I/O register that you write to may not, in some cases, contain the desired value when you execute the FILL

	XY instruction. To ensure that this register contains the correct value for execu- tion, you may want to follow the write to that location with an MWAIT instruction (page 13-178). Refer to Section 4.5.6 on page 4-13 for a description of the potential latency of writes to I/O registers.
Destination Array	The location of the destination pixel array is defined by the contents of the DADDR, DPTCH, CONVDP, OFFSET, and DYDX registers. For a detailed description, refer to Section 12.5, <u>Pixel Array Instructions</u> , on page 12-8.
Pixel Processing	Pixel processing can be used with this instruction. PPOP[[CONTROL]] specifies the pixel-processing operation that is applied to pixels as they are processed with the destination array. There are 16 Boolean and 6 arithmetic operations; the default case at reset is the <i>replace</i> ( $S \rightarrow D$ ) operation. Note that the destination data is read through the plane mask and then processed. The 6 arithmetic operations do not operate with a pixel size of 1 bit per pixel. For more information, refer to Section 12.8, <u>Pixel Processing</u> , on page 12-27.
Window Checking	The window operations can be used with this instruction. You can select window pick, violation detect, or preclipping by setting W[[CONTROL]] to 1, 2, or 3, respectively. For more information, refer to Section 12.7, <u>Window Checking</u> , on page 12-19.
Corner Adjust	There is no corner adjust for this instruction. The direction of the FILL is fixed as increasing linear addresses.
Transparency	You can enable transparency for this instruction by setting T[[CONTROL]] to 1. Select 1 of 3 transparency modes by setting TM[[CONTROL]]. For more information, refer to Section 12.9, <u>Transparency</u> , on page 12-36.
Interrupts	This instruction can be interrupted at a word or row boundary of the destination array. For more information, refer to Section 6.6, <u>Interrupting Graphics Operations</u> , on page 6-13.
Plane Masking	The plane mask is enabled for this instruction. For more information, refer to Section 12.10, <u>Plane Masking</u> , on page 12-39.
Shift Register Transfers	If CST[[DPYCTL]] is set, each memory read or write initiated by the FILL gener- ates a shift-register-transfer read or write cycle at the selected address. This operation can be used for bulk memory clears or transfers. (Not all VRAMs support this capability.) For more information, refer to subsection 9.13.4, <u>VRAM Bulk Initialization</u> , on page 9-47.
Machine States	complex instruction
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>

#### Examples

These FILL examples use the following implied operand setup.

I/O Registers:
CONVDP = 0017h
PSIZE = 0004h
PMASK = 0000h
CONTROL = 0000h
(W=00, T=0, PPOP=00000)
FFFFFFFh

Assume that memory contains the following values before instruction execution.

#### Linear Address Data 00015200h 3210h, 7654h, BA98h, FEDCh, 3210h, 7654h, BA98h, FEDCh 00015300h 3210h, 7654h, BA98h, FEDCh, 3210h, 7654h, BA98h, FEDCh 00015400h 3210h, 7654h, BA98h, FEDCh, 3210h, 7654h, BA98h, FEDCh

## **Example 1** This example uses the *replace* ( $S \rightarrow D$ ) pixel-processing operation. Before instruction execution, PMASK = 0000h and CONTROL = 0000h (T=0, W=00, PPOP=00000).

After instruction execution, memory contains the following values:

#### Linear

AddressData00015200h3210h, F654h, FFFFh, FFFFh, FFFFh, FFFFh, FFFFh, BA9Fh, FEDCh00015300h3210h, F654h, FFFFh, FFFFh, FFFFh, FFFFh, FFFFh, BA9Fh, FEDCh00015400h3210h, F654h, FFFFh, FFFFh, FFFFh, FFFFh, FFFFh, BA9Fh, FEDCh

#### XY Addressing

Ŷ	X Address 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 1 1 1 1	
A d 52 d r 53 e s 54 s	0 1 2 3 4 5 6 F F F F F F F F F F F F F F F F S A B C D E F 0 1 2 3 4 5 6 F F F F F F F F F F F F F F F S A B C D E F 0 1 2 3 4 5 6 F F F F F F F F F F F F F F F F S A B C D E F	

Example 2

This example uses the (*D* XOR S)  $\rightarrow$  *D* pixel-processing operation. Before instruction execution, PMASK = 0000h and CONTROL = 2800h (T=0, W=00, PPOP=01010).

After instruction execution, memory contains the following values:

Y	0 0 0 1	0 2	0 3	0 4	0 5	0 6	0 7	0 8	0 9	0 A	<b>X</b> 0 B	<b>A</b> 0 C	<b>dd</b> 0 D	0 E	ess 1 F	; 1 0	1 1	1 2	1 3	1 4	1 5	1 6	1 7	1 8	1 9	1 A	1 B	1 1 C I	1 D E	1 F	_
A d 52 d r 53 e s 54 s	01 01 01	2 2 2	3 3 3	4 4 4	5 5 5	6 6 6	8 8 8	7 7 7	6 6 6	5 5 5	4 4 4	3 3 3	2 2 2	1 1 1	0 0 0	F F F	E	D D D	c c c	B B B	A A A	9 9 9	8 8 8	7 7 7	9 9 9	A A A	B B	с і с і с і		F	_

Example 3

This example uses window operation 3: the destination is clipped. Before instruction execution, PMASK = 0000h and CONTROL = 00C0h (T=0, W=11, PPOP=00000).

After instruction execution, memory contains the following values:

Y	X Address 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 1 1 1 1 1
A d 52	0 1 2 3 4 5 6 7 8 9 A B F F F F F F F F F 5 6 7 8 9 A B C D E F
r 53	0 1 2 3 4 5 6 7 8 9 A B F F F F F F F F F 5 6 7 8 9 A B C D E F
s 54 s	0 1 2 3 4 5 6 7 8 9 A B F F F F F F F F F 5 6 7 8 9 A B C D E F

#### Example 4

This example uses plane masking: the MSB is masked. Before instruction execution, PMASK = 8888h and CONTROL = 0000h (T=0, W=00, PPOP=00000).

After instruction execution, memory contains the following values:

Y		0	0	02	03	0	05	0	07	0	09	0 A	<b>X</b> 0 B	<b>A</b> 0 C	dd 0 D	0 E	<b>ss</b> 1 F	; 1 0	1	12	13	1 4	1 5	1	1 7	1	1	1 A	1 <sup>-</sup> B (	11 CI	1 D E	1 F
		ŀ		_			-																	_	-	-	-					
A d	52	0	1	2	3	4	5	6	7	F	F	F	F	F	F	F	F	7	7	7	7	7	7	7	7	F	9	A	в	0 נ	ЭE	F
r	53	0	1	2	3	4	5	6	7	F	F	F	F	F	F	F	F	7	7	7	7	7	7	7	7	F	9	A	в	0 0	ΣE	F
e S	54	0	1	2	3	4	5	6	7	F	F	F	F	F	F	F	F	7	7	7	7	7	7	7	7	F	9	A	в	CI	ЭE	F
5																																

 $\backslash$ 

Syntax	FLINE {0   1}												
Execution	The two execution algorithms for FLINE are explained below. These algorithms are similar, varying only in their treatment of the case when $d=0$ .												
Instruction Words	<u>15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0</u>												
	1 1 0 1 1 1 1 0 Z 0 0 1 1 0 1 0												
Fields	The assembler sets bit 7 in the instruction word (the Z bit) to 0 or 1, depending on which FLINE algorithm you select: Z=0 selects algorithm 0 Z=1 selects algorithm 1												
Description	FLINE is a version of LINE. The major differences are:												
	<ul> <li>FLINE is faster than LINE.</li> <li>FLINE does not support windowing.</li> <li>FLINE requires B2 to be linear.</li> </ul>												
	FLINE performs the inner loop of Bresenham's line-drawing algorithm. This type of line draw plots a series of points $(x_i, y_i)$ either diagonally or laterally with respect to the previous point. Movement from pixel to pixel always proceeds in a dominant lateral direction. The algorithm may or may not also increment in the direction with the smaller dimension (this produces a diagonal movement). Two XY-format registers supply the XY increment values for the two possible movements. The FLINE instruction maintains a decision variable, <i>d</i> , that acts as an error term, controlling movement in either the dominant or diagonal direction. The algorithm operates in one of two modes, depending on how the condition <i>d</i> =0 is treated.												
	During FLINE execution, some portion of a line from $(x_0,y_0)$ to $(x_1,y_1)$ is drawn The line is drawn so that the axis with the largest extent has dimension <i>a</i> , an the axis with the least extent has dimension <i>b</i> . Thus, <i>a</i> is the larger (in absolut terms) of $y_1 - y_0$ , or $x_1 - x_0$ and <i>b</i> is the smaller of the two. This means that $a \ge b \ge 0$ .												
	The LINIT instruction (page 13-146) is designed to do most of the setup for FLINE with the exception of the XY to linear conversion of DADDR.												
	The following values must be supplied to draw a line from $(x_0,y_0)$ to $(x_1,y_1)$ :												
	Set the value in DADDR to be the <b>linear</b> address of the first pixel in the line at $(x_0,y_0)$ .												
	Use the line endpoints to determine the major and minor dimensions ( <i>a</i> and <i>b</i> , respectively) for the line draw; then set the DYDX register to this value ( <i>b</i> : <i>a</i> ).												
	Place the signed XY increment for a movement in the diagonal (or minor) direction ( $d \ge 0$ for Z=0, $d > 0$ for Z=1) in the INC1 register.												

- Place the signed XY increment for a movement in the dominant (or major) direction (d < 0 for Z=0,  $d \le 0$  for Z=1) in the INC2 register.
- Set the initial value of the decision variable in register B0 to 2b a.
- Set the initial count value in the COUNT register to a + 1.
- Set the COLOR1 and COLOR0 registers.
- Set the PATTERN register to the required pattern.

FLINE handles the contents of PATTERN in the same way as LINE (unlike PFILL XY). With FLINE, the first pixel drawn is controlled by bit 0 of the PATTERN register.

The PATTERN register contains a 32-bit repeating line-style pattern. If bit 0 of PATTERN is 0, then the first pixel drawn by LINE is a COLOR0 pixel. If bit 0 of PATTERN is 1, then the first pixel drawn by LINE is a COLOR1 pixel. The second pixel drawn by LINE is controlled by bit 1 of B13, and so on. If the line is longer than 32 pixels, the PATTERN is reused cyclically; therefore, the 33<sup>rd</sup> pixel on the line is once again controlled by bit 0 of PATTERN. As each pixel is drawn, the contents of PATTERN are rotated right (circular shifted) by 1 bit. The LSB of the rotated pattern controls the next pixel the instruction puts out.

If PATTERN contains all 1s, the line is drawn in a solid color using the replicated pixel value contained in COLOR1; if PATTERN contains all 0s, the line is drawn in a solid color using COLOR0.

The FLINE instruction may use one of two algorithms, depending on the value of Z.

## Algorithm 0 (Z=0):

While COUNT > 0 COUNT = COUNT - 1 Draw the next pixel If  $d \ge 0$  d = d + 2b - 2aPOINTER = POINTER + INC1 Else d = d + 2b; POINTER = POINTER + INC2

## Algorithm 1 (Z=1):

While COUNT > 0 COUNT = COUNT - 1 Draw the next pixel If d > 0 d = d + 2b - 2aPOINTER = POINTER + INC1 Else d = d + 2b; POINTER = POINTER + INC2

For more information about FLINE, refer to Section 12.4, <u>Line Instructions</u>, on page 12-7.

#### Implied Operands

Regis	ter	Name	Format	Description
B0	†	SADDR	Integer	Decision variable (d)
B2	†	DADDR	Linear	Starting point
B3		DPTCH	Linear	Destination pitch (only used if CONVDP specifies arbitrary pitch)
B7		DYDX	XY	(b: a) = Minor: major dimension
B8	‡	COLOR0	Pixel	COLORO
B9	¶	COLOR1	Pixel	COLOR1
B10	†	COUNT	Integer	Loop count
B11		INC1	XY	Minor axis (diagonal) increment
B12		INC2	XY	Major axis (dominant) increment
B13		PATTERN	Pattern	Pattern register

T These registers are changed by instruction execution.

<sup>‡</sup> This register is only required if there are 0's in the pattern.

§ This register is only required if there are 1's in the pattern.

Address	Name	Description and Elements (Bits)
C00000B0h	CONTROL	PPOPPixel-processing operations (22 options)TTransparency operationTMSets transparency mode
C0000140h	CONVDP	XY-to-linear conversion (destination pitch)
C0000150h	PSIZE	Pixel size (1,2,4,8,16,32)
C0000160h	PMASK (32 bits)	Plane mask — pixel format

Due to the pipelining of memory writes, the *last* I/O register that you write to may not, in some cases, contain the desired value when you execute the FILL XY instruction. To ensure that this register contains the correct value for execution, you may want to follow the write to that location with an MWAIT instruction (page 13-178). Refer to Section 4.5.6 on page 4-13 for a description of the potential latency of writes to I/O registers.

**Pixel Processing**PPOP[[CONTROL]] specifies the operation to be applied to the pixel as it is<br/>written. There are 22 operations; the default case at reset is the pixel-process-<br/>ing replace (S  $\rightarrow$  D) operation. For more information, refer to Section 12.8, <u>Pix-<br/>el Processing</u>, on page 12-27.

*Window Checking* Window checking **cannot** be used with this instruction. The line must be preclipped; for more information, refer to Section 12.7, <u>Window Checking</u>, on page 12-19.

TransparencyYou can enable transparency for this instruction by setting T[[CONTROL]] to1. Select 1 of 3 transparency options by setting TM[[CONTROL]]. For more<br/>information, refer to Section 12.9, <u>Transparency</u>, on page 12-36.

Plane Masking	The p Secti	olane mask is enable on 12.10, <u>Plane Mas</u>	d for this instruction. For more information, refer to king, on page 12-39.
Interrupts	This i refer	instruction can be inte to Section 6.6, <u>Interr</u>	errupted at a pixel boundary. For more information, upting Graphics Operations, on page 6-13.
Machine States	Refe	er to Section 15.1 on	page 15-2.
Status Bits	N ( C ( Z ( V (	Jnaffected Jnaffected Jnaffected Jnaffected	
Example	This i on the and ( LINE FILL	is an example of a C-c e screen. It takes 4 pa (xend, yend), and use or FLINE. It also spe for these.	ompatible assembly routine which draws a solid line rameters from the C parameter stack: (xstart, ystart) as LINIT to initialize the B-file implied operands for acial-cases horizontal and vertical lines and uses a
• •	This have	routine makes the ass been initialized by th	umes that the following B registers and I/O registers e caller:
	B-file I/O re	e registers DPTCH, ( egisters CONTRO	DFFSET, WSTART, WEND and COLOR1 L, CONVDP, PSIZE, and PMASK
STK DADDR DYDX INC1 PATTERN	.set A .set E .set E .set E .set E	A14 ; 32 ; 37 ; 311 ; 313 ;	C-parameter stack pointer Destination address register Delta X/delta Y register Minor axis (diagonal) increment Pattern register
_draw_line:	.globl	_draw_line ; SP,A0,A1 SP,B2,B7,B10,B11 <-A14,A0,1 ;	Reference for external calls ,B12,B13,B14 get xs
	move * sll 1 movy P move *	<pre>*-A14,A1,1 ; L6,A1 A1,A0 *-A14,A1,1 ; </pre>	get ys concatenate ys::xs get xe
	move * sll 1 movy A move A	x-A14,A8,1 ; 16,A8 A8,A1 ; A0,DADDR	get ye concatenate ye::xe
	move A movi - linit jrc e	Al,DYDX -1,PATTERN exit	line is entirely outside window
	jrp c	cliptst	line neither vertical or horizontal

TMS34020 Assembly Language Instruction Set

; Confirmed that line is vertical or horizontal, so FILL can be used ; instead of LINE. Now determine whether line points in the +x or ; +y direction, or in the -x or -y direction. vorh: cmpxy A0,A1 ; a = xe - xs, b = ye - ysjrv negdir ; jump if -x direction jrc neqdir ; jump if -y direction ; Horizontal or vertical line points in +x or +y direction. move A0,DADDR ; DADDR = ys::xs subxy A0,A1 ; a = xe - xs, b = ye - ysmove A1,DYDX ; DYDX = length of line jruc vorhline ; Horizontal or vertical line points in -x or -y direction. negdir: move A1,DADDR ; DADDR = ye::xe subxy A1,A0 ; a = xs - xe, b = ys - yemove A0, DYDX ; DYDX = length of line ; Draw the vertical or horizontal line. ; ++DX, ++DY vorhline: addi 010001h,DYDX fill XY ; draw line jruc exit cliptst: jrv draw ; clipping required use LINE cvdxyl DADDR ; convert to linear address ; does line point up or down? move INC1, INC1 jrlt fdown fline 0 ; draw Bresenham line jruc exit fdown: fline 1 ; draw Bresenham line jruc exit draw: move INC1, INC1 ; does line point up or down? jrlt down line O ; draw Bresenham line jruc exit down: line 1 ; draw Bresenham line exit: mmfm SP, B2, B7, B10, B11, B12, B13, B14 mmfm SP,A0,A1 ; return to caller rets 2

## FPIXEQ

Execution

**Syntax** 

Scan from pixel pointed to by MADDR to find the first pixel equal to COLOR0, up to the number of pixels specified in MPTCH.

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
0	0	0	0	1	0	1	0	1	0	1	1	1	0	1	1

Description

Instruction Words

FPIXEQ compares pixels in memory to the value in the COLOR0 register. The search ends when the current pixel is equal to the COLOR0 pixel of the number of pixels specified in MPTCH has been searched. The search takes place in either an postincrementing or a predecrementing fashion. MPTCH specifies the number of pixels to search. The count is positive in the postincrementing case (*search right*) and negative in the predecrementing case (*search left*).

The *magnitude* of the MPTCH counter decreases as the search continues.

If the instruction finds a pixel, Z is set to 1, the instruction aborts. MADDR is left pointing to the next pixel to check (postincrementing case) or the last pixel checked (predecrementing case).

If the instruction is interrupted, the pointers are set so the FPIXEQ resumes automatically. That is, the PC is decremented to point back to the FPIXEQ, the ST and PC are stacked, MADDR is left pointing to the next pixel to check (postincrementing case) or the last pixel checked (predecrementing case, and before the TRAP routine is started. The RETI at the end of the TRAP resumes the FPIXEQ instruction from the next pixel to be checked.For more information, refer to Section 12.6, Auxiliary Graphics Instructions, on page 12-17.

This instruction is useful for seedfills, data compression, and edge-flag fills.

Register	Name	Format	Description				
B8	COLOR0	Pixel	COLOR0				
B10	MADDR	Linear	Search start pointer				
B11	MPTCH	Integer	Number of pixels to search				
Address	Name	Description and Elements (Bits)					
C0000150h	PSIZE	Pixe	el size				
C0000160h	PMASK (32	2 bits) Plar	ne mask — pixel format				

Due to the pipelining of memory writes, the *last* I/O register that you write to may not, in some cases, contain the desired value when you execute the FPIXEQ instruction. To ensure that this register contains the correct value for execution, you may want to follow the write to that location with an MWAIT (page 13-178) instruction. Refer to Section 4.5.6 on page 4-13 for a description of the potential latency of writes to I/O registers.

#### Implied Operands

- *Pixel Processing* Pixel processing **cannot** be used with this instruction.
- *Window Checking* Window checking **cannot** be used with this instruction.
- *Transparency* Transparency cannot be used with this instruction.
- *Plane Masking* The plane mask is enabled for this instruction. For more information, refer to Section 12.10, <u>Plane Masking</u>, on page 12-39.
- Machine States complex instruction
- Status Bits
- N Unaffected
- **C** Unaffected
- Z 1 if pixel found, 0 otherwise
- V Unaffected

Syntax	FPIXNE											
Execution	Scan from pixel pointed to by MADDR to find first pixel that is <b>not</b> equal to COLOR0, up to the number of pixels specified in MPTCH.											
Instruction Words	15 14 13 0 0 0	12 11 10 0 1 0	9 8 1 0	7         6         5         4           1         1         0         1	3     2     1     0       1     0     1     1							
Description	FPIXNE comp search ends of search takes fashion. MPT( this is <b>not</b> an ( <i>search right</i> )	pares pixels in memory to the value in the COLOR0 register. The when the current pixel is <u>not</u> equal to the COLOR0 pixel. The place in either an postincrementing or a predecrementing CH specifies the maximum number of pixels to search. Note than XY value. The count is positive in the postincrementing case and negative in the predecrementing case ( <i>search left</i> ).										
	If the instruction finds a pixel, Z is set to 1, the instruction aborts. MADDR is left pointing to the next pixel to check (postincrementing case) or the last pixel checked (predecrementing case).											
	The <i>magnitude</i> of the MPTCH counter decreases as the search continues. If the instruction is interrupted, the pointers are set so the FPIXNE resumes automatically. That is, the PC is decremented to point back to the FPIXNE, the ST and PC are stacked, MADDR is left pointing to the next pixel to check (post-incrementing case) or the last pixel checked (predecrementing case, and the TRAP routine is started. The RETI at the end of the TRAP resumes the FPIXNE instruction from the next pixel to be checked. For more information, refer to Section 12.6, <u>Auxiliary Graphics Instructions</u> , on page 12-17.											
Implied Operands	Register	Name	Format	Description								
	B10	MADDR	Linear	Search start po	binter							
	B11	MPTCH	Integer	Number of pixe	els to search							
	B8	COLOR0	Pixel	COLOR0								
	Address	Name	Description and Elements (Bits)									
	C0000150h	PSIZE	Pi	l size								
	C0000160h	PMASK (32	ane mask — pixel forn	nask — pixel format								
	Due to the pipelining of memory writes, the last I/O register that you write to											

Due to the pipelining of memory writes, the *last* I/O register that you write to may not, in some cases, contain the desired value when you execute the FPIXNE instruction. To ensure that this register contains the correct value for execution, you may want to follow the write to that location with an MWAIT (page 13-178) instruction. Refer to Section 4.5.6 on page 4-13 for a description of the potential latency of writes to I/O registers.

**Pixel Processing** 

Pixel processing **cannot** be used with this instruction.

Window Checking

Window checking cannot be used with this instruction.

*Transparency* Transparency cannot be used with this instruction.

*Plane Masking* The plane mask is enabled for this instruction. For more information, refer to Section 12.10, <u>Plane Masking</u>, on page 12-39.

Machine States complex instruction

Status Bits

N Unaffected

**C** Unaffected

- Z 1 if pixel found, 0 otherwise
- V Unaffected

## GETPC Get Program Counter into Register

Syntax	GETPC Rd											
Execution	$PC' \rightarrow Rd$											
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         0       0       0       0       0       1       0       1       0       R       Rd											
Description	GETPC copies completes, Rd c ing the GETPC can use GETPC operations. You position relative	the PC in contains the instruction with the can also to the co	to the c ne addr n. Exec EXGPC use GE ode are	estina ess of cution C and C TPC t a is kr	tion re the ins contir JUMP o acce town a	egiste struct nues instr ess re at as	er (R tion v with t uctio eloca semt	d). W vord i the n ns fo atable oly tir	/hen imme ext ir r quic e data ne.	the in ediate nstrue ck ca a are	nstru ely fo ction Il on j as wl	ction llow- You jump hose
Machine States	1				r.							
Status Bits	<ul><li>N Unaffected</li><li>C Unaffected</li><li>Z Unaffected</li><li>V Unaffected</li></ul>											
Examples	Code GETPC A1 GETPC A1	Before PC 00001BE 00001C1	00h 0h	<u>After</u> A1 00001	IBE0h IC20h							

Syntax	GETPS Rd				
Execution	$PSIZE \to Rd$				
Instruction Words	15 14 13 0 0 0	12 11 10 0 0 0	9 8 7 1 0 1	6 5 4 1 0 R	3 2 1 0 Rd
Description	GETPS fetches destination regi 16, or 32). Fo <u>Instructions</u> , on	the pixel siz ster. The pix r more inform page 12-17	e stored in the P tel size is assum mation, refer to	SIZE register a ned to be a leg Section 12.6,	and loads it into th al value (1, 2, 4, 8 <u>Auxiliary Graphic</u>
Implied Operands	Address C0000150h	Name PSIZE	<b>Description</b> a Pixel Size (*	nd Elements (1 1,2,4,8,16,32)	Bits)
Machine States	2				
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>				
Examples	<u>Code</u> GETPS A0 (PSIZE = 8)	<u>Before</u> A0 00000035	<u>After</u> A0 00000008		

## GETST Get Status Register into Register

Syntax	GETST Rd															
Execution	$ST \rightarrow Rd$															
Instruction Words	15 0	14 0	13 0	12 0	11 0	10 0	9 0	8	7	6 0	5 0	4 R	3	2 R	 d	0
Description	GET	STco	opies	the	conte	ents c	of the	statu	us reę	gister	into	the c	lestir	natio	n regi	ister.
31 30 29 28	26 2	5	22	22	1	*******		******		11	10-	6	5		4—C	)



For more information, refer to Section 4.1, The Status Register, on page 4-2.

Machine States	1		
Status Bits	N Unaffect	ed	
	C Unaffect	ed	
	Z Unaffect	ed	
	V Unaffect	ed	
Examples	Code	<b>Before</b>	<u>After</u>
		ST	A1
	GETST A1	20200010h	20200010h
	GETST A1	0000010h	00000010h

Syntax	IDLE											
Execution	Halt execution until interrupted											
Instruction Words	15     14     13     12     11     10     9     8     7     6     5     4     3     2     1     0       0     0     0     0     0     0     0     1     0     0     0     0     0	]										
Description	IDLE waits for an interrupt to occur. While IDLE waits for an interrupt, an inter- nal microcode loop executes, and the PC continually points to the IDLE instruc- tion. When the TMS34020 takes an interrupt, the PC value that is pushed onto the system stack points to the instruction that immediately follows the IDLE instruction. Upon return from the interrupt, the PC is restored and execution begins at the instruction immediately following the IDLE instruction. An interrupt request that is not enabled is ignored by the IDLE instruction.											
Machine States	Refer to Section 15.1 on page 15-2.											
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>											
Syntax	INC Rd											
-------------------	--------------------------------------------------------------------------------------------	----------------------------------------------------------------------------------	---------------------------------------------------------	---------------------------------------------	-----------------------------------------------	----------------------------------------						
Execution	$Rd + 1 \rightarrow$	Rd										
Instruction Words	15 14	<u>13 12 11 1</u>	0 9 8	7 6	5 4 1 B	3 2 1 0 Bd						
Description	INC adds 1 the destina 1, Rd.	I to the contents ation register. T	s of the desti his instruction	ination reg on is an a	jister and Iternate r	stores the result in nnemonic for ADDK						
	You can ac with the AL	complish multip DDC instruction	ole-precision	arithmetic	by using	INC in conjunction						
Machine States	1											
Status Bits	<ul> <li>N 1 if the</li> <li>C 1 if the</li> <li>Z 1 if the</li> <li>V 1 if the</li> </ul>	result is negati re is a carry, 0 result is 0, 0 of re is an overflo	ive, 0 otherw otherwise therwise w, 0 otherwis	vise se								
Examples	Code INC A1 INC A1 INC A1 INC A1 INC A1	<u>Before</u> A1 00000000h 000000Fh FFFFFFFh FFFFFFFh 7FFFFFFh	After A1 000000011 0000000000000000000000000	NC 1 00 1 00 1 01 Th 10 1 10	<b>2 V</b> 0 0 0 0 1 0 0 0 0 1							

.

Syntax	JAcondition Address				
Execution	If condition <i>true</i> , then Address $\rightarrow$ PC If condition <i>false</i> , then go to next instruction				
Instruction Words	15 14 13 12 11 10 9 8 7	<u>6 5 4 3 2 1 0</u>			
	1 1 0 0 code 1				
	16 LSBs of Addre	55 55			
Fields	code is a 4-bit digit that identifies the cond	lition for the jump within the			
	opcode. (See the condition codes ta	ole below.)			
Description	<ul> <li>The JAcondition instruction conditionally jumps to an absolute address condition is part of a mnemonic that represents the condition for the jump example, if condition is UC, then the instruction is JAUC. (See the commemonics and codes listed below.) If the specified condition is tr TMS34020 jumps to the address and continues execution from that point specified condition is false, the TMS34020 skips the jump and continue cution at the next sequential instruction. Note that the 4 LSBs of the p counter are hardwired to 0.</li> <li>The Address operand in the syntax represents the 32-bit absolute and Note that the second and third instruction words contain the address jump.</li> <li>The JAcondition instructions are usually used in conjunction with the Cl CMPI instructions. The JAV and JANV instructions can also be used to the comment of the table of table of</li></ul>				
	window violations or CPW status.	tions can also be used to detect			
Machine States	window violations or CPW status.	tions can also be used to detect			
Machine States Status Bits	<ul> <li>window violations or CPW status.</li> <li>3 if no jump, else 4</li> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> <li>V Unaffected</li> </ul>	tions can also be used to detect			
Machine States Status Bits Examples <u>Code</u>	<ul> <li>window violations or CPW status.</li> <li>3 if no jump, else 4</li> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> <li>V Unaffected</li> <li>Flags for Branch</li> <li>Code</li> </ul>	Flags for Branch			
Machine States Status Bits Examples Code JAUC HERE	<ul> <li>window violations or CPW status.</li> <li>3 if no jump, else 4</li> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> <li>Flags for Branch NCZV NCZV NCZV x x x x</li> <li>AV HER</li> </ul>	tions can also be used to detect <u>Flags for Branch</u> NCZV NCZV NCZV E x x x 1			
Machine States Status Bits Examples Code JAUC HERE JAP HERE	<ul> <li>window violations or CPW status.</li> <li>3 if no jump, else 4</li> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> <li>V Unaffected</li> <li>Flags for Branch NCZV NCZV</li> <li>NCZV NCZV</li> <li>XXXX</li> <li>JAV HEI 0 X 0 X</li> <li>JANZ HEI</li> </ul>	Elags for Branch         NCZV       NCZV         NCZV       NCZV         NE       x x x 1         RE       x X 0 x			
Machine States Status Bits Examples Code JAUC HERE JAP HERE JALS HERE	<ul> <li>window violations or CPW status.</li> <li>3 if no jump, else 4</li> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> <li>V Unaffected</li> <li>Flags for Branch NCZV NCZV NCZV</li> <li>x x x x</li> <li>X AV HER 0 x 0 x</li> <li>X 1 x x 1 x x</li> <li>X JANN HE</li> </ul>	Flags for Branch         NCZV       NCZV         NCZ × 1         RE       x × 0 x         RE       0 x x x			
Machine States Status Bits Examples Code JAUC HERE JAP HERE JALS HERE JAHI HERE	<pre>window violations or CPW status. 3 if no jump, else 4 N Unaffected C Unaffected Z Unaffected V Unaffected Flags for Branch NCZV NCZV XXXX JAV HER 0X0X JANZ HE XX1X X1XX JANN HE X00X JANV HE</pre>	tions can also be used to detect $\frac{Flags for Branch}{NCZV NCZV NCZV}$ $E x x x 1$ $RE x x 0 x$ $RE 0 x x x$ $RE x x 0$			
Machine States Status Bits Examples Code JAUC HERE JAP HERE JALS HERE JALS HERE JALT HERE	<pre>window violations. The ovv and ovt value over the ovv and over value over the ovv and over value over val</pre>	Flags for Branch         NCZV       NCZV         NCZV       NCZV         NCZV       NCZV         RE       x x 0 x         RE       x x x 0         RE       1 x x         RE       x 1 x x			
Machine States Status Bits Examples Code JAUC HERE JAP HERE JALS HERE JALT HERE JAGE HERE	<pre>window violations. The overall overvents to the overall overvents over the overall overvents to the overall overvents to the visit overvents to the visit overvents over the overvent overvents over the visit overvent overven</pre>	tions can also be used to detect Flags for Branch      NCZV NCZV NCZV      X × 1      RE × X 0 X      RE 0 X X X      RE X X 0      RE 1 X X      RE X 1 X X      RE X 1 X X      RE X 0 X Y      RE X 1 X X      RE X 0 X Y      RE X 1 X X      RE X 0 X Y      RE X 1 X X      RE X 0 X Y      RE X 1 X X      RE X 0 X Y      RE X 1 X X      RE X 0 X Y      RE X 1 X X      RE X 1 X X X      RE X 1 X X X      RE X 1 X X			
Machine States Status Bits Examples Code JAUC HERE JAP HERE JALS HERE JALT HERE JALT HERE JALE HERE JALE HERE	<pre>window violations. The overall overvents take window violations or CPW status.</pre> 3 if no jump, else 4 N Unaffected C Unaffected V Unaffected Flags for Branch NCZV NCZV XXXX JAV HEL 0X0X JANZ HE X1XX JANN HE X00X JANZ HE 0XX1 1XX0 JAN HE 0XX1 1XX0 JAN HE 0XX1 1XX0 JAN HE 0XX1 1XX0 JAN HE 0XX1 JAN	Flags for BranchNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZVNCZV <th <="" colspan="2" th=""></th>			
Machine States Status Bits Examples Code JAUC HERE JAP HERE JALS HERE JALI HERE JALT HERE JALE HERE JALE HERE JAC HERE	window violations or CPW status. 3 if no jump, else 4 N Unaffected C Unaffected V Unaffected V Unaffected Flags for Branch NCZV $x \times x \times x$ $x \times x \times x$ $x \times 1 \times x$ $x \times x$ x	tions can also be used to detect			
Machine States Status Bits Examples Code JAUC HERE JAP HERE JALS HERE JALT HERE JALT HERE JAGE HERE JAGT HERE JAC HERE	window violations or CPW status.         3 if no jump, else 4         N       Unaffected         C       Unaffected         Z       Unaffected         V       Unaffected         Flags for Branch       Code         NCZV       NCZV         X × X ×       JAV         0 × 0 ×       JANZ         × × 1 ×       JANN         × 0 0 ×       JANV         0 × x 1       1 × x 0         0 × x 1       1 × x 0         0 × x 1       1 × x 0         0 × x 1       1 × x 0         0 × x 1       1 × x 0         0 × x 1       1 × x 0         0 × x 1       JANB         0 × x 0       1 × x 1         JANN       HE         0 × x 1       JANB         0 × x 1       JANB         0 × 0 0       1 × 0         x 1 × x       JANB         x 1 × x       JANS         x 1 × x       JANS         x 1 × x       JANS	tions can also be used to detect			

Note that the TMS34020 jumps when any one or more of the *Flags for Branch* listed above are set as indicated.

## **Condition Codes**

	Mnemon	ic			
	Non XY	XY	Result of Compare	Status Bits	Code
Unconditional Compares	JAUC	_	Unconditional	Don't care	0000
Unsigned Compares	JALO (JAC) (JAB)	 JAYN	Dst lower than Src	С	1000
	JALS	JAYLE	Dst lower or same as Src	C + Z	0010
	JAHI	JAYGT	Dst higher than Src	Ē·Ī	0011
	JAHS (JANC) (JANB)	JAYNN	Dst higher or same as Src Dst = Src	Ē	1001
	JAEQ (JAZ)	 JAYZ	Dst ≠ Src	Z	1010
	JANE (JANZ)	 JAYNZ		Z	1011
Signed	JALT	JAXLE	Dst < Src	$(N \cdot V) + (\overline{N} \ \overline{V})$	0100
Compares	JALE		Dst ≤ Src	$(N \cdot \overline{V}) + (\overline{N} \cdot V) + Z$	0110
	JAGT		Dst > Src	$(N\cdotV\cdot\overline{Z})+(\overline{N}\cdot\overline{V}\cdot\overline{Z})$	0111
	JAGE	JAXGT	Dst ≥ Src	$(N \cdot V) + (\overline{N} \ \overline{V})$	0101
	JAEQ (JAZ)	 JAYZ	Dst = Src	Z	1010
	JANE (JANZ)	 JAYNZ	Dst ≠ Src	Z	1011
Compare to	JAZ (JAEQ)	JAYZ	Result = 0	Z	1010
Zero	JANZ (JANE)	JAYNZ	Result ≠ 0	Z	1011
	JAP		Result is positive	$\overline{N} \cdot \overline{Z}$	0001
	JAN	JAXZ	Result is negative	N	1110
	JANN	JAXNZ	Result is nonnegative	N	1111
General	JAZ (JAEQ)	JAYZ	Result is 0	Z	1010
Arithmetic	JANZ (JANE)	JAYNZ	Result ≠ 0	Z	1011
	JAC (JALO) (JAB)	JAYN	Carry set on result	С	1000
	JANC (JAHS) (JANB)	JAYNN	No carry on result	С	1001
	JAB (JALO) (JAC)	JAYN	Borrow set on result	C	1000
	JANB (JAHS) (JANC)	JAYNN	No borrow on result		1001
	JAV †	JAXN	Overflow on result	V	1100

Note: A mnemonic code in parentheses is an alternate code for the preceding code.

Key: † Also used for window clipping

+ Logical OR

· Logical AND

Logical NOT

Syntax		JR	condition	Address							
Execution		lf c lf c	ondition <i>tr</i> ondition <i>fa</i>	<i>ue,</i> then of a <i>lse,</i> then g	fset + PC' - jo to next in	→ PC structior	ı				
Instruction Wo	rds	15 1	14 13 1 0	12 11 0	10 9 code	8 7	65	4 3 offset	2	1	0
Fields		cod	de is a 4 opco	-bit digit th de. (See th	nat identifies ne condition	s the cor I codes t	ndition fo able belo	or the jum ow.)	p withi	n the	;
Description		JR Col exa coc tior nex exe jum	<i>condition</i> c <i>ndition</i> is p imple, if <i>co</i> les listed b a. The asse t instruction cution fro ip and con	onditionall part of a mi pondition is elow.) If the embler cale on (PC') to t m this poin atinues exe	y jumps to a nemonic; it i <u>UC</u> , the inst condition is culates the r the signed w nt. If the con ecution at th	in address represer truction i s <b>true,</b> th new add vord offse ndition is e next se	es that is nts the co s <u>JRUC</u> e TMS3 ress by a et. The T s <b>false,</b> equentia	relative to ondition fo (See the 4020 jump adding the MS34020 the TMS3 I instructi	o the cu or the j e condi os to a e addre 0 then o 34020 on.	rren ump tions new l ess o contil skip:	t PC. For and loca- of the nues s the
		The <i>Address</i> operand is a 32-bit relative address. The assembler calculate the offset as $(Address - PC')/16$ (where PC' is the address of the instructio word immediately following the jump instruction) and inserts the resulting 8-b offset into the opcode. The range for this form of the JR <i>condition</i> instructio is ±128 words (excluding 0)						lates ction 8-bit ction			
		If the offset is outside the range of $\pm 128$ words, the assembler automatica substitutes the longer form of the JR <i>condition</i> instruction. If the offset is 0, the assembler substitutes a NOP. The assembler does not accept an address the is externally defined or an address that is relative to a different section than t PC. Note that the 4 LSBs of the PC are always 0 (word aligned)					cally ), the s that n the				
		The JR <i>condition</i> instructions are often used with the CMP and CMPI instructions. The JRV and JRNV instructions can also be used to detect window violations or CPW status					truc- viola-				
Machine States	,	1 if	no jump, e	else 2							
Status Bits		N C Z V	Unaffecte Unaffecte Unaffecte Unaffecte	ed ed ed							
Examples	<u>Code</u>		Flags for NCZV	<u>Branch</u> N C Z V	NCZV	<u>Code</u>	. <u></u>	Flags fo	r Bran NCZV	ch NC	czv
	JRUC HER JRP HER JRLS HER JRHI HER JRLT HER JRGE HER JRLE HER JRGT HER	EEEEEEE	x x x x 0 x 0 x x x 1 x x 0 0 x 0 x x 1 0 x x 0 0 x x 1 0 x x 0 0 x x 1 0 x 0 0	x 1 x x 1 x x 0 1 x x 1 1 x x 0 1 x 0 1 x 0	x x 1 x	JRC JRNC JRZ JRNZ JRV JRNV JRNV JRN	HERE HERE HERE HERE HERE HERE HERE	x 1 x x x 0 x x x x 1 x x x 0 x x x x 1 x x x 0 1 x x x 0 x x x			

Note that the TMS34020 jumps when any one or more of the *Flags for Branch* listed above are set as indicated.

#### **Condition Codes**

	Mnemor	nic			
	Non XY	XY	Result of Compare	Status Bits	Code
Unconditional Compares	JRUC	—	Unconditional	Don't care	0000
Unsigned Compares	JRLO (JRC) (JRB)	JRYN	Dst lower than Src	С	1000
	JRLS	JRYLE	Dst lower or same as Src	C + Z	0010
	JRHI	JRYGT	Dst higher than Src	Ē·Ī	0011
	JRHS (JRNC) (JRNB)	JRYNN	Dst higher or same as Src Dst = Src		1001
	JREQ (JRZ)	 JRYZ	Dst ≠ Src	Z	1010
	JRNE (JRNZ)	JRYNZ		Z	1011
Signed	JRLT	JRXLE	Dst < Src	$(N \cdot V) + (\overline{N} \ \overline{V})$	0100
Compares	JRLE	—	Dst ≤ Src	$(N \cdot \overline{V}) + (\overline{N} \cdot V) + Z$	0110
	JRGT		Dst > Src	$(N \cdot V \cdot \overline{Z}) + (\overline{N} \cdot \overline{V} \cdot \overline{Z})$	0111
	JRGE	JRXGT	Dst ≥ Src	$(N \cdot V) + (\overline{N} \ \overline{V})$	0101
	JREQ (JRZ)	JRYZ	Dst = Src	Z	1010
	JRNE (JRNZ)	 JRYNZ	Dst ≠ Src	Z	1011
Compare to	JRZ (JREQ)	JRYZ	Result = 0	Z	1010
Zero	JRNZ (JRNE)	JRYNZ	Result ≠ 0	Z	1011
	JRP	—	Result is positive	$\overline{N} \cdot \overline{Z}$	0001
	JRN	JRXZ	Result is negative	N	1110
	JRNN	JRXNZ	Result is nonnegative	N	1111
General	JRZ (JREQ)	JRYZ	Result is 0	Z	1010
Arithmetic	JRNZ (JRNE)	JRYNZ	Result ≠ 0	Z	1011
	JRC (JRLO) (JRB)	JRYN	Carry set on result	С	1000
	JRNC (JRHS) (JRNB)	JRYNN	No carry on result	С	1001
	JRB (JRLO) (JRC)	JRYN	Borrow set on result	C	1000
	JRNB (JRHS) (JRNC)	JRYNN	No borrow on result	ō	1001
	JRV †	JRXN	Overflow on result	V	1100

Note: A mnemonic code in parentheses is an alternate code for the preceding code.

Key: † Also used for window clipping

+ Logical OR

Logical AND

Logical NOT

•

Syntax	JRcondition Address
Execution	If condition <i>true</i> , then offset + PC' $\rightarrow$ PC If condition <i>false</i> , then go to next instruction
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         1       1       0       0       code       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0 <t< th=""></t<>
Fields	<b>code</b> is a 4-bit digit that identifies the condition for the jump within the opcode. (See the condition codes on page 13-138.)
Description	The JR <i>condition</i> instruction conditionally jumps to an address that is relative to the current PC. The <i>condition</i> is part of a mnemonic that represents the condition for the jump; for example, if <i>condition</i> is <u>UC</u> , then the instruction is <u>JRUC</u> . (See the condition mnemonics and codes listed in on page 13-138.) If the specified condition is <b>true</b> , the TMS34020 jumps to a new location. The assembler calculates the address of this location by adding the address of the next instruction (PC') to the signed word offset. The TMS34020 then continues execution from this point. If the specified condition is <b>false</b> , the TMS34020 skips the jump and continues execution at the next sequential instruction. The <i>Address</i> operand in the syntax represents the 32-bit relative address. The address of the instruction word immediately following the jump instruction) and inserts the resulting offset into the second instruction word of the opcode. The range for this form of the JR <i>condition</i> instruction is $-32,768$ to $+32,767$ words (excluding 0). If the offset is 0, the assembler substitutes a NOP instruction. If the address that is relative to a different section than the PC. Note that the 4 LSBs of the program counter are always 0 (word aligned).
Machine States	2 if no jump, else 3
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>

Examples	<u>Code</u>	Flags for Branch			<u>Code</u>	Flags for	Flags for Branch		
-		NCZV	NCZV	NCZV		NCZV	NCZV	NCZV	
	JRUC HERE	хххх			JRV HE	RE XXX1			
	JRP HERE	0 x 0 x			JRNZ HE	RE XXOX			
	JRLS HERE	x x 1 x	x 1 x x		JRNN HE	RE OXXX			
	JRHI HERE	x 0 0 x			JRNV HE	RE XXXO			
	JRLT HERE	0 x x 1	1 x x 0		JRN HE	RE 1xxx			
	JRGE HERE	0 x x 0	1 x x 1		JRB HE	ERE X1XX			
	JRLE HERE	0 x x 1	1 x x 0	x x 1 x	JRNB HE	RE XOXX			
	JRGT HERE	0 x 0 0	1 x 0 1		JRLO HE	RE X1XX			
	JRC HERE	x 1 x x			JRHS HE	RE XOOX	x x 1 x		
	JRNC HERE	x 0 x x			JRNE HE	RE XXOX			
	JRZ HERE	x x 1 x			JREQ HE	RE XX1X			

Note that the TMS34020 jumps when any one or more of the *Flags for Branch* listed above are set as indicated.

## Jump Indirect JUMP

Syntax	JUMP Re	6										
Execution	$Rs\toPC$											
Instruction Words	15 14 0 0	13 12 0 0	11 10 0 0	9 8 0 1	7	6	5 1	4 R	3	2 R	1 d	0
Description	JUMP jump sets the 4 L be used in	os to the a SBs of the conjuncti	address o e progran ion with th	containe n counte he GET	d in th er to 0 ( PC an	e sou word d/or E	rce r align EXGF	egist ed). <sup>-</sup> C in	er. T This i struc	he Tl nstru tions	MS34 uctior 3.	1020 1 can
Machine States	2											
Status Bits	N Unaffee C Unaffee Z Unaffee V Unaffee	cted cted cted cted										
Examples	Code JUMP A1 JUMP A1 JUMP A1	<u>Bef</u> A1 0000 5FF	ore 01EE0h 01EE5h FFFFFh	PC 0055 0055	55550h 55550h 55550h		<u>Afte</u> PC 0000 0000	er D1EE D1EE FFFF	0h 0h -0h			

Syntax	LINE {0   1}				
Execution	The two execution algorithms for the LINE instruction are explained below. These algorithms are similar, varying only in their treatment of $d=0$ .				
Instruction Words	15     14     13     12     11     10     9     8     7     6     5     4     3     2     1     0       1     1     0     1     1     1     1     Z     0     0     1     1     0     1     0				
Fields	The assembler sets bit 7 in the instruction word (the Z bit) to 0 or 1, depending on which LINE algorithm you select:				
	Z=0 selects algorithm 0 Z=1 selects algorithm 1				
Description	LINE is an alternative implementation of the FLINE instruction (page 13-121). The major differences between LINE and FLINE are the following:				
	<ul> <li>LINE is slower than FLINE.</li> <li>LINE supports windowing (FLINE does not).</li> <li>LINE requires B2 to be in XY format.</li> </ul>				
	LINE performs the inner loop of Bresenham's line-drawing algorithm. This type of line draw plots a series of points $(x_i, y_i)$ either diagonally or laterally with respect to the previous point. Movement from pixel to pixel always proceeds in a dominant lateral direction. The algorithm may or may not also increment in the direction with the smaller dimension (this produces a diagonal movement). Two XY-format registers supply the XY increment values for the two possible movements. The LINE instruction maintains a decision variable, <i>d</i> , that acts as an error term, controlling movement in either the dominant or diagonal direction. The algorithm operates in one of two modes, depending on how the condition <i>d</i> =0 is treated.				
	During LINE execution, some portion of a line $[(x_0,y_0)(x_1,y_1)]$ is drawn. The line is drawn so that the axis with the largest extent has dimension $a$ , and the axis with the least extent has dimension $b$ . Thus, $a$ is the larger (in absolute terms) of $y_1 - y_0$ or $x_1 - x_0$ , and $b$ is the smaller of the two. This means that $a \ge b \ge 0$ .				
	The LINIT instruction provides a simple method for setting up the LINE instruc- tion's implied operands. For more information, refer to the LINIT instruction (page 13-146).				
	The following values must be supplied to draw a line from $(x_0,y_0)$ to $(x_1,y_1)$ :				
	Set the value in DADDR to be the <b>linear</b> address of the first pixel in the line at $(x_0,y_0)$ .				
	Use the line endpoints to determine the major and minor dimensions ( <i>a</i> and <i>b</i> , respectively) for the line draw; then set the DYDX register to this value ( <i>b</i> : <i>a</i> ).				

- Place the signed XY increment for a movement in the diagonal (or minor) direction ( $d \ge 0$  for Z=0, d > 0 for Z=1) in the INC1 register.
- Place the signed XY increment for a movement in the dominant (or major) direction (d < 0 for Z=0,  $d \le 0$  for Z=1) in the INC2 register.
- Set the initial value of the decision variable in register B0 to 2b a.
- Set the initial count value in the COUNT register to a + 1.
- Set the COLOR1 and COLOR0 registers.
- Set the PATTERN register to the required pattern.

LINE handles the contents of PATTERN in the same way as FLINE (unlike PFILL XY). With LINE, the first pixel drawn is controlled by bit 0 of the PATTERN register.

The PATTERN register contains a 32-bit repeating line-style pattern. If bit 0 of PATTERN is 0, then the first pixel drawn by LINE is a COLOR0 pixel. If bit 0 of PATTERN is 1, then the first pixel drawn by LINE is a COLOR1 pixel. The second pixel drawn by LINE is controlled by bit 1 of B13, and so on. If the line is longer than 32 pixels, the PATTERN is reused cyclically; therefore, the 33rd pixel on the line is once again controlled by bit 0 of PATTERN. As each pixel is drawn, the contents of PATTERN are rotated right (circular shifted) by 1 bit. The LSB of the rotated pattern controls the next pixel the instruction puts out.

If PATTERN contains all 1s, the line is drawn in a solid color using the replicated pixel value contained in COLOR1; if PATTERN contains all 0s, the line is drawn in a solid color using COLOR0.

The LINE instruction may use one of two algorithms, depending on the value of Z.

### Algorithm 0 (Z=0):

While COUNT > 0 COUNT = COUNT - 1 Draw the next pixel If  $d \ge 0$  d = d + 2b - 2aPOINTER = POINTER + INC1 Else d = d + 2b; POINTER = POINTER + INC2

## Algorithm 1 (Z=1):

While COUNT > 0 COUNT = COUNT - 1 Draw the next pixel If d > 0 d = d + 2b - 2aPOINTER = POINTER + INC1 Else d = d + 2b; POINTER = POINTER + INC2

#### Implied Operands

Regis	ter	Name Format		Description
B0	†	SADDR	Integer	Decision variable (d)
B2	†	DADDR	XY	Starting point $(y_i, x_i)$ , usually $(y_0, x_0)$
B3	‡	DPTCH	Linear	Destination pitch
B4		OFFSET	Linear	Screen origin (0,0)
B5		WSTART	XY	Window starting corner
В		WEND	XY	Window ending corner
B7		DYDX	XY	(b: a) = Minor: major dimension
B8		COLOR0	Pixel	COLOR0
B9		COLOR1	Pixel	COLOR1
B10	†	COUNT	Integer	Loop count
B11		INC1	XY	Minor axis (diagonal) increment
B12		INC2	XY	Major axis (dominant) increment
B13		PATTERN	Pattern	Pattern register

<sup>†</sup> These registers are changed by instruction execution.

<sup>‡</sup> Required only when pitch is an arbitrary, nonpower of 2.

Address	Name	Description and Elements (Bits)
C00000B0h	CONTROL	PPOPPixel-processing operations (22 options)WWindow-clipping operationTTransparency operationTMSets transparency mode
C0000140h	CONVDP	XY-to-linear conversion (destination pitch)
C0000150h	PSIZE	Pixel size (1,2,4,8,16,32)
C0000160h	PMASK (32 bits)	Plane mask — pixel format

Due to the pipelining of memory writes, the *last* I/O register that you write to may not, in some cases, contain the desired value when you execute the LINE instruction. To ensure that this register contains the correct value for execution, you may want to follow the write to that location with an MWAIT (13-177). Refer to Section 4.5.6 on page 4-13 for a description of the potential latency of writes to I/O registers.

# **Pixel Processing**PPOP[[CONTROL]] specifies the operation to be applied to the pixel as it is<br/>written. There are 22 operations; the default case at reset is the pixel-proces-<br/>sing *replace* (S $\rightarrow$ D) operation. For more information, refer to Section 12.8,<br/>Pixel Processing, on page 12-27.

# *Window Checking* Window clipping or picking is selected by setting W[CONTROL] to the appropriate value. The WSTART and WEND registers define the window in XY-coordinate space. For more information, refer to Section 12.7, <u>Window Checking</u>, on page 12-19.

Transparency	You can enable transparency for this instruction by setting T[[CONTROL]] to 1. Select 1 of 3 transparency options by setting TM[[CONTROL]]. For more information, refer to Section 12.9, <u>Transparency</u> , on page 12-36.			
Plane Masking	The plane mask is enabled for this instruction. For more information, refer to Section 12.10, <u>Plane Masking</u> , on page 12-39.			
Interrupts	LINE may be interrupted after every pixel in the line draw except for the last pixel. Note that a LINE instruction that is aborted because of window checking options 1 or 2 does not decrement the PC before pushing it on the stack. In this case, the LINE is not resumed after returning from the interrupt service routine. For more information, refer to Section 6.6, Interrupting Graphics Instructions, on page 6-13.			
Machine States	Refer to Section 15.1 on page 15-2.			
Status Bits	<ul> <li>N Undefined</li> <li>C Undefined</li> <li>Z Undefined</li> <li>V Set depending upon window operation</li> </ul>			
Example	Refer to example for FLINE on page 13-124.			

Syntax	LINIT					
Execution	2b – a (b:a) a + 1 minor axis major axis		XY increme XY increme	ent $\rightarrow$ B11 ent $\rightarrow$ B12		
Instruction Words	15 14 0 0	13 12 11 0 0 1	10 9 1 0	8 7 6 0 0 1 0	5 4 3 2 0 1 0 1	1 0
Description	This line ini up the impl The startpo expects a CVXYL sho	tialization ir ied B-file re bint is assur linear DADI buld be exe	nstruction us gisters as re ned to be in DR, so whe cuted on D	ses the start and equired by the LII B2 and the endp In LINIT is used ADDR before ex	end points for th NE and FLINE ir point in B7. Note in conjunction v ecuting FLINE.	e line to set Instructions. that FLINE vith FLINE,
	The V bit in window. Th between th zontal and that the line	n status is s le N and Z b le two point vertical line e may be tr	et to indica bits are set o s. This allov s as well as ivially reject	te if both start an on the X and Y zen vs for detection c single pixel lines ted.	nd end points lie ro detects on the of the special ca s. The C bit is set	e within the difference ses of hori- to indicate
	For additio	nal informa	tion, refer t	to Section 12.4,	Line Instruction	s, on page
	<u>Window C</u> 12.7.5.2, <u>U</u>	hecking fo	r Line Instr and FLINE for	ructions, on page or Preclipping Lir	3-142, subsectinge 12-23, and the Drawing, on p	subsection age 12-26.
Implied Operands	Window         C           12.7.5.2, U         Register	hecking fo sing LINIT a	Format	nuctions, on page or Preclipping Lir Description	ge 12-23, and the Drawing, on p	subsection bage 12-26.
Implied Operands	I2-7; FLIN           Window C           12.7.5.2, U           Register           B0	he on page hecking fo sing LINIT a Name SADDR	Format	<u>ructions</u> , on page or Preclipping Lir Description	ge 12-23, and the Drawing, on p	on 12.7.5, subsection age 12-26. (output)
Implied Operands	I2-7; FLIN           Window C           12.7.5.2, U           Register           B0           B2	Name SADDR	Format Linear XY	Description Description Decision variable Starting point	9 12-23, and <u>ne Drawing</u> , on p (y <sub>0</sub> , x <sub>0</sub> )	ion 12.7.5, subsection page 12-26. (output) (input)
Implied Operands	Register           B0           B2           B7	Name SADDR DADDR DYDX	Format Linear XY	Description Decision variable Starting point	$x_{142}$ , subsection $y_{12} = 12-23$ , and $x_{12} = 0$ , and $y_{12} = 0$ , $y_{10}$ , $x_{10}$ $(y_{10}, x_{10})$	ion 12.7.5, subsection page 12-26. (output) (input) (input)
Implied Operands	I2-7; FLIN           Window C           12.7.5.2, U           Register           B0           B2           B7	Name SADDR DADDR DYDX DYDX	Format Linear XY XY XY	Description Decision variable Starting point Ending point b:a <i>minor:major</i>	$(y_0, x_0)$ $(y_1, x_1)$	(output) (input) (output) (input) (output)
Implied Operands	Register           B0           B2           B7           B10	Name SADDR DADDR DYDX DYDX COUNT	Format Linear XY XY XY XY Integer	Description Decision variable Starting point Ending point b:a <i>minor:major</i> Count	$(y_0, x_0)$ line dimensions	ion 12.7.5, subsection page 12-26. (output) (input) (input) (output) (output)
Implied Operands	I2-7; FLIN           Window C           12.7.5.2, U           Register           B0           B2           B7           B10           B11	Name SADDR DADDR DYDX COUNT	Format Linear XY XY XY Integer	INE on page ructions, on page or Preclipping Lir         Description         Decision variable         Starting point         Ending point         b:a minor:major         Count         Minor axis (diago	$(y_0, x_0)$ $(y_1, x_1)$ (yon al increment)	(output) (input) (input) (output) (output) (output) (output)
Implied Operands	B0         B2           B7         B10           B11         B12	Name SADDR DADDR DYDX DYDX COUNT	Format Linear XY XY XY Integer XY XY	Description Decision variable Starting point Ending point b:a <i>minor:major</i> Count Minor axis (diago	$(y_0, x_0)$ $(y_1, x_1)$ line dimensions phant increment) (y_1, x_1)	(output) (input) (input) (output) (output) (output) (output) (output)
Implied Operands Machine States	I2-7; FLIN           Window C           12.7.5.2, U           Register           B0           B2           B7           B10           B11           B12           9	Name SADDR DADDR DYDX DYDX COUNT	Format Linear XY XY XY XY Integer XY XY	INE on page 1         ructions, on page 1         or Preclipping Lir         Description         Decision variable         Starting point         Ending point         b:a minor:major         Count         Minor axis (diago         Major axis (domi	$x_{1}$ ( $y_{2}$ , subsection $y_{2}$ ( $z_{2}$ , and $y_{2}$ , $z_{3}$ , and $y_{1}$ , $z_{1}$ , $y_{1}$ , $z_{1}$ ) $y_{1}$ , $z_{1}$ ) $y_{1}$ , $z_{1}$ ) $y_{1}$ , $z_{1}$ ) $y_{2}$ , $z_{1}$ , $z_{2}$ , $z_{2}$ , $z_{1}$ , $z_{2}$ , $z_{2}$ , $z_{2}$ , $z_{1}$ , $z_{2}$	(output) (input) (input) (output) (output) (output) (output) (output)
Implied Operands Machine States Status Bits	I2-7; FLIN         Window C         12.7.5.2, U         Register         B0         B2         B7         B10         B11         B12         9         N = 1 if x         C = 1 if (         Z = 1 if y	Name Name SADDR DADDR DYDX DYDX COUNT (0 = x1 (ver) CPW(y <sub>0</sub> , x <sub>0</sub> ) $(0 = y_1 (hor)$	Format Linear XY XY XY XY Integer XY XY tical line) b) & CPW(y	Description Decision variable Starting point Ending point b:a <i>minor:major</i> Count Minor axis (diago Major axis (domi	$\frac{3-142}{9}, \text{ subsect}$ $\frac{y_0, x_0}{(y_1, x_1)}$ $\frac{y_1, x_1}{y_1, x_1}$ $\frac{y_1, y_1}{y_1, y_1}$	irely ion 12.7.5, subsection page 12-26. (output) (input) (output) (output) (output)
Implied Operands Machine States Status Bits	I2-7; FLIN         Window C         12.7.5.2, U         Register         B0         B2         B7         B10         B11         B12         9         N = 1 if x         C = 1 if (         Z = 1 if y         V = 1 if (	Name Name SADDR DADDR DADDR DYDX DYDX COUNT (0 = x1 (ver) CPW( $y_0, x_0$ ) $(0 = y_1 (hor)$ $y_0, x_0) or (y)$	Format Format Linear XY XY XY Integer XY XY tical line) ) & CPW(y izontal line) 1,x1) lie out	INE on page 1         ructions, on page 1         or Preclipping Lin         Decision variable         Starting point         Ending point         b:a minor:major         Count         Minor axis (diago         Major axis (domi         1,x1)) is nonzero         side the window	$(y_0, x_0)$ ( $y_0, x_0$ ) ( $y_1, x_1$ ) ( $y_1, x_1$ ) ( $y_1, x_1$ ) ( $y_1, x_1$ ) ( $y_1, y_1$ ) ( $y_$	irely isow) itially ion 12.7.5, subsection page 12-26. (output) (input) (input) (output) (output) irely iow)
Implied Operands Machine States Status Bits Examples	Register         B0         B2         B7         B10         B11         B12         9         N = 1        if x         C = 1        if (         Z = 1        if y         V = 1        if (         Refer to S	Name SADDR DADDR DADDR DYDX DYDX COUNT (0 = x1 (ver) COUNT $(0 = y_1 (hor)$ $y_0, x_0) or (y)$ ection 15.1	Format Linear XY XY XY XY XY Integer XY XY tical line) b) & CPW(y izontal line) 1,x1) lie out on page 1	INE on page 1         ructions, on page 1         or Preclipping Lir         Description         Decision variable         Starting point         Ending point         b:a minor:major         Count         Minor axis (diago         1,x1)) is nonzero         side the window         5-2.	$\frac{3-142}{9}, \text{ subsect}$ $\frac{3-142}{9}, \text{ and}$ $\frac{3-142}{9}, \text{ and}$ $\frac{3-142}{9}, \text{ and}$ $\frac{3-142}{9}, \text{ and}$ $\frac{12-23}{9}, \text{ and}$ $\frac{(y_0, x_0)}{(y_1, x_1)}$ $\frac{(y_0, y_0)}{(y_1, x_1)}$ $\frac{(y_0, y_0)}{(y_1, y_1)}$	irely ion 12.7.5, subsection page 12-26. (output) (input) (output) (output) (output) (output) irely low)

Syntax	LMO Rs, Rd									
Execution	31 – (bit numbe	er of leftmost 1 ir	$r Rs) \rightarrow Rd$							
Instruction Words	15 14 13 0 1 1	12 11 10 9 0 1 0	8 7 0 1 Rs	6 5 4 R	3 2 Rd	1 0				
Description	LMO locates the loads the 1s corr of the destination with 0s. Bit 31 of register contain status bit Z is s	MO locates the leftmost (most significant) 1 in the source register. It then oads the 1s complement of the <b>bit number</b> of the leftmost-1 bit into the 5 LSBs of the destination register. The 27 MSBs of the destination register are loaded with 0s. Bit 31 of Rs is the MSB (leftmost) and bit 0 is the LSB. If the source register contains all 0s, then the destination register is loaded with all 0s and status bit Z is set.								
	You can normalize the contents of the source register by following the LMO instruction with an RL Rs,Rd instruction, where Rs is the destination register of the LMO instruction and Rd is the source register.									
Machina States		a de in the same	register me.							
Machine States	I									
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z 1 if the sou</li> <li>V Unaffected</li> </ul>	rce register cont	ents are 0, <i>0</i> d	otherwise						
Examples	<b>Code</b> LMO A0,A1 LMO A0,A1 LMO A0,A1 LMO A0,A1 LMO A0,A1	Before A0 00000000h 00000001h 00000010h 08000000h 8000000h	After NCZV × × 1 × × × 0 ×	A1 00000000h 000001Fh 0000001Bh 00000004h 0000000h						

Syntax	MM	-M -	Rp, r	egist	er lis	t										
Execution	For e	each 32 bit Rp +	regis ts of 32 –	ster F data → Rp	R <i>n</i> in at th	the re e ado	egisto Iress	er lis spe	t, cified	l in R	p →	Rn				
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	1	0	0	1	1	0	1	R		F	۱p	
					b	nary r	eprese	entatio	on of t	ne reg	ister li	st				
Description	MMF	-M lo	ads t	he co	onter	nts of	a spe	ecifie	d list	of <i>ei</i>	ther	A- or	B-file	e rea	isters	; (not

MMFM loads the contents of a specified list of *either* A- or B-file registers (not both) from a block of memory.

- **D** *Rp* is a register that points to the first location in the block of memory.
- The *register list* is a list of registers separated by commas (such as A0, A1, A9). These are the registers that MMFM loads new values into.

MMFM and MMTM are complementary instructions. MMFM reads a list of Aor B-file registers *from* memory, and MMTM writes a list of A- or B-file registers *to* memory. These instructions can be used to save and restore the contents of registers during, for example, subroutine calls and interrupts. All 32 bits of each register in the list are saved and then restored.

MMFM and MMTM use Rp as a pointer register. Rp acts as a stack pointer; MMTM pushes a list of registers onto a stack, and MMFM pops a list of registers from a stack. The stack grows toward lower addresses, similar to the way the SP register points to the system stack. The Rp can be any register that is not included in the register list and that is in the same file as the registers in the list. (SP can be treated as belonging to either register file.)

MMFM and MMTM always leave the Rp register adjusted to point to the new top of the stack following a push or pop operation. MMTM predecrements Rp by 32 prior to pushing each register in the list onto the stack. The last register pushed on the stack by MMTM is the highest numbered register in the list. MMFM postincrements Rp by 32 after popping each register in the list from the stack. The first register popped off the stack by MMFM is the highest numbered register numbered register in the list.

If SP is used as the Rp register, MMTM and MMFM push and pop register values to and from the system stack and leave SP correctly adjusted to point to the new top of the system stack.

Rp and the registers in the list must all be in the same register file. The assembler allows the registers in the list to be specified in any order; the highest numbered register is always restored first (that is, the value at the top of the stack—the lowest address in the stack—is loaded into the highest numbered register). Don't include Rp as one of the registers in the register list, because this produces unpredictable results. For the best performance, the original

contents of Rp should be aligned on a long-word boundary; the alignment of Rp affects the instruction timing as indicated in **Machine States**, below.

The second word of the MMFM instruction is a binary-mask representation of the registers in the list. The R bit (bit 4) in the first instruction word indicates which register file is affected; the bits that are set to 1 in the mask indicate which registers are restored. The bit assignments in the mask are

	SP	A14	A13	A12	A11	A10	A9	A8	A7	A6	A5	A4	A3	A2	A1	A0
or	SP	B14	B13	B12	B11	B10	<b>B</b> 9	B8	B7	<b>B</b> 6	B5	B4	B3	B2	B1	В0

#### (MSB) 15

Machine States

#### machine Otale

Status Bits

- N Unaffected
- C Unaffected
- Z Unaffected
- V Unaffected

Examples

This example restores several B-file registers:

Refer to Section 15.1 on page 15-2.

MMFM B0,B1,B2,B3,B7,B12,B13,B14,SP

This instruction uses register B0 as the stack pointer. Assume that B0 = 00010000h; this is the address of the top of the stack. MMFM moves the data at this location into the LSW of the SP (which is the highest order register listed in this example). Assume that memory contains the following values before instruction execution:

Address	Data	Address	Data
000100F0h	1111h	00010070h	CCCCh
000100E0h	B1B1h	00010060h	BCBCh
000100D0h	2222h	00010050h	DDDDh
000100C0h	0B2B2h	00010040h	BDBDh
000100B0h	3333h	00010030h	EEEEh
000100A0h	B3B3h	00010020h	BEBEh
00010090h	7777h	00010010h	FFFFh
00010080h	B7B7h	00010000h	BFBFh

After the MMFM instruction is executed, the registers in the list have the following values:

B0 = 00010100h	B12 = CCCCBCBCh
B1 = 1111B1B1h	B13 = DDDDBDBDh
B2 = 2222B2B2h	B14 = EEEEBEBEh
B4 = 3333B3B3h	SP = FFFFBFBFh
B8 = 77778787h	

The other B-file registers (which weren't specified in the register list) are not affected by this instruction. Note that B0 now contains the value 10100h; the last part of the data that was restored was for B1, and B0 points to the word past that data.

0 (LSB)

Syntax	MMTM Rp, register list											
Execution	For each register $Rn$ in the register list, $Rp - 32 \rightarrow Rp$ 32 bits of data at the address specified in $Rn \rightarrow Rp$											
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0											
	0         0         0         1         0         0         R         Rp.           binary representation of the register list											
Description	MMTM stores the contents of a specified list of <i>either</i> A- or B-file registers (not both) in memory.											
	$\square$ <i>Rp</i> is a register that points to the first location in a block of memory.											
	The <i>register list</i> is a list of registers that are separated by commas (such as A0, A1, A9). These are the registers that MMTM stores in memory.											
	MMTM and MMFM are complementary instructions. MMFM reads a list of A- or B-file registers <i>from</i> memory, and MMTM writes a list of A- or B-file registers <i>to</i> memory. These instructions can be used to save and restore the contents of registers during, for example, subroutine calls and interrupts. All 32 bits of each register in the list are saved and then restored.											
	MMTM and MMFM use <i>Rp</i> as a pointer register. Rp acts as a stack pointer; MMTM pushes a list of registers onto a stack, and MMFM pops a list of registers from a stack. The stack grows toward lower addresses, similar to the way the SP register points to the system stack. The Rp can be any register that is not included in the register list and that is in the same file as the registers in the list. (SP can be treated as belonging to either register file.)											
	MMTM and MMFM always leave the Rp register adjusted to point to the new top of the stack following a push or pop operation. MMTM predecrements Rp by 32 prior to pushing each register in the list onto the stack. The last register pushed on the stack by MMTM is the highest numbered register in the list. MMFM postincrements Rp by 32 after popping each register in the list from the stack. The first register popped off the stack by MMFM is the highest numbered register in the list.											
	If SP is used as the Rp register, MMTM and MMFM push and pop register val- ues to and from the system stack and leave SP correctly adjusted to point to the new top of the system stack.											
	When MMTM execution is complete, the contents of the lowest order register in the list reside at the highest address in the memory stack, and Rp points to the address of the highest order register in the list.											
	Rp and the registers in the list must all be in the same register file. The assembler allows the registers in the list to be specified in any order; the lowest order register is always saved first. Don't include Rp as one of the registers in the register list, because this produces unpredictable results. For the best performance, the original contents of Rp should be aligned on a long-word											

boundary; the alignment of Rp affects the instruction timing as shown in **Machine States**, below.

The second word of the MMTM instruction is a binary-mask representation of the registers in the list. The R bit (bit 4) in the first instruction word indicates which register file is affected; the bits that are set to 1 in the mask indicate which registers are restored. The bit assignments in the mask are

	A0	A1	A2	A3	A4	A5	A6	A7	A8	A9	A10	A11	A12	A13	A14	SP
or	<b>B</b> 0	B1	B2	<b>B</b> 3	B4	<b>B</b> 5	<b>B</b> 6	B7	<b>B8</b>	B9	B10	B11	B12	B13	B14	SP

### (MSB) 15

Machine States

Refer to Section 15.1 on page 15-2.

Status Bits

N Set to the sign of the result of 0 – Rp. (This value is typically 1 if the original contents of Rp are positive; otherwise, it is 0. The only exceptions to this are when Rp=80000000h and N is set to 0, and when Rp=0 and N is set to 1.)

- **C** Unaffected
- Z Unaffected
- V Unaffected

Examples

This example saves the values of several A-file registers in memory:

MMTM A1,A0,A2,A4,A8,A12,A13,A14,SP

This instruction uses register A1 as the stack pointer. Assume that A1 = 00100000h before instruction execution; this value is decremented by 32 to point to the address where the contents of A0 (the lowest order register in the list) are stored. Assume that the registers in the list contain the following values before instruction execution:

A0 = 0000A0A0h	A12 = CCCCACACh
A2 = 2222A2A2h	A13 = DDDDADADh
A4 = 4444A4A4h	A14 = EEEEAEAEh
A8 = 8888A8A8h	SP = FFFFAFAFh

MMTM saves these register values in memory as shown below:

Address	Data	Address	Data
000FFF00h	AFAFh	000FFF80h	A8A8h
000FFF10h	FFFFh	000FFF90h	8888h
000FFF20h	AEAEh	000FFFA0h	A4A4h
000FFF30h	EEEEh	000FFFB0h	4444h
000FFF40h	ADADh	000FFFC0h	A2A2h
000FFF50h	DDDDh	000FFFD0h	2222h
000FFF60h	ACACh	000FFFE0h	A0A0h
000FFF70h	CCCCh	000FFFF0h	0000h

After instruction execution, register A1 = 000FFF00h; this is the address of the last portion of register data that is saved.

0 (LSE

# MODS Modulus, Signed

Syntax	MODS Rs, R	d								
Execution	Rd mod Rs $\rightarrow$	Rd								
Instruction Words	15 14 13	12 11 10	) 9	87	65	4	3	2	1	0
	0 1 1	0 1	0	F	ls	R		R	d	
Description	MODS perform register by the 3 der in the destinegative, the n for a positive di dend. The origi	ns a 32-bit sig 32-bit divisor nation regis nagnitude of vidend and nal contents	gned divi r in the so ter. Rega the rem divisor. T of the de	de of the ource reg ardless o ainder is he rema estinatio	e 32-bit gister, a of wheth s alway ainder is on regist	divider nd retu ner the s the s the sa er are	nd in rns a resu ame me s alwa	the d 32-b It is p as it sign a ys ov	estina bit ren bositiv woul s the rerwri	ation nain- /e or d be divi- itten.
	Rs and Rd mu	st be in the s	same reg	ister file	9.					
Machine States	40 41 if result = 80 3 if Rs = 0	000000								
Status Bits	<ul> <li>N 0 if Rs is 0 1 if Rs is n 0 if Rs is n</li> <li>C Unaffected</li> <li>Z 0 if Rs is 0 1 if Rs is n 0 if Rs is n</li> <li>V If Rs is 0, t</li> </ul>	ot <i>0</i> and the ot <i>0</i> and the ot <i>0</i> and the ot <i>0</i> and the hen V = 1, c	result in result in result in result in therwise	Rd is – Rd is + Rd is C Rd is r V = 0	-ve + ve ) not <i>0</i>					
Examples	Code MODS A0,A1 MODS A0,A1	Before           A0           00000000h           00000000h           00000000h           00000004h           FFFFFFC           FFFFFFC           FFFFFFC           FFFFFFC           FFFFFFC           FFFFFFC           FFFFFFC           FFFFFFC           FFFFFFC	A1 000 FF 000 000 FF FF 000 h 000 h 000 h 000 h FF	000000h 000007h FFFF9P 000008h 000007h 000000h FFFF8P 000008h 000007h 000000h FFFF9P FFFF8P	A 0 0 0 0 0 1 1 0 0 0 0 1 1 0 0	fter           C Z V           × 0 1           × 0 1           × 0 1           × 0 1           × 1 0           × 1 0           × 1 0           × 1 0           × 1 0           × 1 0           × 1 0           × 1 0           × 1 0           × 1 0           × 1 0           × 1 0           × 1 0           × 1 0           × 1 0           × 1 0           × 1 0	A 00 00 FF 00 00 FF 00 00 00 00 00 00 00	1 00000 FFFF 00000 00000 FFFF 00000 00000 00000 FFFF	00h 07h FF9h 00h 03h 00h FFDh 00h 00h FFDh 00h	

Syntax	MODU Rs, Ra	,					
Execution	Rd mod Rs $\rightarrow$	Rd					
Instruction Words	15 14 13 0 1 1	12 11 10 0 1 1	9 8 7 1 Rs	65 s	4 3 R	2 1 Rd	0
Description	MODU performs tion register by remainder in the register are alw	s a 32-bit uns the 32-bit div e destination ays overwritt	igned divide of t /isor in the sour register. The or en.	the 32-bit ( rce registe iginal cont	dividend er, and r tents of	l in the de eturns a the desti	estina- 32-bit nation
	Rs and Rd mus	t be in the sa	me register file	•			
Machine States	35 3 if Rs = 0						
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z 0 if Rs=0, 1</li> <li>V 1 if divisor F</li> </ul>	if quotient is Rs equals <i>0</i> , (	0, <i>0</i> otherwise ) otherwise				
Examples	Code MODU A0,A1 MODU A0,A1 MODU A0,A1 MODU A0,A1 MODU A0,A1 MODU A0,A1	Before A0 00000000h 0000000h 0000000h 00000004h 00000004h	A1 00000000h 0000007h FFFFFF9h 00000008h 00000007h 0000000h	After N C Z V × × 0 1 × × 0 1 × × 0 1 × × 1 0 × × 0 0 × × 1 0	A1 0000 0000 FFFF 0000 0000 0000	0000h 0007h FFF9h 0000h 0003h 0000h	

\*\*\*\*

- **MOVB Instructions** The **MOVB** instruction is a special form of the MOVE instruction that restricts the field size of the move to 8 bits. MOVB moves a single byte from its source to a specified destination. The following list describes characteristics common to all **MOVB** instructions.
  - MOVB instructions move data from a register to memory, from memory to a register, and between memory locations, but they do not move data between registers.
  - A byte can begin on any bit boundary in memory, although sequential byte moves are more efficient if the byte addresses are aligned on even 8-bit boundaries.
  - All addresses are bit addresses.
  - When a byte is moved into a register, the byte's LSB coincides with the register's LSB; the byte is sign-extended into the 24 MSBs of the register.
  - If the source data is in a register, only the LSbyte is used.
  - Rs and Rd must be in the same register file.
  - The status bits are unaffected unless otherwise noted in the individual descriptions.
  - For machine states information, refer to Section 15.2 on page 15-10.

## Table 13–4. Summary of Operand Formats for the MOVB Instruction

				Destinati	on
		Ro	d *Ra	*Rd(DOff	set) @DAddress
đ	Rs		$\checkmark$	$\checkmark$	$\checkmark$
nrce	*Rs	$\checkmark$	√		
So	*Rs(S	Offset) √	· · ·	· 🗸	
	@SAc	ldress √			$\checkmark$

The MOVB instruction has nine operand combinations, which are listed below with their corresponding instruction words and descriptions.

MOVB Rs, \*Rd

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	0	0	Ō	1	1	0		F	Rs		R		R	d	

Moves the LSbyte of Rs to the memory address contained in the Rd.

TMS34020 Assembly Language Instruction Set

MOVB Rs, \*Rd(offset) -----

_	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	1	0	1	0	1	1	0		F	7s		R		F	ld	
ſ								off	set							

Moves the LSbyte of Rs to the destination memory address. The destination address is formed by adding the signed 16-bit offset to the contents of Rd.

MOVB Rs, @DAddress ----

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
0	0	0	0	0	1	0	1	1	1	1	R		Rs		
	16 LSBs of destination address														
	16 MSBs of destination address														

Moves the LSbyte of Rs to the destination address.

MOVB \*Rs, Rd ---

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	0	0	0	1	1	1		F	₹s		R		R	d	

Moves a byte from the source address contained in Rs into Rd. This instruction also compares the source data to 0.† See **Status Bits** for more information.

MOVB \*Rs, \*Rd -

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	0	0	1	1	1	0		F	₹s		R		R	d	

Moves a byte from the source address contained in Rs to the destination address contained in Rd.

MOVB \*Rs(offset), Rd -

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	0	1	0	1	1	1		F	₹s		R		F	ld	
							off	set							

Moves a byte from the source address to the destination register. The source data's memory address is a bit address and is formed by adding the signed 16-bit offset to the contents of Rs. This instruction also compares the source data to 0. †See **Status Bits** for more information.

## MOVB \*Rs(SOffset), \*Rd(DOffset) \*

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	0	1	1	1	1	0	Rs				R		F	۱d	
	source offset														
	destination offset														

Moves a byte from the source address to the destination address. Both addresses are formed by adding the source or destination signed 16-bit offset to the contents of Rs or Rd, respectively.

## MOVB @SAddress, \*Rd



Moves a byte from the source address to Rd. This instruction also compares the source data to 0. †See **Status Bits** for more information.

#### MOVB @SAddress, @DAddress -

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
0	0	0	0	0	0	1	1	0	1	0	0	0	0	0	0
	16 LSBs of source address														
	16 MSBs of source address														
					16 L	SBs o	of des	tinatio	n addı	ess					
					16 N	1SBs	of des	tinatio	n add	ress					

Moves a byte from the source address to the destination address.

<sup>†</sup>The following status bits information applies only to MOVB instructions with these addressing modes:

MOVB \*Rs, Rd MOVB \*Rs(offset), Rd MOVB @SAddress, Rd

Status Bits

- N 1 if the sign-extended data moved is negative, 0 otherwise
- C Unaffected
- Z 1 if the sign-extended data moved is 0, 0 otherwise
- **V** 0

# MOVB Examples -

Example 1

Assume that memory contains the following values:

	Addres 1000h 1010h	S	<b>Data</b> 0000h 0000h						
	Code MOVB MOVB MOVB MOVB MOVB MOVB	Be A0,*A A0,*A A0,*A A0,*A A0,*A A0,00 A0,00	fore 1 1 1 (1) 1 (-1) 000h 000h	<b>A0</b> 89ABCDEF 89ABCDEF 89ABCDEF 89ABCDEF 89ABCDEF 89ABCDEF	<b>A</b> Th 0 Th 0 Th 0 Th 0 Th x Th x Th x	After 0001000h 0001009h 0001000h 0001001h xxxxxxxx xxxxxxx	@1000h 00EFh DE00h 01DEh 00EFh 00EFh F000h		@1010h 0000h 0001h 0000h 0000h 0000h 0000h
Example 2	Assume	e that m	emory	contains t	he fo	llowing valu	es:		
	<b>Addres</b> 1000h 1010h	S	Data 00EFh 89ABh			-			
	Code MOVB *A MOVB *A MOVB *A MOVB *A MOVB *A MOVB *A MOVB *A MOVB *1 MOVB *1 MOVB *1	A0,A1 A0,A1 A0,A1 A0,A1 A0(0),F A0(0),F A0(8),F A0(-1), .000h,F .00Ch,F	Befc A0 0000 0000 0000 1 0000 A1 0000 A10000 A10000 A1 XXXX	272 1000h 1001h 1008h 100Ch 1000h 1000h 100Dh xxxxx xxxxx	After A1 FFFI 0000 FFFI FFFI 0000 FFFI FFFI FFF	EFFEFh 20077h 20000h FFFB0h FFFEFh 20000h FFFB0h FFFEFh FFFEFh FFFB0h	NC2 1 x 0 0 x 1 1 x 0 1 x 0 1 x 0 1 x 0 1 x 0 1 x 0	2 N 0 0 1 0 1 0 1 0 1 0 1 0 1 0 1 0	
Example 3	Assume	e that m	emory	contains t	he fo	llowing valu	es:		
	Addres 1000h 1010h 2000h 2010h	S	Data CDEFt 89ABh 0000h 0000h	1					
	<u>Code</u>			Befor	e		<u>After</u>		
	MOVB MOVB MOVB MOVB MOVB MOVB MOVB MOVB	*A0,*7 *A0,*7 *A0,*7 *A0,*7 *A0,*7 *A0,*7 *A0(0) *A0(12 @1000f	A1 A1 A1 A1 A1 C,*A1(( 2),*A1 1,02000 1,02009	A0 000010 000010 000010 000010 000010 000010 000010 000010 000010 0000010 0000010 0000010 0000010 0000010 0000010 0000010 0000010	000h 000h 001h 001h 000h 000h 000h xxx xxx	A1 00002000h 0002009h 00002000h 00002000h 00002000h 00002000h 00002000h xxxxxxxx xxxxxxxx	@2000 00EFh 01DEh DE00h 00F7h 01EEh 7800h 00EFh 7800h 00EFh 7800h	n	@2010h 0000h 0000h 0001h 0000h 0001h 0000h 0001h 0000h 0001h

Syntax	MOVE Rs, Rd										
Execution	$Rs \rightarrow Rd$										
Instruction Words	15 14 13	12 11 10	9	87	6	5	4	3	2	1	0
	0 1 0	0 1 1	М	F	Rs		R		Ro	d	
Description	MOVE moves the ister. Note that The source and chip register file source and des performs an important the source and the sour	ne 32 bits of c this is not a l destination e. Note that stination regi plicit compar	lata fro field m registe this is sters to re to 0	m the sou ove; ther ers can be the only o be in d of the re	urce re refore, e any o MOVE lifferer gister	egiste the f of the E ins t file data	er to t field e 31 I truct s. Th	he de size ocati ion ti nis in	estina has i ions i hat a istruc	ation no ef in the llows ction	reg- fect. on- the also
Fields	The assembler the move is with sembler sets M it sets M to 1 if	sets bit 9 (th hin a register to 0 if the southe the registers	e M bit file or v urce ar are in	) in the ir whether i nd destina different	nstruct t cross ation r t files.	tion v ses th egist	vord ne reg ers a	to sp giste are in	)ecify r files the ເ	v whe s. The same	ther as- file;
	The assembler the registers are sets R to 1 if the	sets bit 4 (th e in. The ass e registers a	ne R bi semble re in fil	t) in the i er sets R le B.	instruc to 0 if	tion the r	worc regis	to s ters a	pecil are ir	fy the n file	∍ file A; it
	Note that when specifies the re-	M=0, R spe gister file for	cifies t the <i>sc</i>	he regist <i>urce reg</i>	er file <i>ister.</i>	for b	oth r	egist	ers; i	if M=	1, R
Machine States	1										
Status Bits	<ul> <li>N 1 if the 32-b</li> <li>C Unaffected</li> <li>Z 1 if the 32-b</li> <li>V 0</li> </ul>	bit data move bit data move	ed is no ed is 0,	egative, ( , <i>0</i> otherv	0 othe wise	rwise	9				
Examples	Code MOVE A0, A1 MOVE A0, A1 MOVE A0, A1 MOVE A0, B1 MOVE A0, B1 MOVE A0, B1	Before           A0           0000FFFFh           0000000h           FFFFFFFh           0000FFFFh           0000000h           FFFFFFFFh	<b>A</b> 00 00 Fl xx xx	fter 1 000FFFFt 000000h FFFFFF 000000h FFFFFF 000000 0000000 FFFFFF 00000000	า ่ h	B1 XXXXX XXXXX 0000 0000 FFFF	xxxh xxxh FFFF 0000 FFFF	ו ו ⁼h h Fh	NC2 0x0 0x1 1x0 0x0 0x1 1x0	<b>ZV</b> 0 0 0 0 0	

.

- **MOVE Instructions** The following list describes characteristics common to all **MOVE** instructions (*except MOVE Rs, Rd*). For information on MOVE *Rs, Rd*, refer to page 13-158.
  - □ The MOVE instruction moves a field of 1—32 bits, depending on the selected field size. The optional F parameter determines the field size and extension for the move.
    - **F=0** selects the field size of 0 (FS0).
    - **F=1** selects the field size of 1 (FS1).
    - The SETF instruction sets the field size and extension.
    - If you do not supply a value for F, MOVE uses the value of field 0.
  - The field is right-justified within the source register.
  - **G** Rs and Rd must in the same register file.
  - The status bits are unaffected unless otherwise noted in the individual descriptions.
  - For machine states information, refer to Section 15.2 on page 15-10.
  - The destination address is a bit address.

Table 13–5. Summary of Operand Formats for the MOVE Instruction

					Destina	tion	
•		Rd	*Rd	*Rd+	_*Rd	*Rd(DOffset)	@DAddress
	Rs		$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
	*Rs	$\checkmark$	$\checkmark$				
ece	*Rs+			$\checkmark$			
Sou	*Rs				$\checkmark$		
	* <i>Rs</i> (SOffset)			$\checkmark$		$\checkmark$	
	@SAddress			√		· · · · · · · · · · · · · · · · · · ·	$\checkmark$
· · · · · · · · · · · · · · · · · · ·							

The **MOVE** instruction has 18 operand combinations, which are listed below with their corresponding instruction words and descriptions.

MOVE *Rs,* \**Rd* [,*F*]



Moves a field from Rs to the address specified in Rd.

MOVE *Rs*, \**Rd*+ [,*F*] --

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	0	0	1	0	0	F		F	₹s		R		R	ld	

Moves a field from Rs to the address contained in the destination register. After the move, the contents of Rd are postincremented by the selected field size.

MOVE *Rs*, *–*\**Rd* [,*F*] *–* 

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	0	1	0	0	0	F		F	ls		R		R	d	

Moves a field from the Rs to the address contained in Rd. Before the move, the destination address is determined by subtracting the field size from the contents of Rd. (This value is also the final value for the register.)

## MOVE Rs, \*Rd(offset) [,F] -

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	0	1	1	0	0	F	Rs				R		F	۱d	
							off	set							

Moves a field from the Rs to the destination address. The destination address is formed by adding the signed 16-bit offset to the contents of Rd.

## MOVE Rs, @DAddress [,F] -



Moves a field from Rs to the destination address.

MOVE \*Rs, Rd [,F] -

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	0	0	0	0	1	щ		F	ls		R		R	d	

Moves a field from the source address contained in Rs to the destination address contained in Rd. When the field is moved into the destination register, it is right-justified and sign-extended or zero-extended to 32 bits (depending on the value of FE). This instruction also compares the source data to 0. † See **Status Bits** for more information.

MOVE \*Rs, \*Rd [,F] ----

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	0	0	0	1	0	F		F	٦s		R		F	ld	

Moves a field from a source address contained in Rs to the destination address contained in Rd.

## MOVE \*Rs+, Rd [,F] -

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	0	0	1	0	1	F		F	₹s		R		R	ld	

Moves a field from a source address into Rd. Rs contains the address of the field; after the move, the contents of the source register are incremented by the field size. When the field is moved into Rd, it is right-justified and sign- or zero-extended to 32 bits (depending on the value of the current FE bit). This instruction also performs an implicit compare to 0 of the field data. † See **Status Bits** for more information.

## MOVE \*Rs+, \*Rd+ [,F] -

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	0	0	1	1	0	F		F	٦s		R		R	ld	

Moves a field from one address to another. Rs contains the bit address of the field; Rd contains the bit address of the field's destination. After the move, the contents of both registers are incremented by the field size.

If Rs and Rd specify the same register, the data read from the location pointed to by the original contents of Rs is written to the location pointed to by the incremented value of Rs(Rd).

## MOVE \_\**Rs, Rd* [,*F*] -

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	0	1	0	0	1	F		F	٦s		R		R	d	

Moves a field from a source address into Rd. Rs contains a bit address; before the move, the contents of Rs are decremented by the field size to form the address of the field. (This value is also the final value for the register.) When the field is moved into Rd, it is right-justified and sign- or zero-extended to 32 bits (depending on the value of the current FE bit). This instruction also performs an implicit compare to 0 of the field data.

If Rs and Rd are the same register, the pointer information is overwritten by the data fetched. † See **Status Bits** for more information.

MOVE --\*Rs, --\*Rd [,F] ---

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	0	1	0	1	0	F		F	łs		R		R	d	

Moves a field from one memory location to another. Both registers contain bit addresses; before the move, the contents of both registers are decremented by the field size.

If Rs and Rd are the same register, then the final contents of the register are its original contents decremented by twice the field size.

MOVE \*Rs(offset), Rd [,F] -

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	0	1	1	0	1	F	Rs				R		F	ld	
							off	set			-				

Moves a field from the source address into Rd. The source address is formed by adding a signed, 16-bit offset to the contents of Rs. When the field is moved into Rd, it is right-justified and sign- or zero-extended to 32 bits (depending on the value of the current FE bit). This instruction also performs an implicit compare to 0 of the field data. † See **Status Bits** for more information.

MOVE \**Rs(offset),* \**Rd*+ [,*F*] -----

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	1	0	1	0	0	F	Rs				R		F	۱d	
							off	set							

Moves a field from one memory location to another. The source address is formed by adding the contents of Rs to the signed 16-bit offset. Rd contains the address of the field's destination; after the move, the contents of Rd are incremented by the selected field size.

## MOVE \*Rs(SOffset), \*Rd(DOffset) [,F] -

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	0	1	1	1 0 F Rs R								F	۱d		
						S	ource	e offse	t						
	-					des	stinati	on off	set						

Moves a field from one memory location to another. The source address is formed by adding a signed 16-bit offset to the contents of Rs. The destination address is formed by adding a signed 16-bit offset to the contents of Rd.

## MOVE @SAddress, Rd [,F] -

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
0	0	0	0	0	1	F	1	1	0	1	R	Rs			
16 LSBs of source address															
16 MSBs of source address															

Moves a field from memory to the destination register. SAddress is a 32-bit address. When the field is moved into the destination register, it is right-justified

and sign- or zero-extended to 32 bits (depending on the selected value of FE). This instruction also compares the source data to 0. † See Status Bits for more information.

## MOVE @SAddress, \*Rd+ [,F] -

15	14	13	12	11	10	9	8	7	6	5	4	3 2 1 0			
1	1	0	1	0	1	F	0	0	0	0	R	Rs			
	16 LSBs of source address														
16 MSBs of source address															

Moves a field from one location in memory to another. The source address is a 32-bit address; the destination address is specified by the contents of Rd. After the move, the contents of the destination register are incremented by the field size.

## MOVE @SAddress, @DAddress [,F] -

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
0	0	0	0	0	1	F	1	1	1	0	0	0	0	0	0
16 LSBs of source address															
16 MSBs of source address															
16 LSBs of destination address															
16 MSBs of destination address															

Moves a field from one location in memory to another. Both addresses are 32-bit addresses.

The following status bits information applies only to these MOVEs: + MOVE \*Rs, Rd [,F] MOVE \*Rs+, Rd [,F] MOVE -\*Rs, Rd [,F] MOVE \*Rs(offset), Rd [,F] MOVE @SAddress, Rd [,F]

#### Status Bits

- N 1 if the field-extended data moved to register is negative, 0 otherwise C Unaffected
- Ζ 1 if the field-extended data moved to register is 0, 0 otherwise
- V 0

### **MOVE Examples**

Example 1

This is an example of the following MOVE instructions:

MOVE Rs, \*Rd MOVE Rs, \*Rd+ MOVE Rs, -\*Rd MOVE Rs, \*Rd(offset) MOVE Rs, @DAddress

Assume that memory contains the following value before instruction execution:

	<b>Address</b> 15500h 15510h 15520h	Data 0000h 0000h 0000h		R	egister A0	= FFFFFFF	Fh
<u>Code</u>		<u>Before</u> A1	FS0/1	<u>After</u> A1	@15500h	@15510h	@15520h
MOVE	A0,*A1,0	00015500h	5/x	00015500h	001Fh	0000h	0000h
MOVE	A0,*A1,1	00015503h	x/8	00015503h	078Fh	0000h	0000h
MOVE	A0,*A1,0	00015508h	13/x	00015508h	FF00h	001Fh	0000h
MOVE	A0,*A1,1	0001550Ch	x/24	0001550Ch	F000h	FFFFh	000Fh
MOVE	A0,*A1+,1	0001551Dh	x/16	0001552Dh	0000h	E000h	1FFFh
MOVE	A0,*A1+,0	00015516h	19/x	00015529h	0000h	FFC0h	01FFh
MOVE	A0,*A1+,1	00015500h	x/32	00015520h	FFFFh	FFFFh	0000h
MOVE	A0,-*A1,0	0001530h	5/x	000152Bh	0000h	0000h	F800h
MOVE	A0,-*A1,1	000152Dh	x/8	0001525h	0000h	0000h	1FE0h
MOVE	A0,-*A1,0	0001528h	13/x	000151Bh	0000h	F800h	00FFh
MOVE	A0,*A1(0000h)	,100015500h	x/1	00015500h	0001h	0000h	0000h
MOVE	A0,*A1(0FFFh)	,000014501h	19/x	00014501h	FFFFh	0007h	0000h
MOVE	A0,*A1(7FFFh)	,10000D501h	x/22	0000D501h	FFFFh	003Fh	0000h
MOVE	A0,*A1(8000h)	,00001D500h	27/x	0001D500h	FFFFh	07FFh	0000h
MOVE	A0,@1550Bh,1	XXXXXXXXX	x/16	xxxxxxxx	F800h	07FFh	0000h
MOVE	A0,@15512h,0	XXXXXXXXX	27/x	xxxxxxxx	0000h	FFFCh	1FFFh
MOVE	A0,@1550Ch,1	XXXXXXXXX	x/32	xxxxxxxx	F000h	FFFFh	0FFFh

#### Example 2

This is an example of the following MOVE instructions:

MOVE \*Rs, Rd MOVE \*Rs+, Rd MOVE --\*Rs, Rd MOVE \*Rs(offset), Rd MOVE @SAddress, Rd

Assume that memory contains the following value before instruction execution:

Address	Data	Address	Data
15500h	7770h	15530h	3333h
15510h	7777h	15540h	4444h
15520h	0000h	15550h	5555h

TMS34020 Assembly Language Instruction Set

Code	Before			After		
	A0	FS0/1	FE0/1	A0	A1	NCZV
MOVE *A0,A1,1	00015500h	ı x/1	x/1	00015500h	00000000h	0x10
MOVE *A0,A1,0	00015500h	1 5/x	0/x	00015500h	00000010h	0x00
MOVE *A0,A1,1	00015500h	ı x/5	x/1	FFFFFFF0h	00000000h	1x00
MOVE *A0,A1,0	00015500h	5/x	0/x	00015500h	00000010h	0x00
MOVE *A0,A1,0	00015500h	18/x	0/x	00037770h	00000010h	0x00
MOVE *A0+,A1,0	00015500h	12/x	0/x	0001550Ch	00000770h	0x00
MOVE *A0+,A1,1	00015500h	x/12	x/1	0001550Ch	00000770h	0x00
MOVE *A0+,A1,0	00015500h	27/x	0/x	0001551Bh	07777770h	0x00
MOVE *A0+,A1,1	00015500h	x/27	x/1	0001551Bh	FF777770h	1x00
MOVE -*A0,A1,0	00015520h	31/x	1/x	00015501h	3BBBBBB8h	0x00
MOVE -*A0,A1,0	00015520h	x/31	x/0	00015501h	3BBBBBB8h	0x00
MOVE -*A0,A1,0	00015520h	32/x	X/x	00015500h	77777770h	0x00
MOVE *A0(0020h),A1,1	0001551Ch	1x/ <b>13</b>	x/0	0001551Ch	00000443h	0x00
MOVE *A0(00FFh),A1,0	00015435h	16/x	1/x	00015435h	00004333h	0x00
MOVE *A0(7FFFh),A1,1	0000D531h	1 x/22	x/1	0000D531h	00000443h	0x00
MOVE *A0(8000h),A1,0	0001D530h	127/x	1/x	0001D530h	FC443333h	1x00
MOVE *A0(0FFECh),A1,0	0001554Dh	132/x	0/x	0001554Dh	AAA22219h	1x00
MOVE @15504h,A1,0	xxxxxxx	1/x	18/x	XXXXXXXX	FFFF7777h	1x00
MOVE @15500h,A1,1	XXXXXXXXX	x/0	x/18	XXXXXXXX	00037770h	0x00
MOVE @15501h,A1,0	XXXXXXXX	0/x	30/x	XXXXXXXXX	3BBBBBB8h	0x00
MOVE @15501h,A1,1	XXXXXXXXX	x/1	x/30	XXXXXXXXX	FBBBBBB8h	1x00

Example 3

This is an example of the following MOVE instructions:

MOVE \*Rs, \*Rd MOVE \*Rs+, \*Rd+ MOVE --\*Rs, --\*Rd MOVE @SAddress, @DAddress MOVE @SAddress, \*Rd+

Assume that memory contains the following value before instruction execution:

Address	Data	Address	Data
15500h	FFFFh	15530h	0000h
15510h	FFFFh	15540h	0000h
15520h	FFFFh	15550h	0000h

<u>Code</u>	<u>Before</u>			<u>After</u>					
	A0	A1	FS0/1	A0	A1	@15	530h @1	5540h	@1550h
MOVE *A0, *A1, 1 MOVE *A0, *A1, 0 MOVE *A0, *A1, 1 MOVE *A0, *A1, 1	00015500h 00015500h 00015500h 00015500h	00015530h 00015534h 0001553Ah 0001553Eb	x/1 5/x x/10 19/y	00015500h 00015500h 00015500h 00015500h	0001553 0001553 0001553	80h 84h 8Ah 85h	0001h 01F0h FC00h 8000b	0000h 0000h 000Fh EEEEb	0000h 0000h 0000h 0003h
HOVE "AO, "AI, O	000100000	000100011	13/2	000100000	0001000		000011		000311
MOVE *A0+, *AH, 1 MOVE *A0+, *AH, 0 MOVE *A0+, *AH, 1 MOVE *A0+, *AH, 1	00015510h 00015511h 00015513h 00015510h	00015532h 0001553Ah 0001553Fh 0001553Ah	x/7 13/x x/8 28/x	00015517h 0001551Fh 0001551Bh 0001552Ch	0001553 0001554 0001554 0001555	9h 7h 7h 6h	01FCh FC00h 8000h FC00h	0000h 007Fh 007Fh FFFFh	0000h 0000h 0000h 003Fh
MOVE *-A0, *-A1, 0 MOVE *-A0, *-A1, 1 MOVE *-A0, *-A1, 0 MOVE *-A0, *-A1, 1	00015527h 00015527h 0001552Ah 00015520h	00015555h 00015550h 00015550h 0001555Ah	31/x x/31 32/x x/32	00015508h 00015508h 0001550Ah 00015500h	0001553 0001553 0001553 0001553	6h 1h 0h Ah	FFC0h FFFEh FFFFh FC00h	FFFFh FFFFh FFFFh FFFFh	001Fh 0000h 0000h 03FFh
MOVE @15500h,*A1+ MOVE @15500h,*A1 MOVE @1550Dh,*A1 MOVE @15505h,*A1	,1 0001553A +,00001553A +,10001553A +,00001553A	h xxxxxxxx h xxxxxxxx h xxxxxxxx h xxxxxxxx	x/10 19/x 28/x x/32	00015544h 00015552h 0001554Ch 0001554Dh	xxxxxxxx xxxxxxxx xxxxxxxx xxxxxxxx	( ( (	FC00h 8000h FFFFh FFE0h	000Fh FFFFh 0FFFh 0FFFh	0000h 0003h 0000h 0000h
MOVE @15500h,@15 MOVE @15500h,@15 MOVE @15500h,@15 MOVE @15500h,@15	530h,1 xxxxx 534h,0 xxxxx 530h,1 xxxxx 530h,0 xxxxx	xxx xxxxx xxx xxxxx xxx xxxxx xxx xxxxx xxx xxxxx	(XX) (XX (XX (XX)	x/1 xxxxxx 5/x xxxxxx x/7 xxxxxx 13/x xxxxxx	xx xxxx xx xxxx xx xxxx xx xxxx xx xxxx		00011 01F0 007FI 1FFF	n 0000h h 0000h h 0000h h 0000h	0000h 0000h 0000h 0000h

Example 4

This is an example of the following MOVE instructions:

MOVE \*Rs(offset), \*Rd+ MOVE \*Rs(offset), \*Rd(offset)

Assume that memory contains the following value before instruction execution:

	Address	Data				Addres	s Data	
	15500h	0000h				15530h	3333h	
	15510h	0000h				15540h	4444h	
	15520h	0000h				15550h	5555h	
Code		<u>Before</u>			<u>After</u>			
		A0	A1	FS0/1	A1 .	@15530h	@15540h	@1550h
MOVE *A0(0000h),*A1+	,1	00015530h	0015500h	x/1	00015501h	0001h	0000h	0000h
MOVE *A0(00FFh), *A1+	,1	00015535h	001550Ch	16/x	0001551Ch	3000h	0433h	0000h
MOVE *A0(0FFFh),*A1+	,1	00015531h	00015510h	19/x	00015523h	0000h	3333h	0004h
MOVE *A0(0FFE0h),*A1-	+,1	00015558h	00015508h	x/31	00015527h	3300h	4444h	0055h
MOVE *A0(0001h), *A1(	0000h),0	0001552Fh	00015504h	5/x	00015504h	0130h	0000h	0000h
MOVE *A0(000Fh), *A1(	000Fh),0	0001552Dh	000154FDh	18/x	000154FDh	3000h	0004h	0000h
MOVE *A0(7FFFh),*A1(	B000h),1	0000D531h	0001D508h	x/22	0001D508h	3300h	0433h	0000h
MOVE *A0(0FFF2h), *A1	(7FFFh),	100015540h	0000D501h	x/25	0000D501h	0CCCh	0111h	0000h

Syntax	MOVI <i>IW, Rd</i> [, W]										
Execution	16-bit immediate value	→ Rd									
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         0       0       0       1       0       1       1       1       0       R       Rd         I6-bit value										
Description	MOVI stores a 16-bit, sign- ter. ( <i>IW</i> in the instruction s	extended immed yntax represents	liate value in the destination regis- the 16-bit value.)								
	The assembler uses the short form if the immediate value has been previously defined and is in the range $-32,768$ through $32,767$ . You can force the assembler to use the short form by following the register operand with <b>,W</b> :										
	MOVI IW,Rd,W										
	The assembler truncates message.	the upper bits ar	nd issues an appropriate warning								
Machine States	2										
Status Bits	<ul> <li>N 1 if the data being movies</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V 1 if the data being movies</li> </ul>	ved is negative, ( ved is 0, 0 otherv	9 otherwise vise								
Examples	<u>Code</u>	<u>After</u>									
		A0	NCZV								
	MOVI 32767,A0	00007FFFh	0 × 0 0								
	MOVI 1,A0	00000001h	0 × 0 0								
	MOVI 0,A0	00000000h	0 x 1 0								
	MOVI -1,A0	FFFFFFFh	1 × 0 0								
	MOVI = -32768, AU										
	MOVI 7FFFb A0	00007FFFh	0×00								

Syntax	MOVI <i>IL</i> , <i>Rd</i> [, L]											
Execution	32-bit immediate value $\rightarrow$ f	Rd										
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         0       0       0       0       1       1       1       1       1       Rd         I6 LSBs of IL											
Description	MOVI stores a 32-bit immed instruction syntax represents	diate value in the the 32-bit value	e destination .)	register. (IL in the								
	The assembler uses this opcode if it cannot use the MOVI IW, Rd opcode or if the long opcode is forced by following the register operand with, L:											
	MOVI IL,Rd,L											
Machine States	2 if immediate data is long-word aligned 3 if immediate data is not long-word aligned											
Status Bits	<ul> <li>N 1 if the data being move</li> <li>C Unaffected</li> <li>Z 1 if the data being move</li> <li>V 0</li> </ul>	d is negative, <i>0</i> o d is 0, <i>0</i> otherwis	therwise e									
Examples	Code	After										
		A0	NCZV									
	MOVI 2147483647,A0	7FFFFFFh	0 x 0 0									
	MOVI 32768,A0	00008000h	0 X 0 0									
	MOVI —32769,A0		1 × 0 0									
	$\frac{1}{100} = \frac{12147483648}{8000} = \frac{1214748}{8000} = \frac{1214748}{800} = \frac{1214748}{8000} = \frac{1214748}{8000} = 12$	00000000000000000000000000000000000000										
	MOVI OFFFFFFFA	FFFFFFFh	1 x 0 0									
	MOVI OFFFFh,A0,L FFFFFFh 1x00											

Syntax	MOVK constant, Rd								
Execution	5-bit constant $\rightarrow$	Rd							
Instruction Words	15 14 13 12 0 0 0 1	11 10 1 0	98	7 constan	65 t	4 R	3	2 1 Rd	0
Description	MOVK stores a 5 treated as an unsi opcode correspor extended to 32 bit	i-bit constang gned numb nds to a val	nt in the er in the ue of 3:	e destii e range 2. The	nation reg 1—32, w resulting	giste here cons	r. The const stant v	e cons tant = ( /alue i	stant is 0 in the s zero-
	Note that you can register by XORing for XOR <i>Rs, Rd</i> ) to to 1).	not set a re g the registe o accomplis	gister to er with it sh this. I	o 0 with self; us Both th	n this insti e clr rd ese meth	ructio (an a ods a	on. Yo Iterna alter th	u can ite mne ne Z bi	clear a emonic it (set it
Machine States	1								
Status Bits	<ul><li>N Unaffected</li><li>C Unaffected</li><li>Z Unaffected</li><li>V Unaffected</li></ul>								
Examples	Code           MOVK         1,A0           MOVK         8,A0           MOVK         16,A0           MOVK         32,A0	After A0 00000001h 0000008h 00000010h 00000020h							
Syntax	MOVX Rs, Rd								
-------------------	----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------								
Execution	X half of Rs $\rightarrow$ X half of Rd								
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         1       1       1       0       1       1       0       Rs       R       Rd								
Description	MOVX moves the X half of the source register (16 LSBs) to the X half of the destination register. The Y halves of both registers are unaffected.								
	You can also use the MOVX and MOVY instructions for handling packed 16-bit quantities and XY addresses. You can use the RL instruction to swap the contents of X and Y.								
	Rs and Rd must be in the same register file.								
Machine States	1								
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>								
Examples	Code         Before         After           A0         A1         A1           MOVX A0,A1         0000000h         FFFFFFh         FFFF000h           MOVX A0,A1         12345678h         0000000h         00005678h           MOVX A0,A1         FFFFFFh         0000000h         0000FFFh								

Syntax	MOVY Rs, Rd	,				
Execution	Y half of Rs $\rightarrow$	Y half of Rd				
Instruction Words	15 14 13 1 1 1	12 11 10 0 1 1	9 8 7 6 1 Rs	6 5 4 R	3 2 1 Rd	0
Description	MOVY moves to destination regi	he Y half of the ster. The X halv	source registe ves of both regi	r (16 MSBs) isters are un	to the Y half on affected.	of the
	You can also use quantities and X tents of X and Y	e the MOVX and (Y addresses. \ ⁄.	d MOVY instruc ⁄ou can use the	tions for han RL instructio	Idling packed <sup>-</sup> on to swap the	l 6-bit con-
	Rs and Rd mus	t be in the sam	e register file.			
Machine States	1					
Status Bits	<ul><li>N Unaffected</li><li>C Unaffected</li><li>Z Unaffected</li><li>V Unaffected</li></ul>					
Examples	Code           MOVY A0,A1           MOVY A0,A1           MOVY A0,A1	<u>Before</u> A0 00000000h 12345678h FFFFFFFh	<b>A1</b> FFFFFFFFh 00000000h 00000000h	After A1 0000FFFFh 12340000h FFFF0000h		

Syntax	MP۱	rs f	Rs, R	d												
Execution	lf Ro If Ro	l is a l is a	n eve n ode	en-nu d-nur	mbe nbere	red re ed re	egiste gister	ər, R r, Rs	s × F s × R	{d → d →	Rd: Rd	Rd+	1			
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	1	0	1	1	1	0		F	Rs		R		F	ld	

Description

MPYS performs a signed multiply of a variably sized field in the source register by the 32 bits in the destination register. This produces a 32-bit to a 64-bit result, depending on the register and field definitions. Note that Rs and Rd must be in the same register file.

The value of field size 1 (FS1) defines the size of the multiplier in Rs. FS1 may have any **even** value *n* from 2 to 32 (that is, n = 2, 4, 6...30, 32). The instruction executes a 32-bit-by-*n*-bit multiply — multiplying the 32 bits in Rd by the *n* bits in Rs. All values are signed. The MSB of the source field (bit n - 1 in Rs) defines the sign of the field; the bits to the left of bit *n* are ignored. The MSB of Rd defines the sign of the multiplicand.

Contents	of Rs ( <i>n</i> = FS1); <i>n</i> −1	03	Contents of Rd:
ignored	<i>n</i> -bit multiplier	וֹכ	32-bit multiplicand
1	sign bit		sign bit

MPYS has two modes, depending on whether Rd is even or odd:

## **G** Rd Even:

MPYS multiplies the contents of Rd by the *n*-bit field in Rs, and stores the result in 2 consecutive registers, Rd and Rd+1. (For example, if Rd=B4, the result is stored in registers B4 and B5.) The result is sign-extended and right-justified; the 32 MSBs are stored in Rd, and the 32 LSBs are stored in Rd+1. Note that all 32 bits of both registers are used, regardless of the field size of the multiply.

Do not use A14 or B14 as the destination register, because Rd+1 (A15 or B15) is the stack pointer register (SP). It is not desirable to write over the contents of the SP.

Contents of Bd (even register) Conter	nts of Bd+1 (odd register)
Contents of Au (even register) Conten	
<u>31 n n – 1 0 31</u>	0
sign n MSBs of result	32 LSBs of result

## 🖬 Rd Odd:

MPYS multiplies the contents of Rd by the *n*-bit field in Rs, and stores the 32 LSBs of the result in Rd; neither Rs nor Rd+1 are changed. If the result is greater than 32 bits, the extra MSBs are discarded, regardless of the field size. The N and Z status bits, however, are set according to the full result, including the MSBs that are discarded.

Contents of Rd (odd register) 31 0	
32 LSBs of result	

Machine States 5 + (field size)/2

Status Bits

- N 1 if the result is negative, 0 otherwise
- C Unaffected
- Z 1 if the result is 0, 0 otherwise
- V Unaffected

Example 1

#### MPYS A1, A0

<u>Before</u>			<u>After</u>		
A0	A1	FS1	A0	A1	NCZV
00000000h	00000000h	32	00000000h	00000000h	0 x 1 x
7FFFFFFFh	7FFFFFFFh	32	3FFFFFFFh	00000001h	0 x 0 x
7FFFFFFFh	FFFFFFFh	32	FFFFFFFh	80000001h	1 x 0 x
FFFFFFFFh	7FFFFFFFh	32	FFFFFFFh	80000001h	1 x 0 x
FFFFFFFh	FFFFFFFh	32	00000000h	00000001h	0 x 0 x
80000000h	7FFFFFFFh	32	C0000000h	80000000h	1 x 0 x
80000000h	80000000h	32	40000000h	00000000h	0 x 0 x
80000001h	80000000h	32	3FFFFFFFh	80000000h	0 x 0 x
8040156Fh	7FF3B074h	32	C0262CDCh	53E486F8h	1 x 0 x
8040156Fh	7FF3B074h	24	000624B1h	53E486F8h	0 x 0 x
8040156Fh	7FF3B074h	20	FFFE28B2h	594486F8h	1 x 0 x
8040156Fh	7FF3B074h	16	000027B2h	17EC86F8h	0 x 0 x
8040156Fh	7FF3B074h	14	000007C2h	1C0206F8h	0 x 0 x
8040156Fh	7FF3B074h	8	FFFFFFC6h	1D0766F8h	1 x 0 x
8040156Fh	7FF3B074h	6	00000005h	FCFF3BF8h	0 x 0 x
8040156Fh	7FF3B074h	4	FFFFFFEh	01004158h	1 x 0 x
8040156Fh	7FF3B074h	2	00000000h	00000000h	0 x 1 x

Exam	ple	2
	<b></b>	_

```
MPYS A0,A1
```

<u>Before</u>			<u>After</u>		
A0	A1	FS1	A0	A1	NCZV
00000000h	00000000h	32	unchanged	00000000h	0 x 1 x
7FFFFFFFh	7FFFFFFFh	32	unchanged	0000001h	0 x 0 x
7FFFFFFFh	7FFFFFFFh	32	unchanged	80000001h	1 x 0 x
FFFFFFFFh	7FFFFFFFh	32	unchanged	80000001h	1 x 0 x
FFFFFFFh	FFFFFFFh	32	unchanged	00000001h	0 x 0 x
80000000h	7FFFFFFFh	32	unchanged	80000000h	1 x 0 x
80000000h	80000000h	32	unchanged	00000000h	0 x 0 x
80000001h	80000000h	32	unchanged	80000000h	0 x 0 x
7FF3B074h	80401056h	32	unchanged	53E486F8h	1 x 0 x
7FF3B074h	80401056h	24	unchanged	53E486F8h	0 x 0 x
7FF3B074h	80401056h	20	unchanged	594486F8h	1 x 0 x
7FF3B074h	80401056h	16	unchanged	17EC86F8h	0 x 0 x
7FF3B074h	80401056h	14	unchanged	1C0206F8h	0 x 0 x
7FF3B074h	80401056h	8	unchanged	1D0766F8h	1 x 0 x
7FF3B074h	80401056h	6	unchanged	FCFF3BF8h	0 x 0 x
7FF3B074h	80401056h	4	unchanged	01004158h	1 x 0 x
7FF3B074h	80401056h	2	unchanged	00000000h	0 x 1 x

Syntax	MPY	<b>(U</b> F	₹s, R	d												
Execution	lf Ro If Ro	l is ai I is ai	n eve n ode	en-nu d-nur	ımbe nber	red re	egist giste	er: R r: R	ls × F s × R	Rd → Id →	► Rd Rd	:Rd+	1			
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	1	0	1	1	1	1			Rs		R		F	ld	
Description	MPY	′U pe	rforn	nsan	unsi	gned	multi	iply c	fava	ariabl	y-siz	ed fie	ldin	these	ource	ereg-

MPYU performs an unsigned multiply of a variably-sized field in the source register by the 32 bits in the destination register. This produces a 32-bit to a 64-bit result, depending on the register and field definitions. Note that Rs and Rd must be in the same register file.

The value of field size 1 (FS1) defines the size of the multiplier in Rs. FS1 may have any **even** value *n* from 2 to 32 (that is,  $n = 2, 4, 6 \dots 30, 32$ ). The instruction executes a 32-bit-by-*n*-bit multiply — multiplying the 32 bits in Rd by the *n* bits in Rs. All values are unsigned.

Contents 31 n	of Rs ( <i>n</i> = FS1): <i>n</i> -1 0	3	Contents of Rd: 1 0	
ignored	<i>n</i> –bit multiplier		32-bit multiplicand	

MPYS has two modes, depending on whether Rd is even or odd:

## **Rd Even:**

MPYU multiplies the contents of Rd by the *n*-bit field in Rs and stores the result in 2 consecutive registers, Rd and Rd+1. (For example, if Rd=B4, the result is stored in registers B4 and B5.) The result is zero-extended and right-justified; the 32 MSBs are stored in Rd, and the 32 LSBs are stored in Rd+1. Note that all 32 bits of both registers are used, regardless of the field size of the multiply.

Do not use A14 or B14 as the destination register, because Rd+1 (A15 or B15) is the stack pointer register (SP). It is not desirable to write over the contents of the SP.

|--|--|

# 🖬 Rd Odd:

MPYU multiplies the contents of Rd by the *n*-bit field in Rs and stores the 32 LSBs of the result in Rd; Rs is not changed. If the result is greater than 32 bits, the extra MSBs are discarded, regardless of the field size. The Z status bit, however, is set according to the full result, including the MSBs that are discarded.

	Contents of Rd (odd register)							
				31	LSBs of result	0		
Machine States	Rs nonnegati Rs negative:	ve: 6 + (field size	5 + e)/2	· (field size)/2				
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z 1 if the res</li> <li>V Unaffected</li> </ul>	d d sult is 0, <i>0</i> othe d	rwise					
Example 1	MPYU A1,A0							
	Before           A0           FFFF0000h           FFFF0000h           FFFF0000h           FFFF0000h           FFFF0000h           FFFF0000h	A1 10000000h 10001010h 10001010h 10001010h 10001010h	<b>FS1</b> 32 32 16 8 4	After A0 0FFFF000h 1000000Fh 0000100Fh 0000000Fh 0000000h	A1 00000000h EFF00000h EFF00000h FFF00000h 0000000h	NCZV x x 0 x x x 0 x x x 0 x x x 0 x x x 1 x		
	08001056h 08001056h 08001056h 08001056h 08001056h 08001056h	0003B074h 0003B074h 0003B074h 0003B074h 0003B074h 0003B074h	32 16 14 8 6 4 2	00001D33h 00000183h 00000003h 000000001h 00000000h 00000000h	AB4286F8h A31786F8h A00766F8h A0035178h 20004158h 00000000h	x x 0 x x x 1 x		
Example 2	MPYU A0,A1							
	<u>Before</u> A0 10000000h 10001010h	<b>A1</b> FFFF0000h FFFF0000h	<b>FS1</b> 32 32	After A0 unchanged unchanged	A1 00000000h EFF00000h	NCZV x x 0 x x x 0 x		
	10001010h 10001010h 10001010h 0003B074h 0003B074h	FFFF0000h FFFF0000h FFFF0000h 08001056h 08001056h	16 8 4 32 16	unchanged unchanged unchanged unchanged unchanged	EFF00000h FFF00000h 00000000h DC4486F8h AB4286F8h	x x 0 x x x 0 x x x 1 x x x 0 x x x 0 x		
	0003B074h 0003B074h 0003B074h 0003B074h	08001056h 08001056h 08001056h 08001056h	14 8 6 4	unchanged unchanged unchanged unchanged	A31786F8h A00766F8h A0035178h 20004158h	x x 0 x x x 0 x x x 0 x x x 0 x		
	0003B074h	08001056h	2	unchanged	00000000h	x x 1 x		

TMS34020 Assembly Language Instruction Set

Syntax	MWAIT	MWAIT														
Execution	Wait for o	curren	t me	mory	/ сус	le to	o co	mp	lete							
Instruction Words	15 14	13	12	11	10	9	8	3	7	6	5	4	3	2	1	0
	0 0	0	0	0	0	0	0	)	1	0	0	0	0	0	0	0
Description	MWAIT of to comple execution tion is de	<i>I</i> WAIT delays further instruction execution to allow any pending write cycle o complete. If no write cycle is currently pending, the next instruction begins execution immediately. If a write cycle is pending, execution of the next instruc- ion is delayed until the write cycle completes.														
•• • • • • • • • • • • • • • • • • • •	MWAIT is been con depends a pendin TMS3402 depends a descrip	s typic mplete on the g write 20 has on the otion o	ally u ed pr e valu e to a s con e valu f the	used rior f ues in reg nplet ue in pote	to e to b ister ister ed p the ential	nsui egin I/O in a rior regis Iate	re ti ning reg me to e ster	nat g a iste eme exe xe . Re v of	all p gra ers. I ory-n cutin efer writ	endi phic t ma napp g an to Se es to	ng I/ s in y als ed p inst ectio I/O	O re struc o be periph ruction n 4.8 regis	gister tion usec neral on wl 5.6 or sters.	r upd exec I to e exte hose n pag	ates utior nsur rnal t oper je 4- <sup></sup>	have that to the ration 13 for
Machine States	minimum	minimum of 2														
Status Bits	N Unaf C Unaf Z Unaf V Unaf	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>														
Examples	SETF MOVE MWAIT DRAV	MOVK 16,0 B10, A0,A: ; ;	,0 @C0 2 In in ent	0001 this 1 hi erec	4, 50h s ca dde d. M	B10 se t n st WAIJ	che cato [ w.	; ; 16 = a i11	loa wai bit t tl . tal	d PS t fo t MO ne t ke 2	IZE rwi VE t ime cyc	cite co P: MWA: cles	to o SIZE IT is to o	comp resi s exect	lete ults ute.	
	MOVK SETF MOVE MWAIT DRAV	4, B; 6,0,0 B10, A0,A; ; ;	10 0 22 In in ent	0001 this 2 hi erec	50h ca dde 1. M	se t n st WAIJ	che cate	; ; 6 es	loa wai bit at t	d PS t fo MOV the ce 3	IZE r wi E to time cyc	rite PS: MW2 :les	to o IZE 1 AIT : to o	comp resu is exect	lete lts ute.	

# NEG Negate Register

Syntax	NEG Rd												
Execution	2s comple	ment of F	Rd → I	Rd									
Instruction Words	15 14 0 0	13 12 0 0	11 10 0 0	) 9 ) 1	8 1	7	6 0	5 1	4 R	3	2 R	1 d	0
Description	NEG store into the de	s the 2s c estination	ompler registe	nent of r.	the c	conten	ts of	the c	lestir	natior	reg	isterl	oack
Machine States	1												
Status Bits	<ul> <li>N 1 if the</li> <li>C 1 if the</li> <li>Z 1 if the</li> <li>V 1 if the</li> </ul>	e result is ere is a bo e result is ere is an o	negativ orrow (I 0, <i>0</i> ot overflov	ve, <i>0</i> ot Rd ≠ 0) herwise v (Rd =	herv , <i>0</i> c 9 800	vise otherw 000000	ise Dh), (	) oth	erwis	se			
Examples	Code NEG A0 NEG A0 NEG A0 NEG A0 NEG A0	Before A0 00000000 55555555 7FFFFF 8000000 80000001 FFFFFF	0h 5h Fh 0h 1 h	After NCZV 001 110 110 110 010 010	V 0 0 1 0	A0 00000 AAAA 80000 80000 7FFF	000h AAAE 001h 000h FFFI	3h n					

Syntax	NEGB R	d								
Execution	(2s compl	ement of Rd) –	C →	Rd						
Instruction Words	15 14 0 0	13 12 11 0 0 0	10 9 0	9 8 7 1 1 1	6 5 1 0	4 R	3	2 Ro	1 1	0
Description	NEGB tak rements b register. T instructior	NEGB takes the 2s complement of the destination register's contents and dec- rements by 1 if the borrow bit (C) is set; the result is stored in the destination register. This instruction can be used in sequence with itself and with the NEG instruction for negating multiple-register quantities.								
Machine States	1									
Status Bits	<ul> <li>N 1 if the</li> <li>C 1 if the</li> <li>Z 1 if the</li> <li>V 1 if the</li> </ul>	e result is nega ere is a borrow, e result is 0, 0 c ere is an overflo	tive, C O oth otherw ow, O o	) otherwise erwise vise otherwise						
Examples	Code NEGB A0	<u>Before</u> A0 00000000h	<b>C</b> 0	<u>After</u> N C Z V 0 0 1 0	<b>A0</b> 00000000	)h				
	NEGB AU	000000000 55555555	0	1100		rn ARh				
	NEGB A0	555555555h	1	1100	AAAAAAA	4Ah				
	NEGB A0	7FFFFFFFh	0	1100	80000001	h				
	NEGB A0	7FFFFFFFh	1	1100	80000000	)h				
	NEGB A0	80000000h	0	1101	80000000	)h				
	NEGB A0	80000000h	1	0100	7FFFFFF	Fh				
	NEGB A0	8000001h	0	0100	7FFFFFF	Fh				
	NEGB A0	80000001h	1	0100	7FFFFFF	Eh				
	NEGB A0	FFFFFFFF	0	0100	00000001	n No				
	NEGB A0	FFFFFFFN	1	0110	00000000	m				

## NOP No Operation

Syntax	NOP												
Execution	No operat	ion											
Instruction Words	15 14 0 0	13 12 0 0	11 10 0 0	9	8	7	6 0	5 0	4	3 0	2	1 0	0
Description	The progr processor You can us tions.	am coun status is se the NC	iter is ind otherwis DP instrue	creme e una ction t	nted ffect o pa	i to p ed. d loo	point	to th nd pe	ne ne	ext ir m oth	nstrue ner tir	ction. ming :	The func-
Machine States	1												
Status Bits	N Unaffe C Unaffe Z Unaffe V Unaffe	ected ected ected ected											
Example	Code	<u>Before</u> PC 0002000	E F Oh C	<u>After</u> PC 00200	10h								

Syntax	NOT Rd											
Execution	NOT Rd	→ Rd										
Instruction Words	15 14 0 0	13 12 0 0	11 10 0 0	9	8 7 1 1	6 1	5 1	4 R	3	2 R	1 d	0
Description	NOT store the destin	s the 1s c ation regi	ompleme ster.	ent of th	ie destin	ation	regi	ster's	cont	tents	back	into
Machine States	1											
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z 1 if the result is 0, 0 otherwise</li> <li>V Unaffected</li> </ul>											
Examples	Code NOT A0 NOT A0 NOT A0 NOT A0	Before A0 00000000 55555555 FFFFFF 80000000	Dh x 5h x Fh x Dh x	After N C Z V X X 0 X X X 0 X X X 1 X X X 0 X	<b>A0</b> FFFF AAA4 00000 7FFF	FFFF \AAA/ D000h FFFF	h Գh հ					

### **OR** OR Registers

Syntax	OR Rs, R	d			
Execution	Rs OR Rd	→ Rd			
Instruction Words	15 14 0 1	13 12 11 1 0 1 0	0 9 8 7 1 0 F	654 Rs F	4 3 2 1 0 R Rd
Description	This instruction tents of the	ction bitwise-OF destination reg	Rs the contents of ister; the result is	of the source a stored in th	register with the con- e destination register.
	Rs and Rd	must be in the	same register fil	е.	
Machine States	1				
Status Bits	<ul> <li>N Unaffect</li> <li>C Unaffect</li> <li>Z 1 if the</li> <li>V Unaffect</li> </ul>	cted cted result is 0, <i>0</i> ot cted	herwise		
Examples	Code OR A0,A1 OR A0,A1 OR A0,A1 OR A0,A1	Before A0 FFFFFFFh 00000000h 55555555h 00000000h	<b>A1</b> 00000000h FFFFFFFh AAAAAAAA 00000000h	After A1 FFFFFFFh FFFFFFFh FFFFFFFh 00000000h	N C Z V ××0× ××0× ××0× ××1×

Syntax	ORI	IL, F	Rd													
Execution	32-bit immediate value OR Rd $\rightarrow$ Rd															
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2									2	1	0				
		0	0		•	0	1	61.SBs	sofil	0		_ n			u 	
	16 MSBs of IL															
Description	This i destir synta	This instruction bitwise-ORs a 32-bit immediate value with the contents of the destination register and stores the result in the destination register. ( <i>IL</i> in the syntax represents the 32-bit value.)														
Machine States	2 if immediate data is long-word aligned 3 if immediate data is long-word aligned															
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z 1 if the result is 0, 0 otherwise</li> <li>V Unaffected</li> </ul>															
Examples	Code ORI ( ORI ( ORI ( ORI 0	)FFF1 )0000 )AAA2 0000	FFFF 0000 AAAA 0001	Fh,À h,A0 Ah,A 1,A0	0 0	Befo A0 0000 FFFF 5555 0000	<u>re</u> 0000 FFFI 5555 0000	h =h h	Af A0 FF FF FF	ter FFFF FFFF FFFF	FFh FFh FFh S0h	N X X X X	<b>C Z \</b> x 0 x x 0 x x 0 x x 0 x x 1 x	/		

Syntax	PFILL XY								
Execution	COLOR0 and COLOR1 pixels $\rightarrow$ pixel array (with processing)								
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         0       0       0       0       1       0       1       0       0       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1       1								
Description	PFILL XY fills a pixel array, one row at a time, with a 2-color pattern. The pattern is defined by the PATTERN register. The 2 colors are defined by the COLOR0 and COLOR1 registers. PFILL replaces the 1s in pattern with the pixel value in COLOR1; it replaces the 0s in the pattern with the pixel value in COLOR0. To fill an array with a 2-dimensional pattern, execute PFILL once for each row of the array. If the width of the specified fill region (defined by <i>DX</i> in DYDX) is more than 32 pixels, PFILL replicates the same 32-bit pattern as many times as necessary to fill the row. After each line is drawn, you will typically update the contents of the PATTERN register to define the next row of the pattern. If you do not update the PATTERN register between rows, or if the number of rows in the fill region is 1, then the same 1-dimensional pattern is repeated.								
	for each row of the destination array. If the destination array pitch is a power of 2, and a pattern is drawn to the screen in this manner, then the filled area appears to contain stripes. If the destination array pitch is not a power of 2, then the pattern is defined for only the first line of the array.								

## Aligning a pattern

The contents of the PATTERN register control the pattern. As an example, consider the pixel addressed by the XY address in DADDR at the start of the PFILL instruction. Let that pixel be the n th pixel from the least significant end of a longword boundary, where n is in the range:

$$0 \le n \le \left(\frac{32}{\text{PSIZE}} - 1\right)$$

Bit *n* in the PATTERN register determines if the first pixel drawn is a COLOR0 or COLOR1 pixel. Bit n + 1 determines if the second pixel drawn is a COLOR0 or COLOR1 pixel, and so on. The PATTERN register works cyclically to draw a line. If the *DX* value in DYDX is large, then eventually bit 31 of the PATTERN register will be used to control an output pixel. If a further pixel is drawn, then it will be controlled by bit 0 of the pattern and so on.

## Prealigning a pattern

The last example demonstrated that PFILL does not perform any internal alignment of the PATTERN register. This cuts the overhead time required to start executing and enables you to perform a pattern prealignment to suit your needs.

Consider this case where no prealignment is performed:

PSIZE		= 04h
DADDR	(B2)	= 00000 0000h
DPTCH	(B3)	= 00000 0000h
OFFSET	(B4)	= 00000 0000h
DYDX	(B7)	= 00020 0060h
COLOR0	(B8)	= 00000 0000h
COLOR1	(B9)	= 0FFFFFFFFh
PATTERN	(B13)	= 0FFF000FFh

For this example, PFILL draws a rectangle 96 pixels wide and 32 pixels high. The rectangle contains vertical stripes, alternating between COLOR1 and COLOR1. The first pixel drawn, at bit address 00000000h, is controlled by bit 0 of the PATTERN register.

If the screen is clear and the X part of DADDR is incremented by 1 to 00000001h, PFILL will redraw the rectangle. The first pixel drawn by PFILL, now at bit address 0000004h, will be controlled by bit 1 of the pattern register. The drawn pattern now appears as if it were fixed relative to the screen (not the rectangle edge).

This continues as the X component of DADDR is incremented until DADDR = 000000008h; at this point the first pixel drawn by PFILL, now at bit address 000000020h, will no longer be in the first 32-bit long-word of the screen. In this case, by the argument used above, the first drawn pixel is controlled, once again, by bit 0 of the PATTERN register and thus changing from DADDR = 000000007h to DADDR = 000000008h the pattern will appear to *jump* within the rectangle. This may not always be a desirable way to manage the pattern. You may wish to do one of the following:

- Greate a pattern that appears to be fixed relative to the screen background.
- **Gamma** Create a pattern that appears fixed relative to the edge of the rectangle.

#### Placing the pattern relative to the screen background

There are three ways to fixing or placing a pattern relative to the screen background.

1) Let the number of pixels in a long-word be p, where

$$p = \frac{32}{\text{PSIZE}}$$

If the pattern in the PATTERN register repeats every p pixels, then it will appear fixed with respect to the screen background.

If you set B13 = FF00FF00FFh in the last example, the pattern will not jump.

- 2) Use a pixel size of 1 bit.
- 3) Manually rotate the contents of B13 before executing PFILL. The rotation amount depends on the following two things:
  - Pixel size
  - X component of DADDR

Let the total number of bits controlled by the entire pattern (that is  $32 \times PSIZE$  bits), be known as a *super-word* (range 32 to 1024 bits in size).

Let the long-word containing the pixel addressed by DADDR at the start of the PFILL XY be the  $n^{\text{th}}$  long-word in a super-word (range 0 to 32).

Let the number of pixels in a 32-bit long-word be *p*, where

$$p = \frac{32}{\text{PSIZE}} \quad \text{(range 0 to 32).}$$

Before starting PFILL XY, the pattern should be rotated right by  $m \times p$  bits, before placing it in PATTERN.

This may appear complex, but because pixel size is usually fixed, the prealignment operation can be reduced to a simple sequence of instructions. For example, at 4 bits per pixel, ANDing the XY address in DADDR (available before execution of PFILL) with 018h yields the value  $m \times p$  which can be used to rotate the pattern before placing it into the PATTERN register. At other pixel sizes the following will yield  $m \times p$ :

PSIZE	$m \times p$	
1	000h AND DADDR	always 0 (no rotation required)
2	010h AND DADDR	
4	018h AND DADDR	
8	01Ch AND DADDR	
16	01Eh AND DADDR	
32	01Fh AND DADDR	

#### Placing the pattern relative to the rectangle

Placing or fixing the pattern relative to the rectangle means that the first pixel drawn by PFILL is always controlled by bit 0 of the PATTERN register. (This is how the LINE and FLINE instructions use PATTERN register.)

You can achieve a similar effect for PFILL by rotating the pattern left by a certain amount before placing it into the PATTERN register. The rotation amount depends on the following two things:

- Pixel size
- X component of DADDR

The rotation amount is derived by ANDing DADDR with a constant as follows:

PSIZE	Rotate Amount	
1	01Fh AND DADDR	
2	00Fh AND DADDR	
4	007h AND DADDR	
8	003h AND DADDR	
16	001h AND DADDR	
32	000h AND DADDR	always 0 (no rotation required)

#### Note:

This description describes a striped rectangle, but in practice PFILL is used to pattern-fill a single line followed by a change of pattern before pattern filling a second line, and so on. The reference to a rectangle is made for the purpose of illustration only.

#### Implied Operands

Register Name		Format	Description
B2	DADDR	XY	Destination pixel block address
B3 †	DPTCH	Linear	Destination pixel block pitch
B7	DYDX	XY	Dimensions of drawn rectangle
B13	PATTERN	Binary	Pattern register
B14	POFFSET	Integer	Offset into the pattern
1			

<sup>†</sup> If DY > 1, then DPTCH <u>must</u> be a power of 2, or the pattern will not be well defined.

Address	Name	Description and Elements (Bits)
C00000B0h	CONTROL	PPOPPixel-processing operations (22 options)TTransparency operationTMSets transparency mode
C0000150h	PSIZE	Pixel size (1,2,4,8,16,32)
C0000160h	PMASK (32 bits)	Plane mask — pixel format

Due to the pipelining of memory writes, the *last* I/O register that you write to may not, in some cases, contain the desired value when you execute the PFILL instruction. To ensure that this register contains the correct value for execution, you may want to follow the write to that location with an MWAIT (page 13-177). Refer to Section 4.5.6 on page 4-13 for a description of the potential latency of writes to I/O registers.

*Pixel Processing* Pixel processing can be used with this instruction. For more information, refer to Section 12.8, <u>Pixel Processing</u>, on page 12-27.

*Window Checking* Window checking **can** be used with this instruction.

Transparency		You can e Select 1 o mation, re	You can enable transparency for this instruction by setting T[[CONTROL]] to 1. Select 1 of 3 transparency modes by setting TM[[CONTROL]]. For more information, refer to Section 12.9, <u>Transparency</u> , on page 12-36.				
Plane Masking		The plane Section 1	The plane mask is enabled for this instruction. For more information, refer to Section 12.10, <u>Plane Masking</u> , on page 12-39.				
Corner Ad	just	Corner ac	ust cannot be used with this instruction.				
Machine S	tates	Complex	nstruction				
Status Bits		N Unaffe C Unaffe Z Unaffe V Unaffe	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>				
Examples		This is an gle on the expects 5 a pointer t	example of a C-compatible assembly routine which draws a rectan- screen; the screen is filled with a $16 \times 16$ binary pattern. This routine arguments on the C parameter stack: width, height, xleft, ytop, and o the pattern.				
		This routir caller:	e assumes the following registers were previously initialized by the				
		B-file regi I/O registe	ters DPTCH, OFFSET, WSTART, WEND, COLOR1, COLOR0 rs CONTROL, CONVDP, PSIZE and PMASK				
STK DADDR DYDX PATTERN _fill_pa	.set .set .set .globl tnrect: mmtm	A14 B2 B7 B13 _fill_patnre SP,A0,A1,A2	<pre>;C-parameter stack pointer ;Destination address register ;Delta X/delta Y register ;Pattern register ct ;provide reference for external calls A3 ;save required registers</pre>				
	move move move move move move sll movy	SF, B0, B1, B2, STK, B14 *-B14, DYDX, 1 *-B14, B10, 1 *-B14, DADDR, *-B14, B11, 1 B14, STK *-STK, A3, 1 16, B10 B10, DYDX	<pre>;pop w ;pop h 1 ;pop xleft ;pop ytop ;pop pointer to pattern ;concatenate w, h</pre>				
	sll movy clip jrz move move srl movi	16,B11 B11,DADDR exit DYDX,A1 A1,A2 16,A1 00010000H,A0	;concatenate xleft, ytop ;clip the rectangle to the window ;jump if rectangle outside window ;set up y count				

loop:

A0,A2

B0,B0 B0

в0

A2,DYDX DADDR,A2

movy move

move

rmo

neg

getps

movk	32,B1	
srl	в0,81	number pixels per 32 bit word;
subk	1,B1	;so complement will count pix's wrd
move	DADDR,B0	
andn	в1,80	;address rounded to pix's/word bndry
neg	в0	;shift count = -(LSBs of x)
move	A3,B10	;pattern start address
movk	15,B11	;load 4-bit mask
sll	16,B11	align mask with 4 LSBs of y;
and	DADDR,B11	;isolate 4 LSBs of y
srl	12,B11	;convert y to index value
add	B11,B10	;index into pattern
move	*B10,B10,0	;get 16-bit row of pattern
move	в10,в11	
sll	16,B11	
movy	B11,B10	replicate row to 32 bits;
rl	в0,810	align pattern for x address;
move	B10,PATTERN	;load aligned pattern
pfill	XY	
addxy	A0,A2	
move	A2,DADDR	
dsj	A1,loop	
mmfm	SP,B0,B1,B2,B7,	B10,B11,B13,B14
mmfm	SP,A0,A1,A2,A3	restore required registers;
rets	2	;return

;calculate pattern adjustment

exit:

13-189

- **PIXBLT Instructions** The **PIXBLT** instruction moves a 2-dimensional array of pixels from one memory location to another. Section 12.5, <u>Pixel-Array Instructions</u>, on page 12-8 provides additional information about the PIXBLT instructions. The following list describes characteristics common to all **PIXBLT** instructions. Note that **PIXBLT L,M,L** is discussed independently on page 13-204.
  - The source and destination addresses of the arrays are designated by the SADDR and DADDR registers, respectively.
  - B, L, and XY are not actually operands. Instead, they identify the source or destination array starting addresses as binary, linear, or XY addresses.
     B, L, and XY are referred to as qualifiers.
  - Qualifiers are entered exactly as shown in the syntax; for example, PIXBLT B, L. The first qualifier indicates the format of the starting address of the source array; the second qualifier indicates the format of the starting address of the destination array.
  - You can select a **pixel-processing** option by setting PPOP[[CONTROL]]. When the PIXBLT has binary source data, the pixel-processing operation is applied to *expanded pixels* as they are processed with the destination array; that is, the data is *first expanded* and *then processed*. There are 16 Boolean and 6 arithmetic operations; the default case at reset is the S → D operation. Not that the 6 arithmetic operations do not operate with pixel sizes of 1 bit per pixel. For more information, refer to Section 12.8, <u>Pixel Processing</u>, on page 12-27.
  - ❑ You can enable transparency by setting T[[CONTROL]] to 1. The TMS34020 supports 3 transparency modes; TM[[CONTROL]] selects 1 of 3 transparency options. For more information, refer to Section 12.9, <u>Transparency</u>, on page 12-36.
  - The **plane mask** is enabled. For more information, refer to Section 12.10, <u>Plane Masking</u>, on page 12-39.
  - □ This instruction can be **interrupted** at a word or row boundary of the destination array. For more information, refer to Section 6.6, <u>Interrupting Graphics Instructions</u>, on page 6-13.
  - □ If CST [DPYCTL] is set, each memory read or write initiated by the PIXBLT generates a **shift register transfer** read or write cycle at the selected address. This operation can be used for bulk memory clears or transfers. (Not all VRAMs support this capability.) For more information, refer to subsection 9.13.4, <u>VRAM Bulk Initialization</u>, on page 9-47.
  - The **status bits** are undefined unless otherwise noted in the individual descriptions.
  - The machine states are not presented because the PIXBLT instructions are complex instructions.

		Destination	Array	
		Linear	XY	
<u>ہ</u> ک	Binary	$\checkmark$	$\checkmark$	
ouro	Linear	$\checkmark$	$\checkmark$	
<u>ه</u>	XY	$\checkmark$	$\checkmark$	

Table 13–6. Summary of Array Types for the PIXBLT Instruction

Table 13–7. Summary of B-File Registers for PIXBLT Instructions

	Format							
Reg.	Name	B, L	B, XY	L, L	L, XY	XY, L	XY, XY	Description
B0	SADDR	Linear	Linear	Linear	Linear	XY	XY	Source pixel array starting address
B1	SPTCH	Linear	Linear	Linear	Linear	Linear	Linear	Source pixel array pitch
B2	DADDR	Linear	XY	Linear	XY	Linear	XY	Destination pixel array starting address
B3	DPTCH	Linear	Linear	Linear	Linear	Linear	Linear	Destination pixel array pitch
B4	OFFSET		Linear		Linear	Linear	Linear	Screen origin (0,0)
B5	WSTART		XY		XY		XY	Window starting corner
B6	WEND		XY		XY		XY	Window ending corner
B7	DYDX	XY	XY	XY	XY	XY	XY	Pixel array dimensions (rows:columns)
B8	COLOR0	Pixel	Pixel					Background expansion color
B9	COLOR1	Pixel	Pixel					Foreground expansion color
B14		res	res	res	res	res	res	Reserved register

Note: PIXBLT L,M,L is discussed independently on page 13-204.

Due to the pipelining of memory writes, the *last* I/O register that you write to may not, in some cases, contain the desired value when you execute the PIXBLT instruction. To ensure that this register contains the correct value for execution, you may want to follow the write to that location with an MWAIT. Refer to Section 4.5.6 on page 4-13 for a description of the potential latency of writes to I/O registers.

	Format						_	
Address	Name	B, L	B, XY	L, L	L, XY	XY, L	XY, XY	Description and Elements
C00000B0h	CONTROL	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	V	V	PPOP–Pixel-processing operations (22 options)
			$\checkmark$		$\checkmark$		$\checkmark$	W – Window clipping or pick operation
		$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	T – Enables transparency
		V	V	V	$\checkmark$	$\checkmark$	V	TM – selects 1 of 3 transpar- ency options
				V	$\checkmark$	V	V	PBH – PIXBLT horizontal di- rection
				V	V	$\checkmark$	V	PBV – PIXBLT vertical direc- tion
C0000130h	CONVSP		V		V	V	V	XY-to-linear conversion (source pitch) Used for source preclipping.
C0000140h	CONVDP		V		$\checkmark$	V	V	XY-to-linear conversion (destination pitch)
C0000150h	PSIZE	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	Pixel size (1,2,4,8,16,32)
C0000160h	PMASK (32 bits)	/	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	V	Plane mask — pixel format

Table 13–8. Summary of I/O Registers for the PIXBLT Instructions

Note: PIXBLT L,M,L is discussed independently on page 13-204.

The PIXBLT instruction has 6 combinations, which are listed below with their corresponding instruction words and descriptions. Note that PIXBLT L,M,L is discussed independently on page 13-204.

PIXBLT B, L	
	<u>15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0</u>
	0 0 0 0 1 1 1 1 1 0 0 0 0 0 0 0 0
Description	This instruction expands, transfers, and processes a binary source pixel array; it operates on 2-dimensional arrays of pixels using linear starting addresses for both the source and the destination. The source pixel array is treated as a 1-bit-per-pixel array. As the PIXBLT proceeds, the source pixels are expanded and then combined with the corresponding destination pixels based on the selected graphics operations.
Source Array	The source pixel array for the expand operation is defined by the contents of the SADDR, SPTCH, and DYDX registers. For more details, refer to Section 12.5, <u>Pixel-Array Instructions</u> , on page 12-8.
13-192	TMS34020 Assembly Language Instruction Set

Source Expansion	The actual values of the source pixels are determined by the interaction of the source array with contents of the COLOR1 and COLOR0 registers. In the expansion operation, a 1 bit in the source array selects a pixel from the COLOR1 register for operation on the destination array. A <b>0</b> bit in the source array selects a COLOR0 pixel for this purpose. The pixels selected from the COLOR1 and COLOR0 registers are those that align directly with their interacted partition in the destination array.
	intended position in the destination array word.

**Destination Array** The location of the destination pixel block is defined by the contents of the DADDR, DPTCH, and DYDX registers. For more details, refer to Section 12.5, <u>Pixel-Array Instructions</u>, on page 12-8.

*Corner Adjust* No corner adjust is performed for this instruction. The pixel transfer simply proceeds in the order of increasing linear addresses.

*Window Checking* Window operations are not enabled for this instruction. The contents of the WSTART and WEND registers are ignored.

#### PIXBLT B, XY binary to XY з This PIXBLT instruction expands, transfers, and processes a binary source Description pixel array with a destination pixel array; it operates on 2-dimensional arrays of pixels using a linear starting address for the source and an XY address for

the destination. The source pixel array is treated as a 1-bit-per-pixel array. As the PIXBLT proceeds, the source pixels are expanded and then combined with the corresponding destination pixels based on the selected graphics operations.

*Source Array* The source pixel array for the expand operation is defined by the contents of the SADDR, SPTCH, DYDX, and (possibly) CONVSP registers. For more details, refer to Section 12.5, <u>Pixel-Array Instructions</u>, on page 12-8.

Source Expansion The actual values of the source pixels are determined by the interaction of the source array with contents of the COLOR1 and COLOR0 registers. In the expansion operation, a 1 bit in the source array selects a pixel from the COLOR1 register for operation on the destination array. A **0** bit in the source array selects a COLOR0 pixel for this purpose. The pixels selected from the COLOR1 and COLOR0 registers are those that align directly with their intended position in the destination array word.

**Destination Array** The location of the destination pixel block is defined by the contents of the DADDR, DPTCH, CONVDP, OFFSET, and DYDX registers. For more details, refer to Section 12.5, <u>Pixel-Array Instructions</u>, on page 12-8.

*Corner Adjust* No corner adjust is performed for this instruction. The transfer executes in the order of increasing linear addresses.

Window Checking	You can use window checking with this instruction by setting the W bits in the CONTROL register to the desired value. If you select window checking mode 1, 2, or 3, the WSTART and WEND registers define the XY starting and ending corners of a rectangular window. For more information, refer to Section 12.7, <u>Window Checking</u> , on page 12-19.
Status Bits	<ul> <li>N Undefined</li> <li>C Undefined</li> <li>Z Undefined</li> <li>V 1 if a window violation occurs, 0 otherwise; undefined if window checking is not enabled (W=00)</li> </ul>
PIXBLT L, L	
linear to linear	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0
Description	The PIXBLT instruction transfers and processes a source pixel array with a destination pixel array; it operates on 2-dimensional arrays of pixels using linear starting addresses for both the source and the destination. As the PIXBLT proceeds, the source pixels are combined with the corresponding destination pixels based on the selected graphics operations.
Source Array	The source pixel array for the processing operation is defined by the contents of the SADDR, SPTCH, and DYDX registers. For more details, refer to Section 12.5, <u>Pixel-Array Instructions</u> , on page 12-8.
Destination Array	The location of the destination pixel array is defined by the contents of the DADDR, DPTCH, and DYDX registers. For more details, refer to Section 12.5, <u>Pixel-Array Instructions</u> , on page 12-8.
Corner Adjust	PBH[[CONTROL]] and PBV[[CONTROL]] govern the direction of the PIXBLT. To set up the corner adjust, refer to subsection 12.5.1.2, <u>Selecting the Starting</u> <u>Corner for a PIXBLT</u> , on page 12-10.
Window Checking	Window operations are not enabled for this instruction. The contents of the WSTART and WEND registers are ignored.
PIXBLT L, XY	
inear to XY	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0
	0 0 0 0 1 1 1 1 0 0 1 0 0 0 0 0
Description	The PIXBLT instruction transfers and processes a source pixel array with a destination pixel array; it operates on 2-dimensional arrays of pixels using a linear starting address for the source array and an XY address for the destination array. As the PIXBLT proceeds, the source pixels are combined with the corresponding destination pixels based on the selected graphics operations.

Source Array	The source pixel array for the processing operation is defined by the contents
	of the SADDR, SPTCH, DYDX, and (possibly) CONVSP registers. For more
	details, refer to Section 12.5, Pixel-Array Instructions, on page 12-8.

**Destination Array** The location of the destination pixel array is defined by the contents of the DADDR, DPTCH, CONVDP, OFFSET, and DYDX registers. For more details, refer to Section 12.5, <u>Pixel-Array Instructions</u>, on page 12-8.

 Corner Adjust
 PBH[[CONTROL]] and PBV[[CONTROL]] govern the direction of the PIXBLT.

 To set up the corner adjust, refer to subsection 12.5.1.2, <u>Selecting the Starting</u>

 Corner for a PIXBLT, on page 12-10.

*Window Checking* You can use window checking with this instruction by setting W[CONTROL] to the desired value. If you select window checking mode 1, 2, or 3, the WSTART and WEND registers define the XY starting and ending corners of a rectangular window. For more information, refer to Section 12.7, <u>Window</u> Checking, on page 12-19.

## Status Bits N Undefined

- C Undefined
- Z Undefined
- V 1 if window violation occurs, 0 otherwise; undefined if window checking is not enabled (W=00<sub>2</sub>)

PIXBLT XY, L — XY to linear	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0
	0 0 0 0 1 1 1 1 0 1 0 0 0 0 0 0
Description	The PIXBLT instruction transfers and processes a source pixel array with a destination pixel array; it operates on 2-dimensional arrays of pixels using an XY starting address for the source pixel array and a linear address for the destination array. As the PIXBLT proceeds, the source pixels are combined with the corresponding destination pixels based on the selected graphics operations.
Source Array	The source pixel array for the processing operation is defined by the contents of the SADDR, SPTCH, CONVSP, OFFSET, and DYDX registers. For more details, refer to Section 12.5, <u>Pixel-Array Instructions</u> , on page 12-8.
Destination Array	The location of the destination pixel array is defined by the contents of the DADDR, DPTCH, DYDX, and (potentially) CONVDP registers. For more details, refer to Section 12.5, <u>Pixel-Array Instructions</u> , on page 12-8.
Corner Adjust	PBH[[CONTROL]] and PBV[[CONTROL]] govern the direction of the PIXBLT. To set up the corner adjust, refer to subsection 12.5.1.2, <u>Selecting the Starting</u> <u>Corner for a PIXBLT</u> , on page 12-10.
Window Checking	Window operations are not enabled for this instruction. The contents of the WSTART and WEND registers are ignored.

# PIXBLT XY, XY Pixel Block Transfer Examples

PIXBLT XY, XY	· · · · · · · · · · · · · · · · · · ·
	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         0       0       0       0       1       1       1       0       1       1       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0
Description	The PIXBLT instruction transfers and processes a source pixel array with a destination pixel array; it operates on 2-dimensional arrays of pixels using XY starting addresses for both the source and destination pixel arrays. As the PIXBLT proceeds, the source pixels are combined with the corresponding destination pixels based on the selected graphics operations.
Source Array	The source pixel array for the processing operation is defined by the contents of the SADDR, SPTCH, CONVSP, OFFSET, and DYDX registers. For more details, refer to Section 12.5, <u>Pixel-Array Instructions</u> , on page 12-8.
Destination Array	The location of the destination pixel array is defined by the contents of the DADDR, DPTCH, CONVDP, OFFSET, and DYDX registers. For more details, refer to Section 12.5, <u>Pixel-Array Instructions</u> , on page 12-8.
Corner Adjust	PBH[[CONTROL]] and PBV[[CONTROL]] govern the direction of the PIXBLT. To set up the corner adjust, refer to subsection 12.5.1.2, <u>Selecting the Starting</u> <u>Corner for a PIXBLT</u> , on page 12-10.
Window Checking	You can use window checking with this instruction by setting W[[CONTROL]] to the desired value. If you select window checking mode 1, 2, or 3, the WSTART and WEND registers define the XY starting and ending corners of a rectangular window. For more information, refer to Section 12.7, <u>Window Checking</u> , on page 12-19.
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V 1 if a window violation occurs, 0 otherwise; unaffected if window clipping not enabled</li> </ul>
Transparency examp	le for PIXBLT B, L
	Before executing the PIXBLT instruction, load the implied operand registers with appropriate values. These PIXBLT examples use the following implied operand setup:
	Register File B:         I/O Registers:           SADDR         =         00002030h         PSIZE         =         0010h           SPTCH         =         0000100h         DADDR         =         0010h           DADDR         =         0000100h         DPTCH         =         0000100h           DYDX         =         00020010h         COLOR0         =         FEDCFEDCh           COLOR1         =         BA98BA98h         E         E         E

Additional implied operand values are listed with each example. For this example, assume that memory contains the following data before instruction execution.

Linear Address					Data			
02000h 02080h 02100h 02180h	xxxxh, xxxxh, xxxxh, xxxxh, xxxxh,	xxxxh, xxxxh, xxxxh, xxxxh, xxxxh,	xxxxh, xxxxh, xxxxh, xxxxh, xxxxh,	1234h, xxxxh, 5678h, xxxxh,	xxxxh, xxxxh, xxxxh, xxxxh, xxxxh,	xxxxh, xxxxh, xxxxh, xxxxh, xxxxh,	xxxxh, xxxxh, xxxxh, xxxxh, xxxxh,	xxxxh xxxxh xxxxh xxxxh xxxxh
33000h	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh
33080h	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh
34000h	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh
34080h	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh

**Example 1** This example uses the *replace* ( $S \rightarrow D$ ) pixel-processing operation. Before instruction execution, PMASK = 0000h and CONTROL = 0000h (T=0, PP=00000).

After instruction execution, memory contains the following values:

Linear Address					Data			
33000h	FEDCh,	FEDCh,	BA98h,	FEDCh,	BA98h,	BA98h,	FEDCh,	FEDCh
33080h	FEDCh,	BA98h,	FEDCh,	FEDCh,	BA98h,	FEDCh,	FEDCh,	FEDCh
34000h	FEDCh,	FEDCh,	FEDCh,	BA98h,	BA98h,	BA98h,	BA98h,	FEDCh
34080h	FEDCh,	BA98h,	BA98h,	FEDCh,	BA98h,	FEDCh,	BA98h,	FEDCh

*Example 2* This example uses transparency with COLOR0 = 00000000h. Before instruction execution, PMASK = 0000h and CONTROL = 0020h (T=1, W=00, PP=00000).

After instruction execution, memory contains the following values:

Linear Address					Data			
33000h	FFFFh,	FFFFh,	BA98h,	FFFFh,	BA98h,	BA98h,	FFFFh,	FFFFh
33080h	FFFFh,	BA98h,	FFFFh,	FFFFh,	BA98h,	FFFFh,	FFFFh,	FFFFh
34000h	FFFFh,	FFFFh,	FFFFh,	BA98h,	BA98h,	BA98h,	BA98h,	FFFFh
34080h	FFFFh,	BA98h,	BA98h,	FFFFh,	BA98h,	FFFFh,	BA98h,	FFFFh

## Window-clipping example for PIXBLT B, XY

Before executing the PIXBLT instruction, load the implied operand registers with appropriate values. These PIXBLT examples use the following implied operand setup:

I/O Registers:

#### **Register File B:**

SADDR	=	00002010h	PSIZE	=	0008h
SPTCH	=	0000010h	CONVSP	=	001Bh
DADDR	=	00300022h	CONVDP	=	0013h
DPTCH	=	00001000h			
OFFSET	=	00010000h			
WSTART	Ξ	0000026h			
WEND	=	00400050h			
DYDX	=	00040010h			
COLOR0	=	0000000h			
COLOR1	=	7C7C7C7Ch			

Additional implied operand values are listed with each example. For this example, assume that memory contains the following data before instruction execution.

Linear Data Address 2000h xxxxh, 0123h 4567h, 89ABh, CDEFh, xxxxh, xxxh, xxxh 40000h to 43200h FFFFh

Example 1

This example uses the *replace* (S  $\rightarrow$  D) pixel-processing operation. Before instruction execution, PMASK = 0000h and CONTROL = 0000h (T=0, W=00, PP=00000).

After instruction execution, memory contains the following values:

Linear Address					Data			
41100h	FFFFh,	7C7Ch,	007Ch,	7C00h,	007Ch,	007Ch,	007Ch,	0000h
41180h	007Ch,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh
42100h	FFFFh,	7C7Ch,	7C00h,	7C00h,	7C00h,	007Ch,	7C00h,	0000h
42180h	7C00h,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh
43100h	FFFFh,	7C7Ch,	7C7Ch,	7C00h,	7C7Ch,	007Ch,	7C7Ch,	0000h
43180h	7C7Ch,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh,	FFFFh

#### XY Addressing

											ΧA	ddro	ess									
Y		2	2	2	2	2	2	2	2	2	2	2	2	2	2	2	2	3	3	3	3	3
		0	1	2	3	4	5	6	7	8	9	Α	В	С	D	Ε	F.	0	1	2	3	4
Α																						
d	30	FF	FF	7C	7C	00	00	00	7C	00	00	7C	00	00	00	00	00	00	00	FF	FF	FF
d																						
r	31	FF	FF	7C	7C	7C	00	00	7C	7C	00	7C	00	7C	00	00	00	7C	00	FF	FF	FF
е																						
S	32	FF	FF	7C	7C	00	7C	00	7C	00	7C	7C	00	00	7C	00	00	00	7C	FF	FF	FF
s	33	FF	FF	7C	7C	7C	7C	00	7C	7C	7C	7C	00	7C	7C	00	00	7C	7C	FF	FF	FF

*Example 2* This example uses window operation 3 (clipped destination). Before instruction execution, PMASK = 0000h and CONTROL = 00C0h (T=0, W=11, PP=00000).

After instruction execution, memory contains the following values:

----

XY /	Addre	ssing	
------	-------	-------	--

											ΧA	ddr	ess									
Y		2 0	2 1	2 2	2 3	2 4	2 5	2 6	2 7	2 8	2 9	2 A	2 B	2 C	2 D	2 E	2 F	3 0	3 1	3 2	3 3	3 4
A d	30	FF	FF	FF	FF	FF	FF	00	7C	00	00	7C	00	00	00	00	00	00	00	FF	FF	FF
r e	31	FF	FF	FF	FF	FF	FF	00	7C	7C	00	7C	00	7C	00	00	00	7C	00	FF	FF	FF
S S	32	FF	FF	FF	FF	FF	FF	00	7C	00	7C	7C	00	00	7C	00	00	00	7C	FF	FF	FF
•	33	FF	FF	FF	FF	FF	FF	00	7C	7C	7C	7C	00	7C	7C	00	00	7C	7C	FF	FF	FF

#### Pixel-processing example for PIXBLT L, L -

Before executing the PIXBLT instruction, load the implied operand registers with appropriate values. These PIXBLT examples use the following implied operand setup:

Register File B:		
SADDR	=	00002004h
SPTCH	=	00000080h
DADDR	=	00002228h
DPTCH	=	00000080h
OFFSET	=	00000000h
DYDX	=	0000200Dh

# I/O Registers:

PSIZE = 0004h

Additional implied operand values are listed with each example. For this example, assume that memory contains the following data before instruction execution.

Linear Address					Data			
02000h	000xh,	1111h	2222h,	xx33h,	xxxxh,	xxxxh,	xxxxh,	xxxxh
02080h	000xh,	1111h,	2222h,	xx33h,	xxxxh,	xxxxh,	xxxxh,	xxxxh
02100h	xxxxh,	xxxxh,	xxxxh,	xxxxh	xxxxh,	xxxxh,	xxxxh,	xxxxh
02180h	xxxxh,	xxxxh,	xxxxh,	xxxxh	xxxxh,	xxxxh,	xxxxh,	xxxxh
02200h	xxxxh,	xxxxh,	FFxxh,	FFFFh,	FFFFh,	xFFFh,	xxxxh	xxxxh
02280h	xxxxh,	xxxxh,	FFxxh,	FFFFh,	FFFFh,	xFFFh,	xxxxh	xxxxh
02300h	xxxxh,	xxxxh,	xxxxh,	xxxxh	xxxxh,	xxxxh,	xxxxh,	xxxxh

This example uses the replace (S  $\rightarrow$  D) pixel-processing operation. Before in-Example 1 struction execution, PMASK = 0000h and CONTROL = 0000h (T=0, W=00, PP=00000).

After instruction execution, memory contains the following values:

Linear Address					Data			
02000h	000xh,	1111h,	2222h,	xx33h,	xxxxh,	xxxxh,	xxxxh,	xxxxh
02080h	000xh,	1111h,	2222h,	xx33h,	xxxxh,	xxxxh,	xxxxh,	xxxxh
02100h	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh
02180h	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh
02200h	xxxxh,	xxxxh,	FFxxh,	EEEFh,	DDDEh,	xCCDh,	xxxxh,	xxxxh
02280h	xxxxh,	xxxxh,	00xxh,	1 <b>11</b> 0h,	2221h,	x332h,	xxxxh,	xxxxh
02300h	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh

Example 2

This example uses the  $(D-S) \rightarrow D$  pixel-processing operation. Before instruction execution, PMASK = 0000h and CONTROL = 4800h T=0, W=00, PP=10010).

After instruction execution, memory contains the following values:

Linear Address					Data			
02000h	000xh,	1111h,	2222h,	xx33h,	xxxxh,	xxxxh,	xxxxh,	xxxxh
02080h	000xh,	1111h,	2222h,	xx33h,	xxxxh,	xxxxh,	xxxxh,	xxxxh
02100h	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh
02180h	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh
02200h	xxxxh,	xxxxh,	OFFxxh	, 111Fh,	2221h,	x332h,	xxxxh,	xxxxh
02280h	xxxxh,	xxxxh,	OFFxxh	, 111Fh,	2221h,	x332h,	xxxxh,	xxxxh
02300h	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh,	xxxxh

Plane mask example for L, XY -

Before executing the PIXBLT instruction, load the implied operand registers with appropriate values. This PIXBLT examples uses the following implied operand setup:

#### **Register File B:**

SADDR	=	00002004h
SPTCH	=	0000080h
DADDR	- =	00520007h
DPTCH	=	00000100h
OFFSET	=	00001000h
WSTART	=	0030000Ch
WEND	=	00530014h
DYDX	=	00030016h

#### I/O Registers:

PSIZE	=	0004h
PMASK	=	0000h
CONVDP	=	0017h
CONTROL	=	0000h
(W=00, T=0,	PF	P=00000)

For this example, assume that memory contains the following data before instruction execution.

						Li	nea	ar Ass														Da	ta										
						02 02 02	00 08 10	0h 0h 0h	5	32 32 32	10 10 10	h, h, h,	76 76 76	54  54  54	h, h, h,	BA BA BA	\98 \98 \98	h, h, h,	FE FE	DC DC DC	h, h, h,	32 <sup>-</sup> 32 <sup>-</sup> 32 <sup>-</sup>	10h 10h 10h	, , ,	76 76 76	54  54  54	ז, ז, ז,	E	8A9 8A9 8A9	8h 8h 8h	, F , F , F	ED ED	Ch Ch Ch
						15 15	20 48	0h 0h	to	88	88	h																					
Example	9					Th sti Pl	nis ruc >=(	exa tio 000	am n e 000	ple exe 0).	e us cul	ses tior	the n, F	e re PM	epl AS	ace K :	ə (S = 7	3 77	> D 7H	) pi an	xe d (	l-pi CO	roc NT	es R(	sir OL	ng c . =	ope 00	erat 00ł	tior n (	η. Ε Γ=(	8ef ), \	ore N=	in- 00,
						Af	ter	in	stru	ucti	ion	ex	ec	uti	on,	m	em	ory	y co	onta	ain	s t	he	fol	llov	win	g٧	alu	les	:			
						Li	nea	ar														Da	ta										
						Ac 15 15 15	ddr 520 530 540	<b>es:</b> 0h, 0h 0h	5	88 88 88	88 88 88	h, h, h,	18 18 18	88  88  88	h, h, h,	54 54 54	32  32  32	ז, ז, ז,	98 98 98	76h 76h 76h	1, 1, 1,		CBA CBA CBA	.h, .h, .h,	10 10 10	FE	h, h, h,	5 5 5	43 43 43	2h, 2h, 2h,	8	388( 388( 388(	3h 3h 3h
XY Addı	ressin	g												,		44		~															
Y		0	0	0	0	0	0	0	0	0	0	0	0	°	νд 0	0	0	1	1	1	1	1	1	1	1	1	1	1	1	1 ·	1 1	1	
•		Ō	1	2	3	4	5	6	7	8	9	Α	В	С	D	Ε	F	0	1	2	3	4	5	6	7	8	9	в	С	С	D	E	F
A d d	52	8	8	8	8	8	8	8	1	2	3	4	5	6	7	8	9	A	в	С	D	Е	F	0	1	2	3	4	5	88	38	38	
r	53	8	8	8	8	8	8	8	1	2	3	4	5	6	7	8	9	А	В	С	D	Е	F	0	1	2	3	4	5	8 8	3 8	38	
e S S	54	8	8	8	8	8	8	8	1	2	3	4	5	6	7	8	9	A	В	С	D	Е	F	0	1	2	3	4	5	88	38	38	

Example for PIXBLT XY, L -

Before executing the PIXBLT instruction, load the implied operand registers with appropriate values. This PIXBLT examples use the following implied operand setup:

Register File B:			I/O Registers:	;	
SADDR	=	00400001h	PSIZE	=	0004h
SPTCH	=	00000080h	CONVSP	=	0016h
DADDR	=	00002228h	CONVDP	=	0016h
DPTCH	=	00000080h	PMASK	=	0000h
OFFSET	=	00010000h	CONTROL	=	0000h
WSTART	Ξ	00300009h	(W=00, T=0,	PF	P=00000)
WEND	=	00420012h			
DYDX	=	00030016h			

For this example, assume that memory contains the following data before instruction execution.

Linear Address					Data			
02000h	000xh,	1111h	2222h,	xx33h,	xxxxh,	xxxxh,	xxxxh,	xxxxh
02080h	000xh,	1111h,	2222h,	xx33h,	xxxxh,	xxxxh,	xxxxh,	xxxxh
02100h	xxxxh,	xxxxh,	xxxxh,	xxxxh	xxxxh,	xxxxh,	xxxxh,	xxxxh
02180h	xxxxh,	xxxxh,	xxxxh,	xxxxh	xxxxh,	xxxxh,	xxxxh,	xxxxh
02200h	xxxxh,	xxxxh,	FFxxh,	FFFFh,	FFFFh,	xFFFh,	xxxxh	xxxxh
02280h	xxxxh,	xxxxh,	FFxxh,	FFFFh,	FFFFh,	xFFFh,	xxxxh	xxxxh
02300h	xxxxh,	xxxxh,	xxxxh,	xxxxh	xxxxh,	xxxxh,	xxxxh,	xxxxh

Example

This example uses transparency. Before instruction execution, PMASK = 0000h and CONTROL = 0200h (T=1,W=00, PP=00000).

After instruction execution, memory contains the following values:

Linear Address					Data			
02000h	000xh,	1111h,	2222h,	xx33h,	xxxxh,	xxxxh,	xxxxh,	xxxxh
02000h	000xh,	1111h,	2222h,	xx33h,	xxxxh,	xxxxh,	xxxxh,	xxxxh
02100h	xxxxh,	xxxxh						
02180h	xxxxh,	xxxxh						
02200h	xxxxh	xxxxh,	FFxxh,	111Fh	2221h,	x332h,	xxxxh,	xxxxh
02280h	xxxxh,	xxxxh,	FFxxh,	111Fh,	2221h,	x332h,	xxxxh,	xxxxh
02300h	xxxxh,	xxxxh						

#### Example for PIXBLT XY, XY -

Before executing the PIXBLT instruction, load the implied operand registers with appropriate values. These PIXBLT examples use the following implied operand setup:

Register File B:			I/O Registers	:	
SADDR	=	00200004h	PSIZE	=	0004h
SPTCH	=	00000200h	CONVSP	=	0016h
DADDR	=	00410004h	CONVDP	=	0016h
DPTCH	=	00000200h	PMASK	=	0000h
OFFSET	=	00010000h	CONTROL	=	0000h
WSTART	=	00300009h	(W=00, T=00	), P	P=00000)
WEND	=	00420012h			
DYDX	=	00030016h			

For this example, assume that memory contains the following data before instruction execution.

						Li	ne: ddr	ar 'es:	s													Da	ata										
						04 04 04 18	100 120 140 320	0h 0h 0h 0h	to	32 32 32	210 210 210	h, h, h, h	76 76 76	54 54 54	h, h, h,	BA BA BA	498 498 498	lh, lh, lh,	FE FE		Ch, Ch, Ch,	32 32 32	210 210 210	h, h, h,	76 76 76	654 654 654	h, h, h,	E E E	3A9 3A9 3A9	981 981 981	ו, ו, ו,	FE FE	EDCh EDCh EDCh
Exampl	le					TI in Pl	nis stri P=	ex uct 10(	am ion 001	nple i ex	e u kec	ses uti	s th on,	ne Pl	(D MA	AC .Sk	)D8 ( =	S S 00	5) - 00	→[ ha	) p nd	ixe C(	el-p ON	roc TF	ces IOI	sir _ =	ng ( 44	ope 100	era h (	tio T=	n. :0,	Be W	efore /=00,
XY Add	Iressi	ng				A	fter	' in	strı	uct	ion	ex	ec	uti	on,	m	em	or	ус	ont	tair	าร	the	fo	llov	win	١g \	/alı	ues	s:			
		-										X	Ad	dre	ess																		
Y		0 0	0 1	0 2	0 3	0 4	0 5	0 6	0 7	0 8	0 9	0 A	0 B	0 C	0 D	0 E	0 F	1 0	1 1	1 2	1 3	1 4	1 5	1 6	1 7	1 8	1 9	1 B	1 C	1 C	1 D	1 E	1 : F
Α																																	
d	41	3	3	3	3	7	8	9	А	В	С	D	Е	F	F	F	F	3	4	5	6	7	8	9	А	В	С	3	3	3	3	3	3
d																																	
r	41	З	З	3	З	7	8	9	А	В	С	D	Е	F	F	F	F	3	4	5	6	7	8	9	А	В	С	3	3	3	3	3	3
е		~	~		~	_	~	_		_	~	_	_	_	_	_	_	~		_	~	_	~	~		_	~	~	~	~	~	~	~
s s	41	3	3	3	3	7	8	9	A	в	C	U	E	۲	F	F	F	3	4	5	6	1	8	9	А	В	C	3	3	3	3	3	3

## Syntax PIXBLT L, M, L

*Execution* Linear pixel array to linear pixel array using a binary mask array

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
0	0	0	0	1	1	1	0	0	0	0	1	0	1	1	1

**Description** This instruction transfers a pixel array from the source location specified by a linear address in SADDR to the destination location specified by a linear address in DADDR, which is under the control of the binary mask pixel array specified by a linear address in MADDR. The array dimensions are in DYDX.

Each source pixel is combined with the destination pixel according to the selected pixel-processing option. The resulting pixel can then be written to the destination pixel only if the corresponding bit in the mask array is a *1*.

#### Implied Operands

Instruction Words

Register		Name	Format	Description
B0	†	SADDR	Linear	Source pixel array address
B1		SPTCH	Linear	Source pixel array pitch
B2	†	DADDR	Linear	Destination pixel array address
B3		DPTCH	Linear	Destination pixel array pitch
B7		DYDX	b:a	Dimensions of drawn rectangle
B10	†	MADDR	Linear	Mask pixel array address
B11		MPTCH	Linear	Mask array pitch
B12 & B1	4 †	Reserved	Temporary R	egisters

<sup>†</sup> These registers are changed by instruction execution

Address	Name	Description and Elements (Bits)
C00000B0h	CONTROL	<ul> <li>PPOP Pixel-processing operations (22 options)</li> <li>T Transparency operation</li> <li>TM Sets transparency mode</li> <li>PBH PIXBLT horizontal direction</li> <li>PBV PIXBLT vertical direction</li> </ul>
C0000150h	PSIZE	Pixel size (1,2,4,8,16,32)
C0000160h	PMASK (32 bits)	Plane mask — pixel format

Corner Adjust

PBH[CONTROL] and PBV[CONTROL] govern the direction of the PIXBLT. To set up the corner adjust, refer to subsection 12.5.1.2, <u>Selecting the Starting</u> <u>Corner for a PIXBLT</u>, on page 12-10.

*Window Checking* Window operations are not enabled for this instruction. The contents of the WSTART and WEND registers are ignored.

Pixel Processing	Select a pixel processing option for this instruction by setting PPOP [CON-
	TROL]. The pixel processing option is applied to pixels as they are processed
	with the destination array. Note that the data is read through the plane mask
	and then processed. There are 16 Boolean and 6 arithmetic operations; the
	default case at reset is the <i>replace</i> ( $S \rightarrow D$ ) operation. The 6 arithmetic opera-
	tions do not operate with pixel sizes of 1 or 2 bits per pixel. For more informa-
	tion, refer to Section 12.8, Pixel Processing, on page 12-27.

TransparencyYou can enable transparency by setting T[CONTROL] to 1. The TMS34020<br/>supports 3 transparency modes; TM[CONTROL] selects 1 of 3 transparency<br/>options. For more information, refer to Section 12.9, <a href="https://www.transparency.com">Transparency.com</a>, on page<br/>12-36.

Plane MaskingThe plane mask is enabled for this instruction. For more information, refer to<br/>Section 12.10, Plane Masking, on page 12-39.

Machine States complex instruction

#### Status Bits

- N UndefinedC Undefined
- **Z** Undefined
- V Undefined
- **PIXT Instructions** The **PIXT** instruction transfers a pixel from one location to another. The following list describes characteristics common to all **PIXT** instructions.
  - Rs and Rd must be in the same register file.
  - The **plane mask** is enabled for all PIXT instructions. For more information, refer to Section 12.10, <u>Plane Masking</u>, on page 12-39.
  - The **status bits** are undefined unless otherwise noted in the individual descriptions.
  - For machine states information, refer to Section 15.1 on page 15-2.

Section 12.3, <u>Single-Pixel Instructions</u>, on page 12-6 provides additional information about the PIXBLT instructions.

Table 13–9. Summary of Operand Formats for the PIXT Instructions

		1	Destination Pix	cel
		Rd	*Rd	*Rd.XY
9	Rs		$\checkmark$	$\checkmark$
ourc	*Rs	$\checkmark$	$\checkmark$	
о <u>н</u> ——	*Rs.XY	$\checkmark$		$\checkmark$

Table 13–10. Summary of B-File Registers for PIXT Instructions

			Format		
Reg.	Name	Rs, *Rd. <b>XY</b>	*Rs. <b>XY</b> , Rd	*Rs. <b>XY</b> , *Rd. <b>XY</b>	Description
B1	SPTCH		Linear	Linear	Source pixel array pitch
B3	DPTCH	Linear	Linear	Linear	Destination pixel array pitch
B4	OFFSET	Linear	Linear	Linear	Screen origin (0,0)
B5	WSTART	XY		XY	Window starting corner
B6	WEND	XY		XY	Window ending corner

Due to the pipelining of memory writes, the *last* I/O register that you write to may not, in some cases, contain the desired value when you execute the PIXBLT instruction. To ensure that this register contains the correct value for execution, you may want to follow the write to that location with an MWAIT (page 13-177). Refer to Section 4.5.6 on page 4-13 for a description of the potential latency of writes to I/O registers.

Address	Name	Rs, *Rd	Rs, *Rd.XY	*Rs, Rd	*Hs, *Hd	*Hs.XY, Hd	YX. *Hd.XY	Description and Elements
C00000B0h	CONTROL	√	V		$\checkmark$		V	PPOP — Pixel processing operations (22 options)
			$\checkmark$				$\checkmark$	W — Window clipping or pick operation
		$\checkmark$	$\checkmark$		$\checkmark$		$\checkmark$	T — Enables transparency
		$\checkmark$	$\checkmark$		$\checkmark$		$\checkmark$	TM — Selects transparency options
C0000130h	CONVSP					$\checkmark$	$\checkmark$	XY-to-linear conversion (source pitch) Used for source preclipping.
C0000140h	CONVDP		V				$\checkmark$	XY-to-linear conversion (destination pitch)
C0000150h	PSIZE	$\checkmark$	$\checkmark$	$\checkmark$	√	$\checkmark$	$\checkmark$	Pixel size (1,2,4,8,16,32)
C0000160h	PMASK (32 bits)	1	$\checkmark$	$\checkmark$	V	$\checkmark$	$\checkmark$	Plane mask — pixel format

# Table 13–11. Summary of I/O Registers for the PIXT Instructions

The **PIXT** instruction has 6 addressing modes, which are listed below with their corresponding instruction words and descriptions.

# PIXT *Rs,* \**Rd* register to memory

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	1	1	1	1	0	0		F	Rs		R		F	۱d	

The source pixel is the 1, 2, 4, 8, 16, or 32 LSBs of the source register, depending on the pixel size specified in the PSIZE register. The destination register contains a linear address; the source pixel is transferred to this memory location.

You can select a **pixel processing** option to use with this instruction. For more information, refer to Section 12.8, <u>Pixel Processing</u>, on page 12-27.

Window checking cannot be used with this instruction.

You can enable **transparency** by setting T[[CONTROL]] to 1. The TMS34020 supports 3 transparency modes; TM[[CONTROL]] selects 1 of 3 transparency options. At reset, the default case for transparency is *off.* For more information, refer to Section 12.9, <u>Transparency</u>, on page 12-36.

# PIXT *Rs,* \**Rd*.XY register to memory

	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	1	1	1	1	0	0	0		F	Rs		R		F	łd	
	The ing c cont the ` addr	sourc on the ains a Y val	ce pix e pixe an XN ue o speci	cel is f el siz 7 add ccup fied i	the 1 e spe ress ies t n Rd	, 2, 4, ecifie ; the 2 he 16	8, 16 d in t X valı 5 MS	6, or : he P ue oc Bs.	32 LS SIZE ccupi The	Bs o regi es th sour	f the s ster. e 16 I ce piz	soure The _SBs xel is	ce reg desti s of th s mo	gister natio ne reg ved 1	r, dep en reg gister to the	end- jister , and e XY
	You to th <u>Che</u>	can u le de cking	se <b>w</b> sired , on	<b>indo</b> valu page	<b>w ch</b> ie. F 12-1	ecki or m 9.	<b>ng</b> wi ore ii	ith th nforr	is ins natio	truct n, re	ion by fer to	y set o Sec	ting V Stion	V[[C( 12.7	ONTF , <u>Wir</u>	≀OL]] <u>ıdow</u>
	You infor	can s matic	elect on, re	a <b>pi</b> > efer to	<b>cel p</b> i o Sec	r <b>oce</b> s	<b>ssing</b> 12.8,	opt <u>Pixe</u>	ion to <u>el Pro</u>	use ocess	with t sing, o	his ir on pa	nstruc age 1	ction. 12-27	Forr	nore
	You supp optic 12-3	can e oorts : ons. F 6.	nabl 3 trar For m	e <b>tra</b> i nspar iore i	n <b>spa</b> ency nforr	renc moc natio	y by les; T n, ref	setti ™[[( er to	ng T[ CONT Sec	[CON FROI tion	NTRC _]] sel 12.9,	DL]] to ects <u>Trar</u>	o 1. T 1 of ( Ispar	he T 3 tran ency	MS34 Ispar , on j	4020 ency bage
Status Bits	N C Z V	Jnaff Jnaff Jnaff 1 if pix = 0.	ecteo ecteo ecteo cel is	d d d outsi	de th	ewin	dow	and	W = 1	, 2, 3	; <i>0</i> ot	herw	ise. l	Jnaff	ected	1 if W
PIXT * <i>Rs, Rd</i>	<b></b>															
memory to register	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	1	1	1	1	1	0	1		F	₹s		R	<u>]</u>	F	ld	
	The ferre right in th	sourc d inte -justif e PSI	e reg o the fied a ZE r	gister des ind ze egist	cont tinati ero-e er.	ains ion re xtend	a line egiste ded to	arac er. V 0321	ldres Vhen bits, a	s; the the iccor	e pixe pixel ding t	l at ti is m o the	nis ac novec e pixe	ldres 1 intc 1 size	s is tr Rd, spec	ans- it is ified

Window checking cannot be used with this instruction. The W bits are ignored.

Pixel processing cannot be used with this instruction.

Transparency cannot be used with this instruction.

# Status Bits

- N Unaffected
- C Unaffected
- Z Unaffected
- V Unaffected

# PIXT \*Rs, \*Rd memory to memory

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	1	1	1	1	1	0		F	٦s		R		R	ld	

The source and destination registers both contain linear addresses. The address in Rs is the address of the source pixel; the pixel is moved into the address in Rd.

You can select a **pixel processing** option to use with this instruction. For more information, refer to Section 12.8, <u>Pixel Processing</u>, on page 12-27.

Window checking cannot be used with this instruction.

You can enable **transparency** by setting T[[CONTROL]] to 1. The TMS34020 supports 3 transparency modes; TM[[CONTROL]] selects 1 of 3 transparency options. For more information, refer to Section 12.9, <u>Transparency</u>, on page 12-36.

# PIXT \*Rs.XY, Rd

memory to register

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	1	1	1	0	0	1		F	٦s		R		R	d	

The source register contains an XY address; the X value occupies the 16 LSBs of the register, and the Y value occupies the 16 MSBs. The address in Rs is the address of the source pixel; this pixel is moved into the destination register. When the pixel is moved into Rd, it is right-justified and zero-extended to 32 bits according to the pixel size specified in the PSIZE register.

Pixel processing cannot be used with this instruction.

Transparency cannot be used with this instruction.

# Status Bits

# N Unaffected

- C Unaffected
- Z Unaffected
- V Unaffected

# PIXT \**Rs*.XY, \**Rd*.XY

memory to memory

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
1	1	1	1	0	1	0		F	٩s		R		R	d	

The source and destination registers both contain XY addresses; the X value occupies the 16 LSBs of the register, and the Y value occupies the 16 MSBs. Rs contains the address of the source pixel; Rd contains the address where the pixel is moved.

You can use **window checking** with this instruction by setting W[[CONTROL]] to the desired value. For more information, refer to Section 12.7, <u>Window</u> <u>Checking</u>, on page 12-19.

You can select a **pixel processing** option to use with this instruction. For more information, refer to Section 12.8, <u>Pixel Processing</u>, on page 12-27.

You can enable **transparency** by setting T[[CONTROL]] to 1. The TMS34020 supports 3 transparency modes; TM[[CONTROL]] selects 1 of 3 transparency options. At reset, the default case for transparency is *off.* For more information, refer to Section 12.9, <u>Transparency</u>, on page 12-36.

#### Status Bits

- N Unaffected
- C Unaffected
- Z Unaffected
- V 1 if the pixel lies outside the window and W=1, W=2, or W=3; 0 otherwise. Unaffected if W=0.

Section 12.3, <u>Single-Pixel Instructions</u>, on page 12-6 provides additional information about the PIXT instructions.

### PIXT examples -

Example 1

PIXT A0, \*A1

	<b>Before</b>					<u>Af</u>	<u>ter</u>	
	A0	A1	@20500H	PSIZE	PP	Т	PMASK	@20500h
1)	0000FFFFh	00020500h	0000h	0001h	00000	0	0000h	0001h
1)	0000FFFFh	00020500h	0000h	0002h	00000	0	0000h	0003h
1)	0000FFFFh	00020500h	0000h	0004h	00000	0	0000h	000Fh
1)	0000FFFFh	00020500h	0000h	0008h	00000	0	0000h	00FFh
1)	0000FFFFh	00020500h	0000h	0010h	00000	0	0000h	FFFFh
1)	00000006h	00020508h	0000h	0004h	00000	0	0000h	0600h
2)	00000006h	00020508h	0300h	0004h	01010	0	0000h	0500h
3)	00000006h	00020508h	0100h	0004h	00001	0	0000h	0000h
4)	00000006h	00020508h	0100h	0004h	00001	0	0000h	0100h
5)	0000006h	00020508h	0000h	0004h	00000	0	AAAAh	0400h

#### Notes:

- 1) S replaces D
- 2) (S XOR D) = 0, replaces D
- 3) (S AND D) = 0, transparency is off, D is replaced
- 4) (S + D) = 0, transparency is on, D is not replaced
- 5) S replaces unmasked bit of D

#### Example 2

Before executing a PIXT instruction, load the implied operand registers with appropriate values. These PIXT examples use thefollowing implied operand setup:

#### Register File B: DPTCH OFFSET

WSTART

WEND

= 00000800h = 00000000h = 00300020h = 00500142h

# I/O Registers: CONVDP = 0014h

TMS34020 Assembly Language Instruction Set

#### PIXT A0, \*A1.XY

Be	fore	<u>After</u>							
	A0	A1	@20500H	PSIZE	PP	W	Т	PMASK	@20500h
1)	0000FFFFh	00400500h	0000h	0001h	00000	00	0	0000h	0001h
1)	0000FFFFh	00400280h	0000h	0002h	00000	00	0	0000h	0003h
1)	0000FFFFh	00400140h	0000h	0004h	00000	00	0	0000h	000Fh
1)	0000FFFFh	004000A0h	0000h	0008h	00000	00	0	0000h	00FFh
1)	0000FFFFh	00400050h	0000h	0010h	00000	00	0	0000h	FFFFh
1)	00000006h	00400142h	0000h	0004h	00000	00	0	0000h	0600h
2)	0000006h	00400142h	0300h	0004h	01010	00	0	0000h	0500h
3)	00000006h	00400142h	0100h	0004h	00001	00	0	0000h	0000h
4)	00000006h	00400142h	0100h	0004h	00001	00	0	0000h	0100h
5)	00000006h	00400142h	0000h	0004h	00000	00	0	AAAAh	0400h
6)	00000006h	00400142h	0000h	0004h	00000	00	0	0000h	0600h
7)	00000006h	00400143h	0000h	0004h	00000	00	0	0000h	0000h
8)	0000006h	00400143h	0000h	0004h	00000	00	0	0000h	0000h

XY Address in A1 = Linear Address 20500h

## Notes:

- 1) S replaces D
- 2) (S XOR D) = 0, replaces D
- 3) (S AND D) = 0, transparency is off, D is replaced
- 4) (S + D) = 0, transparency is on, D is not replaced
- 5) S replaces unmasked bit of D
- 6) Window Option = 3, D inside window, S replaces D
- Window Option = 3, D outside window, D not replaced, V bit set in status register
- 8) Window Option = 2, D outside window, D not replaced, WV interrupt

# Example 3

Assume that memory contains the following values:

Address	Data
@20500h	0FFFFh
@20510h	3333h

### PIXT \*A0,A1

<u>Before</u>			<u>After</u>
A0	PSIZE	PMASK	A1
00020500h	0001h	0000h	00000001h
00020500h	0001h	FFFFh	00000000h
00020500h	0002h	0000h	00000003h
00020500h	0002h	5555h	00000002h
00020500h	0004h	0000h	0000000Fh
00020510h	0004h	9999h	00000002h
00020500h	0008h	0000h	000000FFh
00020510h	0008h	5454h	00000023h
00020500h	0010h	0000h	0000FFFFh
00020500h	0010h	BA98h	00004567h
00020510h	0010h	BA98h	00000123h

### PIXT \*A0,\*A1

D . . . . . .

Belore		tore	Atter							
		AO	A1	@20500H	PSIZE	PP	т	PMASK	@20500h	20510h
	1)	00020500h	00020508h	000Fh	0001h	00000	0	0000h	010Fh	XXXX
	1)	00020500h	00020508h	000Fh	002h	00000	0	0000h	030Fh	XXXX
	1)	00020500h	00020508h	000Fh	0004h	00000	0	0000h	0F0Fh	XXXX
	1)	00020500h	00020508h	00EFh	0008h	00000	0	0000h	EFEFh	XXXX
	1)	00020500h	00020508h	1234h	0010h	00000	0	0000h	3434h	xx12h
	2)	00020500h	00020508h	030Fh	0004h	01010	0	0000h	0C0Fh	XXXX
	3)	00020500h	00020508h	010Eh	0004h	00001	0	0000h	000Eh	XXXX
	4)	00020500h	00020508h	020Eh	0004h	00001	0	0000h	020Eh	XXXX
	5)	000020500h	100020508h	000Fh	0004h	00000	0	AAAAh	050Fh	XXXX

#### Notes:

1) S replaces D

. . . . .

- 2) (S XOR D) replaces D
- 3) (S AND D) = 0, transparency is off, D is replaced
- 4) (S + D) = 0, transparency in on, D not replaced
- 5) S replaces unmasked bits of D

#### Example 4

These PIXT examples use the following implied operand setup.

Register File B	3:	I/O Registers:
DPTCH	=800h	CONVSP = 0014h
OFFSET	=0000000h	

Assume that memory address @20500h contains CF3Fh before instruction execution.

#### PIXT \*A0.XY,A1

		<u>After</u>
PSIZE	PMASK	A1
0001h	0000h	0000001h
0001h	FFFFh	00000000h
0002h	0000h	0000003h
0002h	AAAAh	0000001h
0004h	0000h	0000000Fh
0004h	9999h	0000006h
0008h	0000h	0000003Fh
0008h	8989h	00000036h
0010h	0000h	0000CFCFh
0010h	7310h	00008C2F
	<b>PSIZE</b> 0001h 0002h 0002h 0004h 0004h 0008h 0008h 0008h 0010h	PSIZE         PMASK           0001h         0000h           0001h         FFFFh           0002h         0000h           0002h         AAAAh           0004h         0000h           0004h         9999h           0008h         0000h           0008h         8989h           0010h         7310h

#### Note:

The XY addresses stored in register A1 in these examples translate to the linear memory address 20500h. The pitch of the line source was not changed for any of these examples

# Example 5

These PIXT examples use the following implied operand setup.

Register File B:			I/O Registe	rs:	
SPTCH	=	800h	CONVSP	=	0014h
DPTCH	=	800h	CONVDP	=	0014h
OFFSET	=	00000000h			
WSTART	=	00300020h			
WEND	=	00500142h			

#### PIXT \*A0.XY,\*A1.XY

<u>Before</u>		<u>After</u>								
	A0	A1	@20500H	PSIZE	PP	W	Т	PMASK	@20500h	@20510h
1)	00400500h	00400508h	000Fh	0001h	00000	00	0	0000h	010Fh	XXXX
1)	00400280h	00400284h	000Fh	0002h	00000	00	0	0000h	030Fh	XXXX
1)	00400140h	0400142h	000Fh	0004h	00000	00	0	0000h	0F0Fh	XXXX
1)	004000A0h	004000A1h	000Fh	0008h	00000	00	0	0000h	EFEFh	XXXX
1)	0040005Fh	00400051h	00EFh	0010h	00000	00	0	0000h	CDEFh	CDEFh
2)	00400050h	00400142h	0306h	0004h	01010	00	0	0000h	0506h	XXXX
3)	00400140h	00400142h	0106h	0004h	00001	00	0	0000h	0006h	XXXX
4)	00400140h	00400142h	0106h	0004h	10001	00	0	0000h	0106h	XXXX
5)	00400140h	00400142h	0006h	0004h	00001	00	0	0000h	0406h	XXXX
6)	00400140h	00400142h	0006h	0004h	00000	11	0	AAAAh	0606h	XXXX
7)	00400140h	00400142h	0006h	0004h	00000	11	0	0000h	0006h	XXXX
8)	00400140h	00400143h	0006h	0004h	00000	10	0	0000h	0006h	хххху

XY Address in A1 = Linear Address 20500h

# Notes:

- 1) S replaces D
- 2) (S XOR D) replaces D
- 3) (S AND D) = 0, transparency is off, D is replaced
- 4) (S + D) = 0, transparency in on, D not replaced
- 5) S replaces unmasked bits of D
- 6) Window Option = 3, D inside window, S replaces D
- 7) Window Option = 3, D outside window, D not replaced, V bit set in status register
- 8) Window Option = 2, D outside window, D not replaced, WV interrupt generated, V bit set in status register

# **POPST** Pop Status Register from Stack

Syntax	POPST															
Execution	*SP+ → ST															
Instruction Words	15 14 0 0	13 0	12 0	<u>11</u>	10 0	9 0	8 1	7	6	5		4	3 0	2 0	1	0
Description	POPST p after the s	ops th tatus	e sta regi	atus r ster is	egis s rer	ster fi nove	rom 1 ed fro	the som th	tack ne sta	anc ack.	l in	cren	nents	s the	SP b	y 32
31 30 29 28	26 25	22	21						11	•	10—	-6	5		40	
N C Z V	BF IX	SS SS	IE		halling all a	n n n n n n n n n n n n n n n n n n n			FE1		FS	1	FE	0	FS0	
Note: Shaded portions a	Note: Shaded portions are reserved.															
Machine States	For more 6 if the SF 7 if the SF	inform P is ali P is no	natio gneo t alig	n, ref d gned	er to	Sec	tion	4.1,	<u>The</u>	<u>Stat</u>	tus	Rec	lister	; on	page	4-2.
Status Bits	All bits are restored.															
Examples	Assume that memory contains the following values before instruction execu- tion:															
	Address 0FF000001 0FF000101	ו ו	Dat 001 C00	<b>a</b> 0h )0h												
Examples	Code POPST	<u>Befor</u> SP 0FF0	<u>re</u> 0000	)h	<u>A</u> S C	<u>fter</u> T 0000	010h		<b>SP</b> 0FF	000	20h					

Syntax	PUSHST
Execution	ST → _*SP
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0
	0 0 0 0 0 0 0 1 1 1 1 0 0 0 0 0
Description	PUSHST writes the status register contents to the address contained in the SP-32.
31 30 29 28 2	26 25 22 21 11 10-6 5 4-0
	3F IX SS IE FEI FEI FSI FEO FSO

NC	2	VĽ	DF		1 33		Hundred Burner		 F3
Note: Sha	aded	portic	ons are r	eserv	ed.				

For more information, refer to Section 4.1, The Status Register, on page 4-2.

Machine States	2 (1) if the 2 (2) if the	e SP is aligned e SP is not alig	Ined		
Status Bits	N Unaff C Unaff Z Unaff V Unaff	ected ected ected ected			
Example	<u>Code</u> PUSHST	<u>Before</u> SP 0FF00020h	<b>ST</b> C0000010h	<u>After</u> SP 0FF00000h	
	Memory o	contains the fol	lowing values af	ter instruction execu	ition:
	Address 0FF00010	<b>Data</b> h 0010h			
	0FF000201	h C000h			

Syntax	PUTST Rs							
Execution	$Rs \rightarrow ST$							
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         0       0       0       0       0       1       1       0       1       R       Rd							
Description	PUTST copies the contents of the specified register into the status register.							
31         30         29         28           N         C         Z         V           Note:         Shaded portions a	26       25       22       21       11       10—6       5       4—0         BF       IX       IX       IX       IX       FE1       FS1       FE0       FS0         re reserved.							
Mashina Clates	For more information, refer to Section 4.1, <u>The Status Register</u> , on page 4-2.							
machine States	3							
Status Bits	<ul> <li>N Set from bit 31 of Rs</li> <li>C Set from bit 30 of Rs</li> <li>Z Set from bit 20 of Rs</li> </ul>							

- Z Set from bit 29 of Rs
- V Set from bit 28 of Rs

Example	<u>Code</u>	<u>Before</u>	Afte	<u>After</u>		
		A0	ST	ST		
	PUTST A0	C0000010h	xxxxxxxh	C0000010h		

Syntax	RETI					
Execution	$*SP+ \rightarrow ST$ $*SP+ \rightarrow PC$					
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         0       0       0       0       1       0       0       1       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0					
Description	RETI returns to an interrupted routine from an interrupt service routine. The instruction restores the ST and PC to their original values that were stored of the system stack.					
	The stack is located in external memory and the top is indicated by the stack pointer (SP). The stack grows in the direction of decreasing linear address. The ST and PC are popped from the stack and the SP is incremented by 32 after each register is removed from the stack.					
	Note:					
	RETI checks the IX (instruction execution) and BF (bus fault) bits in the restored ST register. If IX or BF is set, the RETI expects to find the internal register values that define the state of the TMS34020 on the stack along with the ST and PC.					
	If this is the case, the RETI restores the additional register values that were pushed on the stack and clears the IX and BF bits in the restored ST value.					
	The CONTROL register and any B-file registers modified by an interrupt routine should be restored before RETI is executed. Otherwise, interrupted instructions may not resume execution correctly.					
Machine States	52 if BF status bit = 1 38 if IX status bit = 1 else 7					
Status Bits	<ul> <li>N Copy of corresponding bit in stack location</li> <li>C Copy of corresponding bit in stack location</li> <li>Z Copy of corresponding bit in stack location</li> <li>V Copy of corresponding bit in stack location</li> <li>IE Copy of corresponding bit in stack location</li> </ul>					
Interrupts	If the IE bit in the restored ST is a 1, interrupts are enabled by the time the RETI instruction finishes executing. If an interrupt request is active during the last state of the RETI instruction, and the interrupt is enabled in the INTENB register, the interrupt will be taken immediately following the RETI. If the source of the interrupt is not cleared automatically, the interrupt service routine should take steps to clear the source of the interrupt. If this is not done, the interrupt will be serviced repeatedly. Sections 6.7, <u>External Interrupts</u> , on page 6-15,					

6.8, <u>Internal Interrupts</u>, on page 6-16, and 6.9, <u>The Bus Fault Interrupt</u>, on page 6-19 discuss each interrupt and the details for clearing the source of the interrupt.

Examples

Assume that memory contains the following values before instruction execution:

Address	Data
CCC0000h	0010h
CCC0010h	C000h
CCC0020h	FFF0h
CCC0030h	0044h

<u>Code</u>	<u>Before</u>	<u>After</u>		
	SP	ST	PC	SP
RETI	CCC0000h	C0000010h	0044FFF0h	CCC0040h

Syntax	RETM					
Execution	*SP+ *SP+	→ ST → PC				
Instruction Words	15 0	14 13 0 0	12 11 0 1	10 9 8 0 0 0	7     6     5       0     1     1	4     3     2     1     0       0     0     0     0     0     0
Description	RETM RETI, read d instruc seque	is used a but RETM irectly from tion is ex nce repea	at the c I forces m mer cecute ats.	end of a single s the next instru mory, that is, it d and the sing	step trap routi uction from the is not read fror gle step trap i	ne. RETM acts similar to interrupted program to be n the cache. The fetched s then taken again; this
	Note:					
	RETM interru are ble cache edly w unsuit	I uses the upted code ocked, so when RE vithout exe able for te	e cache e. Whe if the i TM is ecuting ermina	e read mechan en the single-st next instruction executed, then g any of the ma ating single-ste	ism to access t tep bit (bit 22 in i in the interrup the single step in program opc p traps.	he next instruction in the ST) is set the cache fills ted code is not already in trap will be taken repeat- odes. This makes RETM
Machine States	52 if B 38 if IX else 10	F status b ( status bi )	oit = 1 t = 1			
Status Bits	N         Co           C         Co           Z         Co           V         Co           IE         Co	opy of corr opy of corr opy of corr opy of corr opy of corr	respor respor respor respor respor	nding bit in stac nding bit in stac nding bit in stac nding bit in stac nding bit in stac	ck location ck location ck location ck location ck location	
Examples	Assum tion:	e that me	mory	contains the fo	llowing values	before instruction execu-
	Addres CCC00 CCC00 CCC00 CCC00	<b>ss</b> 00h 10h 20h 30h	<b>Data</b> 0010h C000h FFF0h 0044h	1		
	<u>Code</u>	<u>Before</u> SP		<u>After</u> ST	PC	SP
	RETM	CCC000	0h	C0000010h	0044FFF0h	CCC0040h

\*\*\*\*\*

Syntax	RETS [A	/]								
Execution	*SP → F SP + 32 +	PC (N defaults 16N → SP	s to 0)							
Instruction Words	15 14 0 0	13 12 11 0 0 1	10 9 8 0 0 1	7	6 5 1 1	4	3	2 N	1	0
Description	RETS retu stack and	urns from a su incrementing t	broutine by po he stack pointe	opping er.	, the p	orograi	ກ ເວເ	unter	from	the
	The paran words by is popped explicitly, t	neter <i>N</i> is a va which the stacl from the syster the assembler	lue in the rang c pointer SP is n stack. N is op sets it to the de	e of 0 increi otional efault	to 31; mente l; if the value	it spe d after value of <i>0</i> .	cifies the r of N is	the n eturr s not	umbe addi speci	er of ess fied
	Following address p	completion of ointed to by the	the RETS ins PC popped fi	structi rom th	on, ex ne stac	kecutio ck.	n cor	ntinue	es at	the
Machine States	5 6 if the sta	ack isn't aligned	Ł							
Status Bits	N Unaffe C Unaffe Z Unaffe V Unaffe	ected ected ected ected								
Examples	Assume th tion:	nat memory co	ntains the follo	wing \	alues	before	e insti	ructic	n exe	eu-
	<b>Address</b> 0FF00000F 0FF00010F	Data 0FFF0h 0001h								
	Code RETS 1 RETS 2 RETS 16 RETS 31	Before SP 0FF00000h 0FF00000h 0FF00000h 0FF00000h 0FF00000h	After PC 0001FFF0h 0001FFF0h 0001FFF0h 0001FFF0h 0001FFF0h	<b>SP</b> 0FF( 0FF( 0FF( 0FF(	000201 000301 000401 001201 002101	ר ז ז ז				

Syntax	REV Rd															
Execution	revisio	on nu	imbe	er –	→ Rd											
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	_2	1	0
	0	0	0	0	0	0	0	0	0	0	1	R		F	Rd	
Description	REV s destin	stores	s the 1 reg	e nun ister	nber . The	whicl form	n unio nat of	quely the l	' iden REV	itifies numi	the ber is	revis s:	ion c	of sili	con ir	n the
	bits 0-	2	S	ilicor	n revi	sion	numl	ber								
	bit 3 = 1 if TMS34010 (if bit 3 = 0, then TMS34020; bits 3 and 4 cannot both be 1)															
	bit $4 = 1$ if TMS34020 (if bit $4 = 0$ , then TMS34010; bits 3 and 4 cannot both be 1))															
	bits 5—15 reserved for future generation parts															
	bits 16	6—23	3 s	pin-c	offs											
	bits 24	4—31	l re	eserv	/ed											
Machine States	1															
Status Bits	N U C U Z U V U	naffe naffe naffe naffe	cted cted cted cted													
Examples	REV A0															
	Before	•	A0	= F	FFF	FFF										
	After		A0	= (	00000	010	(TMS	3402	0 revi	sion <sup>.</sup>	1.0)					
	After A0 = 00000011 (TMS34020 revision 2.0)															

# RL Rotate Left, Constant

RL consta	nt, Rd											
left-rotate	Rd by co	nstant -	→ Rd									
15 14 0 0	13 12 1 1	11 10 0 0	9	8 con	7 nstant	6 t	5	4 R	3	2 F	1 Rd	0
RL rotates of bits. RL of the regi in the rang word.	the conto performs ster into t je 0 to 31	ents of th a circula he regista and is sto	e desti r left sh er's LS pred in f	natior nift tha B. Th the 5-	n reg at mo le rol bit co	giste oves tate onst	r left eac cour ant f	by ti h bit ht is iield	he sp shifte speci of the	ecife ed ou ified e RL i	ed nur it the l as a v instru	mber MSB value ction
С — — —		31  MSB			4						0 	 - - -
The asser specified r sets the co	nbler only otation va onstant to	/ accepts alue is gre o its 5 LS	absoli eater th Bs.	ute ex an 31	kpres , the	ssior	ns fo emb	or the ler is	e rota ssues	te co a wa	ount. I arning	f the and
The carry l value is the 0 to clear t	bit is set to e same a the carry	o the valu s the fina and test	ue of th I value a regis	e last of the ter fo	bit the LSI	hat is B). Y simul	s shi 'ou c Itane	fted can u eous	out o Ise a Iy.	f the rotat	MSB te cou	(this int of
1												
<ul> <li>N Unaffe</li> <li>C Set to</li> <li>Z 1 if res</li> <li>V Unaffe</li> </ul>	ected value of sult is 0, <i>(</i> ected	bit [32 — ( ) otherwi	consta se	nt], <i>0</i>	for r	otate	e co	unt c	of cor	nstar	nt = 0	
Code RL 0,A1 RL 1,A1 RL 4,A1 RL 5,A1 RL 30,A1 RL 5,A1	Before A1 0000000 F00000 F00000 F00000 F00000 000000	0Fh 00h 00h 00h 00h	After NCZ × 0 0 2 × 1 0 2 × 1 0 2 × 0 0 2 × 1 0 2 × 1 0 2 × 0 1 2	V A × C × E × C × C × C × C	<b>A1</b> 20000 20000 20000 20000 3C00	000F 0001 000F 001E 0000	h h h h h					
	RL constat left-rotate 15 14 0 0 RL rotates of bits. RL of the regin in the rang word. C C C The asser specified r sets the co The carry I value is the 0 to clear f 1 N Unaffe C Set to Z 1 if res V Unaffe	RL constant, Rdleft-rotate Rd by col $15$ $14$ $13$ $12$ $0$ $0$ $1$ $1$ RL rotates the control bits. RL performs of the register into t in the range 0 to 31 word.C $C$ C $C$ Code $Before$ A1RL $0, A1$ CO000 $C$ RL $1, A1$ F00000 $RL$ RL $2, A1$ F00000 $RL$ RL $5, A1$ F00000 $RL$ S, A1F00000RL $5, A1$ C $C$ <th>RL constant, Rdleft-rotate Rd by constant <math>-15</math>15 14 13 12 11 10001100110001100RL rotates the contents of the of bits. RL performs a circular of the register into the register into the register in the range 0 to 31 and is storword.C31MSBThe assembler only accepts specified rotation value is gressets the constant to its 5 LSThe carry bit is set to the value value is the same as the fina 0 to clear the carry and test1NUnaffectedCSet to value of bit [32 - 421if result is 0, 0 otherwitVUnaffectedCodeBefore A1RL 0,A100000000hRL 1,A1F0000000hRL 3,A1F0000000hRL 3,A1F0000000hRL 3,A1F0000000hRL 5,A10000000h</th> <th>RL constant, Rdleft-rotate Rd by constant <math>\rightarrow</math> Rd1514131211109001100RL rotates the contents of the destion of bits. RL performs a circular left shoof the register into the register's LS in the range 0 to 31 and is stored in tword.C31C31MSBLeft colspan="2"&gt;Colspan="2"&gt;Colspan="2"&gt;C31MSBThe assembler only accepts absolute specified rotation value is greater the sets the constant to its 5 LSBs.The carry bit is set to the value of the value is the same as the final value 0 to clear the carry and test a regise 1NUnaffectedCSet to value of bit [32 - constated the carry and test a regise 1NUnaffectedCSet to value of bit [32 - constated the carry and test a regise 1NUnaffectedCSet to value of bit [32 - constated the carry and test a regise 1NUnaffectedCSet to value of bit [32 - constated the carry and test a regise 1NUnaffectedCSet to value of bit [32 - constated the carry and test a regise 1NUnaffectedCSet to value of bit [32 - constated the carry and test a regise 1NUnaffectedCSet to value of bit [32 - constated the carry and test a regise 1RL 0, A100000000 A to 0 000000000000000000000000</th> <th><b>RL</b> constant, Rdleft-rotate Rd by constant → Rd<math>15</math><math>14</math><math>13</math><math>12</math><math>11</math><math>10</math><math>9</math><math>8</math><math>0</math><math>0</math><math>1</math><math>1</math><math>0</math><math>0</math>conRL rotates the contents of the destination of bits. RL performs a circular left shift the of the register into the register's LSB. The in the range 0 to 31 and is stored in the 5-word.C31Constant of the register into the register's LSB. The in the range 0 to 31 and is stored in the 5-word.The assembler only accepts absolute exspecified rotation value is greater than 31 sets the constant to its 5 LSBs.The carry bit is set to the value of the last value is the same as the final value of the 0 to clear the carry and test a register for 0 to clear the carry and test a register for 1NUnaffectedCSet to value of bit [32 - constant], 0Z1 if result is 0, 0 otherwiseVUnaffectedCodeBeforeA1N C Z VRL 0,A10000000hX 1 0 x 0RL 4,A1F000000hX 1 0 x 0RL 30,A1F000000hX 1 0 x 0RL 5,A10000000hX 1 0 x 0RL 5,A10000000h</th> <th><b>RL</b> constant, Rdleft-rotate Rd by constant → Rd<math>15</math><math>14</math><math>13</math><math>12</math><math>11</math><math>10</math><math>9</math><math>8</math><math>7</math><math>0</math><math>0</math><math>1</math><math>1</math><math>0</math><math>0</math>constantRL rotates the contents of the destination regor of bits. RL performs a circular left shift that more of the register into the register's LSB. The root in the range 0 to 31 and is stored in the 5-bit convord.C<math>31</math>MSBLeft colspan="2"&gt;ConstantMSBThe assembler only accepts absolute express specified rotation value is greater than 31, the sets the constant to its 5 LSBs.The carry bit is set to the value of the last bit to value is the same as the final value of the LS 0 to clear the carry and test a register for 0 set 1N UnaffectedCodeBeforeAfterAfterN C Z V A1RL 0, A10000000Fh× 0 0 × 0000RL 1, A1ForderAfterN C Z V A1RL 0, A10000000Fh× 0 0 × 0000RL 0, A10000000Fh× 0 0 × 0000RL 0, A100000000 × 1 0 × 0000RL 0, A100000000 × 0 × 0000RL 0, A100000</th> <th><b>RL</b> constant, Rdleft-rotate Rd by constant → Rd<math>15</math><math>14</math><math>13</math><math>12</math><math>11</math><math>10</math><math>9</math><math>8</math><math>7</math><math>6</math><math>0</math><math>0</math><math>1</math><math>1</math><math>0</math><math>0</math>constantRL rotates the contents of the destination registerof bits. RL performs a circular left shift that movesof the register into the register's LSB. The rotatein the range 0 to 31 and is stored in the 5-bit constword.Colspan="2"&gt;Colspan="2"&gt;31Colspan="2"&gt;Colspan="2"&gt;Colspan="2"&gt;Colspan="2"&gt;Colspan="2"&gt;All Solute expressionspecified rotation value is greater than 31, the assist sets the constant to its 5 LSBs.The carry bit is set to the value of the last bit that is value is the same as the final value of the LSB). Y0 to clear the carry and test a register for 0 simule1NUnaffectedCodeBeforeAfterN C Z V A1RL 0,A1NO00000FhX 0 0x 000000FhX 0 0x 0</th> <th><b>RL</b> constant, Rdleft-rotate Rd by constant → Rd<math>15</math><math>14</math><math>13</math><math>12</math><math>11</math><math>10</math><math>9</math><math>8</math><math>7</math><math>6</math><math>5</math><math>0</math><math>0</math><math>1</math><math>1</math><math>0</math><math>0</math>constantRL rotates the contents of the destination register left of bits. RL performs a circular left shift that moves each of the register into the register's LSB. The rotate cour in the range 0 to 31 and is stored in the 5-bit constant for word.C<math>31</math>C<math>31</math>C<math>31</math>CMSBThe assembler only accepts absolute expressions for specified rotation value is greater than 31, the assemble sets the constant to its 5 LSBs.The carry bit is set to the value of the last bit that is shivalue is the same as the final value of the LSB). You cold 0 to clear the carry and test a register for 0 simultant of 0 to clear the carry and test a register for 0 simultant of 0 to clear the carry and test a register for 0 simultant of 0 to clear the carry and test a register for 0 simultant of 0 to clear the carry and test a register for 0 simultant of 0 to clear the carry and test a register of 0 simultant of 0 to clear the carry and test a register of 0 simultant of 0 to clear the carry and test a register of 0 simultant of 0 to clear the carry and test a register for 0 simultant of 0 to clear the carry and test a register of 0 simultant of 0 to clear the carry and test a register of 0 simultant of 0 to clear the carry and test a register of 0 simultant of 0 to clear the carry and test a register of 0 simultant of 0 to clear the carry and test a register of</th> <th><b>RL</b> constant, Rdleft-rotate Rd by constant → Rd<math>15</math><math>14</math><math>13</math><math>12</math><math>11</math><math>10</math><math>9</math><math>8</math><math>7</math><math>6</math><math>5</math><math>4</math><math>0</math><math>0</math><math>1</math><math>1</math><math>0</math><math>0</math>constant<math>R</math>RL rotates the contents of the destination register left by t of bits. RL performs a circular left shift that moves each bit of the register into the register's LSB. The rotate count is in the range 0 to 31 and is stored in the 5-bit constant field word.C<math>31</math><math>4</math><math>6</math><math>6</math>MSB<math>4</math><math>6</math><math>6</math><math>6</math>The assembler only accepts absolute expressions for the specified rotation value is greater than 31, the assembler is sets the constant to its 5 LSBs.The carry bit is set to the value of the last bit that is shifted value is the same as the final value of the LSB). You can u 0 to clear the carry and test a register for 0 simultaneous1NUnaffectedCSet to value of bit [32 - constant], 0 for rotate count of 2 1 if result is 0, 0 otherwiseVUnaffectedCodeBeforeA1N CZVN 10 × E000000h × 1 0 × 000000FhRL 0, A1000000Fh × 0 0 × 000000FhRL 1, A1F000000h × 1 0 × 000000FhRL 5, A1F000000h × 0 0 × 000000FhRL 5, A10000000h × 0 0 × 0000000h</th> <th>RL constant, Rdleft-rotateRd by constant → Rd<math>15</math><math>14</math><math>13</math><math>12</math><math>11</math><math>10</math><math>9</math><math>8</math><math>7</math><math>6</math><math>5</math><math>4</math><math>3</math><math>0</math><math>0</math><math>1</math><math>1</math><math>0</math><math>0</math>constant<math>R</math><math>R</math>RL rotates the contents of the destination register left by the sp of bits. RL performs a circular left shift that moves each bit shift of the register into the register's LSB. The rotate count is speci in the range 0 to 31 and is stored in the 5-bit constant field of the word.<math>C</math><math>31</math><math>C</math><math>C</math><math>T</math><math>MSB</math><math>C</math><math>T</math><math>R</math><math>R</math><math>MSB</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><math>R</math><!--</th--><th>RL constant, Rdleft-rotate Rd by constant → Rd<math>15</math>141312111098765432001100constantRrrrRL rotates the contents of the destination register left by the specifie of bits. RL performs a circular left shift that moves each bit shifted ou of the register into the register's LSB. The rotate count is specified in the range 0 to 31 and is stored in the 5-bit constant field of the RL is word.C31</th><th>RL constant, Rdleft-rotate Rd by constant → Rd<math>15</math><math>14</math><math>13</math><math>12</math><math>11</math><math>10</math><math>9</math><math>8</math><math>7</math><math>6</math><math>5</math><math>4</math><math>3</math><math>2</math><math>1</math><math>10</math><math>0</math><math>1</math><math>1</math><math>1</math><math>0</math><math>0</math><math>constant</math><math>R</math><math>R</math><math>d</math>RL rotates the contents of the destination register left by the specified as a volume of the register into the register's LSB. The rotate count is specified as a volume of the range 0 to 31 and is stored in the 5-bit constant field of the RL instruver.C<math>31</math><math>0</math><math>MSB</math><math>LSB</math>The assembler only accepts absolute expressions for the rotate count. I specified rotation value is greater than 31, the assembler issues a warning sets the constant to its 5 LSBs.The carry bit is set to the value of the last bit that is shifted out of the MSB value is the same as the final value of the LSB). You can use a rotate could 0 to clear the carry and test a register for 0 simultaneously.1NUnaffectedCodeBeforeAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfter&lt;</th></th>	RL constant, Rdleft-rotate Rd by constant $-15$ 15 14 13 12 11 10001100110001100RL rotates the contents of the of bits. RL performs a circular of the register into the register into the register in the range 0 to 31 and is storword.C31MSBThe assembler only accepts specified rotation value is gressets the constant to its 5 LSThe carry bit is set to the value value is the same as the fina 0 to clear the carry and test1NUnaffectedCSet to value of bit [32 - 421if result is 0, 0 otherwitVUnaffectedCodeBefore A1RL 0,A100000000hRL 1,A1F0000000hRL 3,A1F0000000hRL 3,A1F0000000hRL 3,A1F0000000hRL 5,A10000000h	RL constant, Rdleft-rotate Rd by constant $\rightarrow$ Rd1514131211109001100RL rotates the contents of the destion of bits. RL performs a circular left shoof the register into the register's LS in the range 0 to 31 and is stored in tword.C31C31MSBLeft colspan="2">Colspan="2">Colspan="2">C31MSBThe assembler only accepts absolute specified rotation value is greater the sets the constant to its 5 LSBs.The carry bit is set to the value of the value is the same as the final value 0 to clear the carry and test a regise 1NUnaffectedCSet to value of bit [32 - constated the carry and test a regise 1NUnaffectedCSet to value of bit [32 - constated the carry and test a regise 1NUnaffectedCSet to value of bit [32 - constated the carry and test a regise 1NUnaffectedCSet to value of bit [32 - constated the carry and test a regise 1NUnaffectedCSet to value of bit [32 - constated the carry and test a regise 1NUnaffectedCSet to value of bit [32 - constated the carry and test a regise 1NUnaffectedCSet to value of bit [32 - constated the carry and test a regise 1RL 0, A100000000 A to 0 000000000000000000000000	<b>RL</b> constant, Rdleft-rotate Rd by constant → Rd $15$ $14$ $13$ $12$ $11$ $10$ $9$ $8$ $0$ $0$ $1$ $1$ $0$ $0$ conRL rotates the contents of the destination of bits. RL performs a circular left shift the of the register into the register's LSB. The in the range 0 to 31 and is stored in the 5-word.C31Constant of the register into the register's LSB. The in the range 0 to 31 and is stored in the 5-word.The assembler only accepts absolute exspecified rotation value is greater than 31 sets the constant to its 5 LSBs.The carry bit is set to the value of the last value is the same as the final value of the 0 to clear the carry and test a register for 0 to clear the carry and test a register for 1NUnaffectedCSet to value of bit [32 - constant], 0Z1 if result is 0, 0 otherwiseVUnaffectedCodeBeforeA1N C Z VRL 0,A10000000hX 1 0 x 0RL 4,A1F000000hX 1 0 x 0RL 30,A1F000000hX 1 0 x 0RL 5,A10000000hX 1 0 x 0RL 5,A10000000h	<b>RL</b> constant, Rdleft-rotate Rd by constant → Rd $15$ $14$ $13$ $12$ $11$ $10$ $9$ $8$ $7$ $0$ $0$ $1$ $1$ $0$ $0$ constantRL rotates the contents of the destination regor of bits. RL performs a circular left shift that more of the register into the register's LSB. The root in the range 0 to 31 and is stored in the 5-bit convord.C $31$ MSBLeft colspan="2">ConstantMSBThe assembler only accepts absolute express specified rotation value is greater than 31, the sets the constant to its 5 LSBs.The carry bit is set to the value of the last bit to value is the same as the final value of the LS 0 to clear the carry and test a register for 0 set 1N UnaffectedCodeBeforeAfterAfterN C Z V A1RL 0, A10000000Fh× 0 0 × 0000RL 1, A1ForderAfterN C Z V A1RL 0, A10000000Fh× 0 0 × 0000RL 0, A10000000Fh× 0 0 × 0000RL 0, A100000000 × 1 0 × 0000RL 0, A100000000 × 0 × 0000RL 0, A100000	<b>RL</b> constant, Rdleft-rotate Rd by constant → Rd $15$ $14$ $13$ $12$ $11$ $10$ $9$ $8$ $7$ $6$ $0$ $0$ $1$ $1$ $0$ $0$ constantRL rotates the contents of the destination registerof bits. RL performs a circular left shift that movesof the register into the register's LSB. The rotatein the range 0 to 31 and is stored in the 5-bit constword.Colspan="2">Colspan="2">31Colspan="2">Colspan="2">Colspan="2">Colspan="2">Colspan="2">All Solute expressionspecified rotation value is greater than 31, the assist sets the constant to its 5 LSBs.The carry bit is set to the value of the last bit that is value is the same as the final value of the LSB). Y0 to clear the carry and test a register for 0 simule1NUnaffectedCodeBeforeAfterN C Z V A1RL 0,A1NO00000FhX 0 0x 000000FhX 0 0x 0	<b>RL</b> constant, Rdleft-rotate Rd by constant → Rd $15$ $14$ $13$ $12$ $11$ $10$ $9$ $8$ $7$ $6$ $5$ $0$ $0$ $1$ $1$ $0$ $0$ constantRL rotates the contents of the destination register left of bits. RL performs a circular left shift that moves each of the register into the register's LSB. The rotate cour in the range 0 to 31 and is stored in the 5-bit constant for word.C $31$ C $31$ C $31$ CMSBThe assembler only accepts absolute expressions for specified rotation value is greater than 31, the assemble sets the constant to its 5 LSBs.The carry bit is set to the value of the last bit that is shivalue is the same as the final value of the LSB). You cold 0 to clear the carry and test a register for 0 simultant of 0 to clear the carry and test a register for 0 simultant of 0 to clear the carry and test a register for 0 simultant of 0 to clear the carry and test a register for 0 simultant of 0 to clear the carry and test a register for 0 simultant of 0 to clear the carry and test a register of 0 simultant of 0 to clear the carry and test a register of 0 simultant of 0 to clear the carry and test a register of 0 simultant of 0 to clear the carry and test a register for 0 simultant of 0 to clear the carry and test a register of 0 simultant of 0 to clear the carry and test a register of 0 simultant of 0 to clear the carry and test a register of 0 simultant of 0 to clear the carry and test a register of 0 simultant of 0 to clear the carry and test a register of	<b>RL</b> constant, Rdleft-rotate Rd by constant → Rd $15$ $14$ $13$ $12$ $11$ $10$ $9$ $8$ $7$ $6$ $5$ $4$ $0$ $0$ $1$ $1$ $0$ $0$ constant $R$ RL rotates the contents of the destination register left by t of bits. RL performs a circular left shift that moves each bit of the register into the register's LSB. The rotate count is in the range 0 to 31 and is stored in the 5-bit constant field word.C $31$ $4$ $6$ $6$ MSB $4$ $6$ $6$ $6$ The assembler only accepts absolute expressions for the specified rotation value is greater than 31, the assembler is sets the constant to its 5 LSBs.The carry bit is set to the value of the last bit that is shifted value is the same as the final value of the LSB). You can u 0 to clear the carry and test a register for 0 simultaneous1NUnaffectedCSet to value of bit [32 - constant], 0 for rotate count of 2 1 if result is 0, 0 otherwiseVUnaffectedCodeBeforeA1N CZVN 10 × E000000h × 1 0 × 000000FhRL 0, A1000000Fh × 0 0 × 000000FhRL 1, A1F000000h × 1 0 × 000000FhRL 5, A1F000000h × 0 0 × 000000FhRL 5, A10000000h × 0 0 × 0000000h	RL constant, Rdleft-rotateRd by constant → Rd $15$ $14$ $13$ $12$ $11$ $10$ $9$ $8$ $7$ $6$ $5$ $4$ $3$ $0$ $0$ $1$ $1$ $0$ $0$ constant $R$ $R$ RL rotates the contents of the destination register left by the sp of bits. RL performs a circular left shift that moves each bit shift of the register into the register's LSB. The rotate count is speci in the range 0 to 31 and is stored in the 5-bit constant field of the word. $C$ $31$ $C$ $C$ $T$ $MSB$ $C$ $T$ $R$ $R$ $MSB$ $R$ </th <th>RL constant, Rdleft-rotate Rd by constant → Rd<math>15</math>141312111098765432001100constantRrrrRL rotates the contents of the destination register left by the specifie of bits. RL performs a circular left shift that moves each bit shifted ou of the register into the register's LSB. The rotate count is specified in the range 0 to 31 and is stored in the 5-bit constant field of the RL is word.C31</th> <th>RL constant, Rdleft-rotate Rd by constant → Rd<math>15</math><math>14</math><math>13</math><math>12</math><math>11</math><math>10</math><math>9</math><math>8</math><math>7</math><math>6</math><math>5</math><math>4</math><math>3</math><math>2</math><math>1</math><math>10</math><math>0</math><math>1</math><math>1</math><math>1</math><math>0</math><math>0</math><math>constant</math><math>R</math><math>R</math><math>d</math>RL rotates the contents of the destination register left by the specified as a volume of the register into the register's LSB. The rotate count is specified as a volume of the range 0 to 31 and is stored in the 5-bit constant field of the RL instruver.C<math>31</math><math>0</math><math>MSB</math><math>LSB</math>The assembler only accepts absolute expressions for the rotate count. I specified rotation value is greater than 31, the assembler issues a warning sets the constant to its 5 LSBs.The carry bit is set to the value of the last bit that is shifted out of the MSB value is the same as the final value of the LSB). You can use a rotate could 0 to clear the carry and test a register for 0 simultaneously.1NUnaffectedCodeBeforeAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfter&lt;</th>	RL constant, Rdleft-rotate Rd by constant → Rd $15$ 141312111098765432001100constantRrrrRL rotates the contents of the destination register left by the specifie of bits. RL performs a circular left shift that moves each bit shifted ou of the register into the register's LSB. The rotate count is specified in the range 0 to 31 and is stored in the 5-bit constant field of the RL is word.C31	RL constant, Rdleft-rotate Rd by constant → Rd $15$ $14$ $13$ $12$ $11$ $10$ $9$ $8$ $7$ $6$ $5$ $4$ $3$ $2$ $1$ $10$ $0$ $1$ $1$ $1$ $0$ $0$ $constant$ $R$ $R$ $d$ RL rotates the contents of the destination register left by the specified as a volume of the register into the register's LSB. The rotate count is specified as a volume of the range 0 to 31 and is stored in the 5-bit constant field of the RL instruver.C $31$ $0$ $MSB$ $LSB$ The assembler only accepts absolute expressions for the rotate count. I specified rotation value is greater than 31, the assembler issues a warning sets the constant to its 5 LSBs.The carry bit is set to the value of the last bit that is shifted out of the MSB value is the same as the final value of the LSB). You can use a rotate could 0 to clear the carry and test a register for 0 simultaneously.1NUnaffectedCodeBeforeAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfterAfter<

Syntax	RL R	s, Rd													
Execution	left-rota	ate Ro	by 5	LSBs	of R	s →	Rd								
Instruction Words	15 1	4 13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	1 1	0	1	0	0			Rs		R			Rd	
Description	RL rota specifie each bi count is of the s	ates th ed in t it shifte s spec source	ne con he sou ed out ified a regist	tents urce i of the s a va ter; th	of th regis MSE alue i ie 27	e de: ter. F 3 of th n the MSE	stinat {L pe e reg range 3s of t	ion rfor iste e 0 the	regis ms a r into to 31 sour	ter le circu the re and i ce rec	ft by Ilar le egiste s tak gister	the eft sl er's L en fr are	num hift tl SB. om t ignc	ber o nat m The ro he 5 L ored.	f bits oves otate .SBs
	c	,		31					◄					0	 1
		1 1 1		N 	ISB 				<b>.</b>					LSB 	ו ו ב – ו
	The ca value is 0 to cle	rry bit s the s ear the	is set t ame a carry	o the s the and t	valu final text F	e of ti value Rd fo	ne las e of th r 0 sir	it bii ne L nuli	t that SB). tanec	is shi You c busly.	fted o an u	out o se a	f the rota	MSB te cou	(this int to
	Rs and	Rdm	nust be	e in th	e sa	me re	egiste	er fil	e.						
Machine States	1														
Status Bits	<ul> <li>N Un</li> <li>C Set</li> <li>Z 1 if</li> <li>V Un</li> </ul>	affecto t to va resul <sup>a</sup> affecto	ed Iue of t is 0, ( ed	bit [3 9 othe	2 — F erwis	Rs], <i>0</i> e	for ro	otat	e coi	int of	0				
Examples	<u>Code</u>		Befor	e	4	After									
	RL A0, RL A0, RL A0, RL A0,	A1 A1 A1 A1	5 LSB 0 0 0 ( 0 0 1 ( 0 0 1 ( 1 1 1	<b>is A0</b> 0 0 0 0 0 1 1 1	)         	<b>A1</b> 00000 =0000 =0000 =0000	)00Fh )000h )000h )000h		NC X 0 X 1 X 0 X 0 X 0	<b>Z V</b> 0 x 0 x 0 x 0 x 0 x	A1 000 000 780	00000 00000 00000	OFh OFh 1Eh OOh		
	NU AU,	<b>AT</b>		~ ~	,				~ 0	. ~	000	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	5011		

# RMO Rightmost One

Syntax	RMO Rs, Rd						
Execution	bit number of rig	ghtmost 1 in Rs	$\rightarrow$ Rd				
Instruction Words	15 14 13 0 1 1	12 11 10 9 1 1 0	9 <u>87</u> 1 F	65 Rs	4 3 R	2 1 Rd	0
Description	The RMO instru- register. It then register. Bit 31 of there are no 1 b status bit Z is se	uction locates the loads the bit nur of Rs is the MSE bits in the source et.	e rightmost nber of the 3 (leftmost) e register, th	(least sig rightmost and bit 0 is nen the de	nificant) 7 1 bit into 1 s the LSB stination	f in the s the desti (rightm result is	ource nation ost). If 0 and
	The rightmost <i>1</i> RMO instruction ter of the RMO i	in the source r with RL <i>Rs, Rd</i> instruction and l	egister can instruction, Rd is the so	be right-ji where Rs ource regis	ustified by is the dea ter.	y followi stination	ng the regis-
	The source and	destination reg	isters must	be in the s	same regi	ister.	
Machine States	1						
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z 1 if the sour</li> <li>V Unaffected</li> </ul>	ce register cont	ents are 0,	0 otherwis	Se.		
Examples	Code RMO A0,A1 RMO A0,A1 RMO A0,A1 RMO A0,A1	Before A0 00000000h 00000001h 00000010h 08000000h	After NCZV xx1x xx0x xx0x xx0x xx0x	A1 000000000 00000000 00000004 0000001B	h h h		
	RMO A0,A1	8000000h	xx0x	0000001F	h		

Syntax	RPIX Rd							
Execution	Rd <sub>new</sub> = Rd <sub>old</sub> LS pixel replicated	( <u>32</u> (PSIZE)	5 times					
Instruction Words	15 14 13 12 0 0 0 0	<u>11 10</u>	98	7 6	5 4 0 F	3	2 1 Rd	0
Description	RPIX replicates the executing the instru size is specified by following completion throughout the 32 b	pixel va ction, yc PSIZE al of the in its of the	lue in the s ou should rig nd must be struction, th register.	pecified ght-justifi 1, 2, 4, 8 e pixel va	destina y the va , 16, or alue will	tion reg alue in 32 bits have b	gister. Pri Rd. The 5. Immedi een replic	or to pixel ately ated
	Given a pixel size of value with 32/ <i>n</i> copie bits to the left of the <u>Auxiliary Graphics In</u>	<i>n</i> bits, th es of the p original p <u>nstructio</u>	e replication bixel. The re bixel. For mo <u>ns</u> , on page	n operation plication ore inforr e 12-17.	on repla proces nation,	aces the s overw refer to	e original vrites the Section	pixel 32 <i>—n</i> 12.6,
Implied Operands	Address Name	e l	Description	and Elem	nents (B	lits)		
	C0000150h PSI	ZE	Pixel size (*	1,2,4,8,16	,32)			
Machine States	<ul> <li>2 if PSIZE = 32</li> <li>4 if if PSIZE = 16</li> <li>5 if PSIZE = 8</li> <li>6 if PSIZE = 4</li> <li>7 if PSIZE = 2</li> <li>8 if PSIZE = 1</li> </ul>	5						
Status Bits	<ul><li>N Unaffected</li><li>C Unaffected</li><li>Z Unaffected</li><li>V Unaffected</li></ul>							
Examples	RPXL A0 PSIZE = 8	Before After	A0 = XXXX A0 = 34343	XX34 43434	Сус	cles = 5		
	RPXL B8 PSIZE = 4	Before After	B8 = XXXX B8 = AAAA	XXXA AAAA	Сус	cles = 6		

# SETC Set Carry Bit

Syntax	SET	С														
Execution	1 →	· C														
Instruction Words	15 0	14 0	13 0	12 0	11	10 1	9 0	8	7	6	5	4	3	2	1	0
Description	SET ister	C set is ur	s the affec	carr cted.	y bit (	(C) in	the s	tatus	regi	stert	o 1. T	he r	esto	fthes	statu	s reg-

31	30	29	28	26	25	22	21	11	10—6	5	40	_
N	C	Ζ	V	BF	IX	SS	IE	FE1	FS1	FE0	FS0	]
Note	: Sh	aded	port	ions are re	eser	ved.						

This instruction is useful for returning a true/false value (in the carry bit) from a subroutine without using a general-purpose register.

Machine States	1				
Status Bits	N Un C 1 Z Un	affected affected			
Evamplas	V Un	Befere		After	
LXamples	Code	ST	NCZV	ST	NCZV
	SETC	B0000010h	1011	40000000h F0000010h	1111
	SETC	4000001Fh	0100	4000001Fh	0100

Syntax	SETCDP														
Execution	Destinatio	n pitch	conve	rsion	facto	or ⊸	• CO	NV	ΌP						
Instruction Words	15 14 0 0	13 12 0 0	2 11 0	10 0	9 1	8 0	7	e 1	3	5 1	4	3 0	2	1	0
Description	SETCDP to linear c	loads th onversi	ne COl on bas	NVDF sed or	P reg	ister DP	with TCH	n the reg	e ap giste	opro er.	priat	e va	lue u	ised	in XY
	Remember ister has b more infor <u>Graphics</u>	er to exe been sel mation Instruct	cute M befor refer <u>ons</u> , c	IWAI e usir to Se on pa	T aften ng its ctior ge 12	er SE valu 12. <sup>-</sup> 2-43.	TCE le in 11, <u>S</u>	DPt aC Setti	o er VX` ng i	nsur YL c up th	e tha or sin <u>ne In</u>	at the nilar nplie	COI instru d Op	NVDF uctior erand	Preg- n. For <u>ds for</u>
Implied Operands	Address	1	lame			De	scri	ptio	n ai	nd E	leme	ents	(Bits)	)	
	B3		DPTC	H (line	linear) Destination array pitch										
	C000014	10h	CONV	/DP		C	estir	natio	on p	itch	conv	ersio	n reg	ister	
Machine States	pitch is	a powe 2 powe arbitra	er of 2 ers of 2 ry	: 4 2: 6 3	(1) (1) (1)								•		
Status Bits	N Unaffe C Unaffe Z Unaffe V Unaffe	ected ected ected ected													
Examples	Before: After:	B3 = 0 C0000	00010 140 =	00h 0013	(51 h	2 × 8	3)								
	Before: After:	B3 = 0 C0000	00004 140 =	00h 0015	(12 h	8 × 8	3)								
	Before: After:	B3 = 0 C0000	00014 140 =	00h 1513	(64 h	0 × 8	3)								
	Before: After:	B3 = 0 C0000	00000 140 =	19h 0000	(25 h	× 1)									

# SETCMP Set CONVMP

Syntax	SETCMP
Execution	Mask pitch conversion factor $\rightarrow$ CONVMP
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         0       0       0       0       0       1       0       1       1       1       1       0       1       1
Description	SETCMP loads the CONVMP register with the appropriate value used in XY to linear conversion based on the MPTCH register.
	Remember to execute MWAIT after SETCMP to ensure that the CONVMP reg- ister has been set before using its value in a CVMXYL or similar instruction. For more information, refer to Section 12.11, <u>Setting up the Implied Operands</u> for Graphics Instructions, on page 12-43.
Implied Operands	Address Name Description and Elements (Bits)
	B11 MPTCH (linear) Mask array pitch
	C0000180h CONVMP Mask pitch conversion register
Machine States	pitch is a power of 2: 4(1) 2 powers of 2: 6(1) arbitrary 3(1)
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>
Examples	Before: B3 = 00001000 (512 × 8) After: C0000180 = 0013
	Before: B3 = 00000400 (128 × 8) After: C0000180 = 0015
	Before: B3 = 00001400 (640 × 8) After: C0000180 = 1513
	Before: B3 = 00000019 (25 × 1) After: C0000180 = 0000

Syntax	SETCSP						
Execution	Source pit	ch conversion fa	actor $\rightarrow C$	ONVSP			
Instruction Words	15 14 0 0	13 12 11 1 0 0 0 0	0 9 E	3 7 6 ) 0 1	5 <u>5</u> 4 0 1	3 2 0 0	1 0 0 1
Description	SETCSP I to linear co	oads the CONV onversion based	SP regist	er with the PTCH reg	e appropria ister.	ite value u	sed in XY
	Remembe register ha For more in for Graphic	r to execute MV s been set befor nformation, refer <u>cs Instructions</u> , o	VAIT afte e using its to Sectio on page 1	r SETCSI value in a n 12.11, <u>S</u> 2-43.	P to ensure CVSXYL ( a ctting up th	e that the or similar in ne Implied (	CONVSP struction. <u>Operands</u>
Implied Operands	Address	Name	[	Descriptio	n and Elem	ents (Bits)	
	B1	SPTCH (	linear)	Source ar			
	C000013	0h CONVSP	•	Source pi	tch conversi	ion register	
Machine States	pitch is	a power of 2: 2 powers of 2: arbitrary	4(1) 6(1) 3(1)				
Status Bits	N Unaffe C Unaffe Z Unaffe V Unaffe	ected ected ected ected					
Examples	Before: B3 After: C0	s = 00001000 0000130 = 0013	(512 × 8)				
	Before: B3 After: C0	= 00000400 0000130 = 0015	(128 × 8)				
	Before: B3 After: C0	= 00001400 0000130 = 1513	(640 × 8)				
	Before: B3 After: C0	= 00000019 000130 = 0000	(25 × 1)				

Syntax	SETF F	<b>SETF</b> <i>FS</i> , <i>FE</i> [, <i>F</i> ]													
Execution	FS, FE → ST														
Instruction Words	15 14 0 0	13 0	12 0	11 0	10 1	9 F	8	7 0	6 1	5 FE	4	3	2 FS	1	0
Description	SETF loa status reg sets the fi the status	ds sp gister eld si s regis	ecifi ; dep ize a ster i	ed fie endii nd ex s not	eld siz ng or ctens affec	ze (F n the ion fo cted.	S) a valu or eit	nd fie le of t her fie	ld ex he <i>F</i> eld 0	tens para or fie	ion ( amet eld 1.	FE) v er, th . (The	value: nis inf e rem	s into <sup>i</sup> orma iainde	the tion er of



For more information, refer to Section 4.1, The Status Register, on page 4-2.

- The *FS* parameter is a value between 1 and 32; it selects the field size. (Note that an FS value of 0 in the opcode corresponds to an actual selected field size of 32.)
- The FE parameter is a value of 0 or 1:

FE=0 selects zero-extension for a field. FE=1 selects sign-extension for a field.

□ The *F* parameter is optional; the default value for *F* is 0. The *F* value determines whether the SETF instruction sets the field size and extension for field 0 or for field 1.

**F=0** selects FS0, FE0 to be altered. **F=1** selects FS1, FE1 to be altered.

Each MOVE instruction also has an *F* parameter that selects the field size and extension of either field 0 or field 1 for the individual move. You can use the SETF instruction to prepare for MOVE instructions.



# Status Bits

N Unaffected

- C Unaffected
- Z Unaffected
- V Unaffected

# Examples

<u>Code</u>		<u>Before</u>	<u>After</u>
		ST	ST
SETF	32,0,0	xxxxx000h	xxxxx000h
SETF	32,1,0	xxxxx000h	xxxxx020h
SETF	31,1,0	xxxxx000h	xxxxx03Fh
SETF	16,0,0	xxxxx000h	xxxxx010h
SETF	32,0,1	xxxxx000h	xxxxx000h
SETF	32,1,1	xxxxx000h	xxxxx800h
SETF	31,1,1	xxxxx000h	xxxxxFC0h
SETF	16,0,1	xxxxx000h	xxxxx400h

Syntax	SEXT Rd [,	<i>F</i> ]												
Execution	field in Rd -	→ sign-ex	tended field F	d										
Instruction Words	15 14 13	12 11	10 9 8	7 6	54	3 2	1	0						
	0 0 0	0 0	1   F   -	0 0	0 R	F	ld							
Description	SEXT sign-e by copying th register. The optional <i>F</i> pa size:	EXT sign-extends the right-justified field contained in the destination register y copying the MSB of the field data into all the nonfield bits of the destination egister. The size of the field is determined by the current field size. The ptional $F$ parameter, which must be specified as a 0 or a 1, selects the field ize:												
	<ul> <li>=0 selects FS0 for the field size.</li> <li>=1 selects FS1 for the field size.</li> </ul>													
	The default value for <i>F</i> is 0.													
Machine States	2													
Status Bits	<ul> <li>N 1 if the re</li> <li>C Unaffect</li> <li>Z 1 if the re</li> <li>V Unaffect</li> </ul>	esult is ne ed esult is 0, ed	gative, 0 othe 0 otherwise	rwise										
Examples	<u>Code</u>	Before	40	<u>After</u> NCZV	۸0									
	SEXT A0.0	17/x	00008000h	0 x 0 x	00008000	)h								
	SEXT A0,0	16/x	00008000h	1 x 0 x	FFFF800	0h								
	SEXT A0,0	15/x	00008000h	0 x 1 x	0000000	Dh								
	SEXT A0,1	x/17	00008000h	0 x 0 x	0008000	Dh								
	SEXT A0,1	x/16	00008000h	1 x 0 x	FFFF800	0h								
	SEXT A0,1	x/15	00008000h	0 x 1 x	00000000	Dh								

Syntax	SLA	col	nstan	it, Ra	1											
Execution	left-s	shift I	Rd by	/ con	stant	. →	Rd									
Instruction Words	15 0	14 0	13 1	12 0	11 0	10 0	9	8 (	7 consta	6 Int	5	4 R	3	2 F	1 Id	0
Description	SLA	left-s	shifts	the d	conte	ents c	of the	e des	tinati	on re	giste	er by	a spe	ecifie	d nui	mber

Description SLA left-shifts the contents of the destination register by a specified number of bits. The shift count is specified by a 5-bit constant; this is a value between 0 and 31.

As shown in the diagram, 0s are shifted into the LSBs. The last bit shifted out of the destination register (the original value of bit [32 - constant]) is shifted into the carry bit. If either the new sign bit (N) or any of the bits shifted out of the register differ from the original sign bit, the overflow bit (V) is set.



The assembler accepts only absolute expressions for the shift count. If the shift count is greater than 31, the assembler issues a warning and sets the constant to its 5 LSBs.

Note that SLA executes slower than SLL because it provides overflow detection.

Machine States

З

Status Bits

- N 1 if the result is negative, 0 otherwise
- C Set to the value of bit [32 constant], 0 for shift count of 0
- Z 1 if a 0 result generated, 0 otherwise
- V 1 if the MSB changes during shift operation, 0 otherwise

Examples	Code		<b>Before</b>	<u>After</u>	
			A1	A1	NCZV
	SLA	0,A1	33333333h	3 3 3 3 3 3	33h 0000
	SLA	0,A1	CCCCCCCCh	CCCCCC	Ch 1000
	SLA	1,A1	CCCCCCCCh	999999	98h 1100
	SLA	2,A1	33333333h	000000	CCh 1001
	SLA	2,A1	CCCCCCCCh	3 3 3 3 3 3	30h 0101
	SLA	3,A1	CCCCCCCCh	666666	60h 0001
	SLA	5,A1	CCCCCCCCh	999999	80h 1101
	SLA	30,A1	CCCCCCCCh	0 0 0 0 0 0	00h 0111
	SLA	31,A1	CCCCCCCCh	0 0 0 0 0 0	00h 0011
	SLA	31.A1	0 0 0 0 0 0 0 0 0 h	0 0 0 0 0 0	00h 0010

# SLA Shift Left Arithmetic, Register

Syntax	SLA Rs, Rd
Execution	left-shift Rd by 5 LSBs of Rs $\rightarrow$ Rd
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         0       1       1       0       0       0       0       Rs       R       Rd
Description	SLA left-shifts the contents of the destination register by a specified number of bits. The shift count is specified by the 5 LSBs of Rs (the 27 MSBs are ignored); this produces a shift count from 0 to 31.
	The last bit shifted out of the destination register (the original value of bit $[32-Rs]$ ) is shifted into the carry bit. If either the new sign bit (N) or any of the bits shifted out of the register differ from the original sign bit, the overflow bit (V) is set.
31 30 29 28 N C Z V	26         25         22         21         11         10—6         5         4—0           BF         IX         SS         IE         FE1         FS1         FE0         FS0

Note: Shaded portions are reserved.

3



Note that SLA executes slower than SLL because it provides overflow detection.

# Machine States

Status Bits

- N 1 if the result is negative, 0 otherwise
- **C** Set to the value of [32 Rs], 0 for shift count of 0
- Z 1 if the result is 0, 0 otherwise
- V 1 if the MSB changes during shift operation, 0 otherwise

Examples

Code	<b>Before</b>		After	
	5 LSBs A0	A1	A1	NCZV
SLA A0,A1	00000	33333333h	33333333h	0000
SLA A0,A1	00000	CCCCCCCCh	CCCCCCC Ch	1000
SLA A0,A1	00001	CCCCCCCCh	99999998h	1100
SLA A0,A1	00010	33333333h	CCCCCCCCh	1001
SLA A0,A1	00010	CCCCCCCCh	33333330h	0101
SLA A0,A1	00011	CCCCCCCCh	6666660h	0001
SLA A0,A1	00101	CCCCCCCCh	99999980h	1101
SLA A0,A1	11110	CCCCCCCCh	00000000h	0111
SLA A0,A1	11111	CCCCCCCCh	00000000h	0011
SLA A0,A1	11111	00000000h	00000000h	0010

TMS34020 Assembly Language Instruction Set

Syntax	SLL	constant, Rd
--------	-----	--------------

*Execution* left-shift Rd by constant  $\rightarrow$  Rd

Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	1	0	0	1		С	consta	nt		R		R	d	

**Description** SLL left-shifts the contents of the destination register by a specified number of bits. The shift count is specified by a 5-bit constant, which is a value between 0 and 31.

The last bit shifted out of the destination register (the original value of bit [32 - constant]) is shifted into the carry bit. Os are shifted into the LSBs. This instruction differs from the SLA instruction only in its effect on the overflow (V) bit.



The assembler only accepts absolute expressions for the shift count. If the specified shift count is greater than 31, the assembler issues a warning and sets the constants to its 5 LSBs.

Machine States

#### Status Bits

N Unaffected

- **C** set to the value of bit [32 constant], *0* for shift count of 0
- **Z** *1* if the result is 0, *0* otherwise
- V Unaffected

Examples	<u>Code</u>	<u>Before</u>	<u>After</u>	
		A1	A1	NCZV
	SLL 0,A1	00000000h	00000000h	x 0 1 x
	SLL 0,A1	88888888h	8888888h	x 0 0 x
	SLL 1,A1	88888888h	11111110h	x 1 0 x
	SLL 4,A1	88888888h	88888880h	x 0 0 x
	SLL 30,A1	FFFFFFCh	00000000h	x11x
	SLL 31,A1	FFFFFFCh	0000000h	x 0 1 x

Syntax	SLL	Rs, Rd
--------	-----	--------

*Execution* left-shift Rd by 5 LSBs of Rs  $\rightarrow$  Rd

Instruction Words

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
0	1	1	0	0	0	1		F	ls		R		R	d	

Description SLL left-shifts the contents of the destination register by a specified number of bits. The shift count is specified by the 5 LSBs of Rs (the 27 MSBs are ignored); this produces a shift count between 0 and 31.

The last bit shifted out of the destination register (the original value of bit [32 - Rs]) is shifted into the carry bit. Os are shifted into the LSBs. This instruction differs from the SLA instruction only in its effect on the overflow (V) bit.



Rs and Rd must be in the same register file.

Machine States	1
Status Bits	N
	<b>^</b>

- N Unaffected
- **C** set to the value of bit [32 Rs], *0* for shift count of 0
- Z 1 if the result is 0, 0 otherwise
- V Unaffected

<u>Code</u>	<b>Before</b>		<u>After</u>	
	5 LSBs A0	A1	A1	NCZV
SLL A0,A1	00000	00000000h	0000000h	x 0 1 x
SLL A0,A1	· 00000	88888888h	8888888h	x 0 0 x
SLL A0,A1	00001	88888888h	11111110h	x10x
SLL A0,A1	00100	88888888h	8888880h	x 0 0 x
SLL A0,A1	11110	FFFFFFCh	0000000h	x11x
SLL A0,A1	11111	FFFFFFCh	00000000h	x 0 1 x

Syntax	SRA	constant, Rd
--------	-----	--------------

*Execution* right-shift Rd by constant  $\rightarrow$  Rd

Instruction Words

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
0	0	1	0	1	0	2	s compl	ement o	of consta	nt	R		R	d	

**Description** SRA right-shifts the contents of the destination register by a specified number of bits. The shift count is specified by constant which is a 5-bit immediate value; this produces a shift count of 0 to 31.

The last bit shifted out of the destination register (the original value of [constant-1]) is shifted into the carry bit. The sign bit (MSB) is extended into the MSBs.



The assembler accepts only absolute expressions for the shift count. If the specified shift amount is greater than 31, the assembler issues a warning, takes the 2s complement of the constant and places it in the opcode.

Machine States

#### Status Bits

- **N** *1* if the result is negative, *0* otherwise
- **C** Set to the value of [constant -1], 0 for shift count of 0
- **Z** *1* if the result is 0, *0* otherwise
- V Unaffected

1

<u>Code</u>	<u>Before</u>	<u>After</u>	
	A1	A1	NCZV
SRA 0,A1	00000000h	00000000h	001x
SRA 0,A1	FFFF0000h	FFFF0000h	100x
SRA 8,A1	7FFF0000h	007FFF00h	0 0 0 x
SRA 8,A1	FFFF0000h	FFFFFF00h	100x
SRA 30,A1	7FFF0000h	0000001h	010x
SRA 31,A1	7FFF0000h	00000000h	011x
SRA 31,A1	FFFF0000h	FFFFFFFh	110x

Syntax	SRA	Rs	, Rd												
Execution	right	-shift	Rd I	oy 2s	com	plem	ent o	of 5 L	_SBs	in R	s →	Rd			
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1
	0	1	1	0	0	1	0		F	٦s		R		F	۲d

Description

SRA right-shifts the contents of the destination register by a specified number of bits. The shift amount is specified by the 2s complement of the 5 LSBs of Rs (the 27 MSBs of Rs are ignored); this produces a shift count between 0 and 31.

0

The last bit shifted out of the destination register (the original value of bit [shift amount - 1]) is shifted into the carry bit. The sign bit (MSB) is extended into the MSBs.



# Machine States

Status Bits

- N 1 if the result is negative, 0 otherwise
- **C** Set to the value of bit [shift amount -1], 0 for shift count of 0
- **Z** 1 if the result is 0, 0 otherwise
- V Unaffected

1

<u>Code</u>	Before		After	
	5 LSBs A0	A1	A1	NCZV
SRA A0,A1	00000	00000000h	00000000h	001x
SRA A0,A1	00000	FFFF0000h	FFFF0000h	100x
SRA A0,A1	11111	7FFF0000h	3FFF8000h	0 0 0 x
SRA A0,A1	11111	FFFF0000h	FFFF8000h	100x
SRA A0,A1	11000	7FFF0000h	007FFF00h	0 0 0 x
SRA A0,A1	11000	FFFF0000h	FFFFFF00h	100x
SRA A0,A1	00010	7FFF0000h	00000001h	010x
SRA A0,A1	00001	7FFF0000h	0000000h	011x
SRA A0,A1	00001	FFFF0000h	FFFFFFFh	110x

3

4

R

2

1

Rd

0

	0	0	1	0	1	1	2	s compl	ement o	f consta	nt
Instruction Words		14	13	12	11	10	9	8	7	6	5
Execution	right	shift	Rd b	y co	nstar	nt →	Rd				
Syntax	SRL	cor	istan	t, Rd							

*Description* SRL right-shifts the contents of the destination register by a specified number of bits. The shift amount is specified by the constant which is a 5-bit immediate value; this produces a shift count between 0 and 31.

The last bit shifted out of the destination register (the original value of [constant-1]) is shifted into the carry bit. Os are shifted into the MSBs.



The assembler accepts only absolute expressions for the shift count. If the specified shift amount is greater than 31, the assembler issues a warning, takes the 2s complement of the constant and places it in the opcode.

Machine States

Status Bits

N Unaffected

1

- **C** Set to the value of [constant 1], 0 for shift count of 0
- Z 1 if the result is 0, 0 otherwise
- V Unaffected

Code	e	<u>Before</u>	After	
		A1	A1	NCZV
SRL	0,A1	00000000h	0000000h	x 0 1 x
SRL	0,A1	7FFFFFFh	7FFFFFFFh	x 0 0 x
SRL	1,A1	7FFFFFFh	3FFFFFFFh	x10x
SRL	8,A1	7FFF0000h	007FFF00h	x 0 0 x
SRL	30,A1	7FFF0000h	0000001h	x 1 0 x
SRL	31,A1	7FFF0000h	0000000h	x11x
SRL	31,Al	3FFF0000h	0000000h	x 0 1 x

Syntax	SRL Rs, Rd											
Execution	right-shift Rd by 2s complement of 5 LSBs in Rs $\rightarrow$ Rd											
Instruction Words	15 14 13	12 11 10	987	65	4 3	2	1	0				
	0 1 1	0 0 1	1	Rs	R	Ro	4					
Description	SRL right-shifts the contents of the destination register by a specified number of bits. The shift amount is specified by the 2s complement of the 5 LSBs of Rs (the 27 MSBs of Rs are ignored); this produces a shift value of 0 to 31.											
	The last bit s [shift amount	shifted out of t – 1]) is shifted	he destination into the carry	register( bit. 0s are	the origi shifted i	inal val nto the	ue c MS	of bit Bs.				
	0s <b>-</b> MSE			L	0  SB		, - <b>-</b> [					
Machine States	1											
Status Bits	<ul> <li>N Unaffected</li> <li>C Set to the value of bit [shift amount - 1], 0 for shift count of 0</li> <li>Z 1 if the result is 0, 0 otherwise</li> <li>V Unaffected</li> </ul>											
Examples	<u>Code</u>	Before	· A4	<u>After</u>			• \ /					
	SPT. 30 31	5 LSBS AU	A1 00000000b	A1 00000	100h		. v					
	SRL A0.A1	00000	7FFFFFFFh	7FFFF	FFFh	× 0 0	x					
	SRL A0,A1	11111	7FFFFFFFh	3FFFF	FFFh	x 1 0	x					
	SRL A0,A1	11000	7FFF0000h	007FF	F00h	x 0 0	x					
	SRL A0,A1	00010	7FFF0000h	00000	001h	x 1 0	x					
	SRL A0,A1	00001	7FFF0000h	00000	000h	x 1 1	x					
	SRL A0,A1	00001	3FFF0000h	00000	000h	x 0 1	x					

Syntax	SUB Rs, Rd											
Execution	$Rd - Rs \rightarrow Rd$											
Instruction Words	15 14 13 0 1 0	12 11 10 9 0 0 1 0	8 7 6 D Rs	5 4 R	3 2 1 0 Rd							
Description	SUB subtracts the contents of the source register from the contents of the destination register and stores the result in the destination register.											
	with the SUBB instruction.											
	Rs and Rd must be in the same register file.											
Machine States	1											
Status Bits	<ul> <li>N 1 if the result is negative, 0 otherwise</li> <li>C 1 if there is a borrow, 0 otherwise</li> <li>Z 1 if the result is 0, 0 otherwise</li> <li>V 1 if there is an overflow, 0 otherwise</li> </ul>											
Examples	Code           SUB A1,A0           SUB A1,A0	Before A0 7FFFFF2h 7FFFFF2h 7FFFFF1h 7FFFFFF1h 7FFFFFFh FFFFFFFDh FFFFFFFDh FFFFFFFDh FFFFFFFF	A1 7FFFFFF1h 7FFFFFF2h 7FFFFFFF5 FFFFFFFF FFFFFFFF FFFFFFFF FFFFFF	After N C Z V 0 0 0 0 1 1 0 0 1 1 0 0 1 1 0 1 1 1 0 1 1 1 0 0 0 0 1 0 0 0 0 0 1 0 0 0 0 0 0 1	<b>A0</b> 00000001h 00000000h FFFFFFFh 7FFFFFFh 80000000h FFFFFFFFh 00000000h FFFFFFFFh 7FFFFFFFh							
# SUBB Subtract Registers with Borrow

Syntax	SUBB Rs, F	?d									
Execution	Rd – Rs – C	→ F	Rd (the carry	bit acts as	a borr	ow)					
Instruction Words	15 14 13	12	11 10 9	87	6	5 4	4	3	2	1	0
	0 1 0	0	0 1 1	F	Rs	1	R		R	d	
Description	SUBB subtraction the contents of register.	ots b of the	oth the conten destination re	ts of the sou gister, and	urce re stores	egister the re	r and esult	d the t in t	e cari he de	ry bit estina	from ation
	You can use extended-pre	this i cisio	instruction with n arithmetic.	n the SUB,	SUBK	K, and	SU	BI ii	nstru	ction	s for
	Rs and Rd m	ust b	e in the same	register file	э.						
Machine States	1										
Status Bits	<ul> <li>N 1 if the result is negative, 0 otherwise</li> <li>C 1 if there is a borrow, 0 otherwise</li> <li>Z 1 if the result is 0, 0 otherwise</li> <li>V 1 if there is an overflow, 0 otherwise</li> </ul>										
Examples	<u>Code</u>	Bet	fore			Afte	r				
		С	A0	A1		NC	zν	A	0		
	SUBB A1,A0	0	0000002h	0000001	h	000	0	00	0000	001h	
	SUBB A1,A0	1	00000002h	00000001	h	001	0	00	0000	)00h	
	SUBB A1,A0	0	0000002h	0000002	?h	001	0	00	0000	)00h	
	SUBB A1,A0	1	00000002h	0000002	!h	110	0	FF	FFF	FFFh	
	SUBB A1,A0	0	00000002h	0000003	h	110	0	FF	FFF	FFFh	
	SUBB A1,A0	0	7FFFFFFEh	FFFFFFF	Fh	010	0	7F	FFF	FFFh	
	SUBB A1,A0	0	7FFFFFFEh	FFFFFFF	Eh	110	1	80	0000	)00h	
	SUBB A1,A0	1	7FFFFFFEh	FFFFFFF	Eh	010	0	7F	FFF	FFFh	
	SUBB A1,A0	0	FFFFFFEh	FFFFFF	Fh	110	0	FF	FFF	FFFh	
	SUBB A1,A0	0	FFFFFFEh	FFFFFF	Eh	001	<b>0</b> ·	00	0000	000h	
	SUBB A1,A0	1	FFFFFFEh	FFFFFF	Eh	110	Ö	FF	FFF	FFFh	
	SUBB A1,A0	0	FFFFFFFEh	FFFFFFF	Dh	000	0	00	0000	001h	
	SUBB A1.A0	1	FFFFFFEh	FFFFFF	Dh	001	0	00	0000	)00h	
	SUBB A1,A0	0	80000001h	00000001	h	100	0	80	0000	)00h	
	SUBB A1,A0	1	80000001h	00000001	h	000	1	7F	FFF	FFFh	

SUBB A1,A0 0 8000001h

0001

7FFFFFFh

00000002h

Syntax	SUBI IW, Rd [, <b>W</b> ]									
Execution	$Rd - IW \rightarrow Rd$									
Instruction Words	15 14 13 12 11 10	9 8 7 6 5	4 3 2 1 0							
	0 0 0 0 1 0	1 1 1 1 1	R Rd							
	L	Is complement of IW	·····							
Description	SUBI subtracts a sign-extended, 16-bit immediate value from the contents of the destination register, and stores the result in the destination register. (The $IL$ in the syntax represents a sign-extended, 16-bit immediate value.)									
·	The assembler uses this form of the SUBI instruction if the immediate value was previously defined and is in the range $-32,768$ to $32,767$ . You can force the assembler to use the short form by following the register operand with <b>W</b> :									
	SUBI IW, Rd, W									
	The assembler truncates any message. You can accomplish conjunction with the SUBB inst	upper bits and issue multiple-precision a truction.	s an appropriate warning rithmetic by using SUBI in							
Machine States	2									
Status Bits	<ul> <li>N 1 if the result is negative,</li> <li>C 1 if there is a borrow, 0 oth</li> <li>Z 1 if the result is 0, 0 other</li> <li>V 1 if there is an overflow, 0</li> </ul>	0 otherwise nerwise vise otherwise								
Examples	Code Before	After								
•	A0	AO	NCZV							
	SUBI 32765,A0 00007FFEh	0000001h	0000							
	SUBI 32766,A0 00007FFEh	00000000h	0010							
	SUBI 32767,A0 00007FFEh	FFFFFFFh	1100							
	SUBI 32766,A0 80007FFEh	8000000h	1000							
	SUBI 32767,A0 80007FFEh	7FFFFFFh	0001							
	SUBI -32766,A0 FFFF8001h	FFFFFFFh	1100							
	SUBI -32767,A0 FFFF8001h	0000000h	0010							
	SUBI -32768,A0 FFFF8001h	0000001h	0000							
	SUBI -32767,A0 FFFF8000h	7FFFFFFh	0100							
	SUBI -32768,A0 7FFF8000h	8000000h	1101							

Syntax	SUBI IL, Rd [, L]									
Execution	$Rd - IL \rightarrow Rd$									
Instruction Words	15 14 13 12 11	10 9 8	765	4	3	2	1	0		
	0 0 0 0 1	1 0 1	0 0 0	R		R	d			
		1s complement of	16 LSBs of IL							
		1s complement of	16 MSBs of IL							
Description	SUBI subtracts a signed nation register, and store syntax represents a sign	SUBI subtracts a signed 32-bit immediate value from the contents of the desti- nation register, and stores the result in the destination register. (The <i>IL</i> in the syntax represents a signed 32-bit immediate value.)								
	The assembler uses this SUBI IW,Rd opcode, or it operand with <b>,L</b> :	The assembler uses this version of the SUBI instruction if it cannot use the SUBI IW,Rd opcode, or if you request the long opcode by following the register operand with <b>,L</b> :								
	SUBI IL, Rd, L									
	You can accomplish mult with the SUBB instruction	iple-precision ar n.	ithmetic by us	sing S	SUBI	in co	njunc	tion		
Machine States	2 if immediate data is long-word aligned 3 if immediate data is not long-word aligned									
Status Bits	<ul> <li>N 1 if the result is negative.</li> <li>C 1 if there is a borrow</li> <li>Z 1 if the result is 0, 0</li> <li>V 1 if there is an overful</li> </ul>	ative, 0 otherwis v, 0 otherwise otherwise ow, 0 otherwise	e							
Examples	<u>Code</u>	<u>Before</u>	After							
		A0	A0		NC	ΖV				
	SUBI 2147483647,A0	7FFFFFFFh	0000000	า	000	) <b>1</b>				
	SUBI 32768,A0	00008001h	00000011	n	000	0 0				
	SUBI 32769,A0	00008001h	00000001	ำ	00	10				
	SUBI 32770,A0	00008001h	FFFFFFF	h	110	0 0				
	SUBI 32768,A0	80008000h	80000001	า	100	0 0				
	SUBI 32769,A0	80008000h	7FFFFFF	ħ	000	) 1				
	SUBI —2147483648,A0	8000000h	00000000	า	00	10				
	SUBI -32769,A0	FFFF7FFEh	FFFFFFF	-h	11(	0 0				
	SUBI -32770,A0	FFFF7FFEh	0000000	า	00.	10				
	SUBI -32771,A0	FFFF7FFEh	00000011	n	000	00				
	SUBI -32770,A0	7FFF7FFDh	7FFFFFFF	h	01(	00				
	SUBI -32771,A0	7FFF7FFDh	80000008	n	11(	J 1				

.

Syntax	SUBK constant, Rd									
Execution	Rd – constant	→ Rd								
Instruction Words	15 14 13 0 0 0	12 11 1 1 0	10 9 1.	8 7 constant	65	4 R	3	2 Ro	1 d	0
Description	SUBK subtracts the 5-bit constant from the contents of the destination register; the result is stored in the destination register. The constant is an unsigned num- per in the range 1—32. Note that constant=0 in the opcode corresponds to the value 32; the assembler converts the value 32 to 0. Using this instruction, the assembler issues an error if you try to subtract 0 from a register.									
	You can accortion with the S	nplish multi UBB instruc	ple-prec ction.	ision arithr	netic by (	using	) SUE	3K in	conji	nc-
Machine States	1									
Status Bits	<ul> <li>N 1 if the res</li> <li>C 1 if there is</li> <li>Z 1 if the res</li> <li>V 1 if there is</li> </ul>	sult is negat s a borrow, sult is 0, <i>0</i> o s an overflo	ive, 0 ot 0 otherv therwise w, 0 oth	herwise vise e erwise						
Examples	Code SUBK 5,A0 SUBK 9,A0 SUBK 32,A0 SUBK 1,A0	Before A0 00000009h 0000009h 0000009h 80000000h	<u>Afte</u> A0 0000 0000 FFF 7FF	<u>r</u> 00004h 00000h FFFE9h FFFFh	N C Z V 0 0 0 0 0 0 1 0 1 1 0 0 0 0 0 1	,				

SUBXY Rs, R	d					
Rd.X – Rs.X – Rd.Y – Rs.Y –	<ul> <li>&gt; Rd.X</li> <li>&gt; Rd.Y</li> </ul>					
15 14 13 1 1 1	12 11 10 0 0 0	9 8 7 1 I	65 Rs	4 3 R	2 1 Rd	0
SUBXY subtrac X and Y values	ts the source ; the result is	X and Y value stored in the o	es individua destination	Illy from th register.	ne destina	ition
You can use th useful for increr in the register f	is instruction nental figure c lle.	for manipulati Irawing. These	ng XY addr e addresses	resses; it i s are store	is particul ed as XY p	larly airs
Rs and Rd mus	t be in the sa	me register fil	e.			
1						
<ul> <li>N 1 if source</li> <li>C 1 if source</li> <li>Z 1 if source</li> <li>V 1 if source</li> </ul>	X field = desti Y field > desti Y field = desti X field > desti	nation X field, nation Y field, nation Y field, nation X field,	0 otherwis 0 otherwis 0 otherwis 0 otherwis	e e e		
Code SUBXY A1,A0 SUBXY A1,A0 SUBXY A1,A0 SUBXY A1,A0 SUBXY A1,A0 SUBXY A1,A0 SUBXY A1,A0 SUBXY A1,A0 SUBXY A1,A0	Before           A0           00090009h	A1 00010001h 00090001h 00010009h 00090009h 00000010h 00090010h 00100000h 00100009h	After A0 00080008h 0000008h 00080000h 00009FFF9f 0000FFF9f FFF90000f FFF90000f FFF9FF9g	N C 00 100 101 10 100 100 100 100 111 h 01	<b>2 V</b> 0 0 1 0 0 0 1 0 0 1 1 1 0 0 0 0 0 0 0 1	
	SUBXY $Rs, R$ Rd.X - Rs.X = Rd.Y - Rs.Y = 15 14 13 1 1 1 1 SUBXY subtract SUBXY subtract X and Y values You can use the useful for increase in the register fit Rs and Rd muse 1 N 1 if source $T$ C 2 1 if source $T$ C 3 if source $T$ C 4 if source $T$ C 5 if source $T$ C 5 if source $T$ SUBXY A1, A0 SUBXY A1 A1 A1 A1 A1 A1 A1 A1 A1 A1	SUBXY $Rs, Rd$ Rd.X - Rs.X $\rightarrow$ Rd.XRd.Y - Rs.Y $\rightarrow$ Rd.Y $15$ 1413121110111000SUBXY subtracts the sourceX and Y values; the result isYou can use this instruction fuseful for incremental figure ofin the register file.Rs and Rd must be in the sa1N1 if source X field = destiC1 if source Y field > destiZ1 if source X field = destiV1 if source X field > destiSUBXY A1, A000090009hSUBXY A1, A000090009h	SUBXY Rs, RdRd.X - Rs.X $\rightarrow$ Rd.XRd.Y - Rs.Y $\rightarrow$ Rd.Y151413121110987111000111SUBXY subtracts the source X and Y valuesX and Y values; the result is stored in the dividenceYou can use this instruction for manipulatiuseful for incremental figure drawing. Thesein the register file.Rs and Rd must be in the same register fild.1N1IN1IN1IN1IN1Source X field = destination X field,CIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII	SUBXY $Rs, Rd$ Rd.X - Rs.X $\rightarrow$ Rd.XRd.Y - Rs.Y $\rightarrow$ Rd.Y $15$ 1413121110987651110001RsSUBXY subtracts the source X and Y values individual X and Y values; the result is stored in the destinationYou can use this instruction for manipulating XY addresses in the register file.Rs and Rd must be in the same register file.1N1 if source X field = destination X field, 0 otherwise CC1 if source Y field > destination Y field, 0 otherwise VZ1 if source X field = destination X field, 0 otherwise VSUBXY A1,A00009009hSUBXY A1,A00009009h0009009h00010001hSUBXY A1,A00009009h0009009h00010000hSUBXY A1,A00009009hSUBXY A1,A00009009hSUB	SUBXY Rs, RdRd.X - Rs.X $\rightarrow$ Rd.XRd.Y - Rs.Y $\rightarrow$ Rd.Y15141312111098765431110001RsRSUBXY subtracts the source X and Y values individually from th <x and="" destination="" in="" is="" register.<="" result="" stored="" td="" the="" values;="" y="">You can use this instruction for manipulating XY addresses; it useful for incremental figure drawing. These addresses are store in the register file.1N1 if source X field = destination X field, 0 otherwiseC1 if source Y field &gt; destination Y field, 0 otherwiseZ1 if source X field = destination X field, 0 otherwiseZ1 if source X field &gt; destination X field, 0 otherwiseV1 if source X field &gt; destination X field, 0 otherwiseV1 if source X field &gt; destination X field, 0 otherwiseV1 if source X field &gt; destination X field, 0 otherwiseV1 if source X field &gt; destination X field, 0 otherwiseV1 if source X field &gt; destination X field, 0 otherwiseSUBXY A1,A00009009hSUBXY A1,A00009009h<t< th=""><th>SUBXY Rs, RdRd.X - Rs.X <math>\rightarrow</math> Rd.XRd.Y - Rs.Y <math>\rightarrow</math> Rd.Y1514131211109876543211110001RsRRdSUBXY subtracts the source X and Y values individually from the destinat X and Y values; the result is stored in the destination register.You can use this instruction for manipulating XY addresses; it is particul useful for incremental figure drawing. These addresses are stored as XY p in the register file.Rs and Rd must be in the same register file.1N1 if source X field = destination X field, 0 otherwiseZ1 if source Y field &gt; destination Y field, 0 otherwiseZ1 if source X field &gt; destination X field, 0 otherwiseV1 if source X field &gt; destination X field, 0 otherwiseZ1 if source X field &gt; destination X field, 0 otherwiseZ1 if source X field &gt; destination X field, 0 otherwiseV1 if source X field &gt; 0009009hSUBXY A1, A00009009h00010001h00080008h0010009h0000000hSUBXY A1, A00009009h0009009h0000000hSUBXY A1, A00009009hSUBXY A1, A00009009h0009009h00000000hSUBXY A1, A00009009h0009009h00000000h0009009h0000000hSUBXY A1, A00009009hSUBXY A1, A00009009hSUBXY A1, A00009009hSUBXY A1,</th></t<></x>	SUBXY Rs, RdRd.X - Rs.X $\rightarrow$ Rd.XRd.Y - Rs.Y $\rightarrow$ Rd.Y1514131211109876543211110001RsRRdSUBXY subtracts the source X and Y values individually from the destinat X and Y values; the result is stored in the destination register.You can use this instruction for manipulating XY addresses; it is particul useful for incremental figure drawing. These addresses are stored as XY p in the register file.Rs and Rd must be in the same register file.1N1 if source X field = destination X field, 0 otherwiseZ1 if source Y field > destination Y field, 0 otherwiseZ1 if source X field > destination X field, 0 otherwiseV1 if source X field > destination X field, 0 otherwiseZ1 if source X field > destination X field, 0 otherwiseZ1 if source X field > destination X field, 0 otherwiseV1 if source X field > 0009009hSUBXY A1, A00009009h00010001h00080008h0010009h0000000hSUBXY A1, A00009009h0009009h0000000hSUBXY A1, A00009009hSUBXY A1, A00009009h0009009h00000000hSUBXY A1, A00009009h0009009h00000000h0009009h0000000hSUBXY A1, A00009009hSUBXY A1, A00009009hSUBXY A1, A00009009hSUBXY A1,

Syntax	SWAPF	*Rs, Rd,0
--------	-------	-----------

ExecutionField specified by \*Rs and FS0  $\rightarrow$  RdRd  $\rightarrow$  field specified by \*Rs and FS0

Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	1	1	1	1	1	1		F	۱s		R		R	d	

**Description** This instruction performs a read (modify) write operation on a field in the memory space. It exchanges the field specified by the contents of Rs and FS0 with Rd. The new contents of Rd are right-justified and either sign- or zero-extended, depending on the value of FE0.

The main reason for the inclusion of this instruction is to allow the implementation of the *test and set* and *test and clear* operations needed for the lowest level of interprocess and interprocessor synchronization.

Note that this instruction does *not* complete until the write is complete (implicit MWAIT). This makes the instruction useful in some I/O register operations. Once the instruction starts, host access requests will not be granted until all the memory SWAPF accesses required are complete. If the read (modify) write is interrupted after the read by a screen refresh or a loss of of bus grant ( $\overline{GI}$  high), or if a retry or bus fault occurs at any time during the cycle, the operation is restarted from the beginning of the read. This makes the operation indivisible. The bus lock status code is output during all SWAPF cycles.

Note:

The following restrictions apply to SWAPF:

- 1) The field must **not** span a 32-bit word boundary. The field is ignored if any part of it is not contained in the same 32-bit word specified by the bit address contained in Rs.
- 2) If SWAPF is used to access 16-bit memories, any part of the field not contained in the first 16-bit word is ignored.

Machine States

Refer to Section 15.1 on page 15-2.

Status Bits

- N 1 if the field-extended data moved to register is negative, 0 otherwise.
- C Unaffected
- **Z** *1* if the field-extended data moved to register 0, *0* otherwise.
- **V** 0

•

#### Examples

			;Test and Set-wait for resource
	SETF	1,0,0	;Single bit
	MOVI	SEMA,A1	;Bit to test and set
WAS_SET:	MOVK	1,A0	;Set is not already set
	SWAPF	*A1,A0	;Test and set
	JRNE	WAS_SET	;Already set-did not get resource
			;Test and Clear-wait for resource
	SETF	1,0,0	;Single bit
	MOVI	SEMA,A1	;Bit to test and clear
WAS_CLR:	CLR	A0	;Clear if not already clear
	SWAPF	*A1,A0	;Test and clear
	JREQ	WAS_CLR	;Already clear-did not get resource
			;Graphics mode save
	MOVI	CONTROL+5,A1	;Point at CONTROL register
	SETF	10,0,0	;Ten bits
	MOVI	NEWMODE, A0	;New value
	SWAPF	*A1,A0	;Read oldmode, set new mode
	CALL	GRAPHOP	;Perform some operation
	MOVE	A0,*A1,0	;Restore old mode

Syntax	TFILL XY					
Execution	COLOR1 pixels fill the horizontal line from $(X_1, Y)$ to $(X_2, Y)$ then					
	$\begin{array}{rcl} X_1 & := & X_1 + DX_1 \\ X_2 & := & X_2 + DX_2 \\ Y & := & Y + 1 \end{array}$					
Instruction Words	<u>15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0</u>					
	0 0 0 0 1 1 1 0 1 1 1 1 1 0 1 0					
Description	TFILL draws a horizontal line and then adjusts its implied operands. The oper- ands are set so that each subsequent call to TFILL will draw one more horizon- tal line, creating lines that build up to form a trapezoid.					

The trapezoid is defined as shown in the diagram:

Figure 13–11. A Trapezoidal Fill



Note that the coordinate parameters for this instruction are specified in the fixed-point format; that is, the 16 MSBs define the signed part of the coordinate and the 16 LSBs define the fractional part of the coordinate. This is the case for both X and Y coordinates, although the Y coordinate will never have a fractional part.

The DX<sub>1</sub> and DX<sub>2</sub> values can have fractional components. This allows for nonintegral slopes at the trapezoid sides. The fractional components are used to determine the new endpoints for the next line. However, only the 16 MSBs are used to determine the XY address of the endpoints.

Note that if  $X_2 \le X_1$  no pixels are drawn, but the contents of  $X_1$  and  $X_2$  are still incremented by  $DX_1$  and  $DX_2$  respectively.

#### Implied Operands

Register	Name	Format	Description				
B0	SADDR	Fixed	X coordinate of X1				
B1	SPTCH	Fixed	DX1 (adjustment for X1)				
B2	DADDR	XY	Used as temp (not user determined)				
B3	DPTCH	Linear	Destination pixel array pitch (usually screen pitch)				
B4	OFFSET	Linear	Screen offset				
B5	WSTART	XY	Window start				
B6	WEND	XY	Window end				
B7	DYDX	Fixed	X coordinate of of X2				
B9	COLOR1	Pixel	Foreground color				
B10	MADDR	Fixed	DX2 (adjustment for X2)				
B11	MPTCH	Fixed	Y coordinate of X1 and X2				
Address	Name	Descript	ion and Elements (Bits)				
C00000B0h	CONTROL	PP — Pixel-processing operations (22 options) W — Window checking operation T — Transparency operation TMODE — Selects 1 of 3 transparency options					
C0000140h	CONVDP	XY-to-li	near conversion (destination pitch)				
C0000150h	PSIZE	Pixel siz	ze (1,2,4,8,16,32)				
C0000160h	PMASK (32 bits)	2 Plane m	nask – pixel format				

To set up the initial values for  $X_1$ ,  $X_2$ , and Y from 2 starting addresses ( $X_1$ , Y) and ( $X_2$ , Y), complete the following steps:

- 1) Use MOVY to copy the Y address into MPTCH.
- Use SLL to shift the 2 XY addresses left by 16 bits. This results in 2 fixedpoint X coordinates.
- Use MOVY to copy the 2 X addresses into SADDR and DYDX, respectively.
- **Pixel Processing**

Pixel processing can be used with this instruction. PPOP [CONTROL] specifies the pixel-processing operation that is applied to pixels as they are processed with the destination array. There are 16 Boolean and 6 arithmetic operations; the default case at reset is the *replace*  $(S \rightarrow D)$  operation. Note that the destination data is read through the plane mask and then processed. The 6 arithmetic operations, do not operate with a pixel size of 1 bit per pixel. For more information, refer to Section 12.8, <u>Pixel Processing</u>, on page 12-27.

*Window Checking* The window operations can be used with this instruction. For more information, refer to Section 12.7, <u>Window Checking</u>, on page 12-19.

Transpare	ncy	You can enabl Select 1 of 3 ti mation, refer	You can enable transparency for this instruction by setting T[[CONTROL]] to Select 1 of 3 transparency modes by setting TM[[CONTROL]]. For more info mation, refer to Section 12.9, <u>Transparency</u> , on page 12-36.							
Interrupts		This instruction can be interrupted at a word or row boundary of the array. For more information, refer to Section 6.6, <u>Interrupting Grap</u> <u>tions</u> , on page 6-13.								
Plane Mas	<i>sking</i> The plane mask is enabled for this instruction. For more informa Section 12.10, <u>Plane Masking</u> , on page 12-39.									
Status Bits	5	<ul><li>N Unaffecte</li><li>C Unaffecte</li><li>Z Unaffecte</li><li>V Unaffecte</li></ul>								
Example		This is an exa ziums on the ments:	mple of a C-compatible assembly routine which draws trape- screen, using the TFILL instruction.This function has 6 argu-							
		(x1a, x2a, ya) (x1b, x2b, yb)	<ul> <li>coordinates of top of trapezoid</li> <li>coordinates of bottom of trapezoid</li> </ul>							
		This routine as	sumes the following registers have been initialized by the caller:							
		B-file registers I/O registers	DPTCH, OFFSET, WSTART, WEND, and COLOR1 CONTROL, CONVDP, PSIZE and PMASK							
STK SADDR SPTCH DYDX MADDR MPTCH _tfill:	.set .set .set .set .set mmtm move sll move sll move sll move sll move sll move sll move sll move sll move	A14 B0 B1 B7 B10 B11 SP, B0, B1, B2, B7, STK, B14 *-B14, SADDR, 1 16, SADDR *-B14, DYDX, 1 16, DYDX *-B14, MPTCH, 1 *-B14, SPTCH, 1 16, SPTCH *-B14, B13, 1 16, B13 *-B14, B12, 1 B14, STK SADDR, SPTCH DYDX, B13 MPTCH, B12 B12, SPTCH	<pre>; C-parameter stack pointer ; Source address register ; Source pitch register ; Delta X/delta Y register ; Mask address register ; Mask pitch register .globl _tfill B10,B11,B12,B13,B14 ;get C-parameter stack into B-file ;pop x1a ;convert to fixed point ;pop x2a ;convert to fixed point ;pop ya ;pop x1b ;convert to fixed point ;pop x2b ;convert to fixed point ;pop yb ;update C-parameter stack ;delta x1 ;delta x2 ;delta y ;dx1</pre>							

•

	divs	B12,B13	;dx1 (cant use MADDR since divs ;requires odd numbered register)
	move	B13,MADDR	;copy into MADDR
	sll	16,MPTCH	;convert y to fixed point
loop:	tfill dsjs mfm rets	B12,loop SP,B0,B1,B2,B7,B10,E 2	311,812,813,814

Syntax	TR	AP N													
Execution	PC ST trap	$\rightarrow -*SP$ $\rightarrow -*SP$ vector N $\rightarrow$	• PC	;											
Instruction Words	15 0	14 13 0 0	12 0	11 1	10 0	9 0	8	7 0	6 0	5 0	4	3	2 N	1	0
Description       TRAP executes a software interrupt. The N parameter is a trap numb         0 to 31 that selects the trap to be executed. During a software interrupt.											nber f rupt:	rom			
	G	The 32-bit return address, PC (the address of the next instruction pushed on the stack.											ction	), is	
		The 32-bit	statu	ıs reg	gister	; ST,	is pu	shec	l on t	he st	ack.				
		The stack p	oint	er, S	P, is	decre	emen	ted b	oy 64						
The IE (interrupt enable) bit in ST is set to 0, disabling maskable and ST is set to 00000010h.									ble ir	nterru	ıpts,				
		Finally, the	trap	vect	or is	loade	ed int	o the	PC.						
The TMS34020 generates the trap vector addresses as shown page:										wn or	nthe	follov	ving		

Trap Number	Address	Name	Description
0	FFFF FFE0h	Reset	Reset
1	FFFF FFC0h	INT1	External interrupt 1
2	FFFF FFA0h	INT2	External interrupt 2
3	FFFF FF80h		
4	FFFF FF60h		Reserved for future
5	FFFF FF40h		hardware or on-chip
6	FFFF FF20h		interrupts
7	FFFF FF00h		
8	FFFF FEE0h	NMI	Nonmaskable interrupt
9	FFFF FEC0h	HI	Host interrupt
10	FFFF FEA0h	DI	Display interrupt
11	FFFF FE80h	WV	Window violation interrupt
12	FFFF FE60h		
13	FFFF FE40h		Reserved for future
14	FFFF FE20h		interrupts
15	FFFF FE00h		•
16	FFFF FDE0h		
			Application defined
29	FFFF FC40h	•	
30	FFFF FC20h	ILLOP	Illegal opcode interrupt
31	FFFF FC00h	Trap 31	Application defined

#### Figure 13–12. Vector Address Map

Note: Traps 0-31 may be accessed by either TRAP or TRAPL instructions.

The stack, which is located in external memory, grows toward lower addresses. Unlike an interrupt, a software trap cannot be disabled.

#### Note:

- 1) TRAP 0 differs from all other traps; it does not save the old status register or program counter. This may be useful in cases where the stack pointer is corrupted or uninitialized; such a situation could cause an erroneous write.
- 2) NMIM[[HSTCTLH]] does not affect the operation of TRAP 8: the PC and ST are always pushed onto the stack.

Machine States

7 if TRAP 0, else 10 if ST aligned, else 12

#### Status Bits

С	0
Ζ	0
v	0

N 0

Examples

Assume that memory contains the following values before instruction execution:

Address	Data
FFFFFC00	0000
FFFFFC10	FFE0
FFFFFC20	0000
FFFFFC30	FFD0
FFFFFFC0	0000
FFFFFFD0	FFB0
FFFFFFE0	0000
FFFFFFF0	FFA0

<u>Code</u>		<u>Before</u>		<u>After</u>								
		PC	SP	PC	SP	ST						
TRAP	0	xxxxxxxh	80000000	FFA00000	80000000h	00000010h						
TRAP	1	xxxxxxxh	80000000	FFB00000	7FFFFFC0h	00000010h						
TRAP	30	xxxxxxxh	80000000	FFD00000	7FFFFFC0h	00000010h						
TRAP	31	xxxxxxxh	80000000	FFE00000	7FFFFFC0h	00000010h						

Syntax	TRA	PL														
Execution	PC ST trap	$PC \rightarrow -*SP$ $ST \rightarrow -*SP$ trap vector $N \rightarrow PC$														
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
·	0	0	0	0	1	0	0	0	0	0	0	0	1	1	1	1
16-bit trap number N																
Description	TRA	PL e	xecu	tes a	soft	ware	inter	rupt.	The	Np	aram	eter i	is a s	signe	d nu	mber

TRAPL executes a software interrupt. The *N* parameter is a signed number from -32,768 to 32,767. The trap address is formed by taking the 16-bit signed immediate operand *N*, shifting it left by 5 bits and then sign-extending it. TRAPL can cover a significantly larger address range than the TRAP instruction. During a software interrupt:

- The 32-bit return address, PC (the address of the next instruction), is pushed on the stack.
- The 32-bit status register, ST, is pushed on the stack.
- The stack pointer, SP, is decremented by 64.
- The IE (interrupt enable) bit in ST is set to 0, disabling maskable interrupts, and ST is set to 00000010h.
- Finally, the trap vector is loaded into the PC.

Note that unlike TRAP 0, the TRAPL 0 is not treated as an exception. That is, TRAPL 0 saves the PC and ST on the stack, whereas TRAP 0 does not.

The TMS34020 generates the trap vector addresses as shown on the following page:

Trap Number	Address	Name	Description
-32768	000F FFE0h	Trap - 32768	Application defined
1	0000 0000h	Trap –1	
0	FFFF FFE0h	Reset	Reset
1	FFFF FFC0h	INT1	External interrupt 1
2	FFFF FFA0h	INT2	External interrupt 2
3	FFFF FF80h	Trap 3	
4	FFFF FF60h	Trap 4	Reserved for future
5	FFFF FF40h	Trap 5	hardware or on-chip
6	FFFF FF20h	Trap 6	Interrupts
7	FFFF FF00h	Trap 7	
8	FFFF FEE0h	NMI	Nonmaskable interrupt
9	FFFF FEC0h	HI	Host interrupt
10	FFFF FEA0h	DI	Display interrupt
11	FFFF FE80h		Window violation interrupt
12	FFFF FE60h	Trap 12	Bosoniad for future
13	FFFF FE40h	Trap 13	hardware or on-chip
14	FFFF FE20h	Trap 14	interrupts
15	FFFF FE00h	Trap 15	
16	FFFF FDE0h	Trap 16	
			Application defined
29	FFFF FC40h	Trap 29	
30	FFFF FC20h	ILLOP	Illegal opcode interrupt
31	FFFF FC00h	Trap 31	Application defined
32	FFFF FBE0h	SS	Single-step/Emulator
33	FFFF FBC0h	BF	Bus fault
34	FFFF FBA0h	Trap 34	
32767	FFF0 0000h	Trap 32767	Application defined
		32 bits	

### Figure 13–13. Vector Address Map

- Notes: 1) Traps (-1)— (-32,768) use the memory at the bottom of the address space for vector addresses. Traps 0—32,767 use the memory at the top of the address space.

  - 3) Traps (-1)--- (-32,768) and 32---32,767 are only accessed by TRAPL.
  - 4) Traps 3-7 and 12-15 are reserved for future interrupts.

#### Machine States

#### 10 if ST is aligned

else 12

#### Status Bits

- **C** 0 **Z** 0

N 0

V Ö

#### Examples

Assume that memory contains the following values before instruction execution:

Address	Data
FFFFFBC0	0000
FFFFFBD0	FF0E
FFFFFC00	0000
FFFFFC10	FFE0
FFFFFFE0	0000
FFFFFFF0	FFA0
0000000	0000
00000010	FF0F

<u>Code</u>		<u>Before</u>		After								
		PC	SP	PC	SP	ST						
TRAPL	-1	xxxxxxxh	80000000	FF0F0000	7FFFFFC0h	0000010h						
TRAPL	0	xxxxxxxh	80000000	FFA00000	7FFFFFC0h	0000010h						
TRAPL	31	xxxxxxxh	80000000	FFE00000	7FFFFFC0h	0000010h						
TRAPL	33	xxxxxxxh	80000000	FF0E0000	7FFFFFC0h	0000010h						

Syntax	VBLT B, L												
Execution	Binary pixel a	$rray \rightarrow lin$	ear pixel a	rray usin	g VRA	AM b	olock	c write	Э				
Instruction Words	15 14 13	12 11	10 9	87	6	5	4	3	2	1	0		
	0 0 0	0 1	0 0	0 0	1	0	1	0	1	1	1		
Description	VBLT moves corresponding expansion. Bo is an expansio data is writter color register) pixel unchang has been load subsection 12	a binary g block de oth SADDF on implicit to the de . Source to ed. Note to ded by the .5.4, <u>VRA</u>	array of p fined by D/ R and DAD in the trans estination a bits of the v this instruc e VLCOL i M Block M	ixels def ADDR an DR conta sfer, such array as a alue 0 lea tion assu instructio iode, pag	ined to a DYI a DYI a that a COL ave th ave th a Fo ge 12-	DX L ears a bit OR e cc hat r mo 14.	ADE Ising tartir Valu Valu I pix orres the \ ore in	DR ai VRA ng ad ue of kel (fr pond /RAN nform	nd D M bl dres 1 in t com t ing d 1 col nation	VDX lock i ses. he so he V lestir or re n, re	to a mode There ource (RAM nation gister fer to		
	Note: 1) DPTCH r 2) this instru	nust be ai uction wor	n integral n ks only if tl	nultiple o ne PSIZE	f 80h, E is 4,8	<b>ano</b> 8,16	d , and	d 32.			1		
Implied Operands	Register	Name	Form	at	Desci	riptio	on						
	B0	SADDR	Line	ar	Sou	rce p	ixel a	array	rray address				
	B1	SPTCH	Line	ar	Sou	rce p	ixel a	array	pitch		·		
	B2	DADDR	Line	ar	Dest	tinati	on pi	ixel ar	ray a	ddres	ss		
	B3	DPTCH	Line	ar	Dest	tinati	on pi	ixel ar	ray p	itch			
	77												
		DYDX	XY		Pixe	l arra	ay dir	nensi	ons	····			
	B14	DYDX TEMP	XY Tem	p	Pixe Inter	l arra med	ay dir iate v	mensi value	ons				
	B7 B14 Address	DYDX TEMP Name	XY Tem	p Descript	Pixe Inter ion an	l arra med	ay dir iate v eme	mensi value nts (E	ons Bits)				
,	B7 B14 Address C0000150h	DYDX TEMP Name PSIZE	XY Tem	p <b>Descript</b> Pixel si:	Pixe Inter ion an ze (4,8	l arra med nd El	ay dir iate v eme 32)	mensi value nts (E	ons Bits)				
	B7           B14           Address           C0000150h           C0000160h	DYDX TEMP Name PSIZE PMASK	XY Tem (32 bits)	p Descript Pixel si: Plane n	Pixe Inter ion an ze (4,8 nask –	l arra med d El	ay dir iate v eme 32) el for	mensi value <b>nts (E</b> rmat	ons Bits)				
<b>x</b>	B7 B14 Address C0000150h C0000160h C00001A0h	DYDX TEMP Name PSIZE PMASK CONFIG	XY Tem (32 bits)	p Descript Pixel si: Plane n Bit 8 (V	Pixe Inter ion an ze (4,8 nask – EN) er	l arra med d El 5,16,3 – pix nable	eme 32) es VF	mensi value nts (E rmat RAM v	ons Bits) vrite r	mask			
	B7 B14 Address C0000150h C0000160h C00001A0h Address	DYDX TEMP Name PSIZE PMASK CONFIG	XY Tem (32 bits)	p Descript Pixel si: Plane n Bit 8 (V	Pixe Inter ion an ze (4,8 nask – EN) er	l arra med d El ,16,3 – pix nable	ay dir iate v eme 32) el for es VF	mensi value nts (E rmat RAM v	ons Bits)	mask			

Pixel processing is not possible with this instruction, because the pixel data is written from the VRAM color register into the VRAM memory array.

#### VBLT B, L Linear VRAM Block Transfer

*Window Checking* Window checking **cannot** be used with this instruction.

*Transparency* Transparency bits are ignored. This instruction has a form of implicit transparency in that source pixels which are 0 correspond to destination pixels which are not changed.

*Plane Masking* The plane mask is implemented in the VRAM using the write mask function, enabled by VEN[[CONFIG]]. For more information, refer to Section 12.10, <u>Plane Masking</u>, on page 12-39.

*Interrupts* This instruction can be interrupted at a word or row boundary of the destination array. For more information, refer to Section 6.6, <u>Interrupting Graphics Instructions</u>, on page 6-13.

*Corner Adjust* Corner adjust **cannot** be used with this instruction.

Machine States complex instruction

Status Bits

N Undefined

- **C** Undefined
- Z Undefined
- V Undefined

#### Example

This is an example of a C-compatible assembly routine which draws a character on the screen using the VBLT instruction. It expects the following arguments on the C parameter stack: width, height, xleft, ytop, and a pointer to the start of the character data. The character data should be a binary representation of the character.

This routine makes the following assumptions:

These B registers and I/O registers have been initialized by the calling program:

B-file registers DPTCH, OFFSET, WSTART, WEND, COLOR0, COLOR1

I/O registers CONTROL, CONVDP, PSIZE, PMASK and CONFIG

- The system contains a global flag \_vblt\_ok which is cleared if the VBLT is not possible. Reasons for this may be:
  - DPTCH is not an integral multiple of 80 hex
  - PSIZE is 1 or 2
  - Pixel processing is not set to replace
  - Transparency is not set to source equals 0
  - The system does not contain VRAMs that support this feature

STK SADDR SPTCH DADDR DPTCH DYDX	.set .set .set .set .set .set .globl	A14 B0 B1 B2 B3 B7 _vblt ; provide	<pre>; C-parameter stack pointer ; Source address register ; Source pitch register ; Destination address register ; Dest. pitch register ; Delta X/delta Y register reference for external calls</pre>
vblt:	.ref	_vblt_ok	; flag to enable VBLTs
_	mmtm move move sll movy move move move sll movy move	<pre>SP,B0,B1,B2,B7,B10,1 STK,B10 *-B10,DYDX,1 DYDX,SPTCH *-B10,B12,1 16,B12 B12,DYDX *-B10,DADDR,1 *-B10,B12,1 *-B10,SADDR,1 B10,STK 16,B12 B12,DADDR @_vblt_ok,A8,1</pre>	<pre>311,B12 ;save required registers ;move c-stack pointer into B-file ;get width ;save the width as source pitch ;get height ;concatenate width &amp; height ;get xleft ;get ytop ;get source address ;restore c-stack pointer ;concatenate xleft &amp; ytop ;get state of vblt flag</pre>
vblt:	jrz	no_vblt	
	clip jrz cvdxyl vlcol vblt	exit DADDR B,L	<pre>;clip to the window ;if outside the window, exit ;convert to linear dest address ;load VRAM color latches ;perform linear vblt</pre>
no vhlt.	jruc	exit	
exit:	pixblt	в,ХҮ	
rets	mmfm 2	SP, B0, B1, B2, B7, B10, B	311,B12 ;restore required registers

Syntax	VFILL L													
Execution	Contents of V	RAM color	<sup>.</sup> latch → ar	ray of p	ixels									
Instruction Words	15 14 13	12 11	10 9	B 7	6	5	4	3	2	1	0			
	0 0 0	0 1	0 1	0 0	1	0	1	0	1	1	1			
Description	VFILL fills an array of pixels defined by DADDR and DYDX using the VRAM block mode writes. Note this instruction assumes that the VRAM color register has been loaded by the VLCOL instruction. For more information, refer to subsection 12.5.7, <u>VRAM Block Mode Fill</u> , on page 12-16.													
	Note:													
	1) DPTCH must be an integral multiple of 80h, and													
	2) this instru	iction work	s only for F	PSIZE's	4,8,1	6, a	nd 3	2.						
Implied Operands														
	Register	Name	Forma	t	Desc	ripti	on							
	B2	DADDR	Linea	r	Des	stinat	ion p	ixel b	ock a	k address				
	B3	DPTCH	Linea	r	Des	stinat	ion p	ixel b	ock p	oitch				
	B7	DYDX	XY		Pixe	el blo	ck dimensions							
	Address	Name		Descrip	tion	and	Elem	ents	(Bits	)				
	C0000150h	PSIZE		Pixel si	ze (4	,8,16	,32)							
	C0000160h	PMASK	(32 bits)	Plane r	nask	— pi	xel fo	ormat						
	C00001A0h	CONFIG	à	Bit 8 (V	ΈN) e	enabl	es V	RAM	write	mask	[			
	Address	Name	Description	า										
	VRAM Color Register	Pixel	Must have	COLOR	1 pixe	els lo	aded	using	J VLC	OL				
	VRAM Write Mask	Pixel	Loaded au VEN = 1	tomatica	lly wh	ien P	MAS	iK is v	vritter	ו and				
Pixel Processing	Pixel processi written from th	ng is not po e VRAM c	ossible with olor registe	this ins r into th	tructi le VF	on, t RAM	oeca men	use ti nory a	ne pi: array	xel da '.	ata is			
Window Checking	Window check	king <b>canno</b>	ot be used v	with this	instr	uctio	on.							
Transparency	Transparency	cannot be	e used with	this inst	tructi	on.								
Plane Masking	The plane may	sk is imple	mented in t	he VRA	M us	sina	the v	vrite	masł	c fund	ction.			

Plane Masking enabled by VEN[[CONFIG]]. For more information, refer to Section 12.10, Plane Masking, on page 12-39.

Interrupts This instruction can be interrupted at a word or row boundary of the destination array. For more information, refer to Section 6.6, Interrupting Graphics Instructions, on page 6-13.

Corner Adjust There is no corner adjust for this instruction.

Machine State	s	complex instruction	
Status Bits		<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>	
Example		This is an example of a on the screen. The rout ytop. Note that the CLIF	C-compatible assembly routine which fills a rectangle tine takes these 4 arguments: width, height, xleft, and instruction is used to clip the rectangle to the screen.
		This routine makes the	following assumptions:
		The calling program B-file registers DP I/O registers CC	n sets up these registers: TCH, OFFSET, WSTART, WEND and COLOR1 NTROL, CONVDP, PSIZE, PMASK and CONFIG
		The system contain is not possible. Real	ns a global flag _vfill_ok which is cleared if the VFILL asons for this may be:
		<ul> <li>DPTCH is not a</li> <li>PSIZE is 1 or 2</li> <li>Pixel processin</li> <li>Transparency is</li> <li>The system doe</li> </ul>	an integral multiple of 80 hex g is not set to replace s not set es not contain VRAMs that support this feature
DADDR DYDX CONTROL	.set .set .set .globl	B2 B7 0C00000B0h _fill_rect	;Destination address register ;Delta X/delta Y register ;Control IO register ;provide reference for external calls
_fill_rect:	mmtm move move move sll		; flag to enable vrints ;save required registers ;move c-stack pointer into B-file ;get width ;get height
	movy move move sll	B12,DYDX *-B10,DADDR,1 *-B10,B12,1 B10,A14 16,B12	;concatenate width & height ;get xleft ;get ytop ;restore c-stack pointer
	movy move jrz	B12,DADDR @_vfill_ok,A8,1 no_vfill	;concatenate xleft & ytop ;get state of vfill flag
	clip jrz cvdxyl vlcol vfill	exit DADDR L	;clip to the window ;if outside the window, exit ;convert to linear dest address ;load VRAM color latches ;perform linear fill
no_vfill: exit:	jruc fill mmfm	exit XY SP,B2,B7,B10,B11,B12	;fill the rectangle using standard fill ;restore required registers

rets

2

•

13-263

Syntax	VLCOL
Execution	COLOR1 $\rightarrow$ Color Registers in all VRAMS
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         0       0       0       0       1       0       1       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0       0
Description	VLCOL writes the value in the COLOR1 register to the color registers in all external VRAMS. The field size is ignored and the flood write outputs to nominal address 0. This instruction should be executed before attempting to use VFILL or VBLT. This instruction performs color expansion in the VRAM as pixels are written. The VRAM color registers are used for this purpose.
Implied Operands	Register         Name         Format         Description
Machine States	2 (1)
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z Unaffected</li> <li>V Unaffected</li> </ul>
Example	This is an example of a C-compatible assembly routine which fills a rectangle on the screen. The routine takes these 4 arguments: width, height, xleft, and ytop. Note that the CLIP instruction is used to clip the rectangle to the screen.
	This routine makes the following assumptions:
	These B-registers and I/O registers have been set up by the calling pro- gram:
	B-file registers DPTCH, OFFSET, WSTART, WEND and COLOR1
	I/O registers CONTROL, CONVDP, PSIZE, PMASK and CONFIG
	<ul> <li>The system contains a global flag _vfill_ok which is cleared if the VFILL is not possible. Reasons for this may be:</li> <li>DPTCH is not an integral multiple of 80 hex</li> <li>PSIZE is 1 or 2</li> <li>Pixel processing is not set to replace</li> <li>Transparency is not set</li> <li>The system does not contain VBAMs that support this feature</li> </ul>

DADDR	.set	B2		;Destination address register
DYDX	.set	В7		;Delta X/delta Y register
CONTROL	.set	0C00000B0h		;Control IO register
	.globl .ref	_fill_rect _vfill_ok	; provid	de reference for external calls ; flag to enable VFILLs
fill r	ect:			
	mmtm	SP, B2, B7, B10, B1	1,B12	;save required registers
	move	A14,B10	;move c-	-stack pointer into B-file
	move	*-B10,DYDX,1		;get width
	move	*-B10,B12,1		;get height
	sll	16,B12		
	movy	B12, DYDX		;concatenate width & height
	move	*		;get xleft
	move	*-B10,B12,1		;get ytop
	move	B10,A14		restore c-stack pointer
	sll	16,B12		_
	movy	B12, DADDR		;concatenate xleft & ytop
	move	<pre>@_vfill_ok,A8,1</pre>		;get state of vfill flag
	jrz	no_vfill		
	clip			;clip to the window
	jrz	exit		; if outside the window, exit
	cvdxyl	DADDR		;convert to linear dest address
	vlcol			;load VRAM color latches
	vfill	L		;perform linear fill
	jruc	exit		
no_vfil	1:			
	fill	XY	;fill th	ne rectangle using standard fill
exit:				
	mmfm	SP, B2, B7, B10, B1	1,B12	;restore required registers
	rets	2		

# XOR Exclusive-OR Registers

Syntax	XOR Rs, Ro	d			
Execution	Rs XOR Rd	→ Rd			
Instruction Words	15 14 13 0 1 0	12 11 10 1 0 1	9 8 7 6 1 Rs	6 5 4 R	3 2 1 0 Rd
Description	XOR bitwise- tents of the de You can use CLR instructi	exclusive-ORs estination registe this instruction on also support	the contents of th er, and stores the to clear registers s this function.	ne source re result in the a (for examp	əgister with the con- destination register. ble, хок во,во); the
	Rs and Rd m	ust be in the sa	me register file.		
Machine States	1				
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z 1 if the red</li> <li>V Unaffected</li> </ul>	ed ed esult is 0, <i>0</i> othe ed	rwise		
Examples	Code XOR A0,A1 XOR A0,A1 XOR A0,A1	<u>Before</u> A0 FFFFFFFFh FFFFFFFh FFFFFFFh	<b>A1</b> 00000000h AAAAAAAAh FFFFFFFh	After NCZV XX0X XX0X XX0X XX1X	<b>41</b> FFFFFFh 55555555h 00000000h

Syntax	XORI IL, Rd					
Execution	$IL XOR Rd \to Rd$					
Instruction Words	15 14 13 12 11 0 0 · 0 0 1	10 9 8 0 1 1 16 LSBs 16 MSBs	7 6 5 1 1 0 of IL of IL	4   R	<u>32</u> Ro	1 0 j
Description	XORI bit-wise exclusive destination register and parameter in the syntax	-ORs a 32-bit im stores the resu above represen	mediate da It in the dea ts a 32-bit i	ta with t stination mmedia	he conter n register ate value	nts of the . (The <i>IL</i> .)
Machine States	2 if the immediate data 3 if the immediate data	is long-word alig is not long-word	ned aligned			
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z 1 if the result is 0, 0</li> <li>V Unaffected</li> </ul>	otherwise				
Examples	Code XORI 0FFFFFFFh,A0 XORI 0FFFFFFFh,A0 XORI 0FFFFFFFh,A0 XORI 0000000h,A0 XORI 0000000h,A0	Before AO 00000000h AAAAAAAAh FFFFFFFh 00000000h FFFFFFFh	After NCZV ××0x ××1x ××1x ××1x ××0x	A0 FFFFFf 555555 000000 000000 FFFFFf	=FFh 55h 00h 00h =FFh	·

maaaam

# ZEXT Zero-Extend to Long

Syntax	ZEXT Rd [, F	7]						
Execution	field in Rd $\rightarrow$	zero-e	extended	l field Rd				
Instruction Words	15 14 13 0 0 0	12 1 0	1 10 0 1	9 8 7 F 1 0	6	5 4 1 R	3 2	1 0 Rd
Description	ZEXT zero-ex all the nonfield size. The option the field size:	tends a l bits in onal <i>F</i> p	right-jus Rd. The paramete	stified field in size of the fie er, which mus	the des Id is def t be spo	stination terminec ecified a	register d by the c s a 0 or a	by zeroing urrent field 1, selects
	F=0 selects F F=1 selects F	S0 for t S1 for t	he field he field	size. size.				
	The default va	lue for	<i>F</i> is 0.					
Machine States	1							
Status Bits	<ul> <li>N Unaffected</li> <li>C Unaffected</li> <li>Z 1 if the res</li> <li>V Unaffected</li> </ul>	d d sult is 0 d	, 0 other	rwise				
Examples	Code ZEXT A0,0 ZEXT A0,0 ZEXT A0,0 ZEXT A0,0 ZEXT A0,1	<b>Before</b> <b>FS0</b> 32 31 1 16 x	<b>FS1</b> × × × × × 16	After A0 FFFFFFFFh FFFFFFFh FFFFFFFh FFFF0000h FFFF0000h	N ( × × × × × × × × × ×	0 x 0 x 0 x 1 x 1 x	A0 FFFFFFF 7FFFFFF 00000001 000000000 000000000	Fh Fh h )h

# Chapter 14

# TMS34082 Pseudo-ops

Many TMS34020 applications require floating-point operations. The TMS34082 Floating-Point Processor is designed specifically to serve as a coprocessor in a TMS34020 system. To extend the TMS34020's direct interface to the TMS34082, the TMS34020 supports a subset of the TMS34082 assembly-language instruction set by supplying a group of **TMS34082 pseudo-ops**. These pseudo-ops are special versions of the TMS34020's general-purpose coprocessor instructions. Instead of designing a protocol for sending instructions and data back and forth between the TMS34020 and TMS34082, you can use these pseudo-ops, which are hard-coded versions of instructions such as the CMOVCG instruction.

This chapter provides a general description of the pseudo-op protocol and provides an alphabetical reference to the TMS34082 pseudo-ops.

	Section	Page
<b>Basic information</b> includes a review of related TMS34020 signals and an overview of the coprocessor interface.	<ul> <li>14.1 Overview and Key Features of the TMS34082</li> <li>14.2 Pseudo-op Format</li> <li>14.3 Register Operands</li> </ul>	14-2 14-3 14-6
Alphabetical reference of pseudo-ops	begins on page	

# 14.1 Overview and Key Features of the TMS34082

The TMS34082 is a high-speed floating-point processor, implemented in TI's advanced 1-micron CMOS technology. On a single chip, the TMS34082 combines a 16-bit sequencer, a 3-operand FPU, and 22 64-bit data registers. An instruction register controls FPU execution, and a status register retains the most recent FPU status outputs. The TMS34082 also contains 8 control registers and a 2-deep stack.

The TMS34082 is fully compatible with IEEE Standard 754–1985 for binary floating-point arithmetic. Floating-point operands can be in either single- or double-precision IEEE format.

Key features and benefits include

- Closely coupled with the TMS34020
  - Direct TMS34020 instruction extension
  - Multiple-TMS34082 capability
- Internal programs for vector, matrix, and graphics operation
- **G** Fast multiply/accumulate cycle time
  - 40 MHz (TMS34082-40) ... 50 ns
  - 32 MHz (TMS34082-32) ... 60 ns
- External memory addressing capability
  - External program storage (up to 64K words)
  - External data storage (up to 64K words)
- □ Full IEEE standard 754–1985 compatibility
  - Addition

Subtraction

Multiplication

Division

- Square root
- Comparison
- Selectable data formats
  - 32-bit integer
  - 32-bit, single-precision floating-point value
  - 64-bit, double-precision floating-point value
- Supported by TMS34020 code-generation tools
- More than 30 complex instructions targeted at graphics math
- Use as a floating-point coprocessor eliminates the need for external logic interface
- Standardized approach to floating-point for full system compatibility
- Eliminates multiple-cycle software implementation
- □ Superior performance for 2-D and 3-D graphics applications

# 14.2 Pseudo-op Format

Section 10.2, <u>Overview of the Coprocessor Interface</u> (page 10-3), lists the TMS34020's general-purpose coprocessor instructions. Section 10.3, <u>Format of Commands Passed to a Coprocessor</u> (page 10-5), describes the general format of these instructions; specific implementations of this format depend entirely on the coprocessor that you choose to include in your system.

The TMS34020 provides a set of TMS34082 pseudo-ops. These pseudo-ops extend the general-purpose format by hard-coding TMS34082 opcodes into certain fields of the general-purpose coprocessor instructions.

Figure 10–1 (which appears in Section10.3, page 10-5) is repeated below as Figure 14–1 (*a*). This figure shows the general format of coprocessor information that is placed on the LAD bus. Figure 10–1 (*b*) shows the TMS34082 implementation of this format.

Figure 14–1. Coprocessor Instruction Information on the LAD bus

(a) General Format

LAD	31	30	29	28	27	26	25	24	23	22	21	20	19	18	17	16	15	14	13	31	12	11	10	9	8	7	6	5	4	3	2	1	0
P111#		ID									со	pro	ces	sor	con	nma	nd									s i z e		I	S		вс	ST	
Key:		ID			3-b	oit c	p	roc	ess	or	D f	ield				s	ize	F	Para	am	net	er	size	e fie	eld								
		Ι			Сс	pro	ce	sso	r pa	ara	me	ter i	nd	əx t	oit	S	;	1	6-t	oit '	w	ord	se	ect	(al	wa	iys d	outp	but	as (	D)		
		B	CSI	Г	4-t	oit k	bus	sta	atus	s (al	wa	ys (	) fc	ric	opr	oce	sso	or c	;yc	les	S)												
		<u>C</u>	opr	oce	ess	or	COI	nm	an	<b>d</b> 2	1-b	it in	str	ucti	on	for	the	СС	pro	Ce	es	sor											
			al.		Re	sei	ve	d (a	alwa	ays	ou	lpul	as	0)																			
Note	e: S	Sect	ion	10	.3 (	(pa	ge	10-	5) (	des	crit	bes	the	se	bit	fie	ds	in	det	ail	Ι.												

#### (b) TMS34082-Specific Format

em.		*****	*****	******	*****	******	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		*****			******	******	******	~~~~~	******	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	******	00000000	******		200000000		,,,,,,,,,,,,,,,,,,,,,,,,,
۱ ا	LAD #																							
13	31—29	28	2	25	24		21	20		16	15	14	13		9	8	7	6	5	4				0
~~~~~	ID		CRs1		(	CRs2			CRd		m	d		fpuop		Т	size	0	Ĩ	0	0	0	0	0
1.																								

- ❑ As Figure 10–1 (b) shows, the bus cycle status code portion is 0000<sub>2</sub>. This indicates that the local-memory cycle generated by this type of instruction is a coprocessor cycle.
- Bit 4, the S (16-bit word select) bit, is also 0; this indicates that only 32-bit accesses will occur.
- The I bit serves the same purpose for the TMS34082 as it does for other processors (refer to subsection 10.3.4, Coprocessor Parameter Index, on page 10-7).

Bit 7 still serves as the size (parameter size) bit. The TMS34082 uses the LSB of the coprocessor command as T (bit 8) that works with the size bit to identify the type and size of the parameter(s) that are passed to the TMS34082:

Т	size	Operand Type
0	0	32-bit integer
0	1	reserved
1	0	single-precision (32-bit) floating-point number
1	1	double-precision (64-bit) floating-point number

- The coprocessor command is divided into 5 fields:
  - The 5-bit fpuop field contains the opcode of a TMS34082 assemblylanguage instruction.
  - The 2-bit md field conveys the coprocessor command's addressing mode:

Mode	Operation
00 <sub>2</sub>	FPU internal operation, no jumps or external moves
012	Transfer to/from TMS34020 registers
102	Transfer to/from TMS34020 local memory
112	External microcode

- CRd is the TMS34082 destination register.
- CRs<sub>2</sub> is the second TMS34082 source register for instructions that use two source operands. CRs<sub>2</sub> also serves as the count operand for instructions that use a count operand. CRs<sub>2</sub> must be a member of the TMS34082 B register file.
- CRs<sub>1</sub> is the TMS34082 source register for instructions that have one source operand; it is the first source register for instructions that use two source operands. CRs<sub>1</sub> must be a member of the TMS34082 A register file.
- The ID field serves the same purpose in the TMS34082 protocol as it serves in the general-purpose protocol. For more details, refer to subsection 10.3.1, Coprocessor ID, on page 10-5. The pseudo-ops default to an ID of 000<sub>2</sub>; to define another ID as the current ID, use the .coproc assembler directive.

For the TMS34082, these bits are hard-coded into special versions of the TMS34020's general-purpose coprocessor instructions. As an example, Figure 14–2 compares the general syntax of the CMOVGC instruction (a general-purpose coprocessor instruction) to the LOAD-and-ADD (ADD) pseudo-op.

Figure 14–2. How General Coprocessor Instruction Syntax Corresponds to TMS34082 Pseudo-ops (a) General syntax for CMOVGC instruction

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
0	0	0	0	0	1	1	0	0	1	0	R		R	's1	
	8 LS	Bs of	copro	cesso	r comr	nand		size	0	0	R		R	's2	
copr	ocess	or ID		-		13	MSB	s of co	oroce	ssor c	omma	ind			

(b) General syntax for TMS34082 ADD pseudo-op

$\frac{14}{10}$					9 1					B	<u> </u>			
$\frac{1}{1}$	0	0	0	0	0	0	0	0	0	R	Rs2			
 ID	CRs1					CF	7 <i>s2</i>	L	CRd					

- **Note:** This portion of the figure shows the ADD pseudo-op. ADD is a special form of the TMS34020 CMOVGC instruction. The coprocessor command and size portions of the CMOVGC instruction are hardcoded to specifically match the needs of the TMS34082's ADD instruction.
- (c) What is output on the LAD bus for the ADD pseudo-op

LAD #		******				*******		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,			******		000000000		
31—29	28 25	24 21	20 16	15 14	13	9	8	7	6	5	4				0
ID	CRs1	CRs2	CRd	01	00000		0	0	0	0	0	0	0	0	0
		-													

Note:

To identify the 3-bit ID, use the .coproc assembler directive (refer to the *TMS340 Family Code Generation Tools User's Guide* for information about the .coproc directive).

# 14.3 Register Operands

The TMS34082 pseudo-ops use register operands only. Table 14–1 lists the register-operand symbols used in the psuedo-op syntaxes in this chapter.

Table 14–1. Symbols Used in Pseudo-op Syntax Listings

Symbol	Description	Symbol	Description
CRs	TMS34082 source register	CRd	TMS34082 destination register
CRs <sub>1</sub>	For pseudo-ops that use 2 TMS34082 source registers, this register supplies the first operand	CRs <sub>2</sub>	For pseudo-ops that use 2 TMS34082 source registers, this register supplies the second operand
	This operand must be a TMS34082 A-file register		This operand must be a TMS34082 B-file register
Rs	TMS34020 source register	Rd	TMS34020 destination register

Note that some pseudo-ops use information from a TMS34020 register or place information into a TMS34020 register. In this case, *Rs* or *Rd* should be a TMS34020 general-purpose register (A0—A14 or B0—B14), just as it would be for a TMS34020 instruction. Most of the pseudo-ops, however, use TMS34082 registers as operands. As Figure 14–3 shows, the TMS34082 contains 2 register banks of 10 64-bit registers, plus 2 feedback registers.

Figure 14–3. TMS34082 Registers That Can Be Used as Pseudo-op Operands



Note: These register files contain TMS34082 registers.

Most pseudo-ops operate on one value from TMS34082 register file A or B, and return the result to file A, file B, or one of the feedback registers. Valid operand/register use includes: 
 CRs or CRs1: RA0—RA9
 CRs2: RB0—RB9

 CRd: RA0—RA9, RB0—RB9, C, and CT

When more than one value is requested from/sent to the TMS34082, the registers are read from/written to in the sequence shown in Figure 14–4. Note that the control, status, and stack registers are in the middle of the list. The sequence bypasses C and CT because they can't be accessed externally.

Figure 14–4. TMS34082 Register Sequence List



Syntax	ABORT														
Execution	Halts coprocessor														
Instruction Words	15 14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	1 1	0	1	1	0	0	0	0	0	1	1	1	1	0	0
	Default ID	)	0	0	0	0	1	0	1	1	0	0	0	0	0
Description	ABORT halts coprocessor in nate.				tion <i>-nex</i> i	of th t <i>-inst</i>	e TN <i>ructio</i>	/IS34 onsta	082 ate. F	copro legis	ocess ter va	sor a lues	nd p are i	laces ndete	; the ermi-
Machine States	2														
Instruction Type	CEXEC, short														
Example	ABORT														
	This example halts the TMS34082 coprocessor.														

Syntax	ABS CRs, CRd															
Execution	$ CRs  \rightarrow CRd$															
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         1       1       0       1       1       0       0       0       0       1       1       1       1       0       0         Default ID       CRs       0       0       1       0       0       CRd       CRd															
Operands	CRsCoprocessor source register containing the 32-bit integer operandCRdCoprocessor destination register															
Description	ABS takes the absolute value of the contents (integer) of CRs and stores the result in CRd.															
	The coprocessor source register, CRs, must be in the A coprocessor file.															
Machine States	2															
Instruction Type	CEXEC, short															
Example	ABS RA6, RB7															
	This example takes the absolute value of the contents of RA6 and stores th result in RB7.															
Syntax	ABS Rs, CRs, CRd															
-------------------	--	-------------------------	------------------------	---------------	----------------	--------------	-----------------	-----------------	-----------------	-----------------	----------------	------------------	----------------	----------------	----------------	----------------
Execution	$Rs \rightarrow CRs$ $ CRs  \rightarrow CRd$															
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3_	2	1	0
	0 0 0 0 0 1 1 0 0 1 R Rs															
	0	1	0	1	1	1	1	0	0	0	0	0	0	0	0	0
	De	efault	ID		CI	Rs		0	0	1	0			CRd		
Operands	Rs	Т	MS3	4020	) sou	rce re	egiste	er for	the 3	32-bit	integ	ger va	alue 1	o co	oroce	essor
	CRs Coprocessor register to contain the 32-bit integer operand															
	CRd Coprocessor destination register															
Description	ABS loads the contents (integer) of Rs into CRs, takes the absolute value o the contents of CRs, and stores the result in CRd.															ue of
	The	copro	ocess	sor s	ource	e regi	ister,	CRs	s, mu	st be	in th	e A c	copro	cess	or fil	ə.
Machine States	3 if th 2 if th	ne fir ne fir	st ins st ins	struct	ion v ion v	vord vord	is lor is no	ng-wa t long	ord a g-wo	ligne rd ali	d gned					
Instruction Type	СМС	VGC	C, on	e reg	ister											
Example	ABS	A5,	RA6	, RE	37											
	This regis resul	exar ter F t in F	nple IA6, 1 RB7.	load: akes	s the the	cont abso	ents lute v	of TI value	MS34 e of th	4020 1e co	regis ntent	ster A s of F	15 int RA6,	o coj and :	oroce store	essor s the

Syntax	ABSD CRs, CRd															
Execution	$ CRs  \rightarrow CRd$															
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1         1       1       0       1       1       0       0       0       0       1														0	
Operands	<ul> <li>CRs Coprocessor source register containing a 64-bit double-precision float ting-point operand</li> <li>CRd Coprocessor destination register</li> </ul>															floa-
Description	ABSD takes the absolute value of the contents of CRs and stores the result CRd.															ult in
	The	copro	ocess	sor so	ource	e regi	ster,	CRs	, mu	st be	in th	e A c	copro	cess	sor fil	e.
Machine States	2															
Instruction Type	CEX	EC, s	short													
Example	ABSD	RAG	5, R	в7												
	This resul	exar t in F	nple 1 RB7.	takes	the	abso	lute	value	e of tl	ne co	nten	ts of	RA6	and	store	s the

## ABSF Absolute Value, Single Precision

Syntax	ABSF CRs, CRd														
Execution	$ CRs  \rightarrow CRd$														
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         1       1       0       1       1       0       0       0       0       1       1       1       1       1       0         Default ID       CRs       0       0       1       0       CRd														
Operands	CRs Coprocessor source register containing a 32-bit single-precision floa- ting-point operand														
	CRd Coprocessor destination register														
Description	ABSF takes the absolute value of the contents (single-precision value) of CR and stores the result in CRd.														
	The coprocessor source register, CRs, must be in the A coprocessor file.														
Machine States	2														
Instruction Type	CEXEC, short														
Example	ABSF RA6, RB7														
	This example takes the absolute value of the contents of RA6 and stores the result in RB7.														

Syntax	ABSF Rs, CRs, CRd															
Execution	$Rs \rightarrow CRs$ $ CRs  \rightarrow CRd$															
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	0	0	1	R		R	S	
	0	1	0	1	1	1	1	1	0	0	0	0	0	0	0	0
	De	fault l	D		CF	Rs		0	0	1	0			CRd		
Operands	Rs TMS34020 source register for the 32-bit single-precision floating-point value to coprocessor															ision
	CRs Coprocessor register containing a 32-bit single-precision floating point operand															ting-
	CRd	С	opro	cess	or de	estina	tion	regis	ter							
Description	ABSI abso	= loa lute v	ds th /alue	e co of th	nteni ne co	ts (si nten	ngle- ts of	prec CRs	ision , and	valu store	e) of es th	<sup>:</sup> Rs i e res	nto C ult in	CRs, 1 CRd	takes	s the
	The c	copro	cess	sor so	ource	e regi	ster,	CRs	, mu	st be	in th	e A c	opro	cess	or file	Э.
Machine States	3 if th 2 if th	ne firs ne firs	st ins st ins	truct truct	ion w ion w	vord i vord i	is Ion is not	ig wo t long	ord-al g wor	ligneo d-alio	d gned					
Instruction Type	СМО	VGC	;, on	e reg	ister											
Example	ABSF	A5,	RA	6, R	в7											
	This takes	exar the	nple abso	load lute	s TM value	IS340 e of th	020 r 1e co	regis ntent	ter A ts of I	5 inte RA6,	o cop and	proce store	ssor s the	regis resu	ster I It in I	7A6, 7B7.

# ADD Add, Integer

Syntax	<b>ADD</b> $CRs_1$ , $CRs_2$ , $CRd$														
Execution	$CRs_1 + CRs_2 \rightarrow CRd$														
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         1       1       0       1       1       0														
Operands	CRs <sub>1</sub> Coprocessor register containing the first 32-bit integer operand CRs <sub>2</sub> Coprocessor register containing the second 32-bit integer operand CRd Coprocessor destination register														
Description	ADD adds the contents (integer) of $\mbox{CRs}_1$ and $\mbox{CRs}_2$ and stores the result in CRd.														
Machine States	2														
Instruction Type	CEXEC, short														
Example	ADD RA5, RB6, RB7														
	This example adds the contents of RA5 and RB6 and stores the result in RB7.														

Syntax	<b>ADD</b> $Rs_1$ , $Rs_2$ , $CRs_1$ , $CRs_2$ , $CRd$														
Execution	$\begin{array}{l} \operatorname{Rs}_1 \to \operatorname{CRs}_1 \\ \operatorname{Rs}_2 \to \operatorname{CRs}_2 \\ \operatorname{CRs}_1 + \operatorname{CRs}_2 \to \operatorname{CRd} \end{array}$														
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0														
	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$														
	Default ID         CRs1         CRs2         CRd														
Operands	Rs <sub>1</sub> TMS34020 source register for the first 32-bit integer value to coprocessor														
	Rs <sub>2</sub> TMS34020 source register for the second 32-bit integer value to coprocessor														
	CRs <sub>1</sub> Coprocessor register to contain the first 32-bit integer operand														
	CRs <sub>2</sub> Coprocessor register to contain the second 32-bit integer operand														
	CRd Coprocessor destination register														
Description	ADD loads the contents (integer) of $Rs_1$ and $Rs_2$ into $CRs_1$ and $CRs_2$ respectively, adds the contents of $CRs_1$ and $CRs_2$ , and stores the result in CRd.														
Machine States	4 if the first instruction word is long word-aligned 3 if the first instruction word is not long word-aligned														
Instruction Type	CMOVGC, two registers														
Example	ADD A5, A6, RA5, RB6, RB7														
	This example loads TMS34020 registers A5 and A6 into coprocessor registers RA5 and RB6 respectively, adds the contents of RA5 and RB6, and stores the result in RB7.														

Syntax	ADD	DC	CRs <sub>1</sub>	, CR	s <sub>2</sub> , C	Rd										
Execution	$CRs_1 + CRs_2 \rightarrow CRd$															
Instruction Words	15	14	13	12	11	10	9	8	_7	6	5	4	3	2	1	0
	1	1	0	1	1	0	0	0	0	0	0	0	0	0	1	1
	De	fault	D		CF	ls₁			CF	Rs <sub>2</sub>				CRd		
Operands	CRs <sub>1</sub>	C fle	opro patin	cess g-po	or re int op	egiste beran	er co d	ontair	ning	the	first	64-b	it do	uble	prec	ision
	CRs <sub>2</sub> Coprocessor register containing the second 64-bit double-precision floating-point operand															ision
	CRd	С	opro	cess	or de	estina	tion	regis	ter							
Description	ADD store	D ad s the	lds tl e resi	ne co ult in	onter CRd	its (d	oubl	e-pre	ecisio	n va	lue)	of Cl	Rs <sub>1</sub> a	and (	CRs <sub>2</sub>	and
Machine States	2															
Instruction Type	CEX	EC, s	short													
Example	ADDD	RA5	, R	в6,	RA7											
	This	exan	nple a	adds	the c	onte	nts o	f RA	5 and	I RB6	6 and	store	es the	e resi	ult in	RA7.

Syntax	ADDF	CRs₁.	CRs2.	. CRd

 $\textit{Execution} \qquad \qquad CRs_1 + CRs_2 \rightarrow CRd$ 

Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	1	1	0	1	1	0	0	0	0	0	0	0	0	0	1	0
	Det	fault I	D		CF	Rs <sub>1</sub>			CF	Rs <sub>2</sub>				CRd		
Operands	CRs <sub>1</sub>	Co flo	opro batin	cess g-poi	or r int of	egist perar	er co nd	ontai	ning	the	first	32-l	oit si	ingle	-prec	ision
	CRs <sub>2</sub> Coprocessor register containing the second 32-bit single-precisi floating-point operand CRd Coprocessor destination register															ision
	CRd Coprocessor destination register															
Description	ADDF adds the contents (single-precision value) of CRs <sub>1</sub> and CRs <sub>2</sub> and stores the result in CRd.														tores	
Machine States	2															
Instruction Type	CEXE	EC, s	hort													
Example	ADDF RA5, RB6, RB7															
	This e	exam	ple a	adds	the o	conte	nts o	f RA	5 anc	I RB6	and	store	es the	e resi	ult in	RB7.

Syntax	ADDF	F	Rs <sub>1</sub> , F	7s <sub>2</sub> , (	CRs <sub>1</sub>	, CR	s <sub>2</sub> , C	Rd								
Execution	$Rs_{1} \rightarrow CRs_{1}$ $Rs_{2} \rightarrow CRs_{2}$ $CRs_{1} + CRs_{2} \rightarrow CRd$ $15  14  13  12  11  10  9  8  7  6  5  4  3  2  1  0$															
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	0	1	0	R		R	<sup>3</sup> 1	
	0	1	0	0	0	0	0	1	0	0	0	R		<u>, R</u>	S2	
	Def	ault I	D		CF	IS <sub>1</sub>			CR	S <sub>2</sub>				CRd		
Operands	<ul> <li>Rs<sub>1</sub> TMS34020 source register for the first 32-bit single-precisio floating-point value to coprocessor</li> <li>Rs<sub>2</sub> TMS34020 source register for the second 32-bit single-precision</li> </ul>															sion
	Rs <sub>2</sub> TMS34020 source register for the second 32-bit single-precision floating-point value to coprocessor															sion
	CRs <sub>1</sub> Coprocessor register to contain the first 32-bit single-precision floating-point operand															sion
	CRs <sub>2</sub>	C flo	opro	cesso g-poi	or re nt op	giste eran	r to o d	conta	in th	e se	cond	32-b	oit si	ngle-	preci	sion
	CRd	С	opro	cesso	or de	stina	tion r	egist	er							
Description	ADDF and C	<sup>:</sup> loa Rs <sub>2</sub>	ds th resp	ie co bectiv	ntent vely, a	s (sir adds	ngle- CRs	preci: 1 anc	sion I CRs	value S <sub>2</sub> , ai	e) of l nd ste	Rs <sub>1</sub> a ores <sup>-</sup>	and i the r	Rs <sub>2</sub> i result	nto C in C	Rs <sub>1</sub> Rd.
Machine States	4 if the 3 if the	e firs e firs	st ins st ins	tructi tructi	ion w ion w	ord i	s lon s not	g woi long	rd-ali word	gned I-alig	ned					
Instruction Type	CMO	VGC	, two	o regi	sters	;										
Example	ADDF	A5,	A6	, RA	5, R	в6,	RA7									
	This e RA5 a	exam and F	iple li RB6 i	oads respe	TMS	3402 ly, ac	20 reg dds th	gister ne co	s A5 a ntent	and A s of F	A6 int RA5 a	o cop and R	oroce B6,	essor and s	regis stores	sters s the

result in RA7.

Syntax	CHE	CK F	Rd													
Execution	If cop F If cop 0	FFF FFF oroce	ssor FFF ssor 0000	is bu Fh — is idi h →	ısy → Rd le Rd											
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0 0 0 0 0 1 1 0 0 1 1 R Rs <sub>1</sub>															
	0	1	0	1	1	1	1	0	0	0	0	0	0	0	0	0
	Default ID         0         0         0         1         1         0         1         0														0	
Operands	Rd Destination register for status information															
Description	CHE is bus CHE	CK c sy, Cl CK s	heck HECl ets a	s the K set II the	s stati s all t bits	us of he bit in Ro	the c ts in F d to C	copro Rd to ).	ocess 1.lft	or. If he TN	the T NS34	FMS3 1082 (	3408 copro	2 cop oces:	oroce sor is	ssor idle,
Machine States	5 if th 4 if th	ne firs ne firs	st ins st ins	truct truct	ion w ion w	vord i vord i	s lon s not	g wo i long	ord-al 9 wor	ignec d-aliç	d gned					
Instruction Type	СМО	VGC	c, one	e reg	ister											
Example	CHEC	K A4														
	lf the A4 to in reg	TMS 1. If gister	63408 the <sup>-</sup> 7 A4 1	82 cc TMS to 0.	oproc 3408	esso 2 cop	r is b proce	usy, 1 essor	this e is id	exam le, th	ple se is ex	əts al ampl	ll the e res	bits i sets a	n reg Ill the	ister bits

Syntax	СМР	Cŀ	7s <sub>1</sub> ,	CRs <sub>2</sub>	?											
Execution	Flags (CRs <sub>1</sub> – CRs <sub>2</sub> ) $\rightarrow$ Coprocessor Status Registers															
Instruction Words	_15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	1	1	0	1	1	0	0	0	0	0	0	0	1	0	0	0
	De	efault l	D		CF	ls <sub>1</sub>			CF	ls <sub>2</sub>		0	0	0	0	0
Operands	CRs <sub>1</sub> Coprocessor register containing the first 32-bit integer operand															
	CRs <sub>2</sub> Coprocessor register containing the second 32-bit integer operand															Ind
Description	CMP subtracts the contents (integer) of CRs <sub>2</sub> from CRs <sub>1</sub> and sets the appropriate status bits in the coprocessor status register.															pro-
Machine States	2															
Instruction Type	CEX	EC, s	short													
Example	CMP	RA5,	RB	6												
	This in the	exan e cop	nple s roce	subtra ssor	acts t statu	the co Is reg	onter jister	nts of	f RA5	from	RB6	6 and	sets	the s	status	bits

Syntax	CMP	Rs <sub>1</sub> , R	s <sub>2</sub> , C	Rs <sub>1</sub> ,	CRs	2										
Execution	$Rs_1 \rightarrow Rs_2 \rightarrow Rs_2 \rightarrow Rs_2$	CRs <sub>1</sub> CRs <sub>2</sub> CRs <sub>1</sub> -	CRs	S₂) →	• Cop	oroce	ssor	Statu	us Re	egiste	ers					
Instruction Words	15 14	13	12	11	10	9	8	7	6	5	4	3	2	1	0	
	0 0	0	0	0	1	1	0	0	1	0	R		Rs	<sup>3</sup> 1		
	0 1	0	0	0	1	0	0	0	0	0	R		Rs	<sup>3</sup> 2		
	Defau	Rs <sub>1</sub> TMS34020 source register for the first 32-bit integer value to cop														
Operands	<ul> <li>Rs<sub>1</sub> TMS34020 source register for the first 32-bit integer value to coprocessor</li> <li>Rs<sub>2</sub> TMS34020 source register for the second 32-bit integer value coprocessor</li> </ul>															
	<ul> <li>Rs<sub>1</sub> TMS34020 source register for the first 32-bit integer value to cop cessor</li> <li>Rs<sub>2</sub> TMS34020 source register for the second 32-bit integer value coprocessor</li> </ul>															
	CRs <sub>1</sub>	Copro	cess	or re	giste	r to c	onta	in the	e first	: 32-b	oit int	eger	oper	and		
	CRs <sub>2</sub>	Copro	cess	or re	giste	r to c	onta	in the	e sec	ond (	32-bi	t inte	ger c	pera	nd	
Description	CMP los tively, s coproce	ads the ubtrac essor s	e cont ts CF tatus	tents Rs <sub>2</sub> fr regi	(inte om C ster.	ger) CRs <sub>1</sub>	of Rs , and	s <sub>1</sub> and d sets	d Rs <sub>2</sub> the	into appro	CRs- opria	<sub>1</sub> and te sta	CRs atus I	<sub>2</sub> res oits ir	pec- 1 the	
Machine States	4 if the 3 if the	first in: first in:	struct struct	tion v	vord vord	is lor is no	ig wo t long	ord-al g wor	igneo d-alio	d gned						
Instruction Type	CMOV	GC, tw	o reg	isters	3											
Example	CMP A5	, A6,	RA5,	, RB	6											
	This exa RA5 an in the c	ample d RB6, oproce	loads subt	TMS racts statu	63402 the c	20 re onte gister	giste nts o	ers A5 of RB6	and from	A6 in 1 RA5	to co 5, and	proce d sets	essoi the s	regis status	sters s bits	

### CMPD Compare, Double Precision

Syntax	CMPD	CRs <sub>1</sub>	, CRs <sub>2</sub>												
Execution	Flags (C	Rs <sub>1</sub> –	CRs <sub>2</sub> ) ·	→ Cop	proce	ssor	Statu	us Re	egiste	ers					
Instruction Words	15 14	13	12 11	10	9	8	7	6	5	4	3	2	1	0	
	1 1	0	1 1	0	0	0	0	0	0	0	1	0	1	1	
	Default ID     CRs1     CRs2     0     0     0       CRs1     Coprocessor register containing the first 64-bit double-preci														
Operands	Default ID       CRs1       CRs2       0														
4 * -	CRs <sub>2</sub>	Copro	cessor i g-point o	egiste operar	r cor nd	ntaini	ng th	ie se	cond	64-t	oit do	uble	-prec	ision	
Description	CMPD si sets the	ubtrac appro	ts the co priate st	ntents atus b	(dou oits in	ible-p the	orecia copro	sion v ocess	value sor st	) of C atus	Rs <sub>2</sub> 1 regis	from ster.	CRs	and	
Machine States	2														
Instruction Type	CEXEC,	short													
Example	CMPD RA	5, RB	6												
	This exa in the co	mple s proces	subtract	s the c tus reo	ontei gister	nts of	f RB6	6 fron	n RAS	5 and	sets	the	status	s bits	

Syntax	CMPF CRs <sub>1</sub> , CRs <sub>2</sub>														
Execution	Flags (CRs <sub>1</sub> – CRs <sub>2</sub> ) $\rightarrow$ Coprocessor Status Register														
Instruction Words	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$														
Operands	CRs <sub>1</sub> Coprocessor register containing the first 32-bit single-precisi floating-point operand CRs <sub>2</sub> Coprocessor register containing the second 32-bit single-precisi floating-point operand														
	CRs <sub>2</sub> Coprocessor register containing the second 32-bit single-precision operand floating-point operand														
Description	CMPF subtracts the contents (single-precision value) of $CRs_2$ from $CRs_1$ and sets the appropriate status bits in the coprocessor status register.														
Machine States	2														
Instruction Type	CEXEC, short														
Example	CMPF RA5, RB6														
	This example subtracts the contents of RB6 from RA5 and sets the status bits in the coprocessor status register.														

Syntax	CMPI	= F	Rs <sub>1</sub> , I	7s <sub>2</sub> ,	CRs	, CF	ls <sub>2</sub>									
Execution	Rs <sub>1</sub> - Rs <sub>2</sub> - Flags	→ CF → CF (CF	Rs <sub>1</sub> Rs <sub>2</sub> Rs <sub>1</sub> —	CRs	;2) →	• Сор	oroce	ssor	Statu	us Re	egiste	er				
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	0	1	0	R		R	s <sub>1</sub>	
	0	1	0	0	0	1	0	1	0	0	0	R		R	S <sub>2</sub>	
	De	fault I	D		CF	ls <sub>1</sub>			CF	Rs <sub>2</sub>		0	0	0	0	0
Operands	<ul> <li>Rs<sub>1</sub> TMS34020 source register for first the 32-bit single-precision floating-point value to coprocessor</li> <li>Rs<sub>2</sub> TMS34020 source register for the second 32-bit single-precision floating-point value to coprocessor</li> </ul>															float-
	<ul> <li>Rs1 TMS34020 source register for first the S2-bit single-precision float ing-point value to coprocessor</li> <li>Rs2 TMS34020 source register for the second 32-bit single-precision float ing-point value to coprocessor</li> <li>CRs1 Coprocessor register to contain the first 32-bit single-precision</li> </ul>															float-
	<ul> <li>Rs<sub>2</sub> TMS34020 source register for the second 32-bit single-precision flo ing-point value to coprocessor</li> <li>CRs<sub>1</sub> Coprocessor register to contain the first 32-bit single-precisi floating-point operand</li> </ul>															ision
	CRs <sub>2</sub>	C flo	opro patin	cess g-poi	or re int op	giste beran	r to Id	cont	ain th	ne se	conc	1 32-	bit s	ingle	-prec	ision
Description	CMPI and C status	F loa CRs <sub>2</sub> s bits	ids th resp in th	ne co pectiv ne co	onten /ely, oproc	ts (si subtr esso	ngle acts r sta	-prec CRs tus re	ision 2 froi egiste	i valu m CF er.	e) of ls <sub>1</sub> , a	Rs <sub>1</sub> and s	and ets t	Rs <sub>2</sub> i he ap	into ( oprop	CRs <sub>1</sub> oriate
Machine States	4 if th 3 if th	e fir: e fir:	st ins st ins	struct struct	ion v ion v	vord i vord i	is lor is no	ng wa t long	ord-a g woi	ligneo rd-alio	d gned					
Instruction Type	СМО	VGC	), two	o reg	isters	5										
Example	CMPF	A5,	A6,	RAS	5, RI	36										
	This e RA5 a RA5,	exan and I and	nple l RB6 i sets	oads respe the :	TMS ective statu	63402 ely, si s bits	20 re ubtra s in tl	giste cts tl ne cc	rs A5 ne co proc	and ntent	A6 in s of I r stat	to co RB6 f us re	proc from egiste	esso the c er.	r regi ontei	sters nts of

Syntax	CVDF	CR	s, CRc	1												
Execution	(CRs)	$\rightarrow CF$	ld													
Instruction Words	15 1 Defa	14 1: 1 0 ault ID	3 12 1	11 1 Cl	10 0 Rs	9 0	8 0 0	7 0 1	6 0 0	5 1 0	4	3 1	2 1 CRd	1	0	
Operands	CRs CRd	Default ID       CRs       0       1       0       0       CRd         CRs       Coprocessor source register containing a 64-bit double-precision floating-point operand       64-bit double-precision       64-bit double-precision         CRd       Coprocessor destination register       0       1       0       0       CRd         CVDF       converts a 64-bit IEEE double-precision floating-point number to       0       0       0       0														
Description	CVDF 32-bit ber res	conv IEEE s sides i	erts a single- n CRs,	64-b precis and t	it IEE sionf hecc	EE de loatir onver	ouble 1g-po ted s	e-pre oint ni ingle	cisioi umbe -prec	n floa er. Th cision	ating- e dou num	poin Ible- Iber r	t nur preci eside	nber sion es in	to a num- CRd.	
	The co	oproce	essor s	source	e regi	ster,	CRs	, mu	st be	in th	e A c	opro	cess	or fil	э.	
Machine States	2															
Instruction Type	CEXE	C, sho	ort													
Example	CVDF	RA5,	RA7													
	This e numbe	xampl er and	e conv stores	verts t s the	he co resuli	nten t in F	ts of A7.	RA5	to a s	single	e-pre	cisio	n floa	ting-	point	

## CVDI Convert, Double Precision to Integer

Syntax	CVD	C	Rs, C	CRd												
Execution	(CRs	$) \rightarrow $	CRd													
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	1	1	0	1	1	0	0	0	0	0	1	1	1	1	1	1
	De	fault	D		CI	Rs		0	1	0	1			CRd		
Operands	Default ID       CRs       0       1       0       1       CRd         CRs       Coprocessor source register containing a 64-bit double-precision floating-point operand       CRd       Coprocessor destination register															ision
	Сна	U	opro	cess	or ae	stina	tion	regis	ter							
Description	CVD integ verte	i con er nu d inte	verts umbe eger	a 64 r. Th num	-bit IE le do ber r	EEE o uble- eside	loub prec s in	le-pre ision CRd.	ecisio num	on flo Iber 1	ating reside	-poir es in	it nur CRs	nber , and	to a 3 I the	2-bit con-
	The c	copro	ocess	sor so	ource	e regi	ster,	CRs	, mu	st be	in th	e A c	opro	cess	or fil	э.
Machine States	2															
Instruction Type	CEXI	EC, s	short													
Example	CVDI	RA5	, RE	57												
	This ( in RE	exan 37.	nple (	conve	erts t	he co	onter	its of	RA5	to ar	n inte	ger a	nd st	ores	the r	esult

Syntax	CVFD C	Rs, C	CRd												
Execution	$(CRs) \rightarrow 0$	CRd													
Instruction Words	15 14 1 1 Default I	13 0 D	12	11 1 CF	10 0 Rs	9 0	8 0 0	7 0 1	6 0 0	5 1 0	4	3 1	2 1 CRd	1 1	0
Operands	Default ID       CRs       0       1       0       0       CRd         CRs       Coprocessor source register containing a 32-bit single-precision floating-point operand       32-bit single-precision floating point value to a 64 line         CRd       Coprocessor destination register       CVED converts a 32-bit IEEE single precision floating point value to a 64 line														
Description	CVFD cor IEEE dou resides in	iverts ible-p CRs	s a 32 precis , and	2-bit sion the	IEEE float conv	E sing ing-p rerteo	gle-p point d dou	recisi valu ıble-p	ion fl e. T precis	oatin he s sion r	g-poi ingle numt	int va -prec per re	alue t cisior eside	o a 6 nur s in C	4-bit nber CRd.
	The copro	cess	or so	ource	regi	ster,	CRs	, mus	st be	in th	e A c	opro	cess	or file	Э.
Machine States	2														
Instruction Type	CEXEC, s	short													
Example	CVFD RA5	, RB	7												
	This exam stores the	nple c resu	onve It in I	erts th RB7.	ne co	nten	ts of	RA5	to a c	doubl	e-pre	ecisio	on nu	mbei	and

Syntax	CVFD R	s, CRs, (	CRd												
Execution	$Rs \rightarrow CRs$ (CRs) $\rightarrow C$	s CRd													
Instruction Words	15 14	13 12	11	10	9	8	7	6	5	4	3	2	1		
		0 0			1	1	0	0	0	<u>п</u> 0	0		0		
	Default I		C	Rs		0	1	0	0	-		CRd			
Operands	Default IDCRs0100CRdRsTMS34020 source register for the 32-bit single-precision float ing-point value to coprocessor32-bit single-precision float ing-precision floating point operand														
	<ul> <li>Rs TMS34020 source register for the 32-bit single-precision float ing-point value to coprocessor</li> <li>CRs Coprocessor register to contain the 32-bit single-precision floating point operand</li> </ul>														
	CRd Co	oprocess	sor de	estina	tion	regis	ter								
Description	CVFD load 32-bit IEEE sion floatin converted	ds the co E single- ng-point v double-	onten precia value orecia	ts (sii sion f . The sion r	ngle- loatir sing numb	preci ng-po le-pre er re	ision) bint va ecisic sides	of R alue t on nu s in C	s inte o a 6 mber Rd.	o CR 4-bit <sup>-</sup> resid	s an IEEE des i	d cor E doul n CR	iverts ble-p s, and	s the reci- d the	
	The copro	cessor s	ource	e regi	ster,	CRs	, mus	st be	in th	e A c	opro	cess	or file	9.	
Machine States	3 if the firs 2 if the firs	st instruc st instruc	tion v tion v	vord i vord i	s lon s no	g wo t long	ord-al g wor	ignec d-aliç	d gned						
Instruction Type	CMOVGC	, one reg	gister												
Example	CVFD B5,	RA5, R	A7												
	This exam verts the c in RA7.	ple loads ontents	STMS	3402 5 to a	20 reg a dou	gister ıble-p	B5 ir Brecis	nto co sion r	proc numb	esso er, a	r reg nd st	ister l ores	RA5, the r	con- esult	

:

Syntax	CVFI	CF	ls, C	Rd												
Execution	(CRs)	$\rightarrow$ C	Rd													
Instruction Words	15 1 Def	14 1 ault IC	13 0 D	12 1	11 1 CF	10 0 Rs	9 0	8 0 0	7 0 1	6 0 0	5 1 1	4	3 1	2 1 CRd	1	0
Operands	CRs CRd	Default ID       CRs       0       1       0       1       CRd         CRs       Coprocessor source register containing a 32-bit single-precision floating-point operand       32-bit single-precision floating-point operand         CRd       Coprocessor destination register         CVFI converts a 32-bit IEEE single-precision floating-point value to a 32-bit														
Description	CVFI intege intege	conv er val er nui	verts ue. 1 mbe	a 32 The s r resi	e-bit ingle des	IEEE -pree	sing cisior Rd.	le-pr ח nun	ecisi nber	on flo resid	oating es in	g-poi CRs	nt va , and	lue t the d	o a 3 conve	2-bit erted
	The c	opro	cess	or so	ource	e regi	ster,	CRs	, mu	st be	in th	e A c	opro	cess	or file	э.
Machine States	2															
Instruction Type	CEXE	C, s	hort													
Example	CVFI	RA5,	RA	7												
	This e in RA	xam 7.	ple c	conve	erts t	he co	onten	ts of	RA5	to ar	inte	ger a	nd st	ores	the r	esult

Syntax	CVFI	Rs,	CRs	s, Cl	Rd											
Execution	Rs → <i>(CRs)</i>	$CRs \rightarrow C$	Rd													
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	0	0	1	R		R	s	
	0	1	0	1	1	1	1	1	0	0	0	0	0	0	0	0
	Der	auit IL	<u> </u>		CI	15		0						Сна		
Operands	Rs TMS34020 source register for the 32-bit single-precision floating-point value to coprocessor CRs Coprocessor register to contain the 32-bit single-precision floating															float-
	<ul> <li>Rs TMS34020 source register for the 32-bit single-precision float ing-point value to coprocessor</li> <li>CRs Coprocessor register to contain the 32-bit single-precision floating point operand</li> </ul>															ating-
	CRd	Co	proc	essi	or de	estina	tion	regis	ster							
Description	CVFI 32-bit single reside	loads IEEE -prec es in (	s the E sing cisior CRd.	cor gle-p n nu	ntents preci mbe	s (sir sion f r resi	igle-i iloati des	oreci ng-po in Cl	sion) oint v Rs, a	of R value ind th	s into to a ( ne co	o CR 32-bi nveri	s an t inte ed ir	d cor ger v ntege	nvert value er nu	s the . The mber
	The c	oprod	cesso	or so	ource	e regi	ister,	CRs	s, mu	st be	in th	e A c	opro	cess	or fil	e.
Machine States	3 if the 2 if the	e firsi e firsi	t inst t inst	ruct	ion v ion v	vord i vord i	is lor is no	ng wo t long	ord-a g woi	ligne rd-ali	d gned					
Instruction Type	CMO	VGC,	one	reg	ister											
Example	CVFI	в5,	RA5,	, RE	37											
	This e verts f	xamp the c	ole lo ontei	ads nts c	TMS of RA	3402 5 to	20 reg an ir	gister itege	r B5 i er, an	nto co d sto	oproc res th	esso ne re:	or reg sult i	ister n RB	RA5, 7.	con-

Syntax	CVID	С	Rs, C	Rd												
Execution	(CRs	$) \rightarrow 0$	CRd													
Instruction Words	15 1 De	14 1 fault I	13 0 D	12 1	11 1 CF	10 0 Rs	9 0	8 0 0	7 0 1	6 0 1	5 1 0	4	3 1	2 1 CRd	1	0
Operands	CRs CRd	Default ID       CRs       0       1       1       0       CRd         CRs       Coprocessor source register containing the 32-bit integer operand         CRd       Coprocessor destination register         CVID converts a 32-bit integer value to a 64-bit IEEE double-precision floating														
Description	CVID point numb	con valu per re	verts e. Th eside	a 32- le int s in (	bit in eger CRd.	tege resic	r valu les ir	ie to a n CR	a 64- s, ar	bit IE Id the	EE d e con	ouble verte	e-pre ed do	cisio uble	n floa -prec	ting- ision
	The c	copro	ocess	or so	ource	e regi	ster,	CRs	, mu	st be	in th	eΑc	copro	cess	or fil	э.
Machine States	2															
Instruction Type	CEXI	EC, s	short													
Example	CVID	RA5	, RB	7												
	This estore	exan s the	nple c resu	onve Ilt in	erts ti RB7.	ne co	nten	ts of	RA5	to a c	loubl	le-pro	ecisio	on nu	mbe	r and

Syntax	CVID	R	s, Cł	Rs, C	Rd											
Execution	Rs <i>→</i> <i>(CRs</i>	→ CR	s CRd													
Instruction Words	_15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	0	1	0	R		R	3	
	0	1	0	1	1	1	1	1	1	0	0	R		R	S	
	De	fault	ID		C	Rs		0	1	1	0			CRd		
Operands	Rs TMS34020 source register for the 32-bit integer values to coprocess CRs Coprocessor source register to contain the 32-bit integer operand															ssor
	Rs TMS34020 source register for the 32-bit integer values to coprocess CRs Coprocessor source register to contain the 32-bit integer operand															۱d
	CRs Coprocessor source register to contain the 32-bit integer operand CRd Coprocessor destination register															
Description	CVIE value resid (Con word	load to es in strai s of	ds the a 64 i CRs nts o the 6	e con -bit s, and f the 4-bit	tents IEEE d the TMS trans	inte dou conv 3408 sfer.)	ger) Ible-p /erte 32 re	of Re precis d dou quire	s into sion uble- e that	CRs floati preci the i	and ing-p sion integ	conv oint num er in	verts a value ber re Rs b	a 32-l e. Th eside e ser	oit int e int s in ( nt as	eger eger CRd. both
	The o	copro	oces	sor s	ource	e reg	ister,	CRs	s, mu	st be	in th	e A d	copro	cess	or file	).
Machine States	4 if fi 3 if fi	rst ir rst ir	istruc istruc	ction ction	word word	l is lo l is no	ng w ot lor	ord-ang wo	aligno ord-a	ed ligne	d					
Instruction Type	CMC	VG	C, two	o reg	ister	S										
Example	CVID	в5,	RAS	5, R#	¥7											
	This verts in RA	exan the 7.	nple l conte	loads ents d	STMS of RA	5342 5 to a	0 reg a dou	ister Jble-	B5 ir preci	nto co sion i	proc numt	esso per, a	r reg and si	ister l tores	RA5, the r	con- əsult

Syntax	CVIF	CRs, (	CRd												
Execution	(CRs)	$\rightarrow$ CRd													
Instruction Words	15 1 Defa	14 13 1 0 ault ID	12 1	11 1 Cl	10 0 Rs	9 0	8 0 0	7 0 1	6 0 1	5 1 0	4	3 1	2 1 CRd	1	0
Operands	CRs CRd	Copro Copro	cess cess	or sc or de	ource estina	regis tion	ster o regis	conta ter	iining	the :	32-bi	t inte	ger o	opera	Ind
Description	CVIF c point v numbe	onverts alue. T er reside	a 32 he in es in (	-bit ir tegei CRd.	ntege r resi	r valı des i	ue to n CF	a 32 Rs, a	-bit IE nd th	EEE s e coi	single nvert	e-pre ed s	cisio ingle	n floa -prec	ting- ision
	The co	proces	sor se	ource	e regi	ster,	CRs	s, mu	st be	in th	e A c	copro	cess	or file	ə.
Machine States	2														
Instruction Type	CEXE	C, shor	:												
Example	CVIF 1	RA5, RA	47												
	This ex stores	kample the res	conv ult in	erts t RA7	he co	onter	nts of	RAS	ō to a	singl	e-pre	ecisio	on nu	mbe	rand

Syntax	CVIF	R	s, CF	Rs, C	Rd											
Execution	Rs – <i>(CRs</i>	>CR	s CRd													
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	0	0	1	R		R	s	
	0	1	0	1	1	1	1	1	0	0	0	0	0	0	0	0
	De	efault	ID		CI	Rs		0	1	1	0			CRd		
Operands	Rs	Т	MS3	4020	sou	rce re	egiste	er for	the 3	32-bit	integ	ger va	alue	to co	oroce	essor
	CRs	С	opro	cess	or so	ource	regi	ster t		ntain	the 3	32-bit	inte	ger o	pera	nd
	CRd	С	opro	cess	or de	estina	tion	regis	ster							
Description	CVIF value in CF	loac to a Rs, a	ls the 32-b nd th	e con it IEE e co	tents E sir nvert	(inte Igle-p ed si	ger) precis ngle	of Rs sion f -prec	into loatir ision	CRs ng-po num	and bint va ber r	conv alue. eside	erts a The i es in	a 32- ntege CRd	bit in er re: ·	teger sides
	The	copro	ocess	sor s	ource	e regi	ister,	CRs	s, mu	st be	in th	e A c	copro	cess	or fil	e.
Machine States	3 if fi 2 if fi	rst in rst in	struc	tion tion	word word	l is lo l is no	ng w ot lor	vord-a	aligno ord-a	ed ligne	d					
Instruction Type	СМС	VGC	C, on	e reg	lister											
Example	CVIF	RA5	, RA	17												
	This store	exan s the	nple ( e resi	conv ult in	erts t RA7	he co	onter	nts of	RA5	i to a	singl	e-pre	ecisio	on nu	mbe	r and

Syntax	DIVD CRs <sub>1</sub> , CRs <sub>2</sub> , CRd
Execution	$\left(\frac{CRs_1}{CRs_2}\right) \rightarrow CRd$
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0
	1 1 0 1 1 0 0 0 0 0 1 0 1 1 1
	Default ID CRs <sub>1</sub> CRs <sub>2</sub> CRd
Operands	CRs <sub>1</sub> Coprocessor register containing the first 64-bit double-precision floating-point operand
	CRs <sub>2</sub> Coprocessor register containing the second 64-bit double-precision floating-point operand
	CRd Coprocessor destination register
Description	DIVD divides the contents (double-precision value) of $\text{CRs}_1$ by $\text{CRs}_2$ and stores the result CRd.
Machine States	2
Instruction Type	CEXEC, short
Example	DIVD RA5, RB6, RA7
	This example divides the contents of RA5 by RB6 and stores the result in RA7.

Syntax	<b>DIVF</b> CRs <sub>1</sub> , CRs <sub>2</sub> , CRd
Execution	$\left(\frac{CRs_1}{CRs_2}\right) \rightarrow CRd$
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0
	1 1 0 1 1 0 0 0 0 0 1 0 1 1 0
	Default ID CRs <sub>1</sub> CRs <sub>2</sub> CRd
Operands	CRs <sub>1</sub> Coprocessor register containing the first 32-bit single-precision floa- ting-point operand
	CRs <sub>2</sub> Coprocessor register containing the second 32-bit single-precision floating-point operand
	CRd Coprocessor destination register
Description	DIVF divides the contents (single-precision value) of $\mbox{CRs}_1$ by $\mbox{CRs}_2$ and stores the result in CRd.
Machine States	2
Instruction Type	CEXEC, short
Example	DIVF RA5, RB6, RA7
	This example divides the contents of RA5 by RB6 and stores the result in RA7.

Syntax	DIVF	Rs	, Rs	5 <sub>2</sub> , C	Rs <sub>1</sub> ,	CRs	2, Cl	Rd								
Execution	Rs₁ –	► CR	ls <sub>1</sub>													
	Rs₂ →	→ CR	S <sub>2</sub>													
	$\left(\frac{CRs}{CRs}\right)$	$\left(\frac{3_{1}}{3_{2}}\right) - \frac{3_{1}}{3_{2}}$	→ CF	٦d												
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	0	1	0	R		Rs	<sup>5</sup> 1	
			0	1	0	0	1	1	0	0	0	R		CBd	<sup>5</sup> 2	
Operands	Rs <sub>1</sub>	TN cis	/IS34	1020 value	sour e to c	ce re opro	giste cess	r for t or	he fir	st 32	-bit fl	oatin	ig-pc	oint si	ngle-	pre-
	Rs <sub>2</sub>	TN pre	∕IS3∠ ∋cisi	1020 on va	sour alue 1	ce re to coj	giste proce	r for t essor	the se	econ	d 32-	bit flo	patin	g-poi	nt sin	gle-
	CRs <sub>1</sub>	Cc tin	proo g-po	cesso int o	or reg pera	gister nd	to c	ontai	n the	e first	32-b	oit sir	ıgle-j	oreci	sion 1	floa-
	CRs <sub>2</sub>	Co flo	oproc ating	cesso g-poi	or reg nt op	gister eran	rtod d	conta	in th	e se	cond	32-t	oit si	ngle-	preci	sion
	CRd	Co	proc	cesso	or de	stina	tion r	egist	er							
Description	DIVF I CRs <sub>1</sub> a the res	oads and ( sult i	s the CRs <sub>2</sub> n CF	con 2 res Rd.	tents pectiv	(sing vely, d	gle pr divide	ecisi esthe	on, fl e cont	oatin tents	g poi of CF	int) o Rs <sub>1</sub> b	fRs- yCR	l and Is <sub>2</sub> , a	Rs <sub>2</sub> nd ste	into ores
Machine States	4 if the 3 if the	e first e first	t insi t insi	tructi tructi	on w on w	ord is ord is	s lon s not	g woi long	rd-ali word	gned 1-alig	ned					
Instruction Type	СМО∖	/GC,	two	regi	sters											
Example	DIVF	A5,	Аб,	RA5	, RB	6, R	A7									
	This ex RA5 ai result	kamp nd R in R/	ole lo B6 re A7.	ads espe	TMS ctive	3402 ly, div	0 reg vides	jister the c	s A5 a ontei	and A nts of	16 int FRA5	o cop i by F	oroce RB6, a	essor and s	regis tores	ters the

### DIVS Divide, Integer

Syntax	<b>DIVS</b> CRs <sub>1</sub> , CRs <sub>2</sub> , CRd
Execution	$\left(\frac{CRs_1}{CRs_2}\right) \rightarrow CRd$
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0
	1 1 0 1 1 0 0 0 0 0 1 0 0 1 0 0
	Default ID CRs1 CRs2 CRd
Operands	CRs1Coprocessor register containing the first 32-bit integer operandCRs2Coprocessor register containing the second 32-bit integer operandCRdCoprocessor destination register
Description	DIVS divides the contents (integer) of $\mbox{CRs}_1$ by $\mbox{CRs}_2$ and stores the result in CRd.
Machine States	2
Instruction Type	CEXEC, short
Example	DIVS RA5, RB6, RB7
	This example divides the contents of RA5 by RB6 and stores the result in RB7.

Svntax	DIVS	Rs1. Rs2. CRs1. CRs2. CRd
Cymax	5110	

**Execution**  $Rs_1 \rightarrow CRs_1$ 

 $Rs_2 \rightarrow CRs_2$ 

$$\left(\frac{CRs_1}{CRs_2}\right) \rightarrow CRd$$

Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	0	1	0	R		R	s <sub>1</sub>	
	0	1	0	1	0	0	1	0	0	0	0	R		R	s <sub>2</sub>	
	D	efault	ID		CF	₹s <sub>1</sub>			CF	ls <sub>2</sub>				CRd		
Operands	Rs <sub>1</sub>	T C	MS3 esso	4020 r	sou	rce re	egiste	er for	the 1	first 3	32-bit	: inte	ger v	value	to co	pro-
	Rs <sub>2</sub>	T C	MS3 oproc	4020 cesso	sou or	rce i	regist	ter fo	or the	e sec	ond	32-b	it in	teger	· valu	ie to
	CRs	1 C	opro	cess	or re	giste	r to c	ontai	n the	first	32-b	oit inte	eger	oper	rand	
	CRs	2 C	opro	cess	or re	giste	r to c	ontai	n the	sec	ond 3	32-bit	inte	ger o	opera	nd
	CRd	С	opro	cess	or de	stina	ition i	regist	ter							
Description	DIVS tively	6 loac /, divi	ls the des 1	e coni the c	tents onter	(inte nts of	ger) ( <sup>:</sup> CRs	of Rs <sub>1</sub> by	1 and CRs	l Rs <sub>2</sub> <sub>2</sub> , an	into ( d sto	CRs <sub>1</sub> ores t	and he re	l CRs esult	s <sub>2</sub> res in CF	pec- d.
Machine States	4 if ti 3 if ti	he fir he fir	st ins st ins	struct struct	ion w ion w	vord i vord i	s Ion s not	g wo : long	rd-ali wor	gnec d-alig	l jned					
Instruction Type	СМС	VGC	C, two	o reg	isters	3										
Example	DIVS	A5,	A6,	RA5	, RE	86, F	RB7									
	This RA5 resu	exan and I It in F	nple l RB6 r RB7.	oads espe	TMS ective	3402 ly, di	20 reę vides	gister the c	s A5 conte	and / nts o	۹6 int f RA	to cop 5 by F	oroce RB6,	and :	r regi: store	sters s the

Syntax	GET	сѕт														
Execution	Copr	Coprocessor Status Register → ST														
Instruction Words	15	15     14     13     12     11     10     9     8     7     6     5     4     3     2     1     0       0     0     0     0     1     1     0     0     1     1     0     0     0     0     0     0														
	0	0	0	0	0	1	1	0	0	1	1	0	0	0	0	0
	0	1	0	0	1	1	1	0	0	0	0	0	0	0	0	1
	De	efault	D	0	0	0	0	0	0	0	0	0	1	1	0	0
<i>Description</i> <i>Machine States</i>	GET TMS 5 if th 4 if th	GETCST loads 4 MSBs of the coprocessor status register (STATUS) into the TMS34020 status register (ST). 5 if the first instruction word is long word-aligned 4 if the first instruction word is not long word-aligned														o the
Instruction Type	СМС	VCS	;													
Example	GETC	ST														
	This TMS in the	exan 3402 e TM	nple 0 tak S340	send es th 20 s	s the e val tatus	e cop ue ar regi:	roce: nd ma ster c	ssor asks corre	statu off th spon	s reg e 4 M ding	jister ISBs to th	to th ; it th e N,	ne TN en st C, Z,	/IS34 uffs t V bit	020. he va ts.	The dues

ι

Syntax	INVD CRs2	, CRd												
Execution	$\left(\frac{1}{CRs_2}\right) \rightarrow$	► CRd												
Instruction Words	15 14	13 12	11	10	9	8	7	6	5	4	3	2	1	0
	1 1	0 1	1	0	0	0	0	0	1	0	1	0	1	1
	Default ID	0	0	0	0		CF	₹s₂				CRd		
Operands	CRs <sub>2</sub> Co floa CRd Co	process ating-po process	or re int op or de	egiste beran estina	er-B <sup></sup> id ation	file c regis	contai	ining	the	64-b	it do	ouble	-prec	ision
Description	INVD divide result in CF	es 1.0 b Rd.	y the	e con	tents	(dou	uble p	orecis	sion)	of C	Rs <sub>2</sub>	and s	store	s the
Machine States	2													
Instruction Type	CEXEC, sh	ort												
Example	INVD RB3,	RA1												
	This examp	ole divid	es 1.	0 by	RB3	and	store	es the	resi	ult in	RA1	•		

Syntax	INVF CRs <sub>2</sub> , CRd
Execution	$\left(\frac{1}{CRs_2}\right) \rightarrow CRd$
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0
	1 1 0 1 1 0 0 0 0 0 1 0 1 0 1 0
	Default ID         0         0         0         CRs2         CRd
Operands	CRs <sub>2</sub> Coprocessor register-B file containing the 32-bit single-precision floa- ting-point operand
	CRd Coprocessor destination register
Description	INVF divides 1.0 by the contents (single precision, floating point) of $\mbox{CRs}_2$ and stores the result in CRd.
Machine States	2
Instruction Type	CEXEC, short
Example	INVF RB3, RA1
	This example divides 1.0 by RB3 and stores the result in RA1.

د. ۱ ۱ ۱ ۱ ۱ ۱ Syntax INVF Rs, CRs<sub>2</sub>, CRd

*Execution*  $Rs \rightarrow CRs_2$ 

$$\left(\frac{1}{CRs_2}\right) \rightarrow CRd$$

Instruction Words

Operands

15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
0	0	0	0	0	1	1	0	0	0	1	R		R	s	
0	1	0	1	0	1	0	1	0	0	0	0	0	0	0	0
D	efault	ID	0	0	0	0		CF	₹s₂				CRd		

Rs TMS34020 source register for the 32-bit floating-point single-precision value to coprocessor

> CRs<sub>2</sub> Coprocessor register-B file to contain the 32-bit single-precision floating-point operand

CRd Coprocessor destination register

*Description* INVF loads the contents (single precision, floating point) of Rs into CRs<sub>2</sub>, divides 1.0 by CRs<sub>2</sub>, and stores the result in CRd.

Machine States3 if the first instruction word is long word-aligned2 if the first instruction word is not long word-aligned

Instruction Type CMOVGC, one register

Example INVF A7, RB3, RA1

This example loads TMS34020 register A7 into coprocessor register RB3, divides 1.0 by RB3, and stores the result in RA1.

#### JUMPC Execute Coprocessor External Instructions

Syntax	JUMPC n											
Execution	Execute external coprocessor instructions found at address $2 \times n$											
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0											
	0 0 0 0 0 1 1 0 0 0 0 0 0 0 0 0											
	1         1         n         0											
	Default ID         0											
Operands	<i>n</i> Specifies the address to which the TMS34082 instruction execution is sent											
Description	JUMPC begins execution of TMS34082 external instructions stored in TMS34082 local memory. The starting address is specified as TMS34082 local memory address $2 \times n$ . Usually, a jump table is stored in these locations to permit complex operations.											
Machine States	3 if the first instruction word is long word-aligned 2 if the first instruction word is not long word-aligned											
Instruction Type	CEXEC, long											
Example	JUMPC 4											
	This example executes TMS34082 instructions stored in the defa TMS34082's local memory. The executed instructions are stored beginning address 8.											

Syntax	MOVD	Rs <sub>1</sub> ,	Rs <sub>2</sub> ,	CRd
--------	------	-------------------	-------------------	-----

*Execution*  $Rs_1, Rs_2 \rightarrow CRd$ 

Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0		
	0	0	0	0	0	1	1	0	0	1	0	R		Rs	91			
	0	1	0	0	1	1	0	1	1	0	0	R	Rs <sub>2</sub>					
	Defa	ault I	D	0	0	0	0	0	0	0	0		CRd					
Operands	Rs <sub>1</sub> TMS34020 source register for the 32 MSBs (sign, exponent, and 20 MSBs of mantissa) of the 64-bit double-precision floating-point value to coprocessor																	
	Rs <sub>2</sub> TMS34020 source register for the 32 LSBs of the 64-bit double-precision floating-point value to coprocessor									64-bit								
	CRd Coprocessor destination register that holds the 64-bit double-precision floating-point value																	
Description	MOVD moves the double-precision value in $Rs_1$ and $Rs_2$ into CRd. $Rs_1$ holds the 32 MSBs, and $Rs_2$ holds the 32 LSBs of the double. You must set the LOAD bit of the TMS34082 configuration register to 0 to indicate that the MSBs are transferred before the LSBs.																	
Machine States	4 if the first instruction word is long word-aligned 3 if the first instruction word is not long word-aligned																	
Instruction Type	CMOVGC, two registers																	
Example	This example uses the MOVD instruction to load a 64-bit double-precision val- ue into register RB5. Note that the 32 MSBs of the value are loaded into a1 and then the 32 LSBs are loaded into A0. Assume that the LOAD bit of the configu- ration register is set to 0, indicating transfers of MSBs before LSBs.																	
		000 000 010 020 040	00	054 05a 0000 05a	10 10 00" 1		m m	ieee etf ove ove	fl 32 @d <sup>,</sup> @d <sup>,</sup>	,0,0 val, val+	a0,0 32,a	1,0						
		050 070 080 090 000	00 8a	0000 064 5f8 1f9 6a51	20" 11 30 95 ad	dval	m •	ovd data doub	al le 3	,a0,: 47.6	rb5 9422	38						
	00000	020	40	75bb	1b		•					-						
Syntax	MO\	/D	*Rs+	, CR	d, Ra	1												
-------------------	---	--	---	--	--	---	--	---	---	--	--	---	--	--	--	---	--	--
Execution	lf T Rep	MS3 eat *Rs Rs *Rs Rs adva	4082 anc 16 tin → C + 32 → C + 32 ance regi	LOA I Rd nes Rd (3 → Rs Rd (3 → Rs to ne ster	<b>D bi</b> = <b>0</b> 2 MS 2 LS 3 xt co	t <b>= 0</b> SBs) Bs) proc	esso	r	If TMS34082 LOAD bit = 1 and Rd = 1 $\rightarrow$ 31 Repeat Rd/2 times *Rs $\rightarrow$ CRd (32 MSBs) Rs + 32 $\rightarrow$ Rs *Rs $\rightarrow$ CRd (32 LSBs) Rs + 32 $\rightarrow$ Rs advance to next coprocessor register									
	If T	MS3 >eat *Rs Rs *Rs Rs adva	4082 and 16 tin → Cl + 32 → Cl + 32 ance regi	LOA Rd (3 $\rightarrow$ Rs Rd (3 $\rightarrow$ Rs to ne ster	<b>D bi</b> = <b>0</b> 2 LS 2 MS 2 MS xt co	t = 1 Bs) SBs) proce	esso	and Rd = 1 → 31 Repeat Rd/2 times *Rs → CRd (32 LSBs) Rs + 32 → Rs *Rs → CRd (32 MSBs) Rs + 32 → Rs advance to next coprocessor register										
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0		
	0	0	0	0	0	1	1	0	1	1	1	R		F	łd			
		ofault		0			0			0	0	<u>н</u>						
Operands	Rs CRd	T a to C s	MS3 ddre o the copro ion fl	4020 ss of copro cess oatin	sou the fi oces: or de g-poi	rce r rst 3 sor stina nt va	egist 2-bits tion i	er (i s of tl regis	ndire he firs ter to	ct po st dou hold	stinc uble- the t	reme preci first 6	ent) sion 34-bi	conta valu t dou	aining e to m ıble-p	the nove reci-		
	Rd	T T	MS3 his v	4020 alue	regis must	ster c in th	ontai Ie rar	ining nge (	the n ) to 3	umbo 1.	erof	32-bi	t trar	sfer	s to m	ake.		
		Ļ	l If	Rd =	0,			ther	1 32 32-bit transfers are made									
			l If	Rd =	1 →	31		ther	n <i>Rd</i> 32-bit transfers are made									
		N n	lote t umb	hat b er in l	ecal Rd w	ise 6 ill giv	4-bit ve un	doul pred	bles r ictabl	equii e res	re tw sults.	o 32·	-bit r	nove	s, an	odd		
Description	MOV addr the c cess sequ tents	/D m ess ii onte or de ience s of F	oves n Rs i nts of estina e list. Id. Th	64-b nto co f Rs a tion i The res	it do oproo re inc s adv numt sults	uble- cesso crem vance oer of will b	preci or reg entec ed to 32-t oe un	ision ister d; aft the r bit tra pred	value s beg er eve next re insfer ictabl	es fro innin ery tw egiste s ma le if F	om m g at ( vo 32 er in t de is Rd is	emo CRd. -bit tr he co dete an o	ry be After ansf oproc ermir dd n	eginn reac ers, t cesso red b umb	ing at h tran the co or reg y the er.	t the sfer, pro- ister con-		

The TMS34082 configuration register LOAD bit determines whether the LSBs or the MSBs will be moved first:

	If the LOAD bit =	= 1, then the LSBs a (32 LSBs of the	re moved first fraction)									
	☐ If the LOAD bit =	= 0, then the MSBs a (sign, exponent	are moved first , and 20 MSBs of the fraction)									
	The LOAD bit default is 0.											
Machine States	If Rd = 0 and If Rd = 0 and If Rd = 1 → 31 and If Rd = 1 → 31 and	Rs is aligned Rs is nonaligned Rs is aligned Rs is nonaligned	36 37 5 + (Rd – 1) 6 + (Rd – 1)									
Instruction Type	CMOVMC, postincre	ement, register count										
Example	MOVD *A5+, RB7, B	37										
	This example move memory location poir After each 32-bit tran transfers, the coproc coprocessor register	es 64-bit double-prec nted to by A5 to coproc nsfer, register A5 is in essor destination is a sequence list. B7 hol	ision values from the TMS34020 essor registers beginning with RB7. ncremented; after every two 32-bit idvanced to the next register in the ds the number of 32-bit transfers to									

be made.

Syntax	MOV	/D	*Rs+	, CR	d, [, c	count	]									
Execution	lf T Rep	MS34 beat ( *Rs Rs *Rs Rs adva	4082 $\rightarrow$ C + 32 $\rightarrow$ C + 32 ance regi	LOA time $Rd$ ( $\Rightarrow$ $\rightarrow$ Rd $\rightarrow$ Rd to ne ster	AD bi as 32 M s 22 LS s ext co	<b>t = 0</b> SBs) Bs) proc	If TMS34082 LOAD bit = 1 Repeat <i>count</i> times Repeat <i>count</i> times *Rs $\rightarrow$ CRd (32 LSBs) Rs + 32 $\rightarrow$ Rs *Rs $\rightarrow$ CRd (32 MSBs) Rs + 32 $\rightarrow$ Rs advance to next coprocessor register									
Instruction Words		14	13	12		10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	1	0	0		tr	ansfers	i	
		0	0	0	1	1	0	1	1	0	0	R		Rs		
Operands	Rs TMS34020 source register (indirect postincrement) containing the address of the first 32-bits of the first double-precision value to move to the coprocessor															
CRd Coprocessor destination register that holds the first 64-bit d cision floating-point value												bit dou	ıble-	pre-		
	count Contains the number of 64-bit transfers to make. This value must in the range 1 to 16; the default value is 1. <i>Count</i> determines the value or transfers:															the le of
	If $count = 16$ , then transfers = 0															
	□ If <i>count</i> = 1 $\rightarrow$ 15, then transfers = 2 × <i>count</i>															
Description	MOVD moves 64-bit double-precision values from memory beginning at the address in Rs into coprocessor registers beginning at CRd. After each transfer the contents of Rs are incremented; after every two 32-bit transfers, the coprocessor destination is advanced to the next register in the coprocessor register sequence list. The number of 64-bit transfers made is determined by the contents of <i>count</i> .														t the sfer, pro- ister con-	
	The or th	TMS: e MS	3408 Bs v	2 cor vill be	nfigur e mov	ation ved fi	regi: rst:	ster L	.OAD	bit d	eterr	nine	s whe	ether t	he L	SBs
		f the	LOA	D bit	= 1,	the (32	n the LSE	e LSE Bs of	3s are the fr	e mov actio	ved f n)	irst				
		fthe	LOA	D bit	= 0,	the (się	n the gn, e	e MSI xpon	Bs ar ent, a	e mo and 2	oved 20 MS	first SBs (	of the	e fracti	ion)	
	The	LOA	D bit	defa	ult is	0.										
Machine States	Rs A Rs N	ligne Ional	d igneo	ť	5 · 6 ·	+ ((Co	ount	× 2) - × 2) -	– 1) – 1)							

Instruction Type	CMOVMC, postincrement, constant count
------------------	---------------------------------------

Example MOVD \*A5+, RB7, 4

This example moves four 64-bit double-precision values from the TMS34020 memory location pointed to by A5 to coprocessor registers beginning with RB7. After each 32-bit transfer, register A5 is incremented; after every two 32-bit transfers, the coprocessor destination is advanced to the next register in the coprocessor register sequence list. *Count* specifies that four 64-bit transfers are made (eight 32-bit transfers).

Syntax	MOVD _*Rs, CRd [, count]														
Execution	If TMS34082 LOAD bit = 0 Repeat <i>count</i> times $Rs - 32 \rightarrow Rs$ *Rs $\rightarrow CRd$ (32 MSBs) $Rs - 32 \rightarrow Rs$ *Rs $\rightarrow CRd$ (32 LSBs) advance to next coprocessor register	If TMS34082 LOAD bit = 1 Repeat <i>count</i> times $Rs - 32 \rightarrow Rs$ *Rs $\rightarrow CRd$ (32 LSBs) $Rs - 32 \rightarrow Rs$ *Rs $\rightarrow CRd$ (32 MSBs) advance to next coprocessor register													
Instruction Words	15 14 13 12 11 10 9 8	3 7 6 5 4 3 2 1 0													
		0 0 0 1 transfers													
	Default ID 0 0 0 0														
Operands	Rs TMS34020 source register (indirect predecrement) containing the address of the bit immediately following the 64-bits used to store the first 64-bit double-precision floating-point value that is transferred														
	CRd Coprocessor destination register that holds the first 64-bit double- precision floating-point value														
	count Contains the number of 64-bit transfers to make. This value must in the range 1 to 16; the default value is 1. <i>Count</i> determines the value of transfers:														
	<b>I</b> If <i>count</i> = 16, th	ien transfers = 0													
	$\Box  \text{If } count = 1 \rightarrow 15,  \text{th}$	ien transfers = 2 × <i>count</i>													
Description	MOVD moves 64-bit double-precision address (Rs –32) into coprocessor is 32-bit transfer, the contents of Rs a ransfers, the coprocessor destination coprocessor register sequence list. Intermined by the contents of <i>count</i>	on values from memory beginning at the registers beginning at CRd. Before each are decremented; after every two 32-bit on is advanced to the next register in the The number of 64-bit transfers made is <i>t</i> .													
	The TMS34082 configuration registe or the MSBs will be moved first:	er LOAD bit determines whether the LSBs													
	If the LOAD bit = 1, then the L (32 LSBs	.SBs are moved first of the fraction)													
	If the LOAD bit = 0, then the M (sign, exp	ISBs are moved first onent, and 20 MSBs of the fraction)													
	The LOAD bit default is 0.														
Machine States	Rs Aligned5 + ((count × 2Rs Nonaligned6 + ((count × 2	2) — 1) 2) — 1)													

Instruction Type	CMOVMC, predecremen	t, constant count
------------------	---------------------	-------------------

Example MOVD --\*A5, RB7, 4

This example moves four 64-bit double-precision values from the TMS34020 memory location pointed to by (A5 - 32) to coprocessor registers beginning with RB7. Before each 32-bit transfer, register A5 is decremented; after every two 32-bit transfers, the coprocessor destination is advanced to the next register in the coprocessor register sequence list. *Count* specifies that four 64-bit transfers are made (eight 32-bit transfers).

Syntax	MOVD CRd, *Rd-	+ [, <i>count</i> ]													
Execution	If TMS34082 LOA Repeat <i>count</i> time CRd (32 MSB Rd + 32 $\rightarrow$ Rd CRd (32 LSBs Rd + 32 $\rightarrow$ Rd advance to ne register	D bit = 0 s s) →*Rd ) → *Rd xt coprocessor	If TMS34082 LOAD bit = 1 Repeat <i>count</i> times CRd (32 LSBs) $\rightarrow$ *Rd Rd + 32 $\rightarrow$ Rd CRd (32 MSBs) $\rightarrow$ *Rd Rd + 32 $\rightarrow$ Rd advance to next coprocessor register												
Instruction Words	15 14 13 12	11 10 9 8	7 6 5 4 3 2 1 0												
			1 0 1 H H0												
	Default ID 0	0 0 0 0	0 0 0 CRd												
Operands	CRd Coprocessor source register for the first 64-bit double-prec floating-point value to the TMS34020 memory Rd TMS34020 register (indirect postincrement) containing the addre														
	the first double-precision value transferred														
	count Contains the number of 64-bit transfers to make. This value must in the range 1 to 16; the default value is 1. <i>Count</i> determines the value of transfers:														
	📮 If count	t = 16, ther	n transfers = 0												
	📮 If count	$t = 1 \rightarrow 15$ , then	n transfers = 2 × <i>count</i>												
Description	MOVD moves the 6 beginning at CRd to transfer, Rd is increr ister is advanced to number of 64-bit tra	64-bit double-precis memory beginning nented, and after ev the next register in t unsfers made is del	sion values from coprocessor registers g at the address in Rd. After each 32-bit very two transfers, the coprocessor reg- he coprocessor register sequence. The termined by the contents of <i>count</i> .												
	The TMS34082 configuration register LOAD bit determines whether the LSBs or the MSBs will be moved first:														
	If the LOAD bit	= 1, then the LSE (32 LSBs of	Bs are moved first the fraction)												
	If the LOAD bit	= 0, then the MS (sign, expon	Bs are moved first hent, and 20 MSBs of the fraction)												
	The LOAD bit defau	ult is 0.													
Machine States	Rd aligned Rd nonaligned	5 + (count*2 – 1 6 + (count*2 – 1	)												
Instruction Type	CMOVCM, postincr	ement, constant co	ount												

### Example

### MOVD RB7, \*A5+, 2

This example moves four 64-bit double-precision values from coprocessor registers beginning at RB7 to TMS34020 memory pointed to by A5. After each 32-bit transfer, register A5 is incremented, and after every two transfers, the coprocessor destination is advanced to the next register in the coprocessor register sequence list. *Count* specifies that two 64-bit transfers are made (four 32-bit transfers).

ExecutionIf TMS34082 LOAD bit = 0 Repeat count times Rd - 32 $\rightarrow$ Rd CRd (32 MSBs) $\rightarrow$ *Rd Rd - 32 $\rightarrow$ Rd CRd (32 LSBs) $\rightarrow$ *Rd Rd - 32 $\rightarrow$ Rd CRd S2 $\rightarrow$ Rd CRd (32 LSBs) $\rightarrow$ *Rd Rd - 32 $\rightarrow$ Rd Rd - 32 $\rightarrow$ Rd CRd (32 LSB $\rightarrow$ CRd - 22 $\rightarrow$ Rd CRd - 2	Syntax	MO	VD	CRd,	-*R	d [, c	ount												
Instruction Words       15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         Image: Instruction Words       15       14       13       12       11       1       0       1       1       1       1       1       0       0       1       1       1       1       1       0       0       1       1       1       1       1       0       0       1       1       1       1       1       1       0       0       1       1       1       1       1       1       0       0       1       0       1	Execution	lf T Re	MS3 peat of Rd - CRo Rd - CRo adva	4082 - 32 - I (32 - 32 - I (32 ance regi	LOA t time → Rc MSB → Rc LSBs to ne ister	(D bi) es (s) → (s) → s) → ext co	<b>t = 0</b> *Rd *Rd proc	If TMS34082 LOAD bit = 1 Repeat <i>count</i> times $Rd - 32 \rightarrow Rd$ $CRd (32 LSBs) \rightarrow *Rd$ $Rd - 32 \rightarrow Rd$ $CRd (32 MSBs) \rightarrow *Rd$ advance to next coprocessor											
Instruction fieldImage: The field of the fie	Instruction Words																		
1       0       0       1       1       1       1       0       0       transfers         Default ID       0		0	0	0	0	0	1	1	0	1	1	0	R	T	Rd	<u>.</u>	$\overline{}$		
Default ID       0 <td< th=""><th></th><th>1</th><th>0</th><th>0</th><th>0</th><th>1</th><th>1</th><th>1</th><th>1</th><th>1</th><th>0</th><th>0</th><th></th><th>ti</th><th>ransfers</th><th></th><th></th></td<>		1	0	0	0	1	1	1	1	1	0	0		ti	ransfers				
Operands       CRd       Coprocessor source register for the first double-precision value to TMS34020 memory         Rd       TMS34020 register (indirect predecrement) containing the address of the bit immediately following the 64-bit used to store the first 64-bit double-precision floating-point value that is transferred         count       Contains the number of 64-bit transfers to make. This value must in the range 1 to 16; the default value is 1. Count determines the value of transfers:         If count = 16,       then transfers = 0         If count = 1 → 15,       then transfers = 2 × count         Description       MOVD moves the 64-bit double-precision values from coprocessor registers beginning at CRd to memory beginning at the address (Rd - 32). Before each 32-bit transfer, Rd is decremented; after every two 32-bit transfers, the coprocessor register is advanced to the next register in the coprocessor register sequence. The number of 64-bit transfers made is determined by the contents of count.         The TMS34082 configuration register LOAD bit determines whether the LSBs or the MSBs will be moved first:       If the LOAD bit = 1, then the LSBs are moved first (32 LSBs of the fraction)         If the LOAD bit = 0, then the MSBs are moved first (sign, exponent, and 20 MSBs of the fraction)       The LOAD bit default is 0.         Machine States       Rd aligned 5 + (count*2 - 1)       Rd anonaligned 6 + (count*2 - 1)         Instruction Type       CMOVCM, predecrement, constant count       If every top of the fraction)			Default	ID	0	0	0	0	0	0	0	0			CRd				
Rd       TMS34020 register (indirect predecrement) containing the address of the bit immediately following the 64-bits used to store the first 64-bit double-precision floating-point value that is transferred         count       Contains the number of 64-bit transfers to make. This value must in the range 1 to 16; the default value is 1. Count determines the value of transfers:         Image 1 to 16; the default value is 1. Count determines the value of transfers:         Image 1 to 16; the default value is 1. Count determines the value of transfers:         Image 1 to 16; the default value is 1. Count determines the value of transfers:         Image 1 to 16; the default value is 1. Count determines the value of transfers:         Image 1 to 16; the default value is 1. Count determines the value of transfers:         Image 1 to 16; the default value is 1. Count determines the value of transfers:         Image 1 to 16; the default value is 1. Count determines the value of transfers:         Image 1 to 200 the 1 to 10; the default value is 1. Count determines the value of transfers:         Image: transfer 1 to 16; the default value is 1. Count determines the value of transfers to 2 × count         Description         MOVD moves the 64-bit double-precision values from coprocessor register is advanced to the next register in the coprocessor register sequence. The number of 64-bit transfers made is determined by the contents of count.         The TMS34082 configuration register LOAD bit determines whether the LSBs or the MSBs will be moved first (32 LSBs of the fraction)         Imathe LOAD bit default is 0. </th <th>Operands</th> <th>CRo</th> <th>d C T</th> <th>Copro MS3</th> <th>cess 4020</th> <th>or so ) mer</th> <th>ource nory</th> <th>regi</th> <th>ster</th> <th>for th</th> <th>ne firs</th> <th>st do</th> <th>uble</th> <th>-prec</th> <th>cision v</th> <th>alu</th> <th>e to</th>	Operands	CRo	d C T	Copro MS3	cess 4020	or so ) mer	ource nory	regi	ster	for th	ne firs	st do	uble	-prec	cision v	alu	e to		
countContains the number of 64-bit transfers to make. This value must in the range 1 to 16; the default value is 1. Count determines the value of transfers:Image: Image: Imag	•	Rd TMS34020 register (indirect predecrement) containing the address of the bit immediately following the 64-bits used to store the first 64-bit double-precision floating-point value that is transferred														ss of 4-bit			
If count = 16,then transfers = 0If count = 1 $\rightarrow$ 15,then transfers = 2 × countDescriptionMOVD moves the 64-bit double-precision values from coprocessor registers beginning at CRd to memory beginning at the address (Rd - 32). Before each 32-bit transfer, Rd is decremented; after every two 32-bit transfers, the copro- cessor register is advanced to the next register in the coprocessor register sequence. The number of 64-bit transfers made is determined by the contents of count.The TMS34082 configuration register LOAD bit determines whether the LSBs or the MSBs will be moved first: (32 LSBs of the fraction)If the LOAD bit = 1,then the LSBs are moved first (32 LSBs of the fraction)Machine StatesRd aligned 6 + (count*2 - 1) Rd nonaligned5 + (count*2 - 1) 6 + (count*2 - 1)Instruction TypeCMOVCM, predecrement, constant count		count Contains the number of 64-bit transfers to make. This value must in the range 1 to 16; the default value is 1. <i>Count</i> determines the value o transfers:													the e of				
□If count = 1 → 15, then transfers = 2 × countDescriptionMOVD moves the 64-bit double-precision values from coprocessor registers beginning at CRd to memory beginning at the address (Rd - 32). Before each 32-bit transfer, Rd is decremented; after every two 32-bit transfers, the copro- cessor register is advanced to the next register in the coprocessor register sequence. The number of 64-bit transfers made is determined by the contents of count.The TMS34082 configuration register LOAD bit determines whether the LSBs or the MSBs will be moved first: □If the LOAD bit = 1, then the LSBs are moved first (32 LSBs of the fraction)□If the LOAD bit = 0, then the MSBs are moved first (sign, exponent, and 20 MSBs of the fraction)Machine StatesRd aligned 6 + (count*2 - 1) Rd nonalignedMachine TypeCMOVCM, predecrement, constant count			Ę	l If	coun	t = 10	6,		then	n trans	sfers	= 0							
DescriptionMOVD moves the 64-bit double-precision values from coprocessor registers beginning at CRd to memory beginning at the address (Rd – 32). Before each 32-bit transfers, Rd is decremented; after every two 32-bit transfers, the copro- cessor register is advanced to the next register in the coprocessor register sequence. The number of 64-bit transfers made is determined by the contents of count.The TMS34082 configuration register LOAD bit determines whether the LSBs or the MSBs will be moved first: I If the LOAD bit = 1, then the LSBs are moved first (32 LSBs of the fraction)If the LOAD bit = 0, then the MSBs are moved first (sign, exponent, and 20 MSBs of the fraction)The LOAD bit default is 0.Machine StatesRd aligned 6 + (count*2 - 1) Rd nonaligned 6 + (count*2 - 1)Instruction TypeCMOVCM, predecrement, constant count			Ę	l if	coun	t = 1	→ 1	5,	ther	n tran	sfers	= 2 :	× coι	ınt					
The TMS34082 configuration register LOAD bit determines whether the LSBs or the MSBs will be moved first:If the LOAD bit = 1, then the LSBs are moved first (32 LSBs of the fraction)If the LOAD bit = 0, then the MSBs are moved first (sign, exponent, and 20 MSBs of the fraction)The LOAD bit default is 0.Machine StatesRd aligned 6 + (count*2 - 1) Rd nonalignedInstruction TypeCMOVCM, predecrement, constant count	Description	MO beg 32-b cess seq of ce	VD m inning bit tran sor re uence bunt.	oves g at C nsfer, egiste e. The	the 6 Rd to , Rd is er is a e nun	64-bil o mer s dec advar nber o	t dou nory reme nced of 64	ble-p begin ented to th bit tr	recis nning ; afte e ne ansfe	sion v g at th er eve xt reg ers m	alues le ado ry tw gister ade is	s fror dress o 32 in th s det	n coj s (Rd -bit tr ne co ermii	oroce l – 32 ransf oproc ned b	essor re 2). Befo ers, the cessor i by the c	egis re e e co regi onte	ters each pro- ster ents		
<ul> <li>If the LOAD bit = 1, then the LSBs are moved first (32 LSBs of the fraction)</li> <li>If the LOAD bit = 0, then the MSBs are moved first (sign, exponent, and 20 MSBs of the fraction)</li> <li>The LOAD bit default is 0.</li> <li>Machine States</li> <li>Rd aligned 5 + (count*2 − 1) Rd nonaligned 6 + (count*2 − 1)</li> <li>Instruction Type</li> <li>CMOVCM, predecrement, constant count</li> </ul>		The or th	TMS ne MS	3408 3Bs v	2 cor vill be	nfigur e mov	atior ved fi	regi rst:	sterl		) bit d	eterr	nine	s wh	ether th	e L	SBs		
<ul> <li>If the LOAD bit = 0, then the MSBs are moved first (sign, exponent, and 20 MSBs of the fraction)</li> <li>The LOAD bit default is 0.</li> <li>Machine States</li> <li>Rd aligned 5 + (count*2 - 1) Rd nonaligned 6 + (count*2 - 1)</li> <li>Instruction Type</li> <li>CMOVCM, predecrement, constant count</li> </ul>			lf the	LOA	D bit	= 1,	the (32	en the 2 LSE	e LSE Bs of	3s are the fr	e mo actio	ved f n)	irst						
Machine States       Rd aligned       5 + (count*2 - 1)         Rd nonaligned       6 + (count*2 - 1)         Instruction Type       CMOVCM, predecrement, constant count			lf the	LOA	D bit	= 0,	the (sig	en the gn, e	e MS xpon	Bs ar ient, a	re mo and 2	oved 20 MS	first SBs (	of the	e fractio	on)			
Machine StatesRd aligned a nonaligned5 + (count*2 - 1) 6 + (count*2 - 1)Instruction TypeCMOVCM, predecrement, constant count		The	LOA	D bit	defa	ult is	0.												
Instruction Type CMOVCM, predecrement, constant count	Machine States	Rd a Rd i	aligne nonal	ed ignec	4	5 6	+ (co + (co	unt*2 unt*2	2 — 1) 2 — 1)	)									
	Instruction Type	CM	OVC	۸, pre	edeci	reme	nt, co	onsta	nt cc	ount									

Example

#### MOVD RB7, -\*A5, 2

This example moves two 64-bit double-precision values from coprocessor registers beginning at RB7 to TMS34020 memory pointed to by (A5-32). Before each 32-bit transfer, register A5 is decremented; after every two 32-bit transfers, the coprocessor destination is advanced to the next register in the coprocessor register sequence list. *Count* specifies that two 64-bit transfers are made (four 32-bit transfers).

Syntax	MOVD CRs <sub>1</sub> , CRd														
Execution	CRs <sub>1</sub> → CRd														
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0														
	1 1 0 1 1 0 0 0 0 1 1 0 1 1 1														
	Default ID         CRs1         0         0         1         CRd														
Operands	CRs <sub>1</sub> Coprocessor source register A that holds the 64-bit double-precision floating-point value														
	CRd Coprocessor destination register														
Description	MOVD moves a 64-bit double-precision value from $CRs_1$ (register A) to CRd														
Machine States	2														
Instruction Type	CEXEC, short														
Example	MOVD RA7, RB4														
	This example moves the 64-bit double-precision value from coprocessor register RA7 to coprocessor register RB4.														

Syntax	MOVD CRs <sub>2</sub> , CRd														
Execution	$CRs_2 \rightarrow CRd$														
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         1       1       0       1       1       0       0       0       0       1       1       1       0       1       1         Default ID       CRs2       0       0       0       1       CRd       CRd														
Operands	CRs <sub>2</sub> Coprocessor source register B that holds the 64-bit double-precision floating-point value														
	CRd Coprocessor destination register														
Description	MOVD moves a 64-bit double-precision floating-point value from CRs <sub>2</sub> (regis- ter B) to CRd.														
Machine States	2														
Instruction Type	CEXEC, short														
Example	MOVD RB3, RB4														
	This example moves the 64-bit double-precision value from coprocessor register RB3 to coprocessor register RB4.														

Syntax	MOVE Rs, CRd														
Execution	$Rs \rightarrow CRd$														
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1														
	0 0 0 0 0 1 1 0 0 0 1 R Rs														
	0 1 0 0 1 1 0 0 0 0 0 0 0 0 0 0 0														
	Default ID         0         0         0         0         0         0         0         CRd														
Operands	<ul><li>Rs TMS34020 source register for 32-bit integer value to coprocessor</li><li>CRd Coprocessor destination register to hold the 32-bit integer</li></ul>														
Description	MOVE moves the contents (integer) of Rs into CRd.														
Machine States	3 if the first instruction word is long word-aligned 2 if the first instruction word is not long word-aligned														
Instruction Type	CMOVGC, one register														
Example	MOVE A5, RA7														
	This example moves the contents of TMS34020 register A5 into coprocessoregister RA7.	or													

Q.)

Syntax	MOV	E f	Rs <sub>1</sub> , I	₽ <i>s</i> 2,	CRd														
Execution	$Rs_1 \rightarrow CRd$ $Rs_2 \rightarrow CRd+1$																		
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	3 2 1 0					
	0	0	0	0	0	1	1	0	0	1	0	R		R	S <sub>1</sub>				
	0	1	0	0	1	1	0	0	0	0	0	R		R	S2				
	De	fault I	D	0	0	0	0	0	0	0	0			CRd					
Operands	Rs <sub>1</sub> TMS34020 source register for the first 32-bit integer value to coprocessor																		
	Rs <sub>2</sub> TMS34020 source register for the second 32-bit integer value to coprocessor														e to				
	CRd Coprocessor destination register that holds the first 32-bit integer value. The second integer will be placed in the next register in the coprocessor register sequence list.														val- pro-				
Description	MOVI	Emo	oves	the c	ontei	nts (i	nteg	er) of	Rs <sub>1</sub>	and	Rs <sub>2</sub> i	nto C	CRd	and (	CRd	+1.			
Machine States	4 if th 3 if th	e firs e firs	st ins st ins	tructi tructi	on w on w	ord is	s Ion s not	g wo Iong	rd-ali word	gned d-alig	ned								
Instruction Type	СМО	VGC	, two	regi	sters														
Example	MOVE	A5,	Аб,	RA7															
	This ir proce	nstru ssor	regis	n mov ster F	res th RA7 a	ie cor and F	ntent RA8,	s of T resp	MS3 ective	4020 ely.	) regi	sters	A5 a	and A	6 into	) CO-			

Syntax	MOV	E *	Rs +	, CR	d, Ro	1									
Execution	lf <b>RE</b> Repo	<b>) = 0</b> eat 3 *Rs - Rs + adva	32 tim → CF - 32 - nce t regi	nes Rd → Rs to ne: ster	s xt co	proce	essol	r	lf <b>R</b> Rej	d = 1 beat / *Rs Rs adva	$  \rightarrow 3$ Rd tin $\rightarrow C$ + 32 ance r	<b>31</b> mes Rd → R to ne regist	s ext ce	oproces	sor
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2 1	0
	0	0	0	0	0	1	1	0	1	1	1	R		Rd	
	1	0	0	0	1	1	0	0	0	0	0	R		Rs	
	De	fault I	D	0	0	0	0	0	0	0	0			CRd	
Operands	Rs	T di	MS3 ress	4020 of the	sour e first	ce re : 32-l	giste pit int	er (ind æger	lirect to m	posti ove t	ncrei o the	ment e cop	) cor roce	itaining I ssor	he ad-
	<ul> <li>CRd Coprocessor destination register to hold the first 32-bit integoperand</li> <li>Rd TMS34020 register containing the number of 32-bit transfers to ma This value must in the range 0 to 31</li> </ul>														nteger
															make.
			l lf <i>i</i>	Rd =	0,			then	<i>32</i> 3	2-bit	trans	sfers	are	made	
			I If /	Rd =	1 →	31,		then	Rd 3	32-bit	tran	sfers	are	made	
Description	MOVI copro ented quent of <i>Rd</i>	E mo ocess I, and ce lis /.	oves sor re d CRe t. The	intege egiste d is ac e nun	er va ers b dvan nber	lues eginr ced t of 32	from ning o the -bit tr	mem at CF next ansfe	iory b Rd. A regis ers m	egin (fter o ter in ade i	ning each the c s det	at the tran copro ermir	e ado sfer, ocess ned b	lress in I Rs is ir sor regis by the co	Rs into icrem- ter se- ntents
Machine States	If Rd	= 0 a	and		Rs	s is a	ligne	d		36					
	If Rd	= 0 a	and		Rs	s is n	onali	gned		37					
	lf Rd If Rd	= 1 - = 1 -	→ 31 → 31	and and	Rs Rs	s is a s is n	ligne onali	d gned		5 + ( 6 + (	Rd – Rd –	· 1) · 1)			
Instruction Type	CMO	VMC	, pos	stincr	eme	nt, re	giste	er cou	unt						
Example	MOVE	*A5	+, R	A7,	в7										
	This pointe transf vance the nu	instr ed to fer, r ed to umbe	uctio by A egist the r er of	n ma 5 to a er As next ra 32-bi	oves copro 5 is ir egist it trar	inte ocess ocrer er in osfer	ger sor re nente the c s to b	value egiste ed, a oproc oe ma	s fro ers be nd th cesso ade.	om T eginn ie co or reg	MS3 ing a proce jister	4020 It RA essor sequ	me 7. Af <sup>.</sup> des ienc	mory lo ter each stination e list. B7	cation 32-bit is ad- 'holds

Syntax	MOV	E *	Rs+,	CRa	l, [, c	ount	l									
Execution	Repe *  F a	at <i>co</i> Rs → Rs + dvar	<i>unt</i> t ► CR 32 – ice to	times d → Rs o the	next	copr	oces	sor r	egiste	ər						
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	1	0	0		tr	ansfer	S	
		0	0	0	1	1	0	0	0	0	0	R		R	S	
Operands	Rs	TI dr	MS34 ress	4020 of the	sour e first	ce re : 32-t	giste bit int	r (ind eger	lirect to m	posti ove t	ncrei o the	nent) copi	) con roce	itainir ssor	ng the	ead-
	CRd Coprocessor destination register to hold the first 32-bit integ operand															eger
	operand count Contains the number of 32-bit transfers to make. This value must the in the range 1 to 32; the default value is 1. <i>Count</i> determines the value of transfers:															st be alue
			lf a	count	t = 32	2,		then	trans	sfers	= 0					
			lf d	count	t = 1	<b>→</b> 31	,	then	trans	sfers	= <i>co</i>	unt				
Description	MOV Rs int creme in the is det	E mo to co enteo copr ermi	oves a proce d, an oces ned l	32-bi essor d the sor ro by the	t inte r regi copi egist e cor	eger v isters roces er se ntents	alue beg sor c quen s of <i>c</i>	s from innin Jestir ce lis <i>ount</i>	m me g at ( natior st. The	mory CRd. 1 is a e nun	v beg After dvan nber	innin eacl ced t of 32	g at h tra to th -bit t	the a nsfer e nex ransf	ddre , Rs i t reg ers m	ss in s in- ister nade
Machine States	Rs Al Rs No	igne onali	d gned	]	5 - 6 -	+ (coi + (coi	unt – unt –	1) 1)								
Instruction Type	СМО	VMC	, pos	stincr	eme	nt, co	onsta	nt co	unt							
Example	MOVE	*A5·	+, R	в7 <b>,</b>	4											
	This e tion p 32-bit advar specit	exam ointe tran nced fies t	ple n ed to sfer, to th hat fe	nove: by A regis e nex our 3	s fou 5 to ter A t reg 2-bit	r 32-l copro 5 is i ister trans	bit int bcess ncrer in the sfers	eger sor re nent e cop are r	value egiste ed, a roces made	es fro ers be nd th ssor r	om TN eginr e cop regist	AS34 ning a proce ter se	1020 at RI essor eque	merr 37. A r dest nce li	nory li fter e tinatio ist. <i>C</i> é	oca- each on is ount

Syntax	MOVE -	Rs, CR	d [, ca	ount]											
Execution	Repeat <i>col</i> Rs – 3 *Rs → advand	unt times 32 → Rs CRd ce to the	s next	сорг	roces	sor r	egist	er							
Instruction Words	15 14	13 12	11	10	9	8	7	6	5	4	3	2	1	0	
	0 0	0 0	1	0	0	0	0	0	1		tr	ransfe	'S		
	1 0	0 0	1	1	0	0	0	0	0	R		F	S		
	Default ID 0 0 0 0 0 0 0 0 0 0 CRd														
Operands	<ul> <li>Rs TMS34020 source register (indirect postincrement) containing the address of the bit immediately after first 32-bit integer to move to the coprocessor</li> <li>CRd Coprocessor destination register to hold the first 32-bit integer</li> </ul>														
	CRd Coprocessor destination register to hold the first 32-bit integrouper and														
	count Contains the number of 32-bit transfers to make. This value must in the range 1 to 32; the default value is 1. <i>Count</i> determines the value transfers:														
	<b>L</b>	lf coun	t = 32	2,		then	tran	sfers	= 0						
	G	lf coun	<i>t</i> = 1	→ 3 <sup>-</sup>	1,	then	tran	sfers	= CO	unt					
Description	MOVE mov (Rs – 32) in the content nation is ad list. The nu <i>count</i> .	ves 32-b nto copro s of Rs a dvanced umber o	it inte ocess re de to th f 32-l	eger v sor re crem e ne: oit tra	value egiste iente xt reg ansfe	s froi ers be d; afte jister rs m	m me eginn er ea in th ade	emor ing a chtra e co is de	y beg at CR ansfe oroce term	innin d. Be r, the essor ined	ig at fore copi regi by th	the a each roces ster s ne co	ddre: i tran sor d seque nten	ss in Isfer, Iesti- ence ts of	
Machine States	Rs Aligned Rs Nonalig	ined	5 · 6 ·	+ (co + (co	unt – unt –	· 1) · 1)									
Instruction Type	СМО∨МС,	predeci	eme	nt, co	onsta	nt co	unt								
Example	MOVE -*A5	, RB7,	4												
	This examp tion pointed each 32-bit sor destina quence list	ble move d to by (A transfer tion is ac . <i>Count</i>	es fou A5 – 3 , regi dvanc speci	r 32- 32) to ster A ced to fies t	bit inf copr \5 is c o the that fe	teger oces decre next our 3	valu sor r men regis 2-bit	es fro egist ted; a ter in trans	om Ti ers b after o the o sfers	MS34 eginr each copro are r	4020 hing a trans bcess made	) men at RB sfer, c sor re e.	iory I 7. Be opro giste	oca- efore ces- r se-	

Syntax	MOV	ΕC	CRd,	*Rd-	+ [, ca	ount]											
Execution	Repe C R a	at <i>co</i> Rs - d + : dvar	ount → *R 32 → ice to	time d Rd o the	s next	copr	oces	sor r	egiste	ər							
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0	
	0	0	0	0	0	1	1	0	1	0	1	R		R	d		
		0	0	0	1	1	1	0	0	0	0		tra	ansfer	S		
Operands	CRd	C T	opro MS34	cess 4020	ssor source register for the first 32-bit integer value to 20 memory												
	Rd TMS34020 register (indirect postincrement) containing the addre for the first integer transferred															ress	
	count Contains the number of 32-bit transfers to make. This value must the range 1 to 32; the default value is 1. <i>Count</i> determines the value of transfers:															st in alue	
			lfo	coun	t = 32	<u>2,</u>		then	trans	sfers	= 0						
			lfo	coun	t = 1	→ 31	<b>I</b> , <sup>1</sup>	then	trans	sfers	= <i>CO</i>	unt					
Description	MOVI at CR Rd is i ter in t is det	E mo d to incre he c ermi	oves men opro ned l	the 3 hory l ted, a cess by th	2-bit begin and th or reg e cor	integ ining ne co gister itents	jer va at th proc sequ s of c	alues e ad esso uence count	from dress r regi e. The	copi in R steri enur	roces d. Af s adv nber	sor r ter e ance of 32	egist ach 3 ed to 1 -bit tr	ters b 32-bit the n ransfe	egin tran extre ersm	ning sfer, gis- iade	
Machine States	Rs Ali Rs No	igne onali	d gned	I	5 - 6 -	+ (coi + (coi	unt – unt –	- 1) - 1)									
Instruction Type	СМО	VCN	l, pos	stincr	eme	nt, co	onsta	nt cc	ount								
Example	MOVE	RB7	<b>,</b> *A	5+,	4												
	This e ginnir fer, re to the that fo	exam Ig at giste nex our 3	ple n RB7 er A5 t regi 2-bit	nove to TM is in ister tran	s fou /IS34 crem in the sfers	r 32-l 020 r ente cop are r	bit int mem d, an roces made	eger ory p d the ssor i e.	value ointe copi regist	es fro d to b roces er se	om co oy A5 ssor c equer	proc Afte destin nce li	esso r eac natio st. <i>C</i>	r regi h 32- n is a ount	isters bit tra dvar spec	be- ans- iced ifies	

Syntax	MOVE CRd, -*Rd [, count]													
Execution	Repeat <i>count</i> times $Rd - 32 \rightarrow Rd$ $CRd \rightarrow *Rd$ advance to the next coprocessor register													
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         0       0       0       0       1       1       0       1       1       1       Rd         1       0       0       0       1       1       0       0       0       transfers         Default ID       0       0       0       0       0       0       0       CRd													
Operands	<ul> <li>CRd Coprocessor source register for the first 32-bit integer value to TMS34020 memory</li> <li>Rd TMS34020 register (indirect predecrement) containing the address of the bit immediately following the 32-bits used to store the first 32-bit</li> </ul>													
	<ul> <li>the bit immediately following the 32-bits used to store the first 32-bits integer value transferred</li> <li>count Contains the number of 32-bit transfers to make. This value must in the range 1 to 32; the default value is 1. <i>Count</i> determines the value of transfers:</li> <li>If <i>count</i> = 32, then transfers = 0</li> </ul>													
Description	■ If <i>count</i> = 1 → 31, then transfers = <i>count</i> MOVE moves the 32-bit integer values from coprocessor registers beginning at CRd to memory beginning at the address (Rd – 32). Before each 32-bit trans- fer, Rd is decremented; after each 32-bit transfer, the coprocessor register is advanced to the next register in the coprocessor register sequence. The num- ber of 32-bit transfers made is determined by the contents of <i>count</i> .													
Machine States	Rs Aligned $5 + (count - 1)$ Rs Nonaligned $6 + (count - 1)$													
Instruction Type	CMOVCM, predecrement, constant count													
Example	MOVE RB7, -*A5, 4													
	This example moves four 32-bit integer values from coprocessor registers be- ginning at RB7 to TMS34020 memory pointed to by $(A5 - 32)$ . Before each 32-bit transfer, register A5 is decremented; after each 32-bit transfer, the co- processor destination is advanced to the next register in the coprocessor regis- ter sequence list. <i>Count</i> specifies that four 32-bit transfers are made.													

Syntax	MOV	E	CRd,	Rd												
Execution	CRd	→ R	d													
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	0	1	1	R		R	d	
	0	1	0	0	1	1	1	0	0	0	0	0	0	0	0	0
	Default ID         0         0         0         0         0         0         0         CRd															
Operands	CRd Coprocessor source register holding the 32-bit integer value															
	Rd TMS34020 destination register															
Description	MOVE moves 32-bit integer from coprocessor register CRd to TMS34020 re ister Rd.															reg-
Machine States	5 if th 4 if th	ie firs ie firs	st ins st ins	truct truct	ion w ion w	vord i vord i	s Ion s not	g wo : long	rd-al 1 wor	igneo d-alio	d gned					
Instruction Type	СМО	VCG	i, one	e reg	ister											
Example	MOVE	RA7	<b>,</b> A5													
	This e regist	exan ter A	nple r 5.	nove	s the	cont	tents	ofco	oproc	esso	or reg	ister	RA7	to T	MS34	1020

## **MOVE** Move, Integer, Coprocessor to Coprocessor

Syntax	MOVE CRs <sub>1</sub> , CRd													
Execution	$CRs_1 \rightarrow CRd$													
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         1       1       0       1       1       0       0       0       0       1       1       0       0       0         Default ID       CRs1       0       0       0       0       1       CRd       CRd													
Operands	CRs <sub>1</sub> Coprocessor source register A that holds the 32-bit integer value CRd Coprocessor destination register													
Description	MOVE moves 32-bit integer value from CRs <sub>1</sub> (register A) to CRd.													
Machine States	2													
Instruction Type	CEXEC, short													
Example	MOVE RA7, RB4													
	This example moves the 32-bit integer value from coprocessor register RA7 to coprocessor register RB4.													

Syntax	MOVE CRs <sub>2</sub> , CRd													
Execution	$CRs_2 \rightarrow CRd$													
Instruction Words	$            \begin{array}{c cccccccccccccccccccccccc$													
Operands	CRs <sub>2</sub> Coprocessor source register B that holds the 32-bit integer value CRd Coprocessor destination register													
Description	MOVE moves a 32-bit integer value from CRs <sub>2</sub> (register B) to CRd.													
Machine States	2													
Instruction Type	CEXEC, short													
Example	MOVE RB3, RB4													
	This example moves the 32-bit integer value from coprocessor register RB3 to coprocessor register RB4.													

## **MOVF** Move, Single Precision, One Register to Coprocessor

Syntax	MOVF	Rs, (	CRd												
Execution	Rs → C	Rd													
Instruction Words	15 14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0 0	0	0	0	1	1	0	0	0	1	R		R	s	
	0 1	0	0	1	1	0	. 1	0	0	0	0	0	0	0	0
	Defau	It ID	0	0	0	0	0	0	0	0			CRd		
Operands	Rs TMS34020 source register for the 32-bit single-precision float ing-point value to coprocessor														
	CRd Coprocessor destination register to hold the 32-bit single-precisi floating-point value														
Description	MOVF	noves	the c	onte	nts (s	single	e-pre	ecisio	on val	ue) o	f Rs	into	CRd.		
Machine States	3 if the 1 2 if the 1	first ins first ins	struct struct	ion w ion w	/ord i /ord i	s Ion s not	ig wo t long	ord-a g wo	ligne rd-ali	d gned					
Instruction Type	CMOVO	GC, on	e reg	ister											
Example	MOVF A	5, RA7	,												
	This exa register	ample RA7.	move	es the	coni	tents	of T	MS3	4020	regis	ster A	\5 int	to cop	oroce	essor

Syntax	MOVF	Rs <sub>1</sub> , I	₽ <i>s</i> 2,	CRd											
Execution	Rs <sub>1</sub> → Rs <sub>2</sub> →	CRd CRd+ <sup>-</sup>	ł												
Instruction Words	15 1	4 13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0 (		0	0	1	1	0	0	1	0	R		Rs	1	
	Defa		0	1	1	0	1	0	0	0	н		CBd	2	
Operands	<ul> <li>Rs<sub>1</sub> TMS34020 source register for the first 32-bit single-precision floa ing-point value to coprocessor</li> <li>Rs<sub>2</sub> TMS34020 source register for the second 32-bit single-precision floa</li> </ul>														loat-
	Rs <sub>2</sub> TMS34020 source register for the second 32-bit single-precision float ing-point value to coprocessor														
	<ul> <li>ing-point value to coprocessor</li> <li>CRd Coprocessor destination register to hold the first single-precision value. The second single-precision value will be placed in the next register in the coprocessor register sequence list.</li> </ul>														
Description	MOVF and CF	moves ld+1.	the c	onte	nts (s	ingle	e-preo	cisior	n valu	ie) of	'Rs <sub>1</sub>	and	Rs <sub>2</sub>	into	CRd
Machine States	4 if the 3 if the	first ins first ins	tructi tructi	ion w ion w	ord is ord is	s Ion s not	g woi Iong	rd-ali word	gned d-alig	ned					
Instruction Type	CMOV	GC, two	o regi	sters	;										
Example	MOVD A	5, A6,	RB7												
	This ex process	ample r sor regi	nove sters	s the RB7	cont and	ents RB8	of TN	MS34	1020	regis	ters	A5 a	nd A6	5 into	) CO-

Syntax	MOVF	*Rs+	, CRa	l, Rd											
Execution	If <b>Rd =</b> Repeat *Rs Rs adv	<b>0</b> 32 tir $\Rightarrow$ C + 32 vance r	nes Rd → Rs to ne egiste	s xt co er	proce	esso	r	lf <b>R</b> ej	d = 1 beat *Rs Rs adva	$  \rightarrow 3$ Rd til $\rightarrow C$ + 32 ance r	<b>31</b> mes Rd → R to ne	ls ext ci ter	oproc	cesso	or
Instruction Words	15 14 0 0 1 0 Defaul	13 0 0 t ID	12 0 0	11 0 1 0	10 1 1 0	9 1 0 0	8 0 1 0	7 1 0 0	6 1 0 0	5 1 0 0	4 R R	3	2 F CRd	1 Id Is	0
Operands	Rs TMS34020 source register (indirect postincrement) containing the a dress of the first 32-bit single-precision floating-point value to move the coprocessor														
	CRd Coprocessor destination register to hold the first 32-bit single-pre- sion floating-point value														
	Rd	TMS3 This v If If	4020 value Rd = Rd =	regis must 0, 1 →	ster c in th 31,	onta e rai	ining nge 0 then then	the n to 3 <i>32</i> 3 <i>Rd</i> 3	umbe 1 2-bit 32-bit	er of ( trans tran	32-bi sfers sfers	t trar are s are	nsfers made mad	storr e e	ake.
Description	MOVF m dress in Rs is incl register the cont	noves Rs inf remer seque ents c	32-bit to cop nted, a ence li of <i>Rd</i> .	t sing proce and C ist. T	le-pr ssor Rd is he ni	ecisi regi: adv umbe	on va sters ance er of 3	lues begir d to tł 32-bit	from nning ne ne : tran	merr at C xt reg sfers	nory t Rd. / gister mac	begir After r in th le is	nning each le cop deter	at the tran proce mine	e ad- isfer, issor ied by
Machine States	If Rd = 0 If Rd = 0 If Rd = 1 If Rd = 1	and and $\rightarrow 3^{\circ}$ $\rightarrow 3^{\circ}$	1 and 1 and	Rs Rs Rs Rs	s is a s is n s is a s is a	ligne onali ligne onali	d gned d gned		36 37 5 + ( 6 + (	Rd – Rd –	- 1) - 1)				
Instruction Type	CMOVM	IC, po	stincr	reme	nt, re	giste	er cou	unt							
Example	MOVF *A	.5+, 1	RВ7,	в7											
	This inst location 32-bit tra advance holds the	ructio pointe insfer d to t e num	n mov ed to b , regis he ne iber o	ves 3 by A5 ster A ext re f 32-	2-bit to co \5 is i giste bit tra	singl proc ncre r in ansfe	e-pre essoi ment the c ers to	cisio regi ed, a oproc be n	n valu sters and th cesso nade	ues fr begi le co or reg	rom T nning proce giste	TMS3 g at F esso r sec	34020 RA7. A r des juenc	0 mei After tinati ce lis	mory each on is t. B7

Syntax	MOV	= *F	?s+,	CRa	1 [, cc	ount]										
Execution	Repe *F F a	at <i>col</i> Rs → Rs + ( dvano	unt CR 32 – Ce to	time d > Rs o the	s next	copr	oces	sor r	egist	er						
Instruction Words	15 0 1	14 0 0	13 0 0	12 0 0	11 0 1	10 1 1	9 1 0	8 0 1	7 1 0	6 0 0	5 0 0	4 R	3 tra	2 ansfers Rs	1	0
	Det	ault ID	)	0	0	0	0	0	0	0	0			CRd		
Operands	<ul> <li>Rs TMS34020 source register (indirect postincrement) containing the address of the first 32-bit single-precision floating-point value to mov to the coprocessor</li> <li>CRd Coprocessor destination register to hold the first 32-bit single-precision floating-point value.</li> </ul>															the ove
	CRd Coprocessor destination register to hold the first 32-bit single-pred sion floating-point value															eci-
	count Contains the number of 32-bit transfers to make. This value must in the range 1 to 32; the default value is 1. <i>Count</i> determines the value of transfers:															the e of
			.lf c	coun	t = 32	2,		then	trans	sfers	= 0					
			lf c	count	t = 1	→ 31	,	then	trans	sfers	= CO	unt				
Description	MOVF addre the co advar numb	= mo ss in onter iced t er of	ves Rsir Its c to th 32-b	32-b nto co of Rs e ne oit tra	it sin oproc s are xt reg ansfe	gle-p cesso incr gister rs ma	recis r reg emei r in th ade is	ion v isters nted, ne co s dete	value s beg and proce ermir	s from innin the esson ned b	m me g at C copr r regi vy the	emor Rd. oces ster con	y beg After ssor sequ tents	ginnir each destir ence of <i>co</i>	ng at trans nation list. <i>unt</i> .	the sfer, n is The
Machine States	Rs Ali Rs No	gned onalig	ined		5 - 6 -	+ (coi + (coi	unt — unt —	1) 1)								
Instruction Type	CMO	/MC,	pos	tincr	eme	nt, cc	onsta	nt co	unt							
Example	MOVF	*A5+	, RI	B7,	4											
	This e memo After e tinatio list. Ce	exam ory loo each ( n is a ount :	ple catio 32-b idva spec	mov on po ittrar ncec cifies	es fo inted nsfer I to th that	our 3 I to by , regis ne ne: four	2-bit y A5 ster A xt reg 32-bi	sing to co \5 is i gister it trar	le-pr proce ncrer in th nsfers	ecisi essor nento e cop s are	on va regi: ed, ar proce mad	alues sters nd the ssor e.	s fror begi ecop regis	n TN nning roces ster se	IS34 at R sor c eque	020 B7. Ies- nce

,

Syntax	MOVF -*R	s, CRa	l [, cc	ount]											
Execution	Repeat <i>count</i> times $Rs - 32 \rightarrow Rs$ *Rs $\rightarrow CRd$ advance to the next coprocessor register														
Instruction Words	15 14 13	12	11	10	9	8	7	6	5	4	З	2	1	0	
	0 0 0	0	1	0	0	0	0	0	1.		t	ransfers	3		
	1 0 0 0 1 1 0 1 0 0 R Rs														
	Default ID         0         0         0         0         0         0         0         0         CRd													]	
Operands	<ul> <li>Rs TMS34020 source register (indirect postincrement) containing th address of the bit immediately after first 32-bit single-precision floating-point value to move to the coprocessor</li> <li>CRd Coprocessor destination register to hold the first 32-bit single-precision floating floating point value.</li> </ul>														
	sion floating-point value														
	count Cont in the of tra u I	ains th range nsfers f <i>count</i> f <i>count</i>	ie nu e 1 to :: t = 32 t = 1	mber 32; tł <u>2,</u> → 3 <sup>-</sup>	r of 3 ne de I,	2-bit fault then then	trans value trans trans	fers e is 1. sfers sfers	to ma . <i>Cou</i> = 0 = <i>co</i>	ake. <sup>-</sup> nt de unt	This etern	value ninest	mus he v	t be alue	
Description	MOVF move address (Rs- transfer, the cessor destin sequence list tents of <i>coun</i>	s 32-b - 32) ir conten ation is . The r	it sin nto c ts of s adv numb	igle-p oproo Rs a vance oer of	orecis cesso re de ed to 32-b	sion v or reg ecren the n it tra	value gister nente ext re nsfer	s fro s beg ed; af egiste s ma	m me ginnii ter e er in t de is	emor ng at ach t he cc dete	y be CRo rans oproo rmin	eginnir d. Befe sfer, th cesso aed by	ng at ore e le co r regi the o	the each pro- ister con-	
Machine States	Rs Aligned Rs Nonaligne	d	5 - 6 -	+ (co + (co	unt – unt –	· 1) · 1)									
Instruction Type	CMOVMC, p	edecr	emei	nt, co	onsta	nt co	unt								
Example	MOVF -*A5,	RB7,	4												
	This example memory local RB7. Before e fer, the copro cessor regist made.	e mov ion po each 32 cessoi er seq	es fo intec 2-bit r des uenc	our 3 I to by trans tinati ce lis	2-bit y (A5 fer, r on is t. <i>Co</i>	sing - 32 egiste adva unt s	gle-pr ) to c er A5 ancec specif	ecisi oproc is de l to tl ies t	on v cesso crem ne ne hat fo	alues or reg nente ext re our 3	s fro jiste d; af giste 2-bi	om TM rs beg fter ea er in th t trans	1S34 innir ch tra ie co sfers	020 Ig at ans- pro- are	

Syntax	MOV	FC	CRd,	*Rd+	+ [, ca	ount]										
Execution	Repeat <i>count</i> times $CRd \rightarrow *Rd$ $Rd + 32 \rightarrow Rd$ advance to the next coprocessor register															
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	1 0 0 0 0 1 1 0 1 0 1 R R 1 0 0 0 1 1 1 1 0 0 0 0 transfer														<u> </u>	
	Default ID         0         0         0         0         0         0         0         0         0         0         CRd															
Operands	CRd Coprocessor source register for the first 32-bit single-precision flo ing-point value to TMS34020 memory														loat-	
	Rd TMS34020 register (indirect postincrement) containing the addres for the first 32-bit single-precision floating-point value transferred															ress d
	count	count Contains the number of 32-bit transfers to make. This value must be in the range 1 to 32; the default value is 1. <i>Count</i> determines the value of transfers:														
	$\Box If count = 32,  then transfers = 0$															
			l lfo	coun	t = 1	<b>→</b> 31	,	then	trans	sfers	= <i>CO</i>	unt				
Description	MOVI begin transf nextr fers n	F ma ning ers, egist nade	oves at C Rd is ter in is de	the 3 Rd to incr the c etern	32-bit men emer oproo nined	: sing nory l nted, cesso l by tl	le-pr begir and or reg ne co	recisi nning the c gister onten	ion va g at th coproc r sequ nts of	alues e ado cesso ience <i>coun</i>	from fress or reg e. The t.	in R ister	oroce d. Af is ac nber	essor i ter ea dvance of 32-l	regis ch 32 ed to bit tra	ters 2-bit the ans-
Machine States	Rs Al Rs No	igne onali	d gned		5 - 6 -	+ (co + (co	unt – unt –	· 1) · 1)								
Instruction Type	CMO	VCN	l, pos	stincr	eme	nt, co	onsta	nt cc	ount							
Example	MOVF	RB7	, *A	5+,	4											
	This e isters 32-bit advar specit	exam beg tran nced fies t	ple n innin sfer, to th hat f	nove g at regis e ne> our 3	s foui RB7 ster A ct reg 2-bit	r 32-t to TN 5 is i ister trans	bit sir MS34 ncrei in the sfers	igle-p 1020 ment e cop are i	precis mem ted, a proces made	ion v iory f nd th ssor i	alue pointe e cop regist	s fror ed to proce er se	n cop by A essor eque	oroces A5. Af desti nce lis	ssor ter e natio st. <i>Co</i>	reg- ach on is ount

Syntax	MOV	Έr	CRd,	*Ra	d [, ca	ount]										
Execution	Repeat <i>count</i> times $Rd - 32 \rightarrow Rd$ $CRd \rightarrow *Rd$ advance to the next coprocessor register															
Instruction Words	_15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0 0 0 0 0 1 1 0 1 1 0 R														ld	
	1         0         0         1         1         1         0         0         0           Default ID         0												ti	CRd	rs	
Operands	CRd Coprocessor source register for the first 32-bit single-precision floring-point value to TMS34020 memory													loat-		
	Rd TMS34020 register (indirect predecrement) containing the address o the bit immediately following the 32-bits used to store the first 32-bits single-precision floating-point value transferred														ss of 2-bit	
	count Contains the number of 32-bit transfers to make. This value must in the range 1 to 32; the default value is 1. <i>Count</i> determines the value transfers:														n the ue of	
			lf	coun	t = 32	2,		then	tran	sfers	= 0					
			If	coun	t = 1	→ 3 <sup>-</sup>	1,	then	tran	sfers	= CO	unt				
Description	MOV begir 32-bi ter is numb	F mo ining t trar adva ber o	oves J at C Isfer, anceo f 32-	the 3 Rd to Rd is d to th bit tra	32-bit o mer s dec ne ne ansfe	t sing nory reme xt re rs m	gle-pi begii entec giste ade i	recisi nning I; afte r in th s det	on va at th r eac e cop ermin	alues e ade h tra proce ned b	fron dress nsfer essor by the	n cop s (Rd r, the regis e con	oroce – 32 copr ster s tents	essor 2). Be roces seque s of <i>c</i>	regis fore e sor re ence. <i>ount</i> .	sters each egis- The
Machine States	Rs A Rs N	ligne onali	d ignec	ţ	5 · 6 ·	+ (co + (co	unt unt	- 1) - 1)					·			
Instruction Type	СМО	VCN	1, pre	edecr	eme	nt, co	onsta	nt co	unt							
Example	MOVF	RB7	,*	Ά5,	4											
	MOVF RB7, -*A5, 4 This example moves four 32-bit single-precision values from coprocessor reg- isters beginning at RB7 to TMS34020 memory pointed to by (A5 – 32). Before each 32-bit transfer, register A5 is decremented; after each 32-bit transfer, the coprocessor destination is advanced to the next register in the coprocessor register sequence list. <i>Count</i> specifies that four 32-bit transfers are made													reg- efore ; the ssor e.		

Syntax	мον	/F (	CRd,	Rd												
Execution	CRd	-→ R	d													
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2 ·	1	0
	0	0	0	0	0	1	1	0	0	1	1	R		R	d	
	0	1	0	0	1	1	1	1	0	0	0	0.	0	0	0	.0
	De	efault	D	0	0	0	0	0	0	0	0			CRd		
Operands	CRd Rd	C ir T	opro Ig-po MS3	cess int va 4020	or so alue desi	ource	rég on re	ister giste	for r	the :	32-bi	t sin	gle-p	precis	ion	float-
Description	MOV	/F mo	oves	the c	onte	nts (s	single	e-pre	cisio	n val	ue) o	f CR	d to	Rd.		
Machine States	5 if th 4 if th	ne fir: ne fir:	st ins st ins	truct truct	ion w ion w	vord i vord i	s Ion s not	g wo : long	ord-al g wor	ligneo d-alio	d gned					
Instruction Type	СМС	VGC	c, one	e reg	ister											
Example	MOVF	RA7	, A5													
	This ter R	exan A7 to	ple r TM	nove S340	s the 20 re	32-b giste	it sin er A5	gle-p	orecis	sion v	alue	from	cop	roces	sorr	egis-

# MOVF Move, Single Precision, Coprocessor to Coprocessor

Syntax	MOVF CRs <sub>1</sub> , CRd														
Execution	$CRs_1 \rightarrow CRd$														
Instruction Words															
	Image: Image in the second s														
Operands	CRs <sub>1</sub> Coprocessor source register A that holds the 32-bit single-precision floating-point operand														
	CRd Coprocessor destination register														
Description	MOVF moves the contents (single-precision value) of $\mbox{CRs}_1$ (register A) to CRd.														
Machine States	2														
Instruction Type	CEXEC, short														
Example	MOVF RA7, RB4														
	This example moves the 32-bit single-precision value from coprocessor register RA7 to coprocessor register RB4.														

Syntax	MOVF CRs <sub>2</sub> , CRd														
Execution	$CRs_2 \rightarrow CRd$														
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0														
	1 1 0 1 1 0 0 0 0 1 1 1 0 1 0														
	Default ID         CRs2         0         0         1         CRd														
Operands	CRs <sub>2</sub> Coprocessor source register B that holds the 32-bit single-precisior floating-point value														
	CRd Coprocessor destination register														
Description	MOVF moves 32-bit single-precision value from CRs <sub>2</sub> (register B) to CRd.														
Machine States	2														
Instruction Type	CEXEC, short														
Example	MOVF RB3, RB4														
	This example moves the 32-bit single-precision value from coprocessor register RB3 to coprocessor register RB4.														

# MPYD Multiply, Double Precision

Syntax	MPYD CRs <sub>1</sub> , CRs <sub>2</sub> , CRd														
Execution	$CRs_1 \times CRs_2 \rightarrow CRd$														
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         1       1       0       1       1       0       0       0       0       1       0       0       1														
Operands	CRs <sub>1</sub> Coprocessor register containing the first 64-bit double-precision float ting-point operand														
	CRs <sub>2</sub> Coprocessor register containing the second 64-bit double-precision floating-point operand														
	CRd Coprocessor destination register														
Description	MPYS multiplies the contents (double-precision value) of CRs $_1$ by the contents of CRs $_2$ and stores the result in CRd.														
Machine States	2														
Instruction Type	CEXEC, short														
Example	MPYD RA5, RB6, RA7														
	This example multiplies the contents of RA5 by RB6 and stores the result in RA7.														

Syntax	<b>MPYF</b> CRs <sub>1</sub> , CRs <sub>2</sub> , CRd															
Execution	$CRs_1 \times CRs_2 \rightarrow CRd$															
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
		1 fouilt l	0	1	1	0	0	0			1	0	0		1	0
Operands	CRs <sub>1</sub>	CRs <sub>1</sub> Coprocessor register containing the first 32-bit single-precision floa- ting-point operand														floa-
	CRs <sub>2</sub> Coprocessor register containing the second 32-bit single-precision floating-point operand														ision	
	CRd	С	opro	cess	or de	stina	tion	regis	ter							
Description	MPYI of CF	= mu Is <sub>2</sub> a	ltiplie nd s	es the tores	con the	tents resuli	(sing t in C	gle-pi Rd.	recis	ion v	alue)	of Cl	Rs <sub>1</sub> b	by the	cont	ents
Machine States	2															
Instruction Type	CEXE	EC, s	hort													
Example	MPYF	RA5	, RB	6, R	A7											
	This example multiplies the contents of RA5 by RB6 and stores the result in RA7.														ult in	

Syntax	MPYF	Rs <sub>1</sub> ,	Rs <sub>2</sub> ,	CRs <sub>1</sub>	, CR	ls <sub>2</sub> , (	CRd								
Execution	$Rs_{1} \rightarrow CRs_{1}$ $Rs_{2} \rightarrow CRs_{2}$ $CRs_{1} \times CRs_{2} \rightarrow CRd$														
Instruction Words	15 14	1 13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0 0 0 0 0 1 1 0 0 1 0 R Rs														
	0 1 0 1 0 0 0 1 0 0 R Rs														
	Default ID CRs <sub>1</sub> CRs <sub>2</sub> CRd														
Operands	Rs <sub>1</sub> TMS34020 source register for the first 32-bit single-precision float ing-point value to coprocessor														loat-
	Rs <sub>2</sub>	Rs <sub>2</sub> TMS34020 source register for the second 32-bit single-precision float ing-point value to coprocessor													
	CRs <sub>1</sub>	CRs <sub>1</sub> Coprocessor register to contain the first 32-bit single-precision floa ting-point operand													
	CRs <sub>2</sub>	Copro floatin	cess g-poi	or re int op	giste eran	r to d	conta	in th	e se	cond	32-ł	oit si	ngle-	prec	ision
	CRd	Copro	cess	or de	stina	tion	regist	ter							
Description	MPYS I and CR	oads ti s <sub>2</sub> resp	he co pectiv	nten <sup>.</sup> vely, r	ts (sii nultip	ngle- olies	preci CRs <sub>1</sub>	sion × Cł	value Rs <sub>2</sub> , a	e) of and s	Rs <sub>1</sub> a store	and I s the	Rs <sub>2</sub> ii resu	nto C It in (	CRs <sub>1</sub> CRd.
Machine States	4 if the 3 if the	first ins first ins	struct struct	ion w ion w	/ord i /ord i	s Ion s not	g wo Iong	rd-ali word	gned d-alig	ned					
Instruction Type	CMOV	GC, two	o regi	isters	6										
Example	MPYF A	5, A6,	, RA5	, RE	36, F	RA7									
	This exa RA5 an the resu	ample I d RB6 ult in R	oads respe A7.	TMS ective	3402 ely, m	20 reg ultipl	jister ies th	s A5 ne coi	and A ntent	\6 int s of F	o cop RA5 I	oroce by Ri	essor 36, ai	regi: nd st	sters ores

Ì.

Syntax MPYS	CRs <sub>1</sub> , CRs <sub>2</sub> , CRd
-------------	---

\*\*\*\*\*

 $\textit{Execution} \qquad \qquad CRs_1 \times CRs_2 \rightarrow CRd$ 

Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0			
	1	1	0	1	1	0	0	0	0	0	1	0	0	0	0	0			
	De	efault	ID		CF	≀s <sub>1</sub>			CF	ls <sub>2</sub>		CRd							
Operands	CRs	1 C	opro	cess	or re	giste	r con	tainiı	ng th	e firs	t 32-l	bit in	teger	ope	rand				
	CRs	2 C	opro	cess	or re	giste	r con	tainii	ng th	e sec	cond	32-b	it inte	eger	opera	and			
	CRd Coprocessor destination register																		
Description	MPY store	່ S mເ es the	ultipli e resu	es th ult in	e cor CRd	ntent:	s (inte	eger)	) of C	Rs <sub>1</sub>	by th	e coi	ntent	s of (	CRs <sub>2</sub>	and			
Machine States	2																		
Instruction Type	CEX	EC, s	short																
Example	MPYS	RA5	, RE	6, F	8В7														
	This RB7.	exan	nple	multi	plies	the o	conte	nts c	of RA	5 by	RB6	and	store	es the	e resi	ult in			
Syntax	MPYS	Rs <sub>1</sub>	Rs <sub>2</sub> ,	CRs <sub>1</sub>	, CR	<sup>1</sup> s <sub>2</sub> , (	CRd												
-------------------	---	----------------------	--------------------	-----------------------------	------------------	---------------------------------	------------------	------------------------------	----------------------------	----------------	--------------------------	----------------	---------------	-----------------	-----------------				
Execution	$Rs_1 \rightarrow CRs_1$ $Rs_2 \rightarrow CRs_2$ $CRs_1 \times CRs_2 \rightarrow CRd$																		
Instruction Words	15 1	4 13	12	11	10	9	8	7	6	5	4	3	2	1	0				
	0	0 0	0	0	1	1	0	0	1	0	R		R	S <sub>1</sub>					
	0	1 0	1	0	0	0	0	0	0	0	R								
	Defa	ult ID		CF	IS <sub>1</sub>			CF	ls <sub>2</sub>			·····	CRd						
Operands	Rs <sub>1</sub> TMS34020 source register for the first 32-bit integer value to copro- cessor																		
	Rs <sub>2</sub> TMS34020 source register for the second 32-bit integer value to coprocessor																		
	CRs <sub>1</sub> Coprocessor register to contain the first 32-bit integer operand																		
	CRs <sub>2</sub> Coprocessor register to contain the second 32-bit integer operand														nd				
	CRd	Сор	ocess	or de	stina	tion i	egist	ter											
Description	MPYS tively, r	loads nultipl	he cor ies CF	ntents Rs <sub>1</sub> ×	s (inte CRs	eger) <sub>2</sub> , an	of Rs d sto	s <sub>1</sub> and ores t	d Rs <sub>2</sub> he re	into sult i	CRs <sub>1</sub> n CF	and d.	CRs	2 res	pec-				
Machine States	4 if the 3 if the	first ir first ir	nstruct Istruct	ion w ion w	ord is ord is	s Ion s not	g wo long	rd-ali word	gned d-alig	l Ined									
Instruction Type	CMOV	GC, tv	vo reg	isters															
Example	MPYS A	.5, AG	, RA5	, RB	6, R	.в7													
	This ex RA5 ar RB7.	ample Id RB	loads 6, mult	TMS iplies	3402 the c	0 reg conte	jister ents c	s A5 of RA	and A 5 by	\6 int RB6,	o cop and	oroce store	ssor s the	regis e resi	aters ult in				

Syntax	NEG CRs, CRd														
Execution	$-CRs \rightarrow CRd$														
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         1       1       0       1       1       0       0       0       0       1       1       1       1       0       0         Default ID       CRs       0       0       1       1       1       CRd       CRd														
Operands	CRsCoprocessor source register containing the 32-bit integer operandCRdCoprocessor destination register														
Description	NEG takes the 2s complement of the contents (integer) of CRs and stores the result in CRd.														
	The source register, CRs, must be in the A coprocessor register file.														
Machine States	2														
Instruction Type	CEXEC, short														
Example	NEG RA5, RB7														
	This example takes the 2s complement of the contents of RA5 and stores the result in RB7.														

Syntax	NEG	i R	s, Cł	Rs, C	Rd																
Execution	$Rs \rightarrow CRs$ CRs $\rightarrow CRd$																				
Instruction Words	_15	14	4 13 12 11 10 9 8 7 6 5								4	3	2	1	0						
	0	0	0	0	0	1	1	0	0	0	1	R		F	s						
	0	1	0	1	. 1	1	1	0	0	0	0	0	0	0	0	0					
	De	efault	ID		CI	Rs		0	0	1	1			CRd							
Operands	Rs TMS34020 source register for the 32-bit integer value to coprocessor													essor							
	CRs Coprocessor register to contain the 32-bit integer operand																				
	CRd	С	opro	cess	or de	estina	ition	regis	ter												
Description	NEG loads the contents (integer) of Rs into CRs, takes the 2s complement of the contents of CRs, and stores the result in CRd.													ent of							
	The	sour	ce reg	giste	r, CR	ls, mi	ust b	e in t	he A	copr	oces	sor r	egist	ter fil	ə.						
Machine States	4 if th 3 if th	ne fir ne fir	st ins st ins	struct struct	ion w ion w	vord i vord i	s lon s noi	ig wo t long	ord-al g wor	igne d-ali	d gned										
Instruction Type	СМС	VGC	C, one	e reg	ister																
Example	NEG	А5,	RA5,	RB7	,																
	This takes	exar s the	nple 2s co	load	s TM emer	IS340 It of th	)20 i ne co	regist onten	ter A ts of	5 inte RA5,	o cop and	oroce store	essor es the	· regi e resi	ster ılt in	RA5 <sub>,</sub> RB7.					

Syntax	NEGD CRs, CRd														
Execution	$-CRs \rightarrow CRd$														
Instruction Words	15 14 1 1	13 0	12 1	<u>11</u>	10 0	9 0	8 0	7	6 0	5. 1	4	3 1	2	1	0
	Defau	it ID		CF	ls		0	0	1	1			CRd		
Operands	CRs Coprocessor register containing the 64-bit double-precision floa- ting-point operand													floa-	
	CRd Coprocessor destination register														
Description	NEGD negates the contents (double-precision value) of register CRs and stores the result in CRd.														and
	The sou	rce re	gister	, CR	s, mı	ust b	e in t	he A	copr	oces	sor r	egist	er file	Э.	
Machine States	2														
Instruction Type	CEXEC	, short													
Example	NEGD RA	5, RE	57												
	This exa	mple	negat	tes th	ie co	nten	ts of	RA5	and	store	es the	e resi	ult in	RB7	•

#### NEGF Negate, Single Precision

Syntax	NEGF CRs, CRd														
Execution	-CRs → CRd														
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0														
	1     1     0     1     1     1     1     1     1     1     1     1     1     0       Default ID     CRs     0     0     1     1     1     CRd														
Operands	CRs Coprocessor register containing the 32-bit single-precision floa- ting-point operand														
	CRd Coprocessor destination register														
Description	NEGF negates the contents (single-precision value) of CRs and stores the re- sult in CRd.														
	The source register, CRs, must be in the A coprocessor register file.														
Machine States	2														
Instruction Type	CEXEC, short														
Example	NEGF RA5, RA7														
	This example negates the contents of RA5 and stores the result in RA7.														

Syntax N	IEGF	Rs, (	CRs, I	CRd
----------	------	-------	--------	-----

Execution $Rs \rightarrow CRs$  $-CRs \rightarrow CRd$ 

Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0		
	0	0	0	0	0	1	1	0	0	0	1	R		R	s			
	0	1	0	1	1	1	1	1	0	0	0	0	0	0	0	0		
	De	Default ID CRs 0 0 1 1												CRd				
Operands	Rs	T in	MS3 Ig-po	4020 int va	sou alue t	irce to co	regis proce	ster essor	for t	he 3	2-bit	sing	gle-p	recisi	on	float-		
	CRs Coprocessor register to contain the 32-bit single-precision floating-point operand															cision		
	CRd Coprocessor destination register																	
Description	NEGF loads the contents (single-precision value) of Rs into CRs, negates the contents of CRs, and stores the result in CRd.															es the		
	The source register, CRs, must be in the A coprocessor register file.																	
Machine States	3 if th 2 if th	ne firs ne firs	st ins st ins	truct truct	ion w ion w	/ord i /ord i	s Ion s not	g wo : long	rd-al 1 wor	igneo d-alio	d gned							
Instruction Type	СМС	VGC	c, one	e reg	ister													
Example	NEGF	A5,	RA5	, RB	7													
	This example loads TMS34020 register A5 into coprocessor register RA5, ne- gates the contents of RA5, and stores the result in RB7.														ō, ne-			

# NOT Not, Integer, 1s Complement

Syntax	NOT CRs, CRd														
Execution	NOT CRs $\rightarrow$ CRd														
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0														
	1 1 0 1 1 0 0 0 0 1 1 1 1 0 0														
	Default ID CRs 0 0 0 1 CRd														
Operands	CRs Coprocessor source register containing the 32-bit integer operand CRd Coprocessor destination register														
Description	NOT takes the 1s complement of the contents (integer) of CRs and stores the result in CRd.														
	The source register, CRs, must be in the A coprocessor register file.														
Machine States	2														
Instruction Type	CEXEC, short														
Example	NOT RA5, RA7														
	This example takes the 1s complement of the contents of RA5 and stores the result in RA7.														

Syntax NOT RS, CRS, CH
------------------------

Execution $Rs \rightarrow CRs$ NOT CRs  $\rightarrow$  CRd

Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0		
	0	0	0	0	0	1	1	0	0	0	1	R		Rs				
	0	1	0	1	1	1	1	0	0	0	0	0	0	0	0	0		
	D	efault	ID	CRs 0 0 1										CRd				
Operands	Rs	Т	MS3	4020	sou	rcere	egiste	er for	the 3	2-bit	integ	jer va	alue t	o cop	proce	ssor		
	CRs	С	opro	cess	or re	giste	r to c	onta	in the	e 32-	bit inf	teger	ope	rand				
	CRd Coprocessor destination register																	
Description	NOT loads the contents (integer) of Rs into the CRs, takes the 1s complement of the contents of register CRs, and stores the result in CRd.															nent		
	The source register, CRs, must be in the A coprocessor register file.																	
Machine States	3 if tl 2 if tl	he fir he fir	st ins st ins	struct struct	ion v ion v	vord i vord i	s lon s not	g wo : long	ord-al g wor	igneo d-alio	d gned							
Instruction Type	СМС	VGC	C, on	e reg	ister					,								
Example	NOT	А5,	RA5,	RAT	,													
	This example loads TMS34020 register A5 into coprocessor register RA5 takes the 1s complement of the contents of RA5, and stores the result in RA7														7A5, 7A7.			

# SQR Square, Integer

Syntax	SQR CRs, CRd														
Execution	$CRs \times CRs \rightarrow CRd$														
Instruction Words	<u>15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0</u>														
	1 1 0 1 1 0 0 0 0 0 1 1 1 1 0 0														
	Default ID CRs 1 0 0 0 CRd														
Operands	CRs Coprocessor source register containing the 32-bit integer operand														
	CRd Coprocessor destination register														
Description	SQR squares the contents (integer) of CRs and stores the result in CRd.														
	The source register, CRs, must be in the A coprocessor register file.														
Machine States	2														
Instruction Type	CEXEC, short														
Example	SQR RA5, RA7														
	This example squares the contents of RA5 and stores the result in register RA7.	ər													

Syntax	SQR	R	s, Cł	7 <i>s</i> , C	Rd											
Execution	Rs – CRs	→ CR × CF	ls ₹s →	CR	ł											
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3											2	1	0		
	0	0	0	0	0	1	1	0	0	0	1	R		F	ls	
	0	1	0	1	1	1	1	0	0	0	0	0	0	0	0	0
	D	efault	ID		С	Rs		1	0	0	0			CRd		
Operands	Rs TMS34020 source register for the 32-bit integer value to coprocesso													essor		
CRs Coprocessor register to contain the 32-bit integer operand																
	CRd	C	opro	cess	or de	estina	ation	regis	ter							
Description	SQR loads the contents (integer) of Rs into CRs, squares the contents of CR and stores the result in CRd.													CRs,		
	The	sour	ce re	giste	r, CF	ls, m	ust b	e in t	he A	сорі	oces	sor r	egist	ter fil	e.	
Machine States	3 if tl 2 if tl	he fir he fir	st ins st ins	struct struct	ion v ion v	vord i vord i	is Ion is no	ig wo t long	ord-al g wor	ligne d-ali	d gned					
Instruction Type	СМС	VG	C, on	e reg	lister											
Example	SQR	А5,	RA5,	RB	7											
	This squa	exar ires t	nple he co	load onter	s TN its of	1S340 RA5	020 i , and	regist I stor	ter A es th	5 int	o cop sult ir	oroce n RB3	essor 7.	regi	ster	RA5 <sub>,</sub>

#### SQRD Square, Double Precision

Syntax	SQR	D	CRs,	CRd												
Execution	CRs	× CF	ls →	CRd	l											
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	1	1	0	1	1	0	0	0	0	0	1	1	1	1	1	1
	Default ID   CRs   1   0   0   0   CRd															
Operands	CRs Coprocessor register containing the 64-bit double-precision floating-point operand															floa-
	CRd Coprocessor destination register															
Description	SQRD squares the contents (double-precision value) of CRs and stores the result in CRd.														s the	
	The s	sourc	ce re	gister	, CR	s, mı	ust b	e in t	the A	copr	oces	sor r	egist	er file	э.	
Machine States	2															
Instruction Type	CEX	EC, s	short													
Example	SQRD	RA5	, RA	.7												
	This	exan	nple	squa	res tł	ne co	nten	ts of	RA5	and	store	es the	e res	ult in	RA7	•

Syntax	SQRF	CRs,	CRd												
Execution	CRs × C	Rs →	CRd												
Instruction Words	15 14 1 1	13 0	12 1	11 1	10 0	9 0	8	7	6 0	5 1	4	3 1	2 1	1	0
Operands	Default ID       CRs       1       0       0       0       CRd         CRs       Coprocessor source register containing the 32-bit single-precision floating-point operand       CRd       CRd														
	CRd Coprocessor destination register														
Description	SQRF squares the contents (single-precision value) of CRs and stores the re- sult in CRd.														e re-
	The sour	ce reç	gister	, CR	s, mı	ust b	e in t	he A	copr	oces	sor r	egist	er file	e.	
Machine States	2														
Instruction Type	CEXEC,	short													
Example	SQRF RAS	5, RB	7												
	This exar	nple s	squar	res th	ne co	nten	ts of	RA5	and	store	s the	e res	ult in	RB7.	

Syntax	SQRF Rs	s, CRs, C	CRd												
Execution	Rs → CRs CRs × CRs	s → CRd	I												
Instruction Words	15 14	13 12	11	10	9	8	7	6	5	4	3	2	1	0	
	0 0	0 0	0	1	1	0	0	0	1	R		R	s		
	0 1	0 1.	1	1	1	1	0	0	0	0	0	· · 0	0	0	
	Default ID     CRs     1     0     0     0     CRd														
Operands	Rs TMS34020 source register for the 32-bit single-precision flor ing-point value to coprocessor														
	ing-point value to coprocessor CRs Coprocessor register to contain the 32-bit single-precision flo ing-point operand														
	CRd Co	process	or des	stina	tion	regis	ter								
Description	SQRF loa (single-pree	ids the cision va	cont lue) c	tents of CF	s of Rs, a	Rs nd si	into tores	CI the r	Rs, result	squa : in C	res Rd.	the	cor	itents	
	The source	e register	, CRs	, ու	ist b	e in t	he A	copr	oces	sor r	egist	er file	ə.		
Machine States	3 if the first 2 if the first	instruct	ion wo ion wo	ord is	s lon s not	g wa : long	ord-al g wor	igneo d-alio	d gned						
Instruction Type	CMOVGC,	one reg	ister												
Example	SQRF A5,	RA5, RB	7												
	This examp squares the	ple loads e conten	s TMS ts of F	3340 7A5,	20 r and	egis stor	ter A	5 inte e res	o cop sult in	oroce RB7	essor 7.	regi	ster	RA5,	

Syntax	SQRT CRs, CRd														
Execution	$\sqrt{CRs} \rightarrow CRd$														
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0														
	1 1 0 1 1 0 0 0 0 1 1 1 0 0														
	Default ID     CRs     1     0     0     1     CRd														
Operands	CRs Coprocessor register containing the 32-bit integer operand														
	CRs Coprocessor register containing the 32-bit integer operand CRd Coprocessor destination register														
Description	SQRT takes the square root of the contents (integer) of CRs and stores th result in CRd.														
	The source register, CRs, must be in the A coprocessor register file.														
Machine States	2														
Instruction Type	CEXEC, short														
Example	SQRT RA5, RB7														
	This example takes the square root of the contents of RA5 and stores the result in RB7.														

#### SQRT Load and Square Root, Integer

Syntax	SQRT	Rs, C	Rs, (	CRd											
Execution	Rs → C √CRs –	Rs • CRd													
Instruction Words	15 14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0 0	0	0	0	1	1	0	0	0	1	R		R	s	
	0 1	0	1	1	1	1	0	0	0	0	0	0	0	0	0
	Default ID CRs 1 0 0 1 CRd														
Operands	Rs TMS34020 source register for the 32-bit integer value to coprocess														
	RsI MS34020 source register for the 32-bit integer value to coprocessCRsCoprocessor register to contain the 32-bit integer operand														
	<ul><li>CRs Coprocessor register to contain the 32-bit integer operand</li><li>CRd Coprocessor destination register</li></ul>														
Description	SQRT lo contents	ads th of CF	ne co Rs, a	ntent nd st	s (inte ores	eger) the r	) of R esult	s into in C	o CRs Rd.	s, tak	es th	e sqi	uare r	ooto	fthe
	The sou	ce re	giste	r, CR	s, mı	ust b	e in t	he A	copr	oces	sor r	egist	er file	Э.	
Machine States	3 if the fi 2 if the fi	rst ins rst ins	struct struct	ion w ion w	/ord i /ord i	s Ion s not	ig wo t long	ord-al g wor	igneo d-alio	d gned					
Instruction Type	CMOVG	C, on	e reg	lister											
Example	SQRT A5	, RAS	5, R#	47											
	This exa takes the	mple squa	load are ro	s TM oot of	IS340 the c	020 r conte	regist ents c	ter A of RA	5 into 5, ar	o cop nd sto	oroce ores t	essor the re	regis esult i	ster I in RA	RA5 <sub>,</sub> \7.

Syntax	SQR	TD	CRs	, CR	d											
Execution	√CR	_ s →	CRd													
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	1	1	0	1	1	0	0	0	0	0	1	1	1	1	1	1
	De	fault	D		CF	₹s		1	0	0	1			CRd		
Operands	CRs Coprocessor register containing the 64-bit double-precision flo ting-point operand															floa-
	CRd Coprocessor destination register															
Description	SQRTD takes the square root of the contents (double-precision value) of CRs and stores the result in CRd.															CRs
	The s	sourc	e reg	gister	, CR	s, m	ust b	e in t	he A	copr	oces	sor	egist	ter file	е.	
Machine States	2															
Instruction Type	CEXE	EC, s	short													
Example	SQRTI	d RA	5, R	A7												
	This e in RA	exam 7.	iple ta	akes	the s	quar	e roo	t of th	ne co	ntent	sofF	RA5 a	and s	tores	ther	esult

# SQRTF Square Root, Single Precision

Syntax	SQRTF CRs, CRd
Execution	$\sqrt{CRs} \rightarrow CRd$
Instruction Words	15 14 13 12 11 10 9 8 7 6 5 4 3 2 1 0
	1 1 0 1 1 0 0 0 0 0 1 1 1 1 1 0
	Default ID CRs 1 0 0 1 CRd
Operands	CRs Coprocessor register containing the 32-bit single-precision floating-point operand
	CRd Coprocessor destination register
Description	SQRTF takes the square root of the contents (single-precision value) of CRs and stores the result in CRd.
	The source register, CRs, must be in the A coprocessor register file.
Machine States	2
Instruction Type	CEXEC, short
Example	SQRTF RA5, RA7
	This example takes the square root of the contents of RA5 and stores the result in RA7.

.

Svntax	SQRTF	Rs.	CRs. CRo	1
- <b>J</b> · · · · · ·				

**Execution**  $Rs \rightarrow CRs$ 

 $\sqrt{CRs} \rightarrow CRd$ 

Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	0	0	1	R		F	ls	
	0	1	0	1	1	1	1	1	0	0.	0	0	0	0	0	0
	De	efault I	D		CF	Rs		1	0	0	1			CRd		
Operands	Rs	T in	MS3 g-po	4020 int va	sou alue 1	irce to co	regis proce	ster essor	for t	he 3	2-bit	sinę	gle-p	recis	ion	float-
	CRs Coprocessor register to contain the 32-bit single-precis floating-point operand CRd Coprocessor destination register															cision
	floating-point operand CRd Coprocessor destination register															
Description	CRd Coprocessor destination register SQRTF loads the contents (single-precision value) of Rs into CRs, takes the square root of the contents of CRs, and stores the result in CRd.															es the
	The s	sourc	e reç	gister	, CR	s, mi	ust be	e in t	he A	copr	oces	sor r	egist	er fil	e.	
Machine States	3 if th 2 if th	ne firs ne firs	st ins st ins	truct truct	ion w ion w	/ord i /ord i	s lon s not	g wo : long	rd-al I wor	igneo d-alio	d gned					
Instruction Type	СМС	VGC	, one	e reg	ister											
Example	SQRT	F A5	, RA	5, R	A7											
	This takes	exan the	nple squa	load: re ro	s TM ot of	S340 the o	020 r conte	egist nts c	er A	5 into 5, an	o cop id sto	broce bres t	ssor he re	regi esult	ster in R	RA5, A7.

#### SUB Subtract, Integer, (A Register – B Register)

Syntax	SUB	CRs	i <sub>1</sub> , Cl	₽ <i>s</i> 2,	CRd											
Execution	CRs <sub>1</sub>	- C	Rs <sub>2</sub>	→ C	Rd											
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
		1	0	1	1	0	0	0	0	0	0	0	0	1	0	0
	CBs <sub>1</sub> Coprocessor A register containing the 32-bit minuend integer opera															
Operands	CRs <sub>1</sub> Coprocessor A register containing the 32-bit minuend integer operand															rand
	<ul> <li>CRs<sub>1</sub> Coprocessor A register containing the 32-bit minuend integer operan</li> <li>CRs<sub>2</sub> Coprocessor B register containing the 32-bit subtrahend integer oper and</li> </ul>															per-
	CRd	С	opro	cess	or de	stina	tion	regis	ter							
Description	SUB in CF	subt Id.	racts	the c	conte	nts (i	nteg	er) of	fCR	s <sub>2</sub> fro	m CF	Rs <sub>1</sub> a	nd st	ores	the r	esuit
Machine States	2															
Instruction Type	CEXE	EC, s	short													
Example	SUB 1	RA5,	RB3	, RA	7											
	This e RA7.	exan	nple s	subtra	acts	he co	onter	nts of	RB3	3 from	n RAS	5 and	stor	es th	e res	ult in

Syntax	SUB	Rs <sub>1</sub> ,	Rs <sub>2</sub>	, CR	s <sub>1</sub> , C	Rs <sub>2</sub> ,	CRa	1								
Execution	Rs <sub>1</sub> - Rs <sub>2</sub> - CRs <sub>1</sub>	→ Cl → Cl – C	Rs <sub>1</sub> Rs <sub>2</sub> Rs <sub>2</sub>	→ Cl	٦d											
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	0	1	0	R		R	S <sub>1</sub>	
	0	1	0	0	0	0	1	0	0	0	0	R		R	S <sub>2</sub>	
	De	fault I	D		CF	IS <sub>1</sub>			CR	IS <sub>2</sub>				CRd		
Operands	<ul> <li>Rs<sub>1</sub> TMS34020 source register for the first (minuend) 32-bit integer value to coprocessor</li> <li>Rs<sub>2</sub> TMS34020 source register for the second (subtrahend) 32-bit integer</li> </ul>															alue
	Rs <sub>2</sub> TMS34020 source register for the second (subtrahend) 32-bit integralue to coprocessor															eger
	value to coprocessor CRs <sub>1</sub> Coprocessor A register to contain the 32-bit minuend integer operar															rand
	CRs <sub>2</sub>	C ol	opro oerar	cesso nd	or B	regi	ster	to co	ontair	n the	32-	bit s	ubtra	aheno	d int	eger
	CRd	С	opro	cesso	or de	stina	tion I	regist	er							
Description	SUB tively,	load: sub	s the tracts	conte s the	ents ( conte	(integ ents c	jer) o of CR	of Rs <sub>1</sub> Is <sub>2</sub> fro	and om C	Rs <sub>2</sub> i Rs <sub>1</sub> ,	into ( and s	CRs <sub>1</sub> store:	and s the	CRs resu	<sub>2</sub> res It in (	pec- CRd.
Machine States	4 if th 3 if th	e firs e firs	st ins st ins	tructi tructi	on w on w	ord is ord is	s Ion s not	g woi long	rd-ali word	gned 1-alig	ned					
Instruction Type	СМО	VGC	, twc	o regi	sters											
Example	SUB A	A0, 3	вб,	RA5,	RB3	, RA	.7									
	This e RA5 a in RA	exam and F 7.	ple k RB3,	bads subti	TMS racts	3402 the c	0 reg conte	ister: nts o	s A0 a f RB	and E 3 fror	86 int n RA	o cop .5, ar	oroce Id sta	essor ores l	regis he re	sters esult

Syntax	SUB	CRs	<sub>2</sub> , Cl	7s <sub>1</sub> ,	CRd											
Execution	CRs <sub>2</sub>	<u>-</u> C	Rs <sub>1</sub>	→ C	Rd											
Instruction Words	15	14	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	1	1	0	1	1	. 0	0	0	0	0	0	0	1	1	0	0
	De	fault I	D		CF	ls <sub>1</sub>			CF	₹s₂				CRd		
Operands	CRs <sub>1</sub> Coprocessor A register containing the 32-bit subtrahend integer oper- and															oper-
	CRs <sub>2</sub> Coprocessor B register containing the 32-bit minuend integer operan															rand
	CRd Coprocessor destination register															
Description	SUB in CF	subti Id.	racts	the c	conte	nts (i	nteg	er) of	fCR	s <sub>1</sub> fro	m CF	Rs <sub>2</sub> a	nd st	ores	the r	esult
Machine States	2															
Instruction Type	CEX	EC, s	short													
Example	SUB 1	RB5,	RA3	, RA	17											
	This e RA7.	exan	ple s	subtr	acts	the co	onter	nts of	f RA3	8 from	n RB	5 and	stor	es th	e res	ult in

Syntax	SUB R	<b>UB</b> Rs <sub>2</sub> , Rs <sub>1</sub> , CRs <sub>2</sub> , CRs <sub>1</sub> , CRd													
Execution	Rs <sub>1</sub> → Rs <sub>2</sub> → CRs <sub>2</sub> –	$s_1 \rightarrow CRs_1$ $s_2 \rightarrow CRs_2$ $Rs_2 - CRs_1 \rightarrow CRd$													
Instruction Words	15 1	4 13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0 0	0	0	0	1	1	0	0	1	0	R		R	<sup>3</sup> 1	
	0 1	0	0	0	1	1	0	0	0	0	R		R	<sup>5</sup> 2	
	Defai	ilt ID		CF	ls <sub>1</sub>			CF	s <sub>2</sub>				CRd		
Operands	Rs <sub>1</sub>	s1 TMS34020 source register for the first (subtrahend) 32-bit integer value to coprocessor													
	Rs <sub>2</sub>	TMS3 value	4020 to co	sou proce	rce re essor	egist	er for	the	seco	nd (r	ninu	end)	32-b	it int	eger
	CRs <sub>1</sub>	Copro opera	cess nd	or A	regi	ster	to co	ontair	n the	932-	bit s	ubtra	ahen	d int	eger
	CRs <sub>2</sub>	Copro	cess	or B r	egist	er to	conta	ain th	e 32-	bit m	inue	nd in	tegei	ope	rand
	CRd	Copro	cess	or de	stina	tion I	regist	er							
Description	SUB loa tively, s	ads the ubtract	conte s the	ents conte	(integ ents d	ger) c of CR	of Rs <sub>1</sub> Is <sub>1</sub> fro	and om C	Rs <sub>2</sub> Rs <sub>2</sub> ,	into ( and s	CRs <sub>1</sub> store	and s the	CRs resu	<sub>2</sub> res It in C	pec- CRd.
Machine States	4 if the 3 if the	first ins first ins	tructi tructi	ion w on w	ord i	s Ion s not	g wor long	rd-ali word	gned d-alig	ned					
Instruction Type	CMOV	GC, two	o regi	sters	i										
Example	SUB B6	, A0,	RB5,	RA3	, RA	.7									
	This exa RB5 an in RA7.	ample l d RA3,	oads subt	TMS racts	3402 the c	0 reg conte	gister: ents o	s B6 a f RA	and A 3 fror	\0 int n RB	o cop 5, ar	oroce Id ste	essor ores t	regis the re	sters esult

Syntax	SUBI	UBD CRs <sub>1</sub> , CRs <sub>2</sub> , CRd														
Execution	CRs <sub>1</sub>	– C	Rs <sub>2</sub> ·	→ Cl	٦d											
Instruction Words	15 1 De	14 1 fault I	13 0 D	12 1	11 1 CF	10 0 Is <sub>1</sub>	9 0	8 0	7 0 CF	6 0 Rs <sub>2</sub>	5 0	4	3 0	2 1 CRd	1	0
Operands	CRs <sub>1</sub>	C si	opro on flo	cesso bating	or A g-poi	regis nt op	ter c eran	ontai d	ning	the r	ninue	end 6	64-bit	t dou	ble-p	reci-
	CRs <sub>2</sub>	рі рі	opro ecisi	cesso on flo	or B patin	regi: g-poi	ster nt op	conta peran	ainin Id	g the	e sub	otrah	end	64-bi	t doi	iple-
	CRd	С	opro	cesso	or de	stina	tion	regis	ter							
Description	SUBI stores	) sub s the	tract resu	s the Ilt in (	cont CRd.	ents	(dou	ble-p	recis	sion v	alue)	ofC	Rs <sub>2</sub> 1	rom (	CRs <sub>1</sub>	and
Machine States	2															
Instruction Type	CEXE	EC, s	hort													
Example	SUBD	RA5	, RB	3, R	A7											
	This example subtracts the contents of RB3 from RA5 and stores the result in RA7.															

Syntax	SUBD	<b>UBD</b> CRs <sub>2</sub> , CRs <sub>1</sub> , CRd													
Execution	CRs <sub>2</sub> –	CRs <sub>1</sub>	→ Cl	Rd											
Instruction Words	15 14 1 1 Defau	13 0 It ID	12 1	11 1 CF	10 0 Is <sub>1</sub>	9 0	8 0	7 0 CF	6 0 Rs <sub>2</sub>	5 0	4 0	3 1	2 1 CRd	1	0
Operands	CRs <sub>1</sub>	Copro cision	cess floati	or A i ing-p	regis oint o	ter co opera	ontai and ontai	ning	the s	ubtra	iheno	d 64-l	bit do	ble-r	-pre-
	0132	sion fl	cess	g-poi	nt op	berar	id .	n in ig		miu		J-+-DI		nie-h	
	Ска	Copro	cess	or ae	stina	ition	regis	ter							
Description	SUBD s stores t	ubtrac he resi	ts the ult in	cont CRd	ents	(dou	ble-p	orecis	sion v	alue)	) of C	Rs <sub>1</sub> 1	irom	CRsź	and
Machine States	2														
Instruction Type	CEXEC	, short													
Example	SUBD R	35, RA	.3, R	A7											
	This exa RA7.	ample s	subtra	acts t	he c	onter	nts of	f RA3	s from	n RB	5 anc	l stor	es th	e res	ult in

Syntax	SUBF CRs <sub>1</sub> , CRs <sub>2</sub> , CRd							
Execution	$CRs_1 - CRs_2 \rightarrow CRd$							
Instruction Words	15       14       13       12       11       10       9       8       7       6       5       4       3       2       1       0         1       1       0       1       1       0       0       0       0       0       0       1       1       0         Default ID       CRs1       CRs2       CRd       CRd       CRd       CRd							
Operands	CRs <sub>1</sub> Coprocessor A register containing the minuend 32-bit single-precision floating-point operand							
	CRs <sub>2</sub> Coprocessor B register containing the subtrahend 32-bit single-precision floating-point operand							
	CRd Coprocessor destination register							
Description	SUBF subtracts the contents (single-precision value) of $\text{CRs}_2$ from $\text{CRs}_1$ and stores the result in CRd.							
Machine States	2							
Instruction Type	CEXEC, short							
Example	SUBF RA5, RB3, RA7							
	This example subtracts the contents of RB3 from RA5 and stores the result in RA7.							

Syntax	SUB	JBF Rs <sub>1</sub> , Rs <sub>2</sub> , CRs <sub>1</sub> , CRs <sub>2</sub> , CRd														
Execution	Rs <sub>1</sub> - Rs <sub>2</sub> - CRs-	$s_1 \rightarrow CRs_1$ $s_2 \rightarrow CRs_2$ $Rs_1 - CRs_2 \rightarrow CRd$														
Instruction Words	15 0 0 De	14 0 1 efault I	13 0 0 D	12 0 0	11 0 0 CF	10 1 0 Rs <sub>1</sub>	9 1 1	8 0 1	7 0 0 CR	6 1 0	5 0 0	4 R R	3	2 R: R: CRd	1 <sup>5</sup> 1 <sup>5</sup> 2	0
Operands	Rs <sub>1</sub> Rs <sub>2</sub> CRs <sub>1</sub> CRs <sub>2</sub>	T si Si C flo si	MS3 on flo MS3 ngle- opro oating opro on flo	4020 pating 4020 prec cess g-poi cess pating	sour g-poi ision or A r nt op or B r g-poi	rce re nt va irce float egist eran regist nt op	egiste lue t regis ing-p er to d ter to eran	er for o cop ster ooint conta conta	the fi for the value ain the	irst (r ssor he s to c e mir ne su	minua ecor oproa nuen btrał	end) nd (s cesso d 32- nend	32-b subtr or bit si 32-b	it sing ahen ngle- it sing	gle-p d) 3 preci gle-p	reci- 2-bit ision reci-
Description	SUBI and C the re	C F loa CRs <sub>2</sub> esult	opro ds th resp in Cf	cesso le col lectiv Rd.	or de ntent ely, s	stina s (sir ubtra	tion ngle- acts t	regisi preci: he cc	sion v onten	value ts of	e) of CRs	Rs <sub>1</sub> a 2 fron	and I n CR	Rs <sub>2</sub> ii Is <sub>1</sub> , a	nto C nd st	Rs <sub>1</sub> ores
Machine States	4 if th 3 if th	ne firs ne firs	st ins st ins	tructi tructi	ion w ion w	ord i ord i	s lon s not	g wo long	rd-ali word	gned J-alig	l Ined					
Instruction Type	СМО	CMOVGC, two registers														
Example	SUBF	A0,	в6,	RA5	, RE	3, F	A7									
	This e RA5 a in RA	This example loads TMS34020 registers A0 and B6 into coprocessor registers RA5 and RB3, subtracts the contents of RB3 from RA5, and stores the result in RA7.														

# SUBF Subtract, Single Precision, (B Register – A Register)

Syntax	SUB	SUBF CRs <sub>2</sub> , CRs <sub>1</sub> , CRd														
Execution	CRs	2 – C	Rs <sub>1</sub>	→ Cl	Rd											
Instruction Words	15 1 De	14 1 efault l	13 0 D	12 1	11 1 CF	10 0 Is <sub>1</sub>	9 0	8 0	7 0 CF	6 0 Rs <sub>2</sub>	5 0	4 0	3 1	2 1 CRd	1 1	0
Operands	CRs	l C si	opro ion fle	cess oatin	or A ı g-poi	regist nt op	er co eran	ontaii d	ningt	he su	ubtra	hend	32-b	oit sin	gle-p	reci-
	CRs <sub>2</sub>	2 C	opro oatin	cess g-poi	or B r nt op	egist beran	er co d	ntair	ningti	he mi	nuer	nd 32-	-bit si	ngle-	prec	ision
	CRd	С	opro	cess	or de	stina	tion	regis	ter							
Description	SUB store	F sut s the	otract e resu	s the ult in	coni CRd	tents	(sing	jle-p	recis	ion va	alue)	of C	Rs <sub>1</sub> f	rom (	CRs <sub>2</sub>	and
Machine States	2															
Instruction Type	CEX	EC, s	short													
Example	SUBF	RB5	, RA	.3, F	A7											
	This RA7.	exan	nple s	subtra	acts f	the co	onter	nts of	f RAG	8 from	n RB	5 and	stor	es th	e res	ult in

Syntax	SUBF	<b>JBF</b> Rs <sub>2</sub> , Rs <sub>1</sub> , CRs <sub>2</sub> , CRs <sub>1</sub> , CRd														
Execution	Rs <sub>1</sub> → Rs <sub>2</sub> → CRs <sub>2</sub> -	$s_1 \rightarrow CRs_1$ $s_2 \rightarrow CRs_2$ $Rs_2 - CRs_1 \rightarrow CRd$														
Instruction Words	15 1	4	13	12	11	10	9	8	7	6	5	4	3	2	1	0
	0	0	0	0	0	1	1	0	0	1	0	R		R	S <sub>1</sub>	
									н							
Operands	Rs <sub>1</sub>	TMS34020 source register for the first (subtrahend) 32-bit single-pre- cision floating-point value to coprocessor														
	Rs <sub>2</sub>	TMS34020 source register for the second (minuend) 32-bit single-pre- cision floating-point value to coprocessor														
	CRs <sub>1</sub>	Co sio	proc n flo	cesso ating	or A r g-poi	egist nt op	er to eran	cont d	ain th	ne su	btrah	end	32-b	it sin	gle-p	reci-
	CRs <sub>2</sub>	Co floa	proc ating	cesso g-poi	or B r nt op	egist eran	erto d	cont	ain th	e mir	nueno	d 32-	bit si	ngle-	preci	sion
	CRd	Co	proc	cesso	or de	stina	tion r	egis	ter							
Description	SUBF and CF the res	load Rs <sub>2</sub> r ult ir	s the respe	e coi ectiv Rd.	ntent ely, s	s (sir ubtra	ngle-  acts t	oreci he co	sion v onten	value ts of	e) of l CRs <sub>1</sub>	Rs <sub>1</sub> a I fron	and I n CR	Rs <sub>2</sub> ii s <sub>2</sub> , a	nto C nd ste	Rs <sub>1</sub> ores
Machine States	4 if the 3 if the	first first	t inst t inst	ructi ructi	ion w ion w	ord is ord is	s lon s not	g wo Iong	rd-ali word	gned d-alig	ned					
Instruction Type	CMOVGC, two registers															
Example	SUBF F	36,	А0,	RB5	, RA	3, R	A7									
	This ex RB5 ar in RA7	This example loads TMS34020 registers B6 and A0 into coprocessor registers RB5 and RA3, subtracts the contents of RA3 from RB5, and stores the result n RA7.														

TMS34082 Pseudo-ops

# **Chapter 15**

# **Instruction Timing**

This chapter summarizes the timings of the TMS34020 assembly-language instruction set. It contains two sections:

	Section	on	Page
These sections are divided between MOVE and MOVB instructions and the remainder of the instructions.	15.1 15.2	Timing for All Instructions Except MOVEs and MOVBs Timing for MOVE and MOVB Instructions	15-2 15-10

Please note these characteristics about the timings listed in this book:

- Numbers identify TMS34020 machine states.
- All timings assume that the cache is enabled and that the instruction is in the cache.
- Numbers in parentheses identify hidden cycles.

The TMS34020 may execute some instructions in parallel, "hiding" some instruction states. Hidden cycles are memory-write cycles that occur at the end of an instruction. The machine states consumed by the instruction that the CPU is executing hide the machine states consumed by the write cycles. These hidden cycles are not counted against he instruction that incurs them, but are counted against subsequent instructions. If an instruction uses the local bus before all of the hidden cycles have been overlapped by subsequent instructions, that instruction must wait for the hidden cycles to complete.

- These timings assume that
  - All memory requests are granted when requested; no higher priority memory requests are pending.
  - When the CPU requests page-mode access, the memory grants it.
  - No wait states occur.
  - No retries occur.

# 15.1 Timing for All Instructions Except MOVEs and MOVBs

This section lists the instructions for all instructions except the MOVE and MOVB instructions. Please note that

- □ If the timing for an instruction states that this is a *complex instruction*, than no simple formula is available for providing the timing for the instruction. The number of machine states consumed by this instruction's execution will vary depending on the circumstances of its execution.
- Instruction timing for graphics instructions varies, depending on the pixelprocessing option you've selected. The timing formulas for graphics instructions (such as DRAV and LINE) ask you to add the values shown in Table 15–1 into your timing calculations.

Table 15–1.	Effects	of Pixel-Processi	ng Options	on Graphics	Instructions
-------------	---------	-------------------	------------	-------------	--------------

	Number of Cycles Required for the Followin Pixel Sizes								
<b>Pixel-Processing Option</b>	1	2 or 4	8, 16, or 32						
Replace	0 (2)	0 (2)	0 (1)						
ADD	—	2 (2)	2 (1)						
ADDS		3 (2)	3 (1)						
SUB		2 (2)	2 (1)						
SUBS	<u> </u>	3 (2)	3 (1)						
MAX	—	3 (2)	3 (1)						
MIN	—	3 (2)	3 (1)						
PPCODE	2 (2)	2 (2)	2 (1)						

Instruction	Number of machine cycles consumed by instruction execution
ABS	1
ADD	1
ADDC	1
ADDI (short)	2
ADDI (long)	<ul><li>2 if the immediate data is long-word aligned</li><li>3 if the immediate data is not long-word aligned</li></ul>
ADDK	1
ADDXY	1
ADDXYI	<ul><li>2 if the immediate data is long-word aligned</li><li>3 if the immediate data is not long-word aligned</li></ul>
AND	1

Instruction	Number of machine cycles consumed by instruction execution
ANDI	<ul><li>2 if the immediate data is long-word aligned</li><li>3 if the immediate data is not long-word aligned</li></ul>
ANDN	1
ANDNI	<ul><li>2 if the immediate data is long-word aligned</li><li>3 if the immediate data is not long-word aligned</li></ul>
BLMOVE	complex instruction
BTST (constant)	1
BTST (register)	1
CALL	3 + (1) if the SP is aligned 3 + (4) if the SP is not aligned
CALLA	<ul> <li>3 if immediate data is long-word aligned, 4 if SP is also long-word aligned</li> <li>3+(3) if immediate data is not long-word aligned, 4+(3) if SP is also not long-word aligned</li> </ul>
CALLR	<ul><li>3 + (1) if the SP is long-word aligned</li><li>3 + (4) if the SP is long-word not aligned</li></ul>
CEXEC (long)	<ul><li>2 (1) if the immediate data is long-word aligned</li><li>3 (1) if the immediate data is not long-word aligned</li></ul>
CEXEC (short)	2 (1)
CLIP	complex instruction
CLR	1
CLRC	1
CMOVCG	Single: 4 if the immediate data is long-word aligned 5 if the immediate data is not long-word aligned
	Double: 5 if the immediate data is long-word aligned 6 if the immediate data is not long-word aligned
CMOVCM (count*+)	5 + [count–1] if the immediate data is long-word aligned 6 + [count–1] if it is not (count is the number of 32-bit transfers)
CMOVCM (count-*)	5 + [count–1] if the immediate data is long-word aligned 6 + [count–1] if it is not (count is the number of 32-bit transfers)
CMOVCS	4 if the immediate data is long-word aligned 5 if it is not
CMOVGC (one register)	2 (1) if the immediate data is long-word aligned 3 (1) if it is not
CMOVGC (two registers)	3 (1) if the immediate data is long-word aligned 4 (1) if it is not

Instruction	Number of machine cycles consumed by instruction execution				
CMOVMC (constant*+)	5 + [constant-1] if the immediate data is long-word aligned 6 + [constant-1] if it is not (constant is the number of 32-bit transfers)				
CMOVMC (constant*-)	5 + [constant–1] if the immediate data is long-word aligned 6 + [constant–1] if it is not (constant is the number of 32-bit transfers)				
CMOVMC (register*+)	5 + [register value–1] if the immediate data is long-word aligned 6 + [register value–1] if it is not (the register value is the number of 32-bit transfers)				
CMP	1				
CMPI (long)	2 if the immediate data is long-word aligned 3 if it is not				
CMPI (short)	2				
СМРК	1				
CMPXY	1				
CPW	1				
CVDXYL	pitch is	a power of 2: 2 2 powers of 2: 3 arbitrary: 14			
CVMXYL	pitch is	a power of 2: 2 2 powers of 2: 3 arbitrary: 14			
CVSXYL	pitch is	a power of 2: 2 2 powers of 2: 3 arbitrary: 14			
CVXYL	pitch is	a power of 2: 3 2 powers of 2: 4 arbitrary: 15			
DEC	1				
DINT	3				
DIVS	Rd Odd: Rd Even:	<ul> <li>39 (normal case)</li> <li>41 (if result = 8000000h)</li> <li>7 (if Rs = 0)</li> <li>40 (normal case)</li> </ul>			
		41 (if result = $8000000$ ) 7 (if Rs = 0 or Rs < Rd)			
DIVU	Rd Odd: Rd Even:	37 (normal case) 7 (if Rs = 0) 37 (normal case) 5 (if Rs = 0,,,,,,,,			
		5 (If HS = 0 or HS ≤ H0)			

Instruction	Number of machine cycles consumed by instruction execution
DRAV	Window option0123inside $4+P+CD$ 5 $4+P+CD$ $4+P+CD$ outside $4+P+CD$ 353
<u>Key:</u> P CD	Selected pixel-processing option; see Table 15–1. Complexity of destination pitch. $CD = 0$ if CONVDP contains a power of 2; $CD = 1$ if CONVDP contains a sum of powers of 2; $CD = 12$ if CONVDP contains an arbitrary pitch.
DSJ	2 if no jump 3 if jump
DSJEQ	2 if no jump 3 if jump
DSJNE	2 if no jump 3 if jump
DSJS	2 if no jump 3 if jump
EINT	3
EMU	8 (more if the TMS34020 enters emulation mode)
EXGF	1 if F0 2 if F1
EXGPC	2
EXGPS	2 (1)
FILL L	complex instruction
FILL XY	complex instruction
FLINE	12 + 3CD + [2 + P]E + 3
	Key:PSelected pixel-processing option; see Table 15–1. If the number of hidden cycles is greater than 1, then $P = P + (hidden cycles - 1)$ .ETotal number of pixels drawn.CDComplexity of destination pitch. $CD = 0$ if CONVDP contains a pow- er of 2; $CD = 1$ if CONVDP contains a sum of powers of 2; $CD = 12$ if CONVDP contains an arbitrary pitch.
FPIXEQ	complex instruction
FPIXNE	complex instruction
GETPC	1
GETPS	2
GETST	1
IDLE	minimum execution time of 1 cycle before taking interrupt EMU: 5 cycles min before responds to halt NMI mode1: 8 NMI mode0, HINT, DPYINT, WINT, INT1, or INT2: 11 if SP aligned, else 13
INC	1

Instruction	Nun con	Number of machine cycles consumed by instruction					
JAcc	3 if	3 if no jump, else 4					
JRcc (short)	1 if	1 if no jump, else 2					
JRcc (long)	2 if	2 if no jump, else 3					
JUMP	2	2					
LINE	Wi	ndow option 0:	13 + 3 <i>CD</i>	+ [3 +P	]E + 2		
	Wi	ndow option 1:	13 + 3 <i>CD</i>	+ [3 + <i>P</i>	] <i>Q</i> + 2		
	Wi	ndow option 2:	13 + 3 <i>CD</i>	+ [3 + <i>P</i>	E + WV	′+2	
	Wi	ndow option 3:	13 + 3 <i>CD</i>	+ [3 + <i>P</i>	]E + 3Q	+ 2	
<u>Кеу:</u> Р WV Q E CD	Selected pixel-proce =3 if there is a wind Total number of pixe Total number of pixe Complexity of destin tains a sum of powe	essing option; see ow violation, = 0 c els calculated but els drawn. nation pitch. $CD$ = 0 ers of 2; $CD$ = 12 i	e Table 15- otherwise. not drawn. ) if CONVD f CONVDF	-1, but i P conta contai	gnore th iins a po ns an ar	wer of 2 bitrary (	en cycles. 2; <i>CD</i> = 1 if CONVDP con- pitch.
LINIT	9	· · · · · · · · · · · · · · · · · · ·					
LMO	1						
MMFM	#	registers moved # cycles	<u>1</u> 6	2 7	<u>3</u> 8	4 9	<u>n</u> n + 5
ММТМ	# Ic	registers moved ong-word aligned byte aligned bit aligned e: Add 1 to all tin	1 4(1) 4(1) 4(2) nings if the	2 6(1) 8(1) 9(2) MMTM	<u>3</u> 7(1) 9(1) 10(2)	4 8(1) 10(1) 11(2) tion is r	$\frac{n}{[4+n](1)}$ [6+n](1) [7+n](1) pot long-word aligned.
MODS	40 41 3	if result = 8000 ( if Rs = 0	0000h				
MODU	35 3	if Rs = 0					
MOVE Rs, F	?d 1						
MOVI (long)	2 3	if immediate data if it isn't	is long-wo	ord aligr	ned		
MOVI (short	2	·					
MOVK	1			-			
MOVX	1						
MOVY	1						
MPYS	Rs Rs	negative: 5 + (fi positive: 6 + (fi	ield size 1) ield size 1)	/2 /2			
MPYU	5 +	(field size 1)/2					

Instruction Timing

Instruction		Number of machine cycles consumed by instruction execution					
MWAIT		minimum of 2					
NEG		1					
NEGB		1					
NOP		1					
NOT		1					
OR		1					
ORI		2 if immediate data is long-word aligned 3 if it isn't					
PFILL		complex instruction					
PIXBLT B, L		complex instruction					
PIXBLT B, XY		complex instruction					
PIXBLT L, L		complex instruction					
PIXBLT L, M, L	•	complex instruction					
PIXBLT L, XY		complex instruction					
PIXBLT XY, L		complex instruction					
PIXBLT XY, XY	<b>,</b>	complex instruction					
PIXT <i>Rs</i> , * <i>Rd</i>		2 + <i>P</i>					
PIXT Rs, *Rd.X	(Y	Window option 0 1 2 3					
		$\begin{array}{rrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrrr$					
PIXT * <i>Rs</i> , <i>Rd</i>		3					
PIXT * <i>Rs</i> , * <i>Rd</i>		4 + <i>P</i>					
PIXT * <i>Rs</i> .XY, F	ld	6 + <i>CS</i>					
PIXT * <i>Rs</i> .XY, */	Rd.XY	$\begin{array}{c cccccc} \hline Window \ option & 0 & 1 & 2 & 3 \\ \hline inside & 7+CS+CD+P & 5 & 7+CS+CD+P & 7+CS+CD+P \\ outside & 7+CS+CD+P & 3 & 5+CD & 3+CD \end{array}$					
<ul> <li>Key: P Selected pixel-processing option; see Table 15–1 (page 15-2).</li> <li>CD Complexity of destination pitch. CD = 0 if CONVDP contains a power of 2; CD = 1 if CONVDP contains a sum of powers of 2; CD = 12 if CONVDP contains an arbitrary pitch.</li> <li>CS Complexity of source pitch. CS = 0 if CONVSP contains a power of 2; CS = 1 if CONVSP contains a sum of powers of 2; CS = 12 if CONVSP contains an arbitrary pitch.</li> </ul>							
POPST		6 if the SP is aligned 7 if it isn't					
PUSHST		2 (1) if the SP is aligned 2 (2) if it isn't					
PUTST		3					
Instruction	Number o consumed	f machine cycles I by instruction e	xecution				
----------------	--	---	--	---	--		
RETI	52 if BF 38 if IX else 7	status bit = 1 status bit = 1					
RETM	52 if BF 38 if IX else 10	status bit = 1 status bit = 1					
RETS	5 6 if the s	stack isn't aligned					
REV	1						
RL (constant)	1						
RL (register)	1						
RMO	1						
RPIX	2 if PSIZE 4 if PSIZE 5 if PSIZE 6 if PSIZE 7 if PSIZE 8 if PSIZE	= 32 = 16 = 8 = 4 = 2 = 1					
SETC	1						
SETCDP	pitch is	a power of 2: 2 powers of 2: arbitrary:	4(1) 6(1) 3(1)				
SETCMP	pitch is	a power of 2: 2 powers of 2: arbitrary:	4(1) 6(1) 3(1)				
SETCSP	pitch is	a power of 2: 2 powers of 2: arbitrary:	4(1) 6(1) 3(1)				
SETF	1	-					
SEXT	2						
SLA (constant)	3						
SLA (register)	3		······································				
SLL (constant)	1						
SLL (register)	1						
SRA (constant)	1		· · · · · · · · · · · · · · · · · · ·		······································		
SRA (register)	1		· · · · · · · · · · · · · · · · · · ·				
SRL (constant)	1			,			
SRL (register)	1	· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·		· · · · · · · · · · · · · · · · · · ·		

Instruction	Number of machine cycles consumed by instruction execution
SUB	1
SUBB	1
SUBI (long)	<ul><li>2 if the immediate data is long-word aligned</li><li>3 if it isn't</li></ul>
SUBI (short)	2
SUBK	1
SUBXY	1
SWAPF	5
TFILL	complex instruction
TRAP	7 if TRAP 0, else 10 if ST aligned else 12
TRAP L	10 if ST aligned else 12
VBLT	complex instruction
VFILL	complex instruction
VLCOL	2 (1)
XOR	1
XORI	<ul><li>2 if the immediate data is long-word aligned</li><li>3 if it isn't</li></ul>
ZEXT	1

# 15.2 Timing for MOVE and MOVB Instructions

This section contains the timing for MOVE and MOVB instructions. These timings are divided into three categories:

- Timings for memory-to-register moves (reads)
- Timings for register-memory moves (writes)
- Timings for memory-to-memory moves

### General assumptions

The timing of the move instructions depends on how the accessed field is aligned in memory. The following cases of field alignment characterize the move instruction timing.

- 1) The field is aligned on the boundaries of a long word or on any byte boundaries.
- 2) At least one end of the field is not aligned to a byte boundary.
- 3) The field crosses a long-word boundary, but both ends are aligned on byte boundaries.
- 4) The field crosses a long-word boundary, and only one end is aligned on a byte boundary.
- 5) The field crosses a long-word boundary, and neither end is aligned on a byte boundary.

### Table 15–2. Cases Table for MOVE and MOVB Timings

Case Number	Number of Read Cycles Required	Number of Write Cycles Required
1	2	2
2	2	3
3	3	3
4	3	4
5	3	5

The timing tables refer to these cases.

## Memory-to-register moves -----

			Case		
Instruction	1	2	3	4	5
MOVB *Rs, Rd	4	4			5
MOVB *Rs(SOffset), Rd	6	6			7
MOVB @SAddress, Rd	5/6	5/6			6/7
MOVE * <i>Rs, Rd</i>	3	3	4	4	4
sign extended:	4	4	5	5	5
MOVE * <i>Rs</i> +, <i>Rd</i>	3	3	4	4	4
sign extended:	4	4	5	5	5
MOVE* <i>Rs, Rd</i>	4	4	5	5	5
sign extended:	5	5	6	6	6
MOVE * <i>Rs(SOffset), Rd</i>	4	4	5	5	5
sign extended:	6	6	7	7	7
MOVE @ <i>Rs, Rd</i>	4/5	4/5	5/6	5/6	5/6
sign extended:	5/6	5/6	6/7	6/7	6/7

# Register-to-memory moves ----

			Case		
Instruction	1	2	3	4	5
MOVB <i>Rs</i> , * <i>Rd</i>	1(1)	1(2)			1(4) 2(3)
MOVB Rs, *Rd	3(1)	3(2)			3(4)
MOVB <i>Rs</i> , @ <i>Rd</i> big endian	2(1)/3(1) 3(1)/3(1)	2(2)/3(2) 3(2)/3(2)			2(4)/3(4) 3(4)/3(4)
MOVE <i>Rs</i> , * <i>Rd</i> big endian	1(1) 2(1)	1(2) 2(2)	1(2) 2(2)	1 (3) 2(3)	1(4) 2(4)
MOVE Rs, *Rd+	1(1) 2(1)	1(2) 2(2)	1(2) 2(2)	1 (3) 2(3)	1(4) 2(4)
MOVE Rs, -*Rd	2(1)	2(2)	2(2)	2(3)	2(4)
MOVE Rs, -*Rd	3(1)	3(2)	3(2)	3(3)	3(4)
MOVE <i>Rs</i> , @ <i>Rd</i> big endian	2(1)/3(1) 3(1)/3(1)	2(2)/3(2) 3(2)/3(2)	2(2)/3(2) 3(2)/3(2)	2(3)/3(3) 3(3)/3(3)	2(4)/3(4) 3(4)/3(4)

### Memory-to-memory moves -

First, look in Table 15–2 (page 15-10) to find the source alignment (case 1–5) and the destination alignment (case1–5). Then, useTable 15–3 to find which column to use in the timing table below.

Table 15–3.	Source/Destination	Alignment for MO	VE and MOVB Timings
	•		

			Destination		
Source	1	2	3	4	5
1	А	С	С	Н	✓ E
2	А	С	С	Н	E
3	В	D	D	G	F
4	В	D	D	G	F
5	В	D	D	G	F

R/W Cycles	2/2	3/2	2/3	3/3	2/5	3/5	3/4	2/4
	Α	В	С	D	Е	F	G	н
MOVB *Rs, *Rd	3(1)	4(1)	3(2)	4(2)	3(4)	4(4)		
MOVB *Rs(SOffset), *Rd(DOffset)	5(1)	6(1)	5(2)	6(2)	5(2)	6(4)	]	
MOVB @SAddress,@DAddress even odd	5(1) 7(1)	6(1) 8(1)	5(2) 7(2)	6(2) 8(2)	5(4) 7(4)	6(4) 8(4)		_
MOVE *Rs, *Rd	3(1)	4(1)	3(2)	4(2)	3(4)	4(4)	4(3)	3(3)
MOVE * <i>Rs</i> +, * <i>Rd</i> +	3(1)	4(1)	3(2)	4(2)	3(4)	4(4)	4(3)	3(3)
MOVE* <i>Rs</i> ,* <i>Rd</i>	4(1)	5(1)	4(2)	5(2)	4(4)	5(4)	5(3)	4(3)
MOVE * <i>Rs(SOffset</i> ), * <i>Rd</i> +	5(1)	6(1)	5(2)	6(2)	5(4)	6(4)	6(3)	5(3)
MOVE *Rs(SOffset), *Rd(DOffset)	5(1)	6(1)	5(2)	6(2)	5(4)	6(4)	6(3)	5(3)
MOVE @Rs, *Rd+ even odd	4(1) 5(1)	5(1) 6(1)	4(2) 5(2)	5(2) 6(2)	4(4) 5(4)	5(4) 6(4)	5(3) 6(3)	4(3) 5(3)
MOVE @ <i>Rs</i> , @ <i>Rd</i> even odd	5(1) 7(1)	6(1) 8(1)	5(2) 7(2)	6(2) 8(2)	5(4) 7(4)	6(4) 8(4)	6(3) 8(3)	5(3) 7(3)

# **Appendix A**

# **Test and Emulation Considerations**

This appendix provides information that you'll need if you're building a TMS34020 target system and you plan to use the TMS34020 Emulator. The TMS34020 Emulator supports realtime in-circuit emulation; key features include

- Serial scan-path technology. The emulator uses TI's revolutionary serial scan-path technology, eliminating the need for the typical emulator target cable, which uses a full device pinout. Instead, the target system needs only a 12-pin header to connect between the TMS34020 and the TMS34020 emulator board through the emulation target cable.
- PC-compatible emulator board. The emulator board is a PC/XT-compatible emulator board. It provides a high-speed communication path between a PC and the TMS34020.
- Symbolic debugger with windowed interface. The emulator's symbolic debugger provides the following features through its windowed interface:
  - Ability to upload/download application code and emulation setup
  - Software breakpoints on selected instructions
  - Single-step execution

\*\*\*\*\*

- Access to registers and memory
- TMS34020 patch assembler/disassembler
- Benchmark timing

	Sect	lion	Page
The remainder of this appendix	A.1	Overview of an Emulation System	. A-2
contains information about setting	A.2	Emulation Connector (12-Pin Header)	. A-3
up your target system.	A.3	Signal Buffering	. A-4
	A.4	Buffer Delays	. A-5
	A.5	Design Considerations	. A-7
	A.6	Mechanical Dimensions	. A-9

# A.1 Overview of an Emulation System

Figure A–1 shows a typical setup using the emulator, target cable, and your target system.

Figure A–1. Typical Setup Using the TMS34020 Emulator and Your Target System



Figure A–2 shows how you connect the emulator and target cable to your target system.

Figure A–2. Connecting the TMS34020 Emulator to Your Target System



# A.2 Emulation Connector (12-Pin Header)

To use the target cable, your target system must have a 12-pin header (2 rows of 6 pins) with the connections that are shown in Figure A–3. The header pins connect directly to the TMS34020 except when the header is farther than 2 inches from the TMS34020 (see Section A.3 on page A-4).

Figure A–3. 12-F	'IN HE	eader	Signal	s
------------------	--------	-------	--------	---

		•			
Header Dimensions:		EMU1	1	2	GND
Pin-to-pin spacing Pin width:	0.025 inches	EMU0	3	4	GND
Pin length:	square post	EMU2	5	6	GND
	nominal	PD (+5V)	7		no pin (key)
		EMU3	9	10	GND
		LCLK1	11	12	GND

Signal	Description	TMS34020 Pin Number	
EMU0	Emulation pin 0	J1	
EMU1	Emulation pin 1	J3	
EMU2	Emulation pin 2	K1	
EMU3	Emulation pin 3	H2	
LCLK1	TMS34020 local clock 1	H1	
PD	Presence detect. Indicates that the cable is connected and target system is powered up. Tie PD to +5 volts in the target system.		

Although you can use other headers, recommended parts include

straight header, unshrouded	DuPont Connector Systems part number 67996–112
right-angle header, unshrouded	DuPont Connector Systems part number 68405–112
right-angle header, 4-wall shrouded	AMP, Incorporated part number 103167–3

# A.3 Signal Buffering

It is extremely important to provide high-quality signals between the emulator and the TMS34020 on the target system. In many cases, the signal must be buffered to produce a high-quality signal. The need for signal buffering and placement of the emulation header can be divided into 3 categories:

□ **No signal buffering.** In this situation, the distance between the header and the TMS34020 should be no more than 2 inches.



■ Buffered transmission signals. In this situation, the distance between the emulation header and the TMS34020 is greater than 2 inches but less than 6 inches. The transmission signals—LCLK1 and EMU3—are buffered through the same package.



❑ All signals buffered. The distance between the emulation header and the TMS34020 is greater than 6 inches but less than 12 inches. All TMS34020 emulation signals—EMU0, EMU1, EMU2, and EMU3—are buffered through the same package.



# A.4 Buffer Delays

The absolute maximum propagation delay for both -32 and -40 TMS34020 devices is 10 ns. The buffer is noninverting, and all emulation signals that are buffered should be buffered through the same package.

The distance between the TMS34020 and the buffers depends on the PWB layout and loading on LCLK1. However, Texas Instruments suggests that the distance be as short as possible and less than 4 inches.

When you buffer LCLK1, don't place another device between the buffer output and header. Connecting another device to this signal could cause false triggering of the device due to cable reflections (see Figure A–4).





Figure A–5 shows a portion of logic in the emulator pod. Note that  $33-\Omega$  resistors are added to EMU0, EMU1, and EMU2; this minimizes cable reflections.

*Figure A–5. Emulator Pod Interface* 



A-6

# A.5 Design Considerations

When designing a TMS34020 target system, please observe these hardware and software emulation constraints. Portions of these design considerations are advanced information and may not apply to all Texas Instruments emulators.

- Reset and interrupts. When an emulator is active, the TMS34020 will service reset and interrupts only if the emulator is in an execution mode. The target system must provide a reset to the TMS34020 before the emulator is activated.
- Host/emulation coordination. If the emulator has stopped execution of the TMS34020 (program execution is halted), the TMS34020 will continue to respond to host port accesses. If TMS34020 program execution is required to provide a response to a host access, the host could hang or timeout. Also, functions such as reset, interrupts, NMI, and HLT will not take effect until the emulator is placed back in an execution mode; this could also hang the host application if a response is required. Emulators and host applications typically use timeouts to keep from hanging if a TMS34020 function is not performed properly. If both the emulator and host are accessing the TMS34020 memory space at the same time, false timeouts could occur in both the emulator and the host.

### Note:

Both the host and emulator can access the same memory space at effectively the same time. Thus, the emulator's memory display could be inaccurate if the host is modifying a memory location within the display range.

To minimize these conflicts, the host can use 3 bits within HSTCTLL to grant access of the TMS34020 to the emulator. These bits are:

- EMR (emulator request),
- EMG (emulator grant), and
- EMIEN (emulator interrupt enable).

The emulator sets EMR when the emulator requires access to the device. If EMIEN is set, a host interrupt is generated via the  $\overline{\text{HINT}}$  pin. When the host sets EMG, the interrupt is cleared and the emulator performs its pending function.

TMS34020 execution will be stopped immediately if an emulation halt condition (such as a breakpoint) is encountered, although emulation access of the TMS34020 will not start until EMG is set. The host processor can use either the host interrupt or the EMR bit to indicate that an emulator halted the TMS34020. When the emulator no longer requires access to the device, the emulator clears EMR. Once again, this causes a host interrupt if EMIEN is set. The host interrupt is deactivated when the host clears EMG.

Using this handshake protocol is optional and should be used in applications that are sensitive to emulation access of the TMS34020. Before attempting to integrate this protocol into your system, consult the *TMS34020 XDS Emulator User's Guide* for additional information.

# A.6 Mechanical Dimensions

Figure A-6 shows the TMS34020 emulator target cable, which consists of

- an emulator connector,
- a 3-foot section of jacketed cable,
- an active cable pod,
- $\Box$  a short section of jacketed cable that connects to the target system, and
- a 12-pin connector that connects to the target system's 12-pin header.





The overall cable length is approximately 3'10". Figure A–7 shows the mechanical dimensions for the target cable pod. The cable pod box is nonconductive plastic with 4 recessed metal screws.





Note: All dimensions are in inches and are nominal dimensions unless otherwise specified.

### *Figure A–8.* 12-Pin Connector Dimensions

(a) Side view



(b) Top view



- **Notes:** 1) All dimensions are in inches and are nominal dimensions unless otherwise specified.
  - 2) Pin-to-pin spacing on the connector is 0.100 inches in both the X and Y planes.

# **Appendix B**

# Glossary

address/status subcycle: First part of a local-memory cycle, sometimes referred to as *row-address time*.

- aliasing: Stairstep effect on a raster display of a line or arc segment.
- **ALTCH:** Address latch signal. You can use the high-to-low transition of ALTCH to capture the address and status present on the LAD bus.
- **antialiasing:** Method for reducing the severity of aliasing effects by adjusting the intensity of a pixel according the pixel's proximity to the line or edge of an object.
- **back porch:** Portion of horizontal or vertical blanking that follows the trailing edge of the horizontal- or vertical-sync pulse.
- **bandwidth:** Number of bits per second that can be transferred by a device.
- **BEN:** Big-endian enable (bit 0 of CONFIG register). BEN=0 (default) selects little-endian addressing mode; BEN=1 selects big-endian addressing mode.
- **big-endian:** Addressing mode in which the "big" or most significant end of an address (bit 31) points to the least significant end (bit 0) of a word of data.
- **binary array:** 2-dimensional bitmap in which each pixel is represented as a single bit (a 0 or a 1).
- **bitblt:** Bit-aligned block transfer. Transfer of a rectangular array of pixel information from one location in a bitmap to another.
- **bitmap:** 1. Digital representation of an image in which bits are mapped to pixels. 2. Block of memory used to hold raster images in a device-specific format.

**bit plane:** Hardware used as a storage medium for a bitmap.

- **black level:** Amplitude of the composite signal at which the beam of the picture tube is extinguished (becomes black) to blank retrace of the beam. This level is established at 75% of the signal amplitude.
- **blanking signals:** Pulses that extinguish the scanning beam during horizontal or vertical retrace periods.
- **breakpoint:** Point within a routine at which the routine may be interrupted by external intervention.
- **BSFLTD:** Bus-fault data registers (32-bit I/O register, address C000 0320h). The TMS34020's memory controller saves the LAD data into BSFLTD when a bus fault occurs on a CPU-initiated memory access.
- **BSFLTDL:** 16 LSBs of BSFLTD, accessed at address C000 320h.
- **BSFLTDH:** 16 MSBs of BSFLTD, accessed at address C000 330h.
- **BSFLTST:** Bus-fault status register (16-bit I/O register, address C000 02D0h). The TMS34020's memory controller saves its state in BSFLTST before it signals that a bus fault occurred.
- **BUSFLT:** Bus fault signal. External logic asserts BUSFLT to indicate that a fault occurred on the current bus cycle.

cache memory: A fast, on-chip memory.

**cache hit:** The cache contains the requested instruction word.

**cache miss:** The cache does not contain the requested instruction word.

CAD: Computer-aided design.

- **CAMD:** Column-address mode. Shifts the column address on the RCA bus to allow mixing of DRAM and VRAM address matrices.
- **CAS:** Column-address strobes (CAS0—CAS3). Drive the CAS inputs of DRAMs and VRAMs.
- **CBP:** Configuration byte protect (bit 4 of CONFIG register). CBP=0 is the default; CBP=1 write-protects the LSbyte of CONFIG until a reset occurs.
- **CD:** Cache disable (bit 15 of CONTROL register). CD=0 (default) enables cache operation; CD=1 forces the TMS34020 to ignore the contents of the cache and to fetch instructions from memory.
- **CF:** Cache flush (bit 14 of HSTCTLH register). Setting CF to 1 flushes and disables the cache. Normal cache operation resumes when CF is cleared to 0.

С

- **clipping:** Removing parts of display elements that lie outside a defined boundary (the boundary is usually a window or a viewport).
- **COLOR0:** Background color register (B8). Identifies the replacement color for 0-value pixels in a source array.
- **COLOR1:** Foreground color register (B9). Identifies the replacement color for pixels that will be altered in the destination array.
- column-address time: See data subcycle.
- **composite video:** Color-picture signal plus all blanking and sync signals. The signals include luminance and chrominance signals, vertical- and horizontal-sync pulses, vertical- and horizontal-blanking pulses, and the color-burst signal.
- **CONFIG:** Configuration register (16-bit I/O register, address C000 01A0h). Contains fields that selectively enable/disable various aspects of system configuration.
- **CONTROL:** Memory control register (16-bit I/O register, addresses C000 00B0h and C000 0190h). Controls various aspects of CPU activity.
- **CONVDP:** Destination pitch conversion factor register (16-bit I/O register, address C000 0140h). Contains a control parameter used for converting an XY destination address to a linear address.
- **CONVMP:** Mask pitch conversion factor register (16-bit I/O register, address C000 0180h). Contains a control parameter used for converting an XY mask address to a linear address.
- **CONVSP:** Source pitch conversion factor register (16-bit I/O register, address C000 0130h). Contains a control parameter used for converting an XY source address to a linear address.
- **coprocessor:** An additional processor in a system; extends the functionality of the main processor. For example, the TMS34082 is a coprocessor for the TMS34020; in a TMS34020 system, the TMS34082 adds floating-point capabilities to the TMS34020's functions.
- **CSD:** Composite-sync direction (bit 2 of DPYCTL register). When the CSYNC/HBLNK pin is configured as CSYNC (CVD=0), CSD determines if CSYNC is configured as in input (CSD=0) or an output (CSD=1).
- **CST:** CPU shift-register transfer enable (bit 11 of DPYCTL register). When CST=1, the TMS34020 converts pixel accesses into VRAM shift-register transfer cycles.
- **CVD:** Composite video disable (bit 3 of DPYCTL register). Controls the functions of the CSYNC/HBLNK and CBLNK/VBLNK pins. CVD=0 selects CSYNC and CBLNK; CVD=1 selects HBLNK and VBLNK.



- DAC: Digital-to-analog converter.
- **DADDR:** Destination address register (B2). Contains the destination array address for graphics instructions.
- **data subcycle:** Second part of a local-memory cycle, sometimes referred to as *column-address time*.
- **DDIN:** Data bus direction input-enable signal. Drives the active-high input enables on bidirectional transceivers.
- **DDOUT:** Data bus direction output-enable signal. Drives the active-low output enables on bidirectional transceivers.
- **DGIS:** Direct graphics interface standard.
- **DIE:** Display interrupt enable (bit 10 of INTENB register). Setting DIE to 1 enables the display interrupt.
- **DIP:** Display interrupt pending (bit 10 of INTPEND register). DIP is set to 1 when a display interrupt is requested.
- **DINC:** Display increment registers (32-bit I/O register, address C000 0240h). Contains the increment value for the DPYNX register.
- DINCL: 16 LSBs of DINC, accessed at address C000 0240h.
- **DINCH:** 16 MSBs of DINC, accessed at address C000 0250h.
- **display area:** Rectangular portion of the physical display screen in which information is visibly displayed; does not include the border area.
- **display element:** Basic graphic element that can be used to construct a display image.
- **display memory:** Area of memory used to hold the graphics image output to the video monitor.
- **display pitch:** Difference in memory addresses between two vertically adjacent positions on the screen.
- **dotclock:** Clock that cycles the rate at which video data is output to a CRT.
- **DPTCH:** Destination pitch register (B3). Defines the linear difference between starting addresses of adjacent rows in a destination array.
- **DPYADR:** Display address register. Provides compatibility with the TMS34010.
- **DPYCTL:** Display control register (16-bit I/O register, address C000 0080h). Controls video timing and VRAM serial-register transfers.

- **DPYINT:** Display interrupt register (16-bit I/O register, address C000 00A0h). Identifies the next scan line (in some circumstances, the next *half* scan line) at which a display interrupt can be requested.
- **DPYNX:** Display next address registers (32-bit I/O register, address C000 0220h). Contains a 32-bit address that is output during a screen-refresh cycle.
- **DPYNXL:** 16 LSBs of DPYNX, accessed at address C000 0220h.
- **DPYNXH:** 16 MSBs of DPYNX, accessed at address C000 0230h.
- **DPYMSK:** Display mask register (16-bit I/O register, address C000 02E0h). When midline reload screen refreshes are enabled, DPYMSK determines which bits of DPYNX & DPYST correspond to the tap-point portion of the address output during screen-refresh cycles.
- **DPYST:** Display start address registers (32-bit I/O register, address C000 0200h). Contains a 32-bit address that points to the pixel at the left of the 1<sup>st</sup> line displayed on the screen.
- **DPYSTL:** 16 LSBs of DPYST, accessed at address C000 0200h.
- **DPYSTH:** 16 MSBs of DPYST, accessed at address C000 0210h.
- **DPYSTRT:** Display start address register. Provides compatibility with the TMS34010.
- **DPYTAP:** Display tap-point address register. Provides compatibility with the TMS34010.
- **DQ:** Data in/data out pin for a VRAM.
- **DRAM:** Dynamic RAM.
- **DRAM refresh:** Maintenance of data stored in dynamic RAMs. Data are stored in DRAMs as electrical charges across a grid of capacitive cells. The charge stored in a cell will leak off over time unless the data is refreshed.
- **DYDX:** Delta Y/delta X register (B7). Defines the X and Y dimensions of a rectangular destination array.

- **EMIEN:** Emulator host-interrupt enable (bit 12 of HSTCTLL register). The value of EMIEN determines if EMG XOR EMR asserts HINT active low (EMIEN=1) or not (EMIEN=0).
- **EMG:** Emulator handshake (bit 11 of HSTCTLL register). In an emulation system, the host sets EMG to 1 to gran the emulator access to TMS34020 memory.

- **EMR:** Emulator handshake (bit 10 of HSTCTLL register). In an emulation system, the emulator sets EMR to 1 to request access to TMS34020 memory.
- **ENV:** Enable video (bit 15 of DPYCTL register). ENV enables (ENV=1) or disables (ENV=0) the video screen.
- field: 1. Group of contiguous bits in a register or memory location, dedicated to a particular function or representing a single entity. 2. Software-configurable data type supported by the TMS34010 and TMS34020; the field length can be programmed to be any value in the range of 1 to 32 bits.
- fill: Solid coloring or shading of a display surface, often achieved as a pattern of horizontal segments.
- **frame:** 1. Time required to refresh an entire screen. 2. Screen image output during a single vertical sweep.
- **frame buffer:** Portion of memory used to buffer raster data to be output to a CRT. Frame buffer contents are often referred to as the bitmap of the display and contain the logical pixels corresponding to the points on the monitor screen.
- **front porch:** Portion of a vertical- or horizontal-blanking pulse that precedes the leading edge of the vertical- or horizontal-sync pulse.
- **GI:** Bus grant input. External bus arbitration logic pulls **GI** low to enable the TMS34020 to gain access to the local-memory bus.
- **GKS:** Graphics kernel system. Application programmer's standard interface to a graphics display.
- gray scale: Scale of light intensities from black to white.
- **GSP:** Graphics system processor. A single-chip device embodying all the processing power and control capabilities necessary to manage a high-performance bitmapped graphics system. The TMS34010 and TMS34020 are GSPs.
- **HA:** Host address input bus (HA5—HA31). A host processor requests an address over these lines.

G

- **HACK:** Halt acknowledge (bit 4 of HSTCTLH register). Setting the HLT bit halts TMS34020 execution at the next interruptible instruction boundary; the TMS34020 sets HACK when the halt actually takes place.
- **HBFI:** Host-bus-fault interrupt (bit 14 of HSTCTLL register). The TMS34020 sets HBFI to 1 if a bus fault occurs on a host access.
- **HBREN:** Host-bus-fault/retry-interrupt (bit 15 of HSTCTLL register). If HBREN=1, the TMS34020 interrupts the host when a retry or bus fault occurs.
- **HBS:** Host byte select-bus (HBS0—HBS3). Identify the bytes to be selected within a specific word.
- **HCOUNT:** Horizontal count register (16-bit I/O register, address C000 01D0h). HCOUNT counts the number of VCLK periods per horizontal scan line.
- **HCS:** Host chip-select signal. A host drives HCS low to latch the current address and byte-select requests.
- HDST: Host data strobe signal.
- **HEBLNK:** Horizontal end blank register (16-bit I/O register, address C000 0030h). HEBLNK identifies the endpoint for the horizontal blanking interval.
- **HESERR:** Horizontal end serration register (16-bit I/O register, address C000 0270h). HESERR determines the endpoint for the composite-sync pulse during the serration region of vertical blanking.
- **HESYNC:** Horizontal end sync register (16-bit I/O register, address C000 0010h). HESYNC identifies the endpoint for horizontal sync.
- **HIE:** Host interrupt enable (bit 9 of INTENB register). Setting HIE to 1 enables the host interrupt.
- **high impedance:** The third state of a three-state output driver, in which the output is driven neither high or low but behaves as an open connection.
- **HIP:** Host interrupt pending (bit 9 of INTPEND register). HIP is set to 1 when a host interrupt is requested.
- **HINC:** Host increment (bit 12 of HSTCTLH register). Setting HINC to 1 enables the TMS34020 to compare the fetched address to the address requested by a host processor, to increment the current address, and to prefetch the contents of the next address.
- **HINT:** Host interrupt signal.
- **HLB0, HLB1:** Host last byte (bits 5&6 of HSTCTLH register). The HLB code tells the TMS34020 which byte of a 32-bit word that a host processor will access last. The TMS34020 uses this information to determine the correct time to prefetch the next word.

- **HLT:** Halt TMS34020 program execution (bit 15 of HSTCTLH register). Setting HLT to 1 suspends TMS34020 instruction processing at the next instruction boundary.
- **HOE:** Host output-enable signal.
- **hold signal:** Signal capable of controlling a processor bus; sent to a bus arbiter to request bus control. Typically, the arbiter grants the request by sending a hold-acknowledge signal to the requestor.
- **horizontal back porch:** Portion of horizontal blanking that follows the trailing edge of the horizontal-sync pulse.
- **horizontal-blanking interval:** Time during which the display is blanked to cover the horizontal retracing of the electron beam on a screen.
- **horizontal front porch:** Portion of a horizontal-blanking pulse that precedes the leading edge of the horizontal-sync pulse.
- **horizontal sync:** Synchronization signal that enables horizontal retrace of the electron beam on a screen.
- **host address bus:** Lines used by a host processor to identify the address of a TMS34020 local-memory location.
- **host processor:** Main processor in a system.
- **HPFW:** Host prefetch-after-write enable (bit 10 of HSTCTLH register). When host prefetches are enabled (HINC=1), the value of HPFW determines if the TMS34020 performs prefetches after reads (HPFW=0) or after writes (HPFW=1).
- **HRDY:** Host ready signal. Driven high when the TMS34020 is ready to complete a host-initiated access.
- **HREAD:** Host read strobe. Driven low during a host's read request.
- **HRYI:** Host-retry interrupt (bit 13 of HSTCTLL register). The TMS34020 sets HRYI to 1 if it retries a host access.
- **HSBLNK:** Horizontal start blank register (16-bit I/O register, address C000 0050h). HSBLNK identifies the startpoint for the horizontal blank-ing interval.
- **HSD:** Horizontal-sync direction (bit 0 of DPYCTL register). Determines if HSYNC is configured as an input (HSD=0) or an output (HSD=1).
- **HSTADRL:** Host address register. Provides compatibility with the TMS34010.
- **HSTADRH:** Host address register. Provides compatibility with the TMS34010.
- **HSTCTLH:** Host control I/O register, high word (16-bit I/O register, address C000 0100h). Controls aspects of host-interface communications.

- **HSTCTLL:** Host control I/O register, low word (16-bit I/O register, address C000 00F0h). Controls aspects of host-interface communications.
- **HSTDATA:** Host data I/O register. Provides compatibility with the TMS34010.
- **HTOTAL:** Horizontal total register (16-bit I/O register, address C000 0070h). Number of VCLK periods per horizontal scan line; defines the startpoint for the horizontal sync pulse.
- **HWRITE:** Host write strobe. Driven low during a host's write request.

**interlaced video:** Video system in which odd-numbered scan lines (odd field) are interlaced with even-numbered scan lines (even field). The odd and even fields constitute one frame. In effect, the number of transmitted pictures is doubled; this reduces flicker.

**IHOST:** Internal host interface address registers (4 32-bit registers: IHOST1, address C000 0308h; IHOST2, address C000 03A0h; IHOST3, address C000 03C0h; IHOST4, address C000 03E0h). The TMS34020 uses these registers for storing information provided by the host.

**implied operand:** A register value that must be supplied for an instruction to execute properly. The B-file registers and several of the I/O registers serve as implied operands for the TMS34020's graphics instructions.

**INTENB:** Interrupt enable register (16-bit I/O register, address C000 0110h). Selective enables /disables external interrupts 1 and 2, the host interrupt, the display interrupt, and the window violation interrupt.

**INTPEND:** Interrupt pending register (16-bit I/O register, address C000 0120h). Identifies the pending/not pending status of external interrupts 1 and 2, the host interrupt, the display interrupt, and the window violation interrupt.

**INTIN:** Interrupt-in (bit 3 of HSTCTLL register).

**INTOUT:** Interrupt-out (bit 7 of HSTCTLL register).

- K: 1) 1024. 2) Approximately 1000. 3) A 5-bit constant for a TMS34020 instruction.
  - **Kbyte:** Approximately 1000 bytes.

Κ

**little-endian:** An addressing mode in which the "little" or least significant end of an address (bit 0) points to the least significant end (bit 0) of a word of data.

long word: 32-bit word.

- **look-up table:** Table used during scan conversion of a digital image that converts color-map addresses into the actual color values displayed.
- **LRDY:** Local ready signal. External circuitry drives LRDY low to stop the TMS34020 from completing a local-memory cycle.
- **LRU:** Least recently used (cache-replacement algorithm). When a cache miss occurs, this algorithm selects the cache segment that will be overwritten, based on the likelihood that the data in the discarded segment will not be needed again for some time. The LRU algorithm selects the segment that was used least recently.
- **LSB:** Least significant bit.

**LSbyte:** Least significant byte.

LSW: Least significant word.

**mask:** Pattern used to control retention or elimination of portions of another pattern.

Mbyte: Megabyte.

**memory map:** Map of memory space, partitioned into functional blocks.

**MPTCH:** Mask pitch register (B11). Defines the linear difference between starting addresses of adjacent rows in a mask array.

**MSB:** Most significant bit.

**MSbyte:** Most significant byte.

**MSGIN:** Message-in (bits 0—2 of HSTCTLL register).

**MSGOUT:** Message-out (bits 4—6 of HSTCTLL register).

**MSW:** Most significant word.

# Ν

M

NIL: Noninterlaced video enable (bit 14 of DPYCTL register). The value of NIL selects interlaced video timing (NIL=0) or noninterlaced video timing (NIL=1).

Appendix B

- NMI: Nonmaskable interrupt (bit 8 of HSTCTLH register). A host processor sets NMI to send a nonmaskable interrupt to the TMS34020.
- NMIM: Nonmaskable interrupt mode (bit 9 of HSTCTLH register). If NMIM=0, the TMS34020 saves the PC and ST contents on the stack before executing a nonmaskable interrupt routine. If NMIM=1, the TMS34020 discards the PC and ST contents before executing the NMI routine.

**nonmaskable interrupt:** Interrupt request that cannot be disabled.

- **NTSC:** National television system committee. Group representing a wide range of interests in the television broadcasting and video industry; NTSC is instrumental in developing graphics and video standards.
- **OFFSET:** XY-address offset register (B4). OFFSET contains the linear address of the 1<sup>st</sup> pixel in the XY-coordinate address space.
- **operand:** Any one of the quantities entering into or arising out of an operation.
- origin: Zero intersection of X and Y axes from which all points are calculated.
- **palette:** Digital look-up table used in a graphics display for translating data from the bitmap into the pixel values to be shown on the screen.
- **pan:** Apparent horizontal or vertical movement of a graphics screen or window over an image contained in a frame buffer that is too large to be completely displayed in a single static picture.

**PATTERN:** Fill-pattern register (B13).

- **PBH:** PIXBLT horizontal direction (bit 8 of CONTROL register). PBH=0 (default) selects left-to-right pixel processing; PBH=1 selects right-to-left processing.
- **PBV:** PIXBLT vertical direction (bit 9 of CONTROL register). PBV=0 (default) selects top-to-bottom pixel processing; PBV=1 selects bottom-to-top processing.
- **pending:** Requested but not yet performed. For example, a pending interrupt is an interrupt that has been requested but has not yet been serviced.

**PGA:** Pin grid array (type of chip package).

**PGMD:** Page-mode signal. Memory decode logic asserts PGMD low if the currently addressed memory supports page-mode accesses.

- **phase:** The time interval for each clock period in a system is divided into phases; one phase corresponds to the time when the clock signal is high, the other to the time that the signal is low.
- **PHIGS:** Programmer's hierarchical interactive graphics standard.
- **pipelining:** Design technique for reducing the effective propagation delay per operation by partitioning the operation into a series of stages, each of which performs a portion of the operation. A series of data is typically clocked through the pipeline in sequential fashion, advancing one stage per clock period.
- **pitch:** Difference in starting addresses of two adjacent rows of pixels in a 2-dimensional pixel array.
- **pixel:** Picture element. 1. Smallest controllable point of light on a display screen. 2. In a bitmapped display, the logical data structure that contains the attributes to be shown at the corresponding physical pixel position on a display screen.
- **pixel-processing option:** Boolean or arithmetic operation for combining two pixel values (source and destination); defined by PPOP[[CON-TROL]].
- **PIXBLT:** Pixel-block transfer. Pixel-array operation in which each pixel is represented by one or more bits. PIXBLTs are a superset of bitblts and include commonly-used Boolean functions as well as integer arithmetic and multi-bit operations.
- **plane:** (also bit plane or color plane) Bitmap layer in a multiple-bit-per-pixel display device. If the pixel size is *n* bits and the bits in each pixel are numbered 0 to n-1, plane 0 is made up of 0-numbered bits in all the pixels, and plane n-1 is made up of n-1-numbered bits in all the pixels. A layered graphics display allows planes or groups of planes to be manipulated independently of the other planes.
- **PMASK:** Plane mask registers (32-bit I/O register, address C000 0160h). PMASK contains a mask of 0s and 1s; the 1s represent protected destination bits, and the 0s represent modifiable destination bits.
- **PMASKL:** 16 LSBs of PMASK, accessed at address C000 0160h.
- **PMASKH:** 16 MSBs of PMASK, accessed at address C000 0170h.
- **PPOP:** Pixel-processing operation (bits 10—14 of CONTROL register). Selects a method for combining source and destination pixels. You can choose from 16 Boolean and 6 arithmetic operations; the default operation is S→D (source pixels replace destination pixels).
- **propagation delay:** Time required for a change in logic level at an input to a circuit to be translated into a resulting change at an output.

- **protocol:** Set of rules, formats, and procedures governing the exchange of information.
- pseudo-op: (pseudo-operation) An operation which is not part of the computer's operation repertoire as realized by hardware; hence, an extension of the set of machine operations.
- **PSIZE:** Pixel size register (16-bit I/O register, address C000 0150h). Defines the current pixel size as 1, 2, 4, 8, 16, or 32 bits.
- **pulse width:** Time interval between specified reference points on the leading and trailing edges of a pulse waveform.

**QFP:** Quad flat package (type of chip package).

**guarter phase:** One-fourth of a local-memory cycle.

- **R0**, **R1**: Bus request and control signals. These signals identify the type of request for use of the bus in a multiprocessor system.
- **RAM:** Random access memory. A memory from which all information can be obtained with approximately the same time delay by choosing an address randomly and without first searching through a vast amount of irrelevant data.

**RAS:** Row-address strobe. Drives the RAS inputs of DRAMs and VRAMs.

- raster: Rectangular grid of picture elements whose intensity levels are manipulated to represent images. In a bitmapped display, the bits within the frame buffer are mapped to the raster pattern of a display screen.
- **raster graphics:** Computed graphics in which a display image is composed of a pixel array arranged in rows and columns.
- **raster-op:** Arithmetic or logical combination that takes place during the transfer of a pixel array from one location to another.
- raster scan: Grid pattern traced by the electron beam on a display screen.
- RCA: Multiplexed row-/column-address bus (RCA0—RCA12). At the beginning of a memory-access cycle, identifies the row address for DRAMs; later in the cycle, the bus identifies the column address.
- RCM0, RCM1: RCA0-RCA12 row address configuration (bits 1&2 of CONFIG register). Determines which bits of the logical address are output on RCA0-RCA12 at row-address time.

- **ready signal:** Signal from a memory or memory-mapped peripheral that informs the processor when a memory cycle is about to complete. Slower memories and peripherals must extend the length of the memory cycle by negating the ready signal (in other words, by sending the processor a "not ready" signal) until the cycle can be completed.
- **REFADR:** Refresh pseudo-address register (16-bit I/O register, address C000 01F0h). Contains the address output during DRAM-refresh cycles.
- **refresh:** Method of restoring the charge capacitance to a memory device (such as a DRAM or VRAM) or of restoring memory contents.
- **request strobe:** Any control signal that begins or ends a read request or a write request.
- **reset:** Restore to normal action and initial conditions.
- **resolution:** Number of visible, distinguishable units in the device coordinate space.
- **retrace:** Line traced by the scanning beam(s) of a display screen as it travels from the end of one horizontal (or vertical) line or field to the beginning of the next horizontal (or vertical) line or field.
- **RGB monitor:** Red-green-blue monitor. Type of monitor capable of displaying colors; has separate inputs for the three signals that drive the red, green, and blue guns of a display.
- **relative coordinates:** Location of a point relative to the location of another point.
- **ROM:** Read-only memory.
- **rotate:** Transform an item or display by revolving it around an axis or center point.

row-address time: See address/status subcycle.

- **RR0—RR2:** Refresh rate (bits 10—12 of CONFIG register). Determines the frequency of DRAM refreshes.
- **RST:** Reset (bit 7 of HSTCTLH register). Setting this bit has the same effect as asserting RESET low; however, *only* the TMS34020 is reset (other devices in the system are not affected).
- **SADDR:** Source address register (B0). Contains the source array address for graphics instructions.
- SAM: Serial access memory or serial data register.

- **scale:** Size change made by multiplying or dividing coordinate dimensions by a scale factor (a constant value).
- **scan line:** Horizontal line traced across a display screen by the electron beam in a monitor or similar raster-scan device.
- **SCOUNT:** Shift clock counter register (16-bit I/O register, address C000 02C0h). During horizontal blanking, SCOUNT is loaded with the right-justified tap-point value and is then incremented once on the rising edge of each SCLK pulse.
- screen refresh: Operation of dumping the contents of the frame buffer to a CRT monitor in synchronization with the movement of the electron beam.
- scrolling: Moving a display vertically or horizontally.
- serial register transfer: Transfer between the RAM storage and internal serial register in a VRAM.
- **SETHCNT:** Set horizontal count register (16-bit I/O register, address C000 0310h). During external horizontal or composite video, SETHCNT is loaded into HCOUNT when HSYNC or CSYNC is pulsed.
- **setup time:** Minimum amount of time that valid data must be present at an input before the device is clocked; ensures proper data acceptance.
- **SETVCNT:** Set vertical count register (16-bit I/O register, address C000 0300h). During external horizontal or composite video, SETVCNT is loaded into VCOUNT when VSYNC or CSYNC is pulsed.
- **SF:** Special-function signal that drives a VRAM's DSF pin.
- **SIZE16:** Bus size signal. Memory decode logic may pull SIZE16 low if the currently addressed memory or port supports only 16-bit transfers.
- **SPTCH:** Source pitch register (B1). Defines the linear difference between starting addresses of adjacent rows in a source array.

**SRAM:** Static RAM.

- **SRE:** Screen-refresh enable (bit 12 of DPYCTL register). Setting SRE to 1 when video is enabled (ENV) enables screen-refresh cycles.
- **SRINC:** Screen-refresh address increment value (bits 5—31 of DINC registers). Defines the amount by which the address in SRNX is incremented after a screen-refresh cycle.
- **SRNX:** Next screen-refresh address (bits 5—31 of DPYNX registers). Represents the long-word address that is output during a screen-refresh cycle.
- **SRST:** Screen-refresh start address (bits 5—31 of DPYST registers). Contains the address of the pixel at the left of the 1<sup>st</sup> line displayed on the screen.

- **SSA:** Cache segment start address register.
- **SSV:** Split-shift-register midline-reload enable (bit 6 of DPYCTL register). Determines whether split-shift-register midline reload is disabled (SSV=0) or enabled (SSV=1 and SRE=1).
- **stairstepping:** Visual effect in bitmapped display devices; produces images by brightening or dimming individual pixels in a pixel array. Also called aliasing.
- **strobe:** Any control signal that begins or ends a memory access.
- **subsegment:** Block of 4 long words in a cache segment. Each of the 4 cache segments contains 8 subsegments, for a total of 32 long words per segment.
- T: Pixel transparency (bit 5 of CONTROL register). T=1 enables transparency; T=0 (default) disables transparency.
- **tap point:** Column address provided to a VRAM during a memory-to-serialregister cycle. The column address specifies the point at which the shift register is to be tapped; in other words, which cell of the serial register is to be connected to the VRAM's serial output.
- **TM:** Transparency mode (bits 0—2 of CONTROL register). Selects the transparency mode for pixel operations.
- trace: Line of the graphics display.
- transformation: Geometric alteration of a graphics display, such as scaling, translation, or rotation.
- **transparency:** Pixel attribute that renders a source pixel invisible so that portions of the destination array show through portions of the source array.
- **TR/QE:** Transfer/output enable signal. Drives the TR/QE input of VRAMs.
- VCE: Video capture enable (bit 7 of DPYCTL register). Selects memory-toregister screen-refresh cycles (VCE=0) or register-to memory screen-refresh cycles (VCE=1).
- **VCOUNT:** Vertical count register (16-bit I/O register, address C000 01C0h). VCOUNT counts the horizontal scan lines in the video display.
- **VEBLNK:** Vertical end blanking register (16-bit I/O register, address C000 0020h). VEBLNK defines the endpoint for the vertical blanking interval.

Appendix B

- VEN: VRAM internal register load enable (bit 8 of CONFIG register). VEN=1 enables the TMS34020 to use VRAMs with internal write-mask and color registers; VEN=0 (default) prohibits this.
- vertical back porch: Portion of vertical blanking that follows the trailing edge of the vertical-sync pulse.
- vertical-blanking interval: Time during which the display is blanked to cover the vertical retracing of an electron beam.
- vertical-blanking pulse: Positive or negative pulse developed during vertical retrace, appearing at the end of each field. Used to blank out scanning lines during the vertical-retrace interval.
- **vertical front porch:** Portion of a vertical-blanking pulse that precedes the leading edge of the vertical-sync pulse.
- **vertical sync:** Synchronization signal that enables vertical retrace of the electron beam of a display screen.
- **VESYNC:** Vertical end sync register (16-bit I/O register, address C000 0000h). VESYNC defines the endpoint of the vertical-sync pulse; in interlaced video, it also defines the endpoint of the 2<sup>nd</sup> equalization region.
- **VRAM:** Video RAM. A dual-ported memory device for computer graphics applications, containing two interfaces: one that allows a processor to read/write data from an internal memory array, a second that provides a serial stream of screen-refresh data to a display screen.
- **VSBLNK:** Vertical start blank register (16-bit I/O register, address C000 0040h). VSBLNK defines the startpoint for the vertical blanking interval.
- **VSD:** Vertical sync direction (bit 1 of DPYCTL register). Determines if VSYNC is configured as in input (VSD=0) or an output (VSD=1).
- **VTOTAL:** Vertical total register (16-bit I/O register, address C000 0060h). Number of horizontal scan lines in the display; defines the startpoint for the vertical-sync pulse.

W.

- W: Window checking (bits 6&7 of CONTROL register). Selects the action that the TMS34020 takes when a pixel operation would write a pixel inside or outside defined window limits.
- wait state: Clock period inserted into a memory cycle in order to permit accesses of slower memories and slower memory-mapped devices.
- **WE:** Write enable signal. Drives the WE inputs of DRAMs and VRAMs.

- **WEND:** Window ending address register (B6). WEND contains the XY address of the most significant pixel of the clipping window.
- **WSTART:** Window starting address register (B5). WSTART contains the XY address of the least significant pixel of the clipping window.
- window: Defined rectangular area of a virtual space on a display screen.
- **window checking:** Checking a pixel's address to see if it lies inside or outside the boundaries of a defined window.
- WVE: Window-violation interrupt enable (bit 11 of INTENB register). Setting WVE to 1 enables the window-violation interrupt.
- **WVP:** Window-violation interrupt pending (bit 11 of INTPEND register). WVP is set to 1 when a window-violation interrupt is requested.
- **X1E:** External interrupt 1 enable (bit 1 of INTENB register). Setting X1E to 1 enables external interrupt 1.
- **X2E:** External interrupt 2 enable (bit 2 of INTENB register). Setting X2E to 1 enables external interrupt 2.
- **X1P:** External interrupt 1 pending (bit 1 of INTPEND register). X1P is set to 1 when an external interrupt 1 is requested.
- **X2P:** External interrupt 2 pending (bit 2 of INTPEND register). X2P is set to 1 when an external interrupt 2 is requested.
- **YZCNT:** Y-zoom count (bits 0—4 of DPYNX registers). Determines when the address in SRNX can be incremented.
- **YZINC:** Y-zoom increment value (bits 0—4 of DINC registers). This value provides the increment value for the Y-zoom feature; valid values include 0, 2, 4, 8, 16, and 32.
- **Y-zoom:** TMS34020 feature that aids in display magnification.
- **zoom:** Scaling a display (or display item) so it is magnified or reduced on the screen.

Appendix B

Χ

Ζ



Phone Numbers -

### TI Customer Response Center (CRC) Hotline: (800) 232–3200

Graphics Hotline: (713) 274-2340

### General-Purpose Register Files -

	Registe	er File A		Register File I	в
	bit 31 MSB	bit 0 LSB	b M	it 31 bi ISB L	it 0 SB
A0			В0	SADDR	
A1			B1	SPTCH	
A2			B2	DADDR	
AЗ			ВЗ	DPTCH	
A4			B4	OFFSET	
A5			B5	WSTART	
A6			B6	WEND	
A7			B7	DYDX	
A8			B8	COLORO	
A9			В9	COLOR1	
A10			B10	MADDR	
A11			B11	MPTCH	
A12			B12	TEMP†	
A13			B13	PATTERN†	
A14			B14	TEMP†	
		SP stack	pointer SP		

The line instructions use these registers for a different purpose. Some graphics instructions use these registers as temporary registers.

### Initial State Following Reset

Immediately following reset,

- All I/O registers are cleared to 0000h. (Possible exceptions are HLT[[HSTCTLH]], REFADR, and SCOUNT).
- General-purpose register files A and B are uninitialized.
- The ST is set to 0000 0010h.
- The PC is uninitialized.
- The cache SSA registers are uninitialized.
- The cache LRU stack is set to the sequence 0, 1, 2, 3.
- All cache P flags are cleared.
- The DRAM refresh-pending counter is set to 9.

Register	Offset	HESYNC	0010h
BSFLTDH	0330h	HSTADRH	00E0h
BSFLTDL	0320h	HSTADRL	00D0h
BSFLTST	02D0h	HSTCTLH	0100h
CONFIG	01A0h	HSTCTLL	00F0h
CONTROL	00B0h or 0190h	HSTDATA	00C0h
CONVDP	0140h	HSBLNK	0050h
CONVMP	0180h	HTOTAL	0070h
CONVSP	0130h	іноѕт	0380h to 03F0h
DINCH	0250h	INTENB	0110h
DINCL	0240h	INTPEND	0120h
DPYADR	01E0h	PMASKH	0170h
DPYCTL	0080h	PMASKI	0160h
DPYINT	00A0h	PRIZE	0150h
DPYNXH	0230h	PSIZE	015011
DPYNXL	0220h	REFADR	01F0n
DPYMSK	02E0h	SCOUNT	02C0h
DPYSTH	0210h	SETHCNT	0310h
DPYSTL	0200h	SETVCNT	0300h
DPYSTRT	0090h	VCOUNT	01C0h
DPYTAP	01B0h	VEBLNK	0020h
HCOUNT	01D0h	VESYNC	0000h
HEBLNK	0030h	VSBLNK	0040h
HESERR	0270h	VTOTAL	0060h

Note: Register address = C000 000h + offset.

### CONTROL Register (C000 00B0h)

15	14 13	12	11	10	9	8	7	6	5	4	3	2	1	0
CD	l	POP	>	F	ΡВΗ	PBV		W	Т	A CALL AND	and a state of the		тм	
тм	000 001 100 101	trans resu trans sour trans trans dest	spar lt=0 spar cce= spar lt=0 spar .=C	rency COL rency rency OLO	/ on .OR / on / on R0	0	PE	3V BH	0 1 0 1	PIXE top t PIXE botto PIXE left to PIXE right	BLT p o bot BLT p om to BLT p o righ BLT p to le	oroco tom oroco oroco nt oroco ft	esse esse esse esse	s s s
т w	0 di 1 ei 00	sabl nable no w	es ti es tr /indo	rans. ans. owin	g		PP CI	ОР <b>)</b>	рі; 0 1	kel-pi enat disal	roces oles d bles (	ssing cach cach	g opt ie ne	ion
	01 10 11	winc winc winc	low low low	hit miss clip	5									

CONFI	G Reais	ter ((	2000	) 01	AOh	) —							
15 14	1 13 12	2 11	10	9	8	7	6	5	4	з	2	1	0
TRAFFIC TO A DESCRIPTION OF THE PARTY OF THE	and the state of t	RR		ARRAN ARRA	VEN	ARABAN ARA	and a state of the	Inder State	PARAL PROPERTY OF	CBP	RC	м	BEN
BEN	0 sele addr 1 sele	cts litt essin cts big	le-er g <i>(de</i> g-en	ndia efau diar	เท <i>เlt)</i> า	C	3P	1	write FIG no v	e-pro 's LSI vrite p	tects byte prote	CC	DN-
RCM	addr determ ical ac output time	essin ines Idress at ro	g whic s bi w-a	ts a ddre	og- are ess	VE RF	EN	0 1 D	syst feat syst cial- RAM	em h ure V em h featu refre	as s RAN as n re V esh r	pec /Is o s RA ate	ial- pe- Ms
DPYCI	TL Regis	ster (l	COO	0 00	)80h)	) —			<del>.</del>				
15 14 E N N I V L	13 12 R E	2 11 C S T	10	9 ************		7 V C E	6 S S V	5	4	3   C   V   D	2 C S D		0 H S D
HSD	0 HSY 1 HSY	NC is NC is	an i an i	npu outp	it but	VC	E	sc 1	reen men	-refre	esh n eg c	nod ycle	e es
VSD	0 <u>VSY</u> 1 <u>VSY</u>	NC is NC is	an i an c	npu outp	it but	CS	ST	0	reg- norn cycle	to-me nal pi es	em c xel-a	ycle acce	es ess
CSD	When 0 0 CSY 1 CSY	NC is	0, an i an c	npu outp	it but			1	pixe becc ter-t	l-acco ome s ransf	ess o seria er cy	cycl I-re vcle	es gis- s
CVD	selects HBLNK 0 selec 1 selec	CYSI ts CS	NC/ SYN BLNI	Ōĸ		SF	RE	0	disa scre enat frest	bles a en re bles s n whe	auto fresi cree en El	mat h en r NV:	ic e- =1
SSV	0 disat reloa 1 enab reloa	oles m d les m d whe	idlir idlin ən S	e RE:	=1	NI EN	L IV	0 1 0 1	inter noni blan enat	laceo nterla ks sc bles c	l vid aced reen lispla	eo vid ay	eo
INTEN	B Regis	ter (C	:000	01	10h)								
15 14	13 12	11	10	9	8	7	6	5	4	3	2	1	0
CARRON PROPERTY AND	NEAR ON DEAR OF THE PARTY OF TH	W V E	D I E	H I E	ALAN AND AND AND AND AND AND AND AND AND A	PART CARACTER	andanan anan	Patroneric Patroneric	AND DE	ANT AND	X 2 E	X 1 E	and the second
IE status bit must be enabled before these interrupts are enabled										led			
X1E	1 enat	les in	it. 1			DI	E	1	enal	bles d	displ	ay i	nt.
X2E HIE	1 enat 1 enat	oles in oles h	it. 2 ost i	nt.		W	VE	1	enal latio	bles v n int.	vind	ow-	vio-
INTPEND Register (C000 0120h)													
15 14	<u>  13 12</u>	11	10	9	8	7	6	5	4	3	2	1	0
and and a state of the state of	ANTAL AND	W V P	D   P	H I P	ALTERNAL AND ALTER	LANDAN BARA	and a state of the	and a state of the	TRANSPORT OF THE PARTY OF THE P	AND	X 2 P	X 1 P	A RANGE AND A

		<u></u>	4
X1P	1 int. 1 pending	DIP 1 display int. pending	
X2P	1 int. 2 pending	WVP 1 window-violation int	
HIP	<ol> <li>host int. pending</li> </ol>	pending	

#### HSTSTLH Register (C000 0100h) 15 14 13 12 11 10 9 8 7 6 5 4 3 2 0 1 R S T н Ν Ν HI HP HA L T CF M M HLB NC FW CK ĬЙ ï HA HP 0 '34020 is running 0 prefetch after any СК 1 '34020 is halted FW access 1 prefetch after writes HLB identifies last byte that HI 0 disables prefetch & host will access NC autoincrement 0 normal operation RST 1 enables prefetch & 1 reset '34020 autoincrement NMI 0 no NMI request 0 no effect CF 1 host requests NMI 1 flush cache NM 0 save context when HLT 0 allow '34020 to run 1 halt '34020 instructhere's an NMI IM 1 discard context

tion execution

### HSTSTLL Reaister (C000 00F0h) .

	J	7					
15 14	13 12 11 10 9 8	7 6	54	3	2	1	0
HB HB REN FI	HR E E E I I I I I I I I I I I I I I I I	NT MS	GOUT	INT IN	N	ISGI	N
MSG IN INT IN	message from host to '34020 0 no interrupt to '34020	EMI EN HRYI	0 no i 1 inte 0 hos trie	interr errupt et acc	upt i to h	to ho lost not	ost re-
MSG OUT	1 host interrupt re- quest to '34020 message from '34020 to host	HBFI	1 hos 0 hos faul 1 hos	it acc it acc ited it acc	ess ess ess	retri not faul	ed ted
INT OUT	<ul> <li>0 no interrupt to host</li> <li>1 '34020 interrupt request to host</li> </ul>	HB REN	If HR 0 no 1 inte	/I or I interr errupt	HBF upt f	I is a to ho nost	set, ost
EMG/ EMR	<ul> <li>00 no request, no interrupt</li> <li>01 host request from EMU, interrupt (if enabled)</li> </ul>	10 ho EN en 11 ho no	st relea /U, inte abled) st grant interru	sed t rrupt to E	oy (if MU,	,	

### LAD Bus Status Codes

Code	Bus Status	Туре
0000 0001 0010 0011	Coprocessor cycle Emulator operation Host cycle DRAM refresh	misc. (00xx)
0100 0101 0110 0111	Video-generated VRAM serial-register trans. CPU-generated VRAM serial-register trans. Write-mask load Color-register load	VRAM (01xx)
1000 1001 1010 1011 1100 1101 1110 1111	Data access Cache fill Instruction fetch Interrupt-vector fetch Bus-locked operation Pixel operation Block write Reserved	CPU (1xxx)

## Memory Map -

Address Range	Size	Use
FFFF FFE0h FFFF FBC0h	34 words	Interrupt & trap vectors
FFFF FBA0h FFFF E000h	222 words	Reserved for interrupt & extended trap vectors
FFFF DFE0h FFF0 0000h	32,512 words	General use & extended trap vectors
FFEF FFE0h C000 2000h	2 <sup>25_</sup> 33,024 words (35,521,408 words)	General use
C000 1FE0h C000 0400h	224 words	Reserved for I/O registers
C000 03E0h C000 0000h	32 words	I/O registers
BFFF FFE0h 0010 0000h	3×2 <sup>25</sup> –32K words (100,630,528 words)	General use
000F FFE0h 0000 0000h	32,768 words	General use & extended trap vectors

### Interrupt Priorities -

Interrupt	Priority	Source	Description
RESET	1	external/ internal	Device reset
BF	2	external	Bus fault interrupt
NMI	3	internal	Nonmaskable interrupt
н	4	internal	Host interrupt.
DI	5	internal	Display interrupt
WV	6	internal	Window violation interrupt
INT1	7	external	External interrupt 1
INT2	8	external	External interrupt 2
SS	9	internal	Single-step interrupt
ILLOP	10	internal	Illegal-opcode interrupt

### Vector Address Map -

Trap#	Address	Desc.	Trap#	Address	Desc.
32768 to 1	000F FFE0h to 0000 0000h	Applica- tion specific	12 to 15	FFFF FE60h to FFFF FDE0h	Re- served
0	FFFF FFE0h	RESET	16	FFFF FDE0h	Applica-
1	FFFF FFC0h	INT1	29	FFFF FC40h	specific
2	FFFF FFA0h	INT2	30	FFFF FC20h	ILLOP
3 to 7	FFFF FF80h to FFFF FE00h	Re- served	31	FFFF FC00h	Applica- tion specific
8	FFFF FEE0h	NMI	32	FFFF FBE0h	SS
9	FFFF FEC0h	н	33	FFFF FBC0h	BF
10	FFFF FEA0h	DI	34 to	FFFF FBA0h	Applica-
11	FFFF FE80h	wv	32767	FFF0 0000h	specific

## TMS34020 Assembly Language Instruction Set —

ABS Rd	CMPI IL,Rd
ADD Rs,Rd	СМРК
ADDC Rs,Rd	CMPXY Rs,Rd
ADDI IW,Rd	CPW Rs,Rd
ADDI IL,Rd	CVDXYL Rd
ADDK K,Rd	CVMXYL Rd
ADDXY Rs,Rd	CVSXYL Rs,Rd
ADDXYI IL,Rd	CVXYL Rs,Rd
AND Rs,Rd	DEC Rd
ANDI IL,Rd	DINT
ANDN Rs,Rd	DIVS Rs,Rd
ANDNI IL,Rd	DIVU Rs,Rd
BLMOVE S,D	DRAV Rs,Rd
BTST K,Rd	DSJ Rd,Address
BTST Rs,Rd	DSJEQ Rd,Address
CALL Rs	DSJNE Rd,Address
CALLA Addr	DSJS Rd,Address
CALLR Addr	EINT
CEXEC size, instruction[,ID]	EMU
CEXEC size, instruction[,ID]	EXGF Rd,F
CLIP	EXGPC Rd
CLR Rd	EXGPS Rd
CLRC	FILL L
CMOVCG Rd <sub>1</sub> [,Rd <sub>2</sub> [size]],	FILL XY
command[,ID]	FLINE {0   1}
CMOVCM *Rd+, transfers,	FPIXEQ
size,commandį, iDj	FPIXNE
command[.ID]	GETPC Rd
CMOVCS command[,ID]	GETPS Rd
CMOVGC Rs,command[,ID]	GETST Rd
CMOVGC Rs1, Rs2, size, com-	IDLE
mand [,ID]	INC
CMOVMC *Rs+,transfers,	JAcc Address
	JRcc Address
command[,ID]	JRcc Address
CMOVMC *Rs+.Rd.size.com-	JUMP Rs
mand [,ID]	LINE {0   1}
CMP Rs,Rd	LINIT
CMPI IW,Rd	LMO Rs,Rd
#### TMS34020 Assembly Language Instruction Set — (continued)

MMFM Rs,[,List]	NEGB Rd							
MMTM Rs,[,List]	NOP							
MODS Rs,Rd	NOT Rd							
MODU Rs,Rd	OR Rs,Rd							
MOVB Rs,*Rd	ORI IL,Rd							
MOVB *Rs,Rd	PFILL XY							
MOVB *Rs(Offset),Rd	PIXBLT B,L							
MOVB *Rs(SOffset),	PIXBLT B,XY							
	PIXBLT L,L							
MOVB AS, @DAddress	PIXBLT L,M,L							
NOVB @SAddress, Hu	PIXBLT L,XY							
@DAddress	PIXBLT XY,L							
MOVE Rs,Rd	PIXBLT XY,XY							
MOVE Rs,*Rd[,F]	PIXT Rs,*Rd							
MOVE Rs,-*Rd[,F]	PIXT Rs,*Rd .XY							
MOVE Rs,*Rd+[,F]	PIXT *Rs,Rd							
MOVE *Rs,Rd[,F]	PIXT *Rs,*Rd							
MOVE _*Rs,Rd[,F]	PIXT *Rs.XY,Rd							
MOVE *Rs+,Rd[,F]	PIXT *Rs.XY,*Rd.XY							
MOVE *Rs,*Rd[,F]	POPST							
<b>MOVE</b>	PUSHST							
MOVE *Rs+,*Rd+	PUTST Rs							
MOVE Rs,*Rd(Offset)[,F]	RETI							
MOVE *Rs(Offset),Rd[,F]	RETM							
MOVE *Rs(Offset),*Rd+[,F]	RETS [N]							
MOVE *Rs(SOffset), *Bd(DOffset)[ E]	REV Rd							
MOVE Rs.@DAddress[ Fl	RL K,Rd							
MOVE @SAddress.Rdl.Fl	RL Rs,Rd							
MOVE @SAddress.*Rd+I.Fl	RMO							
MOVE @SAddress.	RPIX Rd							
@DAddress[,F]	SETC							
MOVI IW,Rd	SETCDP							
MOVI IL,Rd	SETCMP							
MOVK K,Rd	SETCSP							
MOVX Rs,Rd	SETF FS,FE,F							
MOVY Rs,Rd	SEXT Rd,F							
MPYS Rs,Rd	SLA K,Rd							
MPYU Rs,Rd	SLA Rs,Rd							
MWAIT	SLL K,Rd							
NEG Rd	SLL Rs,Rd							

#### TMS34020 Assembly Language Instruction Set -(continued)

SRA K,Rd	SWAPF Rs,Rd,0
SRA Rs,Rd	TFILL XY
SRL K,Rd	TRAP N
SRL Rs,Rd	TRAPL
SUB Rs,Rd	VBLT
SUBB Rs,Rd	VFILL
SUBI IW,Rd	VLCOL
SUBI IL,Rd	XOR Rs,Rd
SUBK K,Rd	XORI IL,Rd
SUBXY Rs,Rd	ZEXT Rd,F

#### Boolean Pixel-Processing Options

00000	Source -> Destination
00001	Source AND Destination -> Destination
00010	Source AND ~Destination → Destination
00011	$0s \rightarrow Destination$
00100	Source OR ~Destination → Destination
00101	Source XNOR Destination → Destination
00110	~Destination -> Destination
00111	Source NOR Destination $\rightarrow$ Destination
01000	Source OR Destination -> Destination
01001	Destination -> Destination
01010	Source XOR Destination → Destination
01011	~Source AND Destination $\rightarrow$ Destination
01100	1s $\rightarrow$ Destination
01101	~Source OR Destination $\rightarrow$ Destination
01110	Source NAND Destination $\rightarrow$ Destination
01111	~Source → Destination

#### Arithmetic Pixel-Processing Options -

10000	Source + Destination -> Destination
10001	ADDS(Source, Destination) $\rightarrow$ Destination
10010	Destination – Source $\rightarrow$ Destination
10011	SUBS(Source, Destination) $\rightarrow$ Destination
10010	MAX(Source, Destination) $\rightarrow$ Destination
10101	MIN(Source, Destination) $\rightarrow$ Destination
10110-11111	Reserved

#### Status Register -

31	30	29	28	27	26	25	24	23	22	21	20	19	18	17	16
N	С	Ζ	V	Cast & Barry	BF	IX	a the same and	AND	SS	IE	and a second	and a substantial substant	ARRAN ARRAN	PERFERENCE STREET	RANGE AND AND AND A
Land Property in	PARTER PARTE	SANATA BARAN	AND DESCRIPTION OF THE OWNER OWNER OF THE OWNER	FE1			FS1			FE0			FS0		
15 Not	14 te:	13 Sha	12 ded	11 port	10 ions	9 are	8 rese	7 erve	6 d.	5	4	3	2	1	0

# Index

12-pin connector, mechanical dimensions, A-10 12-pin header, A-3 3-wire interface, 11-1

### A

A-file registers (A0-A14), 4-6 initial state following reset, 6-23 ABS instruction, 13-32 ABS ('34082 pseudo-op), 14-9, 14-10 ABSD ('34082 pseudo-op), 14-11 ABSF ('34082 pseudo-op), 14-12, 14-13 absolute addresses, 13-3 ADD instruction, 13-33 ADD ('34082 pseudo-op), 14-14, 14-15 ADDC instruction, 13-34 ADDD ('34082 pseudo-op), 14-16 ADDF ('34082 pseudo-op), 14-17, 14-18 ADDI 16-bit (short) version, 13-35 32-bit (long)version, 13-36 ADDK instruction, 13-37 address/status portion (local-memory cycle), 8-8, 8-12 addressing address latch, 2-11 autoincrementing (for host accesses), 4-59, 4-60, 7-12-7-15 big-endian, 3-20-3-25, 4-21, 7-44 comparison feature (for host accesses), 4-60, 7-12 display screen, 4-32, 4-40, 4-46 implicit addressing, 7-12 instruction words, in cache, 5-3, 5-5

linear addressing, 3-3, 3-15 little-endian, 3-20-3-25, 4-21, 7-44 local memory, 3-3 multiplexed addressing, 8-51-8-53 nonmultiplexed addressing, 8-50 modes. 13-2-13-9 absolute addresses, 13-3 constants, 13-2 immediate values, 13-2 register-direct, 13-4 reaister-indirect. 13-5 in XY mode, 13-9 with offset, 13-6 with postincrement, 13-7 with predecrement, 13-8 multiplexing, 4-22 pixel arrays, 4-30, 4-79 prefetching (for host accesses), 4-60, 7-10-7-12 range, 3-3 RCA values at row-address time, 4-21 screen-refresh address, 4-41, 4-42, 4-78 segments within the cache, 5-2, 5-3 subsegments within a cache segment, 5-2, 5-3 tap point, 4-44, 4-45 two 16-bit registers as a 32-bit register, 4-15, 4-32, 4-46, 4-75 window end address, 4-90 start address, 4-91 XY addressing, 3-14, 4-25, 4-28, 4-34, 4-50, 4-73, 4-90, 4-91 XY-to-linear conversion, 3-15---3-17, 4-28, 4-34, 4-72, 4-83 ADDXY instruction, 13-38 ADDXYI instruction, 13-39

algorithms cache control, 5-3 replacement, 5-4 display pitch, 3-13 least-recently-used (cache replacement), 5-4 XY-to-linear conversion, 3-15 ALTCH signal, 2-9, 2-11, 8-2, 10-2 American video standards NTSC, 9-27 RS-170, 9-27 AND instruction, 13-40 ANDI instruction, 13-41 ANDN instruction, 13-42 ANDNI instruction, 13-43 ANSI C, 1-11 applications of the TMS34020, 1-3 arbitration logic examples, 11-15-11-18 multiprocessor systems, 11-13-11-15 archiver. 1-11 arithmetic instructions, 13-24 ABS, 13-32 ADD, 13-33 ADDC, 13-34 ADDI (16 bits), 13-35 ADDI (32 bits), 13-36 ADDK, 13-37 ADDXY, 13-38 ADDXYI, 13-39 DEC, 13-94 DIVS, 13-96-13-97 DIVU, 13-98-13-99 INC, 13-134 MODS, 13-152 MODU, 13-153-13-157 MPYS, 13-172-13-174 MPYU, 13-175-13-176 SUB, 13-241 SUBB, 13-242 SUBI, 13-243, 13-244 SUBK, 13-245 SUBXY, 13-246 arithmetic pixel-processing options, 4-26 array sizes for DRAMs, 8-52 arrays. See pixel arrays assembler, 1-11 assembly-language, tools, 1-10-1-13, 3-24

autoincrementing, 7-12—7-15 disabled, 7-14 legal HBS combinations, 7-13 reads and writes, 7-14 writes only, 7-14 auxilary graphics instructions CLIP, 13-55 FPIXEQ, 13-126—13-127 FPIXNE, 13-128—13-129 PFILL, 13-184—13-189 RPIX, 13-225 TFILL, 13-249—13-252 VBLT, 13-259—13-261 VFILL, 13-262—13-263 VLCOL, 13-264—13-265

## B

B-file registers (B0—B14), 4-6, 4-7, 4-8 COLOR0, 4-18, 4-74 COLOR1, 4-19, 4-74 DADDR, 4-30 **DPTCH**, 4-34 DYDX, 4-50 initial state following reset, 6-23 MADDR, 4-71 MPTCH, 4-72 OFFSET, 4-73 PATTERN, 4-74 SADDR, 4-79 SPTCH, 4-83 WEND, 4-90 **WSTART, 4-91** background color, 4-18, 4-74 bandwidth, host interface, 7-34-7-36 bank selects, 8-57 BEN bit, 3-20, 4-21, 8-4 write protecting the bit, 4-22 BF (bus fault) status bit, 4-3, 6-3, 6-19 big-endian addressing, 3-20-3-25 assembling code for, 3-24-3-25 default at reset, 3-20, 4-21 effect of BEN bit, 3-20, 4-21 host interface, 7-44 instruction timing, 3-25-3-26 processors that use it, 3-20 selecting, 3-20, 4-21 binary PIXBLTs use of COLOR0, 4-18 use of COLOR1, 4-19

blanking composite blanking, CBLNK, 2-15 horizontal blanking, 9-9 ending (HEBLNK), 4-53 HBLNK, 2-15 starting (HSBLNK), 4-66 vertical blanking, 9-9 ending (VEBLNK), 4-86 starting (VSBLNK), 4-88 VBLNK, 2-15 BLMOVE instruction, 13-44-13-45 implied operands DADDR, 4-30 SADDR. 4-79 block of pixels. See arrays block accesses reads of TMS34020 memory (by host), 4-59, 4-60 writes, 4-22 to TMS34020 memory (by host), 4-59, 4-60 with mask, 4-22 block diagram, TMS34020, 1-5 block-write cycles data expansion, 8-42 data mapping, 8-41 status code on local-memory cycle, 8-11 Boolean pixel-processing options, 4-26 branch instructions, effects on PC, 4-4 breakpoints, 6-28 British video standards PAL, 9-27 BSFLTD registers, 4-15-4-17, 6-19 BSFLTDH, 4-15-4-17, 6-19 BSFLTDL, 4-15-4-17, 6-19 BSFLTST register, 4-17, 6-19 BTST constant version, 13-46 register version, 13-47 buffer delays for emulator connections, A-5 bulk initialization, 9-47 bus error/bus fault, 2-11, 7-9 bus-fault interrupt, 6-19-6-20, 7-9 service routine, 6-20 coprocessor cycles, 10-9 CPU-initiated access, 8-14 host-initiated access, 8-14 local-memory cycles, 8-14 on a host-initiated access, 4-64

screen-refresh cycle, 8-14 use of BSFLST to save memory controller state, 4-17 use of BSFLTD to store LAD data, 4-15-4-17 bus-fault interrupt, priority, 6-7 bus-locked operation and dynamic bus sizing, 8-29 status code on local-memory cycle, 8-11 bus-request codes access termination, 11-5-11-12 high-priority request, 11-5-11-12 low-priority request, 11-5-11-12 no request, 11-5-11-12 bus-requests priorities, 2-13, 8-6 bus size signal (SIZE16), 2-11 BUSFLT signal, 2-9, 2-11, 6-2, 6-19, 7-9, 8-2, 8-12, 8-18, 10-2 bus cycle completion codes, 2-12 byte-select strobes, 4-57, 7-2 big-endian addressing, 7-44 little-endian addressing, 7-44 bytes, 3-1



C (carry) status bit, 4-3 C compiler, 1-11, 1-13 cache, 5-1-5-12 accessible words, 5-3 architecture, 5-2 bypassing the cache, 5-8 cache fill, status code on local-memory cycle, 8-11 cache hit, 5-5 cache miss, 5-5 segment miss, 5-6 subsegment miss, 5-5 CD (cache disable) bit, 5-8 CF (cache flush) bit, 5-8 control algorithm, 5-3 disabling the cache, 4-27, 5-8 downloading new code from a host, 5-8 fetching data after a cache miss, 5-6 flushing the cache (CF), 4-61, 5-8 initial state following reset, 6-23 internal parallelism, 5-10 least-recently-used algorithm, 5-4 operation, 5-5-5-8 organization, 5-2

P flags, 5-2, 5-4 performance when enabled vs. disabled, 5-9 reason it's provided, 5-1 replacement algorithm, 5-4 segments, 5-2 self-modifying code, 5-8 setting the CD bit, 5-8 setting the HLT bit, 5-8 size, 5-3 SSA registers, 5-2 subsegments, 5-2 CALL instruction, 13-48 CALLA instruction, 13-49 CALLR instruction, 13-50 CAMD signal, 2-9, 2-12, 8-2, 8-18 capturing a video image, 9-48 Cartesian coordinates, 3-14, 3-19 CAS0---CAS3 signals, 2-9, 2-12, 8-2, 10-2 CBLNK/VBLNK signal, 2-10, 2-15, 9-2 selection, 4-38 CBP bit, 3-20, 4-22, 8-4 CD bit, 4-27, 5-8 CEXEC instruction, 13-51-13-93 CF bit, 4-57, 4-61, 5-8, 7-4 CHECK ('34082 pseudo-op), 14-19 CL30, 1-11 CLIP instruction, 13-55-13-56 implied operands, DADDR, 4-30 CLKIN signal, 2-10, 2-16, 8-2 clocks CLKIN (clock in), 2-16 LCLK1, LCLK2 (local output clocks), 2-16 SCLK (serial data clock), 2-15 VCLK (video clock), 2-15 CLR instruction, 13-57 CLRC instruction, 13-58 CMOVCG instruction, 13-59-13-60 CMOVCM instruction, 13-61-13-62, 13-63-13-65 CMOVCS instruction, 13-66 CMOVGC instruction, 13-67-13-68, 13-69-13-70 CMOVMC instruction, 13-71-13-73, 13-74-13-77, 13-78-13-79 CMP instruction, 13-80 CMP ('34082 pseudo-op), 14-20, 14-21 CMPD ('34082 pseudo-op), 14-22 CMPF ('34082 pseudo-op), 14-23, 14-24

CMPI instruction, 13-81, 13-82 CMPK instruction, 13-83 CMPXY instruction, 13-84 code debugging, single-step mode, 6-28---6-32 downloading new code from a host, 5-8, 7-32 restrictions for compatibility between TMS34010 and TMS34020, 1-17 self-modifying, effects on instruction cache, 5-8 COFF. 1-11 color-latch register loads, status code on localmemory cycle, 8-11 COLOR0 register, 4-18, 4-74 COLOR1 register, 4-19, 4-74 column address bus. 2-12 mode, 2-12 strobes, 2-12 column-address time, 4-21, 8-9 compare instructions, 13-24 BTST (constant), 13-46-13-50 BTST (register), 13-47-13-50 CMP, 13-80 CMPI, 13-81-13-93 CMPK. 13-83-13-93 CMPXY, 13-84 CPW, 13-85-13-86 compatibility with future GSPs local-memory read & write cycles, 8-19 status register values, 4-3 with the TMS34010, 1-16-1-18 code restrictions, 1-17-1-18 CONTROL register, 4-24 DPYADR register, 4-35 DPYSTRT register, 4-48 DPYTAP register, 4-49 HSTADRH register, 4-56 HSTDATA register, 4-65 screen-refresh registers, 9-8 completing a successful local-memory cycle, 8-13 composite video, 9-15-9-17 display example, 9-40-9-42 enabling/disabling, 4-38 equalization pulses, 9-15-9-16 serration pulses, 9-15-9-16

sync direction, 4-37 condition codes for jump instructions, 13-26

Index-4

CONFIG register, 4-20-4-24, 8-4 BEN bit, 3-20, 4-20, 4-21, 8-4 CBP bit, 3-20, 4-20, 4-22, 8-4 RCM bits, 4-20, 4-21, 8-4 RR bits, 4-20, 4-23, 8-4 VEN bit, 8-4 write protecting the register, 4-22 constants, 13-2 context-switching instructions, 13-25-13-27 CALL, 13-48 CALLA, 13-49 CALLR, 13-50 RETI, 13-217-13-218 RETS, 13-220 TRAP L, 13-256—13-258 TRAP N, 13-253-13-255 CONTROL register, 4-24-4-28 CD bit, 4-24, 4-27, 5-8 compatibility with TMS34010, 4-24 PBH bit, 4-24, 4-25 PBV bit, 4-24, 4-26 PPOP bits, 4-24, 4-26-4-27 T bit, 4-24, 4-25 TM bits, 4-24 VEN bit, 4-22 W bits, 4-24, 4-25, 6-17 CONVDP register, 4-28-4-30 SETCDP instruction, 4-28 XY-to-linear conversion, 3-15, 3-16 converting. . . an XY address to a linear address, 3-15-3-17 composite video signals to separate signals, 9-34 pixel access into register transfers, 9-47 separate video signals to a composite signal, 9-34 CONVMP register, 4-28-4-30 SETCMP instruction, 4-28 XY-to-linear conversion, 3-15, 3-16 CONVSP register, 4-28-4-30 SETCSP instruction, 4-28 XY-to-linear conversion, 3-15, 3-16 coprocessor interface, 10-1-10-18 aborts, 10-17 general coprocessor commands command field, 10-6-10-7 format. 10-5-10-7 ID field, 10-5-10-7 parameter size, 10-6-10-7 general coprocessor instructions, 10-3-10-4

local-memory cycles, 10-4, 10-8-10-16 bus faults, 10-9 ending, 10-9 inserting wait states, 10-9 retrying, 10-9 overview, 10-3 passing commands to a coprocessor, 10-8 signals, 10-2-10-18 ALTCH, 10-2 BUSFLT, 10-2 CAS0--CAS3, 10-2 LAD0-LAD31, 10-2 LCLK1, LCLK2, 10-2 LINT1, LINT2, 10-2 LRDY. 10-2 SF, 10-2 WE. 10-2 status checks, 10-17 status code on local-memory cycle, 8-10 system configuration, 10-18 TMS34082, 14-1-14-7 TMS34082 pseudo-ops, 10-3 transferring data, 10-8 coprocessor to local memory, 10-15 coprocessor to TMS34020 register, 10-12 local memory to coprocessor, 10-14 seauence, 10-9 TMS34020 register to coprocessor, 10-11 CPW instruction, 13-85-13-86 implied operands WEND. 4-90 WSTART, 4-91 CSD bit, **4-37**, 9-6 CST bit. **4-39**. 9-6 effect on local-memory cyles, 8-30, 8-33, 8-36 **CSYNC** signal equalization pulses, 9-17 selecting as input or output, 4-37 serration pulses, 9-16 CSYNC/HBLNK signal, 2-10, 2-15, 9-2 selection, 4-38 CVD bit, **4-38**, 9-6 CVDF ('34082 pseudo-op), 14-25 CVDI ('34082 pseudo-op), 14-26 CVDXYL instruction, 13-87-13-88 implied operands CONVDP, 4-29 DPTCH, 4-34 PSIZE, 4-77

CVFD ('34082 pseudo-op), 14-27, 14-28 CVFI ('34082 pseudo-op), 14-29, 14-30 CVID ('34082 pseudo-op), 14-31 CVIF ('34082 pseudo-op), 14-32, 14-33, 14-34 CVMXYL instruction, 13-89-13-90 implied operands CONVMP, 4-29 MPTCH, 4-72 PSIZE, 4-77 CVSXYL instruction, 13-91 implied operands CONVSP, 4-29 PSIZE. 4-77 SPTCH, 4-83 CVXYL instruction, 3-16, 13-92-13-93 implied operands CONVDP, 4-29 DPTCH. 4-34 **OFFSET**, 4-73 PSIZE, 4-77

### D

DADDR register, 4-30 with DYDX for common rectangle function, 4-30 data access, status code on local-memory cycle, 8-11 expansion, 8-37 mapping, during block-write cycles, 8-41 structures, 3-1-3-32 bvtes. 3-1 fields, 3-1, 3-3, 3-5 pixel arrays, 3-1, 3-18-3-19 pixels, 3-1, 3-10-3-13 stacks. 3-26 subcycle (local-memory cycle), 8-12 data portion (local-memory cycle), 8-8, 8-9 DDIN signal, 2-9, 2-11, 8-2 DDOUT signal, 2-9, 2-11, 8-2, 8-18 debugging, A-1 debugging code in single-step mode, 6-28-6-32 DEC instruction, 13-94 delays... buffer delays in emulation, A-5 recognizing interrupts, 6-11 to host accesses, 7-37-7-40 to video synchronization, 9-33 design considerations, for emulation, A-7

destination pitch CONVDP register, 4-28---4-30 conversion factor, 4-28-4-30 DPTCH register, 4-34-4-35 development tools overview, 1-10-1-13 DIE bit, 4-69, 6-3 DINC registers, 3-11, 4-32, 9-7 SRINC bits, 4-32, 4-33, 9-7 YZINC bits, 4-32, 4-33, 9-7 DINCH, DINCL. See DINC registers DINT instruction, 13-95 DIP bit, 4-70, 6-4, 6-17 direct operands, 13-4 display address output during a screen refresh, 4-42 blanking ration (DBR), 9-36 control, 4-36-4-41 increment value, 4-32 interrupt DPYINT register, 4-41 enabling, 4-69 pending indication, 4-70 mask, 4-44-4-46 memory, 8-56 coordinates, 3-13 dimensions, 3-12 requirements for hardware, 8-56 requirements for multiplexed addressing, 8-54 panning, 9-57 pitch, 3-13 screen origin alternate. 3-12 default, 3-12 screen sizes, 9-36 start address, 4-46-4-48 display interrupt, 6-17, 9-37 disabling, 6-6 enabling, 6-6 priority, 6-7 trap number, 6-16 vector address, 6-8, 6-16 DIVD ('34082 pseudo-op), 14-35 DIVF ('34082 pseudo-op), 14-36, 14-37 DIVS instruction, 13-96-13-97 DIVS ('34082 pseudo-op), 14-38, 14-39 DIVU instruction, 13-98-13-99 dot clock. 9-36 downloading new code from a host, 5-8

DPTCH register, 4-34-4-35 XY-to-linear conversion, 3-15 DPYADR register, 4-35 DPYCTL register, 4-36-4-41, 8-4, 9-5 CSD bit, 4-36, 4-37, 9-6 CST bit, 4-36, 4-39, 9-6 CVD bit, 4-36, 4-38, 9-6 ENV bit, 4-36, 4-40, 9-6 HSD bit, 4-36, 9-5 NIL bit, 4-36, 4-40, 9-6 SRE bit, 4-36, 4-40, 9-6 SSV bit, 4-36, 4-38, 9-6 VCE bit, 4-36, 4-39, 9-6 VSD bit, 4-36, 4-37, 9-6 DPYINT register, 4-41-4-42, 6-17 DPYMSK register, 4-44-4-46, 8-58, 9-8 and SRST or SRNX, 9-55 DPYNX registers, 4-42-4-44, 9-7 increment value, 4-32 SRNX bits, 4-42, 4-43, 9-7 YZCNT bits, 4-42, 9-7 DPYNXH, DPYNXL. See DPYNXL registers DPYST registers, **4-46—4-48**, 9-7 SRST bits, 4-46 DPYSTH, DPYSTL. See DPYST registers DPYSTRT register, 3-11, 4-48 DPYTAP register, 4-49 DRAM/VRAM interface, 8-1-8-60 block-mask local-memory cycles, 8-37—8-43 DRAM-refresh local-memory cycles, 8-44-8-45 serial-register transfers, 8-29-8-33 signals, 2-12, 8-2-8-3 CAMD, 2-12, 8-2 PGMD, 8-3 RAS, 2-12, 8-3 RCA0—RCA12, 2-12, 8-3 SF, 2-12, 8-3 SIZE16, 8-3 TR/QE, 2-12, 8-3 WE, 2-12, 8-3 write-mask local-memory cycles, 8-34-8-36 DRAMs array sizes, 8-52 CAS-before-RAS cycles, 4-78 refreshes, 4-78, 8-6, 8-44 status code on local-memory cycle, 8-10 selecting the refresh rate, 4-23

DRAV instruction, 13-100-13-102 implied operands COLOR1, 4-19 CONTROL, 4-27 CONVDP. 4-29 DPTCH. 4-34 OFFSET, 4-73 PMASK, 4-76 PSIZE, 4-77 WEND. 4-90 WSTART, 4-91 DSJ instruction, 13-103 DSJEQ instruction, 13-104-13-105 DSJNE instruction, 13-106-13-107 DSJS instruction, 13-108 DYDX register, 3-18, 4-50-4-52 with DADDR for common rectangle function, 4-30, 4-50 dynamic bus sizing and bus-locked operation, 8-29 data transfers, 8-26 page mode, 8-28 SIZE16 signals, 2-11

## Ε

EINT instruction, 13-109 EMG bit, **4-63**, 7-4, A-7 EMIEN bit, **4-64**, 7-4, A-7 EMR bit, **4-63**, 7-4, A-7 EMU instruction, 13-110 EMU0-EMU3 signals, 2-10, A-3, A-4, A-6 emulation buffer delays, A-5 design considerations, A-1-A-10, A-7 emulator connector. A-3 host communications, 4-63, A-7 inhibiting the host-interface port, 4-63 mechanical dimensions 12-pin connector, A-10 pod, A-9 target cable, A-9 overview of an emulation system, A-2 pod interface, A-6 preventing the host from accessing local memory, 4-63 requesting local memory, 8-7 reset and interrupts, A-7

signals buffering, A-4 EMU0---EMU3, A-3, A-4, A-6 status code on local-memory cycle, 8-10 endian addressing modes. See big-endian addressing ENV bit, 4-40, 9-6 equalization pulses, 9-15-9-16 on CSYNC, 9-17 European video standards PAL (British), 9-27 SECAM (French), 9-27 even field (interlaced video), 9-21 EXGF instruction, 13-111 EXGPC instruction, 13-112 EXGPS instruction, 13-113 extending a local-memory cycle with wait states, 8-12 external interrupts, 6-15 disabling, 6-6 enabling, 4-69, 6-6 pending indications, 4-70 priority, 6-7 recognition delay, 6-11 source, 6-15 vector addresses, 6-8, 6-15 external synchronization, 9-29-9-35 composite sync, 9-30-9-35 conversion, 9-34 horizontal sync, 9-30-9-35 interlaced video, 9-30-9-35 odd/even field alignment, 9-31---9-35 noninterlaced video, 9-30-9-35 vertical sync, 9-30-9-35 external syncrhonization loading the video counters, 9-32 pulse widths, 9-35 syncing to VCLK, 9-32

# F

fast fills, 8-37 FE0 (field extension 0) status bit, **4-2** FE1 (field extension 1) status bit, **4-2** features, of the TMS34020, 1-2 fields, 3-1, 3-3, **3-5—3-9** alignment in memory, 3-7 *aligned to 1-byte boundary, 3-7* 

aligned to 2-byte boundaries, 3-6 straddling a word and aligned on 2 byte boundaries. 3-7 straddling a word and aligned to 1 byte boundary, 3-8 straddling a word and not byte aligned, 3-8 extraction, 3-6 field 0, 3-5 FE0 (field extension) bit, 3-5, 4-2 field size decoding, 3-5 FS0 (field size) bits, 3-5, 4-2 sign-extending, 4-2 zero-extending, 4-2 field 1. 3-5 FE1 (field extension) bit, 3-5, 4-2 field size decoding, 3-5 FS1 (field size) bits, 3-5 sian-extending, 4-2 zero-extendina, 4-2 field extension sign-extending, 4-2 zero-extending, 4-2 in a general-purpose register, 3-5 insertion, 3-6, 3-8, 3-9 pixels, DPYSTRT register, 3-11 PSIZE register, XY-to-linear conversion, 3-15 reading, 3-5 size, 3-5 starting address, 3-5 storage in external memory, 3-6 writing, 3-5 **FILL instructions** FILL L. 13-114-13-116 implied operands COLOR1, 4-19 CONTROL, 4-27 DADDR, 4-30 DPTCH, 4-34 DYDX, 4-50 **PMASK, 4-76** PSIZE, 4-77 FILL XY, 13-117-13-120 implied operands COLOR1, 4-19 CONTROL, 4-27 CONVDP, 4-29 DADDR, 4-30 **DPTCH**, 4-34 DYDX, 4-50 OFFSET, 4-73 PMASK, 4-76 **PSIZE, 4-77** 

WEND, 4-90 WSTART, 4-91 source address, 4-30, 4-79 FLINE instruction, 13-121-13-125 destination address, 4-30 implied operands COLOR0, 4-18 COLOR1, 4-19 CONTROL, 4-27 CONVDP, 4-29 DADDR, 4-30 DPTCH, 4-34 DYDX, 4-50 MPTCH, 4-72 PATTERN. 4-74 PMASK, 4-76 PSIZE, 4-77 SADDR, 4-79 WEND, 4-90 WSTART, 4-91 source address, 4-79 flushing the cache, 4-61, 5-8 foreground color, 4-19, 4-74 FPIXEQ instruction, 13-126—13-127 implied operands COLOR0, 4-18 MPTCH, 4-72 PMASK, 4-76 PSIZE, 4-77 FPIXNE instruction, 13-128-13-129 implied operands COLOR0, 4-18 MPTCH, 4-72 PMASK, 4-76 PSIZE, 4-77 French video standards SECAM, 9-27

# G

general-purpose coprocessor instructions CEXEC, 13-51—13-93 CMOVCG, 13-59—13-60 CMOVCM, 13-61—13-62, 13-63—13-65 CMOVCS, 13-66 CMOVGC, 13-67—13-68, 13-69—13-70 CMOVMC, 13-71—13-73, 13-74—13-77, 13-78—13-79

general-purpose register files. See register files

GETCST ('34082 pseudo-op), 14-40 GETPC instruction, 13-130 GETPS instruction, 13-131 GETST instruction, 13-132 GI signal, 2-9, 2-13, 8-18, 11-2 graphics instructions CPW, 13-85-13-86 CVXYL, 13-92-13-93 destination address, 4-30 DRAV, 13-100-13-102 FILL L, 13-114-13-116 FILL XY, 13-117-13-120 FLINE, 13-121-13-125 interrupts, 6-13-6-14 LINE, 13-142-13-145 LINIT, 13-146 LMO, 13-147 PIXBLT instructions, 13-190-13-205 PIXT instructions. 13-206—13-213 source address, 4-79 graphics operations interrupts, 6-13-6-14 PIXBLT direction, 4-25, 4-26 pixel size, 4-77 pixel-processing operations arithmetic options, 4-26 Boolean options, 4-26 selecting, 4-26-4-27 plane masking, 4-75 transparency, 4-24, 4-25 window checking, 4-25, 4-90, 4-91

## H

HA5—HA31 signals, 2-10, 2-14, 7-2, 7-7
HACK bit, 4-57, 4-57, 7-3
halt latency, 7-39
halting TMS34020 execution, 7-32

acknowledging the halt state, 4-57
HLT bit, 4-61

HBFI bit, 4-64, 6-5, 6-21, 7-5, 7-9
HBREN bit, 4-64, 6-5, 6-21, 7-5, 7-9
HBS0—HBS3 signals, 2-10, 2-14, 7-2, 7-7
HCOUNT register, 4-52—4-53, 9-4

external synchronization, 9-29
loading with the SETHCNT value, 4-81
HCS signal, 2-10, 2-14, 7-2, 7-7
HDST signal, 7-2

HDST signal, 2-10, 2-14 HEBLNK register, 4-53-4-54, 9-4 HESERR register, 4-54-4-55, 9-4 HESYNC register, 4-55-4-57, 9-4 HIE bit. 4-69, 6-3 HINC bit, 4-57, 4-60, 7-4 effects on address comparison, 7-10 effects on autoincrementing, 7-13 effects on prefetching, 7-10 interaction with HPFW, 4-59, 7-10 HINT signal, 2-10, 2-14, 4-64, 6-2, 6-21, 7-2, 7-9 HIP bit, 4-70, 6-4, 6-17 HLB bits (HLB0-HLB1), 4-57, 4-57, 7-3 effects on prefetching, 7-11 HLT bit, 4-57, 4-61, 6-4, 6-22, 7-4 setting for downloading new code, 5-8 software reset, 7-32 HOE signal, 2-10, 2-14, 7-2 horizontal back porch, 9-10 blanking minimum duration, screen refreshes, 9-51 screen refreshes, 9-42 front porch, 9-10 video timing (internal), 9-11-9-12 horizontal blanking, 9-9 VRAM tap point, 4-80 horizontal sync, 9-9 direction, 4-36 horizontal timing HCOUNT register, 4-52 HEBLNK register, 4-53 HESERR register, 4-54 HESYNC register, 4-55 HSBLNK register, 4-66 HTOTAL register, 4-67 SETHCNT register, 4-81 VEBLNK register, 4-86 host interface, 7-1-7-44 access delays, 7-37-7-40 address identification, 4-57, 7-7 autoincrementing, 7-12-7-15 bandwidth, 7-34 optimizing, 7-35 basic communication, 7-7-7-9 big-endian addressing, 7-44 block diagram, 7-6 buffering messages, 4-62

bus fault indication, 4-64 byte-select strobes, 2-14, 4-57, 7-7 illustration, 7-8 chip-select, 2-14 completing a host access, 7-16-7-17 data latch output enable, 2-14 strobe, 2-14 default cycle, 7-15, 8-7 downloading new code from host, 7-32 emulation considerations, A-7 emulator communications, 4-63 features that improve performance, 7-10 address comparison, 4-60-4-61, 7-12 autoincrementing, 4-60-4-61, 7-12 host-default cycle, 7-15 prefetching, 4-59-4-61, 7-10 halt latency, 7-39 implicit addressing, 7-12-7-15 interrupts, 4-58, 4-64, 6-16, 6-21, 7-9 enabling, 4-69 HINT, 2-14 message to host, 4-63 message to TMS34020, 4-62 pending indication, 4-70 little-endian addressing, 7-44 messages, 4-62 multiple-TMS34020 system, 7-40-7-41 prefetching data, 7-10 read cycles, 7-8 back-to-back with autoincrementing, HREAD as strobe, 7-23 back-to-back with prefetching, HCS as strobe, 7-22 single read from I/O register, HREAD as strobe, 7-20 single read, 1 wait state, HCS as strobe, 7-21 single read, HCS as strobe, 7-19 successive reads to same location, HCS and HREAD as strobes, 7-24 read strobe, 2-14 registers CONFIG, 4-20 HSTCTLH, 4-57-4-62, 7-3 HSTCTLL, 4-62-4-65, 7-4 retry indication, 4-64, 7-9 signals, 2-13 BUSFLT, 7-9 HA5-HA31, 2-14, 7-2, 7-7 HBS0-HBS3, 2-14, 7-2, 7-7

HCS, 2-14, 7-2 HDST, 7-2 HDST, 2-14 HINT, 2-14, 7-2, 7-9 HOE, 2-14, 7-2 HRDY, 2-14, 7-2 HREAD, 2-14, 7-2 HWRITE, 2-14, 7-2 LRDY, 7-9 status code on local-memory cycle, 8-10 synchronizing host requests, 7-35 systems with 16-bit memory devices, 7-42-7-43 timing examples, 7-18-7-31 TMS34020 acknowledges halt, 4-57 use of page mode, 8-24 worst-case delay, 7-37 bus-master arbitration, 7-38 CPU cycles, 7-39 DRAM-refresh cycles, 7-38 host request syncronization, 7-38 previous host cycle, 7-38 screen-refresh cycles, 7-38 write cycles, 7-9 back-to-back writes with autoincrementing, HWRITE as strobe. 7-29 back-to-back writes with prefetching & autoincrementing, HREAD and HWRITE as strobe, 7-31 back-to-back writes with prefetching, HCS as strobe, 7-30 back-to-back writes, HCS as strobe, 7-28 single write to I/O register, HWRITE as strobe, 7-26 single write, 1 wait state, HCS as strobe, 7-27 single write, HCS as strobe, 7-25 write strobe, 2-14 host interrupt disabling, 6-6 enabling, 6-6 host-address bus, 2-14, 7-2, 7-7 host-byte selects, legal combinations for autoincrementing, 7-13 host-interface, bus fault indication, 7-9 host-present mode, 6-25 HPFW bit, 4-57, 4-59, 7-3 effects on autoincrementing, 7-13

effects on prefetching, 7-10

interaction with HINC, 4-59, 7-10 HRDY signal, 2-10, 2-14, 7-2 activating for. . . host reads, 7-16 host reads and writes after prefetches, 7-17 host writes, 7-16 HREAD signal, 2-10, 2-14, 7-2, 7-7 HRYI bit, **4-64**, 6-5, 7-5, 7-9 HSBLNK register, 4-66-4-67, 9-4 HSD bit, 4-36, 9-5 HSTADRH, HSTADRL, 4-56, 7-5 HSTCTLH register, 4-57—4-62 CF bit, 4-61, 5-8, 7-4 HACK bit, 4-57, 7-3 HBFI bit, 7-9 HBREN bit, 7-9 HINC bit, 4-60, 7-4, 7-10, 7-13 HLB bit, 7-3, 7-11 HLB bits, 4-57 HLT bit, 4-61, 5-8, 6-4, 6-22, 7-4, 7-32 HPFW bit, 4-59, 7-3, 7-10, 7-13 HRYI bit, 7-9 NMI bit, 4-58, 6-4, 6-16, 7-3 NMIM bit, 4-59, 6-4, 6-16, 7-3 RST bit, 4-58, 6-4, 7-3 HSTCTLL register, 4-62-4-65 EMG bit, 4-62, 4-63, 7-4, A-7 EMIEN bit, 4-62, 4-64, 7-4, A-7 EMR bit, 4-62, 4-63, 7-4, A-7 HBFI bit, 4-64, 6-5, 6-21, 7-5 HBREN bit, 4-62, 4-64, 6-5, 6-21, 7-5 HBYI bit, 4-62 HRYI bit, 4-62, 4-64, 6-5, 6-21, 7-5 INTIN bit, 4-62, 6-5, 6-16, 7-4 **INTOUT, 4-63** INTOUT bit, 4-62, 6-5, 6-21, 7-4 MSGIN, 4-62, 4-63 MSGIN bits, 4-62, 6-5, 6-16, 7-4 **MSGOUT**, 4-63 MSGOUT bits, 4-62, 6-5, 6-21, 7-4 HSTDATA register, 4-65, 7-5 HSYNC signal, 2-10, 2-15, 9-3 selecting as input or output, 4-36 HTOTAL register, 4-67-4-68, 9-4 HWRITE signal, 2-10, 2-14, 7-2, 7-7

#### -----

I/O registers, 4-9-4-13 **BSFLTD**, 4-15 BSFLTST, 4-17 CONFIG. 4-20 CONTROL, 4-24 CONVDP, 4-28 CONVMP, 4-28 CONVSP, 4-28 DINC, 4-32 DPYCTL, 4-36 **DPYINT**, 4-41 DPYMSK, 4-44 DPYNX, 4-42 DPYST, 4-46 **HCOUNT**, 4-52 HEBLNK, 4-53 HESERR, 4-54 HESYNC, 4-55 host accesses, 8-24 HSBLNK, 4-66 HSTCTLH, 4-57-4-62 HSTCTLL, 4-62 HTOTAL, 4-67 IHOST, 4-68 in the memory map, 3-2, 3-3 initial state following reset, 6-23 **INTENB**, 4-69 INTPEND, 4-70 memory map, 4-9 PMASK, 4-77 PMASK registers, 4-75 **REFADR**, 4-78 **SCOUNT, 4-80** SETHCNT, 4-81-4-82 SETVCNT, 4-82-4-83 summary, 4-10 **VCOUNT**, 4-84 **VEBLNK**, 4-86 VESYNC, 4-87 **VSBLNK**, 4-88 **VTOTAL**, 4-89 ID assignments, for coprocessors, 10-6 IDLE instruction, 13-133 IE (global interrupt enable) status bit, 4-2, 6-3, 6-6 IHOST registers, 4-68 illegal opcode interrupt, priority, 6-7 immediate values, 13-2

implicit addressing, 7-12-7-15 implied operands B-file registers, 4-7 summary, 4-8 COLOR0, 4-18, 4-74 COLOR1, 4-19, 4-74 **CONFIG**, 4-20 CONTROL, 4-24 CONVDP, 4-28 CONVMP, 4-28 CONVSP, 4-28 **DADDR**, 4-30 DPTCH, 4-34 DYDX, 4-50 MADDR, 4-71 MPTCH, 4-72 OFFSET, 4-73 PATTERN, 4-74 PMASK, 4-75 **PSIZE**, 4-77 SADDR, 4-79 SPTCH, 4-83 WEND, 4-90 WSTART, 4-91 in-circuit emulation, A-1 host communications, 4-64 INC instruction, 13-134 incrementing . . . automatically for host accesses, 4-60 display address, 4-32 **DPYNX**, 4-32 **HCOUNT**, 4-52 **SCOUNT, 4-80 VCOUNT**, 4-84 incrementing. . ., y-zoom value, 4-32, 4-33, 4-42 indirect operands, 13-5 in XY mode, 13-9 with an offset, 13-6 with postincrement, 13-7 with predecrement, 13-8 instruction cache, 5-1-5-12 accessible words, 5-3 architecture, 5-2 bypassing the cache, 5-8 cache hit, 5-5 cache miss, 5-5 segment miss, 5-6 subsegment miss, 5-5 CD (cache disable) bit, 5-8 CF (cache flush) bit, 5-8

control algorithm, 5-3 disabling the cache, 5-8 downloading new code from a host, 5-8 fetching data after a cache miss, 5-6 flushing the cache, 4-61, 5-8 initial state following reset, 6-23 internal parallelism, 5-10 least-recently-used algorithm, 5-4 operation, 5-5-5-8 organization, 5-2 P flags, 5-2, 5-4 performance when enabled vs. disabled, 5-9 reason it's provided, 5-1 replacement algorithm, 5-4 segments, 5-2 self-modifying code, 5-8 setting the CD bit, 5-8 setting the HLT bit, 5-8 size, 5-3 SSA registers, 5-2 subsegments, 5-2 instruction set. See TMS34020 instruction set instructions fetches, status code on local-memory cycle, 8-11 interrupting execution, 6-13 timings, 15-1-15-12 INTENB register, 4-69-4-70, 6-6 DIE bit, 4-69-4-70, 6-3 HIE bit, 4-69-4-70, 6-3 WVE bit, 4-69-4-70, 6-3 X1E bit, 4-69-4-70, 6-3 X2E bit, 4-69-4-70, 6-3 interlaced video, 9-21-9-28 composite sync equalization region, 9-17 serration region, 9-17 display example, 9-40-9-42 electron beam pattern, 9-22 even field, 9-21 external synchronization, 9-31 odd field, 9-21 programming vertical registers, 9-24 selecting, 4-40 signal combinations, 9-22 interlist utility, 1-11 internal interrupts, 6-16-6-18 display interrupt, 6-17 host interrupt. 6-16 illegal-opcode interrupt, 6-18

NMI, 6-16 single-step interrupt, 6-17 window-violation interrupt, 6-17 internal parallelism, 5-10 interrupt, saving information on the stack, 3-29 interrupts, 6-1-6-32 actions taken, 6-9, 6-10 bus-fault interrupt, 6-19-6-20 delays, 6-11 sources, 6-12 disabling, 6-6 display interrupt, 4-41, 4-69, 4-70, 6-17, 9-37 during instruction execution, 4-2, 6-3, 6-9 effects on PC, 6-9 ST. 6-9 effects on. . . PC. 4-4 SP. 4-5 emulation considerations, A-7 enabling, 4-2, 4-69, 6-6 external interrupts, 4-69, 4-70, 6-15 graphics instructions, 6-10, 6-13-6-14 host interrupt, 2-14, 4-64, 4-69, 4-70, 6-16, 6-21 host interrupts, 4-64 how many supported?, 6-1 illegal-opcode interrupt, 6-18 internal interrupts, 6-16 latency, 6-11 LINT1, LINT2 (local interrupts), 2-16 nonmaskable interrupt, 4-58, 6-16 pending interrupts, 4-70 priorities, 6-7 processing, 6-9 registers, 6-2 HSTCTLH, 4-57, 6-4 HSTCTLL, 4-62, 6-5 INTENB. 4-69-4-70. 6-3 INTPEND. 4-70-4-71. 6-4 ST. 4-2. 6-3 reset, 6-22-6-27 host-present mode, 6-25 self-bootstrap mode, 6-25 RESET (system reset), 2-16 service routines, 6-10 returning, 6-10 single-stepping through, 6-31 signals, 6-2 BUSFLT, 6-2 HINT, 6-2

LINT1, LINT2, 6-2 RESET, 6-2 single-step interrupt, 6-17, 6-28 interaction with other interrupts, 6-30 traps, 6-8, 6-21 numbers. 6-16 vector, fetches, status code on local-memory cycle, 8-11 vector addresses, 6-8, 6-16 window violation, 4-69, 4-70, 6-17 INTIN bit, 4-62, 6-5, 6-16, 7-4 INTOUT bit, 4-62, 4-63, 6-5, 7-4 INTPEND register, 4-70-4-71 DIP bit, 4-70-4-71, 6-4, 6-17 HIP bit, 4-70-4-71, 6-4, 6-17 WVP bit, 4-70-4-71, 6-4, 6-17 X1P bit, 4-70-4-71, 6-4, 6-15 X2P bit, **4-70-4-71**, 6-4, 6-15 INVD ('34082 pseudo-op), 14-41 INVF ('34082 pseudo-op), 14-42, 14-43 IX (interruptible instruction executing) status bit, 4-2, 6-3

# J

JAcc instruction, 13-135-13-136 JRcc (long) instruction, 13-139-13-140 JRcc (short) instruction, 13-137-13-138 JUMP instruction, 13-141 jump instructions, 13-25-13-31 condition codes, 13-26 DSJ, 13-103 DSJEQ, 13-104-13-105 DSJNE, 13-106-13-107 DSJS, 13-108 effects on PC, 4-4 JAcc, 13-135-13-136 JRcc (long), 13-139-13-140 JRcc (short), 13-137—13-138 JUMP, 13-141 JUMPC ('34082 pseudo-op), 14-44

# K

Kernighan and Ritchie, 1-11 key features of the TMS34020, 1-2



LAD0-LAD31 signals, 2-9, 2-11, 8-2, 10-2 connecting to VRAMs, 8-41, 8-43 4-bit VRAMs 4 bits per pixel, 8-41 8 bits per pixel, 8-43 connections to 16-bit host bus, 7-42 data remapping, 8-42, 8-43 LAD4 used as 16-bit word select, 8-25 latching data on the LAD bus, 8-8 saving data during a bus fault, 4-15 status code on LAD0-LAD3, 8-9, 8-10-8-11 values for nonmultiplexed addressing, 8-50 when data is valid, 8-13 which half used during 16-bit accesses, 8-26 latency halt latency, 7-39 host requests, 7-37-7-40 of screen refreshes, 9-50 recoanizing interrupts, 6-11 LCLK1, LCLK2 signals, 2-10, 2-16, 8-2, 8-18, 10-2 LCLK1 effect on external interrupts, 6-15 used in emulation, A-3, A-4, A-5, A-6 LCLK2, identifying valid data on LAD bus, 8-12, 8-18 least-recently-used (cache replacement) algorithm, 5-4 LINE instruction, 13-142-13-145 destination address, 4-30 implied operands COLOR0, 4-18

*COLOR0, 4-18 COLOR0, 4-18 COLOR1, 4-19 CONVDP, 4-29 DADDR, 4-30 DPTCH, 4-34 DYDX, 4-50 MPTCH, 4-72 OFFSET, 4-73 PATTERN, 4-74 PMASK, 4-76 PSIZE, 4-77 SADDR, 4-79 WEND, 4-90 WSTART, 4-91* source address, 4-79 linear addressing, 3-3-3-4, 3-15 advantages, 3-19 array addresses destination address (DADDR), 4-30 source address (SADDR), 4-79 pixels, 3-11 LINIT instruction, 13-146 linker, 1-12 LINT1, LINT2 signals, 2-10, 2-16, 6-2, 6-15, 8-2, 10-2 interrupt pending indication, 4-70 interrupt request, 4-69 little-endian addressing, 3-20-3-25 assembling code for, 3-24-3-25 default at reset. 3-20. 4-21 effect of BEN bit, 3-20 effect of the BEN bit, 4-21 host interface, 7-44 processors that use it, 3-20 selecting, 3-20, 4-21 LMO instruction, 13-147 load-write-mask cycles, 8-34 local-memory interface, 8-1-8-60 addressing mechanisms, 8-50-8-56 cycles address/status portion, 8-8-8-9 bus errors/bus faults. 8-14 completing a successful cycle, 8-13 data portion, 8-8---8-9 ending, 8-12-8-14 extending with wait states, 8-12 general form, 8-8-8-9 page mode, 8-15---8-17 read & write cycles, 8-18---8-24 retrying, 8-13 status codes. 8-10-8-11 with wait states. 8-46-8-48 display examples, 8-57-8-60 dynamic bus sizing, 8-25-8-29 host-default cycles, 7-15, 8-7, 8-49-8-50 LAD0—LAD31 (LAD bus), 2-11 multiplexed addressing, 8-51-8-53 nonmultiplexed addressing, 8-50 page mode, 8-15-8-17 registers, 8-4-8-5 CONFIG, 4-20, 8-4 DPYCTL, 4-36 PMASK. 4-75. 8-5 PSIZE, 4-77

REFADR, 4-78, 8-5 request priorities, 8-6-8-7 signals, 8-2-8-3 ALTCH, 2-11, 8-2, 10-2 BUSFLT, 2-11, 8-2, 8-12, 8-18, 10-2 CAMD, 8-18 CAS0---CAS3, 10-2 DDIN, 2-11, 8-2 DDOUT, 2-11, 8-2, 8-18 Ğl. 8-18 LAD0-LAD31, 2-11, 8-2, 10-2 LCLK1, LCLK2, 8-12, 8-18, 10-2 LINT1. LINT2. 10-2 LRDY, 2-11, 8-2, 8-12, 8-18, 10-2 <u>FGMD, 2-11, 8-3, 8-12, 8-18</u> R0, R1, 8-18 SF. 10-2 SIZE16, 2-11, 8-3, 8-12, 8-18 WE, 10-2 logical instructions, 13-24 AND, 13-40 ANDI, 13-41 ANDN, 13-42 ANDNI, 13-43 LMO, 13-147 NEG, 13-178 NEGB, 13-179 NOT, 13-181 OR, 13-182 ORI, 13-183 RMO, 13-224 XOR. 13-266 XORI, 13-267 loss of bus grant, 8-6 LRDY signal, 2-9, 2-11, 7-9, 8-2, 8-12, 8-18, 10-2 bus cycle completion codes, 2-12

# Μ

MADDR register, **4-71—4-72** SETCMP instruction, 4-71 major interfaces, 2-8 masks display mask (DPYMSK), 4-44—4-46 mask array *address (MADDR), 4-71—4-72 pitch (MPTCH), 4-72 XY-to-linear conversion factor (CONVMP), 4-28* 

pitch conversion factor, 4-28-4-30 CONVMP register, 4-28 MPTCH register, 4-72-4-73 plane mask (PMASK), 4-75 write-mask registers (for VRAMs), 4-20, 4-22 memory address space, 3-3 display memory, 8-56 coordinates, 3-13 dimensions, 3-12 general use, 3-3 I/O registers, 3-3 map of local memory, 3-2 organization, 3-1-3-32 addressing, 3-3---3-4 bank selection, 8-55-8-56 bytes, 3-1 fields, 3-1, 3-3, 3-5 memory map, 3-2 pixel arrays, 3-1, 3-18-3-19 pixels, 3-1, 3-10 stacks, 3-26 reserved, 3-3 system memory, 8-56 vectors, 3-3 memory-to-serial-registers cycles, 8-30 memory-to-split-serial-registers cycles, 8-31 micellaneous instructions, CVDXYL, 13-87-13-88 midline reload, 4-38, 8-58, 9-55-9-56 example display memory dimensions, 8-59 midlines reload. 9-43---9-46 miscellaneous instructions CLR, 13-57 CLRC, 13-58 CVMXYL, 13-89-13-90 CVSXYL, 13-91 REV, 13-221 SETCDP, 13-227 SETCMP, 13-228 SETCSP, 13-229 MMFM instruction, 3-27, 13-148----13-149 MMTM instruction, 3-27, 13-150-13-151 MODS instruction, 13-152 MODU instruction, 13-153-13-157 MOVB instructions, 13-154-13-157 MOVD pseudo-ops instructions, 14-45-14-57

MOVE instructions, 13-158-13-166 move instructions BLMOVE, 13-44-13-45 byte, 13-20 field, 13-20-13-31 MMFM, 13-148-13-149 MMTM, 13-150-13-151 MOVB instructions, 13-154-13-157 MOVE instructions, 13-158-13-166 MOVI (16 bits), 13-167 MOVI (32 bits), 13-168 MOVK, 13-169 MOVX. 13-170 MOVY, 13-171 multiple register, 13-20 register-to-register, 13-19 summary, 13-19-13-23 value-to-register, 13-19 XY, 13-19 MOVE pseudo-ops instructions, 14-58-14-67 MOVF pseudo-ops instruction, 14-68-14-110 MOVI instruction 16-bit (short) version, 13-167 32-bit (long) version, 13-168 MOVK instruction, 13-169 MOVX instruction, 13-170 MOVY instruction, 13-171 MPTCH register, 4-72-4-73 SETCMP instruction, 4-72 XY-to-linear conversion, 3-15 MPYD ('34082 pseudo-op), 14-78 MPYF ('34082 pseudo-op), 14-79, 14-80 MPYS ('34082 pseudo-op), 14-81, 14-82 MSGIN bits (MSGIN0----MSGIN2), 4-62, 4-62, 4-63, 6-5, 6-16, 7-4 MSGOUT bits (MSGOUT0-MSGOUT2), 4-62, 4-63, 6-5, 6-21, 7-4 multiple-TMS34020 system, 7-40-7-41 multiplexed addressing, 8-51-8-53 multiprocessor interface, 11-1-11-20 3-wire interface, 11-1 arbitration logic, 11-13-11-15 2 TMS34020s, 11-15—11-17 examples, 11-15-11-18 with a hold device, 11-17-11-20 bus request codes, 11-5 bus requests, 11-5 initializing multiple TMS34020s, 11-19

local-memory bus passing control, 11-6 releasing control, 11-5 requesting control, 11-5 overview, 11-2 protocols, 11-5 retries, 11-15 signals, 2-13, 11-2 GI, 2-13, 11-2 R0, R1, 2-13, 11-2 system configuration, 11-3-11-4 system with a host processor, 7-40-7-41 wait states, 11-15 with a host processor, 11-20 MWAIT instruction, 13-177 MYPS instruction, 13-172—13-174 MYPU instruction, 13-175-13-176

## Ν

N (negative) status bit, 4-3 NEG instruction, 13-178 NEG ('34082 pseudo-op), 14-83, 14-84 NEGB instruction, 13-179 NEGD ('34082 pseudo-op), 14-85 NEGF ('34082 pseudo-op), 14-86, 14-87 NIL bit, 4-40, 9-6, 9-18, 9-21 NMI bit, 4-57, 4-58, 6-4, 6-16, 7-3 NMIM bit, 4-57, 4-59, 6-4, 6-16, 7-3 noninterlaced video. 9-18-9-20 display example, 9-38-9-39 electron beam pattern, 9-18 programming vertical registers, 9-20 selecting, 4-40 signal combinations, 9-18 nonmaskable interrupt, 6-16 NMI bit, 4-58 NMIM bit, 4-59 priority, 6-7 saving the context, 4-59 nonmultiplexed addressing, 8-50 NOP instruction, 13-180 NOT instruction, 13-181 NOT ('34082 pseudo-op), 14-88, 14-89

## 0

object format, 1-11 object format converter, 1-12 odd field (interlaced video), 9-21 OFFSET register, 4-73-4-74 XY-to-linear conversion, 3-15, 3-16 on-chip registers PC, 4-4 register files, 4-6 status register (ST), 4-2 opcodes, illegal opcodes interrupt, 6-18 range, 6-18 operand formats, 13-2-13-9 optimization, 1-11 OR instruction, 13-182 ORI instruction, 13-183

# Ρ

P flags, 5-2, 5-4 initial state following reset, 6-23 page mode, 8-15-8-17, 8-18 dynamic bus sizing, 8-28 multiple local-memory cycles, 8-15 read cycle timing, 8-20 read/write cycle timing, 8-22 read-modify-write cycle timing, 8-22 selecting page mode, 8-15 signal, 2-11 write cycle timing, 8-20 panning the display, 9-57 parameter size, for coprocessor data, 10-6 PATTERN register, 4-74 PBH bit, 4-25 PBV bit, 4-26 PC. 4-4 and the stack, 3-29 effects of instruction execution, 4-4 effects of interrupts, 6-9 illustration. 4-4 initial state following reset, 6-23 pending... interrupts, 4-70, 6-4 local-memory requests, 8-7 refresh cycles, 4-23, 8-7

PFILL instruction, 13-184-13-189 implied operands COLOR0, 4-18 COLOR1, 4-19 DADDR, 4-30 DPTCH, 4-34 DYDX, 4-51 **OFFSET**, 4-73 PATTERN, 4-74 PGA package pinout, 2-2 PGMD signal, 2-9, 2-11, 8-3, 8-12, 8-18 pin descriptions, 2-1-2-16 by category, 2-9-2-16 DRAM/VRAM interface, 2-9, 2-12 emulation interface, 2-10 host interface, 2-9, 2-13 local-memory interface, 2-9, 2-11-2-16 major interfaces, 2-8 multiprocessor interface, 2-9, 2-13 GI. 2-13 power, 2-16 summarv. 2-9-2-16 system control, 2-10, 2-16 video interface, 2-15 pinouts, TMS34020, 2-2-2-7 PGA package, 2-2-2-7 QFP package, 2-5-2-7 pitches (for pixel arrays) destination array, 4-28-4-30, 4-34-4-35 legal pitch values, 4-29-4-30 mask array, 4-28-4-30, 4-72-4-73 source array, 4-28-4-30, 4-83-4-84 XY-to-linear conversion destination pitch, 4-34-4-35 factor CONVDP register, 4-28-4-30 CONVMP register, 4-28-4-30 CONVSP register, 4-28-4-30 mask pitch, 4-72-4-73 source pitch, 4-83-4-84 PIXBLT instructions, 13-190-13-205 alternate starting corners, 3-18 destination address, 4-30 display pitch, 3-13 horizontal direction, 4-25 PIXBLT B.L. implied operands COLOR0, 4-18 COLOR1. 4-19 CONTROL, 4-27

DADDR. 4-30 DPTCH, 4-34 DYDX, 4-50 PMASK, 4-76 PSIZE, 4-77 SADDR. 4-79 SPTCH. 4-83 PIXBLT B,XY, implied operands COLOR0, 4-18 COLOR1, 4-19 CONTROL, 4-27 CONVDP. 4-29 DADDR, 4-30 DPTCH, 4-34 DYDX, 4-50 OFFSET, 4-73 PMASK, 4-76 PSIZE, 4-77 SADDR, 4-79 SPTCH, 4-83 WEND. 4-90 WSTART, 4-91 PIXBLT L, L, implied operands CONTROL. 4-27 DADDR, 4-31 DPTCH, 4-34 DYDX, 4-50, 4-51 PMASK, 4-76 PSIZE, 4-77 SADDR, 4-79 SPTCH, 4-83 PIXBLT L,M,L, implied operands DADDR, 4-31 DPTCH, 4-34 MADDR, 4-71 MPTCH, 4-72 OFFSET, 4-73 PMASK, 4-76 PSIZE, 4-77 SADDR. 4-79 SPTCH, 4-83 PIXBLT L,XY, implied operands CONTROL. 4-27 CONVDP, 4-29 CONVSP, 4-29 DADDR. 4-31 DPTCH, 4-34 DYDX, 4-50, 4-51 OFFSET. 4-73 PMASK, 4-76

PSIZE, 4-77 SADDR, 4-79 SPTCH, 4-83 WEND, 4-90 WSTART, 4-91 PIXBLT XY,L, implied operands CONTROL, 4-27 CONVDP, 4-29 CONVSP, 4-29 DADDR, 4-31 DPTCH, 4-34 DYDX, 4-50 OFFSET, 4-73 PMASK. 4-76 PSIZE, 4-77 SADDR. 4-79 SPTCH. 4-83 PIXBLT XY,XY, implied operands CONTROL, 4-27 CONVDP. 4-29 CONVSP, 4-29 DADDR, 4-31 DPTCH. 4-34 DYDX, 4-51 OFFSET, 4-73 PMASK. 4-76 PSIZE, 4-77 SADDR, 4-79 WEND, 4-90 WSTART, 4-91 pixel arrays, 3-18 source address, 4-79 vertical direction, 4-26 pixel processing arithmetic options, 4-26 Boolean options, 4-26 selecting, 4-26-4-27 size, 4-77 pixel access, conversion to a VRAM serial-register transfer, 4-39 pixel arrays, 3-1, 3-18-3-19 addresses destination (DADDR), 4-30 source (SADDR), 4-79 binary arrays, 4-18, 4-19 dimensions, 4-50 height (DY), 3-18 illustration, 3-18

mask address, 4-71

operations, window checking, 4-90, 4-91 pitch, 3-18 destination pitch, 4-28-4-30, 4-34-4-35 legal values, 4-29 mask pitch, 4-28-4-30, 4-72-4-73 source pitch, 4-28-4-30, 4-83-4-84 size, 4-50 starting address, 3-18 width (DX), 3-18 window checking, 3-19, 4-50, 4-90, 4-91 XY origin, 3-18 pixel operations color information, 4-18, 4-19 pattern information, 4-74 status code on local-memory cycle, 8-11 pixels, 3-1, 3-10-3-13 DINC register, 3-11 display pitch, 3-13 extraction, 3-11 in memory, 3-10 insertion, 3-11 linear addressing, 3-11 on the screen, 3-11 configurable screen origin, 3-12 PSIZE register, 3-10 starting address, 3-10 storage in memory, 3-10 valid sizes, 3-10 within a general-purpose register, 3-10 XY addressing, 3-11 PIXT instructions, 13-206-13-213 implied operands CONTROL, 4-27 CONVDP, 4-29 CONVSP. 4-29 DPTCH. 4-34 OFFSET, 4-73 PMASK. 4-76 PSIZE. 4-77 SPTCH, 4-83 WEND, 4-90 WSTART, 4-91 plane masking, PMASK register, 4-75-4-77 PMASK registers, 4-75-4-77, 8-5 and VEN, 4-22 block-write cycle (with mask), 8-40 enabling load-write-mask cycles, 8-34 local-memory write cycle (with mask), 8-36 writing 1s complement of PMASK to VRAM writemask registers, 8-34

PMASKH, PMASKL. See PMASK registers POPST instruction, 13-214 power and ground, pins, 2-16 V<sub>CC</sub>, 2-16 V<sub>SS</sub>, 2-16 PPOP bits (PPOP0---PPOP4), 4-26---4-27 prefetching, 7-10-7-12 accessing the correct address, 7-12 after reads, 7-10-7-12 after writes, 7-10-7-12 enabling, 7-10-7-12 size of host data bus, 7-11-7-12 priorities of. . ., memory bus requests, 8-6 program counter. See PC program-control instructions, 13-25-13-27 DINT, 13-95 DSJ, 13-103 DSJEQ, 13-104-13-105 DSJNE, 13-106-13-107 DSJS, 13-108 EINT, 13-109 EMU, 13-110 EXGF, 13-111 EXGPC, 13-112 EXGPS, 13-113 GETPC, 13-130 GETPS, 13-131 GETST, 13-132 IDLE, 13-133 MWAIT, 13-177 NOP, 13-180 POPST, 13-214 PUSHST, 13-215 PUTST, 13-216 RETM, 13-219 SETC, 13-226 SETF, 13-230-13-231 SEXT, 13-232 ZEXT, 13-268 program-control instuctions, SWAPF, 13-247-13-248 PSIZE register, 3-10, 4-77-4-78 PUSHST instruction, 13-215 PUTST instruction, 13-216 single-step interrupt, 6-17

# Q

QFP package pinout, 2-5



R0, R1 signals, 2-9, 2-13, 8-18, 11-2 RAS signal, 2-9, 2-12, 8-3 RCA0---RCA12 signals, 2-9, 2-12, 8-3, 8-53 effect of RCM bits, 4-21 RCM bits (RCM0—RCM1), **4-21**, 4-78, 8-4 effect on local-memory cyles, 8-51, 8-52 write protecting the field, 4-22 read cycles adding wait states. 8-46 general timing, 8-19-8-24 initiated by the host, 8-24 local memory, 8-18 timing with page mode, 8-20 VRAM read transfer, 8-30 read/write cycles, timing with page mode, 8-22 read-modify-write cycles steps in operation, 8-22 timing with page mode, 8-22 with dynamic bus sizing, 8-26 REFADR register, 4-78-4-79, 8-5 address output to RCA and LAD buses, 8-44 refreshes See also screen refreshes address output, 4-42 automatic screen refreshes, 4-40 CAS-before-RAS. 8-44 DRAM refreshes, 4-78 selecting the refresh rate, 4-23 host-access delays, 7-38 pending counter, initial state following reset, 6-23 pseudo-address, 8-44 REFADR register, 4-78 refresh address, 4-78 VRAM screen refreshes, enabling for VRAMs with split serial registers, 4-38 register files, 4-6-4-8 file A, 4-6-4-8 file B, **4-6—4-8** illustration, 4-6 register used as auxiliary stack pointer, 3-29 SP, 4-5, 4-6 storing registers on the stack, 3-27 register-direct operands, 13-4

register-indirect operands, 13-5 in XY mode, 13-9 with an offset, 13-6 with postincrement, 13-7 with predecrement, 13-8 registers, 4-1-4-14, 4-62-4-92 cache registers data, 5-2, 5-3 segment start address, 5-2, 5-3 general-purpose registers, 4-6-4-8 I/O registers, 4-9-4-13 program counter (PC), 4-4 SP. 3-26 stack pointer (SP), 4-5 status register (ST), 4-2 STK, 3-29 reserved. . . bits in the status register, 4-3 memory, 3-3 reset, 6-22-6-27 activity following reset, 6-24 configuring the TMS34020 at reset selecting the endian addressing mode, 4-20, 4-21 selecting the row-/column-address mode, 4-20, 4-21 effects on the cache, 5-4 emulation considerations, A-7 host-present mode, 6-25 how to reset the TMS34020, 6-22-6-27 initial state following reset cache, 6-23 refresh-pending counter, 6-23 registers, 6-23 signals, 6-22 protecting the addressing-mode configuration, 4-22 **RESET** signal, 2-16 self-bootstrap mode, 6-25 software reset using NMI, 4-58 using RST, 4-58 value of ST, 4-2 RESET signal, 2-10, 6-2, 6-22-6-27 effect on HLT bit, 4-61 priority, 6-7 RETI instruction, 6-10, 13-217-13-218 how it differs from RETM, 6-32 single-step interrupt, 6-17

RETM instruction, 6-10, 13-219 how it differs from RETI, 6-32 single-step interrupt, 6-17 retries coprocessor cycles, 10-9 local-memory cycles, 8-13 on a host access, 4-64, 7-9 RETS instruction, 13-220 restrictions, 6-10 REV instruction, 13-221 RL instruction, 13-222, 13-223 RMO instruction, 13-224 rotate/shift instructions RL, 13-222, 13-223 SLA, 13-233, 13-234 SLL, 13-235, 13-236 SRA, 13-237, 13-238 SRL, 13-239, 13-240 row address bus, 2-12 configuration, 4-21 strobe, 2-12 time, 4-21, 8-8 **RPIX** instruction, 13-225 RR bits (RR0-RR2), 4-23, 4-78, 8-4 effect on local-memory cyles, 8-44 RST bit, 4-57, 4-58, 6-4, 7-3

# S

S (select) bit, 8-25 SADDR register, 4-79 scan line duration, 4-67 SCLK signal, 2-10, 2-15, 9-3 SCOUNT register, 4-80-4-81 screen origin alternate, 3-12 default, 3-12 screen refreshes address output during, 4-42 addressing sequence interlaced video, 9-53 noninterlaced video, 9-53 automatic refreshes, 4-40 CAS-before-RAS, 8-44 disabling, 9-49 during horizontal blanking, 9-42 effect of the display mask, 4-44

generating addresses, 9-51 horizontal blanking address generation, 9-52 minimum duration, 9-51 interlaced video, addressing sequence, 9-53 latency, 9-50 memory-to-register cycles, 4-39 midline reload, 9-43, 9-55 noninterlaced video, addressing sequence, 9-53 **REFADR register, 4-78** refresh address, 4-78 register-to-memory cycles, 4-39 registers DINC, 4-32 DPYMSK, 4-44 DPYNX. 4-42 DPYST, 4-46 scheduling, 9-50-9-51 split-serial-register midline reload, 4-38 screen sizes, 9-36 screens, configurable origin, 3-12 SDB, 1-12 segment miss, 5-6 self-bootstrap mode, 6-25 self-modifying code, effects on instruction cache, 5-8 serial registers, 4-38 converting pixel access to serial-register transfer accesses, 4-39 register-to-memory cycles, 8-32, 8-33 split serial registers, 4-38 transfers, 8-6 status-code on local-memory cycle, 8-11 serration, ending (HESERR), 4-54 serration pulses, 9-15-9-16 on CSYNC, 9-16 SETC instruction, 13-226 SETCDP instruction, 13-227 implied operands CONVDP. 4-29 DPTCH, 4-34 SETCMP instruction, 13-228 CONVMP register, 4-28 implied operands CONVMP, 4-29 MADDR, 4-71 MPTCH, 4-72

SETCSP instruction, 13-229 implied operands CONVSP. 4-29 SPTCH, 4-83 SETF instruction, 13-230-13-231 SETHCNT register, 4-81-4-82, 9-4 SETVCNT register, 4-82-4-83, 9-5 SEXT instruction, 13-232 SF signal, 2-9, 2-12, 8-3, 10-2 shift instructions, 13-28 shift/rotate instructions RL, 13-222, 13-223 SLA, 13-233, 13-234 SLL, 13-235, 13-236 SRA, 13-237, 13-238 SRL, 13-239, 13-240 sign-extending field 0. 4-2 field 1, 4-2 signal buffering, for emulator connections, A-4 signal descriptions, 2-1-2-16 by category, 2-9-2-16 DRAM/VRAM interface, 2-12 host interface, 2-13 local-memory interface, 2-11-2-16 major interfaces, 2-8 multiprocessor interface, 2-13 pinouts, 2-2-2-7 PGA package, 2-2-2-7 QFP package, 2-5-2-7 power, 2-16 system control, 2-16 video interface, 2-15 single-step interrupt, 6-17, 6-28 disabling, 6-6 enabling, 6-6 interaction with other interrupts, 6-30 priority, 6-7 SIZE16 signal, 2-9, 2-11, 8-3, 8-12, 8-18 dynamic bus sizing, 8-25 SLA instruction, 13-233, 13-234 SLL instruction, 13-235, 13-236 software development board, 1-12 software libraries 8514 adaptor emulation, 1-12 CCITT data compression, 1-12 font, 1-12 math/graphics, 1-12

software reset, 7-32 using NMI, 4-58 using RST, 4-58 source pitch conversion factor, 4-28-4-30 CONVSP register, 4-28 SPTCH register, 4-83-4-84 SP, 3-26, 4-5 effects of interrupts, 6-9 illustration, 4-5 position in the register files, 4-6 special-function pin, 2-12 SPTCH register, 4-83-4-84 XY-to-linear conversion, 3-15 SQR ('34082 pseudo-op), 14-90, 14-91 SQRD ('34082 pseudo-op), 14-92 SQRF ('34082 pseudo-op), 14-93, 14-94 SQRT ('34082 pseudo-op), 14-95, 14-96 SQRTD ('34082 pseudo-op), 14-97 SQRTF ('34082 pseudo-op), 14-98, 14-99 SRA instruction, 13-237, 13-238 SRE bit, 4-40, 9-6 effect on local-memory cyles, 8-31, 8-32 SRINC bits, 9-7, 9-52, 9-53, 9-54, 9-55 SRINC value, 4-32, 4-33 SRL instruction, 13-239, 13-240 SRNX bits, 9-7, 9-52, 9-53, 9-54, 9-55 SRNX value, 4-42, 4-43 SRST bits, 9-52, 9-53, 9-54, 9-55 SRST value, 4-46 SS (single-step) status bit, 4-2, 6-3, 6-6, 6-17 clearing, 6-29 setting, 6-28 SSA registers, 5-2 illustration, 5-3 initial state following reset, 6-23 SSV bit, 4-38, 9-6 effect on local-memory cyles, 8-31, 8-58 ST. 4-2-4-3 and the stack, 3-29 BF bit, 6-3 definitions of status bits, 4-2-4-4 IE bit, 6-3, 6-6 illustration showing bit positions, 3-5, 4-2 initial state following reset, 6-23 instructions that change it, 6-29 IX bit, 6-3

SS bit, 6-3, 6-6, 6-17 value at reset, 4-2 stack pointer. See SP stacks, 3-26-3-32 auxiliary stacks, 3-29-3-32 growing toward higher addresses, 3-31 growing toward lower addresses, 3-30 system stack, 3-26-3-29 instructions that pop values, 3-27 instructions that push values, 3-27 saving information during a subroutine call, 3-29 saving information during an interrupt, 3-29 saving register values, 3-27-3-29 standards, video NSTC, 9-27 PAL, 9-27 RS-170, 9-27 SECAM, 9-27 starting corner, selecting, 4-30, 4-79 status bits. 4-2 status codes bus cycle completion, 2-12 local-memory cycles block write, 8-11 bus-locked operation, 8-11 cache fill, 8-11 color-latch register load, 8-11 coprocessor cycle, 8-10 data access, 8-11 DRAM refresh, 8-10 emulator operation, 8-10 host cycle, 8-10 instruction fetch, 8-11 interrupt-vector fetch, 8-11 pixel operation, 8-11 serial-register transfer, 8-11 write-mask load, 8-11 status register. See ST strobes byte-select strobes, 7-7 chip-select, 7-7 read strobe, 7-7 write strobe, 7-7 SUB instruction, 13-241 SUB ('34082 pseudo-op), 14-100-14-103 SUBB instruction, 13-242 SUBD ('34082 pseudo-op), 14-104-14-105 SUBF ('34082 pseudo-op), 14-106-14-109

SUBI instruction, 13-243, 13-244 SUBK instruction, 13-245 subroutines effects on PC, 4-4 effects on SP. 4-5 saving information on the stack, 3-29 subsegment miss, 5-5 SUBXY instruction, 13-246 SWAPF instruction, 13-247-13-248 symbolic debugger, A-1 sync signals composite sync, CSYNC, 2-15, 4-37 ending horizontal sync, 4-52 vertical sync, 4-87 horizontal sync ending (HESYNC), 4-55 HSYNC, 2-15, 4-36 vertical sync, VSYNC, 2-16, 4-37 system configuration with a coprocessor, 10-18 with multiple processors, 11-3 configuration (CONFIG register), 4-20-4-24 considerations, bus faults, 6-20 control, signals, 2-16 CLKIN, 2-16, 8-2 LCLK1, LCLK2, 2-16, 8-2 LINT1, LINT2, 2-16, 8-2 **RESET**, 2-16 control (CONTROL register), 4-24-4-28 desian connecting an emulator to a target system, A-2. A-3 emulation considerations, A-1-A-10 multiple processors, 11-1 with multiple TMS34020s, 7-40-7-41 system memory, 8-56

T bit, **4-25** tap point, 4-44 target cable, mechanical dimensions, A-9 target system, setup with XDS emulator, A-2 test and emulation, A-1—A-10 TFILL instruction, 13-249-13-252 implied operands COLOR1. 4-19 CONTROL, 4-27 CONVDP, 4-29 DADDR, 4-31 DPTCH. 4-34 OFFSET, 4-73 PMASK, 4-76 PSIZE, 4-77 SADDR. 4-79 WEND. 4-90 WSTART, 4-91 TIGA. 1-13 TM bits (TM0-TM2), 4-24 TMS34010, registers not used by TMS34020 DPYADR, 4-35 DPYSTRT, 4-48 DPYTAP, 4-49 HSTADRH, 4-56 HSTDATA, 4-65 TMS34020 applications, 1-3 architecture, 1-4-1-9 block diagram, 1-5 compatibility with the TMS34010, 1-16-1-18 development tools, 1-10 in a graphics system, 1-14 instruction set, 13-1-13-31 addressing modes, 13-2-13-9 arithmetic instructions, 13-24 compare instructions, 13-24 condition codes for jumps, 13-26 context-switching instructions, 13-25-13-27 jump instructions, 13-25-13-31 logical instructions, 13-24 move instructions, 13-19-13-23 operand formats, 13-2-13-9 program-control instructions, 13-25-13-27 shift instructions, 13-28 summay, 13-9-13-18 XY instructions. 13-29 internal functions, 1-5 key features, 1-2-1-3 major interfaces, 1-8 overview, 1-1-1-18 TMS34020 Emulator, A-1 TMS34082.14-1-14-7 key features, 14-2-14-7 overview, 14-2-14-7

pseudo-ops, 10-3 See also Chapter 14 format, 14-3-14-5 register operands, 14-6-14-7 sample graphics system, 1-14 TMS44C251 (1M VRAM), 1-15 TR/QE signal, 2-9, 2-12, 8-3 transceivers, used in host interface, 7-6 transparency enabling, 4-25 modes on destination=COLOR0, 4-24 on result=0. 4-24 on source=COLOR0, 4-24 selecting a mode, setting the TM bits, 4-24 T bit, 4-25 TRAP L instruction, 13-256—13-258 TRAP N instruction, 13-253-13-255 traps, 6-21 how many supported?, 6-1 1 vector locations, 6-8

# V

V (overflow) status bit, 4-3 VBLT instruction, 13-259-13-261 enabling the VRAM block-write feature, 4-22 implied operands DADDR, 4-31 DPTCH, 4-34 DYDX, 4-51 PMASK, 4-76 PSIZE, 4-77 SADDR, 4-79 use of VRAM block-write feature, 8-39, 8-40, 8-41 Vcc, 2-10 VCE bit, 4-39, 9-6 effect on local-memory cyles, 8-30, 8-31, 8-32 VCLK signal, 2-10, 2-15, 9-3 VCOUNT register, **4-84—4-86**, 6-17, 9-5 external synchronization, 9-29 loading with the SETVCNT value, 4-82 VEBLNK register, 4-86-4-87, 9-5 interlaced video, 9-26 vector addresses, 6-8 VEN bit, 4-22, 8-4 effect on local-memory cyles, 8-34, 8-36

vertical back porch, 9-10 front porch, 9-10 video timing (internal), 9-13-9-14 vertical blanking, 9-9 interlaced video, 9-23 NTSC and PAL standards, 9-27 vertical sync, 9-9 direction, 4-37 vertical timina SETVCNT register, 4-82 **VCOUNT**, 4-84 VESYNC register, 4-87 VOTAL register, 4-89 VSBLNK register, 4-88 VESYNC register, 4-87-4-88, 9-5 interlaced video, 9-25 VFILL instruction, 13-262-13-263 enabling the VRAM block-write feature, 4-22 implied operands DADDR, 4-31 DPTCH, 4-34 DYDX, 4-51 PMASK, 4-76 PSIZE, 4-77 use of VRAM block-write feature, 8-39, 8-40, 8-41 video American vs. European, 9-21, 9-27 capture enable, 4-39 capture feature, 9-48 composite video, 9-15-9-17 enabling/disabling, 4-38 control CSYNC direction (CSD bit), 4-37 HSYNC direction (HSD bit), 4-36 selecting CBLNK or VBLNK, 4-38 selecting CSYNC or HBLNK, 4-38 VSYNC direction (VSD bit), 4-37 display interrupt, 9-37 display mask, 4-44-4-46 display next address, 4-42 enabling the display, 4-40 equalization region, 9-15-9-17 external synchronization, 9-29-9-35 horizontal timing, 9-11-9-12 interlaced video, 9-21-9-28 interlaced video (selecting), 4-40 midline reload, 9-55 midlines reload, 9-43-9-46

noninterlaced video, 9-18-9-20 noninterlaced video (selecting), 4-40 registers, 9-4-9-8 DINC, 9-7 DPYCTL, 9-5 DPYMSK, 9-8 DPYNX, 9-7 DPYST, 9-7 HCOUNT, 9-4 HEBLNK, 9-4 HESERR. 9-4 HESYNC. 9-4 HSBLNK: 9-4 HTOTAL, 9-4 SETHCNT, 9-4 SETVCNT, 9-5 VCOUNT. 9-5 VEBLNK. 9-5 VESYNC, 9-5 VSBLNK. 9-5 VTOTAL, 9-5 screen refreshes. 9-55 disabling, 9-49 generating addresses, 9-51 scheduling, 9-50-9-51 serration region, 9-15-9-17 signals, 2-15, 9-2 CBLNK/VBLNK, 2-15, 9-2 CSYNC/HBLNK, 9-2 HSYNC, 2-15, 9-3 SCLK. 2-15. 9-3 VCLK, 2-15, 9-3 VSYNC, 2-16, 9-3 standards NTSC, 9-27 PAL. 9-27 RS-170. 9-27 SECAM, 9-27 start address for display, 4-46-4-48 timing examples, 9-38-9-42 timing registers DPYCTL, 4-36 HCOUNT, 4-52 HEBLNK. 4-53 HESERR, 4-54 HESYNC. 4-55 HSBLNK. 4-66 HTOTAL, 4-67 SCOUNT, 4-80 SETHCNT, 4-81-4-82

SETVCNT. 4-82---4-83 VCOUNT, 4-84 **VEBLNK**, 4-86 VESYNC, 4-87 VSBLNK, 4-88 VTOTAL, 4-89 vertical timing, 9-13-9-14 VRAM control, 9-42 video control logic horizontal blanking HEBLNK, 4-53 HSBLNK. 4-66 horizontal timing, HCOUNT, 4-52 scan line duration, 4-67 serration, HESERR, 4-54 sync signals HESYNC. 4-55 VESYNC, 4-87 vertical blanking **VEBLNK**, 4-86 VSBLNK. 4-88 video timing, 9-1-9-58 VLCOL instruction, 13-264—13-265 implied operands, COLOR1, 4-19 VRAMs. 9-42 1M VRAMs, 1-15, 2-12, 9-55 alternate write transfers. 8-33 automatic screen refreshes, 4-40 big-endian addressing, 3-25 block-write cycles, 8-37 data mapping, 8-41 with mask, 8-37, 8-40 without mask. 8-37. 8-39 block-write modes, 4-22 bulk initialization, 9-47 control, 9-42 data expansion, 8-37 display mask, 4-45 fast fills. 8-37 load-color-register cycles, 8-37, 8-38 load-write-mask cycles, 8-34 memory-to-register cycles, 4-40 memory-to-serial register cycle, 8-30 memory-to-split-serial register cycle, 8-31 midline reload, 4-38 pseudo-write transfer, 8-32 screen refreshes, 4-38, 4-39, 4-40 during horizontal blanking, 9-42 serial registers, 4-38

serial-register transfer cycles, 4-39 *adding wait states, 8-48* telling the TMS34020 that the graphics system contains special-function VRAMS, 4-22 TMS44C251, 1-15 write cycles, with mask, 8-36 write transfers, 8-32 write-mask cycles, 8-34 VSBLNK register, **4-88—4-89**, 9-5 interlaced video, 9-26 VSD bit, **4-37**, 9-6 V<sub>SS</sub>, 2-10

VSYNC signal, 2-16, 9-3 pin number, 2-10 selecting as input or output, 4-37 VTOTAL register, **4-89—4-90**, 9-5

interlaced video, 9-26

# W

W bits (W0-W1), 4-25, 6-17 wait states. 8-46---8-48 coprocessor cycles, 10-9 extending a local-memory cycle, 8-12 WE signal, 2-9, 2-12, 8-3, 10-2 WEND register, 4-90-4-91 window checking defining a window end address (WEND), 4-90 start address (WSTART), 4-91 effect of DYDX, 4-50 modes, selecting, 4-25 pixel arrays, 3-19 W bits (W0-W1), 4-25 window-violation interrupt, 6-17 window-violation interrupt, 4-25, 6-17 disabling, 6-6 enabling, 4-69, 6-6 pending indication, 4-70 priority, 6-7 worst-case, delays to host accesses, 7-37 bus-master arbitration, 7-38 CPU cycles, 7-39 DRAM-refresh cycles, 7-38 host request synchronization, 7-38 previous host cycle, 7-38 screen-refresh cycles, 7-38

write cycles adding wait states, 8-46 block-write cycles data mappina. 8-41 with mask, 8-40 without mask, 8-39 general timing, 8-19-8-24 initiated by the host, 8-24 local memory, 8-18 serial-register-to-memory cycles, 8-32, 8-33 timing with page mode, 8-20 VRAM block-write cycles, 8-37 with mask, 8-36 write masks loads, status code on local-memory cycle, 8-11 local-memory cycles, 8-34 write-enable signal, 2-12 write-per-bit (block write) operation, 4-22 write-with-mask, 4-22 WSTART register, 4-91-4-92 WVE bit, 4-69, 6-3 WVP bit, 4-70, 6-4, 6-17



X1E bit, 4-69, 6-3 X1P bit, **4-70**, 6-4, 6-15 X2E bit, 4-69, 6-3 X2P bit, 4-70, 6-4, 6-15 XDS emulator, 1-12, 4-63, A-1 XOR instruction, 13-266 XORI instruction, 13-267 XY addressing, 3-14 array addresses destination address (DADDR), 4-30 source address (SADDR), 4-79 benefits, 3-14 configurable screen origin, 3-11 coordinate range, 3-14 format illustration, 3-14 general-purpose registers, 3-14 instructions that use it, 3-15 limits, 4-50 mapping to on-screen memory, 3-12 **OFFSET** register, 4-73 origin, 3-14 pixels, 3-11, 3-14 window checking, 3-14

XY instructions, 13-29 ADDXY, 13-38 ADDXYI, 13-39 CMPXY, 13-84 CVXYL, 13-92—13-93 FILL XY, 13-117—13-120 XY-to-linear conversion, **3-15—3-17** automatic conversion, 3-15

calculating the Y component, 3-15 CVXYL instruction, 3-16 formula, 3-15 pitch *actual pitch, 3-15 conversion factors, 3-15, 3-16 destination pitch, 4-28, 4-34 mask pitch, 4-28, 4-72 source pitch, 4-28, 4-83*  process, 3-17



Y-zoom, 4-32 increment value, 4-33, 4-42 Y-zoom feature, 9-56—9-57 YZCNT bits (YZCNT0—YZCNT4), 4-42, 9-7, 9-56 YZINC bits (YZINC0—YZINC4), 4-32, 4-33, 9-7



Z (zero) status bit, **4-3** zero-extending field 0, 4-2 field 1, 4-2 ZEXT instruction, 13-268

